

Draft

Rate of return guidelines

Explanatory Statement

July 2018



and an advertised of

© Commonwealth of Australia 2018

This work is copyright. In addition to any use permitted under the Copyright Act 1968, all material contained within this work is provided under a Creative Commons Attributions 3.0 Australia licence, with the exception of:

- the Commonwealth Coat of Arms
- the ACCC and AER logos
- any illustration, diagram, photograph or graphic over which the Australian Competition and Consumer Commission does not hold copyright, but which may be part of or contained within this publication. The details of the relevant licence conditions are available on the Creative Commons website, as is the full legal code for the CC BY 3.0 AU licence.

Requests and inquiries concerning reproduction and rights should be addressed to the:

Director, Corporate Communications Australian Competition and Consumer Commission GPO Box 4141, Canberra ACT 2601

or publishing.unit@accc.gov.au.

Inquiries about this publication should be addressed to:

Australian Energy Regulator GPO Box 520 Melbourne Vic 3001

Tel: (03) 9290 1444 Fax: (03) 9290 1457

Email: <u>AERInquiry@aer.gov.au</u>

Note

This explanatory statement forms part of the AER's draft decision on the rate of return guidelines. It should be read in conjunction with our draft rate of return guidelines.

Request for submissions

The Australian Energy Regulator (AER) invites interested parties to make submissions on this draft decision by **5pm AEST Friday 14 September 2018**.

We prefer that all submissions are in Microsoft Word or another text readable document format. Submissions on our issues paper should be sent to: <u>rateofreturn@aer.gov.au</u>.

Alternatively, submissions can be sent to:

Mr Warwick Anderson General Manager Australian Energy Regulator GPO Box 520 Melbourne Vic 3001

We prefer that all submissions be publicly available to facilitate an informed and transparent consultative process. Submissions will be treated as public documents unless otherwise requested. Parties wishing to submit confidential information should:

- clearly identify the information that is the subject of the confidentiality claim
- provide a non-confidential version of the submission in a form suitable for publication.

We will place all non-confidential submissions on our website. For further information regarding our use and disclosure of information provided to us, see the ACCC/AER Information Policy (October 2008), which is available on our website.

Please direct enquires about this paper, or about lodging submissions to <u>rateofreturn@aer.gov.au</u>.

Short contents

Note					
Request for submissions4					
Sh	Short contents5				
Lo	Long contents6				
Ho	How this explanatory statement is organised11				
1	Overview15				
2	How the guidelines contribute to the legislative objectives69				
3	Form and structure of the rate of return guidelines123				
4	Benchmark gearing ratio161				
5	Overall approach to return on equity174				
6	Risk free rate191				
7	Market risk premium199				
8	Equity beta242				
9	Return on debt approach				
10	Implementation of our return on debt approach				
11	Imputation credits				
Α	Value of imputation credits: detailed analysis401				
В	CEG return on debt memorandum451				
С	Illustration of broad credit band weighting460				

Long contents

Not	Note3				
Red	Request for submissions4				
Sho	ort c	ontents		.5	
Lor	ng c	ontents		.6	
Sho	orte	ned forn	ns	1	
Ηο	w th	is expla	natory statement is organised	13	
1	Ove	erview		15	
	1.1	Introdu	ction	15	
	1.2	The Dra	aft Decision summarised	16	
	1.3	The ob	jectives and principles guiding our decision	8	
	1.4	The co	nsultation process and key stakeholder comments	23	
	1.5	Summa	ary of our reasons in support of our Draft Decision	34	
2	Нο	w the gu	idelines contribute to the legislative objectives	39	
	2.1	Key co	ncepts in the legislative objectives	70	
		2.1.1	National gas and electricity objectives	70	
		2.1.2	Revenue and pricing principles	71	
		2.1.3	Allowed rate of return objective	73	
	2.2	Benchr	nark efficiency	73	
		2.2.1	Benchmarking and incentive regulation	74	
		2.2.2	The NPV=0 condition	76	
		2.2.3	Market cost of capital	77	
		2.2.4	Commensurate with risk	79	
	2.3	Measur	ing success	30	
	2.4	Risk an	nd return	35	
		2.4.1	Compensation for risk	35	

	2.4.2	Gas and electricity	100
	2.4.3	Impact of regulation on risk	104
	2.4.4	Interrelationships	112
	2.5 Summ	nary of submissions	114
3	Form and	structure of the rate of return guidelines	123
	3.1 A nom	ninal, vanilla, weighted average cost of capital	123
	3.2 Auton	natic application	125
	3.2.1	Gearing	127
	3.2.2	Return on equity	128
	3.2.3	Return on debt	130
	3.2.4	Value of imputation credits	131
	3.3 Role o	of financial performance measures	131
	3.3.1	RAB multiples	132
	3.3.2	Historical profitability	143
	3.3.3	Financeability assessments	149
	3.4 Summ	nary of submissions	154
4	Benchma	rk gearing ratio	161
	4.1 Updat	ed empirical estimates	162
	4.2 Estima	ation approach and response to submissions	166
	4.3 Summ	nary of submissions	170
5	Overall ap	proach to return on equity	174
	5.1 Found	lation model approach	174
	5.2 Identi	fy relevant material and determine role (Steps 1 a	and 2)178
	5.3 Implei	ment the foundation model (step 3)	179
	5.4 Step 4	- 6 of the foundation model approach	179
	5.4.1	Step 4 – other information.	181
	5.4.2	Step 5	182

	5.4.	.3	Step 6 – select point estimate	189
6	Risk fr	ee r	ate	191
	6.1 Dra	aft D	Decision	191
	6.1.	.1	Approach to date	191
	6.2 Inp	out c	luring the Guideline review process	192
	6.3 Iss	ues	and the AER's considerations	192
	6.3.	.1	Averaging period length	192
	6.3.	.2	Automatic application of the risk free rate	194
	6.3.	.3	Term of CGS yields	195
	6.4 Su	mm	ary of submissions	197
7	Market	risł	c premium	199
	7.1 Dra	aft d	lecision	199
	7.2 Sta	akeh	older consultation and submission	201
	7.2.	.1	Discussion paper and concurrent expert evidence sessions	202
	7.3 Iss	sues	and the AER's considerations	203
	7.3.	.1	Overall approach to estimating the MRP	203
	7.3.	.2	Historical Excess Returns	209
	7.3.	.3	Dividend Growth Models	215
	7.3.	.4	Survey Evidence	223
	7.3.	.5	Conditioning Variables	227
	7.3.	.6	Other Regulators	231
	7.3.	.7	Wright Approach	234
	7.4 Su	mm	ary of submissions	235
8	Equity	beta	a	242
	8.1 Dra	aft d	lecision	242
	8.2 Sta	akeh	older consultation and submissions	244
	8.2.	.1	Discussion paper and concurrent expert evidence sessions	244

	8.3 Issues and the AER's considerations			247
	8.3	3.1	Conceptual analysis	. 247
	8.3	3.2	Empirical studies	. 249
	8.3	3.3	Comparator firms	. 262
	8.3	3.4	International comparators	. 272
	8.3	3.5	Black CAPM and low beta bias	275
	8.3	3.6	Empirical methodology	. 287
	8.3	3.7	Industry analysis	295
	8.3	8.8	Other information informing range and point estimate	296
	8.3	3.9	Range and point estimate	. 297
	8.4 Su	umma	ary of submissions and responses	.308
9	Return	n on e	debt approach	.329
	9.1 Submissions on transition to the trailing average			
	9.2 N	Γ Pov	ver & Water's proposal for immediate transition	334
	9.3 Ta	sNet	works' proposal to alter its transition path	.335
	9.4 NS	SW a	nd ACT determinations to be re-made	.337
	9.5 Re	eturn	on debt incentive scheme	.337
	9.6 Su	umma	ary of submissions	.338
10	Impler	nenta	ation of our return on debt approach	.340
	10.1	Ber	nchmark credit rating	.341
	10.2	The	e role of the Chairmont report	.343
	10.3	Ber	nchmark term	.345
	10.4	Cho	pice of third party data provider	.352
	10.5	Imp	elementation of the benchmark credit rating	.359
	10.6	Adj	ustments to published data	.367
	10.7	Ave	eraging periods	.369
	10.8	Cor	ntingencies	.371

	10.9	Sur	nmary of submissions376
11	11 Imputation credits		credits
	11.1 Int		oduction
	11.	1.1	Approach to date
	11.2	AEI	R's assessment approach392
	11.	2.1	Regulatory requirements 392
	11.	2.2	Legal processes
	11.	2.3	Definition of a benchmark efficient entity
	11.	2.4	Interrelationships
	11.	2.5	Issues from the concurrent evidence session and submissions. 395
	11.3	Oui	r draft approach396
	11.	3.1	Payout ratio
	11.	3.2	Utilisation rate
Α	Value	of im	putation credits: detailed analysis401
	A.1 Rule requirements401		
	A.2 Recent litigation on the value of imputation credits404		
	A.3 Issues from the concurrent evidence session and submissions40		
A.4 Reasons for approach			ns for approach416
	A.4	l.1	The building block framework 416
	A.4	.2	The conceptual framework for the value of imputation credits 419
	A.4	.3	Evidence underlying our estimate 424
В	CEG r	eturr	on debt memorandum451
	B.1 Ov	veral	I role of the EICSI452
	B.2 Ac	ljust	ment for outturn inflation452
	B.3 CE	G's	other recommended adjustments453
С	Illustra	ation	of broad credit band weighting460

Shortened forms

Shortened form	Extended form
2013 Guidelines	refers to AER, <i>Rate of Return Guidelines</i> , December 2013; AER, <i>Rate of Return Guidelines - Explanatory Statement</i> , December 2013; and/or AER, <i>Rate of Return Guidelines - Explanatory</i> <i>Statement - Appendices</i> , December 2013
AER	Australian Energy Regulator
ARORO	allowed rate of return objective
ASX	Australian Securities Exchange
ΑΤΟ	Australian Taxation Office
САРМ	capital asset pricing model
ССР	Consumer Challenge Panel
	Sub-panel 16 of the CCP.
CCP16	This sub-panel was established to provide advice on our review of the rate of return Guidelines.
CGS	Commonwealth Government securities
CoAG	council of Australian governments
CRG	consumer reference group
DGM	dividend growth model
DRP	debt risk premium
	energy infrastructure credit spreads index
EICSI	This is the index developed in: Chairmont, <i>Aggregation of return on debt data report</i> , 28 April 2018.
ERP	equity risk premium
FAB	ATO franking account balance
GDP	gross domestic product
HER	historical excess returns
IRG	investor reference group
LAD	least absolute deviation
legislative objectives	collectively the NEO, NGO, ARORO, and RPPs
MRP	market risk premium
MSE	mean squared error
NEL	national electricity law
NEO	national electricity objective

Shortened form	Extended form
NER	national electricity rules
NGL	national gas law
NGO	national gas objective
NGR	national gas rules
OLS	ordinary least squares
RAB	regulatory asset base
Regulated services	refers to an electricity prescribed transmission service, an electricity distribution direct control service, and/or a gas reference service
Regulatory period	refers to a regulatory control period and/or an access arrangement period
Regulatory year	refers to a year within a regulatory period
RPPs	revenue and pricing principles
RRG	retailer reference group
Service provider	refers to an electricity transmission network service provider, an electricity distribution network service provider, and/or a gas service provider
SLCAPM	Sharpe-Lintner CAPM
WACC	weighted average cost of capital

How this explanatory statement is organised

This explanatory statement is composed of:

- an Overview, and
- a series of subject matter chapters.

The Overview provides a high level perspective on our draft decision concerning the Guideline review. It contains 5 sections:

- Introduction
- The draft decision summarised
- Objectives and principles guiding our decision.
- A summary of the consultation process and key comments of stakeholders
- Our key reasons in support of our draft decision.

We have set out our full analysis of all the relevant issues in much greater detail in separate subject matter chapters.

Each of the chapters contain a detailed explanation of our analysis and reasons for our draft decision. They also outline the submissions we have received and our considerations of them. The chapters set out the estimation methods, financial models, market data and other relevant evidence that we have taken into account in reaching our draft decision. The chapters also discuss any interrelationships between the financial parameters that are used (or to be used) in the making and implementation of our decisions. Where there are differences between this draft decision and the 2013 Guidelines we explain the reasons for those differences.

The subject matter of each of the chapters is described below:

- Chapter 1 contains the Overview of our decision.
- Chapter 2 discusses the concepts of efficiency, risk and return. These are key concepts within the national electricity and gas objectives, the revenue and pricing principles and the allowed rate of return objective. This chapter therefore sets important context for the way in which we exercise our judgement under the incentive regimes established by the national electricity and gas legislation.
- Chapter 3 addresses the form and structure of the Draft rate of return guidelines to which this explanatory statement applies. The remaining chapters are devoted to particular elements and parameters of the rate of return and value of imputation credits.
- Chapter 4 discusses the benchmark gearing ratio.
- Chapter 5 discusses the overall approach to return on equity.
- Chapter 6 discusses the risk free rate parameter of the return on equity.
- Chapter 7 discusses the market risk premium.

- Chapter 8 discusses the equity beta.
- Chapter 9 then explains our approach to the return on debt.
- Chapter 10 discusses the data and benchmarks for estimating the return on debt.
- And finally, chapter 11 discusses the value of imputation credits.

1 Overview

1.1 Introduction

The Australian Energy Regulator (AER) is responsible for the economic regulation of electricity and gas transmission and distribution services in eastern and southern Australia under the National Electricity Rules (NER) and the National Gas Rules (NGR) (collectively, the rules). We monitor the wholesale electricity and gas markets, and are responsible for compliance with and enforcement of the rules. We also regulate retail energy markets in the ACT, South Australia, Tasmania (electricity only) and New South Wales.

In the economic regulation of electricity and gas transmission and distribution services the allowed return on capital represents the largest component of the revenue determinations. Our rate of return guidelines set out how we will determine the allowed rate of return on capital. The rate of return is a forecast of the cost of funds a network business requires to fund investment in its network. The guidelines also set out the value we propose to assign to imputation credits.

We developed the current guidelines (the 2013 Guidelines) in December 2013. Those Guidelines can be found at <u>https://www.aer.gov.au/networks-pipelines/guidelines-schemes-models-reviews/rate-of-return-guideline-2013</u>.

Under the current legislative framework we must periodically review the guidelines and publish amended guidelines if necessary. We must review the 2013 Guidelines by December 2018.

Starting in mid-2017, the AER initiated a review of the 2013 Guidelines.

In conducting this review, we have engaged in the most extensive consultation process yet undertaken by the AER when formulating an approach to calculating the rate of return and determining the value of imputation credits.

We welcome and are grateful for numerous submissions from consumers, service providers, investors and representative groups received throughout this review process. These submissions have assisted our understanding of the issues and informed the exercise of our judgement.

In addition, we have undertaken new initiatives to better engage with both consumers and industry stakeholders to assist us in reaching our draft decision.

We have had the benefit of assistance from Reference Groups that we have established to help facilitate greater engagement with consumers, investors and retailers in the review process. Our Consumer Challenge Panel has also assisted us in taking into account consumer concerns.

An important new initiative for this review has been the establishment of a 'hot-tub' of experts in concurrent evidence sessions. In these sessions, experts that have been nominated by different consumer and industry stakeholders openly and frankly

discussed with us and each other some of the key issues that apply to the calculation of the rate of return. We conducted two concurrent evidence sessions prior to making our Draft Decision and the concurrent evidence sessions have proved a valuable tool in helping us to arrive at our Draft Decision.

A further significant new initiative we are undertaking for this review is that, following the publication of this Draft Decision, an independent panel will provide us with a report on their assessment of the evidence and our reasons supporting our Draft Decision. We will then take that independent panel report into account when making our final decision.

We also invite the public to make submissions on our Draft Decision. Submissions on this draft decision should be sent by 5pm AEST Friday 14 September 2018, to rateofreturn@aer.gov.au. Alternatively, submissions can be sent to:

Mr Warwick Anderson General Manager Networks Finance and Reporting Australian Energy Regulator GPO Box 3131 Canberra ACT 2601

We prefer that all submissions be sent in an electronic format (Microsoft Word) and are publicly available, to facilitate an informed, transparent and robust consultation process. Accordingly, submissions will be treated as public documents and posted on our website, unless prior arrangements are made with us to treat the submission, or portions of it, as confidential. Those wishing to submit confidential information are requested to:

- clearly identify the information that is the subject of the confidentiality claim, and
- provide a non-confidential version of the submission.

All materials relating to this consultation, and the process for the conduct of the review, are available on the AER's web site at <u>https://www.aer.gov.au/networks-pipelines/guidelines-schemes-models-reviews/review-of-rate-of-return-guideline</u>.

This enhanced process provides for greater transparency and stakeholder engagement. It will assist us significantly in making a final decision that best achieves the long term interests of consumers.

At the outset, therefore, we wish to place on record our thanks to all those who have participated in this new process.

1.2 The Draft Decision summarised

Rate of return

Overall, our approach to gearing, and to estimating the returns on equity and debt, broadly continues our approach to determining the rate of return adopted in our 2013 Guidelines, with updated parameter estimates and data sources.

However, an important difference between our Draft Decision and our 2013 Guidelines, is that under our Draft Decision the way to calculate the rate of return would be capable of being applied automatically at the time of making a future revenue determination.¹ We consider this will help promote investor certainty and business confidence when entering into financing arrangements.

Our Draft Decision is to calculate the rate of return as a nominal vanilla weighted average of an allowed return on equity and an allowed return on debt.

The gearing ratio is 60 per cent for the allowed return on debt and 40 per cent for the allowed return on equity.²

The allowed return on equity will be calculated as an estimated risk free rate plus a market risk premium (MRP) of 6 per cent multiplied by an equity beta of 0.6.³

This equates to an equity risk premium of 360 basis points over the estimated risk free rate. The 2013 Guidelines estimated an equity risk premium of 455 basis points (based on a MRP of 6.5 per cent and an equity beta of 0.7) over the estimated risk free rate.

The risk free rate is to be estimated based on an average of the yield on 10 year Commonwealth Government Bonds (CGS) over an averaging period of between 20 and 60 business days. Service providers are free to choose this averaging period subject to the requirements set out in the Guidelines. If they do not select an averaging period that meets the requirements set out in Guidelines, a default averaging period will apply.

The allowed return on debt will be determined on the basis of the revenue neutral transitional arrangements that we recently determined⁴ for each service provider to move from an 'on-the-day' approach to a 10 year trailing average approach to estimating the return on debt.⁵

As in the 2013 Guidelines, we consider the appropriate benchmark for estimating the return on debt is the yield on debt instruments issued at a BBB+ investment grade rating.

We will estimate a BBB+ yield using a weighted average of 10 year BBB and A rated yield curves published by the Reserve Bank of Australia, Bloomberg, and Thomson Reuters. Two-thirds weight will be placed on the average yield from the BBB curves

¹ Subject to an assessment that this will achieve the allowed rate of return objective.

² The gearing ratio is discussed in detail in chapter 4.

³ The return on equity, risk free rate, MRP and beta are discussed in detail in chapters 5-8.

⁴ We set out our proposed approach in our 2013 Guidelines, and determined the use of the transition and the start date of the transition for each service provider in their first regulatory determination following the 2013 Guidelines.

⁵ The return on debt approach and parameter estimates are discussed in detail in chapters 9 and 10.

over the debt averaging period and one-third weight will be placed on the average yield from the A curves over the debt averaging period. We do not expect the introduction of the Thomson Reuters curve to materially alter the estimated yield, although it should reduce estimate volatility. We consider that adopting a weighted average of A and BBB yield estimates will better reflect a benchmark BBB+ yield.

Each yield estimate is to be calculated over an averaging period between 10 business days and one year in length subject to certain requirements. Where a service provider does not select averaging periods that meet our requirements we will use default averaging periods.

The same methodology for estimating the rate of return applies for all service providers. We consider the degree of risk that applies to each service provider in respect of the provision of regulated electricity or gas services is substantially similar.⁶

Overall, we estimate that our draft guideline will result in a 45 basis point reduction in the overall rate of return compared to the approach we have applied in our regulatory determinations since the 2013 Guidelines. Our estimate of the impact arising from the return on debt reflects estimates over a recent averaging period.⁷ Over the next five years, the total impact could be larger or smaller depending on changes in market conditions.

Imputation credits

We determine a value of imputation credits of 0.5 (or 50 per cent).8

This is the same value we estimated in our 2013 Guideline. However, it represents an increase from the value of 0.4 that we applied in revenue determinations made between 2014 and 2018. In practice, this translates to a 10 per cent reduction in our estimated corporate tax allowances.

Reasons

The reasons that support our approaches to calculating the rate of return and determining the value of imputation credits are summarised in section 1.5 of this Overview and then explained in greater detail in individual subject matter chapters.

1.3 The objectives and principles guiding our decision

The National Electricity Objective (NEO) and the National Gas Objective (NGO) establish the ultimate objective of our decision-making.⁹ In each case, the objective is

⁶ Our consideration of risk and return is in section 2.4.

⁷ The return on debt estimates capture the impact of our draft guideline using an averaging period of all business days in February 2018. The return on equity impacts are driven entirely by changes to the equity risk premium and therefore do not depend on the averaging period.

⁸ Imputation credits are discussed in detail in chapter 11.

⁹ NEL, s. 7; NGL, s. 23.

to promote efficient investment in, and efficient operation and use of, the relevant electricity or gas services, for the long term interests of consumers with respect to the price, quality, safety, reliability and security of supply.¹⁰

When reviewing the 2013 guideline, and in making this draft decision, we must perform our functions in a manner that will or is likely to contribute to the achievement of the NEO and NGO.

In support of the NEO and NGO, the National Electricity Law (NGL) and National Gas Law (NGL) set out Revenue and Pricing Principles (RPPs).¹¹ These principles underlie the achievement of the NEO and NGO and we have had particular regard to these principles in making our draft decision.¹²

The RPPs are in essentially similar terms for both electricity and gas. In summary, those principles are:

- A service provider should be provided with a reasonable opportunity to recover at least the efficient costs the operator incurs in—
 - providing regulated services; and
 - complying with a regulatory obligation or requirement or making a regulatory payment.
- A service provider should be provided with effective incentives in order to promote economic efficiency with respect to the regulated services the operator provides. The economic efficiency that should be promoted includes—
 - efficient investment the network with which the operator provides regulated services; and
 - o the efficient provision of regulated services; and
 - the efficient use of the system with which the operator provides regulated services.
- Regard should be had to the regulatory asset base adopted
 - o in any previous determination or arrangement, or
 - o in the Rules
- A price or charge for the provision of a regulated service should allow for a return commensurate with the regulatory and commercial risks involved in providing the service.
- Regard should be had to the economic costs and risks of the potential for under and over investment by a regulated network service provider in the relevant system.

¹⁰ The NEO contains an additional objective of the reliability, safety and security of network system: see NEL s.7.

¹¹ NEL, s. 7A; NGL, s. 24.

¹² See chapter 2.

• Regard should be had to the economic costs and risks of the potential for under and over utilisation of the relevant system.

Each of these principles has an important guiding role when determining an appropriate way to calculate the rate of return in order to achieve the NEO or NGO. For example, if the rate of return is set at a rate that is too low to promote efficient investment in infrastructure, it will lead to underinvestment. It may not allow a provider a reasonable opportunity to recover at least its efficient costs in providing services or complying with regulatory obligations. It will not provide effective incentives for efficient investment in, or provision for, or use of services. It will not be a rate that provides for a return that is likely to be commensurate with the commercial and regulatory risks. It may lead to various economic costs and risks that might arise from under-investment in the network system. All of these factors would compromise the realisation of the NEO and NGO.

Similarly, if the rate of return is set too high, it will provide an incentive to over-invest in network infrastructure. It will not reflect a return that is commensurate with the regulatory and commercial risks. It will not promote efficient investment in the network system and it is likely to lead to underutilised investment in regulated assets.

There is a balance involved in having regard to these principles. We aim to determine a rate of return and a value for imputation credits that will provide the appropriate investment incentives that will lead to neither over nor under investment in assets, and achieve an appropriate balance of sustainable long term consumer outcomes in respect of price, quality, safety, reliability and security of supply. This task is not one that can be undertaken mechanically. Instead, it is one that requires the exercise of judgement looking to future outcomes. The objectives and principles guide our assessment of the evidence.

In addition to meeting the legislative obligations to achieve the NEO or NGO, and having regard to the RPPs, the National Gas Rules and National Electricity Rules also establish an allowed rate of return objective.¹³ That objective provides that the allowed rate of return is to be commensurate with the efficient financing costs of a benchmark efficient entity with a similar degree of risk as that which applies to the service provider in respect of the provision of its regulated services.¹⁴ This objective needs to be interpreted consistently with the overall NEO and NGO, and the principles set out in the RPPs.

There are certain common repeated concepts within these legislative objectives and principles that are particularly relevant to setting the rate of return and the value of imputation credits – benchmark efficiency, risk and return. We adopt standard, well established regulatory economic approaches to our understanding of each these concepts.¹⁵

¹³ NER cll. 6.5.2(b), 6.5.2(c), 6A.6.2(b), 6A.6.2(c); NGR r. 87(2) and r.87(3).

¹⁴ The allowed rate of return objective is discussed in more detail in section 2.1.3.

¹⁵ See AER, *Risk and judgement Discussion paper*, February 2018 and chapter 2 of this decision.

We estimate a benchmark rate of return which is then applied to a specific service provider, rather than determining the returns of a specific service provider based on all of its specific circumstances.¹⁶

The service providers' actual returns could differ from the benchmark regulatory allowance depending on how efficiently it finances and operates its business. This is consistent with incentive regulation. That is, our rate of return approach drives efficient outcomes by creating the correct incentive by allowing (requiring) service providers to retain (fund) any additional income (costs) from outperforming (underperforming) the efficient benchmark.¹⁷

When estimating our benchmark rate of return we have regard to the degree of risk involved in providing regulated services. This is consistent with the RPPs, which state that a price or charge should allow for a return that matches the regulatory and commercial risks involved in providing the regulated service to which that charge relates. It also contributes to the achievement of the legislative objectives by promoting efficiency – it is well accepted that there is a risk-return trade-off¹⁸ and it would not be efficient to determine an allowed return that is not commensurate with the risks involved.¹⁹

¹⁶ See AER, *Better regulation: Explanatory statement rate of return guideline,* December 2013, ch. 3.

¹⁷ NEL, s. 7A(3); NGL s. 24(2)(b).

¹⁸ Handley, Advice on the Return on Equity, Report prepared for the Australian Energy Regulator, 16 October 2014, p. 4.

¹⁹ Our assessment of the degree of risk involved in providing regulated services and how we have regard to this risk in estimating the benchmark allowed rate of return is set out in more detail in section 2.4.

Dimension of efficiency	Economic meaning	Application to rate of return estimation
Productive efficiency	Achieved when output is produced at minimum cost. This occurs where no more output can be produced given the resources available, that is, the economy is on its production possibility frontier. Productive efficiency incorporates technical efficiency. This refers to the extent that it is technically feasible to reduce any input without decreasing the output or increasing any other input.	Refers to least cost financing (that is, the lowest allowed return on debt and equity) subject to any constraints, such as risk. For our determinations to be productively efficient we need to incentivise service providers to seek the lowest cost financing (all else being equal).
Allocative efficiency	Achieved when the community gets the greatest return (or utility) from its scarce resources.	Allocative efficiency can be achieved by setting an allowed return consistent with the expected return in the competitive capital market (determined by demand and supply) for an investment of similar degree of risk as a service provider supplying regulated services.
Dynamic efficiency	Refers to the allocation of resources over time, including allocations designed to improve economic efficiency and to generate more resources. This can mean finding better products and better ways of producing goods and services.	Refers to the existence of appropriate investment incentives. We can encourage dynamic efficiency by setting an allowance that does not distort investment decisions. Dynamic efficiency is advanced through incentive regulation rather than cost of service regulation that compensates a service provider for its actual costs no matter how inefficient.

Table 1 Application of efficiency concepts to rate of return

Source: AER analysis; Productivity Commission, On efficiency and effectiveness: Some definitions, May 2013; AER, Better regulation: Rate of return guidelines consultation paper, May 2013.

Legislative amendments for a binding rate of return instrument

Finally, one further objective that we wish to highlight in this Overview relates to the legislative amendments that have been recently proposed by the Council of Australian Governments (CoAG) to replace the current Rate of Return Guidelines with a binding legislative instrument.

We understand that the legislative amendments will require the binding legislative instrument to set out how the estimation of the rate of return will be automatically applied in each regulatory determination without any exercise of discretion.

This would be unlike the current legislative framework, which allows both the service providers and ourselves the opportunity to depart from the 2013 Guidelines if the evidence justified that doing so would result in an outcome that better achieves the legislative objectives.

The proposed legislative amendments include provisions to allow this review of the guidelines to also satisfy the process for developing the first binding rate of return instrument. However, we acknowledge that at the time of this decision the legislative amendments have not been made.

In light of CoAG's commitment to implementing a binding rate of return instrument, we have therefore developed guidelines that are capable of either:

- operating as non-binding guidelines under the current legislative framework; or
- being automatically applied as a binding rate of return instrument if the legislative framework changes.

1.4 The consultation process and key stakeholder comments

We have implemented a new enhanced consultation process for this review that is designed to help us better understand consumer and industry views, and to ensure that we are able to take these views into full consideration when deciding how best to contribute to achieving the NEO and NGO through this Draft Decision.

We think it is particularly important to set out in this Overview the key elements of that process, the key concerns raised, and how we have responded to those concerns in this Draft Decision. We discuss stakeholder comments in more detail throughout the chapters and in the following section that sets out our key reasons for our draft decision.

In this section, we summarise:

- the key steps in our consultation process,
- the key concerns of stakeholders,
- submissions we received on the risks and costs associated with a rate of return that is too high or too low, and
- submissions on the scope of the review.

The consultation process

The key steps in our consultation process have included:

- In July 2017 we issued a consultation paper which sought views on how best to run the Guideline review process
- On 18 September 2017 we held a pre-issues paper public forum
- On 31 October 2017 we released an issues paper requesting views on whether our current approach to setting the allowed rate of return remains appropriate.
- On 28 November 2017 we released a positions paper setting out our positions on the process for reviewing the Guidelines.
- On 15 March 2018 we held a concurrent evidence session to discuss gearing, financial performance measures and risk and judgement. Discussion papers on the topics were made available prior to the session on 28 February 2018. Following the first evidence session we published a transcript of the session.
- On 5 April we held a second concurrent evidence session to discuss gamma, equity beta, market risk premium, the risk free rate averaging period and the

automatic application of the guideline. Discussion papers on the topics were made available prior to the session on 15 March 2018. Following the second evidence session we published a transcript.

- On 21 April 2018 we published a joint expert statement that covered view of experts in relation to many topics discussed at each of the concurrent evidence sessions.
- On 10 May 2018 we published a discussion paper addressing return on debt issues and inviting written submissions by 30 May 2018.

In this process we also formed a number of reference groups to input into the review process. These groups have actively and openly engaged with us throughout the process. This has helped us to take their members views into account in this decision. These groups were:

- A consumer reference group (CRG)
- A consumer challenge panel (CCP16)
- An investor reference Group (IRG)
- A retailer reference group (RRG)

Energy Networks Australia (ENA) and the Australian Pipeline Gas Association (APGA) both have also actively and openly engaged in this process. Both sponsored experts to participate in the evidence sessions. A number of consumer groups have also actively participated in the process including Energy Consumers Australia (ECA), Energy Users Association of Australia, Major Energy Users Inc, and the Public Interest Advocacy Centre (PIAC). ECA also sponsored an expert to attend the expert evidence sessions.

We also encouraged our CRG and ENA to engage directly to exchange views, share perspectives and explore potential areas of common ground. The CRG and ENA held a series of meetings which both have indicated were useful.

Throughout the review process we have received public submissions on our various papers, including submissions from the groups listed above.

We have had full regard to the submissions and other information before us (such as the joint statement of experts and transcripts of the concurrent evidence sessions) in making this decision. The extensive engagement from all stakeholders and stakeholder groups has greatly assisted the AER in determining the draft Guidelines that it considers will best contribute to the achievement of the legislative objectives.

Ultimately, we consider our draft decision balances the views and concerns of the service providers, investors in the sector, and consumers. We consider the overall outcome flowing from our draft decision will better achieve the legislative objectives than the alternatives that have been raised throughout this process and discussed in detail in the explanatory statement chapters.

Key comments of stakeholders

Key comments of service providers, investors, and representative organisations of these parties have included:

- The service providers and their investors value certainty and predictability.²⁰
- The current rate of return is about right (if not a little low) and the 2013 Guidelines have been contributing to achieving the NEO and NGO.²¹
- An 'incremental' review of the 2013 Guidelines is appropriate.²²
- There should be a high threshold to change from the 2013 Guidelines.²³
- Changes should be based on changed evidence since the 2013 Guidelines.²⁴
- Where the AER exercises judgement it should be clear and transparent on how it has done so.²⁵

On the individual rate of return input parameter estimates, and the value of imputation credits, it appears many of the service providers and investors do not support a reduction as an acceptable outcome.²⁶ Rather, if a change was to be made, different groups have also provided evidence to support small increases in the market risk premium and beta and a reduction in gamma.²⁷

The ENA submitted²⁸:

Since 2013 Guideline, there have been further material reductions in the allowed return on equity through reduction in the nominal risk free rate. Consequently, maintaining the return on equity parameters from the 2013 Guideline would result in a lower allowed return on equity.

Evidence suggests that current returns, which were significantly lowered in the 2013 Guidelines, are starting to impact investment levels. This is evidenced by lower actual expenditure levels relative to approved allowances.

The APGA submitted²⁹:

²⁰ NSG, Submission on the RORG review, 4 May 2018, p.3.

²¹ ENA, Response to discussion papers and concurrent evidence sessions, 4 may 2018, Section 1; NSG, Submission on the RORG review, 4 May 2018, pp.10-17; SA Power Networks, AGIP, Citipower, United Energy and Powercor, AER discussion papers – review of the rate of return guideline, 4 May 2018.

²² ENA, Response to discussion papers and concurrent evidence sessions, 4 may 2018, Section 2.3; NSG, Submission on the RORG review, 4 May 2018, p.10; APGA, Submission to the AER rate of return guideline review, 04 May 2018, p.2.

²³ NSG, Submission on the RORG review, 4 May 2018, p.5; APGA, Submission to the AER rate of return guideline review, 04 May 2018, p.2.

²⁴ NSG, Submission on the RORG review, 4 May 2018, pp11-13

²⁵ NSG, Submission on the RORG review, 4 May 2018, pp11-13.

²⁶ ENA, Response to discussion papers and concurrent evidence sessions, 4 may 2018, Section 2.3; NSG, Submission on the RORG review, 4 May 2018, p.10; APGA, Submission to the AER rate of return guideline review, 04 May 2018, p.2.

²⁷ NSG, Submission on the RORG review, 4 May 2018, p.15; ENA, Response to discussion papers and concurrent evidence sessions, 4 may 2018, Section 7; ACTO Gas, Submission on rate of return guideline review, 4 May 2018.

²⁸ ENA, Response to discussion papers and concurrent evidence sessions, 4 May 2018, p.3.

Throughout this process, we have supported the notion of incremental change. To us it means that we start from the current position on estimation approaches and outcomes and identify the need to move away from these. For example, in the case of the current cost of debt approach or current foundation model approach (to measure the cost of equity), the AER should consider any material changes in risk or market conditions that require changes to the current method. If the risks and markets conditions have changed material, only then consider looking at alternative approaches, but if they have not, do not make substantive changes to the status quo.

Comments from different consumers and consumer representative groups in this process include:

- Even under an incremental review the starting points for the parameters (estimated in the 2013 Guidelines) need reconsideration.³⁰
- The current rate of return is too high and has not achieved the NEO.³¹
- The AER was conservative (in favour of the service providers) in estimating all WACC input parameter values towards the top of empirical ranges (or in favour of service providers) in its 2013 Guidelines. In this review, the AER should set less conservative estimates.³²
- Consumers themselves are facing large risks associated with increasing energy prices, particularly vulnerable consumers. This should be given consideration when the AER exercises judgment in the determining the rate of return and gamma. In light of these concerns, the AER should not be conservative in exercising its regulatory judgement in favour of service providers.³³
- Relatively lower demand growth and asset utilisation warrant a less conservative approach (less biased towards encouraging investment).³⁴
- With excess capacity present in most networks the balance of risks between too much and too little investment has shifted and consumers suggest that they are not concerned about too little investment in the next period due to the large amounts of investment over the past decade.³⁵

All consumer based submissions supported a decrease in the overall rate of return, although the appropriate decreases varied in consumer submissions. The CRG

²⁹ APGA, Submission to the AER review of the rate of return guideline, 4 May 2018, P.2

³⁰ CCP, Submission on RORG review and concurrent evidence sessions, 4 may 2018, section 3.1.

³¹ CCP, Submission on RORG review and concurrent evidence sessions, 4 may 2018, section 5; CRG, Submission to the AER rate of return guideline review, sections 2 and 3.

³² CCP, Submission on RORG review and concurrent evidence sessions, 4 may 2018, section 3.4; CRG, Submission to the AER rate of return guideline review, p.35.

³³ CRG, Submission to the AER rate of return guideline review, section 3.

³⁴ CCP, Submission on RORG review and concurrent evidence sessions, 4 may 2018, p.24; CRG, Submission to the AER rate of return guideline review, section 3.

³⁵ CRG, Submission to the AER rate of return guideline review, section 3.

supported a reduction in the MRP (from 6.5 per cent to 5.75 per cent), a reduction in beta (from 0.7 to 0.3), and an increase in gamma from 0.4 to 0.83.³⁶ The CRG stated³⁷:

Whilst there is no formal mechanism for testing whether the Guideline serves the long term interests of consumers, available evidence demonstrates the objectives are not being met. The reverse is occurring in an environment of increasing energy prices that could be described as an 'affordability crisis'.

Increasing network charges have been a significant contributor to these unsustainable prices.

Over the last decade the combined regulatory asset base (RAB) of the electricity distribution networks has almost doubled while network utilisation has declined from just under 60 per cent to just over 40 per cent. Despite these changes, network businesses are continuing to enjoy string earnings and are trading at multiples of 1.3 to 1.6 of the RAB. Conversely, there is no evidence of under-investment resulting in a decline in network reliability.

On the other hand CCP16 supported a relatively smaller reduction in the MRP (from 6.5 per cent to 6 per cent), in beta (from 0.7 to 0.6) and an increase in gamma from 0.4 to 0.5.

Consumers also submitted that our cost of debt allowance is too high. CCP 16 submitted³⁸:

The analysis by Chairmont (April 2018) of the debt portfolios and strategies of the networks provides empirical evidence that the actual costs of debt of the network firms may be lower than the ROD that is currently estimated by the AER.

A summary of stakeholder submissions are contained as an appendix in most of the explanatory statement chapters. Submissions are available on our website.³⁹

Submissions on the risks and costs associated with a rate of return that is too high or too low

This review has been undertaken in an environment of heightened consumer concern about increasing energy costs and relatively large (and in some cases underutilised) investment in regulated assets over recent years.

Some consumer submissions referred to 'a conventional wisdom' in Australian energy regulation – namely, that when faced with uncertainty it is better to err by choosing

³⁶ CRG submitted that the MRP should be in the range of 5.5 per cent to 6.0 per cent. CRG submitted that equity beta should be in the range of 0.2 to 0.5, and that a point estimate should be below the midpoint of this range.

³⁷ CRG, Submission to the AER rate of return guideline review, pp.v-vi.

³⁸ CCP 16, Submission to the AER on its allowed rate of return on debt discussion paper, May 2018, p.4

³⁹ https://www.aer.gov.au/networks-pipelines/guidelines-schemes-models-reviews/review-of-rate-of-return-guideline

outcomes that are likely to promote too much investment rather than too little. During the guideline review process, this notion was challenged.

The CCP 16 stated:40

In the past, governments and regulators in Australia have emphasised regulatory outcomes that were designed to ensure sufficient investment. However, in the last 5 years, market circumstances have changed significantly and many networks now have excess capacity, flat demand, and the ability to manage constraints through demand side rebates to influence behaviour.

In our view, the evidence taken as a whole strongly suggests that historical rate of return allowances have been higher than necessary to encourage efficient investment, particularly given the shift away from incentivising investment to consolidating investment decisions of the past.

First, consumers submitted that when we exercised judgement in the 2013 guideline, we did so by choosing outcomes that were systematically in favour of the service providers and their investors. Specifically, consumers point to our choices of beta at the upper end of a possible range (including reference to the Black CAPM), MRP (and our use of DGMs), use of the Wright model, term of debt, choice of debt data series and taxation allowance. It was submitted that these choices have a compounding effect that are likely to lead to too high an overall rate of return and, as such, will not properly take into account the RPPs.

Second, consumers submit that the bias in our judgement has led to an overall rate of return that is too high and this has encouraged excessive investment in networks. Further, the costs of too much investment have been understated. Consumers point to rapid increases in network revenues and asset bases, financial distress experienced by consumers, a chilling effect on the broader economy, but buoyant investment in and acquisitions of network assets.

Third, consumers point to declining demand, declining network utilisation, declining average age of assets, a positive upward trend in measures of network performance and the positive option value that can be realised by deferring network investment. As such, consumers submit that the balance of risks has shifted. There is little risk from under-investment because there is unutilised capacity present in each of the network systems.

Consequently, consumers submit that when we exercise judgement in this current guideline process we should do so in favour of a lower, rather than a higher, rate of return. When put in the context of the NEO and NGO, consumer representatives have clearly indicated, during this consultation process, a willingness to accept a higher level of risk in respect of the rate of return and the investment it is intended to promote in exchange for lower prices.

⁴⁰ CCP 16, Submission on RORG review and concurrent evidence sessions, 4 May 2018, p.5.

However, we also accept submissions made by service providers and investors that we should exercise our judgement with care. There is an ongoing need for investment to replace existing assets, to address locational peak demand and to reconfigure networks in response to changes in the mix of generators. Continued investor confidence is important in achieving these investment outcomes. We are conscious that the rate of return should be set in a manner that is sufficient to attract capital on a long-term sustainable basis, given the opportunity costs, if we are to achieve the NEO and NGO.

Ultimately we are seeking to reach a decision that will promote efficiency in the long term interests of consumers. We consider this requires a degree of caution when exercising our judgement. Nevertheless, we are cognisant of the cumulative effect of choosing high parameter estimates from a reasonable range of estimates and the risks that might follow.

Overall, we accept that these propositions, as highlighted by both consumers and industry representatives, are important considerations. They are relevant factors to be considered in the context of the RPPs and our assessment of how best to achieve the balance of factors set out in the NEO and NGO.

We have taken these considerations into account in this draft decision particularly when considering a best estimate for the market risk premium and equity beta in our calculations for the return on equity. We have also taken these factors into account when deciding how best to estimate the BBB+ yield when estimating the return on debt.

There is a sharp disagreement between consumers on the one hand and networks and investors on the other about how the rate of return should be set in order to advance the NEO and NGO. While there is a level of agreement about the Framework we should apply, its application is strongly contested. We have been conscious to listen carefully to both sides of the debate and to weight the arguments put to us.

Where we exercise judgement, we do so placing our emphasis on market data and avoiding choices that are influenced by any material bias in either promoting or discouraging investment. We consider that the promotion of efficient investment will flow from a decision that reflects well established economic approaches as supported by the available evidence, always having regard to the principles set out in the RPPs and the various elements we are seeking to achieve in the NEO and NGO.

Key stakeholder comments on the scope of the review

We first publically proposed a 'targeted' approach to our review in our 18 September 2017 workshop and in an October 2017 issues paper.⁴¹ The specific questions we asked were: ⁴²

⁴¹ AER, *Issues paper – Review of the rate of return guidelines*, October 2017, p7.

⁴² AER, Issues paper – Review of the rate of return guidelines, October 2017, p

- Should the AER build on the knowledge base gained from development and application of the current guideline or start from a blank slate?
- What are the high, medium and low priority issues for the guideline review?
- Should the AER prepare a prescriptive guideline to enable the mechanistic estimation of the rate of return through a formula?

While most participants agreed that the AER would be unwise to start from a blank slate, several groups noted that the knowledge gained to date on the AER's approach should be open to question.⁴³ In many ways, this also stems from the concerns discussed in the previous section on risks and costs in setting a rate of return that is too high or too low.

In our issues paper we indicated we were reconsidering the weight to give to different pieces of evidence, including whether some information referred to in our 2013 Guideline should be given any weight. For example, we asked stakeholders whether we should use the Black CAPM to inform the equity beta point estimate and the appropriate role of dividend growth models in setting the allowed return on equity.⁴⁴

While submissions on our issues paper were generally supportive of an 'incremental approach' to this review, submissions from consumers suggested the 2013 Guidelines had not achieved the NEO and NGO on the basis that the allowed rate of return from applying it was too high.⁴⁵ They also submitted that it should not be simply assumed that the approaches or the parameter estimates adopted in the 2013 Guidelines would remain appropriate to achieve the rate of return objective, and the NEO and NGO.

We note that there was a general level of agreement amongst stakeholders to:

- Applying a 'utilisation' based post-company tax approach to estimating the value of imputation credits
- Applying the foundation model approach for estimating the allowed return on equity, with the SLCAPM used as the foundation model
- Using a benchmarking approach for estimating all key parameters when estimating the allowed return on equity and allowed return on debt
- Continuing the use of a trailing average cost of debt with a revenue neutral transition
- Using third party data services for estimating the allowed return on debt

In light of these stakeholder comments we have used our 2013 Guidelines as a starting reference point for our analysis, mindful of our obligation to reach a decision that we

⁴³ AER, AER Rate of Return Pubic Workshop: Discussion summary, September 2017, p3.

⁴⁴ AER, *Issues paper – Review of the rate of return guidelines*, October 2017, p9.

⁴⁵ Energy Consumers Australia, *Review of the rate of return guideline*, December 2017, p XX. [add in other submissions here]

are satisfied will, or is likely to contribute to the achievement of the NEO and NGO and which meets the allowed rate of return objective.

At our public forum in September 2017 some stakeholders questioned whether our 2013 Guidelines had contributed to achieving the legislative objectives. Similar submissions were subsequently made by the CRG, ECA, MEU, and EUAA.⁴⁶

The CRG submitted that it considers our 2013 Guidelines have not been contributing to achieving the objectives to the greatest degree, stating:⁴⁷

The CRG strongly believes the Guideline is not meeting its objective.

There is no testing as to whether the Guideline serves the long term interests of consumers. The available evidence demonstrates the objectives are not being met.

The EUAA submitted:48

Our view is that the 2013 Guideline has not met the NEO and NGO objectives.

...

Consumers have received a safe and reliable service. Unfortunately, we believe it has been achieved with far too much capital investment (see the fall in capacity utilisation rates) and poor productivity by many networks with the result that prices are far too high. Many networks are not at "benchmark" efficient levels and yet they continue to earn a secure revenue flow that produces, what limited evidence suggests, to be above normal profits for a regulated natural monopoly.

By contrast, the network shareholder group submitted:49

The framework for incentive-based economic regulation is working as intended, specifically where investment is made by the private sector.

Energex and Ergon Energy submitted that:50

Certainly, there are aspects of the AER's current approach that most stakeholders agree will contribute to the achievement of the NEO and ARORO.

Energy Networks Australia submitted that our current review should build on the 2013 Guidelines, stating:⁵¹

⁴⁶ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 16.

⁴⁷ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 16.

⁴⁸ EUAA, EUAA submission – AER Rate of Return Review Issues Paper October 2017, December 2017, p. 2.

⁴⁹ NSG, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), May 2018, p. 1.

⁵⁰ Energex and Ergon Energy, AER Issues Paper - Review of the Rate of Return Guidelines - Ergon Energy and Energex Submission, December 2017, p. 2.

⁵¹ ENA, AER Rate of Return Guidelines Response to Issues Paper, December 2017, p. 3.

The 2017 Rate of Return Guideline process is an opportunity to build on considerable work undertaken in the 2013 guideline review process, and network businesses support an incremental approach building on this past guideline review

If the rate of return derived from our guideline is too high or too low then it will not promote the achievement of the legislative objectives. Consumer groups have strongly advocated that allowed rate of return determined since our 2013 guidelines has been too high and has therefore not promoted efficient investment in, and efficient operation and use of, electricity services for the long term interests of consumers of electricity.

We acknowledge that this review is being undertaken in an environment of heightened consumer concern about increasing energy costs and relatively large (and sometimes underutilised) investment in regulated assets bases over recent years. While the rate of return is an important contributor to network prices, there are other network and nonnetwork costs that also contribute to overall energy prices. Nonetheless, we are cognisant of the effect that higher energy prices may have on the willingness of consumers to pay for further improvements in network reliability.

We have had regard to these factors when considering the rate of return and value of imputation credits for this decision. We have done so by adopting an approach of focusing on the best empirical estimates for rate of return parameters. In this way we consider we can determine an approach that is most likely to promote the NEO and NGO.

We consider it is important for the achievement of the legislative objectives that we periodically review our rate of return and value of imputation credits. This process of review, combined with the use of incentive regulation, is a key driver of efficiency and the continued achievement of the legislative objectives over time. Therefore, the achievement of the legislative objectives should not be viewed solely in a static sense.

On the use of incentive regulation, the ECA submitted:52

The objective is not to set the rate of return based on a benchmark so that the provider can outperform the rate of return by the way it is financed – the intention is that the rate of return is a constraint so that the provider has maximum incentive to generate higher returns by efficiency in its investments and its operations.

We agree that the objective of the allowed rate of return under an incentive regulatory framework is not to provide a guaranteed degree of outperformance. However, we also note that it is important for allocative and dynamic efficiency that the allowed rate of return provides (in expectation) an opportunity for service providers to recover their efficient costs (without expectation of monopoly rents).

⁵² ECA, Review of the rate of return guideline: Response to the AER Issues Paper, December 2017, p. 11.

We note that we have updated our empirical analysis in a number of areas consistent with incentive regulation. We have reviewed our benchmark gearing, credit rating, debt term, and overall debt costs by examining the recent, actual costs and financial management practices of service providers. In the chapter 10 we set out how we have considered service providers' revealed cost information when deciding on our approach to debt.

1.5 Summary of our reasons in support of our Draft Decision

Actual (realised) returns and the overall allowed rate of return

The CCP16, CRG, ECA, EUAA, and MEU submitted that we should examine information on the actual returns achieved by service providers, through data on actual profitability and RAB multiples, when considering if our rate of return achieves the legislative objectives. The CCP16 submitted that actual profitability and RAB multiples could be used as a cross-check on the selection of parameter values and the overall rate of return.

We consider actual profitability and RAB multiples in chapter 2. In summary, actual profitability and RAB multiples from asset sales may provide some indication of the appropriateness of the allowed total rate of return. However, they may also be indicative of other elements of the firm's cash flows, such as unregulated activities and outperformance on expenditure allowances. Further, sales that have already occurred and past profitability may not reflect changes to the overall rate of return that are occurring at present.

Overall, we consider that these measures cannot be used to directly determine parameter estimates for the allowed rate of return. We agree with the CCP16 submission that there is difficulty in disaggregating the information contained in RAB multiples and historical profitability measures to determine the degree of outperformance of the allowed rate of return.

However, we consider that there may be useful information within the trends in RAB multiples and historical profitability measures over time. Comparisons of RAB multiples and historical profitability measures can provide information on the performance of the regulatory system as a whole. This information may be helpful in considering whether the business' actual rate of return has been systematically lower or higher than the allowed rate of return.

The CCP16 submitted that this information cannot be used at a parameter level but can inform the overall exercise of judgement in setting the rate of return or reviewing other elements of the regulatory regime⁵³. We agree that RAB multiples and historical profitability may provide useful contextual information and cause for further examination of the material we rely on when estimating rate of return parameters (other elements of the regulatory regime are beyond the scope of this review). We have done this as part of this review through further consideration of the impact of regulation on equity beta estimates within our comparator set (see chapter 8), examination of service providers' actual debt issuances (see chapter 10), and further consideration of the most appropriate third-party debt data to reflect our benchmark

⁵³ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 27-28.

credit rating (see chapter 10). Though outside the scope of this review, we are also currently undertaking reviews of other parts of our regulatory regime, such as our review of our regulatory tax approach and our review of profitability measures for gas and electricity businesses.

In addition, we note in chapter 2 that use of market data is an important concept for achieving the legislative objectives, and we do have regard to actual returns in other parts of our rate of return estimation. Our equity beta estimation is based on data on the returns (capital gains and dividends) to listed Australian energy network firms relative to the returns on the ASX300. When estimating the market risk premium we have regard to the historical returns on the Australian stock market. We have also reviewed the actual debt costs of regulated service providers.

Risk

We are required to estimate an efficient rate of return that contributes to the achievement of the NEO and NGO, the RPPs and the ARORO by promoting efficiency in the investment, operation and use of, energy network services for the long term interests of consumers. We must consider how to efficiently compensate for the risk exposure of service providers in supplying regulated services.

It is important to emphasise that the relevant risk is the risk associated with provision of regulated services. It is not the risk of the service provider more generally. This principle is reflected in the NEO/NGO, revenue and pricing principles and the allowed rate of return objective.

Our view is that the only risk that should be compensated through the rate of return is systematic risk. This has agreement from stakeholders and experts. Systematic risks are those that affect the entire economy and cannot be eliminated through diversification. Since non-systematic risks can be eliminated by diversification, investors do not require compensation for them. We also consider that technological risks and regulatory risks as submitted by service providers should not be compensated by the rate of return as these are non-systematic risks.

Stakeholders have submitted divergent views on the risks of supplying regulated energy services, how these risks should be compensated and whether market data is appropriate for measuring these risks. We consider that an efficient return for risk should be estimated through a forward-looking rate of return using relevant market data. We consider this is best achieved through a continued reliance upon the foundation model framework adopted in the 2013 Guideline with the SLCAPM as the foundation model together with the trailing average approach to debt that we have adopted in our revenue determinations during the last 3 years. This approach provides a return for each service provider at the prevailing efficient market rates for their

⁵⁴ The extent to which these risks should be compensated through other aspects of our regulatory determinations will be considered on a case by case basis as part of each determination.

systematic risk at each point in time when the return on equity, or annual update of debt in the trailing average, is determined.

We continue to be of the view that the overall systematic risk of supplying regulated services is low. This is due to the nature of the regulated services being provided as essential services. Risk is also mitigated by the regulatory framework through mechanisms such as cost pass through provisions. Moreover, the regulated asset base is protected, and an allowed return on capital and depreciation is provided via regulated revenues including indexation. Consumers have submitted that systematic risk is lower for regulated services than for other services offered by service providers because regulation mitigates the ability to extract monopoly rents and increases the certainty of the revenue stream, thereby reducing risk. We find that there is support for this proposition as Figure 1 shows a trend of equity beta estimates increasing as the proportion of regulated revenue decreases.



Figure 1 Regulated revenue and beta estimates

Source: Bloomberg; AER analysis

The division between regulated and unregulated is based on the most recent publicly available information on the proportion of regulated and non-regulated activities for a financial year. SKI is based on page 90 of its 2017 Annual Report; DUET is based on its 2016 Annual Report; ⁵⁵ Envestra is based on page 5 of its 2013 Annual Report; AST is based on page 75 of its 2017 Annual Report; GasNet is based on page 3 of its report for the half-year ended 30 June 2016; AAN is based on page 24–25 its 2006 Annual Report; AGL is based

⁵⁵ In our estimation, Dampier to Bunbury Natural Gas Pipeline (DBNGP) is 100 per cent regulated even though DUET's 2016 Annual Report reported that the ERA's regulatory tariff applies to 15 per cent of the capacity to 2020. This is because the ERA still regulates the pipeline and its regulatory tariffs would apply in the absence of renegotiated Standard Shipper Contracts as indicated by DUET (page 11 of 2016 Annual Report).
on page 68 its 2006 Annual Report; APA is based on page 16 of its 2017 Annual Report; HDF is based on page 10 of its 2011 Annual Report.

This approach to risk provides the context for our assessment of the key parameters within the rate of return. It informs our assessment of what amounts to a similar degree of risk to that faced by each service provider in the provision of their respective regulated services for the purposes of the allowed rate of return objective. It informs our assessment of what is an efficient and commensurate return for a service provider's risks.

Overall approach to return on equity

In determining the way to calculate the allowed return on equity we have used our foundation model approach with the Sharpe-Linter Capital asset pricing model (SLCAPM) as the foundation model. The principal reason for this is that SLCAPM is theoretically based, is the most extensively used and accepted asset pricing model, and has survived the test of time. This essentially reflects the same reasons set out in our 2013 Guideline.⁵⁶ It maintains the general approach affirmed by the Australian Competition Tribunal.⁵⁷

The foundation model approach provides a framework for systematically considering relevant information and then exercising our judgement on the appropriate choice of the regulated return on equity. The approach recognises that our task requires us to exercise judgement because we are estimating a forward looking return on equity that will satisfy the national electricity and gas objectives. Further, the information available to inform our decision is imprecise, incomplete and to some extent, conflicting. Figure 2 provides a flow chart of the six steps in our foundation model approach process.

⁵⁶ AER, Explanatory statement, Rate of Return Guideline, 2013, section 5; AER, Explanatory statement appendices, Rate of Return Guideline, 2013, Appendix A – Assessment of models.

⁵⁷ Application by PIAC and AusGrid, [2016] ACompT 1, Return on equity paras 632-814, see use of the SLCAPM paras 714 -804.

Figure 2 Foundation model approach flowchart



Most experts at the concurrent evidence sessions agreed that we should continue to apply the foundation model approach and focus on its application in light of the evidence that has evolved.⁵⁸ No stakeholders submitted that we should review our foundation model approach. This is consistent with our position, accepted by all stakeholders that this review should be an incremental review. In this context, we will update the relevant data and review new evidence so that our judgement can be exercised within the established approach to estimating the allowed return on equity. We consider that this provides the necessary certainty and predictability that stakeholders have said they value whilst allowing us to discharge our regulatory task in a manner that is most likely to contribute to the legislative objectives.

We agree that the foundation model approach should be continued and adopted in this 2018 Guideline review for estimating the return on equity. In this context, we will update the relevant data and review new evidence so that our judgement can be exercised within the established approach to estimating the allowed return on equity. We consider that this provides the necessary certainty and predictability that stakeholders have said they value whilst allowing us to conduct our regulatory task in a manner that is most likely to contribute to the legislative objectives.⁵⁹

Overall, in implementing steps 1 and 2 of our foundation model approach, we have not identified any additional classes of material that we did not consider when preparing our 2013 guideline. Stakeholders did not submit any additional classes of material, but some stakeholders submitted that we should reconsider the weight we give to different classes of material when exercising our judgement. We continue to assign the same roles to each piece of material as we did in 2013. However, when implementing step 3 within our overall framework, and based on new evidence and current material, we are persuaded that we should adjust the relative merit assigned to some pieces of material. The material, role and relative merit in 2018 is shown in Figure 3.

Material (step one)	Role in 2013 (step two)	Role in 2018 and relative merit
Sharpe-Lintner CAPM	Foundation model	No change
Black CAPM	Inform foundation model parameter estimates (equity beta)	No change in role. However, at this time we have diminished confidence in the robustness of the Black CAPM and are therefore not persuaded to select an equity beta towards the top of the observed empirical estimates
Dividend growth models	Inform foundation model parameter estimates (MRP)	No change in role. However, at this time we have diminished confidence in the robustness of DGMs and are therefore not persuaded to select an MRP towards the

Figure 3 Relevant material and role

⁵⁸ Joint Expert Report, RORG review – Facilitation of concurrent evidence sessions, CEPA, 21 April 2018, section 2.1.3, p.19.

⁵⁹ This approach is consistent with the CCP16's submission that, incremental change means continuing to work within the CAPM and foundation model framework, not limiting the exercise of our statutory obligations to satisfy a self-imposed 'incremental approach' and considering all available data. See, CCP 16, *Submission to the AER on its rate of return guideline review concurrent evidence sessions*, 4 May 2018, section 3.1, pp. 14-16.

top of the observed empirical estimates of historical excess returns.

Fama–French three factor model	No role	No change
Commonwealth government securities	Inform foundation model parameter estimates (risk free rate)	No change
Observed equity beta estimates	Inform foundation model parameter estimates (equity beta)	No change
Historical excess returns	Inform foundation model parameter estimates (MRP)	No change
Survey evidence of the MRP	Inform foundation model parameter estimates (MRP)	No change
Implied volatility	Inform foundation model parameter estimates (MRP)	No change
Other regulators' MRP estimates	Inform foundation model parameter estimates (MRP)	No change
Debt spreads	Inform foundation model parameter estimates (MRP)	No change
Dividend yields	Inform foundation model parameter estimates (MRP)	No change
Wright approach	Inform the overall return on equity	We have diminished confidence in the robustness of the Wright approach.
Takeover/valuation reports	Inform the overall return on equity	No change
Brokers' return on equity estimates	Inform the overall return on equity	No change
Other regulators' return on equity estimates	Inform the overall return on equity	No change
Comparison with return on debt	Inform the overall return on equity	No change
Trading multiples	No role	No change
Asset sales	No role	No change
Brokers' WACC estimates	No role	No change
Other regulators' WACC estimates	No role	No change
Finance metrics	No role	No change

Implementing step 3 of the foundation model approach (estimating the SLCAPM) is a key step, in our six step approach and it has been adopted and applied in all our determinations since 2013. We undertook extensive consultation and assessed a number of models including the dividend growth model (DGM), Black CAPM and the

Fama French model for suitability as the foundation model, before deciding on the SLCAPM. Our comprehensive assessment is in the 2013 Guideline documentation.⁶⁰

We received a wide range of submissions supporting the SLCAPM framework.⁶¹The joint expert report noted that there has been no compelling evidence to change our approach.⁶² The Network Shareholder Group and Energy Networks Australia submissions following the concurrent expert sessions supported this view.⁶³ The Consumer Reference Group submitted that a more fundamental review was required due to major issues with the CAPM.⁶⁴ While we acknowledge that all models are simplifications, we agree with Partington and Satchell that the SLCAPM framework is the best approach to reflect the systematic risk of a benchmark efficient entity.⁶⁵

Our positions on the SLCAPM parameter inputs estimated in step 3 and our key reasons are set out below.

Risk free rate

Our draft decision is to use daily 10 year Commonwealth Government Security (CGS) yields as the basis for calculating the return on equity. The CGS yields should be averaged over a period as close as practically possible to the start of the regulatory control period to determine the risk free rate. This is the same as our 2013 Guideline, except that we have now widened the averaging period from 20 business days to a period between 20-60 days as nominated by service providers.

ENA and the CRG supported the change to the averaging period and we had no opposing submissions. ENA submitted:

This proposal [a longer period proposed by the regulated business between 20-60 business days] is outlined in the AER MRP discussion paper, and is supported by both the ENA and CRG.⁶⁶

The CRG stated:

⁶⁰ AER, Explanatory statement, Rate of Return Guideline, 2013, section 5; AER, Explanatory statement appendices, Rate of Return Guideline, 2013, Appendix A – Assessment of models.

⁶¹ Network shareholders group, Submission on the Rate of Return Guideline review, p.9, Energy Networks Australia, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Evidence Sessions, p.8, Australian Pipeline Gas Association, Submission to the AER, Review of rate of return guideline, 4 May 2018, p.2.

⁶² Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, p.23.

⁶³ Network Shareholder Group submission to the Rate of return guideline review, 4 May 2018, p 9, Energy Networks Australia, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Evidence Sessions, p. 8.

⁶⁴ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p.37.

⁶⁵ Partington, G., Satchell, S., Report to the AER: Allowed Rate of Return 2018 Guideline review, May 2018.

⁶⁶ ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.4.

On the basis of that dialogue [between the CRG and ENA], the CRG supports the AER's proposal [a longer period between 20-60 business days] ⁶⁷

Our reasons for lengthening the averaging period as nominated by service providers are:

- that it provides service providers the opportunity to mitigate their exposure to short term volatility in CGS yields
- the longer period does represent a further departure from the theoretical ideal of an on-the-day rate. However, the pragmatic benefit of giving service providers flexibility in choosing the length of the period is likely to reduce volatility and is in the interest of all stakeholders.
- allowing service providers to nominate their own averaging period has a pragmatic benefit also as it provides them with further flexibility in how they mitigate their exposure to short term volatility, through the use of financial arrangements

Our draft decision is to maintain the use of a 10 year CGS term as it is consistent with the 10 year term used for the return on equity parameters. The CRG suggested using a term of five years for the risk free rate, but this would be inconsistent with the term used for the return on equity parameters. ⁶⁸

We have codified the process for determining the risk free rate according to the methodology in our draft decision so that it can be applied during the life of the Guideline without exercising discretion, as required under the draft legislation for a binding guideline.

MRP

The market risk premium (MRP) is the excess return above the risk free rate that investors require (in an ex-ante sense) to invest in the market portfolio. Our MRP estimate for this Guideline is 6.0 per cent selected from a range of 5.0 to 6.5 per cent based on historical excess returns. This point estimate compares with 6.5 per cent in our 2013 Guideline.

We have looked at the evidence available to us and consider that the overall approach for setting the market risk premium in the 2013 Guideline remains appropriate. That is, we continue to give most weight to historical excess returns (HER) and less weight to other evidence.

Stakeholders expressed divergent views on a variety of topics related to the MRP. Service providers and the Network Shareholder Group (NSG) stated that more weight should be given to the DGM and the Wright Approach and less or no weight to geometric averages of HER. ENA submitted:

⁶⁷ CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.39.

⁶⁸ CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.44.

The Dividend Growth Model (DGM) provides useful evidence on the current MRP and should be given explicit and material weight by the AER.⁶⁹

ENA reiterates the view ... that the geometric averages is inappropriate for the purpose of estimating the expected excess return ... the geometric average should not be used and only the arithmetic average should be used for the purpose of setting an allowed rate of return.⁷⁰

Consumer groups submitted that more weight should be given to geometric averages of HER and no weight should be given to DGMs and the Wright Approach. The CCP16 stated:

Whilst recognising that the DGM has a 'solid theoretical basis' and has value in certain circumstances CCP16 remains concerned about the reliance of the DGM in the context of an ex-ante regulatory decision.⁷¹

CCP 16 submitted that the AER has not placed adequate reliance on the outputs of the geometric average nor critically assessed the problems in the arithmetic average given the volatility of the annual Australian equity market returns of 17.7 per cent that Dimson, Marsh and Staunton (2015) report.⁷²

Experts agreed that HER, DGM and surveys can be used to estimate the MRP. They concluded that a long term series of returns are useful for estimating the long-run MRP but disagreed on a range of topics including the role of the DGM and the use of arithmetic versus geometric average returns. Our approach and views on these differences are set out in our key reasons below.

These differences in approach led stakeholders to recommend different values for the MRP. Service providers and the NSG highlighted increased risks and an increase in estimates of HER and recommended use of an MRP of 7 per cent. By contrast, consumer groups submitted the 2013 Guideline MRP was too conservative and pointed to greater stability in market conditions. CRG recommended an MRP of no more than 3.6 per cent based on its view that the geometric mean of the HER between 1958 and 2017 is the most appropriate measure.⁷³ CCP 16 recommends an MRP no higher than 6.0 per cent based on using HER estimates as an anchor to estimating the MRP and giving more weight to geometric averages.⁷⁴

When exercising our regulatory judgement, we rank the utility of different types of evidence at that time and then we qualitatively consider whether our initial estimate of

⁶⁹ ENA, *Response to Discussion papers and concurrent expert evidence sessions*, May 2018, p.10.

⁷⁰ ENA, *Response to Discussion papers and concurrent expert evidence sessions*, May 2018, pp.161-162.

⁷¹ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.89.

⁷² CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.88.

⁷³ CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.57.

⁷⁴ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.87.

the MRP should be moved up or down.⁷⁵ In the 2013 review process, we stated that 6.0 per cent is an appropriate estimate of the historical excess returns (HER) evidence and the starting point for our determination of a point estimate.⁷⁶ We, then moved our estimate up based on the direction of the other evidence, particularly having regard to DGM evidence.

Our key reasons for choosing a market risk premium of 6.0 per cent are:

- We consider that most reliance should be placed on estimates of HER. This material is widely used amongst regulators and market practitioners and is transparent and replicable. We take into account arithmetic and geometric averages over a range of time periods. Since 2013, some time periods now show increased averages. Table 2 below displays the latest HER and the estimates as at 2013. It shows the arithmetic average range of 6 per cent to 6.5 per cent and geometric range of 4.2 per cent to 5.0 per cent. The latest estimates of HER leads us to a range of 5.0–6.5 per cent. This gives consideration to all the information as well as trends in the data and multiple time periods whilst giving most weight to the arithmetic averages of single year returns. From this range, 6.0 per cent is an appropriate HER estimate at this time and the starting point for our determination of a MRP estimate.
- The key reason for the decrease to our MRP estimate from 6.5 to 6 percent per annum is, in this 2018 review process, our diminished confidence in the robustness of DGMs. We are therefore not persuaded to select a MRP towards the top of the observed HER range.
- Our confidence in DGM estimates in the Australian context has decreased since the 2013 Guideline. We note that experts are divided on the level of confidence we should place on DGMs. The major concern with the reliability of estimates from DGMs revolve around the challenge of forecasting the growth rates in dividends, particularly the terminal growth rate. We recognise we had concerns with the use of DGMs in 2013 as well, and note ENA's view that our concerns are not new and therefore we should not adjust our view.⁷⁷ However multiple submissions have stated these issues have become better highlighted since the 2013 guideline. Since 2013, our concerns about possible biases and diverging results have

⁷⁵ ENA, *Regulatory discretion and market risk premium determination*, 26 June 2018.We acknowledge receipt of a late submission from the ENA on 26 June 2018 which has not been subject to a full assessment. The submission included 3 options for exercising regulatory discretion and our initial assessment is as follows. The first option involves assigning specific numerical weights to each relevant piece of evidence. Our experience suggests that this level of precision is not possible. The second option is to set out a ranking of weights to each piece of evidence in terms of relative weight in achieving the final MRP. This option appears to be largely similar to our current approach to exercising regulatory discretion. The third option which recommends setting a 'neutral' long run estimate (for MRP or Total Market Return) as the default and then qualitatively making adjustments, based on a pre-set movement up/down relative to strength of the evidence. Our initial view is that this option would unnecessarily restrict future exercise of regulatory judgment because of 'default' settings becoming a hurdle/onus of persuasion.

⁷⁶ AER, *Explanatory statement - Rate of Return Guideline*, December 2013, p. 97.

⁷⁷ ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.67-71.

increased. Since 2013, we have also received advice about further issues with the DGM and its lack of suitability to reliably track changes in the MRP.⁷⁸

- There are numerous issues surrounding the estimation of dividend growth rates selection and there is a wide variety of potentially acceptable growth rates which could be used in the DGM. With the range of potential growth rates varying from as low as 1 per cent to as high as 5.5 per cent, the DGM based MRP estimate could vary by around 4 per cent purely due to the chosen growth rate.⁷⁹ Given this large potential for error in the MRP estimate driven by growth rate selection, our decision must account for the potential error or unsuitability in the estimate. Further, the models tend to produce results that are out of step with each other. As shown in Figure 4, we note that different constructions and application of the DGM produce a wide range of outcomes.
- Surveys reports remain largely unchanged from 2013 and continue to support an MRP estimate between 6 and 6.5 per cent.
- We continue to use three conditioning variables to inform (or 'condition') our initial MRP estimate. These are implied volatility, dividend yields and credit spreads. Our MRP estimate of 6.0 per cent is consistent with the conditioning variables. In particular, the decreased volatility in equity markets and the material reductions in debt risk premiums over the past 5 years.
- Debt risk premiums have declined materially over the past 5 years from around 340 basis points at 1 January 2013 to 189 basis points as at 29 March 2018.

Sampling period	Arithmetic average	Arithmetic average (2013 guideline)	Geometric average	Geometric average (2013 guideline)
1883–2017	6.3	6.3	5.0	4.8
1937–2017	6.0	5.9	4.2	3.9
1958–2017	6.5	6.4	4.2	3.8
1980–2017	6.4	6.3	4.3	3.8
1988–2017	6.0	5.7	4.5	3.6

Table 2 Historical excess returns

Source: Handley, An estimate of the historical equity risk premium for the period 1883 to 2011, April 2012, p. 6. AER update for 2012–2017 market data. The 2013 guideline values are taken from data up to December 2012.

⁷⁸ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 21 May 2018, p.33; Partington and Satchell, Report to the AER: Cost of Equity Issues 2016 Electricity and Gas Determinations, April 2016, pp.27-33; McKenzie & Partington, Report to the AER: Part A, Return on Equity, October 2014, pp.26-41.

⁷⁹ Previous DGM results have shown a change in growth rate can have an almost one for one inverse impact on the MRP estimate.



Figure 4 MRP estimates from different constructions of the DGM

Source: AER analysis. Variable growth rate equal to 10 year CGS yield but model construction is otherwise identical to the AER's DGM model, IPART February update

Overall, we consider that our HER estimate of 6.0 is consistent with the easing of risk conditions in Australia since 2013 as shown by the conditioning variables, and with our diminished confidence in the robustness of DGM estimates. A value of 6.0 per cent is more in line with market conditions and market practice.

Equity beta

Our draft decision is an equity beta point estimate of 0.6. This is a reduction from the estimate of 0.7 in our 2013 Guidelines.

We have come to this decision by applying the estimation approach that we set out in our 2013 Guidelines and applied in each subsequent regulatory determination. This approach is based on considering the same relevant classes of evidence that were identified in our 2013 Guidelines, namely:

- Giving most weight to empirical estimates of relevant Australian energy network businesses
- Having regard to:
 - conceptual considerations of the risks of energy network businesses relative to the market portfolio
 - o empirical estimates of foreign energy network businesses
 - o the theoretical underpinnings of the Black CAPM
 - o the value of stability and predictability to industry and consumers

Stakeholders supported continuing the approach of giving most weight to empirical estimates of Australian energy networks businesses, and having regard to the other

classes of evidence when determining a final equity beta estimate from within a range of estimates supported by the empirical analysis.⁸⁰ There was also agreement in support of continuing this approach in the expert session.⁸¹

However, stakeholders expressed divergent views on how we should exercise our judgement when evaluating the relative weight that should be afforded to each class of evidence and ultimately, on the final equity beta value we should choose.

Consumer groups submitted that we were too conservative in selecting a value of 0.7 previously and recommended the value should be 0.6 or lower.⁸² The CRG noted:

A point estimate for equity beta should be closer to 0.4 or lower.83

Service providers and the network shareholder group submitted that the equity beta should be at least 0.7 with short term estimates indicating an increase. The NSG submitted:

Equity beta can be maintained at 0.7 for the RORG period.84

Service providers referred to potential low beta bias in empirical estimates (by referencing ex-post studies) and submitted that this supports an uplift beyond the empirical estimates.^{85 86 87} The ENA submitted that:

There is no evidence to support a diminution of low beta bias or the role of the Black CAPM within the foundation model approach.⁸⁸

CCP16 disputed this proposition. It did not consider low beta bias to be particularly suitable for estimating ex-ante the equity beta or for 'adjustment' to the empirical data.⁸⁹ The CCP16 submitted that:

⁸⁰ Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, pp. 3–4; Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 68, 89.

⁸¹ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 23, 28; Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 39.

⁸² CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.68; CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 51.

⁸³ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 51.

⁸⁴ NSG, Submission on the Rate of Return Guideline review, p. 13

⁸⁵ For example, APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.

⁸⁶ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 27.

⁸⁷ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 52.

⁸⁸ ENA, Response to Discussion papers and concurrent expert evidence sessions, May 2018, p.41.

⁸⁹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

There is too much uncertainty around the empirical analysis of the Black CAPM theory for it to play a substantive role in the AER's decisions, and is not generally applied by market practitioners or regulators.⁹⁰

CCP16 also submitted that our empirical estimates overstate the true value of the equity beta because our sample includes firms that undertake unregulated activities that are inherently more risky.⁹¹

Service providers point to the declining size of our comparator set due to recent delistings, and urge caution before departing from our current value.⁹² They propose the inclusion of international energy firms and other Australian infrastructure firms in the comparator set⁹³ but the CCP16 opposes this because these firms are 'not particularly relevant to the [risk of providing regulated services]'.⁹⁴

We have regard to these submissions, the expert evidence session and other relevant material in this decision. We form a range then select a point estimate based on our consideration of the relevant evidence, their relative strengths and weaknesses and suitability for our regulatory task.

We rely mostly on empirical estimates because they provide information from firms that are reasonably comparable to firms in the supply of the regulated energy services. We have updated our empirical analysis⁹⁵ using the same comparator set as that used in our 2013 Guidelines. Results of this update are set out in

⁹⁰ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.9.

⁹¹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 37

⁹² ENA, Response to Discussion papers and concurrent expert evidence sessions, May 2018, p.46.

⁹³ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31; Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, p. 5; Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p. 3; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46, 62.

⁹⁴ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 82

⁹⁵ By including data up to 2 March 2018

Table 3 below. We consider they support a range of 0.4 to 0.8 as all estimates fall within the 0.4-0.8 range except for 1 estimate.

	Average of individual firm estimates					Fixed-we	ight portfoli	o estimates
Portfolio		P1	P2	P3	P4	P5	P6	P7
Firms in portfolio		APA, ENV	AAN, AGL, APA, ENV, GAS	APA, DUE, ENV, HDF, AST	APA, DUE, ENV, HDF, SKI, AST	APA, DUE, ENV, SKI, AST	APA, DUE, SKI, AST	APA, SKI, AST
Equal weighted be	eta estimates							
Longest available period	0.57	0.48	0.50	0.54	0.53	0.43	0.47	0.52
Longest period available (excl. tech boom & GFC)	0.61	0.52	0.51	0.59	0.58	0.50	0.54	0.63
Recent 5 years	0.70	0.71				0.55	0.66	0.79
Value weighted be	eta estimates							
Longest available period ^(a)	n/a	0.52	0.66	0.47	0.47	0.44	0.49	0.54
Longest period available (excl. tech boom & GFC)	n/a	0.56	0.67	0.55	0.55	0.52	0.57	0.66
Recent 5 years	n/a	0.73				0.53	0.72	0.85

Table 3 Re-levered weekly OLS equity beta estimates

Source: AER analysis; Bloomberg

Note: Our comparator firms include AusNet Services (AST). This firm was included in the 2013 Guideline under its former name of SP Ausnet (SPN). It was renamed in 2014.

While we acknowledge the small sample, we consider our comparator set of domestic firms is the best empirical guide currently available. This is because international firms and other Australian infrastructure firms carry different risks and characteristics compared to firms in the supply of regulated energy network services. A range of submissions acknowledged that it would be difficult to adjust international firms and other Australian infrastructure firms to make them comparable to firms supplying the regulated energy services. ^{96 97 98 99 100 101} Experts and the ENA have also agreed that domestic comparators are the most relevant and comparable. ^{102 103 104}

⁹⁶ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 19

Overall, we consider that an equity beta value of 0.6 will best promote the legislative objectives for the following reasons:

• We give most weight to estimates from the longest estimation period (portfolio level and average of firm level estimates). We consider that short term estimates can be influenced by factors such as one-off events (for example, the Global Financial Crisis), shocks and interest rate movements. These factors can obscure the systematic risk of a firm supplying regulated energy services whose exposure is mitigated by regulation and monopoly nature of the service it provides. Estimates from the longest estimation period cluster in the 0.5–0.6 range (see Figure 25) and the average¹⁰⁵ of re-levered OLS estimates is 0.51.

Figure 5 Distribution of 2018 re-levered weekly beta longest estimation period¹⁰⁶



Source: AER analysis, Bloomberg

- ⁹⁷ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70
- ⁹⁸ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 33, 29, 28
- ⁹⁹ Cambridge Economic Policy Associates, Rate of Return Guideline Review Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 46.
- ¹⁰⁰ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 82
- ¹⁰¹ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.
- ¹⁰² AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 28
- ¹⁰³ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 23
- ¹⁰⁴ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62.
- ¹⁰⁵ Average of fixed weight portfolio estimates and averaged firm-level estimates.
- ¹⁰⁶ Based on value weighted and equal weighted portfolio estimates and averaged firm estimates.

- The average¹⁰⁷ of weekly re-levered OLS estimates across all periods is 0.57.
- The point estimate of 0.6 (and empirical range of 0.4–0.8) is consistent with our expectation that the equity beta of a firm supplying the regulated energy services will be below 1.0 due to the overall low risk exposure.
- International estimates support an equity beta value of less than 1.0.¹⁰⁸
- We have had regard to (and further consideration of) the Black CAPM and the
 potential for low beta bias. However, we agree with the CCP16 that there is
 uncertainty around empirical analysis of the Black CAPM and it is not generally
 applied by market practitioners or regulators. Given our diminished confidence, we
 are not persuaded to select an equity beta towards the top of the observed
 empirical estimates.

We observe that recent short-term beta estimates have increased since our 2013 analysis:¹⁰⁹

- The average of firm level estimates have increased since Henry's report with the largest increase for the recent 5 year period (0.459 to 0.70).
- Most portfolio-level estimates rose with the increase being less than 0.05.

However, the overall empirical results (

¹⁰⁷ Average of fixed weight portfolio estimates and averaged firm-level estimates across all three estimation periods (longest, post tech boom excluding GFC and recent 5 years)

¹⁰⁸ The multitude of differences with a supplier of the regulated energy services means that we cannot (reliably) quantify and adjust international estimates to make them comparable to domestic estimates. As a result, they are now used in a qualitative role (similar to conceptual analysis) to indicate if they support a value above or below the market average.

¹⁰⁹ While Professor Henry's report was finalised in 2014, estimates were provided to the AER during 2013 to inform the Rate of Return Guideline review. For example, see: AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, pp. 64–67.

Table 3), particularly the longest estimation period (as seen in Figure 25), support a value of less than 0.7.

We agree with consumer submissions that the inclusion of comparator firms with a high proportion of unregulated activities is likely to result in empirical estimates that are higher than otherwise. This is because the higher level of unregulated activities would increase systematic risk exposure. Of the firms in our sample that are still publicly listed, SKI and AST have a relatively high proportion of regulated revenue while APA has a relatively low proportion. Table 4 compares estimates for SKI and AST against those from the whole comparator set with the following results:

- For the longest estimation period (which we have most regard to), SKI and AST's average firm level estimate is 0.41 and average portfolio estimate of 0.42. These are below estimates from the whole comparator set of 0.57 and 0.5 respectively.
- The average of OLS portfolio estimates and averaged firm estimates for SKI and AST is 0.54 which is below that from the whole comparator set (0.57)

Table 4 Comparison of estimates for listed majority regulatedcomparators (OLS, weekly)

	Whole comparator set	SKI & AST
Average of firm level estimates		
Longest	0.57	0.41
Post tech boom & excl. GFC	0.61	0.52
Recent 5 years	0.70	0.68
Fixed weight portfolio estimates		
Longest	0.43–0.66	0.42–0.43
Post tech boom & excl. GFC	0.50-0.67	0.52–0.53
Recent 5 years	0.53–0.85	0.68–0.7

Source: AER analysis, Bloomberg

Therefore, we consider the above results support an equity beta of less than 0.7 and indicates potential for a value less than that indicated from the whole comparator set. However, we note the 2013 Guideline estimated a reduced equity beta. The 2013 Guidelines estimated an equity beta of 0.7, down from 0.8 estimated in previous determinations, even though longer term estimates were clustered materially below 0.7. In part, we took into account the need to promote stability and predictability and therefore decided not to make a larger change. We adopt similar considerations in this decision and consider a further reduction at this time will provide an estimate that is more in line with empirical estimates and so we have reduced the equity beta by 0.1.

We have considered whether we should employ a different value between gas and electricity businesses. Gas businesses submitted that they are subject to greater risk.

Overall, we conclude that systematic risks between gas and electricity networks are sufficiently similar to warrant a common equity beta.

Return on equity point estimate

Under the 'Return on equity' heading above we outlined our 6-step foundation model approach to estimating the return on equity. Steps 4-6 of this approach are where we bring together the information from step 3, have regard to other information (step 4), and then determine the final return on our equity point estimate (sometimes referred to as 'cross checks').

Our evaluation of the other information that plays a role in informing the overall return on equity (set out in table table) leads us to consider that on the whole they support our foundation model estimate of an equity risk premium (ERP) of 360 basis points. We consider the debt risk premium (DRP)¹¹⁰ to be a valuable relative indicator of the reasonableness of our ERP. This is because we do not expect the return on equity for a benchmark efficient entity with similar degree of risk as a relevant service provider in the provision of regulated energy services to be significantly higher than the return on debt. Figure 6 shows the comparative and relative positions the equity and debt risk premia.

When we finalised our 2013 Guideline the debt risk premium was about 340 basis points (based on the simple average of the RBA and BVAL curves). There has been a material reduction in the DRP since and it is currently around 190 basis points.¹¹¹ In 2013 our ERP was approximately 115 basis points above the DRP. Comparatively, our ERP for this decision is approximately 170 basis points above the DRP. This gives us confidence that service providers, relative to the margin at the start of the 2013 Guideline, have a reasonable opportunity to recover at least their efficient costs of equity over the life of the 2018 Guideline.

¹¹⁰ Spread between BBB+ rated corporate debt and the risk free rate.

¹¹¹ As at 29 March 2018.





Source: AER analysis. The 2013 Guideline estimated the return on debt (and DRP) as a simple average of BBB bond yields from RBA and Bloomberg BVAL data. Updated to 29 March 2018.

We consider that the output of an ERP of 360 basis points from our foundation model SLCAPM using relevant market data is an efficient forward-looking return for risk. Our consideration of 'cross checks' does not lead us to adjust the SLCAPM output. We are satisfied that the ERP of 360 basis points coupled with a risk free rate observed at the time of applying the guideline will achieve the legislative objectives.

Overall approach to estimating the return on debt

Our draft decision is to continue to adopt key elements of our current approach for estimating the return on debt. That is, we will continue to adopt:

- A benchmarking approach based on data from third party data providers and benchmarks for term of debt and credit rating.
- A 10-year trailing average approach with an annual update.
- A 10-year transition from the previous 'on-the-day' to the 10-year trailing average approach. For clarity, our draft decision is to adopt a consistent transition approach across all networks. Where we have commenced the transition in a previous determination for a service provider, we will continue that transition.

These aspects of the approach to estimating debt continue our current approach, which has been affirmed by the Australian Competition Tribunal.¹¹²

The majority of stakeholders also continue to support the adoption of a trailing average approach with a transitioning arrangement. ¹¹³

In contrast, the Network Shareholder Group, ATCO Gas Australia, and NT Power & Water Corporation have indicated preference for an immediate adoption of the trailing average, rather than with transitioning.

Having regard to the evidence before us, the key reasons for our decision are that:

- our reasons for use of a 10 year-trailing average and a 10-year transition to that trailing average have been scrutinised in detail and applied in all decisions made since the current 2013 guidelines, and upheld in appeals before the Australian Competition Tribunal and the Full Federal Court. In our view, the reasons for adopting this combination of approaches remain valid.
- maintaining our current approach provides the necessary certainty and predictability of regulatory arrangements that stakeholders have said they value whilst allowing us to conduct our regulatory task in a manner that is most likely to contribute to the legislative objectives.
- removing the transitioning arrangement will not achieve the national gas and electricity objectives. If we do not employ the transition then the change in methodology would not be revenue neutral and would generate windfall gains or losses depending on how interest rates have moved.

The choice of third party data providers

Through our 2013 guideline and subsequent determinations we decided to source debt data from the RBA and Bloomberg. Since then we have become aware of two additional data sources: Thomson Reuters and S&P Global. For our 2018 guideline our decision is to continue to source data from RBA and Bloomberg and to also source data from Thomson Reuters. We will not use data from S&P Global at this point in time.

Based on evaluation of available information on the curve methodologies, all four curves have strengths and weaknesses and none is clearly superior with respect to either the bond selection criteria or curve fitting methodology. Overall, there is a substantial overlap between the curves in terms of bond selection criteria, though each curve has distinctive characteristics.

¹¹² See: Australian Competition Tribunal, Application by ActewAGL Distribution [2017] ACompT 2, October 2017.

¹¹³ Australian Pipelines and Gas Association, Submission to the Issues Paper, December 2017, p4 – 6; Energy Networks Australia, AER Rate of Return Guideline, December 2017, p16-17, p19-20; Ergon Energy and Energex, Issues paper – review of the rate of return guidelines, December 2017, p4-5; Major Energy Users, Review of the rate of return guidelines, December 2017, p10-11, 15; APA, APA submission responding to AER issues paper, December 2017, p8; Cheung Kong Infrastructure, Submission on rate of return issues paper, December 2017, p3,

In general, our view is that consideration of data providers' methodologies is the most important criteria on which to base our decision on the choice of data series. However, in this case, the S&P Global's Australian-dollar-denominated curves produce outcomes which are materially different to the other curve providers and to our expectations: ¹¹⁴

- Over the data series we have available, the S&P Global broad-A and broad-BBB curves produce very similar results where we would expect a more material difference. In contrast, the BVAL, RBA and Thomson Reuters curves as well as S&P Global's US-dollar-denominated curves exhibit a more material difference.
- For the majority of the period since December 2013, the S&P Global Australiandollar-denominated broad-BBB yield curve produce yields estimates below the 'A' rated curves from the other curve providers.

We recognise that there may be valid drivers of the differences between curve estimates. However, disaggregation of the drivers of these differences is complex due to the proprietary nature of curve estimation and we have not been able to reconcile the differences at this time.

Stakeholder submissions generally supported continued reliance on the Bloomberg and RBA curves but there were mixed views on:

- Whether to include the Thomson Reuters or S&P Global curves—some consumer groups supported inclusion of the new curves, networks and investors submitted that we should maintain the current approach
- What weight to put on the curves—for example, some consumer groups proposed we should give greater weight to the RBA curve compared to the other curves.

Having regard to the available evidence, we consider none of the three curve providers is clearly superior. With respect to the fitness of purpose of the four curves for this application, we agree with the MEU's summary that:¹¹⁵

- While the AER seeks views on the appropriateness of the four different sources of data, it also highlights that all four series have failings to a greater or lesser extent.
- This implies that to overcome these shortcomings, some degree of combination of the data series is not only appropriate but necessary.

Our view is that the combined use of the RBA, Bloomberg and Thomson Reuters curves will contribute to achievement of the NEO and NGO to the greatest degree. Our key reasons for this view are:

• On the bond selection criteria (including approach for identifying outliers) and curve fitting (or averaging) methodologies, we consider that the approaches employed by

¹¹⁴ To assist in our analysis, S&P Global has kindly provided us with a longer historical time series of monthly data than is currently publicly available. We have had regard to this analysis in reaching our decision.

¹¹⁵ Major Energy Users, *Estimating the allowed return on debt—Discussion paper*, May 2018, p. 4.

the RBA, Bloomberg and Thomson Reuters have their unique strengths and weaknesses, but we are not satisfied that any curve is clearly superior.

- All of the curves from all three of the data providers require adjustment from their published form to make them fit for purpose. We are not satisfied that one can be more simply or reliably adjusted to estimate the annual return on debt than another.
- In our view, applying equal weight to each of the three data providers is intuitively
 reasonable and the process of developing a more sophisticated weighting scheme
 would rely on contentious assumptions. In our view, there is no persuasive
 evidence that the likely difference in average from different weighting schemes will
 be material over time.
- An average of the three data providers reduces the impact of movements in any one of the individual curves. This will reduce potential volatility. Further, the use of three data providers incorporates a natural contingency in the event that one of the data providers ceases publication.

The benchmark credit rating

Our draft decision is to adopt a benchmark credit rating of BBB+, consistent with our 2013 Guidelines and 2009 review of WACC parameters. We consider this is consistent with the available empirical evidence.

Stakeholders submitted a range of views about benchmark credit rating. Service providers and investors generally supported use of a BBB+ benchmark credit rating. The ENA supported the use of a BBB+ benchmark credit rating but submitted this would be at the lower end of the range of observed credit ratings. Some customers and retailers submitted we should adopt a higher (A- or A) credit rating.

We note that the service providers within the sample we analysed have generally maintained stable credit ratings over an extended period including the period affected by the GFC and maintained investment grade credit ratings (between BBB– and A–). The table below shows the historical median credit rating for Australian service providers from 2006 to 2017.

Issuer	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017
Industry	BBB/	BBB/		DDD	DDD	DDD	DDD	BBB/				
median	BBB+	BBB+	DDD+	DDD	DDD	DDD	DDD	BBB+	DDD+	DDD+	DDD+	DDD+

Table 5 Median credit ratings over time

Source: Bloomberg (S&P Global), AER analysis

Whilst the above table shows that the median credit rating has moved between BBB and BBB+, the four most recent years of data support a rating of BBB+. We consider that this recent concentration of ratings at BBB+ provides sufficient evidence that this is the appropriate benchmark credit rating.

Implementation of the benchmark credit rating

We will implement the BBB+ benchmark credit rating using a weighted average of:

- 2/3 weighting on broad-BBB curve estimates
- 1/3 weighting on broad-A curve estimates.

This is a departure from our current approach where we have relied on a broad-BBB curve only.

Ideally, we would estimate the return on debt for a benchmark BBB+ rated entity using third party yield curves for specific BBB+ rated bonds. However, due to the relatively small sample of Australian dollar (AUD)-denominated debt from which third party curve providers can develop their estimates, we rely on 'broad' credit rating band estimates which include several credit rating bands within a curve:

- A 'broad-BBB' curve typically includes BBB-, BBB and BBB+ rated debt
- A 'broad-A' curve typically includes A-, A and A+ rated debt.

The CRG summarised this issue as follows:116

It is clear that using a broad-BBB series will over-estimate the allowed return on debt that will be provided to a business with a BBB+ credit rating. This becomes even more egregious when it is realised that the relationship between credit rating and interest rate is not exact and that regulated networks benefit from slightly lower rates than other BBB+ businesses.

Using a broad-A series and averaging it with the broad-B series is one useful approach, and the CRG would support this approach.

Consumer and other stakeholders submitted a range of different views on this proposed approach:

- CCP16 supported the adoption of a 2/3 broad-BBB: 1/3 broad-A average
- The MEU submitted that the benchmark credit rating should be raised from BBB+ to A- or A which would require a different curve mix.
- Energy Australia recommended a 'tiered' benchmark credit rating of A- and BBB.
- While the CRG submission supported the approach, it observed that the proposed 2/3:1/3 average only partially closes the gap between AER approach and actual spreads.

In contrast, networks and investors:

Submitted that the evidence does not support a change from the use of broad-BBB curves only

¹¹⁶ Consumer reference group, Estimating the allowed return on debt—Response to discussion paper, May 2018, pp. 3–4.

- Raised a series of methodological concerns with our analysis of actual debt instruments raised by service providers and with our 'matched-term spreads' analysis
- Submitted that long-dated debt that make up Bloomberg's broad BBB curve has historically been more heavily weighted towards BBB+ bonds

Having regard to the submissions and the evidence available to us, we consider a 2/3 broad-BBB: 1/3 broad-A average will better estimate the required return on debt for a BBB+ rated entity.

Our key reasons for this view are as follows. Firstly, we consider the use of a 'broad-BBB' series alone will, other things held constant, overestimate the return on debt required for a BBB+ rated entity. In regulatory determinations made after the 2013 Guidelines we acknowledged that reliance on a broad-BBB curve only is likely to overestimate the yields for a BBB+ benchmark.¹¹⁷ This is because, to the extent that credit ratings are an informative measure of credit risk, we expect:

- reliance on a broad-BBB curve is likely to overestimate the level of credit risk (and ultimately the required yields) of a BBB+ benchmark credit rating- because the benchmark credit rating (BBB+) is the highest rating band amongst the constituents, the inclusion of any of the lower rated bonds in the sample (BBB or BBB-) would, other things held constant, overestimate the required return on debt for the benchmark credit rating
- reliance on a broad-A curve only would underestimate the level of credit risk (and ultimately the required yields) for a BBB+ benchmark credit rating because all constituents (A-, A, A+) are higher rated than the BBB+ benchmark credit rating
- some combination of broad-BBB and broad-A curves should therefore provide the best fit to a BBB+ benchmark credit rating. As a conceptual expectation, our view is that a 2/3 broad-BBB: 1/3 broad A rating is most likely to match a BBB+ benchmark credit rating.

Secondly, our analysis of actual debt instruments raised by service providers compared to our current approach suggests that:

- When term and date of issuance are controlled, the use of broad-BBB curves has, over 2013–17, overestimated by approximately 29 basis points the spreads at which service providers have issued debt
- When term and date of issuance are controlled, a weighted average of 2/3 broad-BBB : 1/3-broad A curves has, over 2013–17, overestimated by approximately 9 basis points the spreads at which service providers have issued debt

We therefore conclude, that a 2/3 broad-BBB : 1/3 broad-A estimate is a better match for our benchmark credit rating of BBB+. This is supported conceptually and by our

¹¹⁷ See for example: AER, Final determination— AusNet Services transmission determination 2017-2022— Attachment 3: Rate of return, April 2017, p. 340

analysis of debt issuances over the past 5 years. Based on current market observations using a combination of broad A and BBB curves from Bloomberg, the RBA and Thomson Reuters will reduce our estimate of the benchmark return on debt by roughly 10 basis points compared to using a broad BBB curve only.¹¹⁸

Benchmark term

Our benchmark term of debt for these Guidelines is 10 years. This is consistent with the benchmark term of debt in the 2013 Guidelines and the 2009 statement of regulatory intent.

In reaching our view on the benchmark term of debt, we have had regard to information on actual debt instruments collected from the majority of privately owned service providers that we regulate. This information is commercially sensitive, and we engaged Chairmont to assist us with aggregation of this data. We have held individual discussions in confidence with service providers who provided data, and these discussions have informed our conclusions.

We published a discussion paper setting out the results of our analysis and Chairmont's report. Our analysis of debt instruments issued over the past 5 years yielded an average term at issuance of 7.5 years for all instruments included in our sample. Figure 7 illustrates a series of term at issuance where each point in time is an average of debt within the EICSI sample that has been issued in the preceding 12 months, and its relationships with the average spreads for the same instruments.

Figure 7 Spreads to swap on actual debt instruments in the EICSI sample compared to spreads to swap using the AER approach



¹¹⁸ Specifically, we have calculated this estimate over all business days in February 2018.

Source: AER analysis, Chairmont spreadsheet.

Note: This figure uses Chairmont's initial EICSI spreadsheet, but the AER series is updated to include the RBA's June 2018 revisions to its F3 statistical table.

CCP16 noted the evidence that networks on average are issuing debt with a shorter term than 10 years, but did not recommend a change to our 10 year term:¹¹⁹

Clearly, there is a strong argument that the adoption of a benchmark term of 10 years contributes to a conservative (i.e. overestimate) ROD. The question of what to do about this is more complex and difficult. The primary difficulty is the uncertainty and complexity of changing the term while most utilities are in the process of adjusting their debt portfolios as part of transitioning to the trailing average of 10-years. Given the extensive analysis and debate (and administrative and judicial reviews) we would not propose that the AER change its current approach to better approximate a benchmark term based on the actual behaviour of the NSPs as part of the new Guideline – although we would not rule it out as a suitable approach beyond the transition period.

In response to our discussion paper, networks and investors submitted a series of reasons in support of maintaining the 10-year term, including:

- that the 10-year transition to the trailing average return on debt implies a level of regulatory commitment to the approach over a number of regulatory periods
- that there remains a conceptual basis to expect networks to issue longer term debt to match the lives of their assets
- a series of contextual factors such as implementation of the transition and the privatisation of networks within the sample suggesting that the past 5 year sample may not reflect 'business as usual' with respect to the term of debt issuance.
- the use of a simple average of term on issued debt over 2013–17 is not sufficiently robust to support a change of term.

Consumers and retailers submitted a range of different views:

- CCP16 and the CRG pointed to evidence of shorter term issuance over the past 5 years but did not recommend a departure from a 10 year benchmark due to the uncertainty and complexity of changing the term while utilities are in the process of responding to the transition to a trailing average approach.
- Some consumer groups submitted that even a 7.5 year benchmark term to maturity may overstate the required return on debt
- Some consumer groups recommended that our return on debt should be based on the average observed spreads of debt in our sample. This would not require an estimate of the benchmark term.

¹¹⁹ CCP sub-panel 16, Submission to the AER on its allowed rate of return on debt discussion paper, May 2018,

Energy Australia recommended the reduction of the benchmark term of debt to 7.5 years

Having regard to these submissions and the evidence available to us, we consider the benchmark 10 year term remains appropriate. Our key reasons for this view are that:

- We consider that a business will, within the constraints of the market for corporate bonds, aim to match the length of the debt term to the asset life in order to minimise refinancing risk. We note, however, that this is subject to consideration of the increased cost of debt associated with a longer term.
- Consideration of service providers' actual debt raising practices and relevant market circumstances over 2013–17 does not reveal clear conclusions:
 - Over the period for which we have collected actual debt data (2013-17) we have implemented a transition to the trailing average return on debt approach. This is was a material change to the return on debt approach, and we expect it would have impacted debt raising practices to some extent. Based on the data available to us, it is unclear whether or not the observed debt issuance patterns are temporary / cyclical or a transient adjustment in response to our transition to a trailing average approach.
 - A simple average estimate of terms at issuance within the sample of collected actual debt data suggests an average term of 7.4 years. However, we agree with the view expressed by service providers that a simple average across instruments in the sample may understate the 'true' benchmark term of debt.

Nonetheless, we consider empirical evidence to inform the benchmark term of debt is important. As the transition progresses, some of these complexities and uncertainties in the current data may resolve. Accordingly, we will continue to collect actual return on debt information for consideration in future guideline reviews.

Value of imputation credits

Our estimate of the value of imputation credits for this draft instrument is 0.5 selected from a range of 0.3 to 0.6.

Our estimate of 0.5 is rounded to one decimal place from an estimate of 0.53.

The 0.53 estimate is based on the product of an utilisation rate of 0.6 (the proportion of imputation credits paid out by the benchmark efficient entity (BEE) we estimated that would be utilised) multiplied by a distribution rate of 0.88 (our estimate of the proportion of imputation credits created by the BEE that would be distributed).¹²⁰

Our approach, including rounding to one decimal place, is consistent with the approach set out in the 2013 Guideline.

¹²⁰ The term 'distribution rate' is used interchangeably with 'payout ratio' in AER and other parties different documents.

We have carefully considered all submissions and information before us in reaching our draft instrument value of 0.50.

Our 'utilisation' approach to the 'Value' of the imputation credits

In Australia, we operate under an imputation tax system. This means that shareholders can receive an imputation credit for income tax paid at the company level. For eligible shareholders, this credit can either offset their Australian income tax liabilities or be used to obtain a refund from the tax office. Thus, shareholders in Australia factor in three income streams when considering investments: capital gains; dividends; and imputation credits. The National Electricity Rules/National Gas Rules (NER/NGR) recognises that a service provider's allowed revenue does not need to include the value of imputation credits. We reflect this third income stream by discounting the tax allowance we provide to service providers by a factor represented by the Greek letter, γ , 'gamma'. Our regulatory framework employs a vanilla WACC which is a prepersonal costs, pre-personal tax model based on the work of Officer.¹²¹

We interpret the value of imputation credits as an estimate of the proportion of company tax which is expected to be returned to investors through utilisation of imputation credits. That is, we apply a 'utilisation' approach to estimating the post company tax value of imputation credits. This is consistent with our approach for estimating the 'value' of imputation credits proposed in the 2013 Guideline and applied in all regulatory determinations since then.

Our current 'utilisation' approach to determining the value of imputation credits has been found open to us in four recent legal cases including two Full Federal Court cases.¹²²

The utilisation approach to valuing imputation credits has also been extensively considered in submissions we have received and in the discussions at the expert session. Consumers generally support the approach. For example, CCP16 generally supported the overall conceptual framework in which the AER assessed the value of imputation credits (gamma). CCP16 contended that the decision of the Full Federal Court in May 2017 settled the question on the 'value of imputation credits'.¹²³ While industry generally does not agree with the AER's 'utilisation' approach, they appear to accept this approach is legally open to the AER and the AER is likely to use it in making this Guideline. The Joint Energy Networks submitted that the utilisation rate approach to gamma is not being challenged as part of this Guideline review.¹²⁴

¹²¹ R. Officer, *The cost of capital of a company under an imputation tax system*', Accounting and finance, May 1994, vol. 34(1), p. 4

¹²² Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017; Federal Court of Australia, SA Power Networks v Australian Competition Tribunal (No 2) [2018] FCAFC 3, Jan 2018, para. 56; Australian Competition Tribunal, Application by SA Power Networks [2016] ACompT 11, 28 October 2016; Australian Competition Tribunal, Application by ActewAGL Distribution [2017] ACompT 2, 17 October 2017

¹²³ Consumer Challenge Panel 16 - Submission on Rate of Return Guideline Review - 4 May 2018. page 120

¹²⁴ Joint Energy Networks - Submission on Rate of Return Guideline Review - 4 May 2018 page 6

Key reasons for choosing 0.5 in this draft decision

A value of 0.5 is an increase from the value of 0.4 applied in all determinations released since the 2013 Guideline. The increase is driven by updated information and our exercise of judgement in light of this information. Key drivers of the increase include:

- Giving primary weight to Lally's updated estimate of the distribution rate from the top 20 Australian Stock Exchange (ASX) listed firms
- Placing significant reliance upon an estimate of the utilisation rate for 'all equity' (that is data reflecting the ownership of both listed and unlisted Australian equity) from updated ABS equity ownership data

While most stakeholders have accepted our current 'utilisation' interpretation of the value of imputation credits, some stakeholders proposed that the AER should review its overall approach. In particular, the CRG proposed criteria we should employ including: ¹²⁵

An assumption that the utilisation rate of imputation credits (Θ) is 100 per cent. That is, the firms are using the most efficient source of finance, that being Australian investors entitled to make use of imputation credits,

A distribution rate in line with what an efficiently financed form would be expected to distribute, based on the value of the RAB, depreciation and any necessary new investment in the RAB.

A number of stakeholders also proposed that the utilisation rate from the equity ownership approach which uses the ABS equity ownership data is not consistent with the capital asset pricing model (CAPM). Nevertheless, the key issues raised have been around the composition of the empirical evidence and the relative weights to be placed on the different empirical evidence in coming to an estimate for the value of imputation credits.

Some service providers have proposed we use a simple overall national gamma estimate based on ATO data. The Joint Energy Networks submitted:¹²⁶

ATO tax statistics provide a direct estimate of the proportion of company tax paid by the average firm that is returned to its investors through utilization of imputation credits. The items in the ATO data base required for this estimate are reliable and robust and give rise to an estimate for gamma of 0.34.

However, we do not consider an estimate based on this ATO data supports a value of less than 0.5. This is because it reflects a national average imputation credit distribution rate which we consider is below the efficient distribution rate. Once adjusted for the difference in the distribution rate the ATO data supports a gamma

¹²⁵ Consumer Reference Group - Submission on Rate of Return Guideline Review - 4 May 2018 page 59

¹²⁶ Joint Energy Networks - Submission on Rate of Return Guideline Review - 4 May 2018 page 6.

value of 0.5. We also note the ATO has advised us the taxation statistics data should be treated with caution.

The key reasons for choosing an utilisation rate of 0.6

The utilisation rate is the proportion of distributed imputation credits utilised and it is reasonable to expect the majority of eligible shareholders will utilise imputation credits. Given this, the key data we use for estimating the utilisation rate is ABS equity Australian ownership data for 'all equity'. This is also consistent with theoretical models, that the value for the utilisation rate is consistent across firms in the Australian economy.

ABS equity ownership data indicates Australian ownership of Australian equities of somewhere between 60 and 70 percent over recent years and a current Australian ownership rate of around 65 percent. This supports our estimated utilisation rate of 60 per cent given we expect nearly all eligible shareholders to utilise imputation credits distributed to them.

In addition, after adjusting for the higher expected distribution of the BEE than the average corporate firm in Australia, the ATO taxations statistics data also appears to support an utilisation rate of around 60 per cent.

Distribution rate / payout ratio

We have placed primary weight on the estimated distribution rate from the top 20 ASX listed firms from 2001 to end 2017. This has been estimated from audited financial reports by Martin Lally and the analysis has largely been replicated by AER staff. We consider a distribution rate from large listed firms in Australia an appropriate benchmark distribution rate. Considering listed firms is also consistent with our equity beta estimates we place primary weight on taken from Australian listed firms. We consider the top 20 ASX listed firms provide a more accurate picture of a benchmark efficient entity relative to the alternative, as proposed by some, of using a national economy wide measure. Lally also advised us that as most of the regulated firms are either listed or owned by listed firms, a distribution rate from listed equity would be appropriate for the benchmark efficient entity. In comparison to an estimate of the distribution rate from the top 20 ASX listed firms, an economy wide measure inclusive of unlisted firms is likely to have a materially lower average distribution rate due to owners of unlisted firms seeking to avoid taxation at high marginal (personal) tax rates through not distributing earnings.

The estimated distribution rate of the top 20 ASX listed firms from 2000 to 2017 is 0.88. This supports a distribution rate for the BEE of at least 0.88. We consider an efficient service provider¹²⁷ could be expected to be able to maintain a distribution rate of 88 per cent if it elected to do so. We have used 0.83 to be internally consistent with our rounded gamma value of 0.5 and our utilisation rate of 0.60.

Our approach to estimating the distribution rate is a change relative to determinations made since the 2013 Guideline, where we primarily used estimates of the distribution rate based on ATO franking account balance data. We also considered distribution rates from both all equity and only listed equity in these determinations. We now consider an estimate of the distribution rate for listed equity a better benchmark than one for all equity. We also consider audited data from financial reports of the top 20 ASX listed firms superior than estimates based on ATO Franking account balance (FAB) data for listed equity.

The ENA submitted that issues with ATO FAB data identified by the ATO materially impact Lally's estimate of the top 20 ASX firms. This is not correct for several reasons. Firstly, as the top 20 ASX firms Lally examines are constant through time his analysis does not suffer from the material entry and exit problems associated with the use of the ATO FAB data over time (where firms may liquidate for example). Secondly, the data Lally uses is from audited financial accounts and therefore should not suffer from the same potential reliability issues associated with ATO informational reporting data (ATO aggregate FAB data is based on information data not used to calculated tax owing). The raw audited data from financial reports also has the advantage of being largely publicly available and therefore the estimate is replicable and transparent.

¹²⁷ With a similar degree of risk as that which is involved in providing regulated services.

The CCP and Major Energy Users submitted that the actual tax paid by networks is far under the amount of tax assumed by the AER and a value of imputation credits of 0.4 is too conservative and needs to be increased.

Overall, having considered all submissions and other information before us, we consider the most recent and best available data supports an increase in our estimate of the distribution rate to 0.88. However, as noted above, we have used an estimate of 0.83 to be consistent with a rounded value for gamma of 0.50.

2 How the guidelines contribute to the legislative objectives

There are three key limbs to the legislative objectives that set out how we must decide on the allowed rate of return:

- The national gas and electricity objectives
- The revenue and pricing principles, and
- The allowed rate of return objective.

In performing or exercising an economic regulatory function or power, we must do so in a manner that will or is likely to contribute to the national gas and electricity objectives.¹²⁸ The National Electricity Objective (NEO) and the National Gas Objective (NGO) establish the ultimate objective of our decision-making.¹²⁹

In support of the NEO and NGO, the National Electricity Law (NEL) and National Gas Law (NGL) set out Revenue and Pricing Principles (RPPs).¹³⁰ These principles underlie the achievement of the NEO and NGO and we have had particular regard to these principles, set out in

¹²⁸ NEL, s. 16(1)(a); NGL, s. 23.

¹²⁹ NEL, s. 7; NGL, s. 23.

¹³⁰ NEL, s. 7A; NGL, s. 24.

Table 6 below, in making our Draft Decision.

In addition to meeting the legislative obligations to achieve the NEO or NGO, and having regard to the RPPs, the National Gas Rules and National Electricity Rules also establish an allowed rate of return objective.¹³¹ That objective provides that the allowed rate of return is to be commensurate with the efficient financing costs of a benchmark efficient entity with a similar degree of risk as that which applies to the service provider in respect of the provision of its regulated services.¹³² This objective needs to be interpreted consistently with the overall NEO and NGO, and the principles set out in the RPPs.

There are certain common repeated concepts within these legislative objectives and principles that are particularly relevant to setting the rate of return and the value of imputation credits:

- Benchmarking and incentive regulation
- Efficiency, which we consider relates to two important concepts:
 - The NPV=0 condition
 - Market cost of capital
- Return commensurate with risk of providing regulated services

Section 2.1 sets out how the legislative objectives establish these common repeated concepts for the rate of return.

We adopt standard, well established regulatory economic approaches to our understanding of each these concepts.¹³³ Sections 2.2 sets out our consideration of these concepts for the allowed rate of return. Section 2.4 provides further detailed consideration of risk and return, and how our consideration of this concept shapes our determination of the allowed rate of return and value of imputation credits.

2.1 Key concepts in the legislative objectives

In this section we provide further consideration of the:

- national gas and electricity objectives,
- revenue and pricing principles, and
- allowed rate of return objective

in respect of the allowed rate of return.

2.1.1 National gas and electricity objectives

¹³¹ NER cll. 6.5.2(b), 6.5.2(c), 6A.6.2(b), 6A.6.2(c); NGR r. 87(2) and r.87(3).

¹³² The allowed rate of return objective is discussed in more detail in 2.1.3.

¹³³ See AER, *Risk and judgement Discussion paper*, February 2018.

The national gas objective and the electricity objective are broadly comparable, and they state:

National gas objective

The objective of this Law is to promote efficient investment in, and efficient operation and use of, natural gas services for the long term interests of consumers of natural gas with respect to price, quality, safety, reliability and security of supply of natural gas;

National electricity objective

The objective of this Law is to promote efficient investment in, and efficient operation and use of, electricity services for the long term interests of consumers of electricity with respect to—

- (a) price, quality, safety, reliability and security of supply of electricity; and
- (b) the reliability, safety and security of the national electricity system.

We achieve the national gas and electricity objectives through the combined application of all of our regulatory mechanisms, among which the rate of return is an important aspect. The rate of return provides a service provider with a return on capital that is sufficient for the service provider to efficiently raise the capital required to fund efficient investment in its network. The focus of the rate of return is therefore in promoting efficient investment in the network used to provide electricity and natural gas services.

Benchmarking and incentive regulation are important drivers of efficient investment, operation and use of the network. Our regulatory framework employs a benchmark or incentive regulation approach to determining the allowed rate of return contributes our ex ante determination of revenue allowances.

The combination of our expenditure incentive schemes and our service performance incentive scheme provide balanced incentives for efficiency with respect to price and quality of service (including safety, reliability, and security of supply).

Therefore, we consider that the key concept in the national gas and electricity objectives for the allowed rate of return is efficiency.

2.1.2 Revenue and pricing principles

There are six revenue and pricing principles, which are discussed in the table below.

Table 6 Revenue and pricing principles in the NEL and NGL

Revenue and pricing principle	AER consideration				
A service provider should be provided with a reasonable opportunity to recover at least the efficient costs the service provider incurs in:	We consider that a reasonable opportunity to recover efficient costs of providing regulated services is achieved when the rate of return satisfies the 'NPV=0' condition. The NPV=0 condition means that the ex-ante expectation is that over the life of an investment the expected cash flow from the investment meets all the operating expenditure and corporate taxes, repays the capital invested and there is just enough cash flow left over to cover investors' required return on the capital invested.				
 providing regulated services; and complying with a regulatory obligation or requirement 	We consider that the efficient cost of capital is reflected in market rates.				
or making a regulatory payment	We consider that benchmarking and incentive regulation provides appropriate incentives for efficient costs.				
	We note that this principle refers to the efficient costs of providing regulated services, and that an efficient cost of capital must be commensurate with the risk of providing regulated services.				
 A service provider should be provided with effective incentives in order to promote economic efficiency with respect to the regulated services the operator provides. The economic efficiency that should be promoted includes efficient investment the network with which the operator provides regulated services; and the efficient provision of regulated services; and the efficient use of the system with which the operator provides regulated services 	Effective incentives for efficiency are provided through the use of benchmarking and incentive regulation, and the use of market data as benchmarks. An efficient cost of capital must be commensurate with the risk of providing regulated services.				
Regard should be had to the regulatory asset base adoptedin any previous determination or arrangement, orin the Rules	We have regard to the regulatory asset base when determining an allowed rate of return through consideration of the NPV=0 condition. This means that the rate of return should contribute to an ex-ante expectation that over the life of an investment the expected cash flow from the investment repays the capital invested.				
A price or charge for the provision of a regulated service should allow for a return commensurate with the regulatory and commercial risks involved in providing the service	An efficient cost of capital must be commensurate with the risk of providing regulated services. Our consideration of the risk of providing regulated services is set out in greater detail in section 2.4.				
Regard should be had to the economic costs and risks of the potential for under and over investment by a regulated network service provider in the relevant system	A rate of return that is too high may encourage over investment, while a rate of return that is too low may encourage under investment. Over-investment may not be in the long-term interests of consumers with respect to price. Under-investment may not be in the long-term interest of consumers with respect to quality of service.				
Regard should be had to the economic costs and risks of the potential for under and over utilisation of the relevant system	Under-utilisation may be a result of over-investment and over-utilisation may be a result of under-investment. A rate of return that is too high may encourage over investment and a rate of return that is too low may encourage under investment.				

Source: NEL, NGL, AER analysis.
2.1.3 Allowed rate of return objective

The allowed rate of return objective provides that the allowed rate of return is to be commensurate with the efficient financing costs of a benchmark efficient entity with a similar degree of risk as that which applies to the service provider in respect of the provision of its regulated services.¹³⁴

This objective requires consideration of the following concepts that common across the national gas and electricity objectives and the revenue and pricing principles:

- Efficient financing costs
- Benchmarking and incentive regulation
- Return commensurate with risk of providing regulated services

2.2 Benchmark efficiency

Economists typically think of efficiency in three dimensions: productive, allocative and dynamic. Table 7 sets out how this applies in the context of the rate of return.

Dimension of efficiency	Economic meaning	Application to rate of return estimation
Productive efficiency	Achieved when output is produced at minimum cost. This occurs where no more output can be produced given the resources available, that is, the economy is on its production possibility frontier. Productive efficiency incorporates technical efficiency. This refers to the extent that it is technically feasible to reduce any input without decreasing the output or increasing any other input.	Refers to least cost financing (that is, the lowest required return on debt and equity) subject to any constraints, such as risk. For our determinations to be productively efficient we need to incentivise service providers to seek the lowest cost financing (all else being equal).
Allocative efficiency	Achieved when the community gets the greatest return (or utility) from its scarce resources.	Allocative efficiency can be achieved by setting an allowed return consistent with the expected return in the competitive capital market (determined by demand and supply) for an investment of similar degree of risk as a service provider supplying regulated services.
Dynamic efficiency	Refers to the allocation of resources over time, including allocations designed to improve economic efficiency and to generate more resources. This can mean finding better products and better ways of producing goods and services.	Refers to the existence of appropriate investment incentives. We can encourage dynamic efficiency by setting an allowance that does not distort investment decisions. Dynamic efficiency is advanced through incentive regulation rather than cost of service regulation that compensates a service provider for its actual costs no matter how inefficient.

Table 7 Application of efficiency concepts to rate of return

Source: AER analysis; Productivity Commission, On efficiency and effectiveness: Some definitions, May 2013; AER, Better regulation: Rate of return guidelines consultation paper, May 2013.

¹³⁴ NER cll. 6.5.2(c), 6A.6.2(c); NGR r.87(3).

Productive efficiency is promoted through benchmarking and incentive regulation and through setting the rate of return as a market cost of capital reflective of the risks involved in providing regulated services. Allocative efficiency is promoted through estimating the rate of return as a market cost of capital commensurate with the risk involved in providing regulated services. Dynamic efficiency is promoted through benchmarking and incentive regulation, and through adherence to the NPV=0 condition.

2.2.1 Benchmarking and incentive regulation

We apply a benchmark approach and an incentive regulatory framework. We estimate a benchmark rate of return which is then applied to a specific service provider, rather than determining the returns of a specific service provider based on all of its specific circumstances.¹³⁵

The service providers' actual returns could differ from those of the benchmark regulatory allowance depending on how efficiently it operates its business. This is consistent with incentive regulation. That is, our rate of return approach drives efficient outcomes by creating the correct incentive by allowing (requiring) service providers to retain (fund) any additional income (costs) from outperforming (underperforming) the efficient benchmark.¹³⁶

¹³⁵ See AER, *Better regulation: Explanatory statement rate of return guideline,* December 2013, ch. 3.

¹³⁶ NEL, s. 7A(3); NGL s. 24(2)(b).

Table 8 below outlines how we have applied benchmarking and incentive regulation in coming to our draft decision.

Table 8 Use of benchmarking in our allowed rate of return

Element	Application of benchmarking
Gearing ratio	We determine a benchmark gearing ratio from observed gearing ratios of listed Australian energy networks.
Return on equity	The return on equity is benchmarked against market rates via the SLCAPM. The base rate is benchmarked against yields on CGS. The risk premium is benchmarked against the historical returns on the All Ordinaries and the observed equity beta of listed Australia energy networks.
Return on debt - credit rating	We determine a benchmark credit rating that is used in selecting the yield curve benchmark for the return on debt. Our benchmark credit rating is derived from observed credit ratings of privately-owned Australian energy network firms.
Return on debt - term	We determine a benchmark debt term that is used in selecting the yield curve benchmark, and making adjustments to published yields, for the return on debt. Our benchmark debt term is derived from conceptual analysis and observed term of debt issuances of privately owned service providers.
Return on debt - yield	The return on debt is estimated from benchmark yield curves for a given benchmark credit rating and term. This means that the return on debt is benchmarked against market yields on Australian corporate bonds.
Imputation credits - utilisation rate	We consider the utilisation rate is and economy-wide parameter, and accordingly our benchmark utilisation rate is derived from the ABS equity ownership statistics for all Australian equity.
Imputation credits - distribution rate	We consider the distribution rate is an industry-specific parameter. Our benchmark distribution rates is derived from the observed distribution rates of listed Australian equity.

2.2.2 The NPV=0 condition

As the regulatory regime is ex-ante¹³⁷, we consider a rate of return that meets the objectives must provide ex-ante compensation for efficient financing costs. This return would give a benchmark efficient entity a reasonable opportunity to recover at least its efficient financing costs. This is a zero net present value (NPV) investment condition, which is described as follows:¹³⁸

¹³⁷ The AEMC describes, 'allowed revenues for network businesses are now set using the expenditure required by prudent, efficient operators as a benchmark. Companies have incentives to beat the benchmarks so they can keep some of their savings and pass the rest on to customers'. See AEMC, *Overview 2014–15*.

¹³⁸ Partington, G., Satchell, S., *Report to the AER: Discussion of the allowed cost of debt*, 5 May 2016, p. 14.

The zero NPV investment criterion has two important properties. First, a zero NPV investment means that the ex-ante expectation is that over the life of the investment the expected cash flow from the investment meets all the operating expenditure and corporate taxes, repays the capital invested and there is just enough cash flow left over to cover investors' required return on the capital invested. Second, by definition a zero NPV investment is expected to generate no economic rents. Thus, ex-ante no economic rents are expected to be extracted as a consequence of market power. The incentive for investment is just right, encouraging neither too much investment, nor too little.

During the first concurrent evidence session, the experts agreed that setting an allowed return to achieve a zero NPV outcome achieves efficient investment incentives, and is in the long term interest of consumers.¹³⁹

2.2.3 Market cost of capital

Because the market for capital finance is competitive, an efficient service provider is expected to face competitive prices in the market for funds. Therefore, we consider efficient financing costs are reflected in the prevailing market cost of capital (or WACC) for an investment with a similar degree of risk as that which applies to a service provider in respect of the provision of regulated services.¹⁴⁰ As Alfred Kahn stated, 'since the regulated company must go to the open capital market and sell its securities in competition with every other would-be issuer, there is clearly a market price (a rate of interest on borrowed funds, an expected return on equity) that it must be permitted and enabled to pay for the capital it requires'.¹⁴¹

We consider employing a rate of return that is commensurate with the prevailing market cost of capital (or WACC) is consistent with the zero NPV investment condition (see above). We also consider economic efficiency more generally is advanced by employing a rate of return that reflects rates in the market for capital finance. Similarly, Partington and Satchell interpret efficient financing costs as the opportunity cost of capital, which is a market rate of return for assets with a given level of risk.¹⁴²

¹³⁹ AER Concurrent Evidence Session 1 – Proofed Transcript, p.15

¹⁴⁰ See Partington, G., Satchell, S., *Report to the AER: Discussion of the allowed cost of debt*, 5 May 2016, p. 15. We note the cost of capital (from a firm's perspective) is also known as investors' required rate of return (from an investors' perspective).

¹⁴¹ Kahn, A.E., 'The economics of regulation: Principles and institutions', The MIT Press, Massachusetts, 1988, p. 45.

¹⁴² Partington, G., Satchell, S., *Report to the AER: Discussion of the allowed cost of debt*, 5 May 2016, p. 15.

Table 9 below outlines how we have applied benchmarking and incentive regulation in coming to our draft decision.

Table 9 Use of market data in our allowed rate of return

Element	Application of benchmarking	
Gearing ratio	In coming to a benchmark gearing ratio we have had regard to observed gearing levels of listed Australian energy networks. These gearing levels are the result of these firms managing their financing practices as part of their operations in competitive capital markets.	
Return on equity - risk free rate	We estimate the risk free rate from market yields on CGS.	
	Our market risk premium benchmark is informed by market data on:	
	• the historical returns on the All Ordinaries	
Return on equity - market risk premium	 analyst forecasts and market prices of equities that are used in dividend growth models 	
	 conditioning variables derived from market prices and dividends 	
Return on equity - beta	Our equity beta estimate is informed by market prices and dividends of listed Australian energy networks relative to the market prices and dividends for the ASX300.	
Return on debt - credit rating and term	Our benchmark credit rating is derived from observed credit ratings of privately-owned Australian energy network firms. Our benchmark debt term is informed by from observed term of debt issuances of privately owned service providers. These firms are managing their financing practices as part of their operations in competitive capital markets.	
Return on debt - yield	The return on debt is estimated from market yields on Australian corporate bonds.	
Imputation credits - utilisation rate	Our benchmark utilisation rate is derived from the ABS equity ownership statistics for all Australian equity. This ownership data is the result of the operation of equity markets.	
Imputation credits - distribution rate	Our benchmark distribution rates is derived from the observed distribution rates of listed Australian equity. These listed Australian firms are determining their distribution rates as part of their operations in competitive equity markets.	

2.2.4 Commensurate with risk

When estimating our benchmark rate of return we have regard to the degree of risk involved in providing regulated services. This is consistent with the RPPs, which state that a price or charge should allow for a return that matches the regulatory and commercial risks involved in providing the regulated service to which that charge relates. It also contributes to the achievement of the legislative objectives by promoting efficiency – it is well accepted that there is a risk-return trade-off¹⁴³ and it would not be efficient to determine an allowed return that is not commensurate with the risks involved.

Further consideration of the risks involved in providing regulated services is set out in section 2.4 below.

2.3 Measuring success

In sections 2.1 and 2.2 we set out the key concepts that are particularly relevant to setting the rate of return and the value of imputation credits that contribute to the achievement of the legislative objectives to the greatest degree. In the remainder of this draft decision we set out an approach to estimating the rate of return and value of imputation credits that we consider is likely to achieve the legislative objectives to the greatest degree by focusing on these key concepts.

At our public forum in September 2017 some stakeholders questioned whether our 2013 Guidelines had contributed to achieving the legislative objectives. Similar submissions were subsequently made by the CRG, ECA, MEU, and EUAA.¹⁴⁴

The CRG submitted that it considers our 2013 Guidelines have not been contributing to achieving the objectives to the greatest degree, stating:¹⁴⁵

The CRG strongly believes the Guideline is not meeting its objective.

There is no testing as to whether the Guideline serves the long term interests of consumers. The available evidence demonstrates the objectives are not being met.

The ECA submitted:146

On the available evidence Energy Consumers Australia concludes that the current approach to setting the allowed rate of return is not consistent with the relevant objectives. We believe the rate of return is set in excess of the efficient financing costs of an efficient entity.

The Major Energy Users Inc submitted that:147

The MEU considers that the current approach to setting the RoR has provided a massive transfer of wealth from consumers to network asset owners, not only directly but also in the incentive it has provided to inefficient network investment causing future consumers to inherit a massively overstated value for the network assets they will need

¹⁴³ Handley, Advice on the Return on Equity, Report prepared for the Australian Energy Regulator, 16 October 2014, p. 4.

¹⁴⁴ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 16.

¹⁴⁵ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 16.

¹⁴⁶ ECA, Review of the rate of return guideline: Response to the AER Issues Paper, December 2017, p. 13.

¹⁴⁷ MEU, Submission by The Major Energy Users Inc, December 2017, p. 12.

The EUAA submitted:148

Our view is that the 2013 Guideline has not met the NEO and NGO objectives.

•••

Consumers have received a safe and reliable service. Unfortunately, we believe it has been achieved with far too much capital investment (see the fall in capacity utilisation rates) and poor productivity by many networks with the result that prices are far too high. Many networks are not at "benchmark" efficient levels and yet they continue to earn a secure revenue flow that produces, what limited evidence suggests, to be above normal profits for a regulated natural monopoly.

CRG submitted that, as an important component of network prices, the rate of return has contributed to an energy affordability crisis, stating:¹⁴⁹

The Review is occurring in an environment of increasing energy prices that could be described as an 'affordability crisis.' The impact has been particularly severe on low-income households, young families and trade exposed energy intensive businesses including agriculture, manufacturing and catering. Increasing network charges have been a significant contributor to these unsustainable prices.

•••

Faced with massive retail price increases, consumers have responded where able to do so by switching fuels, adopting and investing in energy efficiency, and investing in distributed generation in the form of rooftop solar photovoltaic (PV) generation.

By contrast, the network shareholder group submitted:¹⁵⁰

The framework for incentive-based economic regulation is working as intended, specifically where investment is made by the private sector.

Energex and Ergon Energy submitted that:¹⁵¹

Certainly, there are aspects of the AER's current approach that most stakeholders agree will contribute to the achievement of the NEO and ARORO.

Energy Networks Australia submitted that our current review should build on the 2013 Guidelines, stating:¹⁵²

¹⁴⁸ EUAA, EUAA submission – AER Rate of Return Review Issues Paper October 2017, December 2017, p. 2.

¹⁴⁹ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, pp. 16, 18.

¹⁵⁰ NSG, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), May 2018, p. 1.

¹⁵¹ Energex and Ergon Energy, AER Issues Paper - Review of the Rate of Return Guidelines - Ergon Energy and Energex Submission, December 2017, p. 2.

¹⁵² ENA, AER Rate of Return Guidelines Response to Issues Paper, December 2017, p. 3.

The 2017 Rate of Return Guideline process is an opportunity to build on considerable work undertaken in the 2013 guideline review process, and network businesses support an incremental approach building on this past guideline review

Similarly, CitiPower, Powercor, SA Power Networks, United Energy, and Australian Gas Infrastructure Group submitted that the 2013 Guidelines were the outcome of significant review:¹⁵³

The AER and stakeholders have generally agreed that the review of the Guideline should be incremental in nature. This is appropriate given a current Guideline already exists that itself was the outcome of significant review.

If the rate of return derived from our guideline is too high or too low then it will not promote the achievement of the legislative objectives. Consumer groups have strongly advocated that the rate of return derived from the approach set out in our 2013 guidelines is too high and it has therefore not promoted efficient investment in, and efficient operation and use of, electricity services for the long term interests of consumers of electricity.

The CRG, ECA, MEU, and EUAA referred to the following indicators that the 2013 Guidelines have not been contributing to achieving the legislative objectives to the greatest degree:¹⁵⁴

- The parameter values we have chosen in our decisions are too high given the low risk in providing regulated services
- Energy prices have increased substantially resulting in price-stress for some customers, increased disconnections, and switching to alternative fuels.
- Regulatory asset bases (RABs) have increased substantially while utilisation has declined and reliability has not been an issue
- Service providers have been able to achieve supernormal profits
- Recent sales and acquisitions of service provders have been at RAB multiples materially greater than 1

We acknowledge that this review is being undertaken in an environment of heightened consumer concern about increasing energy costs and relatively large (and sometimes underutilised) investment in regulated assets bases over recent years. While the rate of return is an important contributor to network prices, there are other network and nonnetwork costs that also contribute to overall energy prices. Nonetheless, we are

¹⁵³ CitiPower, Powercor, SA Power Networks, United Energy, and Australian Gas Infrastructure Group, Submission on the AER Issues Paper, December 2017, p. 2.

¹⁵⁴ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 16 - 34; Energy Consumers Australia, Review of the rate of return guideline, December 2017, p. 6 -14; MEU, Submission by The Major Energy Users Inc, December 2017, p. 14, Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p. 2

cognisant of the effect that higher energy prices may have on the willingness of consumers to pay for further improvements in network reliability.

We have had regard to these factors when considering the rate of return and value of imputation credits for this decision. We have done so by adopting an approach of focusing on the best empirical estimates for rate of return parameters. In this way we consider we can determine an approach that is most likely to promote the NEO and NGO.

We have also had regard to the arguments submitted by the CRG, ECA, MEU, and EUAA on each parameter value. Our considerations of these matters are set out in the remainder of this draft decision. In a number of areas we have determined parameter values or estimation approaches that differ from those in our 2013 Guidelines. We have done so on the basis that the updated parameter values will contribute to the legislative objective to the greatest degree given the current environment. We consider the current environment is different in a number of ways to the environment at the time of our 2013 Guidelines. Most notably we consider that market volatility has subsided and risk premiums have reduced since 2012-13 (this is discussed further in the chapter 7).

We consider it is important for the achievement of the legislative objectives that we periodically review our rate of return and value of imputation credits. This process of review, combined with the use of incentive regulation, is a key driver of efficiency and the continued achievement of the legislative objectives over time. Therefore, the achievement of the legislative objectives should not be viewed solely in a static sense.

On the use of incentive regulation, the ECA submitted:155

The objective is not to set the rate of return based on a benchmark so that the provider can outperform the rate of return by the way it is financed – the intention is that the rate of return is a constraint so that the provider has maximum incentive to generate higher returns by efficiency in its investments and its operations.

We agree that the objective of the allowed rate of return under an incentive regulatory framework is not to provide a guaranteed degree of outperformance. However, we also note that it is important for allocative and dynamic efficiency that the allowed rate of return provides (in expectation) an opportunity for service providers to recover their efficient costs (without expectation of monopoly rents).

We note that we have updated our empirical analysis in a number of areas consistent with incentive regulation. We have reviewed our benchmark gearing, credit rating, debt term, and overall debt costs by examining the recent, actual costs and financial management practices of service providers. In chapter 10 we set out how we have

¹⁵⁵ ECA, Review of the rate of return guideline: Response to the AER Issues Paper, December 2017, p. 11.

considered service providers' revealed cost information when deciding on our approach to debt.

The CCP16, CRG, ECA, EUAA, and MEU submitted that we should examine information on the actual returns achieved by service providers, through data on actual profitability and RAB multiples, when considering if our rate of return achieves the legislative objectives.¹⁵⁶ The CCP16 submitted that actual profitability and RAB multiples could be used as a cross-check on the selection of parameter values and the overall rate of return.¹⁵⁷

We consider actual profitability and RAB multiples in 3. In summary, actual profitability and RAB multiples from asset sales may provide some indication of the appropriateness of the allowed total rate of return. However, they may also be indicative of other elements of the firm's cash flows, such as unregulated activities and outperformance on expenditure allowances. Further, sales that have already occurred and past profitability may not reflect changes to the overall rate of return that are occurring at present. Ultimately, this information does not provide a definitive answer to the specific return investors require.

Overall, we consider that these measures cannot be used to directly determine parameter estimates for the allowed rate of return. We agree with the CCP16 submission that there is difficulty in disaggregating the information contained in RAB multiples and historical profitability measures to determine the degree of outperformance of the allowed rate of return.

However, we consider that there may be useful information within the trends in RAB multiples and historical profitability measures over time. Comparisons of RAB multiples and historical profitability measures can provide information on the performance of the regulatory system as a whole. This information may be helpful in considering whether the business' actual rate of return has been systematically lower or higher than the allowed rate of return.

The CCP16 submitted that this information cannot be used at a parameter level but can inform the overall exercise of judgement in setting the rate of return or reviewing other elements of the regulatory regime¹⁵⁸. We agree that RAB multiples and historical profitability may provide useful contextual information that may inform our exercise of regulatory judgment necessary when determining the allowed rate of return.

¹⁵⁶ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 59 - 63; Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 20, Energy Consumers Australia, Review of the rate of return guideline, December 2017, p. 13, Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p. 4; Major Energy Users, Review of the rate of return guidelines, December 2017, p. 14-15.

¹⁵⁷ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 60- 61.

¹⁵⁸ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 27-28.

Further, we note in section 2.2.3 that use of market data is an important concept for achieving the legislative objectives, and we do have regard to actual returns in other parts of our rate of return estimation. Our equity beta estimation is based on data on the returns (capital gains and dividends) to listed Australian energy network firms relative to the returns on the ASX300. When estimating the market risk premium we have regard to the historical returns on the Australian stock market. We have also reviewed the actual debt costs of regulated service providers.

2.4 Risk and return

We are to determine the allowed rate of return such that it contributes to the achievement of the NEO and NGO (the Objectives). The appropriate risk compensation is also crucial in achieving the NGO and NEO¹⁵⁹.

2.4.1 Compensation for risk

In this section we discuss concepts of risk, WACC and SLCPM, summarise past discussions, and consider stakeholder submissions on risk.

As part of reviewing the allowable rate of return, we are required to consider how to compensate for the risk exposure of service providers in supplying regulated services. Appropriate risk compensation is an important part of the rate of return regulatory framework and is integral to achieving the legislative objectives.

What is risk?

Risk is the degree of uncertainty about an event—such as the uncertainty around the expectation of the return on an investment.¹⁶⁰ As explained by Handley, the risk-return trade-off is one of the most fundamental paradigms of finance.¹⁶¹ The trade-off states that a risk averse investor will want a higher expected return when faced with a higher risk. This concept is illustrated by Brealey et al. below:¹⁶²

Investments A and B both have an expected return of 10%, but because investment A has the greater spread of possible returns, it is more risky than B. We can measure this spread by the standard deviation. Investment A has a standard deviation of 15%; B, 7.5%. Most investors would prefer B to A. Investments B and C both have the same standard deviation, but C offers a higher expected return. Most investors would prefer C to B.

¹⁵⁹ As set out in NER cl.6; NGR cl. 6A

¹⁶⁰ Bishop, S., Faff, R., Oliver, B., Twite, G., 'Corporate Finance', Ed. 5 Pearson Prentice Hall, 2004, p. 577.

¹⁶¹ Handley, J., 'Advice on the return on equity: report prepared for the AER', 16 October 2014, p. 4.

¹⁶² Brealey, R., Myers, S., Allen, F., 'Principles of corporate finance', 2011, Ed. 10, McGraw-Hill Irwin, Figure 8.2, p. 187.

Figure 8 Risk versus expected return



Source: Brealey, Myers, Allen (2011), Figure 8.2.

The above example explains the relationship between risk and return for a single investment. For a given expected return, risk averse investors are assumed to prefer an investment with a lower variance. However, we note that for an investment in an investment portfolio, the risk relevant to its price is the risk it will add to the portfolio. Therefore, under the assumption that investors hold fully diversified 'efficient' market portfolios, only an investment's systematic risk is relevant. In finance, there are two distinct types of risk—systematic risk (market risk or nondiversifiable risk) and non-systematic risk (firm-specific risk or diversifiable risk). That is, in finance:¹⁶³

The risk of any share can be broken down into two parts. There is the *unique risk* that is peculiar to that share, and there is the *market risk* that is associated with market-wide variations. Investors can eliminate unique risk by holding a well-diversified portfolio, but they cannot eliminate market risk. *All* the risk of a full diversified portfolio is market risk.

Similarly, McKenzie and Partington have advised:164

Modern finance theory specifies that the risk to be compensated via the WACC is the non-diversifiable, or systematic, component of total risk (in simple terms, that risk which cannot be eliminated by holding stocks in a well diversified portfolio). This risk is measured as covariance, or equivalently beta, risk.

The appropriate risk compensation is also crucial in achieving the NGO and NEO¹⁶⁵. The Objectives require the promotion of efficient investment in, and efficient operation and use of, energy network services for the long-term interests of energy consumers.¹⁶⁶

Efficiency

When considering an efficient return for risk, it is important to differentiate between risk that is efficiently compensated through the allowed rate of return (compensable risk) and non-compensable risk. When developing the 2013 Guideline, we commissioned Frontier to explore these risks and to provide advice on what risks we should compensate service providers for through the allowed rate of return.¹⁶⁷

In estimating an efficient allowed rate of return, we highlight five factors that we consider important:

- Estimating a forward-looking return
- Estimating a market return through use of market data
- Providing a return for systematic risk
- The role of incentive regulation
- Similar degree of risk as a service provider in the provision of regulated services

¹⁶³ Brealey, R., Myers, S., Partington, G., Robinson, D., 'Principles of corporate finance', 2007, The McGraw-Hill Companies Inc., 2007, p. 201.

¹⁶⁴ McKenzie, M., Partington, G., *Risk, asset pricing models and WACC*, June 2013, p. 10.

¹⁶⁵ As set out in NER cl.6; NGR cl. 6A

¹⁶⁶ The National Electricity Objective is in section 7 of the National Electricity Law and the National Gas Objective is in section 23 of the National Gas Law.

¹⁶⁷ Frontier, Assessing risk when determining the appropriate rate of return for regulated energy networks in Australia, July 2013.

Forward looking return

Risk is the degree of uncertainty about an event—such as the uncertainty around the expectation of the return on an investment.¹⁶⁸ It is strictly a forward-looking concept, as no event is uncertain after it has occurred.

Use of market data

Because the market for capital finance is competitive, an efficient service provider will face competitive prices in the market for funds. We continue to be of the view that efficient financing costs are reflected in the prevailing market cost of capital (or WACC) for an investment with a similar degree of risk as that which applies to a service provider in respect of the provision of standard control, prescribed transmission, or reference services.¹⁶⁹

As Alfred Kahn stated: 'since the regulated company must go to the open capital market and sell its securities in competition with every other would-be issuer, there is clearly a market price (a rate of interest on borrowed funds, an expected return on equity) that it must be permitted and enabled to pay for the capital it requires'.¹⁷⁰

We maintain the view that economic efficiency is advanced by employing a rate of return that reflects rates in the market for capital finance.¹⁷¹ Partington and Satchell have also interpreted efficient financing costs as the opportunity cost of capital, which is a market rate of return for assets with a given level of risk.¹⁷²

We have received divergent submissions that discuss the use of market data.

Australian Pipeline and Gas Association's submission to the evidence session supports the use of market data to estimate costs.¹⁷³

Energy Users Association of Australia submitted concerns with transparency in regulation as well as the use of available data.¹⁷⁴ We consider that the use of market data assists in addressing the Energy Users Association of Australia's concerns about transparency.

¹⁶⁸ Bishop, S., Faff, R., Oliver, B., Twite, G., 'Corporate Finance', Ed. 5 Pearson Prentice Hall, 2004, p. 577.

¹⁶⁹ See Partington, G., Satchell, S., *Report to the AER: Discussion of the allowed cost of debt*, 5 May 2016, p. 15. We note the cost of capital (from a firm's perspective) is also known as investors' required rate of return (from an investors' perspective).

Kahn, A.E., 'The economics of regulation: Principles and institutions', The MIT Press, Massachusetts, 1988, p. 45.

¹⁷¹ For example, see: AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 12, 16. AER, Final decision United Energy distribution determination 2016 to 2020 Attachment 3–Rate of return, May 2016, pp. 281–292.

¹⁷² Partington, G., Satchell, S., *Report to the AER: Discussion of the allowed cost of debt*, 5 May 2016, p. 15.

¹⁷³ Australian Pipeline Gas Association, Submission to the AER, Review of rate of return guideline, 4 May 2018, p.3

¹⁷⁴ Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p. 5–6.

Energy Consumers Australia highlighted their opposition to the use of market data, believing that it reflects risks (and returns) greater than that of service providers.¹⁷⁵ Instead, the Energy Consumers Australia suggested to go into more depth, and consider how much risk individual assets owned by the companies have.¹⁷⁶ Our view is that while conceptually Energy Consumers Australia's proposal may provide a finer estimate of risk, it may not be entirely feasible in reality. This is because data for estimating systematic risk on an individual asset basis is unlikely to be observable.

The Consumer Reference Group submitted that the volatility in market data/share price for the firms included in our comparator set may be related to market sentiment rather than the fundamental characteristics.¹⁷⁷ Additionally, they raised that the comparator firms have unregulated sources of revenue and therefore contain more risk than if they were only providing regulated services.¹⁷⁸ This view was also submitted by the Major Energy Users.¹⁷⁹

We acknowledge that stocks can experience short-term fluctuations due to market movements and one-off events. To mitigate the effects of short-term movement, we consider regressions using multiple estimation periods when considering the equity beta parameter (see section 8.3.2 for more discussion).¹⁸⁰

We consider that, ideally firms that share all or most of the key characteristics of a benchmark efficient entity would be used in our estimates. However, in practice, few firms would fully reflect this benchmark. Therefore, we use market data for domestic businesses that are considered to be reasonable comparators to a benchmark efficient entity with a similar degree of risk as an entity providing regulated services to inform the equity beta estimate. We discuss this point in detail in section 0.

Diversification and compensation for systematic risk

The rate of return allows a service provider to compensate investors for the risk of committing capital to fund investments in its network. While we agree with the view that all risks should be accounted for in the risk framework¹⁸¹, we maintain the view that investors do not require compensation via the rate of return for all risks involved in investing in a service provider.

¹⁷⁵ Energy Consumers Australia, Review of the rate of return guideline, December 2017, p. 20.

¹⁷⁶ Energy Consumers Australia, Review of the rate of return guideline, December 2017, p. 15–18.

¹⁷⁷ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 47

¹⁷⁸ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 26.

¹⁷⁹ Major Energy Users, Review of the rate of return guidelines, December 2017, p 7–8.

¹⁸⁰ We consider an estimate from the longest estimation period available, as well as a period that excludes the 'technology bubble' and the global financial crisis. AER, Better Regulation Explanatory Statement Rate of Return Guideline (Appendices), December 2013, p. 49

¹⁸¹ Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p.12, Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, p.10, Energy Networks Australia, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Evidence Sessions, p.7

Since investors can eliminate non-systematic risk by holding a well-diversified portfolio, then it is unlikely that investors will require compensation for these risks. It follows that it would be inefficient to compensate investors for this non-systematic risk in the allowed rate of return.

The view that only systematic risks should be compensated through the allowed rate of return is an area of agreement between experts other than David Johnstone who has concerns with the CAPM framework but did not propose an alternative.¹⁸²

In setting the allowed return on equity in our previous decisions, we provided compensation for the systematic risk that an efficient service provider would face through the equity beta.¹⁸³ The equity beta under the SLCAPM measures systematic risk as the sensitivity of an asset or business to the overall movements in the market. ¹⁸⁴ It does this by measuring the standardised correlation between the returns on this asset or business with that of the overall market.¹⁸⁵

We have since received submissions questioning the appropriateness of beta and CAPM, which we consider in chapter 8.

The Consumer Challenge Panel submitted that the risk of network businesses not achieving their expected rate of return is low due to regulation insulating businesses from asset impairment.¹⁸⁶ This submission is consistent with our view and with the method of estimating the expected allowed return of a benchmark efficient entity's costs and incentives of efficient investment.

In setting the allowed return on debt, we provide the efficient compensation for the risks that an investor in the service provider's debt faces, as they are included in the promised returns we observe using our debt data sources.¹⁸⁷ Further, since we provide a return on debt based on the promised yield, our allowed rate of return will be slightly above the expected return. This is because we consider that promised returns will exceed expected returns, as the expected return is the promised return less the default

¹⁸² Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, p.23.

¹⁸³ For example, AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 20.

¹⁸⁴ Theoretically, this asset or business is 'a benchmark efficient entity'. In practice, we use a sample of businesses we consider comparable to a benchmark efficient entity to calculate equity beta. See: AER, Better regulation explanatory statement rate of return guideline, December 2013, pp. 83–86.

¹⁸⁵ McKenzie, M., Partington, G., *Risk, asset pricing models and WACC*, June 2013, p. 21; Brealey, R., Myers, S., Partington, G., Robinson, D., 'Principles of corporate finance', 2007, The McGraw-Hill Companies Inc., 2007, p. 107.

¹⁸⁶ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p.5

¹⁸⁷ We observe the promised returns of debt issued by a sample of firms we consider comparable to a benchmark efficient entity based on the benchmark credit rating and term. In practice, we may have overcompensated a benchmark efficient entity for these risks as we observe broad BBB debt whereas we consider a benchmark efficient entity would issue BBB+ debt. In this draft decision we have refined our method — see chapter 10.

risk.¹⁸⁸ This means the overall allowed return includes some compensation for the expected default loss on debt due to non-systematic risk.

Benchmarking and incentive regulation

The service providers' actual returns could differ from the allowed return depending on how efficiently it operates its business. This is consistent with incentive regulation. That is, our rate of return approach drives efficient outcomes by creating the correct incentive by allowing (requiring) service providers to retain (fund) any additional income (costs) as a result of outperforming (underperforming) the efficient benchmark.¹⁸⁹

Similar degree of risk

The ARORO aims to set the efficient financing costs of a benchmark efficient entity (BEE) with a similar degree of risk as that which applies to the service provider in respect of the provision of (in the case of the NER standard control or prescribed transmission services) or (in the case of the NGR reference services). Given this, an essential concept to consider in giving effect to the ARORO is 'risk'.

The risk of providing regulated services is a core element of the rate of return due to the important relation between risk and required returns in finance theory. This ensures that the allowed rate of return provides a reasonable opportunity to recover efficient financing costs relative to risks, and promotes efficiency in investment, provision and use of regulated services.

There were divergent views on the concept of a benchmark efficient entity:

- Illan Sadeh and Greg Houston considered a benchmark efficient entity an important concept, highlighting its impact on judgement and discretion since the use of a benchmark removed the need for the regulator to make decisions about a service provider's financing decisions.¹⁹⁰
- David Johnstone suggested that the regulator should be highly interested in service providers' financing decisions and therefore uncertain about using a BEE.¹⁹¹
- The Network Shareholder Group submitted that the concept of a benchmark efficient entity is critical to the incentive framework and establishing an effective rate of return. The opportunity to outperform the benchmark encourages investment, efficiency and innovation.¹⁹²

¹⁸⁸ For further explanation, see McKenzie and Partington, Report to the AER: The relationship between the cost of debt and the cost of equity, March 2013, pp. 7, 21; AER, Final decision: Access arrangement final decision— Multinet Gas (DB No. 1) Pty Ltd, Multinet Gas (DB No. 2) Pty Ltd 2013-17, March 2013, Part 3, p. 48

¹⁸⁹ NEL, s. 7A (3); NGL s. 24(2) (b).

¹⁹⁰ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.12-16.

¹⁹¹ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 17

¹⁹² Network Shareholder Group, , Response to issues paper on the review of the Rate of Return Guidelines, December 2017, p.5

We consider that the concept of a benchmark is important given we apply an incentive regulatory framework rather than a cost of service approach which is generally consistent with evaluating service providers' actual financing decisions. We also note that under the current rate of return provisions of the NER our regulatory task is to determine the allowed rate of return such that it achieves the ARORO, which includes the term 'benchmark efficient entity' in its definition:¹⁹³

In setting the allowed return on equity, we provide compensation through the equity beta for the systematic risk that a benchmark efficient entity would face. In setting the allowed return on debt, we provide compensation using a benchmark credit rating for a benchmark efficient entity with a similar degree of risk as that which applies to the service provider in respect of providing regulated services. We use market data for firms that we consider are most similar to the regulated service providers risk profile. We consider that this market data reveals the required compensation for an investment with a similar degree of risk as that which applies to a service provider in respect of the provision of standard control, prescribed transmission, or reference services.

We maintain the view that the Australian market is the market within which a benchmark efficient entity for each service provider operates, and this is appropriate to make it properly comparable in degree of risk to the service providers. This recognises that the location of a business determines the conditions under which the business operates and these include the regulatory regime, tax laws, industry structure and broader economic environment. As most of these conditions will be different from those prevailing for overseas entities, the risk profile of overseas entities is likely to differ from those within Australia. Consequently, the returns required are also likely to differ. Hence, when estimating input parameters for equity we have placed most reliance on Australian market data whilst using overseas data informatively.

We have received stakeholder submissions that proposed including international energy firms in our comparator set for estimating the equity beta parameter. This is discussed in more detail in section 8.3.4.

Technological risk

We recognise that disruptive technologies such as solar panels, smart technology and power storage are beginning to change how consumers produce and consume electricity.¹⁹⁵ We also recognise this could affect how consumers use network infrastructure and may change some risks faced by service providers.

We have received submissions on technological risk's impact on the rate of return:

¹⁹³ NER, cl. 6.5.2(c); NER, cl. 6A.6.2(c); NGR r. 87(3).

¹⁹⁴ See Partington, G., Satchell, S., *Report to the AER: Discussion of the allowed cost of debt*, 5 May 2016, p. 15. We note the cost of capital (from a firm's perspective) is also known as investors' required rate of return (from an investors' perspective).

¹⁹⁵ For example, AER, Final decision SA Power Networks determination 2015–16 to 2019–20, Attachment 3 - Rate of Return, October 2015, p. 447.

- Service providers have stated that we have not adequately accounted for the recent risks arising from disruptive technologies.¹⁹⁶
- APA views that limited data and effects of regulatory change make explicit consideration of the effects of disruptive technologies on systematic risk difficult. Additionally it also submitted that technology will also affect gas fired power generation and the transportation of gas.¹⁹⁷
- Evoenergy submitted that any change in systematic risk arising from technological changes will take some time to be reflected in beta estimates, emphasising the importance of these changes in determining the direction of beta estimates.¹⁹⁸

We continue to be of the view that the technological risks submitted to us should not be compensated through the rate of return for the following reasons:

In determining whether this risk needs to be accounted for in the equity beta (or the rate of return generally), we have to consider whether the risk is systematic. We consider developments in the sort of technologies (such as distributed generation, smart technology and power storage) are unlikely to have significant effects outside the energy sector. Investors in the market would be able to diversify such risks by investing in other industries. This point was highlighted by Graham Partington in the expert evidence sessions¹⁹⁹.

- Jemena agreed that risks stemming from technological change may be nonsystematic and may require compensation through cash follows rather rate of return allowance.²⁰⁰
- The New Zealand Commerce Commission also noted that technological risk is 'generally non-systematic in nature and so is not relevant to WACC'.²⁰¹
- Energy Networks Australia sought clarification of our view on stranding risk.²⁰² To the extent that there are genuine risks of extreme changes in demand for specific service providers which present the potential for asset stranding, the regulatory regime for gas and electricity can mitigate the risk by providing prudent discount and accelerated depreciation provisions. Further, the rules protect capital invested.²⁰³

See, for example, Cheung Kong Infrastructure – Submission on rate of returns issue paper – 12 December 2017, p. 3-4; Energy Networks Australia – Submission on rate of returns issue paper – 12 December 2017, P.7, Consumer challenge panel 16 – submission on rate of returns issue paper – 12 December 2017, p.37.

¹⁹⁷ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p.16

¹⁹⁸ Evoenergy, Review of rate of return guideline –evidence sessions, 4 May 2018, p. 2. , APA,

¹⁹⁹ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 47

²⁰⁰ Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p. 3.

 ²⁰¹ New Zealand Commerce Commission, Input methodologies review decisions, Topic paper 4 cost of capital issues,
 20 December 2016, p.109

²⁰² Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 7, 30

²⁰³ NER, cl, 6A.26.

 The effects of technological risk can be uncertain and changes to technology may also lead to positive impacts for businesses. Graham Partington and Stephen Satchell use the example of an increase in electric cars likely creating large demand for electricity charging points.²⁰⁴ Therefore, it is not clear to us that the final impact of the sort of technology risk noted by service providers require compensation through the rate of return.

Graham Partington and Stephen Satchell also noted that 'to the extent that technological risks is systematic it will be reflected in the equity beta, without the need to undertake additional analysis.'²⁰⁵

We note Illan Sadeh, Stephen Gray and Evoenergy expressed the view that the market did not fully factor in technological risk.²⁰⁶ However, it is not clear that is the case. As previously noted in this section 2.2.3, we consider that market data is appropriate for informing our rate of return parameters. It is also standard practice to use market data to estimate equity beta. Our analysis of broker and valuation reports show that there has been no explicit adjustments to rate of return parameters for the technological risks highlighted in the above submissions. Further, to the extent a technological risk is not reflected in the market data, then this may be non-systematic and does not warrant compensation through the rate of return on an ex-ante basis.

In respect of the view that technological risk is systematic, the Consumer Challenge Panel considered this view is tenuous and unproven.²⁰⁷ Additionally, the Consumer Challenge Panel also stated that consumers bear the risk of the underutilised assets, as the full costs of the assets must continue to be reflected in the regulated revenues and prices. The view that consumers bear stranded asset risk was also highlighted by Energy Users Association of Australia.²⁰⁸ The Consumer Challenge Panel also noted that many industries are affected by changes in technological trends but do not receive adjustment assistance.²⁰⁹

After reviewing all evidence presented to us, our view is that a specific adjustment to the rate of return should not be made to account for technological change.

Catastrophic and other risk

Submissions and experts discussed the treatment of catastrophic risks that are of low probability, but have the potential for significant damage (such as natural disasters).

²⁰⁴ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, May 2018, p.4

²⁰⁵ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, May 2018, p.10

²⁰⁶ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.52. Evoenergy, Review of rate of return guideline – evidence sessions – May 2018, p.2

²⁰⁷ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 44-46.

²⁰⁸ Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p.8

²⁰⁹ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p.44

Our view is that these risks should not be compensated through the rate of return via the equity beta parameter for the following reasons:

- Catastrophic risk cannot be considered systematic risks, as the risk is unrelated to the market and only a portion of the market is affected.²¹⁰ This was an area of agreement in the first concurrent evidence session.²¹¹
- The risks can be mitigated by the option to purchase insurance, with insurance costs flowing through to customers through our opex allowance. This view was highlighted by experts, investors and consumer advocates.²¹²
- Catastrophic risks can be mitigated via the potential to pass the costs to relevant users and shipwreck clauses. However, Graham Partington and Stephen Satchell note the risk of automatically allowing the costs of such events to be passed onto consumers, effectively transferring the risk away from providers.²¹³

We have received submissions from the Network Shareholder Group and APA suggesting Australian governments have increased perceived and actual intervention in regulation which has translated into increased regulatory risk and will see a negative impact on capital investment.²¹⁴ Energy Consumers Australia submitted the opposite view, stating that political, regulatory and sovereign risk in Australia is low. ²¹⁵

The Consumer Challenge Panel submitted that it considered policy risks are not systematic. It stated that: ²¹⁶

- The AER should regulate under the existing rules where the consumer not the utility bears the policy risks
- Compensation should not be provided ex ante for the uncertain possibility of a policy change by governments.
- If there is to be compensation for the financial impacts of a policy change, that is a matter for consideration by governments at the time of the policy change, not by the regulator in anticipation of the policy change. This view was also highlighted by Ian McAuley.²¹⁷

We are of the view that the regulatory risks (government intervention and uncertainty in policy) should not be compensated through the rate of return for the following reasons:

²¹⁰ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, May 2018, P.11

²¹¹ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.47

²¹² AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.46, p.50, p.59, Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p.8, Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p.3

²¹³ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, May 2018, p.11

²¹⁴ Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p.7, APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p.16

²¹⁵ Energy Consumers Australia, Review of the rate of return guideline, December 2017, p.26

²¹⁶ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 44–46.

²¹⁷ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 3

- Similar to technology risks, the regulatory risks are unlikely to have significant effects outside the energy sector. Investors in the market would be able to 'diversify' away such risks by holding a market portfolio. As a result, we do not consider the risk arising from policy changes can be reasonably classified as systematic risk²¹⁸ and be accounted for in the equity beta (or the rate of return generally) for a benchmark efficient entity on an ex-ante basis.²¹⁹
- To the extent any compensation is deemed necessary, we consider it should be determined by the government, as our regulatory task is to set rate of return to efficiently compensate for the systematic risk of supplying regulated energy network services.

CAPM

In setting the allowed return on equity, we provide compensation through the foundation model approach for the systematic risk that an efficient service provider would face. The equity beta under the SLCAPM measures systematic risk as the sensitivity of an asset or business to the overall movements in the market.²²⁰ It does this by measuring the standardised correlation between the returns on this asset or business with that of the overall market.²²¹

There was a degree of disagreement with the SLCAPM being an appropriate framework for reflecting systematic risk and the required rate of return for a service provider:

- The Energy Consumers Australia suggested a "direct risk-based approach".²²² The submission did not contain further details regarding this alternative approach, including a suggested methodology.
- Ian McAuley and Major Energy Users pointed to beta as an indicator of short-term volatility, which is largely irrelevant for long-term systematic risk.²²³

²¹⁸ For example, AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 447–448;

²¹⁹ In our April/June 2015 final and preliminary decisions, we considered that, 'Even if the risk arising from disruptive technologies has increased the systematic risk of a benchmark efficient entity, we consider this will be captured in our empirical equity beta estimates to the extent that investors are aware of the risk' (see, for example, AER, Jemena Gas Networks final decision 2015-20: Attachment 3—Rate of return, June 2015, p. 406). We do not consider the risk arising from disruptive technologies can be reasonably classified as systematic risk. As Partington and Satchell stated in their October 2015 report, 'Since we do not consider the impact of disruptive technology to be a systematic risk we do not consider that it would be captured by estimates of beta, however recent they are' (see Partington and Satchell, Report to the AER: Analysis of criticism of 2015 determinations, October 2015, p. 39).

²²⁰ Theoretically, this asset or business is 'a benchmark efficient entity'. In practice, we use a sample of businesses we consider comparable to a benchmark efficient entity to calculate equity beta. See: AER, Better regulation explanatory statement rate of return guideline, December 2013, pp. 83–86.

²²¹ McKenzie, M., Partington, G., *Risk, asset pricing models and WACC*, June 2013, p. 21; Brealey, R., Myers, S., Partington, G., Robinson, D., 'Principles of corporate finance', 2007, The McGraw-Hill Companies Inc., 2007, p. 107.

²²² Energy Consumers Australia, Review of the rate of return guideline, December 2017, p.26

- Ian McAuley and The Australian Institute also noted that a service provider could be uncorrelated with the overall market and receive only the risk free rate.²²⁴
- The Consumer Reference Group submitted that a more fundamental review was required due to major issues with the CAPM.²²⁵
- We also received a wide range of submissions supporting the SLCAPM framework.²²⁶

While we acknowledge that all models are simplifications, we agree with Graham Partington and Stephen Satchell's view that the SLCAPM framework is the best approach to reflect the systematic risk of a benchmark efficient entity.²²⁷ We reached this view for the following reasons:

 There is widespread agreement for an incremental review of the Guideline implies continued use of the foundational model approach (including the SLCAPM).²²⁸ This is unless material changes in risks or market conditions justify an alternative approach. The joint expert report noted that there has been no compelling evidence to change our approach.²²⁹

We have received agreement from many submissions that the foundational model approach should continue to be applied.²³⁰ This includes the use of the SLCAPM.

 Financial market practitioners, academics and other regulators consistently use the SLCAPM for estimating the expected return on equity²³¹. The Network Shareholder

- ²²⁷ Partington, G., Satchell, S., *Report to the AER: Allowed Rate of Return 2018 Guideline review, May 2018*
- ²²⁸ For example, see Energy Networks Australia, AER Review of the Rate of Return Guideline Response to Discussion Papers and Concurrent Evidence Sessions , p44, or , Australian Pipeline and Gas Association, Submission to issues Paper: AER Review of the rate of return guideline, 12 December 2017, p.2, Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, , p.8, Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p.3, Public Interest Advocacy Centre, Submission on rate of return guideline review issues paper, 18 December 2017, p.2, Network Shareholder Group, Response to issues paper on the review of the Rate of Return Guidelines, December 2017, p.9,
- ²²⁹ Cambridge Economic Policy Associates, Rate of Return Guideline Review Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, p.23
- ²³⁰ Energy Networks Australia, AER Review of the Rate of Return Guideline Response to Discussion Papers and Concurrent Evidence Sessions, p.8, Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p.3, Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p.3 Public Interest Advocacy Centre, Submission on rate of return guideline review issues paper, 18 December 2017, p.2 Network Shareholder Group, Response to issues paper on the review of the Rate of Return Guidelines, December 2017, p.9

²²³ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 2 Major Energy Users, Review of the rate of return guidelines, December 2017, p 7–8.

²²⁴ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 2 The Australian Institute, Submission to the review of the rate of return guidelines, December 2017, p. 5.

²²⁵ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p.37

²²⁶ Network shareholders group, Submission on the Rate of Return Guideline review, p.9, Energy Networks Australia, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Evidence Sessions, p.8, Australian Pipeline Gas Association, Submission to the AER, Review of rate of return guideline, 4 May 2018, p.2

Group and Energy Networks Australia submissions following the concurrent expert sessions supported this view.²³²

- The SLCAPM reflects the risk-return relationship in a clear and simple relationship. We have not received robust evidence that service providers are uncorrelated with the market.
- Has well-accepted and unbiased methods for estimating its parameters, and these
 parameters can be estimated with tolerable accuracy.²³³ The availability and
 potential for bias in analysis was highlighted as a concern by the National Seniors
 Australia.²³⁴
- We have regard to the prevailing conditions in the market for equity funds.²³⁵ We use other relevant sources of information to cross-check the foundation model estimate. The triangulation of estimates from relevant market participants broadly supports our foundation model estimate of the return on equity.

Stephen Gray, Greg Houston and Illan Sadeh submitted the view that the long term systematic risk is generally constant over time.²³⁶ Combined with our emphasis on long term estimation periods, the effects of short term volatility is mitigated.

 Analysis by Graham and Harvey found that the market factor proposed by Sharpe was the dominant factor in asset pricing models.²³⁷

We note that Stephen Gray and Energy Networks Australia expressed the view that the rate of return should be set above the SLCAPM estimate to reflect potential shortfalls over time.²³⁸ However, it is not clear that this is warranted on the basis of expost performance. We set an ex-ante return on equity that efficiently compensates for the risks of supplying regulated energy network services over a relevant regulatory period.

We note that the CCP proposed that the rate of return should be adjusted for upside risks in the regulatory framework (for example incentive schemes).²³⁹ However, we note that firms benefit/loses from incentive schemes on an ex-post basis. As we set the

²³⁴ National Seniors Australia, Issues Paper: Review of the rate of return guidelines, December 2017, p.2

²³¹ See AER, *Explanatory statement to the rate of return guideline (appendices)*, 17 December 2013, pp. 12–13.

²³² Network Shareholder Group submission to the Rate of return guideline review, 4 May 2018, p 9, Energy Networks Australia, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Evidence Sessions, p. 8

²³³ Partington, G., Satchell, S., *Report to the AER: Allowed Rate of Return 2018 Guideline review*, May 2018, p.12

²³⁵ NER, cl 6.5.2(g); NER, cl 6A.6.2 (g); NGR, r. 87 (7).

²³⁶ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.22

²³⁷ Graham, J. and Harvey C. (2001) The Theory and Practice of Corporate Finance: Evidence from the Field, *Journal* of *Financial Economics* 60.

²³⁸ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 44, Energy Networks Australia, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Evidence Sessions, p.30

²³⁹ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p.43

rate of return on an ex-ante basis, we do not consider incentive schemes warrant inclusion.

We note that David Johnstone challenged the use of the SLCAPM²⁴⁰. However, no alternative was presented as a replacement.

Weighted Average Cost of Capital

The National Gas and Electricity Rules require us to have regard to relevant estimation methods, financial models, market data and other evidence. The Rules also highlight the desirability of consistent application of financial parameters that are relevant or common to the return on equity and debt.²⁴¹

We use an estimated weighted average return on capital (equity and debt) that a benchmark efficient entity would require to finance investment in its network.²⁴² The WACC model utilises market data, and allows the incorporation of equity and debt models to predict an average of the return on equity and debt, fulfilling these requirements.

We have received submissions that suggest that the use of WACC is inappropriate:

- Consumer groups submitted that the model is set in a competitive market and does not reflect the risk environment of a regulated market.²⁴³
- Evoenergy and Energy Networks Australia submitted that we should identify risks relevant to the allowed rate of return and assess how each risk is addressed in the framework.²⁴⁴
- Public Interest Advocacy Centre has also submitted that we should examine risks through a bottom up analysis.²⁴⁵

We continue to be of the view that the WACC model is appropriate to estimate the allowed rate of return for the following reasons:

• We note that, regardless of the industry, firms need to offer a competitive (that is market-based) rate of return to attract and retain investor funds. Further, as discussed in section 2.4.3, the regulatory framework can mitigate the impact of (systematic) risks but does not completely eliminate these risks.

²⁴⁰ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.47

²⁴¹ NER, cl. 6.5.2(e), NER cl, 6A.6.2 (e); NGR, r. 87(5).

²⁴² The term service provider relates to service providers that provide gas and electricity transmission and distribution services.

²⁴³ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 1, Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 36, Canegrowers, Submission to AER review of the rate of return guideline, December 2017, p.4, Major Energy Users, Review of the rate of return guidelines, December 2017, p 7

²⁴⁴ Energy Networks Australia – AER Rate of Return Guidelines Response to Issues Paper – 12 December 2017, p.7, Evoenergy, Review of rate of return guideline –evidence sessions, 4 May 2018, p.2,

²⁴⁵ Public Interest Advocacy Centre, Submission on rate of return guideline review issues paper, 18 December 2017, p. 2.

- The WACC model avoids the need to calculate the required compensation for each risk faced by the networks and the assessment of whether each risk is systematic.
- We have considered the risks affecting firms in the supply of regulated energy network services as part of our conceptual consideration of risk-this is not dissimilar to Public Interest Advocacy Centre's proposal. We concluded that the systematic risk for a benchmark efficient entity with a similar degree of risk as a service provider in the provision of regulated energy network services would be below the market average of 1.0 (see section 8.3.1 for more detail). We have identified the key risk affecting the return on equity is systematic risk. This is measured by the equity beta parameter in the SLCAPM and it is standard practice to use market data to estimate this parameter.
- We consider employing a rate of return that is commensurate with the prevailing market cost of capital (or WACC) is consistent with the zero NPV investment condition (see above). Experts at our concurrent evidence sessions agree that only systematic risk is explicitly compensable through WACC and non-systematic risks are reflected through expected cash flows.²⁴⁶ There was also agreement that consumer interests are best achieved by setting the allowed return to be the opportunity cost of capital for investors²⁴⁷ and that WACC estimates the expected return that investors would require.²⁴⁸
- In setting the allowed return on debt, we provide the efficient compensation for the risks that an investor in the service provider's debt face. These risks are included in the promised returns we observe using our debt data sources.²⁴⁹ Further, since we provide a return on debt based on the promised yield, our allowed rate of return will be slightly above the expected return. This also means the overall allowed return includes some compensation for the expected default loss on debt due to non-systematic risk. The key risks for debt holders are systematic risk, credit risk (the risk of default and credit rating downgrades) and liquidity risk.²⁵⁰

2.4.2 Gas and electricity

²⁴⁶ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, Page 23

²⁴⁷ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, Page 10-11

²⁴⁸ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, Page 23

²⁴⁹ We observe the promised returns of debt issued by a sample of firms we consider comparable to a benchmark efficient entity based on the benchmark credit rating and term. In practice, we may have overcompensated a benchmark efficient entity for these risks as we observe broad BBB debt whereas we consider a benchmark efficient entity would issue BBB+ debt.

²⁵⁰ McKenzie, M., Partington, G., *Risk, asset pricing models and WACC*, June 2013, p. 14.

We estimate the efficient financing costs of a BEE that has a similar degree of risk as that which applies to the service provider in respect of the provision of regulated services.²⁵¹

If the systematic risk of providing different network services by gas and electricity networks are different then we may need to recognise different benchmarks. In assessing whether more than one benchmark is required, the key issue is whether there is a difference in risks between gas and electricity regulated network services.

We have received divergent views on this topic:

- APA, Australian Pipeline and Gas Association and Ian McAuley have submitted that there are differences in risk between gas pipeline businesses and electricity network companies.²⁵²
- Most experts agreed in the second evidence session that there were no strong theoretical reasons for believing that the beta of regulated electricity and gas should be the same. Stephen Gray and Greg Houston stated that there are differences in risk between gas and electricity businesses and that they warrant different equity betas.²⁵³ However, there was no agreement if equity beta would be different for different types of businesses.²⁵⁴
- APA submitted that there are differences in the size and composition of user bases and contractual agreements between companies.²⁵⁵ For example, gas is a more dominant fuel source in industries with higher exposure to economic cycles such as construction, while electricity is primarily for residential consumers.²⁵⁶
- APGA submitted that there are differences in price elasticity of demand due to the availability of substitutes²⁵⁷.
- Experts at the evidence session mentioned a 2016 New Zealand Commerce Commission decision in support of a higher equity beta for regulated gas businesses to regulated electricity businesses.²⁵⁸

²⁵¹ Under the different rules, regulated network services for electricity transmission is 'prescribed transmission services', electricity distribution is 'standard control services' and the gas sector refers to 'reference services'. See NER, cl. 6.5.2(c), NGR r. 87(3).

²⁵² APA, AER Rate of return issues paper, 12 December 2017, p. 5, Australian Pipeline and Gas Association, Submission to issues Paper: AER Review of the rate of return guideline, 12 December 2017, p.3, Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 3,

²⁵³ AER, Concurrent evidence session 1- Proofed transcript of proceedings, 15 March 2018, Australian Pipeline and Gas Association, Submission to the AER, Review of rate of return guideline, p 17

²⁵⁴ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 49.

²⁵⁵ APA, submission responding to discussion papers and expert evidence, p.17, Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 1

²⁵⁶ APA, submission responding to discussion papers and expert evidence, p.17

²⁵⁷ Australian Pipeline and Gas Association, Submission to the AER: Review of the rate of return guideline, May 2018, p.5

 Graham Partington and Satchell's view is that it would be difficult to justify separate equity betas to different sectors of the industry and would be even more difficult to quantify this difference. The view that differences in risk will be hard to distinguish is supported by the CCP.²⁵⁹

We have not started from the position that there should be only one benchmark for both gas and electricity (or a one size fits all). Having reviewed the submissions,²⁶⁰ we do not consider they provide substantively new material or information to that considered in the 2013 Guideline and subsequent regulatory decisions:²⁶¹

We consider the regulatory framework for gas and electricity service providers are similar. Both gas and electricity service providers face limited competition risk by virtue of being regulated natural monopolies. Generally, competition risks for service providers are low. While electricity transmission service providers are required to use a revenue cap, there is a range of controls available for electricity distribution and gas service providers to propose (as part of the Framework and Approach Paper). We currently apply revenue caps to all except one electricity distributor.²⁶² The current controls for those areas can mitigate demand risk.²⁶³ Refer to

²⁵⁸ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.60, Australian Pipeline and Gas Association, submission to the issues paper, AER review of the rate of return guideline, December 2017, p.3, AER Concurrent Evidence Session 2 – Proofed Transcript, April 2018 p.40

²⁵⁹ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 82.

²⁶⁰ Energy Consumers Australia, Review of the rate of return guideline, December 2017, P.15-18

²⁶¹ Submissions in the 2013 Rate of return guideline highlighted different regulatory regimes, commercial arrangements, price elasticity of demand, user bases, and availability of substitutes (p.183-184). These topics were similar to the recent submissions. See APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p 17-18.

²⁶² We apply an average revenue cap to EvoEnergy. Going forward, Evoenergy will transition into a revenue cap for the 2019-24 regulatory control period. Evoenergy, Attachment 11: Control Mechanisms, Regulatory Proposal for the ACT electricity distribution network 2019-2024, January 2018, p.1

²⁶³ See: NER, cl. 6A.4.2 (a) (1); NER, cl. 6.2.5(b); NGR, r. 97(2).

- Table 10, in section 2.4.3.
- To the extent that there are genuine risks of extreme changes in demand for specific service providers which present the potential for stranding of an asset, the regulatory regime for gas and electricity can mitigate this risk by providing prudent discount and accelerated depreciation provisions.²⁶⁴
- Our Australian empirical analysis is based on a comparator set which includes gas service providers. Therefore, if there are differences in the systematic risks of electricity and gas service providers, this may be captured in our Australian empirical estimates of equity beta. See section 0 for a list of firms in our comparator set.

It is also not clear that the experts supported different betas for gas and electricity businesses: ²⁶⁵

- Illan Sadeh noted that differences in businesses do not necessarily translate into the rate of return, but rather the opex allowance.²⁶⁶ The point that differences may be reflected in opex was supported by Stephen Gray.²⁶⁷
- There was no agreement on whether different benchmarks were warranted.²⁶⁸ Partington noted difficulty in reliably measuring the risk differences, Johnstone noted the possibility of upside risks and Gray noted there may be discussions on whether risks are partially non-systematic.

New Zealand Commerce Commission Decision

In its 2016 review, the NZCC adopted a gas beta 0.05 higher than electricity after aggregating four factors: $^{\rm 269}$

- Results of an analysis on gas beta subsample relative to energy and electricity
- Gas having a higher income elasticity of demand than electricity
- Low gas penetration in New Zealand relative to other countries in comparator sample
- Overseas regulatory precedent

However, we do not find the NZCC's rationale to be sufficiently persuasive to warrant different equity betas for gas and electricity betas in Australia for the following reasons:

²⁶⁴ For prudent discounts, see NER, cl. 6A.26, NGR r. 96; for accelerated depreciation provisions see NER, cl. 6.5.5(b)(1), 6A.6.3(b)(1); NGR, r.89(1).

AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.57-58

AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p.63.

AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 58

²⁶⁸ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence Expert Joint Report, April 2018, p.49

²⁶⁹ New Zealand Commerce Commission, Input methodologies review decisions, topic paper 4 cost of capital issues, December 2016, Paragraph 374

The NZCC has acknowledged that "neither of these factors are sufficient in supporting an uplift in isolation".²⁷⁰

The NZCC's reasons do not appear to be relevant to the firms we regulate as

- The beta analysis was based on a comparator sample of NZ, Australian, UK and US utility firms, which included vertically integrated utilities. This conflicts with our decision to use a domestic pure-play comparator set due to differences in risk and regulatory environments.²⁷¹
- The low gas penetration is less relevant in the Australian market, with 56 per cent of Australia connected to gas compared to only 21 per cent of North Island.²⁷² NZCC viewed the low level of penetration as an indication of high expansion potential and asset stranding (deemed partly systematic).²⁷³
- It is also not clear whether gas has a higher price elasticity than electricity. The ACCC's east coast gas inquiry concluded that suppliers had market power over gas users.²⁷⁴

We note APGA's submission that the NZCC report shows the asset betas of US regulated gas businesses lie above electricity businesses.²⁷⁵ However, the NZCC report also noted the following observations:

- Ofgem uses the same equity beta for distribution and similar betas for transmission.
- European evidence also provides mixed direction, with half of the regulators in the NZCC sample use the same asset beta, or a lower asset beta for gas.²⁷⁶
- Further, NZCC notes that given differences in context, regulatory frameworks and environments, decisions by international regulatory entities provide limited benefit.²⁷⁷

2.4.3 Impact of regulation on risk

All else being equal, we have concluded in past decisions that an entity providing unregulated services in a competitive market is likely to have a higher risk and more

²⁷⁰ New Zealand Commerce Commission, Input methodologies review decisions, topic paper 4 cost of capital issues, December 2016, Paragraph 344

²⁷¹ See AER, Final decision AusNet distribution determination – attachment 3 – rate of return – May 2016, pp 38

²⁷² New Zealand Commerce Commission, Input methodologies review decisions, topic paper 4 cost of capital issues, December 2016, Paragraph 418

²⁷³ New Zealand Commerce Commission, Input methodologies review decisions, topic paper 4 cost of capital issues, December 2016, Paragraph 371.3

²⁷⁴ ACCC Inquiry into the east coast gas market, April 2016, P.18-19

Australian Pipeline Gas Association, Submission to the AER, Review of rate of return guideline, 4 May 2018, page
 5

²⁷⁶ New Zealand Commerce Commission, Input methodologies review decisions, topic paper 4 cost of capital issues, December 2016, Paragraph 434

²⁷⁷ New Zealand Commerce Commission, Input methodologies review decisions, topic paper 4 cost of capital issues, December 2016, Paragraph 442

variable expected returns than a monopoly business such as the service providers in the provision of regulated services.²⁷⁸ This is because regulation:²⁷⁹

- mitigates monopolies from being able to extract monopoly rents, thereby constraining potential profits
- increases the certainty of the revenue stream, thereby reducing risk.

This gave us insight into the equity beta for a benchmark efficient entity relative to the average equity beta across all firms in the market, which is 1.0 by definition.²⁸⁰

We maintain the view that incentive regulation allows service providers to earn more stable cash flows with periodic resets of revenues reflecting changes in actual expenditure.²⁸¹ As most unregulated businesses do not have the same protections or restrictions, they are likely to face different risk environments.²⁸²

Frontier has also recognised the role of regulation in affecting risk in advising:283

The form and nature of regulation applicable to Australian energy networks mitigates most of the business risks they face as compared to the business risks faced by other types of firms in the economy. Regulated revenues are set on a periodic basis and changes in volumes may only affect the timing of revenues (under a revenue cap). Even where revenues fall short of expectations due to lower volumes (as under a price cap), the lower volumes imply that costs would probably also have been lower than expected. Unanticipated or poorly-managed changes in costs are partly borne by customers and only partly by the network business through the building block form of incentive regulation that applies. Stranding and optimisation risks are minimal for energy networks, a complete contrast to businesses operating in other sectors.

For example see: AER, Better regulation explanatory statement rate of return guideline, December 2013, pp. 36–46; AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 24.

 ²⁷⁹ For example see: AER, Better regulation explanatory statement rate of return guideline, December 2013, pp. 36–46; AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 24.

²⁸⁰ More precisely, the value weighted average across all firms in the market is 1.0. As pointed out by McKenzie and Partington, the equal weighted average may not be 1.0, since larger firms may be unevenly distributed above or below 1.0. See: McKenzie and Partington, *Estimation of the equity beta (conceptual and econometric issues) for a gas regulatory process in 2012*, April 2012, p. 21. (McKenzie and Partington, *Estimation of equity beta*, April 2012)

²⁸¹ For example see: AER, Better regulation explanatory statement rate of return guideline, December 2013, pp. 36–46; AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 25.

²⁸² For example see: AER, Better regulation explanatory statement rate of return guideline, December 2013, pp. 36–46; AER, Better regulation explanatory statement rate of return guideline (appendices), December 2013, pp. 39–46; AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 25.

²⁸³ Frontier Economics, Assessing risk when determining the appropriate rate of return for regulated energy networks in Australia, July 2013, p. 4.

For clarity, regulation of the kind embodied in the national electricity and gas legislation reduces risks compensated through the rate of return (for example, demand risk). Regulation also reduces uncompensated risks for example, by allowing cost pass throughs for non-systematic risks such as industry-specific tax changes or geographic-specific natural disasters.

We have previously determined that regulation of energy network services reduces compensable risks such as:

- Demand risk: the revenue or price setting mechanism mitigates demand risk. Under a price cap, service providers may mitigate the risk of forecast e by restructuring tariffs, such that higher fixed charges are set to offset falls in demand. Under a revenue cap, where forecast quantity demanded differs from actual quantity demanded, service providers have the possibility to recover for variation through price adjustments in subsequent years.
- Inflation risk: service providers of regulated energy network services face less inflation risk than unregulated businesses, as movements in actual inflation are reflected in the CPI-X mechanism. We reviewed our treatment of inflation in 2017²⁸⁴, after receiving stakeholder submissions on the issue.
- Interest rate risk: The regulatory framework effectively moves risk of interest rate movements affecting financing costs onto customers. Service providers may further limit their exposure to this risk by raising capital during the averaging period/s that they know in advance.²⁸⁵ To the extent they are unable to raise capital over the averaging periods; they can still materially reduce their exposure to interest rate risk by hedging the base rate.

AER – Final Position paper – Regulatory treatment of inflation – December 2017. Available at:

https://www.aer.gov.au/networks-pipelines/guidelines-schemes-models-reviews/review-of-expected-inflation-2017
 Service providers are able to propose future averaging periods for the returns on debt and equity as part of their regulatory proposal for the upcoming regulatory period.

Table 10 below summarises a selection of provisions in the NER and NGR that we consider likely to mitigate various systematic and non-systematic risks.

Table 10 Key clauses in the NGR and NER that mitigate systematic risk

NER	NGR	Effect on risk
6.3.2(b)	50	The term of each regulatory control period is at least 5 years, providing a fixed duration in which a service provider has a regulated return on its assets, cashflow certainty, and fixed terms of access for its services.
6.2.6, 6.5.9	92	This control mechanism automatically accounts for indexation and annual increases in efficient costs. Smooths cashflows from year to year to provide stable level of cash flow, reducing risks of short-term revenue.
6.18	97(5)	The prices service providers may charge annually are certain.
6.4.3(a)(1)- (3), 6.5.1, 6.5.2, 6.5.5, S6.2.1, S6.2.2B, S6.2.3,	76, 77, 78, 87(1), 90	The cashflow that the AER determines incorporates a return on and of the service provider's asset base. The historical asset base rolls forward from one regulatory control period to the next and from year to year within each regulatory control period. Guarantees recovery of historical asset costs through depreciation, the earning of a return on the asset base, indexation and recovery of future efficient capex. This substantially lessens risks in capital investment that might otherwise apply to a business operating in a workably competitive market.
6.5.2	87	The AER sets the rate of return on the asset base by reference to the risks faced by the service provider. The AER updates this each regulatory control period to account for changed market conditions.
6.5.3	87A	Provision for tax in determining total revenue is required regardless of whether the service provider pays tax.
6.5.6, 6.5.7	79, 91	The AER assesses expenditure requirements for each service provider by reference to the amount necessary to meet a set of standards and objectives. These include the need to meet the expected demand for services and to meet quality, reliability, security, and safety standards. The AER does not assess expenditure by reference to the capacity of consumers to pay. This removes risks that could otherwise arise in providing a reliable and safe service. The AER reassesses the requirements of service providers for each regulatory period to account for changes in market conditions and trends.
6.5.10	97 (1)(c)	Allows service providers to pass through certain costs to consumers in circumstances where this might not be possible in a workably competitive market. For instance, the pass through provisions provide for a pass through of costs that arise through regulatory change.
6.5.7(f), 6.6A, chapter 5	80-86, and 103- 104	Assists in appropriate planning for changes in the commercial environment, including provision for new projects during a regulatory period.
6.20, 6.21, 6.6.1(a1)(d), and RoLR provisions	Parts 19- 21	Provides for a statutory billing and settlements framework with prudential requirements (and other similar provisions) to minimise financial risk associated with providing and charging for services. There is also provision for dealing with potential risks associated with retailer insolvency.
6.6.5, 6A.7.1		Provides an opportunity to apply for a reopening of a determination for capital expenditure if an event that is beyond reasonable control of the service provider and the occurrence of the event could not have reasonably been foreseen by the service provider at the time of the making of the determination.

Source: NER & NGR, AER analysis.

We have received submissions indicating differing opinions on the impact regulation has on the systematic risks faced by service providers. Consumer groups submitted
the view that regulation reduces risks while increases financial benefits.²⁸⁶ APA submitted that more analysis is required before the characteristics of electricity networks and gas pipelines can be shown to operate with low betas.²⁸⁷

We have therefore, looked at the impact of regulation on empirical equity beta estimates. The figure below shows a general trend of increasing beta estimates as the proportion of regulated revenue decreases. This is consistent with the conclusion that regulation lowers a firm's equity beta estimate.



Figure 9 Regulated revenue and beta estimates

Source: Bloomberg; AER analysis

The division between regulated and unregulated is generally based on the most recent publicly available information on the proportion of regulated and non-regulated activities for a financial year. SKI is based on page 90 of its 2017 Annual Report; DUET is based on its 2016 Annual Report;²⁸⁸ Envestra is based on page 5 of its 2013 Annual Report; AST is based on page 75 of its 2017 Annual Report; GasNet is based on page 3 of its report for the half-year ended 30 June 2016; AAN is based on page 24–25 its 2006 Annual Report;

²⁸⁶ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p.28, Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p. 8-9. National Seniors Australia, Issues paper: Review of the rate of return guidelines, December 2017, p.2-3

²⁸⁷ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p.16,

²⁸⁸ In our estimation, Dampier to Bunbury Natural Gas Pipeline (DBNGP) is 100 per cent regulated even though DUET's 2016 Annual Report reported that the ERA's regulatory tariff applies to 15 per cent of the capacity to 2020. This is because the ERA still regulates the pipeline and its regulatory tariffs would apply in the absence of renegotiated Standard Shipper Contracts as indicated by DUET (page 11 of 2016 Annual Report).

AGL is based on page 68 its 2006 Annual Report; APA is based on page 16 of its 2017 Annual Report; HDF is based on page 10 of its 2011 Annual Report.

Business risk

Business risk in this context refers to the systematic risk exposure of the underlying business assets.²⁸⁹ It is generally accepted that a benchmark efficient entity has lower business risk than the market average firm.²⁹⁰ We maintain our previous view that business risk for a benchmark efficient entity will be low for the following reasons:²⁹¹

- There are a number of inherent characteristics of an energy transportation network that lead to low systematic risk exposure. For example, operation of a natural monopoly and provision of an essential service with low price elasticity of demand.
- The structure of the regulatory regime insulates service providers from systematic risk. For example, this provides for revenue cap regulation, tariff variation mechanisms and cost pass through mechanisms. This also provides for tariff structures that include fixed charges and protection of sunk investment through rolling forward the regulatory asset base (RAB).

In their 2012 report to the AER, McKenzie and Partington divided business risk into intrinsic risk, and operational risk.²⁹² Intrinsic risk describes how the business cycle impacts on a firm's sales and operational risk relates to a firm's operating leverage (the proportion of fixed to variable costs).

McKenzie and Partington previously considered that operational risk for the efficient entity would be above the market average given the high proportion of fixed costs for energy networks.²⁹³ However, the overall business risk would be low because a benchmark efficient entity could mitigate the effect of this cost structure through the

We note business risk in this context is only systematic/market risk and does not include firm specific risk that can be diversified away.

²⁹⁰ McKenzie and Partington, Estimation of equity beta, April 2012, pp. 6, 10; SFG, Equity beta: Report for Jemena Gas Networks, ActewAGL and Networks NSW, May 2014, pp. 17–18. (SFG, Equity beta, May 2014); SFG, Estimating the required return on equity: Report for Energex, 28 August 2014, p. 60; SFG, Beta and the Black capital asset pricing model: Report for Jemena Gas Networks, Jemena Electricity Networks, ActewAGL, Ausgrid, Ausnet Services, Australian Gas Networks, CitiPower, Endeavour Energy, Energex, Ergon, Essential Energy, Powercor, SA Power Networks and United Energy, 13 February 2015, p. 42 (SFG, Beta and the Black capital asset pricing model, 13 February 2015); SFG, Equity beta report prepared for APT Petroleum Pipelines Ltd, October 2011, p. 11; McKenzie and Partington, Report to the AER: Risk, asset pricing models and WACC, June 2013, p. 11; Frontier Economics, Assessing risk for regulated energy networks, July 2013, p. 64. McKenzie and Partington, Report to the AER, Part A: Return on equity, October 2014, p. 11. Origin Energy, Submission to NSW distribution network service providers regulatory proposals for 2014–19, August 2014, p. 7.

²⁹¹ AER, Explanatory statement to the rate of return guideline (appendices), December 2013, pp. 40–41. Also see: Frontier Economics, Assessing risk for regulated energy networks, July 2013; McKenzie and Partington, Estimation of equity beta, April 2012, p. 6.

²⁹² McKenzie and Partington, *Estimation of equity beta*, April 2012, pp. 5–6. See also: McKenzie and Partington, *Report to the AER: Risk, asset pricing models and WACC*, June 2013, p. 11.

²⁹³ McKenzie and Partington, *Estimation of equity beta*, April 2012, pp. 7, 14.

use of fixed charges. Similarly, intrinsic risk would be very low because a benchmark efficient entity is insulated from the business cycle for reasons listed above.

McKenzie and Partington concluded that the intrinsic risk of a firm is the 'primary, if not sole, driver of its systematic risk.' ²⁹⁴ They reiterated this view in their subsequent reports. ²⁹⁵

Financial risk

Financial risk relates to the increased systematic risk for equity holders arising from debt holdings, since debt payments take precedence over equity payments. Given their low risk cash flows, service providers might issue a higher proportion of debt than if they were operating in a competitive market.²⁹⁶ This reduces their cost of capital if debt is cheaper than equity.

It is generally accepted that a benchmark efficient entity has higher financial risk than the market average firm does.²⁹⁷ The key cause is the relatively high financial leverage for a benchmark efficient entity (60 per cent), relative to the market average firm (about 30 to 35 per cent).

However, the high financial leverage does not necessarily result in an equivalently high financial risk:

- For instance, in their 2014 (and 2015) report, McKenzie and Partington noted for energy network businesses, the likelihood of bankruptcy as leverage increases is low (to the extent that the business is able to pass on borrowing costs to consumers).²⁹⁸
- In their 2013 report, McKenzie and Partington also noted that, given the low default risk in regulated energy network businesses, the financial risk effects are 'unlikely to be substantive in normal market conditions'.²⁹⁹
- Frontier has also acknowledges that our implementation of a trailing average approach would reduce interest rate reset risk.³⁰⁰

Overall risk

²⁹⁴ McKenzie and Partington, *Estimation of equity beta*, April 2012, p. 14

²⁹⁵ McKenzie and Partington, *Report to the AER, Part A: Return on equity*, October 2014, p. 12; Partington, *Report to the AER: Return on equity (Updated)*, April 2015, p. 32.

²⁹⁶ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, May 2018, P.4

²⁹⁷ McKenzie and Partington, *Estimation of equity beta*, April 2012, pp. 7, 10; SFG, *Equity beta*, May 2014, pp. 17–18; SFG, *Estimating the required return on equity: Report for Energex*, 28 August 2014, p. 60; SFG, *Beta and the Black capital asset pricing model*, 13 February 2015, p. 42; SFG, *Equity beta report prepared for APT Petroleum Pipelines Ltd*, October 2011, p. 11.

²⁹⁸ McKenzie and Partington, *Report to the AER, Part A: Return on equity*, October 2014, p. 11; Partington, *Report to the AER: Return on equity (Updated)*, April 2015, pp. 31–32.

²⁹⁹ McKenzie and Partington, *Report to the AER: Risk, asset pricing models and WACC*, June 2013, pp. 11–12.

³⁰⁰ Frontier Economics, Assessing risk for regulated energy networks, July 2013, p. 74.

We maintain the view that the above assessment suggests that the intrinsic business risk of a firm is the main driver of its systematic risk. We expect a benchmark efficient entity to have low intrinsic risk exposure (relative to the market average). We also consider the high financial leverage of a benchmark efficient entity (relative to the market average) does not necessarily correspond to an equivalently high exposure to financial risk.

The Consumer Reference Group submitted that risks are substantially reduced by regulation and are being continuously reduced and reviewed.³⁰¹ It noted, as an example, the recent network tariff reform under the AEMC's Demand Side Participation Review increasing inter-annual and intra-annual revenue smoothing. We acknowledge the CRG's view that risks will continually be reviewed in the future, and that this may impact risks in the future.

Based on this information, we consider there are reasonable conceptual grounds to expect the overall systematic risk for a benchmark efficient entity to be below that of the market average firm. This leads to our expectation that the equity beta of a benchmark efficient entity will be below 1.0. This view is supported in Partington and Satchell's latest (2018) report to the AER. ³⁰²

2.4.4 Interrelationships

In determining the allowed rate of return, we must have regard to any interrelationships between estimates of financial parameters that are relevant to the estimates of the return on equity and the return on debt.³⁰³ The 2013 Guideline described these interrelationships in detail where we have had regard to them in developing our approach.³⁰⁴

We maintain the view that one should not view any component or relevant parameter adopted for estimating the rate of return in isolation.³⁰⁵ This view supported in recent submissions by the CCP16 and NSG.³⁰⁶ In developing our approach and implementing it to derive the overall rate of return, we are cognisant of a number of interrelationships relating to the estimation of the return on equity and debt and underlying input parameters.

³⁰¹ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p.28.

³⁰² Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, May 2018, P.3.

³⁰³ NER, cl. 6.5.2(e); NER, cl. 6A.6.2 (e); NGR r. 87(9).

 ³⁰⁴ For example, see: AER, Better regulation explanatory statement rate of return guideline, December 2013, pp. 15, 20, 158; AER, Better regulation explanatory statement rate of return guideline (appendices), December 2013, pp. 12, 25–26, 51, 78, 166

³⁰⁵ For example, see: AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 40

 ³⁰⁶ Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p.12, Consumer
 Challenge Panel, Submission to the AER on its Rate of Return Guideline review concurrent evidence sessions, p.
 6

Importantly, the principles set out in this paper regarding the efficient compensation of risk through the allowed rate of return should be applied consistently in the estimation of all rate of return parameters. However, while agreed principles should be applied consistently, the availability of particular data may mean that the consistent application of these principles may result in different datasets being used for different parameters.

2.5 Summary of submissions

This section details the submissions we have received from stakeholders as part of our review process to date, and notes how we have had regard to each submission.

Key Point	Submission	Stakeholder	AER response
Different Beta for gas and electricity	Gas and electricity businesses face different risks	APA, ³⁰⁷ Greg Houston ³⁰⁸ , Ian McAuley ³⁰⁹	We view that while gas and electricity businesses may face different risks, the regulatory framework mitigates risk sufficiently so that the differences in residual risk after regulation are immaterial. While there is no theoretical reason that the beta of regulated gas and electricity should be the same, there was no agreement from the concurrent evidence session on whether equity beta should be different for different types of businesses. Experts noted that differences in businesses do not necessarily translate into rate of return but potentially opex allowance.
	Differences in gas and electricity risks cannot be measured reliably	Partington ³¹⁰ CCP. ³¹¹	
	There are upside risks to consider.	Johnstone ³¹²	
	Differences in risk may be both systematic and non- systematic and may be differences in Opex.	Gray ³¹³	
	Empirical evidence from NZCC's 2016 Decision provides new support for a gas differential.	APGA ³¹⁴	
	Differences in businesses (such as labour costs) do not have to translate into rate of return, but rather opex.	Sadeh ³¹⁵	
	No theoretical reason that beta of regulated gas and electricity should be the same. Nor is this required by the NER or the NGR	APGA ³¹⁶ ,	based on a comparator set which includes gas service providers, which will capture any potential differences in systematic
	NZCC shows the asset	APGA ³¹⁷	electricity service

Table 11 Summary of submissions on compensable risk

- ³⁰⁷ APA, AER Rate of return issues paper, 12 December 2017, p. 5
- ³⁰⁸ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 56
- ³⁰⁹ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 1
- ³¹⁰ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018. p. 57
- ³¹¹ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 82.
- ³¹² AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 57
- ³¹³ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 58
- ³¹⁴ Australian Pipeline and Gas Association, Submission to issues Paper: AER Review of the rate of return guideline, 12 December 2017, p.3
- ³¹⁵ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 58
- ³¹⁶ Australian Pipeline and Gas Association, Submission to the AER, Review of rate of return guideline, 4 May 2018, p. 17

Key Point	Submission	Stakeholder	AER response
	betas of US regulated gas		providers.
	electricity businesses.		We reviewed the NZCC decision document and do not view it to be relevant to our task. The NZCC itself highlighted that the factors supporting an uplift in the gas beta are not sufficient in isolation. Additionally, the decision appears to be based on a number of NZ specific factors.
	Transmission and distribution pipelines service different user bases. The risks of service provision using transmission pipelines are different from those of distribution pipelines with their extensive, and usually diversified, end-user populations. The costs of financing pipeline investment are therefore different across transmission pipelines, and between transmission pipelines and gas distribution pipeline systems.	APA ³¹⁸ Ian McAuley ³¹⁹	
	Issue was considered in 2013, no new evidence presented	CCP ³²⁰ ECA ³²¹	Did not include due to new evidence of NZCC.
Technological risk	Emerging tech has increased structural changes in energy markets	Cheung Kong Infrastructure ³²²	We consider technological, risks are not compensable through the rate of return
	2013 Guidelines were based on assumption that energy markets face limited competition	ENA ³²³	because they are not systematic and are diversifiable. To the extent that there are impacts on systematic risks, beta will reflect the systematic component.
	Seems diversifiable	Partington ³²⁴ , CCP16 ³²⁵	
	Potential to be covered in OPEX	Gray ³²⁶	Additionally, we agree with the CCP that should these

- ³¹⁸ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p.17
- ³¹⁹ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p.1
- ³²⁰ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 45
- ³²¹ Energy Consumers Australia, Review of the rate of return guideline, December 2017, p.15-18
- ³²² Cheung Kong Infrastructure, Submission to issues paper: AER Review of rate of return guideline, December 2017, p. 34
- ³²³ Energy Networks Australia, Submission to issues paper: AER Review of rate of return guideline, December 2017, p. 7
- ³²⁴ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 47
- ³²⁵ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 37
- ³²⁶ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 52

³¹⁷ Australian Pipeline and Gas Association, Submission to the AER, Review of rate of return guideline, 4 May 2018, p. 5

Key Point	Submission	Stakeholder	AER response
	Investors haven't factored in technology risks because they don't know.	Sadeh ³²⁷	risks result in stranding risk, the risks are borne by consumers.
	There is upside risks	Gray ³²⁸ , Johnstone ³²⁹	
	Some elements may affect markets as a whole and should be reflected	Sadeh ³³⁰	
	If systematic component will come through beta, we have to look at the more recent evidence	Gray ³³¹	
	Borne by consumers rather than service providers as assets cannot be stranded.	CCP16 ³³²	
	Tech risk identified partially due to attitude to reduce carbon emission and unrelated to economic cycle	CCP16 ³³³	
	Investors in other industries affected by changes in tech do not receive assistance.	CCP16 ³³⁴	
	The view that technological risk is systematic is unproven.	CCP16 ³³⁵	
	Any change in systematic risk from tech will take time to reflect in beta.	Evoenergy ³³⁶ APA ³³⁷	
	Tech risk will affect gas businesses too.	APA ³³⁸	
	Technological risks may be	Jemena ³³⁹	

- ³²⁷ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 50
- ³²⁸ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 53
- ³²⁹ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 58
- ³³⁰ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 18
- ³³¹ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 54
- ³³² Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 37
- ³³³ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 44
- ³³⁴ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 44
- ³³⁵ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 44-46
- ³³⁶ Evoenergy, Review of rate of return guideline –evidence sessions, 4 May 2018, p. 2
- ³³⁷ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p. 16
- ³³⁸ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p. 16

Key Point	Submission	Stakeholder	AER response
	non-systematic and may require compensation through cash flows		
Catastrophic risks	Entities can buy insurance, passing into operating costs.	Houston ³⁴⁰ Ian McAuley ³⁴¹	We consider catastrophic risks are not compensable through the rate of return because they are not systematic and are diversifiable. The risks can also be mitigated by the option to purchase insurance, with costs flowing through to Opex allowance. In extreme situations, catastrophic risks can be mitigated via the potential to pass the costs to relevant users and shipwreck clauses.
	Compensating for risks unrelated to the market would be departing from the CAPM framework	Johnstone ³⁴²	
	Non-systematic risks can be captured elsewhere in framework, e.g. Opex. Otherwise should be	NSG ³⁴³	
	compensated in AROR		
	Entities can buy insurance, passing into operating costs.	Houston ³⁴⁴	
	Ensure all risks are accounted for in establishing returns required by investors whether through beta, cash flow or the regulatory framework.	NSG ³⁴⁵ , Gray ³⁴⁶ , Houston ³⁴⁷ ENA ³⁴⁸	We acknowledge that risks will be continually reviewed in the future, and that this may impact risks in the future.
Regulatory risk	Australian Governments have increased threat of intervention. If risks are not reflected in rate of return, there will be impacts to investment.	NSG ³⁴⁹ APA ³⁵⁰	We consider regulatory risks are not compensable through the rate of return because they are not systematic and are diversifiable. To the extent
	Compensation for policy changes is a decision for	CCP ³⁵¹ Ian McAuley ³⁵²	compensation is deemed necessary, we consider it

- ³³⁹ Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p. 3
- ³⁴⁰ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 59
- ³⁴¹ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p.3
- ³⁴² AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 48
- ³⁴³ Network Shareholder Group, Submission on the RoRG review, May 2018, p.8
- AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 59
- $^{\rm 345}$ $\,$ Network Shareholder Group, Submission on the RoRG review, May 2018, p.10 $\,$
- AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 9
- ³⁴⁷ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 19
- ³⁴⁸ Energy Networks Australia, AER Review of the Rate of Return Guideline Response to Discussion Papers and Concurrent Evidence Sessions, May 2018, p. 7
- ³⁴⁹ Network Shareholder Group, Submission on the RoRG review, May 2018. P.7
- $^{\rm 350}$ $\,$ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p.16 $\,$
- ³⁵¹ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 45
- ³⁵² Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 3

Key Point	Submission	Stakeholder	AER response
	government, not the regulator.		should be determined by the government.
	Prospect of rule change is highly uncertain. Need to go through separate powers between AER, AEMC and ministerial council and governments.	CCP ³⁵³	Additionally, we agree with CCP's submission that the argument that policy risks are systematic is tenuous and unproven.
	Political, regulatory and sovereign risk is low.	ECA ³⁵⁴	
WACC/ CAPM framework	CAPM is not appropriate to model risk	Johnstone ³⁵⁵	We are of the view that the SLCAPM framework is the
	Current model and rules should be maintained.	NSG ³⁵⁶ ENA ³⁵⁷ APGA ³⁵⁸	 best approach to reflect the systematic risk of a BEE. We consider efficient financing costs are reflected in the prevailing market cost of capital (or WACC) for an investment with a similar degree of risk as that which applies to a service provider in respect of the provision of regulated energy network services. Beta and systematic risk are relatively stable over the long term. This view is shared by a variety of stakeholders and the majority of experts. We acknowledge that short term fluctuations may arise from risks unrelated to that of a service provider, hence we place more weight on long term
	Support WACC, SLCAPM, BEE. These principles lowers cost of capital	NSG ³⁵⁹	
	A more fundamental review is required. Major issues with CAPM. Likely to overestimate the required rate of return as it ignores revenue from other sources (such as incentives).	CRG ³⁶⁰	
	CAPM draws from competitive markets and reflects competitive markets, not monopolistic regulated markets	CRG ³⁶¹ Ian McAuley ³⁶² Canegrowers ³⁶³ MEU ³⁶⁴	
	Whether using an equity beta based on the relative volatility of share prices in the stock market (where investors face all of the	CRG ³⁶⁶	

- ³⁵⁴ Energy Consumers Australia, Review of the rate of return guideline, December 2017, p. 26
- ³⁵⁵ AER Concurrent Evidence Session 1 Proofed Transcript, April 2018, p. 47
- ³⁵⁶ Network Shareholder Group, Submission on the RoRG review, May 2018, p. 8

³⁵⁷ Energy Networks Australia, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Evidence Sessions, May 2018, p. 8

- ³⁵⁸ Australian Pipeline and Gas Association, Submission to the AER, Review of rate of return guideline, 4 May 2018, p. 2
- ³⁵⁹ Network Shareholder Group, Submission on the RoRG review, May 2018, p. 7
- ³⁶⁰ Consumer Reference Group, Submission to the AER RoRG review, May 2018, p. 37
- ³⁶¹ Consumer Reference Group, Submission to the AER RoRG review, May 2018, p. 36
- ³⁶² Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 3
- ³⁶³ Canegrowers, Submission to AER review of the rate of return guideline, December 2017, p. 4
- ³⁶⁴ Major Energy Users, Review of the rate of return guidelines, December 2017, p. 7

³⁵³ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 44

Key Point	Submission	Stakeholder	AER response
	systematic risks) is an appropriate measure to assess the underlying risk faced by the network firm where these systematic		estimation periods. It is not clear that we should adjust the rate of return above or below the SLCAPM estimate to reflect potential shortfalls as the adjustment reflects
	risks have been mitigated through the rules.365		
	AER should examine risks through a bottom up analysis.	PIAC ³⁶⁷	ex-post performance.
	The stocks used in AERs sample are considered defensive. Volatility may be market sentiment more than fundamental risk.	CRG ³⁶⁸	
	SLCAPM is an average expected return. Allowed return should be increased to cover for uninsurable risks.	ENA ³⁶⁹	
	Need to consider upside risks. Such as incentive bonuses schemes.	CCP ³⁷⁰	
	Beta is an indicator of volatility, and irrelevant over long term.	Ian McAuley ³⁷¹	
	A Beta can be 0	Ian McAuley ³⁷²	
	There is potential for bias in analysis and some useful data is unavailable due to imperfect information.	NSA ³⁷³	
	The CAPM implies that the returns the network firm generates from its normal activities would have to, at least, match the returns that	CRG ³⁷⁴	

- ³⁶⁶ Consumer Reference Group, Submission to the AER RoRG review, May 2018, p. 36
- ³⁶⁵ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 36
- ³⁶⁷ Public Interest Advocacy Centre, Submission on rate of return guideline review issues paper, 18 December 2017, p. 2
- ³⁶⁸ Consumer Reference Group, Submission to the AER RoRG review, May 2018, p. 47
- ³⁶⁹ Energy Networks Australia, AER Review of the Rate of Return Guideline Response to Discussion Papers and Concurrent Evidence Sessions, May 2018, p. 30
- ³⁷⁰ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 43
- ³⁷¹ Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 2
- ³⁷² Ian McAuley, Submission to AER on rate of return guidelines, December 2017, p. 2
- ³⁷³ National Seniors Australia, Issues Paper: Review of the rate of return guidelines, December 2017, p. 2
- ³⁷⁴ Consumer Reference Group, Submission to the AER RoRG review, May 2018, May 2018, p. 37

Key Point	Submission	Stakeholder	AER response
	firm might gain from investing in the stock market. This concept is flawed.		
BEE	BEE is important to the framework. Opportunity to outperform is crucial to incentives.	Sadeh ³⁷⁵ Houston ³⁷⁶ NSG ³⁷⁷	
	No such thing as allowed return equal to efficient cost of a BEE, as efficient cost of financing depends on characteristics of the cash stream, which is determined by the regulator.	Johnstone ³⁷⁸	We consider the concept of a BEE is an important reference point for our regulatory task.
	Unsure what efficient financing structure of a BEE is. You can't rebalance every 5 years. Some firms make the choice to hedge, some don't.	Partington ³⁷⁹	We define a BEE as having a similar degree of risk as a service provider in the provision of regulated energy network services. We use market data for
	Important to judgement/discretion. Benchmarking removes the AER from making decisions and getting involved.	Sadeh ³⁸⁰ , Johnstone ³⁸¹	firms that we consider are reasonable and closest comparators of a BEE.
	Companies used in estimating a BEE have unregulated revenues. This affects risk reductions.	CRG ³⁸² MEU ³⁸³	
Regulation mitigates risk	the very substantial risk reductions under the rules mean that economic and financial risk (the top left boxes) for regulated entities	CRG ^{384 385} EUAA ³⁸⁶ NSA ³⁸⁷	We agree that regulation mitigates substantial amounts of systematic risks and have analysed

³⁷⁹ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 20

- ³⁸² Consumer Reference Group, Submission to the AER RoRG review, May 2018, p. 26
- Major Energy Users, Review of the rate of return guidelines, December 2017, p. 7-8
- ³⁸⁴ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 28.

AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 13

AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 13

 $^{^{\}rm 377}$ Network Shareholder Group, Submission to evidence session, May 2018, p. 5

AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 18

AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 12

³⁸¹ AER Concurrent Evidence Session 1 – Proofed Transcript, April 2018, p. 16

³⁸⁵ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 37

³⁸⁶ Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p. 8-9

³⁸⁷ National Seniors Australia, Issues paper: Review of the rate of return guidelines, December 2017, p. 2-3

Key Point	Submission	Stakeholder	AER response
	are almost entirely neutralised the very substantial risk reductions under the rules mean that economic and financial risk (the top left boxes) for regulated entities are almost entirely neutralised		regulation's impact on beta
	more careful analysis is required before the characteristics of electricity networks and gas pipelines can be shown to operate, in the circumstances of a specific network or pipeline, to deliver a low β	APA ³⁸⁸	
Stranding risk	The Guideline should be explicit about its treatment of the risk of stranding risk	ENA ³⁸⁹	To the extent that there are genuine risks of extreme changes in demand for
	Stranding risk is borne by consumers.	CCP ³⁹⁰ EUAA. ³⁹¹	which present the potential for asset stranding, the
	More emphasis need to be on an examination of the risks networks face and whether the market risk premium and beta accurately reflect that risk allocation	EUAA. ³⁹²	regulatory regime for gas and electricity can mitigate the risk by providing prudent discount and accelerated depreciation provisions. the CCP also stated that consumers bear the risk of the underutilised assets, as the full costs of the assets must continue to be reflected in the regulated revenues and prices We have consistently and transparently set out our approach for examining risk, MRP and beta. For example, AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, we list out our criteria of assessing

- APA, Submission responding to discussion papers and expert evidence, 4 May 2018, p. 16
- ³⁸⁹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 7, 30
- ³⁹⁰ Consumer Challenger Panel 16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 44-46.
- ³⁹¹ Energy Users Association of Australia, EUAA submission AER Rate of Return Review Issues Paper, October 2017, p. 8
- ³⁹² Energy Users Association of Australia, EUAA submission AER Rate of Return Review Issues Paper, October 2017, p. 5–6.

Key Point	Submission	Stakeholder	AER response
			information.
The use of market data	Concerned with transparency in regulation and use of available data	EUAA ³⁹³	We maintain the view that economic efficiency is advanced by employing a rate of return that reflects rates in the market for capital finance. ³⁹⁴ Partington and Satchell have also interpreted efficient financing costs as the opportunity cost of capital, which is a market rate of return for assets with a given level of risk. ³⁹⁵
	Market data reflects risk/returns greater than that of regulated assets	ECA ³⁹⁶	
	Consider risk on an asset basis and not at a business level	ECA ³⁹⁷	
	Volatility in market data/Share price may be related to market sentiment rather than fundamental characteristics	CRG ³⁹⁸	
Interrelationships	Components and parameters for the rate of return should not be viewed in isolation.	CCP16 ³⁹⁹ NSG ⁴⁰⁰	We maintain the view that one should not view any component or relevant parameter adopted for estimating the rate of return in isolation. In developing our approach and implementing it to derive the overall rate of return, we are cognisant of a number of interrelationships relating to the estimation of the return on equity and debt and underlying input parameters

³⁹³ Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p. 5–6.

³⁹⁴ For example, see: AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 12, 16. AER, Final decision United Energy distribution determination 2016 to 2020 Attachment 3–Rate of return, May 2016, pp. 281–292.

³⁹⁵ Partington, G., Satchell, S., Report to the AER: Discussion of the allowed cost of debt, 5 May 2016, p. 15.

³⁹⁶ Energy Consumers Australia, Review of the rate of return guideline, December 2017, p. 20.

³⁹⁷ Energy Consumers Australia, Review of the rate of return guideline, December 2017, p. 15–18.

³⁹⁸ Consumer Reference Group, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 47

³⁹⁹ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline review concurrent evidence sessions, p. 6

⁴⁰⁰ Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p.12

3 Form and structure of the rate of return guidelines

We have implemented the approach outlined in section 2 to achieve the legislative objectives by developing rate of return guidelines that:

- calculate an allowed rate of return in the form of a nominal, vanilla, weighted average cost of capital that is consistent with the value of imputation credits and does not include transaction costs involved in raising capital
- are capable of being automatically applied without exercise of discretion.

These aspects about the form and structure of the rate of return guidelines are discussed further in the sections 3.1 and 3.2 below.

In addition to estimating the rate of return as a nominal vanilla WACC, stakeholders submitted that we should consider other information that could act as a cross check on the overall level of the rate of return. We consider the role of this information in section 3.3 below.

3.1 A nominal, vanilla, weighted average cost of capital

Our decision is to determine the benchmark allowed rate of return for a regulatory year as a weighted average of the return on equity for the regulatory period in which that regulatory year occurs and the return on debt for that regulatory year, weighted by our benchmark gearing ratio. The rate of return is calculated as follows:

$$WACC = E(k_e)(1-G) + E(k_d)G$$

Where:

- E(ke) is the expected return on equity
- E(kd) is the expected return on debt
- G is the proportion of debt in total financing, otherwise referred to as the gearing ratio

Our allowed rate of return is determined on a nominal vanilla basis that is consistent with our estimate of the value of imputation credits.

Under the current regulatory framework we must determine the rate of return on this nominal, vanilla, weighted average basis that is consistent with the estimate of the value of imputation credits.⁴⁰¹ We also understand that we will be required to determine

⁴⁰¹ NER, cl. 6.5.2(d); NER, cl. 6A.6.2(d); NGR, r, 87(4).

the allowed rate of return on this basis under COAG's draft legislation for the implementation of a binding rate of return instrument.⁴⁰²

In any case, we consider that a nominal, vanilla, weighted average of the return on equity and return on debt, without adjustment for capital raising costs, would best contribute to the achievement of the legislative objectives, for the following reasons:

- The use of a weighted average of the returns on equity and debt allow for the relative risks involved in investing as an equity-holder or debt-holder to be reflected in the overall rate of return.
- A nominal, vanilla rate of return provides for a simpler rate of return estimation, and a more transparent and detailed modelling of the impacts of inflation and tax costs on regulated cashflows.
- This has been our long-standing approach that we have applied consistently over a number of years. We have not received any submissions suggesting that we should change any of these aspects of our rate of return estimation approach.

We also estimate an allowed rate of return that does not include the transaction costs involved in raising debt and equity capital. Instead, we will assess efficient compensation of these costs through expenditure allowances at each regulatory determination. Similar to the treatment of inflation and tax, this approach is consistent with our current approach, provides for a simpler estimate of the allowed rate of return, and a more transparent and detailed modelling of capital raising transaction costs.

Estimating a weighted average of the returns on debt and equity that is consistent with the value of imputation credits requires us to estimate:

• The return on equity

Our estimation approach for this parameter is set out in chapter 5, and our estimation of individual return on equity parameters are set out in chapters 6 (risk free rate), 7 (market risk premium), and 8 (equity beta).

• The return on debt

Our estimation approach for this parameter is set out in chapters 9 and 10.

• The benchmark gearing ratio

Our estimation of this parameter is set out in chapter 4.

• The value of imputation credits

Our estimation of this parameter is set out in chapter 11.

In the next section we outline how the guidelines can be automatically applied to calculate each of these elements.

⁴⁰² CoAG EC, *Draft legislation to create a binding rate of return instrument*, 02 March 2018; and CoAG EC, *Bulletin, Binding rate of return guideline, June 2018.*

3.2 Automatic application

Legislative amendments have been proposed that would make these guidelines binding. In the draft legislation we are required to make a binding rate of return instrument that either sets a value for the rate of return on capital and value for imputation credits, or sets a formula for the calculation of the rate of return and the value of imputation credits. If we set a formula rather than a value then the formula must be capable of being automatically applied during the life of the guideline, without any exercise of discretion.⁴⁰³ We cannot set different methodologies or a band of values from which we can choose at the time of applying the guideline in a regulatory determination.⁴⁰⁴

In light of CoAG's commitment to implementing a binding rate of return guideline, we have developed guidelines that are capable of either:

- operating as non-binding guidelines under the current legislative framework; or
- being automatically applied as a binding rate of return instrument if the legislative framework changes.

Implementing this approach, our decision is to make guidelines that set:

- The rate of return as a formula, being the weighted average of the return on debt and return on equity, weighted by the gearing ratio (as set out in section 3.1 above). For each input into this formula, we set:
 - The return on equity as a formula, being the Sharpe-Lintner Capital Asset Pricing Model (SLCAPM) formula.
 - The return on debt as a formula, being the trailing average portfolio approach, with a transition from an on-the-day approach to a trailing average, and based on third part debt data.
 - A fixed value for the benchmark gearing ratio.
- A fixed value of imputation credits (gamma).

Within these formulas, we also set out how the calculation of the allowed rate of return and its various elements will change under certain events or contingencies, such as if a particular data source is no longer available.

We will implement our formula for calculating the rate of return by setting out:

• The use of the SLCAPM to calculate the return on equity with a fixed value for the market risk premium of 6 per cent and a fixed value for equity beta of 0.7. This results in a fixed equity risk premium of 3.6 per cent, which will be added to the risk free rate calculation.

⁴⁰³ CoAG EC, *Draft legislation to create a binding rate of return instrument*, 02 March 2018; and CoAG EC, *Bulletin, Binding rate of return guideline, June 2018.*

⁴⁰⁴ Statutes Amendment (National Energy Laws) (Binding Rate of Return Instrument) Bill 2018, cl.18 and 30.

Our discussion paper on this topic noted that this was the first time we are applying this automatic approach for the life of the instrument. That paper did not set out how we might assess whether we should set a value or a formula. We asked whether it is appropriate to include self-executing formulas (mechanistic/automatic) where only the data is entered at the time of application. We set out our initial views on whether our current approach to return on debt, return on equity, gamma, and gearing ratio is amenable to mechanistic application over the life of the guideline. Our initial view in the discussion paper was, other than the approach to estimating the return on equity, all other aspects were amenable to mechanistic application.

The expert joint report on the concurrent evidence sessions reported the following agreed positions:

- According to the draft legislation the AER cannot exercise discretion in the way it calculates the rate of return during individual price/revenue determinations, but it could exercise discretion during the development of the Instrument. This means that the AER is required to either
 - o fix parameters as part of the Instrument, or
 - provide a way for a parameter to be computed mechanistically and objectively so that it is commensurate with the market conditions at that time.⁴⁰⁶
- Parameters that are relatively stable over a long period (regulatory period or more) should be fixed. Also, where data taken at a given point in time is not suited for estimating the parameter value, then such should also be fixed. Hence, equity beta and gearing should be fixed. That is, equity beta is relatively stable over a long period of time and gearing information at a specific point in time is not suitable for estimating the value of gearing. ⁴⁰⁷

Where market variables influence the appropriate value at a given time then such parameters should be set via a prescriptive methodology. Hence, the risk free rate and cost of debt should be a prescriptive methodology based on market evidence.⁴⁰⁸

In relation to the market risk premium, the expert joint report appeared to indicate agreement that the market risk premium is neither constant nor directly inversely related to the risk free rate. However, given the uncertainty on how to model the correlation between the market risk premium and risk free rate, most experts considered it appropriate to fix the value of the market risk premium. One expert's view

⁴⁰⁵ AER, *Discussion paper, MRP, risk free rate averaging period and the automatic application of the rate of return,* March 2018, section 7.

⁴⁰⁶ Joint Expert Report, *RORG review – Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 2.1.1, p.12.

⁴⁰⁷ Joint Expert Report, RORG review – Facilitation of concurrent evidence sessions, CEPA, 21 April 2018, section 2.10, p.17.

⁴⁰⁸ Joint Expert Report, *RORG review – Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 2.10, p.17.

was that a formula for calculating the market risk premium that is dependent on the risk free rate should be developed. This expert was concerned that a fixed value of the market risk premium may result in the allowed return on equity being too high when the risk free rate is high and too low when the risk free rate is low.⁴⁰⁹

We also considered other stakeholder submissions in arriving at our draft decision (see section 3.4 for a summary of submissions). We agree with the consensus in the expert joint report that parameters that are relatively stable over a long period (such as a regulatory period or more) should be fixed. We also agree that where market variables influence the appropriate value at a given time then such parameters should be set via a prescriptive methodology. Other stakeholder submissions also support the experts' consensus opinion.

We applied the above assessment approach agreed to by the experts in exercising our judgement on what should be fixed as a value or as a prescriptive methodology. Applying this assessment approach will provide for the rate of return guidelines to be commensurate with efficient finance cost and most likely contribute to the achievement of the legislative objectives. The following subsections set out how we have applied this assessment approach when considering each of the parameters for the rate of return and value of imputation credits: the benchmark gearing ratio, the return on equity, the return on debt, and the value of imputation credits.

3.2.1 Gearing

Our decision is to set a fixed gearing value in the guidelines.

All experts agreed that conceptually gearing should not change regularly as the core capital structure decisions of companies are stable and there is a cost to changing capital structure in response to regulatory gearing changes.⁴¹⁰ The experts agreed that spot gearing values are distorted by short term market fluctuations and therefore data should be averaged over a longer period of time.⁴¹¹ On this basis, the experts agreed that gearing data at a point (spot values) in time is not appropriate to be used in setting the rate of return. Rather, a historical average should be determined and fixed for the life of the guidelines.⁴¹² Other stakeholders also support the experts' consensus opinion.⁴¹³

⁴⁰⁹ Joint Expert Report, RORG review – Facilitation of concurrent evidence sessions, CEPA, 21 April 2018, section 6.13, p.64.

⁴¹⁰ Joint Expert Report, *RORG review – Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 3.02, p.28.

⁴¹¹ Joint Expert Report, RORG review – *Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 3.06, p.30. Dr. Martin Lally noted that the optimum historical averaging period is unclear but getting it 'wrong' and consequential over or under forecasting gearing would not materially affect gearing.

⁴¹² Joint Expert Report, RORG review – *Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 2.10, p.17.

⁴¹³ NSG, Submission on the RORG review, 4 May 2018, p.13; Evoenergy, Review of the rate of return Instrument – evidence sessions, 4 May 2018; CCP, Submission to the AER on its RORG review concurrent evidence session, 4 May, 2018, pp.16-17.

Our gearing estimate in the guidelines is based on a historical average (see section 4 for a detailed discussion on our gearing estimate). We agree with the experts that conceptually the capital structure of companies is stable. We also agree that gearing should not be determined based on spot values during the life of the instrument as short term gearing data can be distorted by market fluctuations in share prices. We therefore consider it appropriate to fix a value for gearing in the guidelines.

3.2.2 Return on equity

Our decision is set in the guidelines a formula - based on the SLCAPM - to calculate the return on equity. Within the SLCAPM formula, our decision is to set fixed values for market risk premium and equity beta, and set a formula for calculating the risk free rate.

In our 2013 guidelines our approach to estimating the return on equity was based on our foundation model approach.⁴¹⁴ Our draft decision continues this approach through use of the SLCAPM formula to calculate the return on equity and through our approach to determining the inputs into the SLCAPM formula (see chapter 5 for further detail on our return on equity approach).

In our discussion paper we set out three options on how we could automate the return on equity:

- Setting a fixed value for the rate of return on equity.
- Setting a fixed value for the equity risk premium⁴¹⁵ to be used with a risk free rate that is commensurate with the timing of each regulatory determination.
- Setting out a methodology that allows some or all of the SLCAPM parameter inputs to vary during the period of the guidelines.

The concurrent expert evidence sessions, joint expert report, and submissions in response focussed on option 2. Within that option the discussion was largely on whether the market risk premium should be fixed. Experts unanimously agreed that:

- The value for equity beta should be fixed as it is stable over long periods.
- The risk free rate for the SLCAPM should be set as a methodology as market variables influence its appropriate value.⁴¹⁶

We consider that setting a fixed value for equity beta in the guidelines will best contribute to the legislative objectives and we have not received any submissions that hold a contrary view. We also consider empirical equity beta data is relatively stable

⁴¹⁴ AER, Rate of return guideline, December 2013, section 5. Available at: <u>https://www.aer.gov.au/networks-pipelines/guidelines-schemes-models-reviews/rate-of-return-guideline-2013</u>.

⁴¹⁵ We refer to the product of the MRP and equity beta in the SL-CAPM formula as the equity risk premium.

⁴¹⁶ Joint Expert Report, RORG review – Facilitation of concurrent evidence sessions, CEPA, 21 April 2018, section 2.10, p.17.

over long periods, consistent with the experts' views.⁴¹⁷ We also agree that the risk free rate is influenced by variations in the market and should be determined by a prescribed methodology that captures these variations. This is also consistent with submissions to us.⁴¹⁸

The experts at our concurrent expert evidence sessions considered whether it was appropriate to set a formula for calculating the market risk premium that would be applied over the life of the guidelines. The reason why they considered this was because all experts agreed that the market risk premium is neither constant nor directly inversely related to the risk-free rate. However, given the lack of an accepted model of the correlation between the market risk premium and the risk free rate, most experts considered it more appropriate to fix the market risk premium.⁴¹⁹

At the concurrent expert evidence sessions, Prof Stephen Gray stated that we have 3 options. These are:

- fix the market risk premium, which is added to the observed risk free rate
- fix the total market return, such as uses by UK regulators
- fix the total market return at the commencement of the instrument and apply a formula that adjusts the total market return in proportion to movements in the observed risk free rate

In respect of the third option, Prof. Gray suggested that for example we adjust total market return by a fixed percentage of the movement in the risk free rate.⁴²⁰ Prof. Gray noted that experts might consider whether the AER's current approach to date, which in his view is to fix the market risk premium, results in a return that is too high (low) when the risk free rate is high (low).⁴²¹

We agree with most of the experts that it is more appropriate to set a fixed value for the market risk premium in the guidelines. There is no accepted theoretical basis to support a market risk premium that varies within lock-step with the risk free rate nor is there a robust basis by which to calculate the appropriate adjustment to the market risk premium in line with changes to the risk free rate. More importantly, we consider that an approach which promotes a stable return on equity may not be suitable for a regulatory model which resets every 5 years. A fixed value of the market risk premium

⁴¹⁷ Joint Expert Report, RORG review – *Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 2.10, p.17.

⁴¹⁸ NSG, Submission on the RORG review, 4 May 2018, p.13; Jemena, Submission on concurrent evidence sessions and discussion papers, 4 May 2018.

⁴¹⁹ Joint Expert Report, RORG review – *Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 6.13, p.64.

⁴²⁰ AER, *CES* session 1 proofed transcript, 15 March 2018, p.11.

⁴²¹ Joint Expert Report, RORG review – *Facilitation of concurrent evidence sessions*, CEPA, 21 April 2018, section 6.13, p.64.

which is reset every 5 years is more likely to remain unbiased and reduce risk of over/under investment.⁴²²

In the absence of a robust basis to determine how the market risk premium should vary over time, we consider setting a formula for the market risk premium is not appropriate. Such an approach would lead to arbitrary changes in the market risk premium, which is not founded on good finance principles nor supported by robust analysis. Outcomes from such an approach are not commensurate with efficient financing costs.

We do not use the approach of setting a fixed value for the total market return as our foundation model approach and do not consider we should adopt it in a guideline that may be applied automatically.⁴²³ Setting a fixed value for the total market return is equivalent to adopting a particular formula to setting the market risk premium based on the total market return and the (variable, market-driven) risk free rate. As noted above, we consider that there is no accepted theoretical basis for modelling the correlation between the risk free rate and the market risk premium. Further, if the risk free rate increases (decreases) when the total market return is a fixed value, then the resulting shrinking (increasing) equity risk premium may not compensate (over compensate) service providers for systematic risk. We consider a return on equity comprising a variable risk free rate and a fixed equity risk premium provides the certainty and predictability that all stakeholders require.

3.2.3 Return on debt

Our decision is to set a formula for calculating the allowed return on debt. This is the same approach as that adopted in our 2013 guidelines. Under the current rules framework, an annually updating return on debt is required to be given effect by automatic application of a formula. For this reason, our view is that implementation of a binding rate of return instrument requires only incremental changes to our current approach.

Our decision is to set a formula that calculates the return on debt reflective of a benchmark credit rating and term to maturity, and based on data from third party data providers.

In our discussion paper, our initial view was that the return on debt approach in our 2013 guidelines is amenable to automatic application. We did not receive any submissions indicating that our approach to date is not suitable for automation.

Debt data from third party data providers fluctuates over time with market conditions. However, the availability of the data providers themselves, the benchmark credit rating, and benchmark term of debt are relatively stable over time. Observed values for these parameters, particularly credit rating and debt term, may change over time, as service

⁴²² See our chapter 7 of this draft decision.

⁴²³ This total market return approach is sometimes referred to as the Wright approach, which we discuss in chapter 7 of this draft decision.

providers react to market conditions, deal with legacy debt arrangements, and take time to adjust to financing practices to target levels. We have observed changes in actual debt term of service providers since previous reviews in 2009 and 2013. However, we consider that the benchmark efficient term is likely to be relatively stable.

3.2.4 Value of imputation credits

Our approach to date to estimating the value for imputation credits (gamma) is set out as the product of the payout ratio (the proportion of imputation credits generated by the benchmark efficient entity that are distributed to investors) and the utilisation rate (the extent to which investors can use the imputation credits they receive to reduce their personal tax). We choose a value of 0.5 for gamma from within a range of 0.3 to 0.6 based on an utilisation rate of 0.6 and a distribution rate of 0.83.⁴²⁴

We consider the utilisation rate and the payout ratio parameters do not change quickly and also see no reasons to expect movement up or down. Therefore, it is appropriate to fix the value of imputation credits for the life of the guideline.

3.3 Role of financial performance measures

Stakeholders submitted that we should consider information that could act as a cross check on the overall reasonableness of the rate of return:

- the CCP16, CRG, MEU, Ergon Energy and Energex submitted that we should consider RAB multiples
- the CCP16, CRG, MEU, PIAC, ECA, EUAA, QCOSS, Origin Energy, Agricultural Industries Energy Task Force, QCOSS, ATCO Gas Australia, Ergon Energy and Energex submitted that we should consider historical profitability measures
- the CCP16, ENA, Origin Energy, Ergon Energy and Energex and ATCO Gas Australia submitted that we should consider financeability assessments.

These three types of overall financial performance measures are discussed in turn below.

Overall, we consider that these measures cannot be used to directly determine parameter estimates for the allowed rate of return. We agree with the CCP16 submission that there is difficulty in disaggregating the information contained in RAB multiples and historical profitability measures to determine the degree of outperformance of the allowed rate of return.

However, we consider that there may be useful information within the trends in RAB multiples and historical profitability measures over time. Comparisons of RAB multiples and historical profitability measures can provide information on the performance of the regulatory system as a whole. This information may be helpful in considering whether

⁴²⁴ See our chapter 11 for further detail.

the business' actual rate of return has been systematically lower or higher than the allowed rate of return.

The CCP16 submitted that this information cannot be used at a parameter level but can inform the overall exercise of judgement in setting the rate of return or reviewing other elements of the regulatory regime⁴²⁵. We agree that RAB multiples and historical profitability may provide useful contextual information and cause for further examination of the material we rely on when estimating rate of return parameters (other elements of the regulatory regime are beyond the scope of this review). We have done this as part of this review through further consideration of the impact of regulation on equity beta estimates within our comparator set (see sections 2.4.3 and 0), examination of service providers' actual debt issuances (see sections 10.2, 10.3, and 10.5), and further consideration of the most appropriate third-party debt data to reflect our benchmark credit rating (see section 10.1). Though outside the scope of this review, we are also currently undertaking reviews of other parts of our regulatory regime, such as our review of our regulatory tax approach and review of profitability measures for gas and electricity businesses.

3.3.1 RAB multiples

RAB multiples are the enterprise value of a firm divided by its Regulatory Asset Base (RAB). It can be calculated using two main sources of data to evaluate the market value of equity in service providers:

- Acquisition data the purchase price when a transaction⁴²⁶ of the service providers occurs, or
- Trading data the existing share price of a business that has an equity ownership in a service provider.

Subject to satisfying several conditions, a RAB multiple of 1 may indicate that the present value of the future stream of expected cash-flows of the firm is equal to its RAB. This means that investors are compensated exactly at a level to encourage efficient investment.

Our 2013 Guidelines do not incorporate RAB multiples, nor have we used RAB multiples in our decision making for any regulatory determinations. In developing our 2013 Guidelines we considered the potential use of RAB multiples to inform our decision making, however, we decided not to use them. In our 2013 Rate of Return Guidelines we stated that:

We propose to not apply levels and changes in RAB acquisition and trading multiples as a direct reasonableness check on the overall rate of return at the

⁴²⁵ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 27-28.

⁴²⁶ A transaction may only involve the purchase of a certain portion of equity in a service provider, in which case, an implied RAB multiple can be calculated based on the price paid for the percentage of shares acquired.

time of a particular revenue determination or access arrangement. Instead, we propose to use these multiples as part of a set of indicators that we monitor over time and across network businesses to help inform us of potential areas of inquiry and research. This more general use of these multiples reflects the fact that there are many potential influences on RAB acquisition and trading multiples, such as changes in the expectations and the realisations of business revenues, expenditures and rates of return. Given these many potential influences, any changes in these multiples may not be immediately attributable to any one factor.⁴²⁷

In our Issues Paper for the 2018 rate of return guideline review, we noted stakeholder submissions from previous consultations, that asset sales should be used as potential tests of whether the allowed rate of return is achieving the national gas and electricity objectives⁴²⁸. We also discussed the difficulties in using RAB multiples from asset sales as they may be indicative of other elements of the firm's cash flows and that they do not provide a definitive answer to the specific return investors require. However, we did note that if RAB multiples significantly and persistently differ from one, then it may be informative of the reasonableness of our overall building block allowances and our overall rate of return estimates over time.

We therefore invited submissions on whether information on asset sales should be used when assessing outcomes against the NEO, NGO and the related RPPs.

Given the submissions from consumer groups in support of using RAB multiples to inform the rate of return, we decided to revisit the issue. Therefore, we also included RAB multiples for discussion in our first Concurrent Expert Evidence Session, and released a Discussion Paper⁴²⁹ prior to the session and invited stakeholder submissions.

Throughout our consultation process, the network businesses have not been in favour of the use of RAB multiples to inform the rate of return – primarily due to the difficulty in disaggregating the RAB multiples into its different sources of value. However, as mentioned previously, consumer groups have been in favour, mainly highlighting that the rate of return is a large component of the building block revenue, and the large multiples observed, particularly in recent transactions, are suggestive that the rate of return has been too generous.

Our draft decision is not to use RAB multiples to inform our rate of return. This is consistent with the approach we adopted in the 2013 Guidelines. After considering all the submissions and experts' views provided as part of this review, we consider that there is still much subjectivity and no agreement from experts, in terms of the appropriate assumptions to use to disaggregate these multiples. We also note recent developments in the UK, with the decomposition of RAB multiples for particular utilities,

⁴²⁷ AER, Rate of Return Guideline - Explanatory Statement, December 2013, p 48.

⁴²⁸ AER, Review of the rate of return guidelines - Issues Paper, October 2017, p 16.

⁴²⁹ AER, Financial performance measures - Discussion Paper, February 2018.

where there has been contention regarding the assumptions that should be used and the appropriate factors to consider.

However, our draft position is to monitor RAB multiples (as part of our separate review into reporting profitability measures⁴³⁰). This may assist us in gauging the overall impact of all our decisions (including expenditure allowances) on investment in network businesses.

In the subsections below, we outline our consideration and responses to the issues that have been raised by stakeholders in submissions (including the CRG) and by experts in the Concurrent Expert Evidence Session.

Stakeholders' submissions in response to our Issues Paper

In response to our Issues Paper:

- Consumer groups submitted that we should consider RAB multiples to inform our decision making – specifically:
 - CCP16 submitted that RAB multiples provide directly observable evidence on whether the outcomes for the allowed rate of return match the expectations of investors and the requirements of the NEO and NGO in practice – it also submitted that RAB multiples can be used to inform the Rate of Return but not mechanistically⁴³¹; and
 - the Major Energy Users submitted that it supported the AER in examining asset sales measures given that it claimed that little information is currently available on the service providers⁴³².
- However, in contrast some of the network businesses, such as APA, ENA and Spark Infrastructure were against the use of RAB multiples and submitted they were irrelevant given various considerations such as, the price that investors pay for an asset can be impacted by a wide range of different factors, and that the AER should uphold its benchmark approach⁴³³. Although, we note that Ergon Energy and Energex submitted that information on asset sales is potentially useful in testing the reasonableness of the AER's determinations.⁴³⁴

Potential use of RAB multiples to inform the Rate of Return

As indicated above, in its submission, CCP16 supported the use of RAB multiples to inform the rate of return. This was because it considered that they provide the most

⁴³⁰ AER, Draft Position Paper - Profitability measures for regulated gas and electricity network businesses, April 2018.

⁴³¹ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p 6.

⁴³² Major Energy Users, Review of the rate of return guidelines, December 2017, p 14.

⁴³³ APA, APA submission responding to AER Issues Paper, December 2017, p 3; Energy Networks Australia, AER Rate of Return Guidelines – Response to Issues Paper, December 2017, p13; Network Shareholder Group, Response to Issues Paper on the review of the Rate of Return Guidelines, December 2017, p 13.

⁴³⁴ Ergon Energy and Energex, Issues Paper – Review of the Rate of Return Guidelines, December 2017, p 3.

direct information available on the relativity of allowed and expected returns on capital or equity, and are easily observed at the time of transactions⁴³⁵. CCP16 also provided detailed information on the potential use of RAB multiples including their use by other regulators.

CCP16 noted that with the three most recent electricity network transactions (TransGrid network 2015, Ausgrid network 2016, Endeavour network 2017), the winning bidders paid RAB multiples of 1.6, 1.4 and 1.58 respectively. It considered that a very conservative interpretation of the recent transactions is that they provide strong evidence that the combined allowances for the cost of capital and tax under the AER's current framework and recent decisions are not too low. It suggested that, given the magnitude of the multiples in absolute terms and relative to multiples in other regulatory jurisdictions, one could conclude that it provides evidence that the allowances are more likely to have exceeded investors' expectations for the required return on investment⁴³⁶.

However, it noted that the weakness of RAB multiples is that further analysis is required to make the best use of the information on the relativity of expected and actual returns. As such, it considered that RAB multiples cannot be used in a mechanical manner. It noted that in the case of TransGrid, the purchasing consortium indicated that TransGrid's two unregulated business units – a telecoms arm and connection of renewable energy to the grid – can provide growth opportunities to warrant the high price. CCP16 also noted that it is likely that the bidder who makes the most optimistic assessment of these opportunities will be the likely winner and this will be reflected in its bid, adding to the premium above TransGrid's RAB⁴³⁷.

As mentioned previously, CCP16 also provided information on how RAB multiples have been used by other regulators and advisors, as a cross-check on the overall rate of return⁴³⁸:

- New Zealand Commerce Commission (NZCC) States that its focus is not on isolating the individual sources of excess returns, but rather its objective is to assess whether the existing WACC uplift is too generous. It considered that irrespective of the cause of a high RAB multiple, the existence of such multiples is strong evidence that the WACC is not too low.
- The Chairman of the UK Office of Water (Ofwat) has referred to high RAB multiples for UK water utilities as evidence that the regulator's allowed WACC is too high, and noted that the continuing trend for water companies is for them to be sold at

⁴³⁵ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p 29.

⁴³⁶ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p 18.

⁴³⁷ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p 18.

⁴³⁸ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp118-119.

prices around 130 per cent of regulated asset value. This suggests that the regulator's adopted cost of capital is too high and the premia reflects excess demand for these assets.

- PwC, in its 2013 advice to Ofwat on the approach to reviewing the appropriate returns for water companies, reported an average market-to-asset ratio (MAR, analogous to RAB multiples) in the UK water sector of 1.23. It identified the WACC being set too high relative to the actual costs of financing as a possible driver, in addition to outperformance in the unregulated business (which is generally small), synergies available to a new entity and outperformance in operational allowances.
- Grant Samuel, in 2014 prepared an independent expert's report relating to APA Group's proposal to acquire Australian gas distribution company Envestra. In his report, Grant Samuel commented that a common rule of thumb parameter used in the valuation of energy infrastructure assets are RAB multiples, and noted that most assets generally trade at a premium to RAB and that the precise reasons for this are uncertain but contributing factors could include, amongst others, a cost of capital being less than that assumed by the regulators.
- Deloitte, published in 2011, a paper exploring a number of valuation issues concerning regulated infrastructure assets. When describing factors for Australian utilities trading at a premium, Deloitte stated that the effective cost of capital borne by the asset owner may be lower than that assumed by the regulator due to either a cheaper cost of capital and/or greater leverage.

In response to CCP16's submission, we consider that RAB multiples do provide valuable information and can potentially provide information on the relativity between our determined cost of capital and investors' required cost of capital. This is because when valuing the network businesses, investors would incorporate their ability to outperform on any, or all, of the regulatory benchmarks, including the rate of return building block allowance.

However, as noted by CCP16, the difficulty is in disaggregating the RAB multiples into their different sources of value. The numerous assumptions that must be made regarding the different sources of value, and then the extent of outperformance, and value creation in those items adds considerable complexity and contention to the process of determining the rate of return.

In our consideration of the potential use of RAB multiples, we have also investigated other regulator's use of RAB multiples, as identified by CCP16 in its submission, to consider whether we could also apply a similar approach.

As discussed above, the NZCC has used RAB multiples as one of its considerations to reduce its WACC uplift from the 75th percentile of its WACC range to the 67th percentile⁴³⁹. It analysed the RAB multiples for 4 of its regulated electricity and gas

⁴³⁹ Other considerations included empirical analysis of the expected losses to consumers from under- and overestimating the 'true' cost of capital.

distribution businesses by making various adjustments to understand why the businesses were acquired (or trading) at a premium to their RAB values. Thus, the NZCC accounted for unregulated revenues, certain financial obligations such as deferred taxes and capital works in progress which were not reflected in the RAB at the time etc⁴⁴⁰. However, we note the limitations the NZCC identified in its use of RAB multiples⁴⁴¹:

- limited data points NZCC used 3 acquisition multiples and 1 trading multiple for its analysis
- the possibility of there being a range of factors that could affect the RAB multiples, including outperformance of operating and capital expenditure – the NZCC did not incorporate outperformance of regulatory expenditure allowances in its analysis, and
- difficultly in separating the unregulated activities when decomposing the RAB multiples – whilst the NZCC did incorporate a separation of unregulated business in its analysis, the extent of growth and cost efficiencies that can be achieved in the unregulated portion can be uncertain and hence quite subjective.

We also considered PwC's advice to Ofwat in 2013 as mentioned by CCP16. However, as outlined in our Discussion Paper, we also note PwC's subsequent advice in 2017 to Ofwat in decomposing MARs for two UK water companies (United Utilities and Severn Trent)⁴⁴². PwC's advice was to inform Ofwat's methodology paper for the upcoming 2019 water price reviews. PwC concluded that its analysis of MARs for these two water companies suggested that the allowed return on equity was higher than investors' expected cost of equity.

However, we note that following PwC's analysis, National Grid in the UK commissioned NERA to consider the evidence on MARs for National Grid and UK water companies, including the analysis undertaken by PwC on United Utilities and Severn Trent⁴⁴³. NERA considered that PwC had not accurately adjusted for non-regulated businesses, non-wholesale businesses, outperformance opportunities and pension deficits/surpluses. NERA found that the MARs for the businesses, after adjusting for the identified issues, can be approximately 1. Therefore, NERA considered that there was no evidence to suggest that investors' expected cost of equity is lower than the allowed returns for the water sector. NERA also made a similar conclusion in its decomposition of National Grid's MAR.⁴⁴⁴

⁴⁴⁰ NZCC, Amendment to the WACC percentile for price-quality regulation for electricity lines services and gas pipeline services – Reasons paper, October 2014, pp 155-156.

⁴⁴¹ NZCC, Amendment to the WACC percentile for price-quality regulation for electricity lines services and gas pipeline services – Reasons paper, October 2014, p 113.

⁴⁴² Delivering Water 2020: Our methodology for the 2019 price review, Appendix 12: Aligning risk and return – December 2017, pp 51-52.

⁴⁴³ NERA, Implications of Observed Market-to-Asset Ratios for Cost of Equity at RIIO-T2, December 2017.

⁴⁴⁴ We note that earlier in July 2017, CEPA analysed the RAB multiple for the sale of National Grid's 61% equity stake in its gas distribution business. CEPA calculated a potential range for the implied cost of equity, which suggested that investors may be willing and able to finance gas distribution assets at an actual cost of equity below the

In our investigation of the potential use of RAB multiples, we also considered the literature available on the different assumptions that can be applied to the different sources of value. We found that they can be wide-ranging, eg:

- Control premiums Buyers typically pay a premium for a controlling interest in a business but this can vary considerably eg, between 20 per cent to 40 per cent⁴⁴⁵.
- Unregulated revenues Different businesses have differing amounts of unregulated revenue and differing scope for potential growth. We note that for recent transactions the amount and growth in unregulated revenues at the time of the acquisitions were quite varied⁴⁴⁶:
 - TransGrid (RAB value of about \$6.2bn to \$6.5 billion in 2015): unregulated revenue was about \$44 million in 2015,⁴⁴⁷ and had increased by about 170 per cent over the preceding 5 years⁴⁴⁸.
 - Ausgrid (RAB value of about \$15.3bn to \$15.8 billion in 2016): unregulated revenue was up to \$200 million in 2016 but had not experienced such growth over the preceding years compared with TransGrid⁴⁴⁹.
 - Endeavour Energy (RAB value of around \$6.2bn to \$6.4 billion in 2017): unregulated revenue was up to \$125 million, and similarly, had not experienced such growth over the preceding years compared with TransGrid⁴⁵⁰.
 - Further, bidders may have differing levels of optimism in the opportunities and cost efficiencies that can be achieved, as noted by CCP16 regarding the potential growth in TransGrid's unregulated revenues.

Concurrent expert evidence session

We also considered experts' views from the Concurrent Expert Evidence Session. In summary, most experts agreed that it is not practicable for observations of RAB multiples to be decomposed in order to draw inferences as to the rate of return

⁴⁵⁰ Endeavour Energy, Annual Reports 2012 to 2016. Its total income for 2016 was about \$1.5 billion.

regulatory allowance. CEPA, Key questions for RIIO-T2 and GD2 - Lessons from the sale of National Grid Gas Distribution, July 2017.

⁴⁴⁵ As noted by Frontier Economics (Frontier Economics, Why do regulated assets sell for more than the RAB? -IPART 25th Anniversary Conference, October 2017). Grant Samuel & Associates, Takeover Offer from BaoSteel and Aurizon – Independent Expert Report for Aquila Resources Limited, 20 June 2014, p 65; Lonergan Edwards & Associates, Takeover Offer for Country Road Limited – Independent Expert Report, 21 July 2014, p 45; EY, Independent Expert's Report and Financial Services Guide – PanAust Limited Takeover Offer from Guangdong Risking H.K. (holding) Limited 24 April 2015, p 65.

⁴⁴⁶ The TransGrid, Ausgrid and Endeavour Energy transactions were undertaken in November 2015, October 2016 and May 2017, respectively.

⁴⁴⁷ This was about 5% of its total income of about \$887 million in 2015. TransGrid, Annual Report 2015, p 22.

⁴⁴⁸ TransGrid, Annual Report 2015, p 7.

⁴⁴⁹ Ausgrid, Annual Reports 2011 to 2016. Its total income for 2016 was about \$2.6 billion.

required by the market⁴⁵¹. Hence, it is difficult for them to be used by the AER in the process of setting the rate of return.

Most experts also agreed that there can be several reasons as to why RAB multiples can be higher than 1, and that it is difficult to disaggregate the multiples into their different sources of value. However, two experts disagreed, with one expert expressing their view that high RAB multiples can still indicate that returns are too high⁴⁵².

We discuss below our consideration of the issues raised in the session.

Consideration of economic conditions at the time of the transactions

Some experts noted the current structural decline in interest rates, and that they are lower than when the regulatory determinations were made for the recent transactions. Therefore, a potential source of value is a lower cost of debt, which NSW tax payers have benefited from, through a higher sale price. It was also suggested that once interest rates started increasing, there could be the opposite effect ie, where there is no regulatory outperformance of items other than the cost of debt, the RAB multiple could decrease to less than 1.⁴⁵³

In response, we agree that it is likely that RAB multiples incorporate economic circumstances at the time of the transactions (they would also incorporate investors' views of future economic conditions). Therefore, depending on when the transaction occurred, the magnitude and the cause of the difference between the investor's required cost of capital and the regulatory rate of return may differ, which further adds to the subjectivity in the assumptions that can be used.

Likelihood of outperformance of the Rate of Return as a source of value

In terms of outperformance of the actual Rate of Return itself, one expert indicated that valuations for the transactions would have been done for over a 99-year period (given the duration of the leases). Therefore, it would be difficult for buyers to definitively know the value of any rate of return outperformance beyond the existing regulatory period for each transaction⁴⁵⁴.

However, we note that this argument could also apply to each of the building block items eg, it would be difficult for buyers to definitively know the value of any expenditure outperformance beyond the existing regulatory period. This suggests that RAB multiple premia are largely derived from sources outside of the building block components and due to factors such as control premia and unregulated revenue etc, which we consider to be unlikely.

⁴⁵¹ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 35-36.

⁴⁵² CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 35-36.

⁴⁵³ AER, Concurrent Expert Evidence Session 1 – Proofed Transcript, March 2018, pp 111-113.

⁴⁵⁴ AER, Concurrent Expert Evidence Session 1 – Proofed Transcript, March 2018, p 113.

We consider that outperformance of the rate of return can potentially be a source of value, as RAB multiples would incorporate the buyer's view of potential outperformance on any, or all, of the regulatory benchmarks.

Limited dataset to develop robust conclusions

One of the experts noted that transactions are relatively infrequent, and there is a risk of inappropriately applying learnings from one transaction generally across all the network businesses eg, if a buyer paid a relatively large premium because it considered it could extract value through improved efficiencies then it would be inappropriate to apply this across the board to other networks as though the rate of return was too generous⁴⁵⁵.

We note that there are few acquisition multiples, particularly under the 2013 Rate of Return Guidelines, to draw definitive conclusions. Therefore, we agree with the view expressed at the Concurrent Expert Evidence Session, and so we would be cautious about applying findings from an investigation of any sample set with few observations, into a guideline that would apply to all the service providers.

Also, for analysis of RAB multiples, we consider that acquisition multiples would be preferable over trading multiples. This is because trading multiples would conceivably incorporate more factors that would need to be adjusted for eg, trading multiples are likely to include shareholders' views of managements' ability to deliver outperformance, whereas with acquisition multiples, the purchaser would be assessing their own ability to deliver outperformance – hence trading multiples are likely to have added information asymmetry.

High RAB multiples can still indicate that returns are too high

One of the experts disagreed with the view that it is not practicable for observations of RAB multiples to be used to inform the Rate of Return. His view was that high RAB multiples can still indicate that returns are too high⁴⁵⁶.

We consider that RAB multiples incorporate a multitude of information, including the possibility that the allowed Rate of Return is higher than the expected return required by investors. However, the outperformance may be in part (or wholly) due to outperformance in other building block items or sources of value outside of regulated revenues. As mentioned previously, the difficulty is in disaggregating the RAB multiples to determine the degree of outperformance of the allowed rate of return. Hence, we are cautious about drawing conclusions purely based on the observation that RAB multiples are high.

⁴⁵⁵ AER, Concurrent Expert Evidence Session 1 – Proofed Transcript, March 2018, p 115.

⁴⁵⁶ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 35-36.

Submissions in response to the Concurrent Expert Evidence Session

We have also considered submissions in response to our Discussion Paper and the Concurrent Expert Evidence Sessions:

- Network businesses reiterated their general view that RAB multiples should not be used to inform the rate of return⁴⁵⁷
- The Consumer Reference Group (CRG) commented that the network businesses are continuing to enjoy strong earnings and are trading at multiples of 1.3 to 1.6 of their RABs⁴⁵⁸.
- CCP16 reiterated its view that RAB multiples provide information on expected returns that is directly relevant to the AER's task of determining a fair rate of return. It again noted that while other factors affect RAB multiples, it considers that there are sound regulatory and commercial precedents for disaggregating the impacts of these factors, and that the implied ROE can be used in a directional manner in setting the ROE and ROR – not in a mechanical manner. CCP16 also noted that at the Concurrent Expert Evidence Session there was agreement that RAB multiples contain information on expected returns, but disagreement on if and how RAB multiples can be used in determining the Rate of Return⁴⁵⁹.

CCP16 also referred to its previous submission regarding analysis by Credit Suisse of the TransGrid transaction which suggested that even after allowing for the outperformance on the efficiency incentives, zero tax payments for the foreseeable future, and growth in unregulated business there is an additional \$1bn (or around 15 per cent of RAB) in unaccounted value. CCP16 considered that the most likely remaining explanation for this is that the allowed Rate of Return exceeds the required Rate of Return⁴⁶⁰.

In response, we note that for the TransGrid transaction, it could be that the allowed rate of return is indeed higher than the rate of return required by investors. However, we also note that it could be in part (or wholly) due to other reasons such as:

• Differences in views of the potential growth and cost efficiencies achievable in the unregulated portion of TransGrid's revenues.

⁴⁵⁷ Evoenergy, Rate of return guideline – evidence sessions, May 2018, p 5; Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p 10; APA, APA submission responding to discussion papers and expert evidence, May 2018, p 3; ENA, AER Review of the Rate of Return Guideline – Response to Discussion Paper and Concurrent Expert Evidence Sessions, May 2018, pp 12-13; APGA, Submission to the AER – Review of rate of return guideline, May 2018, p 21.

⁴⁵⁸ RoR CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p vi.

⁴⁵⁹ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 8, 54, 58 – 64.

⁴⁶⁰ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p 64.

 Possibility that the buyers overpaid for their acquisition due to optimism in assumptions⁴⁶¹, or unobserved assumptions (that may be considered optimistic or unrealistic) but have actually been incorporated into the purchase price⁴⁶²

CCP16 also noted that it broadly supports Professor Johnstone's position, that while the implied return in RAB multiples cannot be estimated with precision, it is reasonable to draw qualitative conclusions from very high values⁴⁶³. However, as mentioned above, we are cautious about drawing conclusions, purely based on the observation that RAB multiples are high.

AER assessment of the use of RAB multiples

After considering all the submissions and experts' views, in order to use RAB multiples to inform our rate of return, we consider that we would need to be able to control for at least the following items (including any interactions between the factors):

- Outperformance in regulatory benchmarks, including expenditure allowances (operating expenditure and capital expenditure) and tax allowances
- Unregulated revenue potential growth and cost efficiencies that can be achieved
- Control premium if the acquisition results in a majority share ownership; also value may be placed on perceived real options which may be easier to exercise with a majority share ownership
- Economic circumstances at the time of the transactions differences between the rate of return set at the time of the business' determination versus market conditions when the transactions took place
- Possibility of over-optimism in assumptions.

However, there is much subjectivity and no agreement from experts on the appropriate assumptions to use to disaggregate these multiples. We also note recent developments in the UK, with the decomposition of RAB multiples for particular utilities, where there has been contention regarding the assumptions that should be used and the appropriate factors to consider.

Therefore, given the subjectivity and uncertainty in assumptions required to disaggregate RAB multiples, our draft decision is that the use of these multiples is not an appropriate method to inform our decision on the rate of return.

Rather, our draft position is to examine:

⁴⁶¹ Optimism in the efficiencies that can be achieved in the expenditure allowances.

⁴⁶² Wright, Burns, Mason & Pickford, Estimating the cost of capital for implementation of price controls by UK Regulators – March 2018, p 13.

⁴⁶³ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p 60.

- the latest empirical evidence for each of the parameters in the Sharpe-Linter CAPM, that we are using as our foundation model
- the level of conservatism that we applied in selecting our parameter estimates when developing the 2013 Rate of Return Guideline and the continuing appropriateness of this, and
- the decisions we have made in subsequent determinations following the 2013 Guidelines

to determine the parameter estimates that best achieve the NEO, NGO and related RPPs, to promote the efficient delivery of services by the network businesses for the long-term interests of consumers.

Our draft decision is consistent with the approach we adopted in the 2013 Guidelines.

3.3.2 Historical profitability

Analysis of historical profitability refers to the use of financial statements to compare:

- free cash flows to equity, with the
- estimated cash flows to equity (inputted into the rate of return building block).

In our 2013 Guidelines, we did not include the use of historical profitability measures to inform our rate of return estimate. Service providers are incentivised to outperform regulatory benchmarks for opex, capex, debt, tax and service performance. Thus, the ability for a service provider to earn an actual return on equity higher than the allowed return on equity, may be due to the outperformance of these benchmarks - importantly outperformance does not necessarily imply that the regulatory rate of return is incorrect.⁴⁶⁴ Further, historical profitability measures provide information on historical profitability, which may not reflect investor expectations about future profitability.

However, during recent determination processes, consumer groups have made submissions raising concerns around excessive profitability of network businesses, and hence the need for ongoing profitability reporting and assessment⁴⁶⁵. In response to those submissions, we commenced a separate consultation process exploring the use of profitability measures for these businesses - AER review into profitability measures for regulated gas and electricity network businesses.⁴⁶⁶ We have also received submissions on this topic as part of our rate of return guideline review.

In our draft Position Paper for our profitability measures review, we identified the following measures to potentially report on: Return on Assets, Return on Equity, EBIT/customer numbers and RAB multiples.⁴⁶⁷

⁴⁶⁴ AER, Rate of Return Guideline - Explanatory Statement, December 2013, p 48.

⁴⁶⁵ AER, Discussion Paper - Financial performance measures, February 2018, p 11.

⁴⁶⁶ AER, Draft Position Paper – Profitability measures for regulated gas and electricity network business, April 2018.

⁴⁶⁷ AER, Draft Position Paper – Profitability measures for regulated gas and electricity network business, April 2018.

Our draft decision is not to use the historical profitability measures identified in our draft Position Paper to inform our rate of return for this guideline review. This is because we currently do not have a robust data set to calculate these measures. This approach is consistent with our 2013 Guidelines.

However, going forward, for the purposes of determining the rate of return, we consider that careful consideration of profitability measures (such as those identified from financial statements) may be helpful in identifying whether the business' actual cost of debt has been systematically lower or higher than the cost of debt applied in the rate of return.

The subsections below set out further detail on our review into reporting profitability measures, submissions received, and our assessment of the role of historical profitability measures for our 2018 rate of return guideline.

Separate AER review into reporting of profitability measures

In November 2017, we released a discussion paper on profitability measures for regulated electricity and gas network businesses⁴⁶⁸. The paper also included a study undertaken by McGrath Nicol on the measures of financial performance that could be applied to the businesses we regulate.

In April 2018, we released a Draft Position Paper identifying the suite of profitability measures that we intend to report on: Return on Assets, Return on Equity, EBIT/customer numbers and RAB multiples.⁴⁶⁹

In our Draft Position Paper, we noted that in response to our discussion paper, there was general support for our intent to collect and report data on measures of profitability. Consumer groups stated profitability analysis formed an essential post-implementation review of the regulatory framework, providing an important check on how service providers performed against the regulatory determination. They considered the measures would assist consumers to identify if the networks were achieving excessive profits relative to the risks they faced and investigate the causes of any excessive profits.⁴⁷⁰

We stated that our primary purpose in reporting the measures is to provide transparency for stakeholders on the profitability of the service providers, and that we intend to publish the measures (including our analysis of the outcomes and any relevant caveats) in annual performance reports for gas and electricity businesses.

⁴⁶⁸ AER, Discussion Paper – Profitability measures for regulated gas and electricity network business, November 2017.

⁴⁶⁹ AER, Draft Position Paper – Profitability measures for regulated gas and electricity network business, April 2018.

 ⁴⁷⁰ AER, Draft Position Paper – Profitability measures for regulated gas and electricity network business, April 2018, p
 2.
We consider this additional information will assist stakeholders in making submissions to our regulatory determination processes and other regulatory reviews.⁴⁷¹

We also stated that we would have regard to profitability outcomes of the service providers as part of our regulatory determination processes. However, the information would not be used in a mechanistic way to make adjustments to allowed revenues. Rather, the information would be contextual, along with other information such as expenditure and service performance outcomes from previous regulatory periods.

We held a public forum on the draft position paper on 16 May 2018, and submissions were due by 30 May 2018. We intend to release our final position in August 2018.

In the sections below, we outline the submissions we have received in relation to profitability measures, as part of this Rate of Return guideline review. We also discuss our consideration of the issues raised (including views expressed by experts) in reaching our draft decision on the rate of return.

From the submissions received, we found that stakeholders were generally in support of our work in investigating profitability measures. However, there were mixed views on how useful profitability measures are in informing our decision on the rate of return.

Stakeholders' submissions in response to our Issues Paper

In response to our Issues Paper, service providers submitted that:472

- Profitability measures are potentially useful in testing the reasonableness of the AER's overall revenue determination. However, it is difficult to use this information directly to inform the rate of return.
- Examination of a firm's profitability provides a useful cross-reference regarding the relationship between regulatory returns and the broader performance of the business. However, the principles set out in the rate of return guideline must maintain primacy in determining how the actual return is derived.
- APA and Spark Infrastructure submitted that information on profitability is irrelevant to assessing the allowed rate of return⁴⁷³.

The Agricultural Industries Energy Task Force submitted that it supported the AER using a performance measurement framework and having access to detailed financial data from companies. The task force referred to work completed by the Sapere Group

 ⁴⁷¹ AER, Draft Position Paper – Profitability measures for regulated gas and electricity network business, April 2018, p
 5.

⁴⁷² Ergon Energy and Energex, Issues Paper – Review of the Rate of Return Guidelines, December 2017, p 3; Origin Energy, Review of rate of return guidelines, December 2017, p 2.

⁴⁷³ APA, APA submission responding to AER Issues Paper, December 2017, p 3; Network Shareholder Group, Response to Issues Paper on the review of the Rate of Return Guidelines, December 2017, p 13.

which claimed that under the current guideline, network owners can exceed efficient costs, prices and profits.⁴⁷⁴

Consumer groups such as the MEU, ECA and EUAA supported the use of profitability measures⁴⁷⁵. The EUAA submitted that it is a weak argument to state that 'there are many reasons for actual profitability to be above the Allowed Rate of Return Objective so we should not be worried about actual profitability¹⁴⁷⁶.

CCP16 submitted that historical profitability measures can be used to assess whether actual profitability has been lower or higher than:⁴⁷⁷

- the allowed Rate of Return under the determination, and
- other comparable regulated and unregulated businesses.

CCP16 also indicated that we would need to identify the underlying reasons for variations in actual profitability against the allowed rate of return, and compare profitability using multiple measures. It also stated that for example, if EBIT/RAB is comparable to the allowed rate of return, but the return on equity is significantly above the rate inputted into the rate of return, it may suggest that:⁴⁷⁸

- the tax allowed is higher than the actual tax paid
- the actual gearing levels are significantly different from the assumed level and/or
- actual debt costs are significantly below the benchmark debt costs assumed.

It noted that lower debt costs could be because:

- the utilities have better credit ratings than assumed, or
- lenders perceive that regulated utilities have lower business risks that are not fully
 reflected in the ratings and are willing to lend at lower rates than the benchmark for
 comparable businesses.

Concurrent Expert Evidence Session

Whilst not explicitly discussed at the Concurrent Expert Evidence Session, it was however addressed in the Joint Expert Report. In the report, most experts agreed that ex post firm-specific profitability data contains no information that assists in estimating the rate of return required by the market⁴⁷⁹.

⁴⁷⁴ Agricultural Industries Energy Taskforce, Submission on rate of return issues paper, December 2017, p 3.

⁴⁷⁵ Major Energy Users, Review of the rate of return guidelines, December 2017, p 14; Energy Consumers Australia, Review of the rate of return guideline, December 2017, p 13;

⁴⁷⁶ Energy Users Association of Australia, EUAA submission - AER Rate of Return Review Issues Paper, October 2017, p 4.

⁴⁷⁷ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 26-27.

⁴⁷⁸ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 27-28.

⁴⁷⁹ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, p 35.

However, one expert expressed the view that profitability measures could be used to assess whether company or industry returns were too high, and that ex post reviews (actual financial performance outcomes) provides an indication of whether the idealised building block approach has done something "reasonable" and sustainable⁴⁸⁰. If not, then the building block approach can be replaced with something possibly simpler and more transparent (eg, CPI increases only) or the input parameters could be changed, to achieve a realistic level of "good" regulation.

Submissions in response to the Concurrent Expert Evidence Session

In response to the Concurrent Expert Evidence Session and our Discussion Paper service providers submitted that analysis of historical performance measures should not be used in determining the rate of return.⁴⁸¹

However, consumer groups continued to support the use of profitability measures. Both the Consumer Reference Group (CRG) and the Queensland Council of Social Services (QCOSS) considered that the profitability of the Queensland networks were excessive. QCOSS submitted that the returns for Powerlink, Energex and Ergon were substantially higher than other energy supply industry participants (eg, CS Energy and DUET Group).⁴⁸² The CRG submitted that we should use actual profitability data to compare against modelled returns, but noted that there is currently no reporting being undertaken under the 2013 Rate of Return Guideline.⁴⁸³

CCP16 reiterated its view submitted in response to the Issues Paper, and stated that general financial performance measures can inform the overall judgement on the Rate of Return, which will in turn be reflected in the values for the underlying parameters such as the MRP and beta, around which there is considerable uncertainty.⁴⁸⁴

However, CCP16 submitted that as they are backward looking measures they provide limited guidance on expected returns, but comparisons of historical profitability measures can provide information on the performance of the regulatory system as a whole. CCP16 considered that this information cannot be used at a parameter level,

⁴⁸⁰ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 35-36.

⁴⁸¹ Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p 10; APA submission responding to discussion papers and expert evidence, May 2018, p 3; ENA, AER Review of the Rate of Return Guideline – Response to Discussion Paper and Concurrent Expert Evidence Sessions, May 2018, pp 12-13;

⁴⁸² Both submissions referred to a report undertaken by ResponseAbility which showed that the average annual profit margins over the past 5 years for Queensland networks had been between 25% to 30%, whereas for CS Energy it had been a little over 10% and for DUET Group less than 10% over the same period. QCOSS, Submissions on review of Rate of Return Guideline, May 2018, pp 16-17; RoR CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p 20.

⁴⁸³ RoR CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p 20.

⁴⁸⁴ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 8, 54.

but can inform the overall exercise of judgement in setting the rate of return, or reviewing other elements of the regulatory regime⁴⁸⁵.

AER assessment of the role of historical profitability measures

Our draft decision is not to use the historical profitability measures identified in our draft Position Paper to inform our rate of return for this guideline review. This is because we currently do not have a robust data set to calculate the measures identified in our draft Position Paper. This approach is consistent with the 2013 Guidelines.

As indicated in our draft Position Paper for the profitability measures review, our intention is to collect the information required, and commence calculating and presenting the profitability measures identified. We consider that it is important to collect information on the actual profitability of the network businesses that we regulate. This can help inform us on the effectiveness of our regulatory framework and identify areas that require further investigation. For example, if investigation of actual profitability against the allowed rate of return identifies that the main driver of the higher profits is due to systematically lower than expected expenditures, then we may need to further investigate our approach to setting the expenditure allowances.

However, we do consider that there are limitations in the use of historical profitability measures to inform our rate of return. We do not consider that such measures could directly inform us as to whether the allowed return on equity was too high or too low. This is because, if a particular business' revenue and actual costs (eg, operating and capital expenditure, tax and debt costs including gearing) were to equal its regulatory allowances (or assumptions), then theoretically the business should only be able to earn its allowed return on equity (excluding any unregulated cashflows).⁴⁸⁶

Therefore, variations between actual profit and return on equity could be indicative of variations between actual and expected: revenues; expenditure allowances; interest costs; gearing; depreciation and tax allowances. Under incentive based regulation, if businesses are able to lower their actual costs whilst meeting their required service standards (ie, are able to achieve efficiencies) then they are able to keep part of the benefits, which would manifest in their actual return on equity, being higher than their allowed return on equity.

However, we consider that careful consideration of profitability measures (such as those identified from financial statements) may be helpful in identifying whether the business' actual cost of debt has been systematically lower or higher than the cost of debt applied in the rate of return.⁴⁸⁷ In this regard, while we agree with the majority view expressed at the Concurrent Expert Evidence Session that ex post profitability

⁴⁸⁵ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 27-28.

⁴⁸⁶ For simplicity, we have set aside the differences between nominal and real returns in our regulatory framework.

⁴⁸⁷ This is because through careful consideration of financial statements we may potentially be able to compare the actual interest costs paid by businesses against the allowed return on debt.

data should not directly inform our rate of return estimate, we disagree that historical profitability measures contain no useful information.

In response to CCP16 comments about using financial performance measures to exercise judgement on the overall rate of return, or even at the parameter level, we consider it appropriate to first understand the drivers behind any financial performance measures before exercising any form of judgement.

We would also like to note that we regularly review our approach to setting all our regulatory allowances, to examine whether they remain appropriate. We have commenced a review of our tax allowance, given preliminary advice from the ATO indicated an apparent material discrepancy between our tax allowances and the actual tax payments made to the ATO by the service providers. If the actual tax paid by the businesses is lower than our regulatory tax allowance, then this can be a driver of actual returns being higher than our expected returns. We are currently seeking submissions to our tax review, and we are intending to release our final report and recommendations by December 2018.⁴⁸⁸

3.3.3 Financeability assessments

Financeability refers to a service provider's ability to meet its financing requirements and to efficiently raise new capital. In the regulatory context, it often refers to the service provider's ability to achieve the benchmark credit rating applied in the estimation of the rate of return. This is typically assessed through examining the key financial ratios used by credit rating agencies and testing if these ratios support the benchmark credit rating, based on a service provider's allowed cashflows. As it involves testing the benchmark credit rating against allowed cashflows, various stakeholders have argued that it may be viewed as a cross-check on the assumptions underpinning those allowed cashflows.

Our 2013 Rate of Return Guideline does not include a financeability assessment as part of determining the rate of return, nor does it include it as a cross-check on the reasonableness of the rate of return.⁴⁸⁹ In subsequent consultations some stakeholders submitted that financeability should be used as a potential test of whether the allowed rate of return is achieving the legislative objectives.⁴⁹⁰

Our draft decision is not to use financeability assessments to inform our rate of return. We consider that a financeability assessment would not be helpful in a regulatory context if it were to be undertaken using the assumptions (eg, gearing and interest costs) underpinning the allowed revenue. We also do not consider it appropriate to undertake a financeability assessment using the actual costs of a service provider as these costs may not be efficient.

⁴⁸⁸ AER, Issues Paper – Review of regulatory tax approach, May 2018; AER, Tax Review 2018 - Initial Report, June 2018.

⁴⁸⁹ AER, Rate of Return Guideline - Explanatory Statement, December 2013, p 59.

⁴⁹⁰ AER, Issues Paper - Review of the rate of return guidelines, October 2017, p 16.

Therefore, there is no clear guidance on the assumptions that should be used in any financeability assessment as a cross check on the benchmark parameters in the Sharpe-Linter CAPM that we are using as our foundation model. Consideration of the appropriateness of these parameters should continue to be based on the evidence examined in determining those parameters.

Stakeholders' submissions in response to our Issues Paper

In response to our Issues Paper, service providers were generally in favour of financeability assessments as a cross-check on the reasonableness of the AER's decisions, and to ensure consistency with the AER's benchmark credit rating assumption.⁴⁹¹ However, some businesses indicated that it is difficult to use financeability assessments to directly inform the rate of return, and that the debt and equity principles applied in the rate of return must retain primacy.⁴⁹²

The ENA submitted that if an issue is identified then it would not necessarily mean that an adjustment in the rate of return is required – rather, it could be addressed through re-profiling of regulated cash flows from future periods.⁴⁹³

CCP16 considered that financeability assessments could serve as a cross-check of the regulator's decision but not as a driver of decisions. It also submitted that the primary responsibility for addressing financing issues should rest with the business (including through equity injections). It also commented that in applying tests, regulators commonly assume benchmark gearing and capex and opex in line with the regulator's assumptions so that it does not provide for poor management. It also indicated that other regulators (eg, Ofwat and Ofgem) have adopted the approach of testing the impact of a range of scenarios on the likely outcome for the return on equity. It considered that this was another potentially useful means of testing the sustainability of a regulatory decision.⁴⁹⁴

Concurrent Expert Evidence Session

In summary, most experts at the session agreed that financeability analysis may provide insights on the time profile of cash flows. However, such analysis would not provide objective information for use in setting the rate of return required by the market.⁴⁹⁵

⁴⁹¹ APA, APA submission responding to AER issues paper, December 2017, p 3; ATCO Gas Australia, Review of rate of return guideline – Issues Paper, December 2017, p 4;

⁴⁹² Network Shareholder Group, Response to issues paper on the review of the Rate of Return Guidelines, December 2017, p 4; Origin Energy, Review of rate of return guidelines, December 2017, p 2; Ergon Energy and Energex, Issues Paper – Review of the Rate of Return Guidelines, December 2017, p 3;

⁴⁹³ ENA, AER Rate of Return Guidelines – Response to Issues Paper, December 2017, p 13;

⁴⁹⁴ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 6, 30

⁴⁹⁵ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 36-37.

We note that one expert suggested that financeability analysis could be used as a test of the internal consistency of the regulatory determination. That is, a certain credit rating is assumed when determining the allowed cashflows, however, if key financial ratios using the allowed cashflows, do not support the credit rating that was initially assumed, then assumptions may need to be revisited.⁴⁹⁶

Some experts considered that if analysis of cash flow profiles were to be undertaken (for a business and its capex program – especially if it is particularly large) and there is a breach of credit metrics leading to a downgrade, then this would indicate an issue of the timing of cash flows. The expert joint report suggested that to remedy the situation:⁴⁹⁷

- a company's management would be able to raise equity and/or defer dividends
- alternatively, a regulator may have flexibility to adjust depreciation schedules in a NPV neutral way.

The report also noted that the analysis of cash flow profiles should use a gearing assumption based on that of the benchmark efficient entity rather than the actual gearing of the company. Also, the report noted that a range of other stylised assumptions would need to be made (eg, borrowing at the rate assumed in the trailing average debt calculation).⁴⁹⁸

Submissions on our discussion paper and concurrent evidence session

In response to the Concurrent Expert Evidence Session and our Discussion Paper the ENA submitted that financeability assessments should be considered to ensure that the allowed return is sufficient to support the assumed credit rating used to determine the cash flows.⁴⁹⁹ APA on the other hand submitted that financial performance measures are irrelevant to the setting of the Rate of Return.⁵⁰⁰

CCP16 reiterated its views submitted in response to our Issues Paper – mainly that the AER should not use financeability analysis as a direct input into the consideration of the Rate of Return. It also highlighted that regulators, when undertaking financeability assessments, typically use the ratios used by rating agencies and that these are well accepted, but ratings also depend on the qualitative evaluation of the business environment. It noted that for regulated businesses, the quality of the regulatory framework is critical and that it considered that the AER's current regulatory framework

⁴⁹⁶ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 36-37.

⁴⁹⁷ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 36-37.

⁴⁹⁸ CEPA, Rate of Return Guideline Review – Expert Joint Report, April 2018, pp 36-37.

⁴⁹⁹ ENA, AER Review of the Rate of Return Guideline – Response to Discussion Paper and Concurrent Expert Evidence Sessions, May 2018, p 14.

⁵⁰⁰ APA, APA submission responding to discussion papers and expert evidence, May 2018, p 3.

is more than achieving this objective, as the framework has delivered relative stable ratings at or above investment grade.⁵⁰¹

CCP16 also noted that at the Concurrent Expert Evidence Session the primary issues addressed by financeability tests are matters of timing (that the building block model, properly applied, provides a revenue stream over the asset's life that matches the costs). It commented that to address any financeability issues, other regulators have:⁵⁰²

- placed primary responsibility on the utility for managing cash flows across regulatory periods, and
- required that any financeability adjustment is NPV neutral.

CCP16 also considered that if a financeability issue arose, it is likely to require a utility specific adjustment which is not possible within a framework that:⁵⁰³

- requires NPV neutrality within each regulatory period
- a common Rate of Return determined in accordance with a binding instrument.

AER assessment of financeability tests

Our draft decision is not to use financeability assessments to inform our rate of return. This is consistent with the approach we adopted in the 2013 Guidelines. Whilst the rate of return does comprise a large component of the building block revenue, it is not the sole determinant of cashflows. Also, assessing whether cashflows are sufficient to maintain a credit rating involves assessing more than the allowed rate of return. It typically involves:

- a quantitative assessment of a service provider's actual and projected revenue, expenditure and finance costs, and
- a qualitative assessment of a service provider's regulatory environment and ownership model
- to form an opinion on the ability and willingness of a service provider to meet its financial obligations.

Therefore, a service provider may have a higher or lower credit rating than our benchmark depending on the credit rating agency's qualitative assessment of the service provider, and/or whether the service provider is projected to outperform or underperform relative to regulatory allowances, including expenditure allowances.

⁵⁰¹ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 64-65.

⁵⁰² Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 64-65.

⁵⁰³ Consumer Challenge Panel, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 64-65.

Thus, a lower credit rating than our benchmark rating is not necessarily an indication that the rate of return is insufficient (and vice versa).

Further, we do not consider that a financeability assessment would be helpful in a regulatory context if it were to be undertaken using the assumptions (eg, gearing and interest costs) underpinning the allowed revenue. This is because the cashflows assumed under such a financeability assessment, would be equal to the cashflows provided in calculating the allowed revenues in the first place – which means that there would be no cashflow timing issues under such an assessment.⁵⁰⁴

We also do not consider it appropriate to undertake a financeability assessment using the actual costs of a service provider. This is because we are aiming to provide benchmark allowances to allow for efficient service delivery – not actual costs which may be inefficient.

We consider that financeability assessments may be useful to service providers internally when they consider how to best finance their capital expenditure, particularly if there is a relatively large forward capex program. Given its actual costs, the business can undertake financeability assessments to decide how to finance the expenditure (eg, the mix of borrowings, raising equity and/or deferring dividends) and any implications the financing may have on its credit rating.

We also note that empirical evidence over the past 10 years, as presented in our Discussion Paper, suggest that credit ratings for service providers have been stable and have been at around BBB+.⁵⁰⁵ Further, there has been no indication from service providers (or stakeholders), or sufficient evidence put forward, that the allowed rate of return has been insufficient and hence is the cause of adversely affecting service providers' financeability.

Also, we note our decision to adjust certain parameters downwards, which would decrease the rate of return, holding all else constant. However, we have made these decisions using the latest empirical evidence available for each of those parameters. Hence, given that these adjustments are in line with market conditions, we do not consider that these incremental changes would adversely affect a service provider's financeability.

Stakeholders identified that financeability assessments are typically undertaken alongside considerations of whether or not to re-profile cashflows and bring forward cashflows from future regulatory periods (for example, through changing depreciation profiles). We note that this requires consideration of the entire profile of cashflows

⁵⁰⁴ We note that we also provide a return of capital (regulatory depreciation) allowance, and that the credit metrics treat depreciation as a non-cash item, which provides some headroom. However, it is necessary for a return of capital to be incorporated into the building block revenues as it is appropriate for funds to be returned to the provider of capital.

⁵⁰⁵ AER, Discussion Paper – Financial performance measures, February 2018, p 32.

resulting from a regulatory determination, which is beyond the scope of the rate of return guideline and more appropriately considered at each determination.

3.4 Summary of submissions

This section details the submissions we have received from stakeholders as part of our review process to date, and notes how we have had regard to each submission.

Table 12 Summary of submissions on automatic application

Key point	Submission	Stakeholder	AER response
Rationale for fixing parameters or prescriptive methodology	Supports fixing parameter values to maximise investor certainty where there has been no material change since they were last set and no material changes are expected over the next Instrument period. Materiality must be assessed against the level of precision and judgment required in establishing the initial level.	Network Shareholder Group ⁵⁰⁶	The evidence considered in setting each parameter value is discussed in the relevant sub sections throughout this document. In this section we have considered whether we should fix the parameter value for the life of the guideline.
	Supports experts joint report position that parameters that are relatively stable over a long period (regulatory control period or more) or data at a given point in time is not suited for estimating the parameter value should be the ones that are fixed.		We agree
	Re-openers must be limited where parameters are fixed. If Material changes occur, they are better addressed via re-openers.		The legislation does not allow re-openers ⁵⁰⁷
	Relatively stable parameters should be fixed. Other parameters should be set with reference to market data at the time of the determination and accordingly the Instrument should set out the methodology for determining the value.	Evoenergy ⁵⁰⁸	We agree
	Strongly encourages the AER to make the new binding Instrument as prescriptive as possible with appropriate formula to vary the actual rate of return as necessary rather than re-opening.	CCP 16 ⁵⁰⁹	The legislation does not allow re-openers. We have adopted an approach to assessing whether we should fix parameters or set a formula to vary the returns consistent with the experts' consensus.

⁵⁰⁹ CCP 16, Submission to the AER on its RORG review concurrent evidence session, 4 May, 2018, pp.16-17.

⁵⁰⁶ NSG, Submission on the RORG review, 4 May 2018, p.13.

⁵⁰⁷ CoAG EC, *Bulletin, Binding rate of return guideline,* June 2018.

⁵⁰⁸ Evoenergy, *Review of the rate of return Instrument – evidence sessions*, 4 May 2018.

Fixed parameters	Gearing (60 per cent), equity beta (0.7) and MRP (7.0) should be fixed.	Network Shareholder Group ⁵¹⁰	The value for gearing, equity beta and MRP are discussed in the relevant chapters. Our decision to fix these parameters during the life of the Instrument has been determined consistent with the rationale proposed by the NSG.
	If the Instrument does not set a value for the MRP for the life of the Instrument then it should identify the weights given to each source of information.	Evoenergy ⁵¹¹	We set a value for the MRP for the life of the Instrument
	Fix the equity beta and MRP to promote stability, predictability and consistency of the allowed rate of return.	Jemena ⁵¹²	Equity beta and MRP are fixed
	The AER's decision to give weight to the DGM should be done in the context of a binding Instrument. The generally volatile nature of the DGM outputs is a major issue in setting a MRP for a binding Instrument.	CCP 16 ⁵¹³	Our decision on the weight given to the DGM is set out in chapter 7. We agree with most experts that it is more appropriate to fix the MRP for the life of the Instrument
Prescriptive methodology	The risk free rate be set as a prescriptive methodology	Network Shareholder Group ⁵¹⁴	We agree

Table 13 Summary of submissions on financial performance measures

Key point	Submission	Stakeholder	AER response
RAB multiples			
Relevance of RAB multiples in determining the RoR	RAB multiples are largely irrelevant (not necessary) in assessing the allowed RoR. Spark Infrastructure also added that it supports the AER's current benchmark approach (which provides businesses with incentives to operate efficiently).	APA ⁵¹⁵ , Spark Infrastructure ⁵¹⁶ APA ⁵¹⁷	We consider that RAB multiples may potentially provide information on the differences between the allowed RoR and investors' required RoR. However, the difficulty is in extracting the information given the subjectivity in assumptions involved in disaggregating the multiples.

⁵¹⁰ NSG, Submission on the RORG review, 4 May 2018, p.13.

⁵¹¹ Evoenergy, *Review of the rate of return Instrument – evidence sessions*, 4 May 2018.

⁵¹² Jemena, Submission on concurrent evidence sessions and discussion papers, 4 May 2018.

⁵¹³ CCP 16, Submission to the AER on its RORG review concurrent evidence session, 4 May, 2018, p.116.

⁵¹⁴ Submission on the RORG review from the Network Shareholder Group (NSG), 4 May 2018, p.13.

⁵¹⁵ APA, *APA Submission responding to the AER issues paper*, December 2017, p 3.

⁵¹⁶ Network Shareholder Group, *Response to Issues Paper on the review of the Rate of Return Guidelines*, December 2017, p 13.

⁵¹⁷ APA, APA submission responding to discussion papers and expert evidence, May 2018, pp 3-4.

We agree that a range of factors can affect RAB multiples. We also note that each transaction may have unique characteristics eg, businesses may have differing scope for growth in unregulated revenues. Hence, any findings would need to be carefully examined.	ENA ⁵¹⁸ ENA ⁵¹⁹ , Evoenergy ⁵²⁰ , Network Shareholder Group ⁵²¹	No weight should be given to asset sales, as the price investors pay can be impacted by a wide range of different factors – considered that most experts agreed with this view at the CEES. The ENA also added that every transaction is unique, hence it would be wrong to extrapolate from one particular transaction across the entire industry.	
Our draft position is to monitor RAB multiples (as part of our profitability reporting framework). This may assist us in gauging the overall impact of all our decisions (including expenditure allowances) on investment in network businesses.	Australian Pipelines and Gas Association ⁵²²	RAB multiples play little role in setting an allowed RoR ex-ante, but can play an important role ex-post in monitoring outcomes.	
As mentioned above, we are mindful of the subjectivity in assumptions required to disaggregate RAB multiples.	MEU ⁵²³	Supports the use of asset sales, as there is little information currently available on the service provider sector.	
We note that RAB multiples have been either used (or referred to) as one of several considerations. However, we also note the limitations in their use, as identified by other regulators, and recent developments in the UK that highlight the subjectivity in the assumptions involved in disaggregating the multiples.	CCP ⁵²⁴ CCP ⁵²⁵	RAB multiples provide direct observable evidence on whether the outcomes for the allowed RoR match the expectations of investors and the requirements of the NEO/NGO in practice. They are also considered by other regulators (eg, NZCC and Ofwat).	
Our draft decision is not to use RAB multiples to adjust parameters given the subjectivity in assumptions involved, and there being no clear agreement from experts on the assumptions that should be used.	CRG ⁵²⁶	Transactions for business with regulated entities have RAB multiples of 1.3 to 1.6. This could be used to adjust the observed beta range from $0.4 - 0.7$ to $0.2 - 0.5$.	
Due to the subjectivity in assumptions required to disaggregate the RAB multiples, it is difficult to objectively	CCP16 ⁵²⁷ , Ergon Energy and Energex ⁵²⁸	AB RAB multiples should not be used in a s as a mechanical manner (or to directly heck' on inform the RoR), but rather as a	Using RAB multiples as a 'cross-check' on

- ⁵²⁰ Evoenergy, *Review of rate of return guideline evidence sessions*, May 2018, pp 4-5.
- ⁵²¹ NSG, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), May 2018, pp 12-13.
- ⁵²² APGA, Submission to the AER Review of rate of return guideline, May 2018, p 21.
- ⁵²³ Major Energy Users, *Review of the rate of return guidelines*, December 2017, p 14.
- ⁵²⁴ CCP16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p 6.
- ⁵²⁵ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 54, 57, 61, 62.
- ⁵²⁶ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p vii.

⁵¹⁸ ENA, AER Rate of Return Guidelines – Response to Issues Paper, December 2017, p 13.

⁵¹⁹ ENA, AER Rate of Return Guidelines – Response to Issues Paper, December 2017, p 13; ENA, AER Review of the Rate of Return Guideline – Response to Discussion Papers and Concurrent Expert Evidence Sessions, May 2018, pp 92-93.

the RoR	reasonableness check on the RoR decision.	CCP ⁵²⁹	determine how much of the premium in the multiples are due to investors' required RoR being different from the allowed RoR.
	Considers that there is no role for RAB multiples to be used as a 'cross- check', as it agrees with most experts that there is no objective basis for the application of an overall test of the reasonableness of the RoR decision. Also, using RAB multiples as a 'cross- check' could result in a further avenue for backdoor discretion, which does not promote confidence in the stability of the process; also the 'cross-check' information itself, if used, would then be elevated above all other considerations, circumventing the rigour and transparency that may have preceded it.	Network Shareholder Group ⁵³⁰	As mentioned above, we consider that it is difficult to objectively determine how much of the premium in RAB multiples is due to outperformance of the allowed RoR. Therefore, we do not consider it is an appropriate method to use in a binding instrument.
Financeability			
Relevance of financeability in determining the RoR	Supports the use of financeability assessments - noted that they are routinely used by other regulators, and that they are an important consideration in order to ensure the financial viability of service providers. ENA also submitted that such an assessment would ensure that the allowed RoR is sufficient to support the assumed credit rating used to determine the cash flows. APA also submitted that financeability could be important given that providers of finance have regard to allowed rates of return.	ATCO Gas Australia ⁵³¹ , ENA ⁵³² ENA ⁵³³ , APA ⁵³⁴	We consider that financeability assessments would not assist in determining the RoR: the RoR allowance is not the sole determinant of cashflows; credit ratings typically include more than just a quantitative assessment of cashflows (eg, they include the rating agency's own qualitative assessment of the business' ownership model). Further, undertaking an assessment using the assumptions (eg, expenditure allowances and gearing) underpinning the building block revenue would not reveal any cashflow issues, as future capex is assumed to be financed according to the assumptions in the RoR allowance.
	Submitted that it supported the AER having access to additional data sets	Public Interest Advocacy	We have credit rating information for service providers. We note that the

- ⁵²⁷ CCP16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 6, 29.
- ⁵²⁸ Ergon Energy and Energex, *Issues Paper Review of the Rate of Return Guidelines*, December 2017, p 3.
- ⁵²⁹ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 54, 57, 61, 62.
- ⁵³⁰ NSG, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), May 2018, pp 12-13.
- ⁵³¹ ACTO Gas Australia, Review of rate of return guideline issues paper, December 2017, p 4.
- ⁵³² ENA, AER Rate of Return Guidelines Response to Issues Paper, December 2017, p 13.
- ⁵³³ ENA, AER Review of the Rate of Return Guideline Response to Discussion Paper and Concurrent Expert Evidence Sessions, May 2018, p 14.
- ⁵³⁴ APA, APA submission responding to AER issues paper, December 2017, p 3.

	on areas such as financeability. MEU also supports the use of financeability measures as it considers that there is little information currently available on the service provider sector.	Centre (PIAC) ⁵³⁵ MEU ⁵³⁶	credit ratings of service providers are currently around BBB+.
	Considers the use of financeability is not necessary, and supports the AER's current benchmark approach (which provides businesses with incentives to operate efficiently).	Spark Infrastructure ⁵³⁷	As discussed above, we consider that the use of financeability assessments is not informative in determining the allowed RoR.
Using financeability as a 'cross-check' on the RoR	Supports the use of financeability as it provides a useful cross-reference regarding the relationship between regulatory returns and the broader performance of the business. However, the principles set out in the guideline must maintain primacy in determining how the actual return is derived.	Origin Energy ⁵³⁸	Whilst the RoR comprises a large component of building block revenue, it is not the sole determinant of cashflows. Also, financeability assessments that credit rating agencies undertake, incorporate the actual performance of the businesses, which may include differences from our regulatory benchmark allowances.
	Considers that such tests can be used to examine the financial sustainability of a proposed determination, but they should not be used to directly determine the return on equity or the overall rate of return allowed.	CCP ⁵³⁹ , Ergon Energy and Energex ⁵⁴⁰ CCP ⁵⁴¹	Undertaking a financeability assessment of our determination would not be informative, if based on the assumptions underpinning the building block revenues. Also, if based on the actual costs of the business then we could be introducing potential inefficiencies.
Analysis of historic	al profitability		
Relevance of profitability in determining the RoR	Information on profitability is largely irrelevant (has no role) in assessing the allowed rate of return. The ENA also added that there is no clear link with any RoR parameter; historical measures are not relevant to a forward-looking incentive regime; and a large number of factors can affect the actual profitability of a business.	APA ⁵⁴² , Network Shareholder Group ⁵⁴³ APA ⁵⁴⁴ , Energy Networks Association ⁵⁴⁵	We consider that there are limitations to the use of historical profitability analysis to inform our RoR decision - if a business' actual costs were to equal its regulatory allowances, then it should only be able to earn its allowed return on equity. However, for debt costs, careful consideration of profitability measures may inform

- ⁵³⁸ Origin Energy, *Review of rate of return guidelines*, December 2017, p 2.
- ⁵³⁹ CCP16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p 6.
- ⁵⁴⁰ Ergon Energy and Energex, Issues Paper Review of the Rate of Return Guidelines, December 2017, p 3.
- ⁵⁴¹ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p 64.
- ⁵⁴² APA, APA Submission responding to the AER issues paper, December 2017, p 3.
- ⁵⁴³ Network Shareholder Group, Submission on the Rate of Return Guideline review, May 2018, p 10;
- ⁵⁴⁴ APA, APA submission responding to discuss papers and expert evidence, May 2018, pp 3-4.
- ⁵⁴⁵ ENA, AER Review of the Rate of Return Guideline Response to Discussion Paper and Concurrent Expert Evidence Sessions, May 2018, pp 12-13.

⁵³⁵ PIAC, Submission on rate of return guideline review issues paper, December 2017 p 1.

⁵³⁶ Major Energy Users, *Review of the rate of return guidelines*, December 2017, p 14.

⁵³⁷ Network Shareholder Group, *Response to Issues Paper on the review of the Rate of Return Guidelines*, December 2017, p 13.

us as to whether a business' actual cost of debt has been systematically higher or lower than the rate assumed in the RoR.

	Submitted that a potential profitability framework should not disrupt the incentive framework that businesses have to outperform targets.	APGA ⁵⁴⁶	We agree - under incentive based regulation if a business is able to achieve efficiencies then it is able to keep part of the benefits.
	Submitted that it supported the AER using a performance measurement framework and having access to detailed financial data from companies. It referred to work completed by the Sapere Group which claimed that under the current guideline, network owners can exceed efficient costs, prices and profits.	Agricultural Industries Energy Task Force ⁵⁴⁷	We intend to commence collecting and reporting on the profitability measures identified in our Draft Position Paper on our Profitability Reporting Framework. However, if a business achieves greater efficiencies than our allowances then it is able to retain part of the benefits, which would manifest in its actual return on equity being higher than its regulatory allowance.
	Supports the use of overall financial performance measures, particularly given year-to-year performance can be influenced by many factors.	ATCO Gas Australia ⁵⁴⁸	We agree that financial performance measures can be affected by many factors and so we would need to carefully understand the drivers before using them in decision making. Also, as noted above, we consider that careful consideration of performance measures may be useful in informing the cost of debt inputted into the RoR.
	Supports the use of profitability measures and submitted that it considered the profitability of the Queensland networks were substantially higher than other energy supply industry participants.	Queensland Council of Social Services ⁵⁴⁹	As noted above, we consider that careful consideration of financial performance measures may be useful in informing the cost of debt that should be applied in the RoR.
	Supports the use of profitability measures to monitor the actual performance of the RoR. Notes that there is currently no data on the returns actually achieved against which to compare modelled returns.	CRG ⁵⁵⁰	As outlined in our Draft Position Paper, we intend to commence collecting and reporting on profitability measures. This may help inform analysis of our overall regulatory framework. However, as discussed above, whilst careful consideration of profitability measures (from financial statements) may be useful in informing the cost of debt to apply in our RoR, it would not be helpful in setting the return on equity.
Using analysis of historical	Supports the use of profitability as it provides a useful cross-reference	Origin	As discussed above, we consider that profitability measures have limited

⁵⁴⁶ APGA, Submission to the AER – Review of rate of return guideline, May 2018, p 21.

⁵⁴⁷ Agricultural Industries Energy Taskforce, Submission on rate of return issues paper, December 2017, p 3.

⁵⁴⁸ ACTO Gas Australia, *Review of rate of return guideline – issues paper,* December 2017, p 4.

⁵⁴⁹ QCOSS, Submissions on review of Rate of Return Guideline, May 2018, pp 16-17;

⁵⁵⁰ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p 29.

use in informing the RoR (the exception potentially being the cost of debt). However, it may assist in setting expenditure allowances and other regulatory allowances eg, if businesses are systematically incurring costs that are lower than our allowances.	Energy ⁵⁵¹	regarding the relationship between regulatory returns and the broader performance of the business. However, the principles set out in the guideline must maintain primacy in determining how the actual return is derived.	profitability as a 'cross-check' on the RoR
As per our response above, we consider that profitability measures have limited use to inform our RoR. We agree that there are a range of factors that may impact the profitability of businesses.	Ergon Energy and Energex ⁵⁵²	Information on profitability is potentially useful in testing the reasonableness of the AER's determinations. However, considers that it is difficult to use this information directly to inform the rate of return, as there are a range of factors that impact profitability.	
As discussed above, whilst historical profitability measures may be useful in informing the cost of debt to apply in our RoR, it would not be helpful in setting the return on equity – nor would it assist in applying judgement in deciding the return on equity.	CCP ⁵⁵³ CCP ⁵⁵⁴	Historical profitability measures can be used to assess whether actual profitability has been higher or lower than the allowed RoR, and other comparable businesses. However, as they are backward looking measures, they provide limited guidance on expected returns. Whilst this information cannot be used at a parameter level, it can inform the overall exercise of judgement in setting the RoR, or reviewing other elements of the regulatory regime. Ergon Energy and Energex additionally submitted that there are a range of factors that can impact profitability, hence it is difficult to use this information to directly inform the rate of return.	

⁵⁵¹ Origin Energy, *Review of rate of return guidelines*, December 2017, p 2.

⁵⁵² Ergon Energy and Energex, *Issues Paper – Review of the Rate of Return Guidelines*, December 2017, p 3.

⁵⁵³ CCP16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp 26-28.

⁵⁵⁴ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, pp 66-67.

4 Benchmark gearing ratio

In chapter 3 we considered that the allowed rate of return should be calculated as the weighted average of the return on debt and return on equity (the weighted average cost of capital or WACC). The gearing ratio is used to weight the expected required returns on debt and equity to derive the WACC.

Our decision is to adopt a gearing ratio for the purpose of deriving the WACC⁵⁵⁵ to be 60 per cent. This is based on a benchmarking approach and examining relevant empirical evidence based on market data. We are satisfied that a 60 per cent gearing ratio, and our empirical benchmarking approach to estimating this ratio, will contribute to the achievement of the legislative objectives to the greatest degree.

We consider an empirical benchmarking approach will contribute to the achievement of the legislative objectives because it both provides incentive for service providers to adopt efficient gearing structures and also prevents exposing consumers to variabilities of gearing levels adopted by individual service providers.⁵⁵⁶ Empirically estimating gearing ratios is also consistent with our empirical estimation of equity beta and credit rating.⁵⁵⁷

Our empirical estimation of a benchmark gearing ratio is based on evidence from a comparator set of listed Australian service providers. We consider that the gearing ratios of Australian service providers will most closely reflect the regulatory and commercial risks involved in providing regulated services. Benchmarking against listed service providers allows us to consider market gearing values. Market values have been accepted by our engaged experts as being more appropriate than book values.⁵⁵⁸ The use of market and book values, and of the appropriate comparator set, is further considered in section 4.2 below.

As outlined in chapter 2, we consider that an empirical benchmarking approach using market data and a comparator set reflective of the risks involved in providing regulated services will contribute to the achievement of the legislative objectives.

⁵⁵⁵ We note that we are currently reviewing our regulatory tax approach. Under our current approach the benchmark gearing ratio is an input into our regulatory tax allowances as it determines the tax interest shield. Our consideration of gearing in this decision is only for the purposes of weighting the returns on debt and equity, and have not considered the appropriate interest tax shield here.

⁵⁵⁶ All else equal, variabilities in gearing levels lead to different rates of return and consequently different prices across service providers.

⁵⁵⁷ In addition to weighting the returns on debt and equity to form a WACC, the gearing ratio can affect the leverage risk of a firm. We expect leverage risk to have an effect on equity beta and be a factor in the considerations of credit rating agencies.

⁵⁵⁸ CEPA, *Expert Joint Report,* April 2018, p.27.

Results of our empirical analysis are presented in section 4.1 below.⁵⁵⁹ This analysis shows that the average gearing level of our comparator set over the past 5 years is 54 per cent, and over the past 10 years is 61 per cent, based on market values. We have estimated averages over the past 5 and 10 years as we consider that this provides enough observations to make a reliable estimate. Having regard to both 5 and 10 year historical averages allows us to consider both the relevancy of more recent data and the reliability of a longer time period. The sample period and data set is further considered in section 4.2 below.

Our estimation method is consistent with the approach adopted in the 2013 Rate of return guidelines and 2009 review of WACC parameters.⁵⁶⁰

4.1 Updated empirical estimates

⁵⁵⁹ Our empirical evidence was based on the financial reports of closely related comparators along with the data provided by Bloomberg. The estimates from Bloomberg were broadly consistent with our estimates.

⁵⁶⁰ AER, Rate of return guideline, December 2013; AER, Final Decision, Review of the WACC parameters, May 2009.

Table 14 presents gearing estimates for five comparator businesses⁵⁶¹ over the past ten years using market values of equity and debt (with book value of debt used as a proxy for the market value of debt). The average gearing level of our comparator set over the 10 years to 2017 is 61 per cent, and 54 per cent in the last 5 years to 2017.

⁵⁶¹ Choice of the comparator businesses is discussed in section 2.4.

	ENV	ΑΡΑ	DUE	AST	SKI	AVE
2006	66%	51%	79%	56%	60%	62%
2007	65%	59%	67%	55%	57%	61%
2008	77%	73%	76%	59%	70%	71%
2009	75%	68%	80%	70%	70%	73%
2010	74%	61%	80%	64%	65%	69%
2011	66%	53%	79%	64%	62%	65%
2012	63%	47%	72%	59%	59%	60%
2013	53%	46%	71%	57%	62%	58%
2014	47%	45%	64%	58%	55%	54%
2015	N/A	50%	62%	59%	59%	58%
2016	N/A	49%	51%	57%	53%	52%
2017	N/A	49%	N/A	52%	51%	51%
5 year average	50%	48%	62%	56%	56%	54%
10 year average	65%	54%	70%	60%	60%	61%

Table 14 AER gearing estimates based on market values

Notes: ENV is Envestra Limited, APA is APA Group, DUE is DUET Group, AST is AusNet Services, and SKI is Spark Infrastructure. SKI estimates are as at 31 December each year. AST estimates are as at 31 March each year. All other estimates are as at 30 June each year. The average for all firms in a year does not make any adjustment for these timing differences.

Source: Annual reports and financial statements for each company, AER analysis

Table 15 presents gearing estimates for five comparator businesses over the past ten years using book values of both equity and debt. The average gearing level of our comparator set over the 10 years to 2017 is 70 per cent, and 68 per cent in the last 5 years to 2017.

	ENV	APA	DUE	AST	SKI	AVE
2006	91%	67%	82%	57%	81%	76%
2007	90%	69%	75%	57%	80%	74%
2008	82%	71%	76%	58%	89%	75%
2009	80%	70%	79%	67%	85%	76%
2010	79%	68%	79%	62%	66%	71%
2011	78%	63%	77%	60%	69%	70%
2012	78%	64%	77%	61%	68%	70%
2013	71%	63%	79%	61%	68%	68%
2014	71%	65%	76%	64%	67%	69%
2015	N/A	68%	74%	69%	69%	70%
2016	N/A	71%	65%	66%	68%	67%
2017	N/A	71%	N/A	64%	68%	68%
5 year average	71%	68%	73%	65%	68%	68%
10 year average	77%	68%	76%	63%	72%	70%

Table 15 AER gearing estimates based on book values

Source: Annual reports, AER analysis

4.2 Estimation approach and response to submissions

The approach to estimating a benchmark gearing ratio set out in our 2013 Guidelines was discussed in our February 2018 gearing discussion paper.⁵⁶² In this decision we have maintained this approach. We note that there was a high degree of agreement among stakeholders⁵⁶³ and experts participating in the concurrent expert evidence sessions that this approach to estimating gearing remains appropriate.

CCP16, SA Power Networks, CitiPower, Powercor, Australian Gas Infrastructure Group, United Energy, Ergon Energy, Energex, ENA, Evoenergy, APA, APGA, Jemena, and the Network Shareholder Group all supported maintaining this approach. In contrast, the Agricultural Industries Energy Taskforce,⁵⁶⁴ Canegrowers,⁵⁶⁵ and MEU⁵⁶⁶ submitted that the gearing approach may result in a gearing ratio that is too

⁵⁶² AER, Discussion Paper – Gearing, February 2018.

⁵⁶³ See section 4.3 - Summary of submissions

⁵⁶⁴ Agriculture Industries Energy Taskforce, Submission on rate of return issues paper, Dec 2017, p. 10.

⁵⁶⁵ Canegrowers, Submission on rate of return issue paper, Dec 2017, p. 3.

⁵⁶⁶ MEU, Submission on rate of return issue paper, Dec 2017, p. 3.

high. The CCP16 submitted that the gearing ratio is unlikely to have a material effect on the overall rate of return, stating:⁵⁶⁷

changes in the gearing have a relatively small effect on the ROR. This is because as gearing rises so does the equity beta due to the greater concentration of risk on equity. Hence, a higher gearing does not necessarily result in a lower ROR as the benefit of greater use of lower-cost debt is offset by the higher cost of equity.

Given this, while the AER should update the underlying empirical analysis and review its approach, we do not expect, a priori, that the AER should change its approach.

These submissions are considered in the sections below and in section 4.3.

Market and book values of equity

In our 2013 Guidelines we considered gearing estimates from both market and book values of equity, with primary weight on estimates from market values.

The Network Shareholder Group, APA, APGA, and the ENA all submitted that market values should be used to estimate gearing.⁵⁶⁸ The Network Shareholder Group noted that book values are "simply a historical value and will almost never have an impact on the cost of financing debt or equity".⁵⁶⁹ Similarly, the ENA submitted that the rate of return reflects the market-clearing cost of capital and other rate of return parameters are based on market values, therefore gearing should also be derived from market values.⁵⁷⁰

CCP16 submitted that we should consider both book and market values, that primary weight should be on market values, but that book values allow for a larger sample of firms.⁵⁷¹

In the concurrent expert evidence sessions the experts agreed that market-based estimates are the only appropriate measure of gearing.⁵⁷²

We consider that primary weight should be placed on gearing estimates from market values. We note that the use of market values promotes consistency between our benchmark gearing ratio and other rate of return parameters that are typically informed by market data. We consider this is important given the relationship between leverage risk and equity beta, and the estimation of equity beta from returns data of listed equity.

⁵⁶⁷ CCP16, Submission of Rate of Return Guideline review, May 2018, p. 50.

⁵⁶⁸ Network shareholder group, Submission of Rate of Return Guideline review, May 2018, p. 14; APA, Submission of Rate of Return Guideline review, May 2018, p. 20; APGA, Submission of Rate of Return Guideline review, May 2018, p. 14; ENA, Submission of Rate of Return Guideline review, May 2018, p. 32

⁵⁶⁹ Network shareholder group, Submission of Rate of Return Guideline review, May 2018, p. 14.

⁵⁷⁰ ENA, Submission of Rate of Return Guideline review, May 2018, p. 32.

⁵⁷¹ CCP16, Submission of Rate of Return Guideline review, May 2018, p. 47.

⁵⁷² AER, Evidence session 1 & 2, Expert Joint Report, April 2018, p. 27.

Comparator set

The ENA and the Network Shareholder Group submitted that are current comparator set remains appropriate.⁵⁷³ The Network Shareholder Group submitted that a comparator set of Australian listed energy networks provides a reasonable reflection of the degree of risk involved in providing regulated services, and that Australian firms are most relevant to ensure equivalence of tax regime and legal framework for bankruptcy.⁵⁷⁴

CCP16 submitted that given the small number of service providers for which data is available, we should also consider the gearing benchmarks used by other regulators. CCP16 submitted that while gearing may be affected by country and sector specific factors, such as thin capitalisation rules, length of the regulator period, and the design of incentive mechanisms, the gearing assumptions used by other regulators in the sector can help establish a plausible range for the gearing benchmark.⁵⁷⁵

We consider that our current comparator set and sample period provides sufficient data for a reliable gearing estimate, and do not consider that adding gearing estimates from other sectors or countries is required. We also note that there are interrelationships between leverage risk and equity beta, and leverage risk and the return on debt. The overall level of risk of providing regulated services may be an important consideration for investors and managers of service providers. As the overall level of (systematic) risk of providing regulated services may differ between sectors and countries, we consider it is appropriate to place greater weight on gearing estimates from Australian listed service providers.

Sample period

In our 2013 Guidelines we considered gearing estimates from comparable businesses over a historical ten year period.⁵⁷⁶ CCP16 submitted that long term averages of at least 5 years should be used.⁵⁷⁷ APGA submitted that the benchmark gearing ratio should be based upon a long-run average.⁵⁷⁸

In the concurrent expert evidence sessions the experts agreed that, conceptually, gearing should not change regularly as the core capital structure decisions of companies are stable.⁵⁷⁹ The experts agreed that gearing choices typically reflect a long-term investment strategy so market evidence should be averaged over 5-10

⁵⁷³ ENA, Submission of Rate of Return Guideline review, May 2018, pp. 32; Network shareholder group, Submission of Rate of Return Guideline review, May 2018, pp. 14.

⁵⁷⁴ Network shareholder group, Submission of Rate of Return Guideline review, May 2018, pp14.

⁵⁷⁵ CCP16, Submission of Rate of Return Guideline review, May 2018, pp51.

⁵⁷⁶ AER, Explanatory Statement - Draft rate of return guideline, August 2013, pp. 179-180.

⁵⁷⁷ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p. 52.

⁵⁷⁸ APGA, Submission of Rate of Return Guideline review, May 2018, pp14.

⁵⁷⁹ CEPA, Evidence session 1 & 2, Expert Joint Report, April 2018, pp. 28.

years. The experts also agreed that share price movements and changes in the market capitalisation of a listed company could distort short run gearing estimates.⁵⁸⁰

We agree with these submissions for the reasons they have provided, and have considered long term (five and ten year) averages of annual gearing estimates. We also consider that it is generally desirable to have a consistent approach to estimating rate of return parameters and have therefore tended to adopt a consistent period and frequency across all rate of return parameters (where possible). We note that our empirical analysis of equity beta and credit ratings involves consideration of data over a relatively long time period of five to ten years or more.

Hybrid securities

Hybrid securities are securities that have characteristics of both debt and equity. In our 2013 Guidelines we noted that Envestra and Spark Infrastructure had shareholder loan notes that are included as debt for accounting purposes but had the following characteristics of equity:

- they were stapled to each share, with no separate existence without the share (that is, they cannot be traded independently),
- they were subordinate to all other creditors; and
- returns on the notes were not guaranteed and only payable to the extent to which there is available cash.

In our 2013 Guidelines we considered that these loan notes should be treated as equity.

Since publishing our discussion paper we have become aware of additional loan notes attributable to Spark Infrastructure that were not adjusted for in the gearing estimates in the discussion paper. Consistent with the treatment of other loan notes, we have removed these additional loan notes from our measures of debt when estimating gearing ratios in this decision.⁵⁸¹

More recently, AusNet Services has successfully priced two hybrid security issues in the form of non-convertible subordinated notes. These subordinated notes differ from shareholder loan notes of Envestra and Spark Infrastructure in that they are not stapled to the shares of AusNet Services.

The CRG submitted that given the significantly reduced risk of default in regulated utilities, the risk profile of subordinated and unsubordinated debt is very similar and should be afforded the same return. CRG submitted that subordinated debt has a lower risk profile than equity, and can't simply be afforded the same return as equity.

⁵⁸⁰ CEPA, Evidence session 1 & 2, Expert Joint Report, April 2018, pp30.

⁵⁸¹ We have also updated our gearing estimates for Spark Infrastructure to account for their ownership stake in Transgrid, which was not accounted for in estimates presented in our discussion paper.

The CRG submitted that this would over-reimburse infrastructure owners and may require a change to the binary debt v equity treatment of returns.⁵⁸²

Given the relative size of AusNet Services' current level of debt and hybrid securities, we note that adjusting for these hybrid securities is unlikely to have a material impact on the overall gearing estimates. For this reason, and noting that the subordinated notes are not stapled to shares, we have not removed these hybrid securities from measures of AusNet Services debt.

4.3 Summary of submissions

This section details the submissions we have received from stakeholders as part of our review process to date, and notes how we have had regard to each submission.

Key Point	Submission	Stakeholder	AER Response
Benchmark gearing ratio	Benchmark gearing ratio should remain at 60 per cent	Joint Energy Networks, APA, APGA, CCP16, CRG, ENA, Evoenergy, NSG, Jemena, Ergon Energy, Energex ⁵⁸³	Our decision is for a benchmark gearing ratio of 60 per cent.
	Incorrect gearing assumptions will result in the AER providing 'return on capital' allowances well above the required levels	Agricultural Industries Energy Taskforce ⁵⁸⁴	We consider that our benchmark gearing ratio of 60 per cent, as based on an empirical benchmarking approach, will contribute to the achievement of the legislative objectives.
	Our current estimate appears to be a departure from the gearing that service providers apply to the regulated parts of their businesses.	Canegrowers ⁵⁸⁵	Our benchmark gearing estimate is based on empirical analysis of listed Australian service providers. We note that for SKI, ENV, AST, and DUE a relatively large proportion of total revenue is (was) regulated.
	The gearing approach needs to be modified to reflect the reality that the RAB includes a significant element of	MEU ⁵⁸⁶	This issue was considered in detail in our 2017 review of our regulatory treatment of inflation. We consider that our current treatment of inflation, including indexation of the

Table 16 Summary of submissions on benchmark gearing

⁵⁸² CRG, Submission of Rate of Return Guideline review, May 2018, pp. 38-39

⁵⁸³ CCP16, Submission of Rate of Return Guideline review, May 2018, p. 50; CRG, Submission of Rate of Return Guideline review, May 2018, pp21; APA, Submission of Rate of Return Guideline review, May 2018, p.21; Joint Energy Network, Submission of Rate of Return Guideline review, May 2018, p. 32; APGA, Submission of Rate of Return Guideline review, May 2018, p.14; ; ENA, Submission of Rate of Return Guideline review, May 2018, p.7; Evoenergy, Submission of Rate of Return Guideline review, May 2018, p.14; ; Jemena, Submission of Rate of Return Guideline review, May 2018, p.14; ; Ergon Energy & Energex, Submission of Rate of Return issues paper, December 2017, pp. 3-4

⁵⁸⁴ Agricultural Industrial Energy Taskforce, Submission of Rate of Return Guideline review, December 2017, p.10.

⁵⁸⁵ Canegrowers, Submission on Rate of return issues paper, December 2017, p.3.

⁵⁸⁶ MEU, Submission on Rate on return issues paper, December 2017, p.10

	retained inflation and assets that do not contribute to the provision of the services.		regulatory asset base, will contribute to the achievement of the legislative objectives.
Market vs RAB	Gearing should be estimated on market value.	APA, APGA, CCP16, ENA, NSG, Ergon Energy, Energex ⁵⁸⁷	We consider that the use of market values is most reflective of the market's assessment of risks involved in providing regulated services.
Parent company debt	The approach to parent company debt and 'double leverage' used in our 2013 Guidelines is appropriate.	CCP16 ⁵⁸⁸	Our decision is based on an empirical approach that is consistent with the treatment of double leverage used in our 2013 Guidelines. That is, we have adjusted Spark Infrastructure's reported gearing measure to account for its minority ownership in SA Power Networks, Victoria Power Networks, and Transgrid.
Sample period	Gearing should be measured over the long term and the benchmark based on observed levels of gearing over at least 5- 10 years	CCP16, NSG, APA ⁵⁸⁹	We have considered both 5 and 10 year empirical estimates.
Hybrid securities	The approach adopted in our 2013 Guidelines regarding hybrid securities and loan notes should be continued.	CCP16 ⁵⁹⁰	Our decision is based on an approach to hybrid securities and loan notes that is consistent with that of our 2013 Guidelines.
	The risk profile of subordinated and unsubordinated debt is very similar and should be afforded the same return.	CRG ⁵⁹¹	We consider that subordinated debt that is stapled to shares (such as Envestra and Spark Infrastructure's loan notes) must be treated as equity. However, we have not adjusted our measure of AusNet Services debt to remove its unstapled subordinated debt.
	Subordinated debt has a lower risk profile than equity, and can't simply be afforded the same	CRG ⁵⁹²	We consider that subordinated debt that is stapled to shares (such as Envestra and Spark Infrastructure's loan notes) must be treated as

⁵⁸⁷ CCP16, Submission of Rate of Return Guideline review, May 2018, p.52; APA, Submission of Rate of Return Guideline review, May 2018, pp 20-21; NSG, Submission of Rate of Return Guideline review, May 2018, p.14; APGA, Submission of Rate of Return Guideline review, May 2018, p.14; ENA, Submission of Rate of Return Guideline review, May 2018, p. 32-34; Ergon Energy & Energex, Submission of Rate of Return issues paper, December 2017, pp. 3-4

⁵⁸⁸ CCP16, Submission of Rate of Return Guideline review, May 2018, pp. 52-53.

⁵⁸⁹ CCP16, Submission of Rate of Return Guideline review, May 2018, p.52; APA, Submission of Rate of Return issues paper, December 2017, p 4; NSG, Submission of Rate of Return Guideline review, May 2018, p.14

⁵⁹⁰ CCP16, Submission of Rate of Return Guideline review, May 2018, p.52.

⁵⁹¹ CRG, Submission of Rate of Return Guideline review, May 2018, pp. 38-39

⁵⁹² CRG, Submission of Rate of Return Guideline review, May 2018, pp. 38-39

	return as equity.		equity. However, we have not adjusted our measure of AusNet Services debt to remove its unstapled subordinated debt.
Appropriate comparators	Given the small number of service providers for which data is available, we should also consider the gearing benchmarks used by other regulators to define a plausible range for the benchmark gearing.	CCP16 ⁵⁹³	We consider that a comparator set of Australian service providers will best reflect the regulatory and commercial risks involved in providing regulated services. We consider that there is sufficient data from this comparator set.
	The AER should consider different gearing ratios for different industries and/or sectors.	APA ⁵⁹⁴	As set out in section 2.4, we consider that the regulatory and commercial risks involved in providing regulated gas and electricity services are sufficiently similar to warrant a single benchmark.
	It is unclear on how the AER has approached the task of stripping from the market data the consequence of these firms not being pure play providers of regulated services.	ECA, EUAA ⁵⁹⁵	We acknowledge that the firms in our comparator set have varying degrees of unregulated activities. We take this into account when exercising our regulatory judgment in determining a benchmark gearing ratio. However, we have not calculated any explicit adjustments to the data for this issue, and we consider that this comparator set is the best available.
	It is unclear to what extent the AER has assessed that each of the entities is 'efficient.'	ECA, EUAA ⁵⁹⁶	 We have not formed a view on the efficiency of the firms in our comparator set. Rather, as set out in chapter 2, we consider that a regulatory gearing ratio that is: set ex ante, benchmarked to firms with a similar degree of regulatory and commercial risks as those involved in providing regulated services, and based on market data, will promote efficiency and the contribute to the achievement of the legislative objectives.

- ⁵⁹³ CCP16, Submission of Rate of Return Guideline review, May 2018, p.52
- ⁵⁹⁴ APA, Submission of Rate of Return Guideline issues paper, December 2017, p. 5.
- ⁵⁹⁵ ECA, Submission of Rate of Return Guideline issues paper, December 2017, p. 17; EUAA, Submission of Rate of Return Guideline issues paper, December 2017, pp. 7-8.
- ⁵⁹⁶ ECA, Submission of Rate of Return Guideline issues paper, December 2017, p. 17; EUAA, Submission of Rate of Return Guideline issues paper, December 2017, pp. 7-8.

Gearing and tax Gearing and tax The approach to benchmarking gearing is providing an excessive tax allowance. ECA ⁵⁹⁷	gearing ratio for the purposes of weighting the returns on debt and equity. Our review of tax will consider the role of a gearing ratio in calculating the allowed tax building block.
--	---

Source: Agriculture Industries Energy Taskforce, Submission on rate of return issues paper, Dec 2017, p. 10. Canegrowers, Submission on rate of return issue paper, Dec 2017, p. 3. MEU, Submission on rate of return issue paper, Dec 2017, p. 3.

⁵⁹⁷ ECA, Submission of Rate of Return Guideline issues paper, December 2017, p. 11

5 Overall approach to return on equity

We estimate the expected return on equity using the approach we developed in our 2013 Guidelines after extensive stakeholder consultation and implemented since then in all of our regulatory determinations. It was affirmed by the Australian Competition Tribunal.^{598 599}

5.1 Foundation model approach

Our approach is known as the foundation model approach and comprises of six steps as set out below.

Step 1 - identify relevant material

Step 2 – determine role/ how best to employ relevant material including determining the foundation model (SLCAPM)

Step 3 – implement foundation model. Determine SLCAPM input parameter ranges and point estimates.

Step 4 – other information. Estimate other information used to inform overall return on equity

Step 5 – evaluate information from step 3 and 4.

Step 6 – distil return on equity point estimate. Use SLCAPM point estimate as starting point and select final return on equity value having regard to information from steps 4 and 5.

Most experts at the concurrent evidence sessions agreed that we should maintain the foundation model approach and focus on its application in light of the evidence that has evolved. David Johnstone noted the need to consider the whole rate of return rather than focussing only on the model outcome.⁶⁰⁰

Implementing the foundation model is a key step in our six step approach and has survived the test of time. We undertook extensive consultation and assessed a number of models including the dividend growth model (DGM), Black CAPM and the Fama

⁵⁹⁸ Since the Tribunal decision in 2016, disagreements amongst stakeholders on the allowed return on equity was largely driven by differences in opinion on how best to exercise judgment given the uncertainty/imprecision of the evidence, rather than the six step foundation model approach.

⁵⁹⁹ The Australian Competition Tribunal reviewed our return on equity estimate based on our foundation model approach on appeal by stakeholders and found that there was no reviewable error. See, *PIAC – AusGrid*, [2016] *ACompT* 1.

⁶⁰⁰ Cambridge Economic Policy Associates, *Joint Expert Report, RORG review – Facilitation of concurrent evidence* sessions, 21 April 2018, section 2.1.3, p.19.

French model for suitability as the foundation model before deciding on the SLACPM. Our comprehensive assessment is in the 2013 Guidelines.⁶⁰¹

The foundation model approach provides a framework for systematically considering relevant information and then exercising our judgement on the appropriate choice of the regulated return on equity. The approach recognises that our task requires us to exercise judgement because we are estimating a forward looking return on equity that will satisfy the national electricity and gas objectives. Further, the information available to inform our decision is imprecise, incomplete and, to some extent, conflicting.

Submissions were generally supportive of maintaining the foundation model approach:

- The Network Shareholders Group (NSG) supported the continued use of our foundation model approach in this 2018 Guideline.⁶⁰³
- The Energy Networks Australia (ENA) agreed with the general view at the expert concurrent evidence sessions that the foundation model should be a given and the focus should be on updating parameter estimates in light of new evidence since 2013.⁶⁰⁴
- Evoenergy, considered the adoption of the foundation model for the new Guideline appropriate.⁶⁰⁵
- APA accepts the continued use of our foundation model approach at this time but notes that a thorough review of the approach is warranted at the next review (in 2022).⁶⁰⁶ The Australian Pipeline & Gas Association (APGA) submits that our review should start from the current foundation model approach and change only if risks and market conditions have changed materially.⁶⁰⁷
- The CRG raised concerns about the efficacy of using the CAPM but noted that thorough investigation would be required prior to determining an alternative approach.⁶⁰⁸

We agree that the foundation model approach should be continued and adopted in this 2018 Guideline review for estimating the return on equity. In this context, we will update the relevant data and review new evidence so that our judgement can be exercised within the established approach to estimating the allowed return on equity. We consider that this provides the necessary certainty and predictability that

https://www.aer.gov.au/networks-pipelines/guidelines-schemes-models-reviews/rate-of-return-guideline-2013

⁶⁰¹ AER, Explanatory statement, Rate of Return Guideline, 2013, section 5; AER, Explanatory statement appendices, Rate of Return Guideline, 2013, Appendix A – Assessment of models. Available at:

⁶⁰² Section 2.4 of this decision discusses submissions relating to our use of the CAPM in this 2018 review process.

⁶⁰³ NSG, Submission on the RORG review, 4 May 2018, p.12.

⁶⁰⁴ ENA, Response to discussion papers and concurrent evidence sessions, 4 May 2018, 44.

⁶⁰⁵ Evoenergy, *Review of the rate of return guideline – evidence sessions*, 4 May 2018

⁶⁰⁶ APA, Submission in response to discussion papers and expert evidence, 4 May 2018, p.1.

⁶⁰⁷ APGA, Submission to the AER review of the rate of return guideline, 4 May 2018, p.2.

⁶⁰⁸ CRG, Submission to the AER rate of return guideline review, May 2018, p.37.

stakeholders have said they value whilst allowing us to conduct our regulatory task in a manner that is most likely to contribute to the legislative objectives.⁶⁰⁹

We acknowledge that these Guidelines could be binding unlike the 2013 Guidelines, for which the current legislative framework allows both the service providers and ourselves the opportunity to depart if the evidence justified that doing so would result in an outcome that better achieves the allowed rate of return objective. Although departures were available we stated that we would not do so lightly as this may undermine certainty and predictability and also undermine the extensive consultation process undertaken in developing the guidelines. We continue to place high importance on certainty and predictability and consider that the estimate of expected return on equity based on our foundation model approach is unbiased, recognises uncertainties and has appropriate regard to all relevant information. Therefore, we consider our foundation model approach is suitable for determining return on equity parameters to use under binding Guidelines.

⁶⁰⁹ This approach is consistent with the CCP16's submission that, incremental change means continuing to work within the CAPM and foundation model framework, not limiting the exercise of our statutory obligations to satisfy a self-imposed 'incremental approach' and considering all available data. See, CCP 16, *Submission to the AER on its rate of return guideline review concurrent evidence sessions*, 4 May 2018, section 3.1, pp. 14-16.



Figure 10 Foundation model approach flowchart

5.2 Identify relevant material and determine role (Steps 1 and 2)

Overall, we have not identified any additional classes of material that we did not consider when preparing our 2013 Guidelines. Therefore, the list of material we employed in 2013 remains appropriate. This conclusion was supported in the concurrent evidence sessions and in submissions.

We also consider that we should assign the same roles to each piece of material as we did in 2013. However, when implementing step 3 within our overall framework, and based on new evidence and current material, we are persuaded that we should adjust the relative merit assigned to some pieces of material in exercising our judgement to determine a return on equity that will contribute to achieving our legislative objectives.

The list of relevant material, the roles assigned and the relative merit in 2018 are set out in Table 17.

Material (step one)	Role in 2013 (step two)	Role in 2018 and relative merit
Sharpe-Lintner CAPM	Foundation model	No change
Black CAPM	Inform foundation model parameter estimates (equity beta)	No change in role. However, at this time we have diminished confidence in the robustness of the Black CAPM and are therefore not persuaded to select an equity beta towards the top of the observed empirical estimates
Dividend growth models	Inform foundation model parameter estimates (MRP)	No change in role. However, at this time we have diminished confidence in the robustness of DGMs and are therefore not persuaded to select an MRP towards the top of the observed empirical estimates of historical excess returns.
Fama–French three factor model	No role	No change
Commonwealth government securities	Inform foundation model parameter estimates (risk free rate)	No change
Observed equity beta estimates	Inform foundation model parameter estimates (equity beta)	No change
Historical excess returns	Inform foundation model parameter estimates (MRP)	No change
Survey evidence of the MRP	Inform foundation model parameter estimates (MRP)	No change
Implied volatility	Inform foundation model parameter estimates (MRP)	No change
Other regulators' MRP estimates	Inform foundation model parameter estimates (MRP)	No change
Debt spreads	Inform foundation model parameter	No change

Table 17 Relevant material and role

	estimates (MRP)	
Dividend yields	Inform foundation model parameter estimates (MRP)	No change
Wright approach	Inform the overall return on equity	We have diminished confidence in the robustness of the Wright approach
Takeover/valuation reports	Inform the overall return on equity	No change
Brokers' return on equity estimates	Inform the overall return on equity	No change
Other regulators' return on equity estimates	Inform the overall return on equity	No change
Comparison with return on debt	Inform the overall return on equity	No change
Trading multiples	No role	No change
Asset sales	No role	No change
Brokers' WACC estimates	No role	No change
Other regulators' WACC estimates	No role	No change
Finance metrics	No role	No change

Source: AER analysis.

5.3 Implement the foundation model (step 3)

In summary, after assessing the relevant evidence we consider the best estimates for the SLCAPM parameters are:

- A formula for calculating the risk free rate based on yields on 10-year Commonwealth Government Securities (CGS)
- A value of 0.6 for equity beta
- A value of 6 per cent for market risk premium.
- These parameter input point estimates and reasons are discussed in chapters 6, 7, and 8.

5.4 Step 4 - 6 of the foundation model approach

Steps 4-6 are where we have regard to other information, evaluate and then determine the final return on our equity point estimate (sometimes referred to as 'cross checks'). Our approach to other information was set out in our 2013 Guidelines.⁶¹⁰

⁶¹⁰ AER, *Explanatory statement appendices, Rate of Return Guideline*, 2013, Appendix B – Other information.

Whilst there was consensus that we should apply our foundation model approach, we note that experts and stakeholders submitted divergent views on the role of cross checks. These concerns were mainly focused on the role of additional information to inform the AER in exercising its judgement on the appropriate overall rate of return. Nevertheless, these comments although not directly concerning the return on equity, have some relevance to our foundation model approach steps 4-6, as we apply cross checks.

Experts at our concurrent evidence session discussed whether there is an objective basis for an overall test of reasonableness of the rate of return. Some considered that all relevant evidence should be considered together rather than reserving some for 'cross checking' and noted the difficulty in finding a consistent and objective method for 'cross checking'. Some also expressed concern that 'cross checking' could result in 'backdoor' discretion and limit transparency. Graham Partington considered that there is a role for reasonableness checks and notes that doing otherwise could lead to unreasonable estimates of the cost of capital.⁶¹¹

CCP16 submitted that while the rate of return should be considered as a whole, cross checks are essential to judgement and discretion and inclusion would improve the overall rate of return and reduce biases at the parameter level.⁶¹² It added that the current foundation model approach provides a sound framework that already provides for transparent and reasoned consideration of cross-checks in the exercise of discretion to determine the overall rate of return and parameter values.⁶¹³ The point was also made that without a sensibility check, too much faith is placed on the CAPM model and the precision of parameter models, than can be rationally supported ⁶¹⁴

The Network Shareholder Group agreed with most experts and noted that 'cross check' can either be 'elevated above all other considerations' or have no effect in determining the allowed rate of return.⁶¹⁵ They also noted that 'cross checks' weaken the quality of the process because, if action is taken, then the final adjustment is elevated above all other considerations circumventing rigour and transparency. If no action is taken, the Network Shareholder Group submitted that the cross check has no effect.

We consider our steps 4-6 are an integral part of our return on equity approach, although we acknowledge their limitations. These steps—in conjunction with the use of a foundation model—provides an appropriate balance between a relatively replicable and transparent process and providing flexibility to consider market circumstances. We consider this provides scope for more openness and flexibility to test the

⁶¹¹ CEPA, *Expert Joint Report, RORG review – Facilitation of concurrent evidence sessions*, 21 April 2018, section 2.1.3.

⁶¹² CCP 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 24.

⁶¹³ CCP 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 25.

⁶¹⁴ CCP 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 28.

⁶¹⁵ Network Shareholder Group (NSG), Submission on the Rate of Return Guideline review, 4 May 2018, p.12
reasonableness of the final return on equity point estimate. This recognises that, ultimately our rate of return must meet legislative objectives and requires the exercise of judgement. Any potential adjustments will be reasoned against our legislative objectives. Based on the evidence in steps 4 and our evaluation under step 5, we may reconsider the foundation model input parameter estimates, or more fundamentally, we may also reconsider the foundation model itself. That said, we consider it reasonable to expect our final return on equity estimate, in most market circumstances, to fall within the foundation model range.

5.4.1 Step 4 – other information.

Under step 4, we set out the form of other information that will inform the overall return on equity estimate. As noted above under steps 1-2, we have assessed the information currently before us and found that no new material was presented leading us to change the list of material we employed or the roles we allocated to that material in 2013.

In section 3.3 we considered the information we received relating to profitability analysis, financeability analysis and RAB multiples as potential relevant material for estimating the rate of return. Our conclusion is that this material should not be given a role in estimating the expected return on equity. However, we agree that RAB multiples and historical profitability may provide useful contextual information and cause for further examination of the material we rely on when estimating rate of return parameters.

The additional information we will consider under step 4 is in Table 18 and is consistent with our 2013 Guidelines.⁶¹⁶

Table 18 Other relevant information

Additional information	Form of information
Wright approach	Point in time
Other regulators' return on equity estimates	Point in time
Brokers' return on equity estimates	Point in time and directional
Takeover/valuation reports	Directional
Comparison with return on debt	Relative

Table 19 sets out the return on equity and equity risk premium range derived from other information.

⁶¹⁶ AER, *Better Regulation, Rate of Return Guideline*, December 2013, p.16.

	Return on equity		Equity risk premium	
	Minimum	Maximum	Minimum	Maximum
AER Foundation Model*	4.84	7.84	2.2	5.2
Wright approach CAPM	5.6	10.6		
Independent Valuation reports617	9.03	9.23	4.44	4.62
Broker Reports - Unadjusted	6.2	8.4	3.5	5.0
Broker Report – Adjusted for Imputation	6.7	9.0	4.1	4.9
Other regulators decisions**	6.8	11.85	4.2	9.36

Table 19 Return on equity range from other information

* Risk free rate of 2.64 percent. Equity beta range 0.4–0.8 and MRP of 5.5-6.5 percent.

** Period from March 2017 to March 2018.

5.4.2 Step 5

Under step 5, we evaluate the other information set. In undertaking this evaluation, consistent with our 2013 Guidelines, we may have regard to matters including:

- Patterns shown in other information
- The strengths and limitations of the other information
- The magnitude by which the other information suggests that the foundation model point estimate under or over estimates the expected return on equity (if at all)⁶¹⁸.

5.4.2.1 The Wright Approach

The Wright approach is in effect, a model that assumes a stable total market return, and by implication a perfect inverse relationship between the market risk premium and the risk free rate. Since the 2013 Guidelines we have received evidence that the model has no theoretical basis in Australia and is not an appropriate tool for regulatory use, nor is it used by market practitioners.⁶¹⁹ This view was supported at the expert sessions, but the converse view was also noted, that is, the return on equity does not

⁶¹⁷ Based on the most recent valuation report which is a KPMG report for DUET released on 7 March 2017.

⁶¹⁸ AER, *Better Regulation, Rate of Return Guideline*, December 2013, p.16.

⁶¹⁹ Rankin and Idil, A century of Stock-Bond Correlations, September 2014, Partington and Satchell, Cost of Equity issues 2016 Electricity and Gas Determinations, April 2016, pp30-31; Partington and Satchell, Report to the AER, May 2018, p.34-35, AER, Draft decision - Multinet Gas access arrangement 2018-22, Attachment 3 - Rate of return, p.220. Our analysis of independent valuation reports for the 2018 rate of return guideline review also indicated no reports appeared to use the Wright CAPM.

necessarily move one-for-one with movements in the risk free rate.⁶²⁰ We therefore continue to observe results from the Wright model.

We estimate the return on equity under the Wright approach using a range for the longterm historical average return on the market. We use a range because the estimated return on the market will vary depending on the time period used.⁶²¹ The table below sets out our estimates of historical returns on the market portfolio. The nominal return ranges from 10.1 per cent to 12.6 per cent.

Sampling period	Market return (real)	Market return (nominal)
1883–2017	8.6	11.3
1937–2017	7.4	10.1
1958–2017	8.9	11.6
1980–2017	9.8	12.6
1988–2017	9.2	11.9

Table 20 Historical returns on the market portfolio (per cent)

Source: Handley, *An estimate of the historical equity risk premium for the period 1883 to 2011*, April 2012, p. 6. AER update for 2012–2017 market data.

Notes Historical market returns are estimated using arithmetic averages, assuming a theta value of 0.6, and assuming an inflation rate of 2.5 per cent. Nominal figures calculated by the AER using the Fisher equation: $1+i=(1+r)x(1+\pi)$ where r denotes the real return, i denotes the nominal return and π denotes the inflation rate.

We estimate a return on equity under the Wright CAPM⁶²² by combining the historical nominal market return with our prevailing risk free rate estimate⁶²³ and equity beta estimate.⁶²⁴ As shown in Table 21, our estimated range for equity beta and market return results in Wright CAPM return on equity estimates ranging from 5.6 to 10.6 per cent.

Table 21 Wright CAPM return on equity (per cent)

AER equity beta estimate	Wright CAPM return on equity based on 10.1 market return	Wright CAPM return on equity based on 12.6 market return
0.4	5.6	6.6
0.8	8.6	10.6

Source: AER analysis.

Notes: Based on a final risk free rate estimate of 2.64 per cent.

⁶²⁰ AER, Second Concurrent Evidence Session, 5 April 2018, p. 69, 72.

⁶²¹ AER, Explanatory statement: Rate of return guideline (appendices), December 2013, pp. 26–27.

⁶²² See section 0 for details on the Wright CAPM.

⁶²³ Our risk free rate estimate is 2.64 per cent.

⁶²⁴ Our estimated range for equity beta is 0.4 to 0.8.

5.4.2.2 Other regulators return on equity estimates

Figure 11 shows our estimate of the equity risk premium of 3.6 per cent is below those from other regulators. However, we note that, with the exception of the ERA, other Australian regulators do not set revenue determinations for regulated distribution and transmission energy network services.

Figure 11 Equity risk premium estimates from other regulators' decisions



Source: AER analysis of other Australian regulators since 2017

5.4.2.3 Brokers return on equity estimates

The table below shows the estimates of return on equity and premium above the risk free rate contained in the broker reports dated from March 2017 to May 2018.

Table 22 Recent broker reports (per cent)

		Return on equity	Equity risk premium
Broker estimate—no imputation adjustment	Minimum	6.2	3.5
Broker estimate—no imputation adjustment	Maximum	8.4	5.0
Broker estimate—adjusted for imputation	Minimum	6.7	4.1
Broker estimate—adjusted for imputation	Maximum	9.0	4.9

Source: AER analysis of broker reports, dated 30 March 2017 to May 2018 that include a valuation for AusNet Services, Spark Infrastructure, APA Group, and/or DUET Group.

We observe from the table above that the equity risk premium from the AER's foundation model of 3.6 per cent is at the lower end of the range of the average premiums estimated by brokers. Directionally, as shown in Figure 12 below, the broker equity risk premium has remained within similar range for the duration of 2017 and 2018 although there has been a slight downward movement since the end of 2017. Our equity risk premium estimate is below the bottom of the unadjusted range.



Figure 12 Equity risk premium estimates from broker reports

Source: AER analysis of broker reports that include a valuation for AusNet Services, Spark Infrastructure, APA Group, and/or DUET Group.

Notes: Average broker ERP is the mean of estimates from all brokers and for all businesses available at the time

Figure 13 shows the equity risk premium in broker reports since 2017. We note that they cluster between 3.5–4.2 per cent which is consistent with our value of 3.6 per cent.



Figure 13 Clustering of equity risk premium estimates from broker reports

Source: AER analysis of broker reports that include a valuation for AusNet Services, Spark Infrastructure, APA Group, and/or DUET Group.

5.4.2.4 Takeover and valuation reports

The figure below outlines the range of return on equity and equity risk premium estimates from relevant independent valuation reports. The most recent report for a regulated energy business is KPMG's report for DUET released on 7 March 2017. This report implies an equity risk premium of 4.44 to 4.62 per cent (without adjustment for dividend imputation).



Figure 14 Equity risk premium from relevant valuation reports over time

Source: AER analysis of reports from Thomson Reuters

Notes: We have shown the equity risk premium based on a nominal vanilla WACC, expert reports using a different WACC form have been adjusted accordingly. This equity risk premium ('Valuers estimate-high') also reflects the impact of any discretionary uplifts applied by the independent valuer.

We have considered information from independent valuation reports. There have been only 19 relevant independent valuation reports spanning a period going back to 1991.⁶²⁵ Only 13 reports included a discounted cash flow analysis with information on a return on equity estimate. These 13 reports were provided by only four independent valuation firms, with 9 of the 13 reports being provided by Grant Samuel & Associates.

As a result, we consider that the number of reports is too low and the concentration of reports among only a few valuers is too high to be able to place significant reliance on the evidence from valuation reports.

We also note that the ranges for return on equity and equity risk premium estimates contained in Figure 14 include the final values used in the independent valuation reports and reflect any uplifts applied. However, as noted in previous decisions we have concerns about the applicability of these uplifts to the legislative objectives.⁶²⁶ These uplifts may reflect a range of factors that do not warrant inclusion/consideration in the rate of return (for example, non-systematic risks, term structure of the chosen equity proxies, the relevant investment period exceeding the term of the proxies).

⁶²⁵ Our search found reports going back to 1991, but contains no reports between 1991 and 1998 for comparable electricity or gas network businesses.

⁶²⁶ For example, see: AER, Draft decision Multinet Gas Access Arrangement 2018–2022 Attachment 3–Rate of return, July 2017, p. 102. Available at: <u>https://www.aer.gov.au/networks-pipelines/determinations-accessarrangements/multinet-gas-access-arrangement-2018-22/draft-decision</u>

We also have concerns that the adjustment for dividend imputation may not be appropriate as it is not clear the extent to which these estimates may be based on third party estimates that already account for the value of imputation credits.⁶²⁷ Given the absence of sufficient information to support any precise adjustment, the risk premium appropriately reflecting dividend imputation is likely somewhere between the adjusted and unadjusted premiums, but we are unable to distil a precise estimate due to a lack of transparency in valuation reports.

5.4.2.5 Comparison with return on debt

We consider the debt risk premium (DRP) to be a valuable relative indicator of the reasonableness of our ERP. ⁶²⁸ Both provide an indication of the relative risk faced by investors. The advantage of the DRP is that it is easier to observe. Figure 15 shows the comparative and relative positions of the equity and debt risk premia.⁶²⁹



Figure 15 Comparison of ERP to DRP

Source: AER analysis, Bloomberg (BVAL) and RBA data.

Note: data updated to 29/03/18.

⁶²⁷ For example, see: AER, Draft decision Murraylink transmission determination 2018 to 2023 Attachment 3–Rate of return, September 2017, p. 94. Available at: <u>https://www.aer.gov.au/networks-pipelines/determinations-accessarrangements/murraylink-determination-2018-23/draft-decision</u>

⁶²⁸ The spread between BBB+ rated corporate debt and the risk free rate. The DRP is calculated using the 2013 Guidelines' approach to estimate the return on debt (and DRP) as a simple average of BBB bond yields from RBA and Bloomberg BVAL data.

⁶²⁹ At December 2013, the ERP estimated in the 2013 Guidelines was 4.55 per cent, which was 1.14 per cent higher than the prevailing DRP at the time of 3.41 per cent. At 29 March 2018, the 2013 ERP of 4.55 per cent is now 2.66 per cent higher than the prevailing DRP of 189 per cent. Our ERP for this decision of 3.6 per cent is 1.71 per cent above this DRP.

The DRP is around 1.9 per cent as at the end of March 2018. When we finalised our 2013 Guidelines the debt risk premium was about 3.4 per cent. Comparatively, our 2013 Guidelines ERP estimate was 4.55 per cent, resulting in equity investors expecting to receive a premium of approximately 1.1 per cent above debt investors. Our ERP for this decision is 3.60 per cent which results in the expected return on equity having a premium of about 1.7 per cent above expected return on debt.

Thus, even though our ERP is lower in this decision compared to our 2013 Guidelines it represents a greater margin above the cost of debt.⁶³⁰

5.4.2.6 Evaluation of information

Our evaluation of the other information in section 5.4.2 leads us to consider that on the whole we do not need to revisit our estimate of an equity risk premium of 3.6 per cent.

We recognise the equity risk premium ranges from the Wright approach, valuers' and other regulators' decisions are above the ERP we have estimated. By contrast, our ERP for this decision represents an increase in comparison to the DRP. Once their strengths and weaknesses of the available cross checks are considered, we do not see a case for making further adjustment to the result calculated using the SLCAPM.

We also note stakeholders' disagreement on the usefulness of other information 'cross checks' and the concern that any adjustments based on other information could undermine transparency and rigour of process.

5.4.3 Step 6 – select point estimate

We are satisfied that an expected return on equity:

- derived from our foundation model approach and calculated via the SLCAPM,
- using a market risk premium of 6.0 per cent,
- using an equity beta of 0.6, and
- a risk free rate observed at the time the Guideline is applied,
- will contribute to the achievement of the legislative objectives. That is, using a wellestablished forward looking asset pricing model to compensate for systematic risk and populating it with parameter value estimates based on market data reflects a good estimate of expected market cost of capital. When capital is priced via a competitive market, the opportunity to beat the benchmark creates incentives to seek efficiencies. In a similar manner, providing a benchmark return on equity for

⁶³⁰ We are currently adjusting our approach for estimating the return on debt. In the 2013 Guidelines and subsequent determinations we used a simple average of BBB-rated Bloomberg BVAL and RBA yield curves. Following further considerations in this decision, the return on debt is now based on a weighted average of BBB and A-rated Bloomberg BVAL, RBA and Thomson Reuters yield curves. If the new debt methodology is applied consistent across this comparison, the margin between the 2018 ERP and DRP would be higher compared to the 2013 debt approach.

service providers, reflecting a market rate of return for the risk of providing regulated services furthers the revenue and pricing principles and is in the long term interests of energy users.

6 Risk free rate

In our section 5, we stated our decision to use the SLCAPM to calculate the allowed return on equity. The risk free rate is one of three parameters that contribute to the SLCAPM's formation of the return on equity. In the SLCAPM, the risk free rate measures the expected return from a riskless asset.

Our current approach to estimating a forward looking risk free rate uses the yield on Commonwealth Government Securities (CGS) with a 10 year term averaged over 20 consecutive business days as close as practicably possible to the commencement of the regulatory control period. ⁶³¹

6.1 Draft Decision

Our decision is to continue to estimate the risk free rate through the application of a formula for calculating the yield to maturity on 10 year CGS yields , based on data available over an averaging period nominated by service providers. Our decision is to refine the criteria that averaging periods should satisfy, based on review of submissions made to us and to accommodate the possibility of these guidelines becoming a binding rate of return instrument. Our decision is that service providers should have the ability to nominate an averaging period start and end date that meets the following requirements:

- Starts no earlier than 7 months prior to the commencement of the regulatory period in the case of electricity determinations, or the revision commencement date in the case of gas access arrangements, as specified in the current access arrangement.
- Ends no later than 3 months prior to the commencement of the regulatory period in the case of electricity determinations, or the revision commencement date in the case of gas access arrangements, as specified in the current access arrangement.
- Has between 20 and 60 consecutive business days in the period between the nominated start and end date, at the time the proposal is received by the AER.
- Is nominated prior to the start of the averaging period and contained in the initial proposal.

Where a service provider has failed to nominate an averaging period in accordance with the criteria above (or choses not to nominate one), we will use a period of 20 consecutive business days ending 3 months prior to the commencement of the regulatory control period or revision commencement date.

6.1.1 Approach to date

⁶³¹ AER, *Explanatory Statement* - *Rate of Return Guideline*, December 2013, p.15

The 2013 Rate of Return Guideline⁶³² with supporting explanatory statements⁶³³ and appendices⁶³⁴ detailed the methodology for calculating the risk free rate over the past five years, and recent regulatory decisions confirm our position where challenged.⁶³⁵ The methodology was to use an average of CGS yields taken over a 20 business day period, nominated by the service provider in advance of the period.

6.2 Input during the Guideline review process

We released an issues paper in October 2017 inviting input on the appropriate risk free rate methodology.⁶³⁶ We summarised and considered submissions on the issue paper in our discussion paper.⁶³⁷ A group of experts considered the averaging period in their statement of agreed positions prior to the concurrent evidence sessions.⁶³⁸ Stakeholders made further submissions on the Guideline before 4 May 2018.

We have summarised the key points made by stakeholders and experts in Appendix A.

6.3 Issues and the AER's considerations

We identified the averaging period length, automatic application of the risk free rate and the term of CGS yields used to estimate the risk free rate as the key issues to consider in this review. We outline our consideration of these issues in sections 6.3.1, 6.3.2 and 6.3.3.

6.3.1 Averaging period length

We outline the different factors that informed our draft decision on the averaging period length in this section.

Relevance to the on the day risk free rate

We apply the SLCAPM pragmatically, considering the theoretically ideal methodology and allowing departures where it provides a sufficient practical benefit. Ideally the SLCAPM uses values that represent market conditions as close as possible to the commencement of the regulatory control period or the revision commencement date, this would be the on the day CGS yields for the risk free rate.⁶³⁹ However, we have not

⁶³² AER, *Rate of Return Guideline*, December 2013

⁶³³ AER, *Explanatory Statement* - *Rate of Return Guideline*, December 2013

⁶³⁴ AER, *Explanatory Statement -* Appendices - *Rate of Return Guideline*, December 2013

⁶³⁵ See for example AER, APA VTS 2018 Access Arrangement Final Decision - Attachment 3, 30 November 2017; AER, AusNet Services 2017-22 Final Decision - Attachment 3, 11 May 2017;

⁶³⁶ AER, Rate of return issues paper, 31 October 2017, p. 18-19

⁶³⁷ AER, MRP Risk Free Rate Averaging Period and Automatic Application Discussion paper, March 2018, p. 38-39

⁶³⁸ AER, Concurrent Evidence Session 2 – Facilitator's Note, 4 April 2018, p. 42

⁶³⁹ This ideal was established and maintained in Federal Court cases, (for example Federal Court of Australia, ActewAGL Distribution v The Australian Energy Regulator [2011] FCA 639, 8 June, 2011, paragraph 119), the 2013 Guidelines, and a report by Dr Martin Lally (Lally, Risk free rate and the present value principle, August 2012, p.7).

used an on the day rate CGS yield as a proxy for the risk free rate in our 2013 or 2009 guidelines.⁶⁴⁰ We decided not to use the on the day rate, because there was a benefit in using a longer averaging period.

We do carefully consider whether departures from the SLCAPM ideal are sufficiently justified by a benefit. We conveyed the benefit for a 20 day averaging period in our 2013 Guidelines, as reduced exposure to volatility⁶⁴¹ and intentional distortions⁶⁴² of the CGS yields. However, we did not accept an averaging period of 8 months in a later regulatory determination, as it was more than necessary to overcome the shortcomings of a 20 day averaging period.⁶⁴³

Exposure to risk free rate fluctuations

Stakeholder input in the review of the Guidelines identified a potential benefit in allowing a service provider greater flexibility in how it mitigates its exposure to fluctuations in the on the day rate.⁶⁴⁴ This included agreement between networks⁶⁴⁵ and consumers⁶⁴⁶ that an averaging period of between 20 and 60 business days would provide this flexibility. Other stakeholders⁶⁴⁷ and the concurrent evidence session experts also supported the change. ⁶⁴⁸ We consider that based on stakeholder input to the guideline review process, there is support from diverse stakeholders to use a longer averaging period.

Comparison of different averaging period lengths

Figure 16 shows the impact of different averaging period lengths on volatility in the on the day rate.

⁶⁴⁰ AER, Review of the WACC parameters, May 2009, p.132

⁶⁴¹ AER, Explanatory statement - rate of return guideline, December 2013, pg. 77

⁶⁴² Lally, *Risk free rate and present value*, August 2012, p. 7

⁶⁴³ AER draft decision – AusNet Services gas access arrangement 2018-2022 – Attachment 3, July 2017, pg 69-71

⁶⁴⁴ See Table 23.

⁶⁴⁵ ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.4

⁶⁴⁶ CRG, ROR submission final, 04 May 2018, p.39

⁶⁴⁷ See Table 23.

⁶⁴⁸ AER, Evidence session 1& 2 Expert Joint Report, 21 April 2018, page 64



Figure 16 Impact of different lengths of averaging CGS yields

Source: RBA interest rate statistics f16, AER analysis

The 20 day averaging period reduces the impact of individual days in the on the day rate. It does not however remove short-term fluctuations in the on the day rate. The 60 day averaging period in comparison reduces the impact of short-term fluctuations but still follows the underlying trends of the on the day rate. In contrast, the 250 day average departs significantly from the on the day rate.

6.3.2 Automatic application of the risk free rate

This guideline applies automatically, removing the need for us to exercise discretion by codifying the process for nominating the risk free rate.

Averaging period nomination window

We will need to specify the window within which the service provider can nominate the averaging period. We cannot have an averaging period end any later than three months prior to the regulatory control period commencement and revision commencement dates, to give us sufficient time to come to a final decision.⁶⁴⁹ We are

⁶⁴⁹ Our final decisions for determinations and access arrangement are generally produced at least 2 months prior to the commencement of their regulatory control period or revision commencement date. Please see AER, 7 year regulatory determination calendar 2015-2022, February 2018 for more guidance on reset timeframes

prepared to allow an averaging period that starts up to seven months prior to the commencement of the regulatory control period, to allow service providers a range of dates they can nominate for a 60 day averaging period to provide confidentiality of the nominated averaging period.⁶⁵⁰ As such, we require that a nominated averaging period must start and end between 7 months and 3 months prior to the commencement of the regulatory control period of revision commencement date.

If the final decision is delayed

We have considered how delays in the final decision or a remittal interact with our risk free rate methodology. We do not see it necessary to require a business to nominate a revised risk free rate averaging period. We make delayed determinations and access arrangements as if they were in effect from the original commencement of the regulatory control period⁶⁵¹ or revision commencement date. ⁶⁵² Therefore, the nominated averaging period would remain appropriate and we would not require a revised nomination.

Nominated averaging periods that don't meet the criteria

We have included a mechanism for addressing circumstances where service providers fail to meet the nominated averaging period criteria (this includes failing to nominate a period). We will use a default averaging period that is 20 days in length ending three months prior to the commencement of the regulatory control period or revision commencement date. We will not reveal whether the service provider has failed to meet the averaging period criteria until after the default averaging period has ended. If the service provider fails to meet the nominated averaging period criteria then we will calculate the risk free rate using the default averaging period.

Situations where the number of business days change

We have considered how changes to public holidays may cause nominated averaging periods to fail to meet the criteria. We consider it appropriate that the nominated averaging period need merely meet the criteria at the time of the proposal. This will avoid forcing service providers to use the default averaging period due to unforeseeable changes in the number of business days. To clarify, this does not include public holidays that are public knowledge at the time of the proposal.

6.3.3 Term of CGS yields

The term of the CGS yields informs which series of yields should be used in approximating the risk free rate.

⁶⁵⁰ We have tried to protect confidentiality of averaging period where possible, see for example, AER preliminary decision Powercor distribution determination - Attachment 3 - Rate of return, October 2015

⁶⁵¹ See for example, AER final decision Powercor distribution determination - Attachment 3 - Rate of return, October 2015

⁶⁵² See for example, AER, Draft Decision Roma to Brisbane Gas Pipeline Access Arrangement - Overview, p.60

Appropriate term

We require that the return on equity parameters have a consistent investment horizon to ensure they are estimating a reasonable rate of return. We used a 10 year term for CGS yields in our 2013 Guidelines⁶⁵³, as it was consistent with the term used for the return on equity parameters⁶⁵⁴, including MRP. ⁶⁵⁵ Consumers submitted that a shorter CGS yield term of five years was appropriate for the risk free rate. They made this recommendation based on the following points:⁶⁵⁶

- The return on equity is set for a 5 year period (i.e. over the course of the regulatory period) and the risk free rate should reflect this,
- The equity beta and the market risk premium in the CAPM comes from an index of share price volatility measured over shorter periods but averaged over a longer period.
- Investors reassess their portfolios on a much shorter basis than 10 years.

We consider the consumer suggestion is advocating for using a different term for the risk free rate of five years instead of ten. This change would require the term for estimating all parameters of the return on equity to be revised to five years. The consumer submission did not provide sufficient justification or support for changing the SLCAPM parameters term from 10 to 5 years.

⁶⁵³ AER, Explanatory statement, rate of return guideline, December 2013, pg. 79

⁶⁵⁴ AER, Explanatory statement, rate of return guideline, December 2013, p.48

⁶⁵⁵ AER, Explanatory statement, rate of return guideline, December 2013, p. 81

⁶⁵⁶ See Table 23.

6.4 Summary of submissions

This section details the submissions we have received from stakeholders as part of our review process to date, and notes how we have had regard to each submission.

Key Point	Submission	Stakeholder	AER response
Methodology	The methodology should be prescribed by the AER in the guideline.	EUAA ⁶⁵⁷ MEU ⁶⁵⁸ Network Shareholder Group ⁶⁵⁹	We agree that the methodology should be prescribed by us in the guideline rather than specifying the rate.
Averaging period length	The risk free rate averaging period should be extended up to 60 consecutive business days - to increase stability of the risk free rate.	CKI ⁶⁶⁰ Jemena ⁶⁶¹ Victorian electricity distributors ⁶⁶² APGA ⁶⁶³ CRG ⁶⁶⁴ ENA ⁶⁶⁵⁶⁶⁶ Concurrent evidence session agreed position ⁶⁶⁷ ATCO ⁶⁶⁸	We see there is overwhelming support for a move from 20 to 60 days in the length of the averaging period. Given that there is little opportunity for gaming, we are comfortable with moving from 20 to 60 consecutive business days.
	There should be no changes in the setting of averaging periods as this creates unnecessary inflexibility	APA ⁶⁶⁹ Network Shareholder Group ⁶⁷⁰ Network Shareholder Group ⁶⁷¹	We see that allowing an averaging period between 20 and 60 consecutive business days will allow service providers flexibility in how they mitigate their exposure to volatility in the risk free rate
	The current averaging period approach is	CCP ⁶⁷² Ergon Energy and Energex ⁶⁷³	We agree that an averaging period of 20 days is appropriate.

Table 23 Summary of submissions on the risk free rate

- ⁶⁵⁷ EUAA, Submission on the rate of return issues paper, 12 December 2017, pg. 8
- ⁶⁵⁸ MEU, Submission on the rate of return issues paper, 12 December 2017, pg.15
- ⁶⁵⁹ Network Shareholder Group, Submission on Rate of Return Guideline Review, 04 May 2018, pp. 13
- ⁶⁶⁰ CKI, Submission on the rate of return issues paper, 12 December 2017, pg.4
- ⁶⁶¹ Jemena, Submission on the rate of return issues paper, 12 December 2017, pg. 3
- ⁶⁶² Victortian Distributors, Submission on the rate of return issues paper, 12 December 2017, pg. 1
- ⁶⁶³ APGA, Submission on Rate of Return Guideline Review, 04 May 2018, pp.3
- ⁶⁶⁴ CRG, Submission on Rate of Return Guideline Review, 04 May 2018, p.39
- ⁶⁶⁵ ENA, Submission on the rate of return issues paper, 12 December 2017, pg. 18
- ⁶⁶⁶ ENA, Submission on Rate of Return Guideline Review, 04 May 2018, pp.4
- ⁶⁶⁷ AER, *Evidence session 1& 2 Expert Joint Report*, 21 April 2018, page 64
- ATCO Australia, Submission on the rate of return issues paper, 12 December 2017, pg.5
- ⁶⁶⁹ APA, Submission on the rate of return issues paper, 12 December 2017, pg. 6
- ⁶⁷⁰ Network Shareholder Group, Submission on the rate of return issues paper, 12 December 2017, pg. 8
- ⁶⁷¹ Network Shareholder Group, Submission on Rate of Return Guideline Review, 04 May 2018, pp. 14
- ⁶⁷² CCP, Submission on the rate of return issues paper, 12 December 2017, pg. 38
- ⁶⁷³ Ergon Energy, Submission on the rate of return issues paper, 12 December 2017, pg. 4

	satisfactory		
	There is scope to review the length of the averaging period	APGA ⁶⁷⁴	We agree that the averaging period length should be reviewed
Service providers allowed to nominate the averaging period	The service provider should have discretion as to which averaging period they would like to use	ATCO ⁶⁷⁵	We agree with this position as it gives service providers an opportunity to mitigate fluctuations in the risk free rate.
	Service providers should not be allowed to choose the averaging period	MEU ⁶⁷⁶ Jemena ⁶⁷⁷	We see little benefit in preventing service providers from nominating a period. Service providers have to nominate the period in advance, which reduces the possibility of bias.
Appropriate CGS term	The appropriate CGS term is 5 years as opposed to 10	CRG ⁶⁷⁸	We disagree, the WACC we are using is to estimate an appropriate rate of return over a 10 year period.

 $^{^{\}rm 674}$ $\,$ APGA, Submission on the rate of return issues paper, 12 December 2017, pg. 6 $\,$

⁶⁷⁵ ATCO Australia, Submission on the rate of return issues paper, 12 December 2017, pg.5

⁶⁷⁶ MEU, Submission on the rate of return issues paper, 12 December 2017, pg.15

⁶⁷⁷ Jemena, Submission on the rate of return issues paper, 12 December 2017, pg. 3

⁶⁷⁸ CRG, Submission on Rate of Return Guideline Review, 04 May 2018, p.44

7 Market risk premium

The market risk premium (MRP) is the difference between the expected return on a market portfolio and the return on the risk free asset. The MRP compensates an investor for the systematic risk of investing in the market portfolio or the 'average firm' in the market. Systematic risk is that which affects all firms in the market (such as macroeconomic conditions and interest rate risk) and cannot be eliminated or diversified away through investing in a wide pool of firms

As we use an Australian domestic SLCAPM, the relevant MRP is the expected Australian dollar return on the Australian market portfolio less the return on Australian dollar risk free asset.

Our regulatory task is to determine an overall rate of return (or WACC) for a Benchmark efficient entity that is commensurate with its efficient financing costs. To achieve this, our estimate of MRP used in the SLCAPM should be a good estimate of the expected Australian domestic MRP.

The expected MRP is not directly observable, although realised excess equity returns can be observed after the fact. These can then be used as information to assist with estimating the MRP.

In addition to past observed market returns you can use other information to inform your estimate of the MRP. This includes estimates from dividend growth models and from observed risk premiums on other assets such as debt.

7.1 Draft decision

Our draft decision is to set an MRP of 6.0 per cent per annum over the yield to maturity on Australian Commonwealth Government Bonds with a term to maturity of 10 years (10 year CGS). In estimating the MRP we have considered all relevant evidence available to us from the review, including evidence from historical excess return data and potential methods of forward estimation of the MRP.

Our estimate of 6.0 per cent per annum is a decrease from the MRP of 6.5 per cent per annum (also over 10 year CGS) estimated during the 2013 Guidelines process and subsequent regulatory determinations.

- We consider the evidence before us continues to support the overall approach used for estimating the MRP in the 2013 Guidelines process. That is, in estimating the MRP we have given the most weight to historical excess returns and less weight to other relevant evidence. The key reasons for continuing to give most weight to historical excess returns are:
 - o These are directly observable, easily replicable and transparent
 - We expect required risk premiums to change relatively slowly through time.

When exercising our regulatory judgement, we rank the utility of different types of evidence at that time and then we qualitatively consider whether our initial estimate of

the MRP should be moved up or down.⁶⁷⁹ In the 2013 review process, we stated that 6.0 per cent is an appropriate estimate of the historical excess returns (HER) evidence and the starting point for our determination of a point estimate.⁶⁸⁰ We, then moved our estimate up based on the direction of the other evidence we consider in estimating the MRP, particularly having regard to dividend growth models (DGMs) evidence.

In this review, we continue to give greater wieght to HER for informing the market risk premium. This material, along with other relevant evidence, support a value of 6 per cent.

The key reason for a decrease in the MRP from 6.5 to 6 percent per annum is, in this 2018 review process, evidence from DGMs has not persuaded us to move the point estimate derived from HER. While we have considered a range of results that DGMs, as submitted through the consultation process, we have diminished confidence in the estimates from DGMs. We have received considerable expert advice since the 2013 Guidelines raising significant concerns with MRP estimates from DGMs.⁶⁸¹ The DGM evidence does not give us sufficient confidence to move the HER estimate.

We consider a decrease in MRP to 6 per cent per annum is also consistent with decreased volatility in equity markets since 2013 and material reductions in debt risk premiums over the past 5 years.

We note that moving the MRP to 6 per cent per annum is consistent with the MRP used by the AER immediately prior to the 2013 Guidelines review.

In this review we have updated the data and also considered any new evidence before us. We have then in determining an MRP of 6 per cent exercised our regulatory judgment based on the updated information. Continuing our existing approach to determining the MRP was broadly supported in submissions and in the expert sessions. However, there were different views on the merits of the different pieces of evidence and the values that should be derived from each.

⁶⁷⁹ ENA, *Regulatory discretion and market risk premium determination*, 26 June 2018.We acknowledge receipt of a late submission from the ENA on 26 June 2018 which has not been subject to a full assessment. The submission included 3 options for exercising regulatory discretion and our initial assessment is as follows. The first option involves assigning specific numerical weights to each relevant piece of evidence. Our experience suggests that this level of precision is not possible. The second option is to set out a ranking of weights to each piece of evidence in terms of relative weight in achieving the final MRP. This option appears to be largely similar to our current approach to exercising regulatory discretion. The third option which recommends setting a 'neutral' long run estimate (for MRP or Total Market Return) as the default and then qualitatively making adjustments, based on a pre-set movement up/down relative to strength of the evidence. Our initial view is that this option would unnecessarily restrict future exercise of regulatory judgment because of 'default' settings becoming a hurdle/onus of persuasion.

⁶⁸⁰ AER, *Explanatory statement - Rate of Return Guideline*, December 2013, p. 97.

⁶⁸¹ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 21 May 2018, p.33; Partington and Satchell, Report to the AER: Cost of Equity Issues 2016 Electricity and Gas Determinations, April 2016, pp.27-33; McKenzie & Partington, Report to the AER: Part A, Return on Equity, October 2014, pp.26-41.

The service providers and the Network Shareholder Group (NSG) stated that more weight should be given to estimates from DGMs and from the Wright Approach. They also submitted less or no weight should be given to geometric averages of historical excess returns. For example, the ENA submitted:

The Dividend Growth Model (DGM) provides useful evidence on the current MRP and should be given explicit and material weight by the AER.⁶⁸²

ENA reiterates the view ... that the geometric averages is inappropriate for the purpose of estimating the expected excess return ... the geometric average should not be used and only the arithmetic average should be used for the purpose of setting an allowed rate of return.⁶⁸³

Consumer groups submitted that in estimating the MRP more weight should be given to geometric averages of historical excess returns and no weight should be given to estimates from DGMs and the Wright Approach. For example, CCP 16 stated:

Whilst recognising that the DGM has a 'solid theoretical basis' and has value in certain circumstances CCP16 remains concerned about the reliance of the DGM in the context of an ex-ante regulatory decision.⁶⁸⁴

CCP 16 also stated in relation to the weight to put on geometric average excess returns:

The AER has not placed adequate reliance on the outputs of the geometric average nor critically assessed the problems in the arithmetic average given the volatility of the annual Australian equity market returns of 17.7% that Dimson, Marsh and Staunton (2015) report. ⁶⁸⁵

We have considered the different views put to us by stakeholders and experts in arriving at our draft decision. These considerations are set out in section 7.3.

The observed arithmetic MRP since 1988 of 6 per cent, combined with current (relatively low volatility) market conditions and some evidence of a decreasing MRP through time, supports our view that an MRP of 6 per cent per annum will give investors an opportunity to recover their efficient costs and contribute to achieving the NEO/NGO, revenue and pricing principles and the allowed rate of return objective.

7.2 Stakeholder consultation and submission

We released an issues paper in October 2017 outlining key areas of the Guideline review process. In response, stakeholders made submissions summarised in our MRP

⁶⁸² ENA, *Response to Discussion papers and concurrent expert evidence sessions*, May 2018, p.10.

⁶⁸³ ENA, *Response to Discussion papers and concurrent expert evidence sessions*, May 2018, pp.161-162.

⁶⁸⁴ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.89.

⁶⁸⁵ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.88.

discussion paper.⁶⁸⁶ They are discussed as part of our considerations of issues in section 7.3.

7.2.1 Discussion paper and concurrent expert evidence sessions

We released an MRP discussion paper in March 2018 to provide background on likely MRP estimation issues we expected to be discussed at a concurrent evidence session on 5 April 2018. ⁶⁸⁷ The MRP discussion paper also contained questions to frame the discussions.

A statement of agreed positions was published following the concurrent evidence session.⁶⁸⁸ We have reviewed this and the transcript and found the following positions to be accepted by the majority of experts:

Agreed Position	Reasons for agreement		
3 methods of market return estimation will be considered	Historic Excess Returns (HER), Dividend Growth Models (DGM) and market surveys are the most commonly used methods of estimation and there are currently no others which have been rigorously tested enough to be considered suitable for the regulatory task. It was also noted that surveys can also include valuation and broker reports as well as decisions of other regulators.		
The DGM is a useful source of evidence	The DGM is considered a useful source of evidence by almost all market practitioners, however there are stark differences in opinion when it comes to how it is best used.		

Table 24 Agreed positions on MRP from concurrent evidence sessions

The statement of agreed position contained a reference indicating a particular 'NERA' adjustment should be used when estimating historical excess returns. This is not included in the table above because we consider this was subject to some disagreement. Not all experts were (fully) available over the course of preparing the expert joint report to present their views. The author of the joint statement also confirmed agreed positions may have been taken if no one objected rather than by requiring positive agreement and that assessing the views was not a quantitative voting exercise.⁶⁸⁹

⁶⁸⁶ AER, Discussion paper, *MRP, risk free rate averaging period and automatic application of the rate of return*, March 2018.

⁶⁸⁷ AER, Discussion paper, *MRP, risk free rate averaging period and automatic application of the rate of return*, March 2018.

AER, Evidence session 1& 2 Expert Joint Report, 21 April 2018.

⁶⁸⁹ See CEPA, AER RORG Expert Joint Report, 21 April 2018, p. 11: "It should be noted that not all experts were present in all the sessions and may therefore not have given views on all issues. The issues on which experts contributed were set out above in Section 1.1. Graham Partington (GP) was unavailable due to overseas commitments from 14 April 2018 and provided limited input from that date, but did have sight of the final draft. David Johnstone provided input on drafts until 10 April 2018.

We have also received submissions on the expert concurrent evidence sessions. The submissions and our considerations of these are contained in the next section. A full summary of submissions is provided in the section 7.4.

7.3 Issues and the AER's considerations

We have received divergent submissions on a range of issues from stakeholders as part of this review. In forming a view on our approach, we have taken into account these submissions and further consideration of the relevant evidence. Our consideration of our approach, issues and submissions are in the sections below under the following sections:

- Overall approach
- Historical excess returns
- Dividend growth model
- Survey evidence
- Conditioning variables
- Other regulators
- Wright approach

7.3.1 Overall approach to estimating the MRP

Our approach for estimating the market risk premium was published in our 2013 Guidelines as part of the foundational model approach for estimating the return on equity. We consider a wide range of relevant evidence and estimation methods for informing estimates of the MRP based on their strengths, weaknesses and suitability for our regulatory task which we distilled to the following:

- Historical returns we gave Historical Excess Returns (HER) estimation the first priority in our MRP estimation, as the method is transparent, replicable and widely used across Australia.⁶⁹⁰
- Dividend Growth Models (DGM) We gave our own construction of the DGM some directional weight in MRP estimation process in 2013, but evidence from our own construction of the DGM did not persuade us that the point estimate derived from HER should be moved in this 2018 review. Overall, we have not been persuaded to rely upon other DGM evidence that has been submitted to us during the review process to move the point estimate derived from HER. The DGM method has forward looking properties, but it is highly sensitive to assumptions.⁶⁹¹

[&]quot;The report indicates when most experts held a particular view. However, assessing the views was not a quantitative voting exercise, but a way of identifying alternative views and the reasons for them. Dissenting views of any expert were considered to be of value and may inform the views of the AER."

⁶⁹⁰ AER, Explanatory statement - Rate of Return Guideline, December 2013, p. 90.

⁶⁹¹ AER, *Explanatory statement - Rate of Return Guideline* , December 2013, p. 90.

- Survey Evidence Survey results use the direct theoretical link between expected excess returns and stated expectations, but vary in usefulness dependent on the design, timeliness and representativeness of the respondents.⁶⁹²
- Conditioning Variables These were selected variables such as dividend yields, credit spreads and implied volatility which we determined had properties which revealed information about current market conditions. However, there is an indirect link between these indicators and the MRP. ⁶⁹³
- Other Australian Regulators Decisions We reviewed other regulator's decisions to inform ourselves of practices employed elsewhere in Australia. However, there is potential for this approach to be circular. ⁶⁹⁴

7.3.1.1 Inverse relationship between MRP and risk free rate

Having considered the evidence before us, we remain of the view that, while the MRP may vary over time, there is no estimable inverse relationship between the MRP and risk free rate.

The view that MRP varies over time is generally accepted by all stakeholders. However there is uncertainty whether there is an inverse relationship between the risk free rate and MRP. This was evident by the comments in the second evidence session by multiple experts including Graham Partington, Ilan Sadeh, Simon Wheatley and Jim Hancock and Stephen Gray.⁶⁹⁵ For example, Stephen Gray supported the view that MRP and the risk free rate had a negative relationship, noting that as government bond yields decreased, the required return on equity will decrease, but not one-for-one.⁶⁹⁶ Ilan Sadeh was less supportive of this relationship, highlighting that the MRP has been fairly constant for the last two decades, and that the market views it as an accepted outcome.⁶⁹⁷

We have previously stated and continue to hold the view that the evidence for an inverse relationship between the MRP and the risk free rate is weak.⁶⁹⁸ We consider there is neither strong theoretical reasons, nor strong empirical evidence, to support an ongoing and consistent inverse relationship. There are also a number of academic reports which suggest there may actually be a positive relationship between the risk free rate and the MRP at times.⁶⁹⁹

⁶⁹² AER, *Explanatory statement - Rate of Return Guideline*, December 2013, p. 90.

⁶⁹³ AER, *Explanatory statement - Rate of Return Guideline*, December 2013, p. 91.

⁶⁹⁴ AER, *Explanatory statement - Rate of Return Guideline*, December 2013, p. 91.

⁶⁹⁵ AER, Concurrent Evidence Session 2 – Proofed Transcript, April 2018, p. 69-71.

⁶⁹⁶ AER, Concurrent Evidence Session 2 – Proofed Transcript, April 2018, p. 69.

⁶⁹⁷ AER, Concurrent Evidence Session 2 – Proofed Transcript, April 2018, p. 70, 75.

⁶⁹⁸ For example, see: AER. Multinet draft decision, July 2017, p. 104; AER, Final decision SA Power Networks determination 2015–16 to 2019–20 Attachment 3–Rate of return, October 2015, p. 410–423, AER, *Final Decision AusNet distribution determination - Attachment 3 - rate of return*, May 2016, p. 82; AER, *APA VTS Final Decision: Attachment 3 - Rate of Return*, November 2017, p.299.

⁶⁹⁹ Li, *Time-varying risk aversion and asset prices*, Journal of Banking and Finance, 2007; Kim & Lee, *Stock returns, asymmetric volatility, risk aversion and business cycle: Some new evidence*, July 2007,; Damodoran, *Equity Risk*

During the concurrent evidence sessions there was discussion about the 'ends of the argument', one being a stable MRP and the other being a stable return on equity. This was presented as the bounds between which the true MRP exists.⁷⁰⁰ Stakeholders submitted various views on whether a fixed MRP contributes towards our regulatory objectives. The CCP 16 supported fixing the MRP, highlighting the view that MRP should be the slowest WACC parameter to change.⁷⁰¹ Energy Networks Australia submitted that fixing MRP has produced volatility in the allowed return on equity that varies one for one with changes in the risk free rate.⁷⁰² Similarly, Evoenergy viewed that fixing MRP may result in equity estimates inconsistent with market evidence.⁷⁰³

The expert joint report appeared to indicate agreement that the MRP is neither constant nor directly inversely related to the risk free rate.⁷⁰⁴ However, most experts at our concurrent evidence session considered there was no proper model to quantify the relationship.⁷⁰⁵ We agree that there is no proper model and therefore there is no robustly estimable relationship between the MRP and risk free rate. The proposition for an upper bound is essentially a view that that the Wright approach should be given consideration in estimating the MRP. We discuss the Wright approach in section 0. More importantly, we consider that an approach which promotes a stable return on equity may not be suitable for a regulatory model which resets every 5 years.⁷⁰⁶

Reports have previously been submitted to the AER using debt yields, and their relationship with the yield on the risk free asset. These have been used to support the position that there is more stability in the overall yield on debt than either the yield on risk free asset or than the debt risk premium. This has then been used to support the argument the return on equity is likely to be more stable than the SLCAPM implies (if applied using a fixed MRP over a variable risk free return).⁷⁰⁷ However extending the

Premiums (ERP): Determinants, Estimation and Implications – the 2012 Edition; Rankin and Idil, A century of Stock-Bond Correlations, September 2014; Antti Ilmanen, Expected Returns on Stocks and Bonds, Journal of Portfolio Management, Winter 2003.

- AER, Concurrent Evidence Session 2, 5 April 2018, pp.62-64.
- ⁷⁰¹ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, p.
 95.
- ⁷⁰² Energy Networks Australia, Response to Issues Paper, December 2017, p.22, 24.
- ⁷⁰³ EvoEnergy, Submission on Rate of Return Guideline Evidence Session, 04 May 2018, p.4.
- ⁷⁰⁴ Joint Expert Report, RORG review Facilitation of concurrent evidence sessions, CEPA, 21 April 2018, section 6.13, p.64.
- ⁷⁰⁵ Joint Expert Report, RORG review Facilitation of concurrent evidence sessions, CEPA, 21 April 2018, section 6.13, p.64.
- ⁷⁰⁶ Mr Sadeh also stated: "And a lot of investors recognise that as a feature of the current framework which is, you know, a fixed MRP over a bond rate that moves, and that is seen particularly for long-term investors, superannuation funds, they want their members to have exposure to Australian macro-economic variables. They see this as a resetting bond in that circumstance. So they understand that in absolute sense, even though they are investing their equity for a long time, 99 years, they accept that during different five-year periods they are going to get an absolute return that is a function of the bond rate, and that's priced into the way the investment works". See, AER, Concurrent Evidence Session 2 Proofed Transcript, April 2018, p. 70, 75.
- ⁷⁰⁷ SFG, Response to the QCA Discussion Paper on risk free rate and market risk premium, 19 March 2013; HustonKemp, The Cost of Equity, Response to the AER's Draft Decisions for the Victorian Electricity Distributors, January 2016, p.44.

results in a similar study we have shown in Figure 17 that the yield to maturity on BBB debt has fallen since the 2013 Guidelines, along with both the DRP and the yield to maturity on the risk free asset.



Figure 17 Cost of BBB debt and its composition since 2013

Source: RBA extrapolated 10 year BBB debt yield average with BVAL 10 year BBB, RBA 10 year CGS yield.

McKenzie and Partington have also previously advised that any relationship between the MRP and risk free rate is an open question, and any relationship that may exist is not sufficiently well established to form the basis for regulatory adjustment to the MRP.⁷⁰⁸ We have also received advice that a negative correlation between the risk free rate and the MRP has no well accepted theoretical support and is not used much in practice.⁷⁰⁹

Nevertheless, we accept there is evidence that some brokers and valuation experts effectively create a more stable overall return on equity by uplifting the risk free rate. HoustonKemp, in a 2016 paper, highlights this to suggest there is clear evidence that valuation firms accept an inverse relationship between the MRP and the risk free rate.⁷¹⁰ This is shown in Figure 18.⁷¹¹

⁷⁰⁸ McKenzie and Partington, *Review of the AER's Overall Approach to the risk free rate and MRP*, February 2013, p28.

⁷⁰⁹ Partington and Satchell, Cost of Equity issues 2016 Electricity and Gas Determinations, April 2016, pp30-31; Partington and Satchell, Report to the AER, May 2018, pp34-35.

⁷¹⁰ HustonKemp, *The Cost of Equity, Response to the AER's Draft Decisions for the Victorian Electricity Distributors*, January 2016.

⁷¹¹ HustonKemp, *The Cost of Equity, Response to the AER's Draft Decisions for the Victorian Electricity Distributors*, January 2016, p.44.



Figure 18 Plot of expert valuation reports' chosen RFR vs 10 year CGS yield

Source: HoustonKemp, The Cost of Equity, Response to the AER's Draft Decisions for the Victorian Electricity Distributors, January 2016, p.44.

Note: Data from Thompson Reuters, the ASX and the RBA. The 10-year CGS yields are interpolated from the RBA files f16.xls, f16hist.xls and f16hist2013.xls.

Having considered all material before us, we remain of the view that uplifts applied by brokers and valuers to initial estimates may be inconsistent with the allowed rate of return objective. We have also been advised that uplifting returns when applying the SLCAPM may not be suitable in a regulatory context as it is too ad-hoc.⁷¹²

Uplifts may reflect non-systematic risks, or be designed to account for risks not addressed in cash flow forecasts, or (to the extent there is any) the expectation of outperformance of regulatory allowances. They may also reflect the relevant investment period exceeding the term of our regulatory control period. It is relevant that we are estimating a required return on capital to be applied in regulatory determinations applying for five years and which will be subsequently reset (including we expect for adjustment of the risk free rate) in the next regulatory control period.

⁷¹² Partington and Satchell, Report to the AER: Discussion of estimates of the return on equity, 12 April 2017, p16.

In terms of bounds on the MRP, we note that there is no reason (or evidence) to support movements in MRP being bounded to a relationship with the return on the risk free asset given the lack of an estimable relationship.

We accept that the true MRP may vary through time. For example the MRP may change in times of both changed risk and changed investor risk aversion. Quite distinct from risk aversion, the price of market risk may also change through time.

We note the MRP is the extra expected return investors require for investing in the full diversified 'market' portfolio. It can be thought of as the required (ex ante) compensation for bearing a unit of market risk. Because of the market wide nature of the MRP and its non-observable nature, it may change more slowly than other SLCAPM input parameters. This was acknowledged in the concurrent evidence session.⁷¹³

There is some suggestion that over time, as investing in a global portfolio becomes easier and investors achieve greater diversification, the market risk premium falls. This was raised by the CRG in its submission on the discussion paper and the evidence sessions⁷¹⁴ and is also referred to in a report sponsored by Challenger Limited authored by Bianchi, Drew and Walk.⁷¹⁵ This is alongside evidence from Dimson, Marsh and Staunton (2015) that there is downward trend over time in the realised MRP. Figure 19 shows a graph of Dimson et al.'s trend of the achieved equity risk premium over time.



Figure 19 Trend in Equity Risk Premium in Australia

Source: Dimson, Marsh and Staunton (2015); Calculations from Bianchi, Drew & Walk, *The Unpredictable Equity Risk Premium*, November 2015.

⁷¹³ AER, Second Concurrent Evidence Session, 5 April 2018, p.65.

⁷¹⁴ CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.53.

⁷¹⁵ Bianchi, Drew & Walk, *The Unpredictable Equity Risk Premium*, November 2015.

Whilst we must consider evidence as presented, acknowledging trends with solid theoretical foundations assists us in estimating a forward looking MRP.

7.3.2 Historical Excess Returns

We give most weight to historical excess returns in estimating the MRP. Consistent with the approach used in the 2013 Guidelines and updated evidence, we have set a range of 5.0 - 6.5 per cent from the historical excess returns data, with a point estimate of 6.0 per cent. In deriving our observed estimate of historical excess returns, we:

- use the unadjusted Brailsford, Handley, Maheswaran (BHM) data. This is because the data is from a reputable source (the ASX) and BHM investigated the underlying data and concluded that it was reasonable.
- consider both arithmetic and geometric averages over multiple time periods. We
 consider there are strengths and weaknesses to each method and Partington and
 Satchell have advised to continue to use both.

Annual historical excess returns are estimated by measuring realised annual excess market returns (above the annualised return on the risk free asset). These annual excess returns on the market are then averaged over varying time periods to give different estimates of a forward looking MRP. In doing this we acknowledge the expected MRP may differ to the realised MRP and may change through time.

Historical excess returns have been our main source of information used for estimating the MRP since the 2013 Guidelines. The method is easily replicable, transparent and widely used in both regulation and by market practitioners. We intend to continue to use historical estimates of realised excess returns on the market as our primary basis for MRP estimation.

In relying on the historical excess returns we have given relatively less weight to estimates using data before 1958. This is due to concerns with data reliability given the age of the data. There is also a debate around how to adjust the pre 1957 data for biases it contains and whether to use the NERA adjustment of BHM adjustment.

There were a number of points raised in submissions both to our 2017 Issues Paper as well as the March 2018 discussion paper and the concurrent evidence session.⁷¹⁶ These are covered below.

7.3.2.1 NERA adjustment vs BHM use of the ASX adjustment

There are potential problems with the realised Australian market (All Ordinaries) return data series before 1962. The issue relates to the fact that the underlying yield series on the Australian market over the periods from 1882 to 1955 and 1956 to 1961 were

AER, Transcript of Second Concurrent Evidence Session, 5 April 2018, p52.

constructed retrospectively by Lamberton and the Sydney Stock Exchange.⁷¹⁷ These series reflect a simple unweighted average yield on dividend paying stocks only. The unweighted yields are biased towards smaller high yielding stocks and excluding non-dividend paying stocks also overstates the realised yields.

The Australian Stock Exchange (ASX) subsequently made an adjustment to the Lamberton/Sydney Stock Exchange return series to correct for these biases. In doing so the ASX reduced the yields by 25 per cent in the early years of the accumulation index where it did not have any other dividend yields to guide it.⁷¹⁸

Brailsford, Handley and Maheswaran confirmed the ASX applied an adjustment factor of 0.75 for the period 1882 to 1964 and that the adjustment appeared reasonable. Based on their assessment, they determined to use this ASX constructed yield series for the pre 1974 period for estimating the excess return on the market over this period.⁷¹⁹ A full explanation of this is contained in the academic paper by Brailsford, Handley and Maheswaran.⁷²⁰

Some submissions and the concurrent evidence session stated that we should be using a return series adjusted by NERA Economic Consulting (NERA) instead of the earlier return series in the paper by Brailsford, Handley and Maheswaran that reflects the ASX adjustment.⁷²¹ The expert report from the concurrent evidence session also stated that experts suggested we should use the NERA adjustment moving forward. However they did not give reasons to support to this position. Having reviewed the report, we do not consider that the use of the NERA adjustment was agreed by all experts.⁷²²⁷²³

⁷¹⁷ From July 1961, the series was calculated by the SSE on a monthly basis. (see Brailsford, Handley, Maheswaran, 'Re-examination of the historical equity risk premium in Australia', Accounting and Finance, Vol. 48, 2008, p. 79). See AER, Final decision SA Power Networks determination 2015–16 to 2019–20 Attachment 3–Rate of return, October 2015, p. 375.

⁷¹⁸ Email correspondence from the ASX to Brailsford et al. dated 26 May 2004, reported in Brailsford, Handley, Maheswaran, 'Re-examination of the historical equity risk premium in Australia', Accounting and Finance, Vol. 48, 2008, p. 80; See AER, Final decision SA Power Networks determination 2015–16 to 2019–20 Attachment 3–Rate of return, October 2015, p. 375.

⁷¹⁹ Brailsford, Handley, Maheswaran, 'Re-examination of the historical equity risk premium in Australia', *Accounting and Finance*, Vol. 48, 2008, p. 81.

⁷²⁰ Brailsford, Handley, Maheswaran, 'Re-examination of the historical equity risk premium in Australia', *Accounting and Finance*, Vol. 48, 2008

⁷²¹ Brailsford, Handley and Maheswaran, Re-examination of the historical equity risk premium in Australia, Accounting and Finance, vol. 48, 2008, pp. 85–86.

CEPA, AER RORG Expert Joint Report, 21 April 2018, p.59

⁷²³ See CEPA, AER RORG Expert Joint Report, 21 April 2018, p. 11: "It should be noted that not all experts were present in all the sessions and may therefore not have given views on all issues. The issues on which experts contributed were set out above in Section 1.1. Graham Partington (GP) was unavailable due to overseas commitments from 14 April 2018 and provided limited input from that date, but did have sight of the final draft. David Johnstone provided input on drafts until 10 April 2018.

The report indicates when most experts held a particular view. However, assessing the views was not a quantitative voting exercise, but a way of identifying alternative views and the reasons for them. Dissenting views of any expert were considered to be of value and may inform the views of the AER."

During the development of the 2013 Guidelines the ENA engaged NERA. NERA proposed an alternative adjustment to the Lamberton/Sydney Stock Exchange data set.⁷²⁴

We have considered the use of the NERA yield series at length in prior AER regulatory determinations.⁷²⁵ Following the expert session we have considered this issue again.

We chose to use the BHM data that contains the ASX adjustment in the 2013 Guidelines and continue to consider this is the appropriate data to use. The reasons we continue to hold this view are:⁷²⁶

- BHM investigated and confirmed that the adjustment applied by the ASX to address the upward bias in dividend yields for 1882 to 1961 return data was reasonable
- The ASX has provided and adjusted the data.
- Adjustments such as those previously proposed by NERA have not been adequately justified by NERA and do not lead to a material improvement in the quality of data. For example, John Handley stated that 'it is unreasonable to draw a conclusion about three-hundred data points from a sample of only seven of those data points. Second and more fundamentally, NERA has not reconciled their data back to the Lamberton data as illustrated below'.⁷²⁷
- NERA's estimates contain adjustments that do not lead to complete reconciliation with the Lamberton data and did not establish that there was a downward bias in the BHM data set.

As such, we consider that our historical return analysis should be based on the updated BHM data set containing the ASX adjustment, consistent with the approach in our 2013 Guidelines.⁷²⁸ In line with this view in this decision we use updated BHM data.

7.3.2.2 Arithmetic vs Geometric Averages

The debate between using arithmetic averages and geometric averages has been a common feature of submissions to us over recent years and in submissions during this guideline review process.

CCP 16 contends that the AER has not placed adequate reliance on the outputs of the geometric average nor critically assessed the problems in the arithmetic average given the volatility of the annual Australian equity market returns of 17.7 per cent that

⁷²⁴ AER, Better regulation explanatory statement rate of return guideline (appendices), December 2013, p. 83; NERA Economic Consulting, The Market, Size and Value Premiums – A report for the Energy Networks Association, June 2013, p. 8 - 17).

⁷²⁵ See for example, AER, Ausgrid Final Determination, Attachment 3 - Rate of Return, April 2015, pp 321-326.

AER, Final Decision TransGrid transmission determination 2015-2018, 30 April 2015, pp.312-316.

⁷²⁷ Handley, *Further advice on the return on equity*, April 2015, p. 8.

⁷²⁸ AER, December 2013 Guidelines Explanatory Statement – Appendices,pp.83-84.

Dimson, Marsh and Staunton (2015) report.⁷²⁹ The service providers and the Network Shareholder Group (NSG) stated less or no weight should be given to geometric averages of HER.

Currently we base our estimate primarily on arithmetic returns, and have regard to the highest value from our set of geometric averages when forming a 'floor' of a potential point estimate. We acknowledge the potential downward bias of the geometric returns in this method⁷³⁰, but also take into account the extra information the geometric average returns when determining an estimate for the MRP.⁷³¹

Some submissions state that we should not give any weight to the geometric return.⁷³² One of the reasons this is argued is because the geometric average return compounds returns over a number of years, whereas the AER is estimating a single year return on capital to use in the AER's post tax revenue model (PTRM). This was also discussed in the concurrent evidence session.⁷³³

There have also been submissions during the guideline review process which suggest the AER should not be giving the arithmetic average any preference over the geometric average in estimating the MRP.⁷³⁴

Partington and Satchell have also advised that it is appropriate for the AER to consider return periods of more than one year.⁷³⁵ They noted that investors compound returns and whether or not the AER compounds returns is not relevant to the return that investors require. Given investor holding periods of more than one year they noted that it is appropriate for the AER to have regard to the geometric average. In the concurrent evidence session Satchell stated that he was disinclined to remove the use of either method entirely given the information available.⁷³⁶ Blume and Jacquier et al also show that where the holding period is more than one year, then the arithmetic mean of one year returns is an upward biased measure.⁷³⁷

This topic was also the subject of previous consideration in the Australian Competition Tribunal which concluded it was appropriate to rely on both sets of estimates.⁷³⁸

⁷²⁹ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, May 2018, p.88.

⁷³⁰ AER, APA VTS Final Decision: Attachment 3 - Rate of Return, November 2017, p.76.

⁷³¹ AER, APA VTS Final Decision: Attachment 3 - Rate of Return, November 2017, p.206.

⁷³² For example, ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.79-80.

AER, Concurrent evidence session 05 April 2018 transcript, April 2018, pp.55-60.

⁷³⁴ For example, CRG, *ROR submission final*, 04 May 2018, pp.53-54.

⁷³⁵ Partington and Satchell, *Report to the AER: Allowed Rate of Return Guideline Review*, 21 May 2018, p. 34.

⁷³⁶ AER, Concurrent evidence session 05 April 2018 transcript, April 2018, p. 57

⁷³⁷ Blume ME, Unbiased Estimators of Long-Run Expected Rates of Return, Journal of the American Statistical Association, vol. 69, 1974, pp. 634–638; Jacquier E, Kane A, Marcus AJ, Geometric or Arithmetic Mean: A Reconsideration, Financial Analysts Journal, 59, pp.46-53.

⁷³⁸ Australian Competition Tribunal, Application by Envestra Ltd (No 2) [2012] ACompT4, 11 January 2012, paragraph 157

On balance, we consider there is sufficiently robust evidence to continue to consider geometric averages. For this reason we have maintained our approach of giving most weight to arithmetic averages but using the geometric range to set the lower bound of the overall HER range.

7.3.2.3 Time Periods

In the expert statement from the concurrent evidence session there was a statement which suggested only periods of at least 50 years should be considered when using historical excess returns to estimate the MRP.⁷³⁹ This would mean removing the 1980 and 1988 onwards periods from consideration.

Following the expert session we have considered this issue further.

Pink Lake Analytics, in a 2017 report to the ERA, raised issues about potential uncertainty with older data, as highlighted by the different possible adjustments, and suggested a simple solution would be to exclude the pre-1937 period.⁷⁴⁰ It also suggested testing for structural breaks would be a more rigorous test of which periods were relevant.⁷⁴¹ It also acknowledged that the older time periods have lower standard errors and minimising this is desirable, but making decisions to exclude periods because of simple heuristics should be done with caution because they can be abused.⁷⁴²

We accept that standard errors are improved by using a longer data set. However, as financial conditions change through time, so do investors required risk premiums, we consider that estimates using more recent data are likely to be less biased. Therefore, we consider that we should continue to observe estimates from a range of time periods including more recent periods.

7.3.2.4 Time Horizon

A point raised in the submissions has been that the MRP should not be calculated using the geometric return because our one year return is not compounded within our modelling, and so should not use compounded return averages.⁷⁴³ However there is debate over what time period the MRP should be estimated over.

We accept that the PTRM is a single year model, and calculates the allowed revenue based on yearly estimates. However the return on equity, and the MRP, is implicitly estimated over a 10 year forward looking holding period given our anchor for the SML is the yield to maturity on Commonwealth Government Bonds with term to maturity of 10 years.

AER, Concurrent evidence session 05 April 2018 transcript, April 2018, p. 60

⁷⁴⁰ Pink Lake Analytics, *Estimation of the Market Risk Premium*, December 2017, p.9

⁷⁴¹ Pink Lake Analytics, *Estimation of the Market Risk Premium*, December 2017, p.9

⁷⁴² Pink Lake Analytics, Estimation of the Market Risk Premium, December 2017, p.8

⁷⁴³ ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.80-82

During the second concurrent evidence session, Hancock also stated that a single year arithmetic estimate, without any adjustment for previous return volatility, would not be sufficient. He therefore recommended looking at several time periods in order to arrive at a satisfactory MRP estimate.⁷⁴⁴

We have also been advised that it is appropriate for us to consider return periods of more than one year as investors holding periods are more than one year.⁷⁴⁵

In light of the discussions above we remain of the view it is appropriate to have regard to both arithmetic and geometric historical excess returns, over various time periods.

7.3.2.5 Updated Data

Table 25 displays data taken as single year historical excess returns over various time periods. From this we reach the arithmetic historical excess return range of 6 per cent to 6.5 per cent and the geometric excess return range of 4.2 per cent to 5.0 per cent. We were advised the final return period of 2000-2017 presented in the discussion paper is likely not large enough to be statistically reliable.⁷⁴⁶ Therefore, we have not given it any consideration.

It is important to note that whilst more recent data is likely more represented of current market conditions and investor expectations, averages over longer sampling periods will have lower standard errors all else equal.

Whilst there is an increase in the arithmetic average between the period starting in the 1937 and the periods starting in 1958 and 1980, the geometric average remains fairly consistent. We recognise that the geometric average value is likely to be below that of a best forward looking annual risk premium estimate. However, the consistency in the geometric values from different time periods implies annual return volatility may have caused the observed arithmetic excess returns to have increased over the periods starting in 1958 and 1980. This is also supported by the second column in Table 25that shows the arithmetic annual return volatility over each period

Large differences between the arithmetic and geometric averages arise when there is high volatility in annual returns over the sampling period. As annual return volatility decreases over a sampling period, the gap between the arithmetic average annual return and the geometric average annual return decreases. Our discussion in 7.3.5 shows that we are in a time of relatively low volatility in relation to the long term average. Because of this, we might expect a forward looking estimate of the current MRP to be somewhere between the geometric and arithmetic historical returns. CRG submitted that we should set the MRP at 3.6 per cent by giving more weight to the geometric mean as measured for the period 1984–2017.⁷⁴⁷ We have regard to the

⁷⁴⁴ Hancock, AER second concurrent evidence session, 5 April 2018

⁷⁴⁵ Partington and Satchell, Report to the AER: Allowed Rate of Return Guideline Review, 21 May 2018, p. 34

⁷⁴⁶ AER, Concurrent Evidence Session 2, Proofed Transcript, p.60.

⁷⁴⁷ CRG, Submission to the AER rate of return guideline review, 04 May 2018, pp.57.

extra information the geometric average returns provide. However, we do not consider that they provide evidence of such significance to directly estimate the MRP based on a single historical period's geometric average.

Sampling period	Arithmetic average	Arithmetic return Standard Deviation	Arithmetic average (2013 guideline)	Geometric average	Geometric average (2013 guideline)
1883–2017	6.3	1.63	6.3	5.0	4.8
1937–2017	6.0	1.92	5.9	4.2	3.9
1958–2017	6.5	2.17	6.4	4.2	3.8
1980–2017	6.4	2.12	6.3	4.3	3.8
1988–2017	6.0	1.72	5.7	4.5	3.6

Table 25 Historical excess returns (per cent)

Source: Handley, An estimate of the historical equity risk premium for the period 1883 to 2011, April 2012, p. 6. AER update for 2012–2017 market data. The 2013 guideline values are taken from data up to December 2012.

Notes: Calculated using an assumed imputation value (or theta value) of 0.6.

In the 2013 Guidelines we estimated a range of possible MRP values of 5 to 6.5 percent based on arithmetic and geometric average annual returns. We then estimated a point estimate of 6.5 per cent.

Based on expert and stakeholder advice we acknowledge that the geometric average is downwardly biased, however it does provide information and should be considered in our range.

7.3.3 Dividend Growth Models

Dividend Growth Models (DGMs) can use analyst forecasts of current dividends combined with estimate of dividend growth and the current price to estimate an implied MRP. A basic constant growth dividend growth model is algebraically expressed as follows:

$$P = \frac{D_1}{(r-g)}$$

Where:

P is the share price

D1 is expected dividend in the next period

r is cost of equity

g is expected growth rate

In the 2013 Guidelines we used estimates from DGMs to inform our point estimate of the MRP, where the point estimate came from within a possible MRP range of 5 per cent to 6.5 per cent, which in turn was based on observed historical excess returns. Relatively high results from our DGM resulted in the MRP point estimate being at the top or our range and set at 6.5 per cent.

Having reviewed submissions, the expert evidence session and further analysis, our view is to not move the market risk premium estimate based on DGM. We acknowledge that this places less reliance on the DGM than the 2013 Guidelines. This is because since 2013 our concerns about biases of the model and the divergent results from alternative versions of the model have increased.

7.3.3.1 Use of the DGM

We note the DGM divides opinion across stakeholders and experts. A number of submissions stated that the DGM is the only method readily available to the AER which is forward looking.⁷⁴⁸ It is considered forward looking because it relies on estimates of future growth rates, both in the long and short term. A number of submissions stated that the DGM should be given equal, or near equal, weight with the historical excess returns.⁷⁴⁹ Other submissions have stated that because of issues regarding assumptions and biases we should place no substantial weight on MRP estimates from the DGM.⁷⁵⁰

We recognise we had concerns with the use of DGMs in 2013 as well, and note ENA's view that our concerns are not new and therefore we should not adjust our view.⁷⁵¹ However multiple submissions have stated these issues have become better highlighted since the 2013 Guidelines. Since 2013, we have also received advice about further issues with the DGM and its lack of suitability to reliably track changes in the MRP.⁷⁵² In light of this advice about concerns we consider valid, we consider DGM

⁷⁴⁸ For Example, Ergon Energy and Energex, Ergon Energy and Energex submission on AER Issues Paper, 12 December 2017, p.6; Queensland Treasury Corporation, QTC submission to the RoR Guideline Review Issues Paper, December 2017, p.5.

⁷⁴⁹ APGA, Submission on AER Review of ROR Guideline, 04 May 2018, pp.12-13; EvoEnergy, Submission on Rate of Return Guideline Evidence Session, 04 May 2018, p.4; APA, Submission on Rate of Return Review, 04 May 2018, pp.28-32; ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.67-71; ATCO Gas Australia, Response to Revie of Rate of Return Guideline – Issues Paper, 12 December 2017, p.8; Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December 2017, p.4; Network Shareholder Group, Submission on rate of return issues paper, 12 December 2017, p.10; Queensland Treasury Corporation, QTC submission to the RoR Guideline Review Issues Paper, December 2017, p.5.

⁷⁵⁰ ECA, *Response to the AER Issues Paper*, December 2017, p.22; EUAA, AER Rate of Return Issues Paper October, December 2017, p.9; Major Energy Users, *Submission by the MEU to the review of the rate of return guideline*, 18 December 2017, p.16; CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.109-111.

⁷⁵¹ ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.67-71.

⁷⁵² Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 21 May 2018, p.33; Partington and Satchell, Report to the AER: Cost of Equity Issues 2016 Electricity and Gas Determinations, April 2016, pp.27-33; McKenzie & Partington, Report to the AER: Part A, Return on Equity, October 2014, pp.26-41.
based estimates of the MRP no longer sufficiently reliable to warrant increasing the MRP above 6 per cent per annum.

The major concern with the reliability of estimates from DGMs revolves around the challenge of forecasting the growth rates in dividends, particularly the terminal growth rate. This is discussed in detail below.

7.3.3.2 Model Selection

When referring to DGM evidence it is also important to note that there are multiple forms of DGM. Each of them use the inputs differently in order to estimate the market risk premium. There is no 'correct' dividend growth model and all DGMs seem likely to suffer from the same general limitations.

As DGMs come in different forms, there is an issue of which configuration is likely to lead to the best estimate of the MRP. For example, IPART use 5 separate DGMs in its estimation process as it considers this will lead to the best result.

In the 2013 guideline we arrived at a version of the DGM we believed was best suited to our regulatory task, and that is displayed below:

$$P_c = \frac{m \times E(D_c)}{(1+k)^{m/2}} + \sum_{t=1}^{N} \frac{E(D_t)}{(1+k)^{m+t-0.5}} + \frac{\frac{E(D_N)(1+g)}{k-g}}{(1+k)^{m+N-0.5}}$$

Where:

- Pc is the current price of equity, for which we use the S&P/ASX 200 index as the proxy
- E(Dc) is expected dividends per share for the current financial year⁷⁵³
- E(Dt) is expected dividends per share for the financial year t years after the current financial year
- m is the fraction of the current financial year remaining, expressed as a decimal point
- N is the time period after which dividend growth reverts to its long-term rate (for the two stage model, N = 2, for the three stage model N = 9)
- g is the expected long term growth rate in nominal dividends per share
- k is the expected return on equity for the market portfolio

We see no reason to move away from this particular construction of the DGM we have used since the 2013 Guidelines. We have also had no issues raised with our specific

⁷⁵³ We sourced dividend forecasts from Bloomberg. We have been informed by Bloomberg that its convention for reporting dividend forecasts on an index is to use calendar year forecasts as the relevant financial year forecasts.

model in submissions. As such all results quoted here, unless otherwise specified, are from the above model.

7.3.3.3 Long Term Dividend Growth Rate

As indicate by the algebra above, DGM estimates are highly influenced by the terminal, or long term, dividend growth rate.

In the 2013 Guidelines we estimated a central terminal growth rate of 4.6 per cent with high growth rate of 5.1 per cent and a low growth rate of 3.78 per cent.

Our central dividend growth rate of 4.6 per cent was estimated by Martin Lally based on an expected nominal GDP growth for Australia of 5.6 per cent and an equity leak of 1 per cent.⁷⁵⁴ The upper bound was formed from the same advice, and the 3.78 per cent was from the advice of Partington and McKenzie.⁷⁵⁵

In his 2013 advice Partington suggested our growth rates may be too high given economic conditions.⁷⁵⁶ We note economic conditions do not appear to have improved markedly since 2013.

We also noted in our March 2018 discussion paper that those models which use a variable growth rate based on the risk free rate now generate results that are substantially below our model, whereas before they were well within an accepted range.⁷⁵⁷ Whilst growth rates based on GDP estimates may still be suitable, it is important to consider if they are still the best estimate of long term dividend growth for use in our DGM, and how reliable the different estimates might be.

Different sources give GDP growth estimates that are materially different and there is a significant margin of error in those estimates. ⁷⁵⁸ A research discussion paper by the RBA also indicates its own GDP forecasts may biased.⁷⁵⁹ Previous advice from Partington also suggests that the GDP growth rate may not be a good measure of dividend growth.⁷⁶⁰

⁷⁵⁴ Martin Lally, *The Dividend Growth Model*, March 2013, pp. 13-20.

⁷⁵⁵ Partington & McKenzie, *DGM Final report*, December 2013, p.4,15.

⁷⁵⁶ Partington & McKenzie, *DGM Final report*, December 2013, pp.4-15.

AER, MRP Discussion Paper, March 2018, pp.20-21.

⁷⁵⁸ RBA, May Statements on Macroeconomic Policy - Economics Outlook, May 2018 states growth will be above 3%; The IMF forecasts Australian growth will fall to 2.6% in 2023 <u>http://www.imf.org/en/Countries/AUS</u>; Economist Intelligence Unit puts long term growth between 2018 and 2030 as 2.2% and between 2018 and 2050 as 2.1%. <u>http://country.eiu.com/article.aspx?articleid=156657399&Country=Australia&topic=Economy&subtopic=Long-term+outlook&subsubtopic=Summary</u>; the OECD forecasts growth around 2.7% between now and 2060 <u>https://data.oecd.org/gdp/gdp-long-term-forecast.htm</u>;

⁷⁵⁹ Reserve Bank of Australian, Research discussion paper - Estimates of Uncertainty around the RBA's forecasts, Peter Tulip and Stephanie Wallace, Nov 2012.

⁷⁶⁰ Partington & McKenzie, *DGM Final report*, December 2013, p.13.

A report by Bianchi, Drew and Walk sponsored by Challenger Limited also stated that long term real dividend growth rates in Australia have been around 1 per cent.⁷⁶¹ An expected 1 per cent per annum real dividend growth rate implies an expected nominal dividend growth rate of around 3.5 per cent per annum if expected inflation is 2.5 per cent per annum.

In addition, Damodaran and Fenebris suggests, based on finance theory, that GDP Growth is materially similar to the risk free rate, and as such the current risk free rate holds as a suitable anchor for the long term dividend growth rate.⁷⁶² Damodaran suggests a suitable estimate for the long term dividend growth is simply the current risk free rate, whilst Fenebris considers and appropriate terminal dividend growth rate is the risk free rate less 2 per cent per annum.

It was noted in the submission from the ENA that Fenebris' growth rate can lead to results that seem implausible, and advice from Partington and Satchell agreed with that.⁷⁶³ The advice from Partington and Satchell also stated that whilst the general trend of the CGS rates may follow the GDP growth it has the potential to be a poor predictor when the risk free rate is particularly high or low.⁷⁶⁴

We accept that there are a wide range of possible growth rates and that at times a growth rate based on the risk free rate may not produce suitable results, especially in times of extreme growth or high/low risk free rates. We also note the ERA stated in its recent decision that the 5.1 per cent growth rate may not be suitable in the current economic climate⁷⁶⁵.

Ultimately, there are numerous issues surrounding the estimation of dividend growth rates selection and there is a wide variety of potentially acceptable growth rates which could be used in the DGM. With the range of potential growth rates varying from as low as 1 per cent to as high as 5.5 per cent, the DGM based MRP estimate could vary by around 4 per cent purely due to the chosen growth rate.⁷⁶⁶ Given this very large potential for error in the MRP estimate driven by growth rate selection, our decision must account for the potential error or unsuitability in the estimate. In this decision we have taken this into account by giving relatively lower weight to estimates of the MRP from dividend growth models.

⁷⁶¹ Bianchi, Drew & Walk, *The Unpredictable Equity Risk Premium*, November 2015, p.24.

⁷⁶² Fenebris, Determination of a Market-Wide Implied Cost of Capital, May 2016; Damodoran, Closure in Valuation, <u>http://people.stern.nyu.edu/adamodar/pdfiles/eqnotes/dcfstabl.pdf</u>

⁷⁶³ ENA, Submission to the AER Review of Rate of Return Guideline, 4 May 2018, p74.

⁷⁶⁴ Partington & Satchell, *Final Report to the AER 2018*, 22 May 2018, pp.31-33.

⁷⁶⁵ ERA, Draft Decision on Proposed Revisions to the Access Arrangement for the Western Power Network, 2 May 2018, p.34.

⁷⁶⁶ Previous DGM results have shown a change in growth rate can have an almost one for one inverse impact on the MRP estimate.

7.3.3.4 Other issues raised with the DGM

Whilst no stakeholders consider the DGM is a model without issues, ENA has submitted that because there are no new issues with the model we should not adjust our view on the model's regulatory suitability.⁷⁶⁷ However, multiple submissions have stated these issues have become better highlighted since the 2013 Guidelines, and diverging results shows their potential unsuitability for use in a regulatory setting.⁷⁶⁸

This section covers other issues raised with estimating the MRP using dividend growth models.

Analyst Forecasts

We have previously observed that analyst forecasts are upwardly biased.⁷⁶⁹ We have received advice that this is a real and present issue for the DGM, and there has not been any counter argument to this except for some analysis on the ASX 20 submitted by Frontier in its 2016 report to the AER. Frontier claimed the analysis actually showed a downward bias in forecasts, however it was a single survey with a small sample and as such we previously gave the argument little weight as per expert advice.⁷⁷⁰

During the second concurrent evidence session, Partington raised a related point which he referred to as incurable optimism.⁷⁷¹ This references the point that in most applied examples of the DGM in use the short term dividend growth rate must decrease by a significant amount to reach the long term dividend growth rate. Partington's point states that at some point these short term dividend growth rates must be equal to long term rates, or at times be below them in order for the short term and long term growth rates to fulfil their purposes.

As analyst forecasts are an essential component of the DGM, having an upward bias of an unknown amount makes it hard for us to place much weight on the estimates from these models.

Inflation

The construction of dividend growth models regularly uses expected inflation to reach the nominal growth figure. This can lead to differing final growth rates depending on the estimate of expected inflation used. This was raised by Partington in the concurrent evidence session on 5 April 2018.⁷⁷² This represents a further issue with assumptions being made in order to arrive at a DGM based MRP estimate.

⁷⁶⁷ ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.67-71.

 ⁷⁶⁸ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, p.
 115; CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.56.

⁷⁶⁹ AER, APA VTS Final Decision, Novemebr 2017, p.217.

AER, APA VTS Final Decision, November 2017, pp.217-218.

Partington, AER concurrent evidence Session, 05 April 2018, p81.

⁷⁷² Partington, AER concurrent evidence Session, 05 April 2018, p81.

Term Structure

Previous advice from Lally has stated that the DGM may produce upwardly biased results when the risk free rate is low due to the term structure of equity.⁷⁷³ This point was raised within the concurrent evidence sessions as well.

We acknowledge that biases in our DGM may exist due to the term structure of equity and may vary over time due to movements in the risk free rate.

Sticky Dividends

A point that has been raised in previous decisions⁷⁷⁴ and was highlighted in the second concurrent evidence session⁷⁷⁵ is that firms display traits of altering their dividend payouts in ways that are not symmetrical. When markets encounter poor returns firms are less likely to lower their dividend payout ratio than they are to increase them during good times.

This can cause the DGM to have further upward bias, as dividends may remain artificially high for a period after conventional economic theory suggests the return on capital should have dropped. As such, MRP estimates from the DGM could well be upwardly biased at times due to this behaviour from firms.⁷⁷⁶

Frontier, in its 2018 report to the AER, submit that because the RBA data shows earnings forecasts have not fallen as much as expected in recent years that Sticky Dividend concerns should not be considered.⁷⁷⁷ Whilst this concern may not be an issue at the current time, it remains a concern about the model more generally and highlights its dependency on forecasts and business practices.

Stable Return on Equity

We analysed the historical results from our construction of the DGM and found that there is as much as an 80 per cent negative correlation between the MRP estimates from the DGM and the risk free rate. This means the DGM implicitly (in its application) assumes a stable return on equity. This raises two concerns for us

Firstly, this is inconsistent with our view that there is a lack of support for an inverse relationship between the risk free rate and the MRP. This was discussed at length in the second concurrent evidence session, and is covered in more detail in 0.

Secondly, even if the DGM was reliable, it seems overly complicated to use a model like a DGM when the movements in the MRP could largely explained by movements in the risk free rate.

⁷⁷³ Lally, *The Dividend Growth Model*, 4 March 2013, pp.5-9.

AER, APA VTS Final Decision – Rate of Return, November 2017, pp. 216-217,212.

Partington, AER concurrent evidence Session, 05 April 2018, p81.

Partington & Satchell, *Report to the AER, Return on equity (Updated)*, April 2015, p.49.

⁷⁷⁷ Frontier, *Rate of Return for Ausgrid 2018*, April 2018, pp.152-154.

Dividend Reinvestment

Due to the DGM's reliance on dividend payout forecasts, any scheme which alters the value of dividends paid out is important to take account of when considering its suitability for use by the AER. Partington highlighted in the second concurrent evidence session that some firms operate dividend re-investment plans.⁷⁷⁸ As such firms that may advertise a 6 per cent dividend distribution may only have a true dividend yield of 4 per cent if there is a 30 or 40 per cent participation in these dividend re-investment plans. This is likely to lead to upward bias in the MRP estimation from the DGM.

7.3.3.5 Updated Data

Taking note of the results from various growth rates which can be applied to the DGM and the other models which are in regulatory use in Australia it is clear that there is a wide range of results for the MRP estimate which the DGM could support.

The table below shows MRP estimates using dividend growth rates from 3.78 per cent to 5.1 per cent, although we note the best estimate of the dividend growth rate may be lower than 3.78 per cent for the reasons discussed earlier

Sensitivity	Two stage model	Three stage model
Baseline		
4.6 per cent long-term growth rate	7 47	7 38
2 month average to end April 2018		
unadjusted analysts' forecasts		
5.1 per cent long-term growth rate	7.94	7.77
3.78 per cent long-term growth rate	6.70	6.73
6 months to end April 2018	7.42	7.37
12 months to end April 2018	7.52	7.46
Analysts' forecast + 10 per cent	8.05	7.95
Analysts' forecast - 10 per cent	6.90	6.81
Combined - low	6.08	6.15
Combined - high	8.56	8.42

Table 26 Dividend Growth Model Results with Sensitivity Analysis

Source: Bloomberg, AER analysis.

Notes:All market risk premium estimates are based on an assumed theta of 0.6.Combined - low is based on 3.78 per cent growth, 6 month averaging, analysts' forecasts - 10 per cent.Combined - high is based on 5.1 per cent growth, 12 month averaging, analysts' forecasts + 10 per cent.

Partington, AER concurrent evidence Session, 05 April 2018, p81.

As further illustration of the DGM's sensitivity to choice of inputs and wide range of results, Figure 20 displays results from different constructions and applications of the DGM produce. The variable growth rate model replaces the static growth rate of the AER's regular DGM with the monthly 10 year CGS yield. The IPART's DGMs are based on its approach discussed in section 7.3.6. The wide range of results also highlights how MRP estimates from the same DGM can diverge under different assumptions even using the same model.



Figure 20 Results from various DGM constructions from 2013-2018

Source: AER own analysis, Variable growth rate equal to 10 year CGS yield but model construction is otherwise identical to the AER's DGM model, IPART February update

7.3.4 Survey Evidence

We consider market surveys continue to support an MRP between 5.5 per cent and 6.5 per cent.

We note survey evidence comes from market practitioners who are asked what they expect the MRP to be in the Australian market. These surveys take on different forms and can vary in different ways, including questions asked, type of participants and number of participants. As such it is important to view each piece of evidence in the context it is presented.

In the approach to date we have used the survey evidence to inform our MRP estimate. It informs us about investors' and market practitioners' expectations and/or what they apply in practice

Submissions to this guideline process were split on the use of survey evidence, with some stakeholders supporting increasing the use of surveys, such as Energy Australia⁷⁷⁹, whilst other stakeholders such as the APGA⁷⁸⁰ opposed any use of surveys at all. Proponents argued they were direct results from market practitioners, but opponents countered saying that surveys are easily skewed and the headline data can be misleading.

In the concurrent evidence session there was disagreement as to how the surveys should be considered. Gray and Wheatley stated that there should be no consideration given to survey evidence whilst Partington stated that their forecasting ability is limited but they are a good measure of expectations across the market.⁷⁸¹

We recognise that surveys have limitations and are not at a level of reliability as to give it weight as a direct estimation method of the MRP. However, we consider that it has some value and use it to inform us of investor expectations. There has not been any significant change in evidence regarding surveys which persuade us that we should change our 2013 Guidelines position on the role of survey data.

⁷⁷⁹ Energy Australia, Submission to AER Rate of Return Evidence Sessions, 04 May 2018, p.2.

⁷⁸⁰ APGA, Submission on AER Review of ROR Guideline, 04 May 2018, p.12.

⁷⁸¹ Partington, Gray & Wheatley, *AER Concurrent Evidence Session*, 5 April 2018, pp.89-90.

Table 27 shows that since the 2013 Guidelines surveys have largely supported an MRP between 5.5 per cent and 6.5 per cent. We note that several factors play a part in the overall value of each survey such as, timing of survey, the sample, wording of the questions and response rate and non-response bias. In this context, we continue to recognise that triangulation across surveys can reduce the limitations associated with particular surveys.⁷⁸² Only the 2017 and 2018 Fernandez surveys⁷⁸³ indicate that expectations are higher than our range. We do not consider that on its own these surveys move us to think that our range of 5.5 to 6.5 per cent is unreasonable or that we should choose a point estimate at the top of the range.⁷⁸⁴

⁷⁸² McKenzie and Partington, Supplementary report on the MRP, February 2012, pp.17-20.

⁷⁸³ Frontier Economics acknowledges that KPMG 2017 survey's median MRP is 6.0 but suggest that more should be read into the KPMG website comments regarding valuers adjustments to the current risk free rate. See, Frontier, *Ausgrid's Rate of Return*, April 2018, pp126-128, section 18.5.1.

⁷⁸⁴ It is worth noting that 2017 survey had markedly fewer responses than previous and subsequent surveys conducted by the same author. The subsequent survey (2018) appears to also indicate a downward movement relative to 2017.

Table 27 MRP Survey Results

Survey	Numbers of responses	Mean (per cent)	Median (per cent)	Mode (per cent)
Fernandez et al (2012)	73	5.9	6.0	N/A
KPMG (2013)a	19	N/A	6.0	6.0
Fernandez et al (2013)	17	6.8	5.8	N/A
Asher and Hickling (2013)	46	4.8	5.0	6.0
Fernandez et al (2014) b	93	5.9	6.0	N/A
Asher and Hickling (2014) c	27	4.4	4.6	6.0
Fernandez et al (2015)	40	6.0	5.1	N/A
KPMG (2015) d	~27	N/A	6.0	6.0
Asher and Carruther (2015)	29	4.9	N/A	N/A
Fernandez et al (2016)	87	6.0	6.0	N/A
Carruther (2016)	24	5.3	N/A	N/A
Fernandez et al (2017)	26	7.3	7.6	N/A
KPMG (2017)	45	N/A	6.0	6.0
Fernandez et al (2018)	74	6.6	7.1	N/A

Sources: Several survey reports.785

Notes: a) While this survey had 23 market participants, 19 specified what market risk premium they used.

b) The 2014 survey did not report the response rate. AER staff obtained this information from Professor Fernandez via email correspondence on 22 July 2014.

c) The response rate for this survey is lower than the response rate in previous Asher and Hickling surveys because the survey took place from 5 December 2014 to 14 December 2014, which was very close to Christmas. AER staff obtained the mode from Associate Professor Anthony Asher via email correspondence on 17 September 2015.

d) The KPMG (2015) survey had 29 market participants, but figure 24 indicates that not all the market participants gave a response for the market risk premium. However, visual inspection indicates that the response rate was approximately 27.

⁷⁸⁵ Fernandez, Ortiz, Acín, Market risk premium used in 71 countries in 2016: a survey, May 2016; KPMG, Australian valuation practices survey 2015, May 2015; Fernandez, Ortiz, Acín, Discount rate (risk-free rate and market risk premium) used for 41 countries in 2015: a survey, April 2015; Asher and Hickling, Equity Risk Premium Survey 2014, Actuaries Institute, April 2015; Fernandez, Linares, Acín, Market Risk Premium used in 88 countries in 2014, IESE Business School, June 2014; Asher and Hickling, Equity Risk Premium Survey, Actuary Australia, December 2013; Fernandez, Arguirreamalloa and Linares, Market Risk Premium and Risk Free Rate used for 51 countries in 2013, IESE Business School, June 2013; KPMG, Valuation Practices Survey 2013, February 2013; Fernandez, Arguirreamalloa and Corres, Market Risk Premium used in 82 Countries in 2012, IESE Business School, January 2013; Asher and Carruther , *Equity Risk Premium Survey 2015, Actuaries Digital*, May 26 2016; David Carruthers, Equity Risk Premium Survey 2016, 8 March 2017; Fernandez, Linares, Acín, Discount Rate (Risk-Free Rate and Market Risk Premium) used for 41 Countries in 2017: a survey, April 2017; KPMG, *KPMG Valuation Practices Survey*, July 2017, Fernandez et al, Market Risk Premium used for 59 countries in 2018, April 4 2018.

7.3.5 Conditioning Variables

Having evaluated the evidence presented below in the context of their limitations, we consider that conditioning variables, on balance, do not support any adjustment to our estimate of MRP based on HER data.

Dividend yields do not provide a discernible trend relative to long term averages and therefore does not indicate any directional movement. However, credit spread data appears to be trending lower since 2015 but on its own is insufficient to move our estimate of MRP from 6.0 per cent.

The 2013 Guidelines set out three conditioning variables that we use to inform (or 'condition') our initial MRP estimate. These are implied volatility, dividend yields and credit spreads. We take these into account when estimating the MRP because they provide an indication of changes in market conditions.⁷⁸⁶ We are cautious on how we use this evidence given their limitations but consider these are relevant and give them some consideration.

Frontier Economics submits that in the absence of formal econometric mapping to a point estimate of the MRP, it is unclear how to interpret the evidence. Therefore, Frontier Economics suggests that we should not be considering conditioning variables to estimate the MRP.⁷⁸⁷ Frontier Economics' concerns appear to stem from its interpretation on how we use conditioning variables. In particular, it appears to consider that we give conditioning variables weight as evidence in their own right. We reiterate that, consistent with our 2013 Guidelines, we consider that conditioning variables do not provide reliable estimates on their own.⁷⁸⁸ Rather, we use conditioning variables to inform our point estimate derived from HER. In this context, given we do not use conditioning variables to develop our initial point estimate, econometric mapping to a point estimate is not a pre-requisite.

We agree with CCP 16 that conditioning variables can provide useful evidence by showing investors' perceptions of risks, but we are aware of the limitations and cautious in using this evidence.⁷⁸⁹ As noted in our 2013 Guidelines, we consider conditioning variables should be used symmetrically through time to avoid bias. That is, irrespective of whether each conditioning variable indicates a higher or lower MRP at any given time, we will continue to consider them consistently over time to inform our estimate.^{790 791}

⁷⁸⁶ AER, Explanatory statement – Rathe of return guideline, December 2013, pp.94-97.

⁷⁸⁷ Frontier, *Ausgrid's Rate of Return*, April 2018, pp128-131, section 18.5.2.

⁷⁸⁸ AER, Explanatory statement – Rathe of return guideline, December 2013, pp.94.

⁷⁸⁹ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.112-113.

⁷⁹⁰ AER, Explanatory statement – Rathe of return guideline, December 2013, pp.92.

⁷⁹¹ AER, APA VTS Final Decision – Rate of Return, November 2017, pp.220-228.

We consider that the submissions have not presented us with evidence to persuade us that we should change our 2013 Guidelines position on the role of conditioning variables.

7.3.5.1 Implied volatility

The implied volatility approach assumes that the MRP is the price of risk multiplied by the volume of risk (volatility).⁷⁹²

Figure 21 shows volume of risk in the market portfolio estimated using the implied volatility index up to 1 May 2018. Implied volatility has been lower over the past 12 months with an average of 12.76 compared to a long term average (since 1997) of 17.8. Between the start of 2010 and the time we finalised our 2013 Guidelines the implied volatility average was 18.66.

During 2015 and 2016 there was a slight increase in volatility which increase was also seen in other conditioning variables. There are spikes around that time in almost all of the conditioning variables shown, however they have all flattened out again and suggest volatility is below the long run average. This is due to the ASX (and global markets) posting major lows from August 2015 to February 2016⁷⁹³ from a range of factors such as falls in the US stock market, concerns with quantitative easing and falling oil prices.⁷⁹⁴

Volatility has been below the long term average for most of the period since our 2013 Guidelines, and has been significantly below the average of the 5 years that led up to the 2013 Guidelines. These consistently lower volatility values indicate that there has been less risk in the market over the past years. We have previously received advice that it is unlikely to have a relatively high MRP when implied market volatility is low.⁷⁹⁵ Whilst there is no guarantee volatility will remain low, the current persistent trends suggest we are below the long term average with no indication that the series is mean reverting. It is also important to note that an increase in volatility alone would not necessarily drive an increase in the MRP.

⁷⁹² This was based on Merton, R.C., On estimating the expected return on the market: An explanatory investigation, *Journal of Financial Economics*, 1980, Vol 8, pp.323–361.

⁷⁹³ https://afma.com.au/data/afmr/2016%20AFMR.pdf;

⁷⁹⁴ https://www.reuters.com/article/us-usa-markets-idUSKCN0VL0XO

⁷⁹⁵ Partington and Satchell, *Report to the AER: Discussion of submissions on the cost of equity*, 29 May 2017, p.47.



Figure 21 Volatility Index of ASX200



7.3.5.2 Dividend yields

Consistent with our 2013 Guidelines, we use dividend yields as a directional indicator of the MRP. We do this by comparing current dividend yields with the average over time.⁷⁹⁶. Figure 22 shows the dividend yields against their historical average up to 1 May 2018. As shown, dividend yields are currently sitting around their long term average of 4.25 per cent. It also shows that dividend yields increased in 2015, decreased in 2016 and again saw an increase in early 2018. Overall, there is no clear trend away from the long term average and therefore does not provide an indication about the current direction of the MRP.

⁷⁹⁶ AER, Explanatory statement – Rate of return guideline (appendices), December 2013, p.94.





Source: AER analysis; sourced via Bloomberg code AS51. Long term average taken from the start of the data series in 2000.

7.3.5.3 Credit spreads

Consistent with our 2013 Guidelines, we use credit spreads as a directional indicator to inform our MRP estimate.⁷⁹⁷ Credit spreads are the spreads between the risk free rate (the yield on Australian government securities) and the return on debt for different debt instruments. We look at whether the spreads are widening, stabilising or narrowing as an indicator of changes in market conditions

Figure 23 shows the spread between state government debt and the Australian government debt up to 1 May 2018.⁷⁹⁸ As shown, credit spreads peaked during the 2008 global financial crisis (GFC) and then increased significantly in 2011-12. The spreads have been slowly trending lower since 2015 and are now around pre GFC levels.

⁷⁹⁷ AER, Explanatory statement – Rathe of return guideline, December 2013, pp.96.

⁷⁹⁸ Using debt with 3 years term to maturity as more data are available.



Figure 23 Spread of State Government Debt

Source: AER analysis; Spreads from Australian government securities to state government bonds with 3 years term to maturity, sourced via Bloomberg interest rate statistics.

Figure 24 shows the credit spreads for A-rate and BBB rated corporate debt instruments over yields on Australian government securities. These credit spreads spiked during the GFC and again increased in 2012. The spreads are showing a clear downward trend since 2015 and currently sitting around pre GFC levels. The swap rate spread is sitting lower than pre GFC levels.



Figure 24 Australian Bond spreads over Government Yields

Source: RBA, Chart Pack, download May 2018

7.3.6 Other Regulators

We consider our point estimates of 6 per cent is within the range from other regulators' decisions.

As part of the 2013 Guidelines we looked at other regulators' estimates for the MRP.⁷⁹⁹ In response to that we have received submissions that stated the exercise was largely circular and achieved little.⁸⁰⁰

For the upcoming guideline we propose to continue looking at other regulators' decisions. However, we would look at the rationale and arguments put forward in support of/opposition to the evidence and methods.

We also want to ensure that there is no evidence in consideration elsewhere that we have not been made aware of or considered.

The table below summarises MRP values set by other Australian regulators.

Table 28 MRP from past deci	sions by other regulators
MRP (per cent)	

	MRP (per cent)
min	6.00
max	7.75

Source: AER Analysis of regulatory decisions from IPART, ERA, QCA and ESCV from March 2017 to March 2018.801

We do not consider foreign regulators as part of our decision because they often have different regulatory objectives and regimes, as such it is hard to give a direct comparison on rate of return parameters such as MRP. This is the same position we have taken in previous decisions and we see no reason to depart from this with no new evidence submitted.⁸⁰²

7.3.6.1 IPART

IPART currently sets an MRP based on 50 per cent weighting to historic averages and 50 per cent on current estimates. To build the historic average it looks at excess returns, whereas for the current estimates it relies largely on estimates from 5 versions of the DGM. There is also consideration of a method which uses 4 economic indicators when estimating the current estimate of MRP.⁸⁰³

⁷⁹⁹ AER, *Explanatory Statement - Rate of Return Guideline (Appendices)*, December 2013, p100.

⁸⁰⁰ Frontier, *The Market Risk Premium*, September 2016, p.32.

⁸⁰¹ IPART, Review of prices for rural bulk water services from 2 July 2017 to 30 June 2021 – Final Report, June 2017, IPART, Sydney Desalination Plant Pty Ltd – Review of prices from 1 July 2017 to 30 June 2022 – Final Report, June 2017, IPART, Prices for wholesale water and sewerage services, Sydney Water Corporation and Hunter Water Corporation – Final Report, June 2017, IPART, Draft Report – Review of Fares for Private Ferry Services, September 2017, IPART, Draft Report - Maximum fares for rural and regional bus services from 1 January 2018, October 2017, ERAWA, Determination on the 2017 WACC for the Freight and Urban Railway Networks, and for Pilbara railways, October 2017, QCA, Seqwater Bulk Water Price Review 2018-21, March 2018, IPART, Review of fares for private ferry services – Final Report, December 2017, IPART, WACC Bimanual update, February 2018

⁸⁰² AER, APA VTS Final Decision - Attachment 3, November 2017, p.87

⁸⁰³ IPART, IPART Review of our WACC Methodology, February 2018, p.52

We have stated previously that taking a simple average of multiple equity models, especially in this case where most have the same input of growth rate with no consideration for potential error, does not improve the validity of the model's results. This averaging also does not improve the potential biases that we consider are prevalent in the models. We have received advice that this would only be appropriate if that these were different estimates which were on average unbiased.⁸⁰⁴

When considering a historical MRP, IPART agree that excess returns data leads to an MRP estimate of 6 per cent.⁸⁰⁵

7.3.6.2 ERA

The ERA, in its recent draft decision for the Western Power network, used historic returns, dividend growth models and conditioning variables in order to arrive at its estimated of the MRP.⁸⁰⁶ Their MRP is calculated over a 5 year risk free rate, and arrived at a value of 6.2 per cent.⁸⁰⁷

In forming a lower bound it used a combination of the arithmetic and geometric averages and arrived at a historic estimate of the MRP at 5.6 per cent.⁸⁰⁸ It relies largely on the 2 stage DGM in forming its upper bound based on the DGM, and arrived at 7.6 per cent. The ERA then use conditioning variables to support an MRP towards the lower end of this range, arriving at an estimate of 6.2 per cent.⁸⁰⁹

The ERA presents evidence that we have considered and take a similar stance, using supporting evidence from our own previous decisions. We note that its historic estimate of the MRP and final point estimate are within our HER range of 5.0–6.5 per cent.

7.3.6.3 International Evidence

The 4 May 2018 submission from the Joint Energy Networks suggested that evidence from international firms put our equity risk premium as materially lower than other jurisdictions.⁸¹⁰ We do not take international regulator data into account because of the issues surrounding differences in regulatory procedures and tasks. This has been covered in previous decisions.⁸¹¹

⁸⁰⁴ Partington and Satchell, Report to the AER: Discussion of estimates of the return on equity, April 2017, p. 27.

⁸⁰⁵ IPART, IPART Review of our WACC Methodology, February 2018, p.47

⁸⁰⁶ ERA, Draft Decision on Proposed Revisions for the Western Power Network, 2 May 2018, pp.20-46

⁸⁰⁷ ERA, Draft Decision on Proposed Revisions for the Western Power Network, 2 May 2018, p. 63

⁸⁰⁸ ERA, Draft Decision on Proposed Revisions for the Western Power Network, 2 May 2018, p.29

⁸⁰⁹ ERA, Draft Decision on Proposed Revisions for the Western Power Network, 2 May 2018, pp.20-46

⁸¹⁰ AGIG, CitiPower, Multinet, Powercor, SA Power Networks, United Energy, Submission to Rate of Return Guideline Review, 4 May 2018, p.5;

AER, APA VTS Final Decision – Rate of Return, November 2017, pp. 296-297.

7.3.7 Wright Approach

We continue to observe results from the Wright model at the overall equity level but consistent with our 2013 Guidelines we will not consider its outcome to estimate the MRP.

The Wright approach uses a constant return on the market portfolio, estimated via either historical market returns or methods such as the DGM, and the prevailing risk free rate to estimate the MRP. As such it does not calculate the MRP as an observable parameter, but as the difference between the historic market returns and the prevailing risk free rate.

Under the 2013 Guidelines the Wright approach was given consideration at the overall return on equity level and no consideration in estimating the MRP.⁸¹² The view that the Wright Approach is valid and should be used by us in our MRP estimation processes was presented by several stakeholders including APA, the ENA and Cheung Kong Infrastructure.⁸¹³ They submitted, and were supported in the concurrent evidence session by Gray and Wheatley, that the Wright Approach should form one of the two bounds we consider when starting out on our estimate of the MRP.⁸¹⁴

We also received submissions from stakeholders such as the CCP and the ECA stating that the Wright approach should not be given weight at any level, due to lack of evidence of its theoretical underpinning in practice.⁸¹⁵ During the concurrent evidence session this view was also supported.⁸¹⁶ It was raised in the concurrent evidence session by Mr Sadeh that there would need to be evidence as to the inverse relationship between MRP and the risk free rate to consider the Wright approach.⁸¹⁷

The Wright Approach relies heavily on the assumption of a perfect, or at least near perfect, negative relationship between the risk free rate and the MRP. That is, for every basis point the risk free rate decreases the MRP increases the same amount, or vice versa, in order to keep the return on equity constant. Since the 2013 Guidelines we have received evidence that the model has no theoretical basis in Australia and is not an appropriate tool for regulatory use, nor is it used by market practitioners.⁸¹⁸

⁸¹² AER, Explanatory statement: Rate of return guideline (appendices), December 2013, pp. 24–27.

⁸¹³ ENA, *Response to AER Rate of Return Discussion Papers and Expert Sessions*, 04 May 2018, pp.77-79; Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, p.4.

⁸¹⁴ AER, Second Concurrent Evidence Session, 5 April 2018, p. 69.

⁸¹⁵ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.103-112.

AER, Second Concurrent Evidence Session, 5 April 2018, p. 69.

⁸¹⁷ AER, Second Concurrent Evidence Session, 5 April 2018, p. 72.

⁸¹⁸ Rankin and Idil, A century of Stock-Bond Correlations, September 2014, Partington and Satchell, Cost of Equity issues 2016 Electricity and Gas Determinations, April 2016, pp30-31; Partington and Satchell, Report to the AER, May 2018, p.34-35, AER, Draft decision - Multinet Gas access arrangement 2018-22, Attachment 3 - Rate of return, p.220. Our analysis of independent valuation reports for the 2018 rate of return guideline review also indicated no reports appeared to use the Wright CAPM.

As discussed above, we consider there is neither strong theoretical reasons, nor strong empirical evidence, to support an ongoing and consistent inverse relationship between the MRP and the risk free rate. We also note the evidence since 2013 has increased our concerns about relying on the Wright approach.

7.4 Summary of submissions

The table below contains responses to a discussion paper in March 2018 on market risk premium to provide background on the matters to be discussed at the concurrent evidence session, alongside questions to frame that discussion.⁸¹⁹

Key point	Submission	Stakeholder	AER response
		We disagree.	
The Construction of MRP The Construction The Construction of MRP The MRP is not an exogenous variable and should be formed as the difference between expected total return and the prevailing risk free rate	The MRP is not an exogenous variable and should be formed as the difference between expected total return and	APA ⁸²⁰	We consider that our approach for estimating the MRP is consistent with common practice and has been previously been acknowledged by APA and APTPPL as used by practitioners and in financial text books and literature. ⁸²¹ Partington and Satchell have also previously advised that it is common market practice to 'treat the MRP as the exogenous variable' ⁸²² to the Sharpe-Lintner CAPM.
		We continue to note APA's proposal suggests an inverse and perfectly offsetting relationship between the MRP and risk free rate which we disagree with. APA's proposal also bears resemblance to the Wright approach. We have regard to this material and concluded that its weaknesses limit its use in estimating the MRP.	
	Estimating MRP directly is likely to overstate the risks faced by a natural monopoly firm	CANEGROWERS, The Australia Institute, PIAC ⁸²³	Estimating the realised MRP directly remains appropriate to the SLCAPM and removes as much subjectivity from the regulatory process as possible.
The Dividend Growth Model	The DGM is a useful estimation tool and should be used by the AER in its MRP estimation process	APA, ATCO, APGA, Cheung Kong, ENA, Ergon Energy and Energex, QTC,	Conceptually, the DGM has some merit as a theoretical model to estimate the MRP but issues surrounding inputs, potential biases and sensitivities have limited its use by us.

Table 29 Summary of submissions on market risk premium

AER, APA VTS final decision, November 2017, p. 78; APA VTS, Access arrangement revision proposal submission, 14 August 2017, p. 75; APTPPL, Access arrangement submission, 14 August 2017, p. 68.

⁸²³ CANEGROWERS, *Letter to AER re:RoR Review*, 19 December 2017, p.4; Australia Institute, *Submission on rate of return issues paper*, 12 December 2017, p. 6; PIAC; *PICA letter to the AER*, December 2017, p.2.

⁸¹⁹ AER, MRP Risk Free Rate Averaging Period and Automatic Application Discussion Paper, March 2018.

⁸²⁰ APA, *APA submission responding to AER issues paper*, 12 December 2017, pp. 7-9.

Partington and Satchell, Report to the AER: Discussion of estimates of the return on equity, April 2017, pp. 17, 34.

	Spark Infrastructure ⁸²⁴	
	APGA, NSG, APA, ENA, Joint Energy Networks ⁸²⁵	
	(Frontier) ⁶²⁶	
The DGM should not be used by the AER in MRP estimation	ECA, Energy Users Association of Australia, MEU, CCP ⁸²⁷ CCP ⁸²⁸	We do not propose to use the DGM to directly inform the MRP estimate in this guideline review. Whilst it has potential to be a useful tool, the range of results that 'suitable' models can produce does not give confidence in selecting a precise estimate looking forward.
The DGM should only use growth rates formed from GDP growth.	APGA, ENA ⁸²⁹	Whilst we consider other growth rates may be applicable we will not be changing them in this review, keeping the previous growth rates which were based on GDP growth rates.
There is evidence to suggest appropriate DGM growth rates are falling	CRG ⁸³⁰	We agree that the range of appropriate DGM growth rates may be falling and have considered issues with growth forecasts in exercising our judgement.
The DGM's overall return on equity estimate has been more stable than	ENA ⁸³¹ (Frontier) ⁸³²	We have maintained since the 2013 Guidelines that the DGM's estimate of the overall RoE is overly stable and not

- ⁸²⁴ APA, APA submission responding to AER issues paper, 12 December 2017, pp.11-12; ATCO Gas Australia, Response to Revie of Rate of Return Guideline – Issues Paper, 12 December 2017, p.8; APGA, Submission to the Issues Paper, 12 December 2017, pp. 9-10; Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December 2017, p.4; ENA, Response to AER Issues Paper, 12 December 2017, p.30; Ergon Energy and Energex, Ergon Energy and Energex submission on AER Issues Paper, 12 December 2017, p.6; Queensland Treasury Corporation, QTC submission to the RoR Guideline Review Issues Paper, December 2017, p.5.
- APGA, Submission on AER Review of ROR Guideline, 04 May 2018, pp.12-13; Network Shareholder Group, NSG Submission to the RORG review, 04 May 2018, pp. 15-17; APA, Submission on Rate of Return Review, 04 May 2018, pp.28-32; ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.67-71; AGIG, CitiPower, Multinet, Powercor, SA Power Networks, United Energy, Submission to Rate of Return Guideline Review, 4 May 2018, p.6.
- ⁸²⁶ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.
- ⁸²⁷ CCP 16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p.98; ECA, Response to the AER Issues Paper, December 2017, p.22; EUAA, AER Rate of Return Issues Paper October, December 2017, p.9; Major Energy Users, Submission by the MEU to the review of the rate of return guideline, 18 December 2017, p.16.
- ⁸²⁸ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, p. 115.
- APGA, Submission on AER Review of ROR Guideline, 04 May 2018, pp.12-13; ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.73-75.
- ⁸³⁰ CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.56.
- ⁸³¹ ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.67-71.
- ⁸³² Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.

	the AER's allowed RoE and should be considered appropriate		applicable for use in a regulatory framework. We do not change this position.
	The DGM's MRP estimates are too volatile for use in a regulatory context and do not line up with other market data	CCP ⁸³³	Our analysis shows that DGM-based MRP estimates move inversely to changes in the risk free rate, with the risk free rate accounting for almost 80 per cent of the changes in some cases.
	The Variable Growth Rate DGM has merit, and should be investigated further as to its regulatory uses	CCP ⁸³⁴	We will continue to investigate and monitor the reasonableness of the variable growth rate DGM.
Fixing the MRP	Fixing the Return on Equity should be the aim rather than fixing the MRP	ATCO, ENA ⁸³⁵	We disagree and note that sufficient evidence has not been provided by ATCO and ENA to persuade us that the return on equity is stable over time.
	The MRP could be fixed for the duration of the guideline as long as method is clear, explicit and replicable	APGA, CCP, EUAA, MEU, Ergon Energy, Jemena, NSG ⁸³⁶ EvoEnergy ⁸³⁷	With a binding guideline now in (draft) legislation we consider a fixed MRP/ERP would be the best way to achieve the legislative objectives. We have consistently and transparently set out and explained our approach for estimating the MRP.
	The MRP should be formulaically applied throughout the guideline, but not fixed	QTC, ENA ⁸³⁸ EvoEnergy, Jemena ⁸³⁹ Frontier ⁸⁴⁰	Our position is to apply a fixed MRP over the life of the guideline, as discussed in section 3.2.
	A fixed MRP should be subject to strict re-	APGA, Ergon Energy and	Draft legislation on the binding guideline does not allow for re-openers.

⁸³³ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.109-111.

⁸³⁴ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.111-112.

⁸³⁵ ENA, *Response to AER Issues Paper*, 12 December 2017, p.23; ATCO Gas Australia, *Response to Revie of Rate of Return Guideline – Issues Paper*, 12 December 2017, p.7.

APGA, Submission to the Issues Paper, 12 December 2017, p. 8; CCP 16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp.81-89; EUAA, AER Rate of Return Issues Paper October, December 2017, p.9; Ergon Energy and Energex, Ergon Energy and Energex submission on AER Issues Paper, 12 December 2017, p.5; Jemena, Submission on Rate of Return issues paper, December 2017, pp.3-4; Network Shareholder Group, Submission on rate of return issues paper, 12 December 2017, p.9.

⁸³⁷ EvoEnergy, Submission on Rate of Return Guideline Evidence Session, 04 May 2018, p.4.

⁸³⁸ Queensland Treasury Corporation, QTC submission to the RoR Guideline Review Issues Paper, December 2017, p4; ENA, Response to AER Issues Paper, 12 December 2017, p.30.

⁸³⁹ EvoEnergy, Submission on Rate of Return Guideline Evidence Session, 04 May 2018, p.4; Jemena, Submission on concurrent expert sessions and discussion paper, 4 May 2018, p.3.

⁸⁴⁰ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.

	openers throughout the guideline period	Energex, ENA, NSG ⁸⁴¹	
		NSG ⁸⁴²	
	The AER should consider not having re-openers in the guideline, or limit their potential to re-open automatically	CCP, MEU ⁸⁴³	Draft legislation on the binding guideline does not allow for re-openers.
	The AER's previous approach of setting a fixed MRP has resulted in return on equity estimates inconsistent with market evidence.	EvoEnergy ⁸⁴⁴	We note that our regulatory decisions made since the 2013 Guidelines do not apply a fixed MRP. Rather, at each regulatory determination we considered whether we should depart from our 2013 Guidelines to achieve the allowed rate of return objective.
	Applying a formula to MRP estimation is not appropriate as it could be taken out of context by volatile estimates	CCP ⁸⁴⁵	We do not apply a formula. See section 3.2.
	Of all the WACC parameters MRP should be the slowest to change, so fixing the MRP is in line with expectations	CCP ⁸⁴⁶	Our position is to apply a fixed MRP over the life of the guideline which is discussed in section 3.2.
Historical Excess Returns	Geometric Averages are not appropriate in the MRP estimation process	ENA ⁸⁴⁷ APGA, APA, ENA, Jemena ⁸⁴⁸ (Frontier) ⁸⁴⁹	We consider that geometric averages can offer useful information for estimating the MRP.
	Geometric averages	Ian McAuley ⁸⁵⁰	We consider both geometric and arithmetic

- ⁸⁴¹ APGA, Submission to the Issues Paper, 12 December 2017, p. 8; ENA, Response to AER Issues Paper, 12 December 2017, p.30; Ergon Energy and Energex, Ergon Energy and Energex submission on AER Issues Paper, 12 December 2017, p.5; Network Shareholder Group, Submission on rate of return issues paper, 12 December 2017, p.9.
- ⁸⁴² Network Shareholder Group, *NSG Submission to the RORG review*, 04 May 2018, pp. 15-17.
- ⁸⁴³ Major Energy Users, Submission by the MEU to the review of the rate of return guideline, 18 December 2017, p.16; CCP 16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp.81-89.
- ⁸⁴⁴ EvoEnergy, Submission on Rate of Return Guideline Evidence Session, 04 May 2018, p.4
- ⁸⁴⁵ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.116.
- ⁸⁴⁶ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, p. 95.
- ⁸⁴⁷ ENA, *Response to AER Issues Paper*, 12 December 2017, p.27.
- ⁸⁴⁸ APGA, Submission on AER Review of ROR Guideline, 04 May 2018, pp. 11-12; APA, Submission on Rate of Return Review, 04 May 2018, pp.28-32; ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.79-80; Jemena, Submission on concurrent expert sessions and discussion paper, 4 May 2018, p.3.
- ⁸⁴⁹ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.

	should be given more weight in the MRP estimation process	CRG, CCP ⁸⁵¹	averages and take into account their strengths and weaknesses in exercising our judgement.
	Only single year arithmetic returns from periods of over 50 years should be used	ENA ⁸⁵² (Frontier) ⁸⁵³	We accept that standard errors are improved by using a longer data set. However, as financial conditions change through time, as do investors required risk premiums, we consider that estimates using more recent data are likely to be less biased relative to the true forward looking MRP there is likely to be useful information contained in estimates from more recent periods. Therefore, we consider that we should continue to observe estimates from a range of time periods including more recent periods.
The Wright Approach	The Wright Approach should be given more weight	CKI, APA, ENA ⁸⁵⁴ ENA, Joint Energy Networks ⁸⁵⁵ (Frontier) ⁸⁵⁶	Given there is neither strong theoretical reasons, nor strong empirical evidence, to support an ongoing and consistent inverse relationship between the MRP and the risk free rate we continue to be of the view that the Wright approach should not inform our MRP estimate.
	The Wright Approach should be given no weight	CCP ⁸⁵⁷	We do not give the Wright approach any weight in estimating the MRP.
MRP selection	The AER was overly cautious in 2013, selecting higher parameter estimates than advised leading to a higher ROR than	CCP, ECA ⁸⁵⁸ Energy Australia, CCP ⁸⁵⁹	We have set out to select the best estimate of each parameter at the current time in an unbiased manner, given the regulatory aim of estimating a 10 year rate of return.

- ⁸⁵⁰ Ian McAuley, Submission to Australian Energy Regulator on Rate of Return Guidelines, December 2017, p.4; CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.103-107.
- ⁸⁵¹ CRG, Submission to the AER rate of return guideline review, 04 May 2018, pp.53-54.
- ⁸⁵² ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.80-82.
- ⁸⁵³ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.
- ⁸⁵⁴ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, p.4; APA, *APA submission responding to AER issues paper*, 12 December 2017, pp. 7-9; ENA, *Response to AER Issues Paper*, 12 December 2017, p.30.
- ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.77-79; AGIG, CitiPower, Multinet, Powercor, SA Power Networks, United Energy, Submission to Rate of Return Guideline Review, 4 May 2018, p.6.
- ⁸⁵⁶ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.
- ⁸⁵⁷ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.103-112.
- ⁸⁵⁸ CCP 16, Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p.81; ECA, Response to the AER Issues Paper, December 2017, p.24.
- ⁸⁵⁹ Energy Australia, Submission to AER Rate of Return Evidence Sessions, 04 May 2018, p.2; CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.103-107.

	necessary		
	An MRP of 7.0 per cent is more appropriate than the 6.5 per cent set previously which does not account for increased risk and forward looking estimates	Network Shareholder Group, ENA, Joint Energy Networks ⁸⁶⁰ (Frontier) ⁸⁶¹	For the reasons discussed in this chapter, we do not agree.
	The MRP selected by the AER in the CAPM should be discounted to account for the lack of risks faced by the networks compared to other market based companies	CRG ⁸⁶²	We adopt the SLCAPM as the foundation model for estimating the return on equity. In this model, systematic/compensable risks are reflected in the equity beta parameter.
	The MRP could be set as low as 3.6 per cent based on the current set of data and risks facing firms	CRG ⁸⁶³	We consider that the current data would not support an MRP of 3.6 per cent.
	Trends in historic data suggest globalisation has reduced market risk in Australia	CRG, CCP ⁸⁶⁴	We have considered evidence that suggests a downward trend in realised MRP.
Survey Evidence	Survey Evidence highlights market expectation and is a useful source of evidence	Energy Australia ⁸⁶⁵	Expert advice and discussion in the expert session suggests that whilst its forecasting ability is limited, survey data is useful for measuring expectations. Survey evidence continues to inform our MRP estimate.
	Evidence from surveys is extremely limited and should not be considered in a regulatory context, at least without extreme caution	APGA, APA ⁸⁶⁶ (Frontier) ⁸⁶⁷	We recognise that surveys have limitations and are not at a level of reliability as to give it weight as a direct estimation method of the MRP.
Other Regulator	Other regulators have	NSG, ENA ⁸⁶⁸	We propose to continue looking at other

- ⁸⁶² CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.56.
- ⁸⁶³ CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.57.

⁸⁶⁵ Energy Australia, Submission to AER Rate of Return Evidence Sessions, 04 May 2018, p.2.

⁸⁶⁰ Network Shareholder Group, NSG Submission to the RORG review, 04 May 2018, pp. 15-17; AGIG, CitiPower, Multinet, Powercor, SA Power Networks, United Energy, Submission to Rate of Return Guideline Review, 4 May 2018, p.6; ENA, Response to AER Rate of Return Discussion Papers and Expert Sessions, 04 May 2018, pp.65-66.

⁸⁶¹ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.

⁸⁶⁴ CRG, Submission to the AER rate of return guideline review, 04 May 2018, p.53; CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp. 99-102.

⁸⁶⁶ APGA, Submission on AER Review of ROR Guideline, 04 May 2018, p. 12; APA, Submission on Rate of Return Review, 04 May 2018, pp.28-32.

⁸⁶⁷ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.

Evidence	increased their MRP estimates since the 2013 Guidelines	(Frontier) ⁸⁶⁹	regulators' decisions and give these appropriate consideration.
	International regulator evidence shows the AER is not in line with other jurisdictions	Joint Energy Networks ⁸⁷⁰	We consider that international regulators should not be included when considering other regulators' decisions. Their rate of return decisions are unlikely to be comparable due to a number of differences. For example, they are for firms regulated under different regulatory frameworks.
Market Evidence	Debt Risk Premiums are an obvious cross check and indicate the risk in the equity market has decreased	CCP ⁸⁷¹	We continue look at credit spreads as one of the conditioning variables that inform our estimate of the MRP.
	Market evidence should be used by the AER to consider whether any change in MRP fits with market conditions	CCP ⁸⁷²	We continue to have regard to conditioning variables as directional indicators for the MRP. This allows us to incorporate information from market conditions when estimating the MRP estimate.

⁸⁶⁸ Network Shareholder Group, *NSG Submission to the RORG review*, 04 May 2018, pp. 15-17; ENA, *Response to AER Rate of Return Discussion Papers and Expert Sessions*, 04 May 2018, pp.65-66.

⁸⁶⁹ Frontier submitted a report as part of AusGrid's reset. This is not part of the guideline process but is still considered.

⁸⁷⁰ AGIG, CitiPower, Multinet, Powercor, SA Power Networks, United Energy, Submission to Rate of Return Guideline Review, 4 May 2018, p.5.

⁸⁷¹ CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, p. 95.

⁸⁷² CCP16, Submission to the AER Rate of Return Guideline Review concurrent evidence sessions, 4 May 2018, pp.112-113.

8 Equity beta

The equity beta is a key parameter within the Sharpe–Lintner CAPM. It measures the 'riskiness' of a firm's returns compared with that of the market. Specifically, the equity beta measures the standardised correlation between the returns on an individual risky asset or firm with that of the overall market.⁸⁷³

Generally, investors are assumed to be able to diversify away non-systematic (or business-specific risk). Therefore investors do not require compensation for business specific risk.⁸⁷⁴ Compensation is only required for bearing systematic risk. Sources of systematic risk include changes in real GDP growth, inflation, currency, prices, commodity prices and real long term interest rates. A firm's sensitivity or exposure to these risks will depend on its business activities and its level of financial leverage.⁸⁷⁵

The relevant 'risk' is a similar degree of risk to that faced by a firm in the provision of its regulated energy network services.⁸⁷⁶ Accordingly, we use the data from such firms that are relevant to estimates of beta for our regulatory task.

8.1 Draft decision

Based on the balance of evidence, we have selected a point estimate of 0.6 which is towards the middle of our range of 0.4–0.8 and reflects the information available.

We have maintained the overall approach to estimating the equity beta parameter. That is, we give most weight to empirical estimates of relevant Australian energy network businesses and less weight to other relevant evidence. This is supported by a number of submissions.⁸⁷⁷

We have come to this decision by applying the estimation approach that we set out in our 2013 Guidelines and applied in each subsequent regulatory determination. This approach is based on considering the same relevant classes of evidence that were identified in our 2013 Guidelines, namely:

⁸⁷³ R. Brealey, S. Myers, G. Partington and D. Robinson, *Principles of corporate finance*, McGraw–Hill: First Australian edition, 2000, pp. 186–188 (Brealey et al, *Principles of corporate finance*, 2000).

⁸⁷⁴ G. Pierson, R. Brown, S. Easton and P. Howard, *Business Finance*, 8th Edition, p. 214.

⁸⁷⁵ M. McKenzie and G. Partington, Report to the AER: Estimation of the equity beta (conceptual and econometric issues) for a gas regulatory process in 2012, 3 April 2012, p. 5 (McKenzie and Partington, Estimation of equity beta, April 2012). This report is available on the AER website at: http://www.aer.gov.au/sites/default/files/RBP%20gas%20transmission%202012%20-20Equity%20Beta%20report%20-%20McKenzie%20and%20Partington%20(Public)%20-%203%20April%202012_0.pdf

⁸⁷⁶ NER 6.5.2(c), 6A.6.2© and NGR 87(3)

⁸⁷⁷ Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, pp. 3–4; Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 89; Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68; Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 39.

- Giving most weight to empirical estimates of relevant Australian energy network businesses
- Having regard to:
 - conceptual considerations of the risks of energy network businesses relative to the market portfolio
 - o empirical estimates of foreign energy network businesses
 - o the theoretical underpinnings of the Black CAPM
 - o the value of stability and predictability to industry and consumers

We received divergent submissions on the merits of each type of evidence. For example, networks proposed giving (more) weight to international estimates while consumer groups opposed this proposal.

We have reviewed the relevant evidence for their strengths, weaknesses and suitability for our regulatory task in light of these submissions.

We continue to give most weight to empirical estimates because they provide information from firms that are reasonably comparable to firms in the supply of the regulated energy services. We have updated our empirical analysis⁸⁷⁸ using the same comparator set as that used in our 2013 Guidelines. Our update of the empirical analysis to 2018 supports an empirical range of 0.4–0.8 with clustering in the 0.5–0.6 range. We consider our comparator set of domestic firms is the best empirical guide currently available. This is because international firms and other Australian infrastructure firms carry different risks and characteristics compared to firms in the supply of regulated energy network services.

From a range of 0.4–0.8, we consider that an equity beta value of 0.6 will best promote the legislative objectives for the following key reasons:

- We give most weight to estimates from the longest estimation period because short term estimates can be unduly influenced by factors such as one-off events (for example, the Global Financial Crisis), shocks and interest rate movements. Estimates from the longest estimation period cluster in the 0.5–0.6 range (see Figure 25) and the average⁸⁷⁹ of re-levered OLS estimates is 0.51.
- The average⁸⁸⁰ of weekly re-levered OLS estimates across all periods is 0.57.
- We use conceptual analysis and international estimates as cross checks for our empirical range. The point estimate of 0.6 (and empirical range of 0.4–0.8) is consistent with our expectation that the equity beta of a firm supplying the

⁸⁷⁸ By including data up to 2 March 2018

⁸⁷⁹ Average of fixed weight portfolio estimates and averaged firm-level estimates.

⁸⁸⁰ Average of fixed weight portfolio estimates and averaged firm-level estimates across all three estimation periods (longest, post tech boom excluding GFC and recent 5 years)

regulated energy services will be below 1.0 due to the overall low risk exposure. Our update of international estimates support an equity beta below 1.0.

- We have had regard to (and further consideration of) the Black CAPM and the potential for low beta bias. However, we agree with the CCP16 that there is uncertainty around empirical analysis of the Black CAPM and it is not generally applied by market practitioners or regulators. As such, we do not consider that an uplift beyond the empirical estimates for our comparator set is warranted.
- We observe some increase in estimates since the 2013 Guidelines. However, the overall empirical results, particularly the longest estimation period, support a value of less than 0.7.
- We recognise that the inclusion of some comparator firms with a high proportion of unregulated operations would lead to higher empirical equity beta estimates than otherwise would be. Although this would likely support a value below those from the whole comparator set, we consider that a reduction of 0.1 (from 0.7 to 0.6) is warranted at this time to provide an estimate that is more in line with empirical estimates. We note that we reduced the equity beta from 0.8 to 0.7 in 2013 even though longer term estimates were clustered below those values. In part, we were conscious of the need to promote stability and predictability and therefore decided not to make a larger change.
- We have considered whether we should employ a different value between gas and electricity businesses. Gas businesses submitted that they are subject to greater risk. Overall, we conclude that systematic risks between gas and electricity networks are sufficiently similar to warrant a common equity beta.

Our consideration is set out in section 8.3.

8.2 Stakeholder consultation and submissions

We released an issues paper in October 2017 outlining key areas of the Guideline review process. In response, stakeholders have made submissions which were summarised in our equity beta discussion paper.⁸⁸¹ A summary of the submissions is in our March 2018 Discussion Paper.⁸⁸² We do not repeat them here but they are discussed in section 8.3 as part of our consideration of issues.

8.2.1 Discussion paper and concurrent expert evidence sessions

We released a discussion paper in March 2018 on equity beta to provide background on the matters to be discussed at the concurrent evidence session and questions to frame that discussion.⁸⁸³

⁸⁸¹ AER, Discussion paper equity beta, March 2018

⁸⁸² AER, Discussion paper equity beta, March 2018

⁸⁸³ AER, Discussion paper equity beta, March 2018.

CEPA published a statement of agreed positions with areas of agreement and disagreement from the sessions.⁸⁸⁴ A summary is provided in the Table 30 below:

⁸⁸⁴ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018

Table 30 Areas of agreement and disagreement on beta

Areas of agreement	Areas of disagreement
 Areas of agreement Beta should be assessed from stock market data⁸⁸⁵ Technological risk does not need to be considered separately in estimating equity beta⁸⁸⁶ No simple adjustments can be made to make international data comparable to domestic data⁸⁸⁷ Long periods of data for estimating β are likely to produce the most statistically reliable results. ⁸⁸⁸ 	 Areas of disagreement De/re-levering raw equity beta estimates⁸⁸⁹ Assuming a debt beta of zero⁸⁹⁰ Feedback loop between the AER's decision and equity beta estimates⁸⁹¹ Transparency in the AER's application of the 2013 Guidelines⁸⁹² The appropriateness of the current comparator set⁸⁹³ The inclusion of de-listed firms in the comparator set⁸⁹⁴ Use of non-market data to estimate equity beta⁸⁹⁵ Differential beta for different types of energy businesses⁸⁹⁶ Transparency in the AER's selection of point estimate⁸⁹⁷ If equity beta has changed since 2013⁸⁹⁸ Black CAPM and low beta bias⁸⁹⁹
	Adjustment for low beta blasses Stability ⁹⁰¹

Source: Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018

Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 42

 ⁸⁸⁶ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 43
 ⁸⁸⁷ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 46

⁸⁸⁷ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 46

⁸⁸⁸ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 50

⁸⁸⁹ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 39–40

 ⁸⁹⁰ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 41
 ⁸⁹¹ Cambridge Economic Policy Associates, Expert Joint Poport, 21 April 2018, p. 41

 ⁸⁹¹ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 41
 ⁸⁹² Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 42

 ⁸⁹² Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 42
 ⁸⁹³ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 43

 ⁸⁹⁴ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 47

 ⁸⁹⁵ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 48

⁸⁹⁶ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 49

⁸⁹⁷ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 52

⁸⁹⁸ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 52

⁸⁹⁹ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 52

⁹⁰⁰ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 53

⁹⁰¹ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 54

The expert joint report stated that experts agreed that the weight placed on the estimates should decline in line with the length of the time since delisting.⁹⁰² This is not included in the table above because we consider this was subject to some disagreement. Not all experts were (fully) available over the course of preparing the expert joint report to present their views. The author of the joint statement also confirmed agreed positions may have been taken if no one objected rather than by requiring positive agreement and that assessing the views was not a quantitative voting exercise.⁹⁰³

We have also received submissions on the expert concurrent evidence sessions. The submissions and our considerations are discussed in section 8.3. A summary of submissions is provided in section 8.4.

8.3 Issues and the AER's considerations

We have received divergent submissions on a range of issues from stakeholders as part of this review. In forming a view on our approach, we have taken into account these submissions and further consideration of the relevant evidence. Our consideration of our approach, issues and submissions are in the sections below under the following sections:

- Conceptual analysis
- Empirical studies
- Comparator firms
- International comparators
- Black CAPM and low beta bias
- Empirical methodology
- Industry analysis
- Other information informing range and point estimate
- Range and point estimate

8.3.1 Conceptual analysis

⁹⁰² Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 47.

⁹⁰³ See CEPA, AER RORG Expert Joint Report, 21 April 2018, p. 11: "It should be noted that not all experts were present in all the sessions and may therefore not have given views on all issues. The issues on which experts contributed were set out above in Section 1.1. Graham Partington (GP) was unavailable due to overseas commitments from 14 April 2018 and provided limited input from that date, but did have sight of the final draft. David Johnstone provided input on drafts until 10 April 2018.

The report indicates when most experts held a particular view. However, assessing the views was not a quantitative voting exercise, but a way of identifying alternative views and the reasons for them. Dissenting views of any expert were considered to be of value and may inform the views of the AER."

In the 2013 Guidelines we used conceptual analysis as a cross check of Australian empirical estimates.⁹⁰⁴ This was because it allowed us to form a prior expectation of where the equity beta of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services sits relative to the market average.⁹⁰⁵

Having reviewed all the materials submitted to date, we remain of the view that it is possible to determine a conceptual expectation of the systematic risk of a benchmark efficient entity relative to the market average firm. Conceptual analysis yields insight into where the equity beta for a benchmark efficient entity sits relative to the average equity beta across all firms in the market, which is 1.0 by definition.⁹⁰⁶

However, our conceptual analysis is necessarily qualitative in nature and is therefore used as a cross–check against the empirically derived range.⁹⁰⁷

In section 2.4.3, we consider the assessment of business risk and financial risk for the benchmark efficient entity suggests that the intrinsic business risk of a firm is the main driver of its systematic risk. We expect the benchmark efficient entity to have low intrinsic risk exposure (relative to the market average). We also consider the high financial leverage of the benchmark efficient entity (relative to the market average) does not necessarily correspond to an equivalently high exposure to financial risk.

On the basis of this information, we consider there are reasonable conceptual grounds to expect the overall systematic risk for the benchmark efficient entity to be below that of the market average firm. This leads to our expectation that the equity beta of the benchmark efficient entity will be below 1.0. Our consideration of the comparative systematic risk of gas and electricity service providers in section 2.4.2 leads us to conclude that a single benchmark is still appropriate.

We have considered the risks to service providers arising from technological risk, catastrophic and policy risks in section 2.4.1. Our assessment is that the sort of risks noted by service providers are not systematic and do not warrant compensation in the return on equity.

Both the ENA and Spark Infrastructure submitted that there has been changes in the risks faced by service providers which needs to be accounted for in the regulatory framework.⁹⁰⁸

⁹⁰⁴ AER, *Explanatory statement to the rate of return guideline*, December 2013, pp. 86

⁹⁰⁵ For example, see: AER, Better regulation equity beta issues paper, October 2013, p. 11; AER, SAPN final decision, p. 94, 434.

⁹⁰⁶ AER, Explanatory statement to the rate of return guideline (appendices), December 2013, pp. 39

⁹⁰⁷ AER, Explanatory statement to the rate of return guideline (appendices), December 2013, pp. 43

⁹⁰⁸ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31; Network Shareholder Group, Re: Response to issues paper on the review of the rate of return guideline, 12 December 2017, pp. 9-10

It is not clear to us that there has been material changes in the systematic risk faced by service providers of regulated energy network services. The type of risks noted by service providers and network associations revolve around technology risk, policy risk and natural disasters (as noted in section 2.4.1). We do not consider these types of risk should be compensated through the rate of return as they are non-systematic and investors can diversify them away. Our view is that only systematic risk should be compensated through the rate of return and non-systematic risks should be compensated through the rate of return and non-systematic risks should be compensated through the rate of return and non-systematic risks should be

8.3.2 Empirical studies

Our empirical estimates of equity beta are based on regressions that relate the returns on a set of comparator firms to the return on the market.

When we estimated equity beta in the 2013 Guidelines the main determinant of our estimate was empirical estimates from a comparator set of relevant firms. This comparator set was made up of Australian energy network firms with a similar degree of risk as a service provider in the provision of regulated services. We considered that empirical estimates for this comparator set best met the criteria we set out in the 2013 Guidelines for assessing materials and their relevance/suitability for determining the rate of return.⁹⁰⁹ That is, these empirical estimates are:

- Based on available market data and derived with sound, econometric techniques.
- Fit for purpose as they are based on businesses that most closely, albeit imperfectly, meet our definition of the benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services.
- Implemented in accordance with good practice as they are derived from robust, transparent and replicable regression analysis. We note that consistent results are derived from different studies using different econometric techniques and sampling periods.
- Based on quantitative modelling in that they are derived using regression techniques with no arbitrary adjustment to the data.
- Based on market data that is credible, verifiable, comparable, timely and clearly sourced.
- As a result, we considered empirical studies likely to contribute to a rate of return estimate that achieves the allowed rate of return objective and a source of evidence that should be used as the primary determinant of equity beta.

⁹⁰⁹ AER, Better Regulation Explanatory Statement Rate of Return Guideline, December 2013, pp. 23–26, 83-84,

The 2013 Guidelines used Professor Olan Henry's 2014 study to inform our empirical estimates.⁹¹⁰ This report presented empirical estimates on equity beta for a comparator set of nine Australian energy network firms, using available data from 29 May 1992 to 28 June 2013.⁹¹¹ This report also presented estimates for individual firms as well as various portfolio specifications, and used a range of different estimation methods and time periods. We assessed the equity beta estimates presented in Henry's empirical analysis and concluded that they support a range of 0.4 to 0.7. We also updated Henry's study–using the same methodology–by including data up to 30 April 2017. The results continued to support a range of 0.4-0.7.⁹¹²

We consider that Australian empirical estimates should continue to be the main determinant our equity beta estimate for the following reasons:

- We have observed in regulatory decisions since the 2013 Guidelines that service providers and their consultants have relied on empirical Australian studies to inform the equity beta.⁹¹³
- The ENA has submitted that our standard approach to estimating equity beta from domestic comparators is appropriate.⁹¹⁴ Submissions did not object to the use of empirical Australian estimates.
- Experts have agreed that the overall framework for estimating equity beta is appropriate.⁹¹⁵
- Partington and Satchell have supported using our empirical estimates for informing the equity beta to be applied in a Sharpe-Lintner CAPM for a benchmark efficient entity with a similar degree of risk as a relevant provider of regulated energy network services.⁹¹⁶
- They align with our rate of return criteria which we use to assess materials and their relevance/suitability for determining the rate of return.⁹¹⁷

Updated empirical estimates

We have further updated our 2017 study by including data up to 2 March 2018 for this review.

⁹¹⁰ While Professor Henry's report was published in 2014, estimates were provided to the AER during 2013 to inform the Rate of Return Guideline review. For example, see: AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, pp. 64–67.

⁹¹¹ Henry, *Estimating* β : *An update*, April 2014, p. 9.

⁹¹² In this update, we estimated Ordinary Least Squares (OLS) and Least Absolute Deviation (LAD) estimates of equity beta for our comparator firms just as Professor Henry did in his 2014 study. See AER, staff beta analysis, June 2017.

⁹¹³ For example see: Frontier, An equity beta estimate for Australian energy network businesses, December 2016; CEG, Replication and extension of Henry's beta analysis, 21 September 2016

⁹¹⁴ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46, 50

⁹¹⁵ Cambridge Economic Policy Associates, p. 39

⁹¹⁶ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 15.

⁹¹⁷ AER, Better Regulation Explanatory Statement Rate of Return Guideline, December 2013, pp. 23–26, 83-84,

We consider the most useful empirical estimates:

- use the Ordinary Least Squares (OLS) estimator (with the Least Absolute Deviation (LAD) estimator used as a robustness check for outliers in the underlying data)
- are measured over multiple estimation periods
- use weekly return intervals (with monthly returns used as a robustness check)
- use the Brealey–Myers formula to de- and re-lever raw⁹¹⁸ estimates to a benchmark gearing of 60 per cent, although we consider both raw and re-levered estimates
- are based on averages of individual firm estimates and fixed weight portfolios (equal weighting and value weighting)
- do not apply a Blume or Vasicek adjustment.⁹¹⁹

Table 13 set out updated re-levered OLS equity beta estimates for the individual comparator firms (averaged across firms) and fixed weight portfolios⁹²⁰ respectively. The results show that:

- The re-levered individual firm estimates (averaged across firms) range from 0.57– 0.70.
- The re-levered fixed weight portfolio estimates range from 0.43 to 0.85.

⁹¹⁸ Raw equity beta estimates are those that are observed from the initial regression

⁹¹⁹ Henry does not apply a Blume or Vasicek adjustment of any of his estimates, as specified in our terms of reference.

⁹²⁰ Equally weighted and value weighted portfolios

Table 31 Re-levered weekly equity beta estimates from AER update (OLS, weekly)

		P1	P2	P3	P4	P5	P6	P7
Firms	Avg of firm estimates	APA, ENV	AAN, AGL, APA, ENV, GAS	APA, DUE, ENV, HDF, AST	APA, DUE, ENV, HDF, SKI, AST	APA, DUE, ENV, SKI, AST	APA, DUE, SKI, AST	APA, SKI, AST
Equal weighted								
Longest available period	0.57	0.48	0.50	0.54	0.53	0.43	0.47	0.52
Post tech boom & excl. GFC	0.61	0.52	0.51	0.59	0.58	0.50	0.54	0.63
Recent 5 years	0.70	0.71				0.55	0.66	0.79
Value weighted								
Longest available period	n/a	0.52	0.66	0.47	0.47	0.44	0.49	0.54
Post tech boom & excl. GFC)	n/a	0.56	0.67	0.55	0.55	0.52	0.57	0.66
Recent 5 years	n/a	0.73				0.53	0.72	0.85

Source: AER analysis; Bloomberg

Note: Our comparator firms include AusNet Services (AST). This firm was included in the 2013 Guidelines under its former name of SP Ausnet (SPN). It was renamed in 2014.

Henry's report presented LAD (weekly) estimates as a robustness check for outliers in the underlying data. He also presented OLS estimates using monthly return intervals as a robustness check of the estimates using weekly return intervals. We have updated the LAD and OLS monthly estimates and the results are as follows:

- the re-levered LAD estimates range from 0.37-0.91.
- the OLS estimates using monthly return intervals range from 0.46 to 1.14.⁹²¹

⁹²¹ Henry did not present raw estimates for monthly return intervals. Henry also did not present LAD estimates using monthly return intervals. Henry did present time varying portfolio OLS estimates using monthly return intervals, and these estimates range from 0.39 to 0.47. See: *Henry, Estimating β: An update*, April 2014, p. 58. Henry also suggested that the individual firm estimates based on monthly returns be treated with a degree of caution because some estimates are statistically insignificant. See: *Henry, Estimating β: An update*, April 2014, p. 27.
We continue to recognise that there is generally a trade–off in determining the length of the estimation period.⁹²² On one hand, older data might be considered less reflective of current systematic risk assessments (which would suggest a shorter period). On the other hand, in order to obtain a robust and statistically reliable equity beta estimate we need to have sufficient number of observations (which would suggest a longer period).

We use our empirical analysis to form a range for equity beta which yields a range of 0.4–0.8. We arrive at our range for the following reasons:

• Figure 25 shows that re-levered estimates⁹²³ for the longest estimation period fall in the range 0.4–0.8.

Figure 25 Distribution of 2018 re-levered weekly beta by range (OLS & LAD)



Source: AER analysis, Bloomberg

- All estimates (re-levered weekly portfolio level OLS estimates and averaged firm level estimates) fall within the 0.4–0.8 range except for 1 estimate from Figure 25
- LAD (weekly) estimates are used as a robustness check for outliers in the underlying data. Averaged firm-level estimates (0.51–0.71) are consistent with OLS estimates. Portfolio-level LAD estimates also fall within 0.4–0.8 except for 2 estimates.

⁹²² AER, Better Regulation Explanatory Statement Rate of Return Guideline (Appendices), December 2013, p. 49.

⁹²³ Based on OLS and LAD estimates of portfolios and the average of firm-level estimates

We consider the empirical estimates support a point estimate towards the middle of our range.

- We give most weight to the longest estimation period. As observed in Figure 25, most estimates cluster in the 0.5–0.6 range over the longest period.
- The average of weekly re-levered OLS estimates across all periods is 0.57.

Weekly re-levered OLS estimates cluster in the 0.5–0.6 range as observed from

• Table 31.

We give most weight to the longest estimation period because the benchmark gearing ratio has remained at 60 per cent since the 2009 WACC guideline.⁹²⁴ We also consider that short term estimates can be unduly influenced by factors such as one-off events (for example, the Global Financial Crisis), shocks and interest rate movements (discussed in more detail in section 8.3.9). These factors can obscure the systematic risk of a firm supplying regulated energy services whose exposure is mitigated by regulation and monopoly nature of the service it provides. Submissions from consumers have generally cautioned against reliance on short term estimates of equity beta.⁹²⁵

We note there were divergent views on empirical estimates of the equity beta parameter:

- The ENA considered that the inclusion of de-listed firms does not fully reflect increases in equity beta estimates since the 2013 Guidelines.⁹²⁶
- The CRG was concerned that the inclusion of some comparator firms may overestimate equity beta because they contain substantial unregulated revenue.⁹²⁷

In response, we have estimated equity betas for the still-listed firms and compared them against estimates from the whole comparator set in

⁹²⁴ AER, Final decision Electricity transmission and distribution network service providers review of the weighted average cost of capital (WACC) parameters, May 2009, p. v.

⁹²⁵ Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 96; Ian McAuley, Submission to Australian Energy Regulator on rate of return guidelines, December 2017, p. 3; Canegrowers, Submission to AER review of the rate of return guideline, 19 December 2017, p. 4; Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 83.

⁹²⁶ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 53, 62

⁹²⁷ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 51.

Table 32.

Table 32 Comparison of estimates for entire comparator set to listed comparators (OLS, weekly)

	Whole comparator set	Still listed firms (APA, SKI, AST)	Still listed majority regulated firms (SKI, AST)
Average of firm level estimates			
Longest	0.57	0.50	0.41
Post tech boom & excl. GFC	0.61	0.59	0.52
Recent 5 years	0.70	0.80	0.68
Fixed weight portfolio estimates			
Longest	0.43–0.66	0.52–0.54	0.42–0.43
Post tech boom & excl. GFC	0.50-0.67	0.63–0.66	0.52–0.53
Recent 5 years	0.53–0.85	0.79–0.85	0.68–0.7

Source: AER analysis, Bloomberg

We consider the still listed firm support our empirical range of 0.4–0.8 and a point estimate towards the middle of this range for the following reasons:

- The range of all estimates for the still listed firms (0.5–0.85) is broadly consistent with our range of 0.4–0.8.
- We place most reliance on the longest estimation period. The corresponding estimates of 0.5 (average of firm estimates) and 0.52–0.54 (portfolio estimates) support our earlier view of a value towards the middle of our empirical range.

We agree with consumer submissions that the inclusion of comparator firms with a high proportion of unregulated activities is likely to result in empirical estimates that are higher than otherwise. This is because the higher level of unregulated activities would increase systematic risk exposure and would not be representative of firms in the supply of the regulated energy services. We consider results from

Table 14 support an equity beta of less than 0.7 and indicates potential for a value less than that from the whole comparator set:

- The average of estimates for SKI and AST⁹²⁸ is 0.54 across all periods which is consistent with the 2018 update clustering in the 0.5–0.6 range and supports a point estimate towards the middle of the empirical range.
- For the longest estimation period (which we give most weight to), SKI and AST's average firm level estimate is 0.41 and average portfolio estimate of 0.42. These are below estimates from the whole comparator set of 0.57 and 0.50 respectively.
- The average of portfolio estimates and averaged firm estimates for SKI and AST is 0.54 which is below that from the whole comparator set (0.57).

Given concerns with the number of comparator firms in section 0,⁹²⁹ we have maintained the use of our comparator set. However, we have regard to the above information when selecting a range and point estimate in section 8.3.9.

Comparison to 2013 Guidelines

The 2013 Guidelines considered Henry's empirical analysis supported a range of 0.4–0.7 after considering re-levered OLS equity beta estimates for the individual comparator firms (averaged across firms) and fixed weight portfolios:⁹³⁰

- The re-levered individual firm estimates (averaged across firms) range from 0.46 to 0.56. The corresponding raw (that is, observed market gearing level) estimates range from 0.48 to 0.50.⁹³¹
- The re-levered fixed weight portfolio estimates range from 0.39 to 0.70. The corresponding raw estimates range from 0.42 to 0.58.

We also considered that Henry's results indicate a best empirical estimate of approximately 0.5 for the benchmark efficient entity. This was principally because most of the estimates were clustered around 0.5.⁹³²

⁹²⁸ Simple average of portfolio estimates and averaged firm-level estimates

⁹²⁹ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31; Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, p. 5; Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p. 3; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46, 62.

⁹³⁰ For example, see: AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 473–465.

⁹³¹ The raw equity beta estimates are those that are observed from the initial regression. They have not been delevered and re-levered to a benchmark gearing of 60 per cent. These estimates are not presented but can be found at: Henry, *Estimating* β : *An update*, April 2014, pp. 87–89.

⁹³² For example, see: AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 473–465. Based on all averages of individual firm estimates and fixed weight portfolio estimates presented in Henry's 2014 report (95 estimates in total). This includes OLS and LAD estimates, raw and re-levered estimates, weekly and monthly return intervals and all estimation periods.

The figure below compares weekly re-levered equity beta estimates from the 2018 update with Henry's estimates.⁹³³ We observe that estimates have seen some increase and now cluster in the 0.5–0.6 range.

Figure 26 Comparison of weekly re-levered equity beta estimates from Henry's 2014 report (average of individual firm estimates and fixed weight portfolio estimates)



Source: AER analysis, Bloomberg

The tables below compare weekly re-levered equity beta estimates using the longest estimation period (which we give most weight to) for Henry's results and the 2018 update.

Table 33 Comparison of re-levered weekly average firm equity beta estimates (OLS)

	Average of firm-level estimates	
	Henry	2018 update
Longest period	0.52	0.57
PTEG	0.56	0.61

⁹³³ This is based on re-levered weekly OLS and LAD estimates for the following periods: the longest estimation period, post technological boom and excluding GFC and the most recent 5 years. Henry did not estimate portfoliolevel values for the most recent 5 year period.

5 years	0.46	0.70	
---------	------	------	--

Source: AER analysis, Bloomberg

Note: PTEG is Scenario 2 which is post tech boom excluding GFC

Table 34 Comparison of re-levered weekly portfolio equity beta estimates- longest period (OLS)

	Equal weighted portfolio estimates - Longest period		Value weighted portfolio estimates - Longest period	
	Henry	2018 update	Henry	2018 update
P1	0.46	0.48	0.50	0.52
P2	0.52	0.50	0.70	0.66
P3	0.50	0.54	0.44	0.47
P4	0.48	0.53	0.42	0.47
P5	0.39	0.43	0.39	0.44

Source: AER analysis, Bloomberg

Table 35 Comparison of re-levered weekly portfolio equity beta estimates - PTEG (OLS)

	Equal weighted portfolio estimates - PTEG		Value weighted portfolio estimates - PTEG	
	Henry	2018 update	Henry	2018 update
P1	0.49	0.52	0.54	0.56
P2	0.52	0.51	0.70	0.67
P3	0.55	0.59	0.52	0.55
P4	0.53	0.58	0.50	0.55
P5	0.45	0.50	0.48	0.52

Source: AER analysis, Bloomberg

Note: PTEG is Scenario 2 which is post tech boom excluding GFC

We do observe some increase since 2013:

- The average of firm level estimates have increased since Henry's report with the largest increase for the recent 5 year period (0.46 to 0.70).
- Most portfolio-level estimates rose with the increase being less than 0.05.

However, Table 33 still support an equity beta less than 0.7 as all updated estimates remain below 0.7. Further, estimates from the longest estimation period have shown marginal increases. We give most weight to the longest estimation period because short term estimates may be unduly influenced by one-off events, market volatilities and interest rate movements. See section 8.3.9 for more discussion on how interest rate movements impact empirical beta estimates for our comparator firms.

8.3.3 Comparator firms

To ensure as close a match to the efficient required compensation, we use market data for firms we consider to be reasonably comparable to a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services

During the development of the 2013 Guidelines, nine firms were identified that may be considered as reasonable comparators to a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services. They are ASX listed firms that provide regulated electricity and/or gas network services operating within Australia.⁹³⁴

Firm (ASX ticker)	Time / trading period	Sectors
AGL Energy Limited (AGK)	January 1990 – October 2006	Electricity, Gas
Alinta (AAN)	October 2000 – August 2007	Gas
APA Group (APA)	June 2000 – present	Gas, Minority interest in other energy infrastructure
DUET Group (DUE)	August 2004 – April/May 2017	Electricity, Gas
Envestra Ltd. (ENV)	August 1997 – October 2014	Gas
GasNet (GAS)	December 2001 – November 2006	Gas
Hastings Diversified Utilities Fund (HDF)	December 2004– November 2012	Gas
Spark Infrastructure Group (SKI)	March 2007935 – present	Electricity, Gas
AusNet Services (AST), formerly SP AusNet (SPN)	December 2005 – present	Electricity, Gas

Table 36 Firms in the AER's comparator set

Source: AER analysis

Suitability of the current comparator set

We have received submissions stating that our comparators are not representative of the benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services:

• The CRG stated that some firms operate both regulated and non-regulated activities—the non-regulated component can be substantial.⁹³⁶ It concluded that the

⁹³⁴ For example, see: AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, pp. 64–73; AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463.

⁹³⁵ The SKI data is available from December 2005, but the data prior to March 2007 reflects stapled securities traded as instalment receipts—these instalments requires further leverage adjustment and makes beta estimation difficult.

⁹³⁶ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 26, 45.

beta should not be based on unregulated firms and the number of firms that have some match to the benchmark efficient entity (BEE) is limited to just AusNet Services.⁹³⁷ The MEU also stated that the data is too much compromised by unregulated revenue biasing the measure.⁹³⁸

- The APGA noted that the current sample used by the AER has only one stillexisting firm that represents the risk faced by the gas businesses (APA).⁹³⁹ APA and the ENA also noted that there is only gas firm in the sample.⁹⁴⁰
- The CCP 16 submitted that the weight given to the comparators should be reduced as their congruence with characteristics of the BEE reduces.⁹⁴¹ The QCOSS noted that reliance on Australian regulated energy utility information makes the calculated WACC inherently less reliable.⁹⁴²

As in the 2013 Guidelines, we consider that, ideally, firms that share all or most of the key characteristics of the benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services would be used when conducting our empirical analysis to estimate the equity beta. However, in practice, few firms would fully reflect this. Therefore we use market data for domestic businesses that are considered to be reasonable comparators to the benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services to inform the equity beta.

Having considered the relevant evidence and submissions, our draft decision is to maintain the existing comparator set for the following reasons:

- The existing comparator firms reflect information from firms that are most comparable to a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services we regulate. This has agreement from the expert concurrent evidence session.⁹⁴³
- International energy network estimates and other Australian infrastructure firms
 possess a range of differences to a service provider in the provision of regulated
 energy services. We are not persuaded that they should be included in our
 comparator set or used to inform a point estimate within our range.

⁹³⁷ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 45.

⁹³⁸ Major Energy Users Inc, Review of the rate of return guidelines issues paper submission by the Major Energy User Inc, December 2017, p. 12.

⁹³⁹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 4.

⁹⁴⁰ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 18, ENA, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9.

⁹⁴¹ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.

⁹⁴² Queensland Council of Social Service, Submission on review of Rate of Return Guideline, May 2018, p. 18.

⁹⁴³ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 23, 24, 28

- De-listed firms still carry useful and (historically) reliable information. They provide information on the systematic risk of firms that are most comparable to the firms we regulate. Experts also agreed that they should be included in the comparator set.⁹⁴⁴
- Experts have noted that systematic risk and equity beta (for firms in the provision of regulated energy networks services) are relatively stable and change slowly.⁹⁴⁵We consider this provides additional support for the relevance and inclusion of de-listed firms in the comparator set.
- As observed in Table 36, our comparator set contains firms that provide gas and/or electricity services.
- The expert joint report stated that experts agreed that the weight to place on the estimates should decline in line with the length of the time since delisting.⁹⁴⁶

As noted in section 8.2.1, we consider this was subject to some disagreement. James Hancock considered that 'the case for reducing [de-listed firms'] weight with time since listing has not been made' and better alternatives have not been identified.⁹⁴⁷ Further, not all experts were (fully) available over the course of preparing the expert joint report to present their views. The author of the joint statement also confirmed agreed positions may have been taken if no one objected rather than by requiring positive agreement and that assessing the views was not a quantitative voting exercise.⁹⁴⁸

Expanding the comparator set

We received divergent submissions on the topic of expanding the comparator set:

 Service providers stated that the current comparator set should be expanded to include international energy firms and/or other Australian infrastructure firms because the current sample is too small.⁹⁴⁹ The ENA noted that as the sample of

quantitative voting exercise, but a way of identifying alternative views and the reasons for them. Dissenting views of any expert were considered to be of value and may inform the views of the AER."

⁹⁴⁴ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 47

⁹⁴⁵ Cambridge Economic Policy Associates, Expert Joint Report, April 2018, p. 51.

⁹⁴⁶ Cambridge Economic Policy Associates, Expert Joint Report, April 2018, p. 47.

⁹⁴⁷ Cambridge Economic Policy Associates, Expert Joint Report, April 2018, p. 47.

⁹⁴⁸ See CEPA, AER RORG Expert Joint Report, 21 April 2018, p. 11: "It should be noted that not all experts were present in all the sessions and may therefore not have given views on all issues. The issues on which experts contributed were set out above in Section 1.1. Graham Partington (GP) was unavailable due to overseas commitments from 14 April 2018 and provided limited input from that date, but did have sight of the final draft. David Johnstone provided input on drafts until 10 April 2018. The report indicates when most experts held a particular view. However, assessing the views was not a

⁹⁴⁹ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31; Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, p. 5; Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p. 3; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46, 62.

close domestic comparator reduces, relatively more weight must be given to the other relevant evidence.⁹⁵⁰

- Stephen Gray supported expanding the comparator set to include overseas energy networks and Australian infrastructure firms due to a reduction in the comparator firms.⁹⁵¹
- The CCP16 opposed expanding the comparator set. They observed that international firms are very poor comparators and the considerable difficulties to adjust additional firms to improve comparability with the benchmark efficient entity.⁹⁵²

The expert joint report noted that the range of comparators may be extended but care must be taken to ensure inferences drawn are appropriate.⁹⁵³ This is consistent with our view that to include additional firms in the comparator set, we must first be satisfied that they bear a sufficiently similar degree of risk as a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services after a careful assessment of their risks, operations, regulatory framework, etc.

This is supported by the Consumer Challenger Panel sub-panel 16 (CCP16) which noted that it is important to investigate the characteristics of any comparator firms as well as the changes in the structures of the existing network firms.⁹⁵⁴

Our view is that the set of comparator firms need to reflect information from firms that are most comparable to a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services. The most relevant data for estimating equity beta comes from domestic energy network firms.

This has agreement from the ENA and experts:

- Stephen Gray acknowledged that the remaining domestic firms are most relevant⁹⁵⁵
- Ilan Sadeh noted that domestic energy network firms are the most comparable.⁹⁵⁶
- The ENA observed that the best available evidence is from domestic NSPs and that this evidence should receive material weight.⁹⁵⁷

⁹⁵⁰ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

⁹⁵¹ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 23.

⁹⁵² Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 82

⁹⁵³ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 44.

⁹⁵⁴ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

⁹⁵⁵ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 28

⁹⁵⁶ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 23

 This is supported by the ECA which noted that it 'remains unclear how when the benchmark efficient entity is an Australian entity international estimates would have any relevance'.⁹⁵⁸ The CCP16 also cautioned the inclusion of international estimates and other Australian infrastructure firms if the new data is not particularly relevant to the benchmark efficient entity.⁹⁵⁹

Stephen Gray and Simon Wheatley submitted that expanding the comparator set to include international energy firms and other Australian infrastructure firms would improve statistical reliability.⁹⁶⁰ QCOSS submitted that the problem is that the equity beta is set based on an unavoidably narrow comparator group and can easily be distorted by a few data points.⁹⁶¹

We consider that, as noted by APA, it is necessary to weigh up the potential statistical improvement from expanding the comparator set against the suitability of the additional firms.⁹⁶²

We consider that a small set of comparators does not necessarily justify expanding the comparator set in itself. If the additional firms do not carry a similar degree of risk or cannot be appropriately adjusted to be comparable to a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services then they can bias estimates.

Experts and submissions have also noted that a small sample for firms does not necessarily require expanding the comparator set:

- Partington and Satchell's previous advice indicate that it would be preferable to use a small sample of comparable firms instead of a larger sample of firms with different risks to that of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services.⁹⁶³
- The NSG noted that a 'narrow set of firms is of itself an [insufficient] rationale to include additional comparators'.⁹⁶⁴
- Graham Partington and Ilan Sadeh disagreed with expanding the comparator set.⁹⁶⁵

⁹⁵⁷ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62.

⁹⁵⁸ Energy Consumers Australia, Review of the rate of return guideline response to the AER Issues Paper, December 2017, p. 23.

⁹⁵⁹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70.

⁹⁶⁰ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 44.

⁹⁶¹ Queensland Council of Social Service, Submission on review of Rate of Return Guideline, May 2018, p. 18

⁹⁶² Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62.

⁹⁶³ Partington and Satchell, Report to the AER: discussion of comparator firms for estimating beta, June 2016, p. 9.

⁹⁶⁴ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.

Further, we observe in the section below that international energy and other Australian infrastructure firms differ from a supplier of regulated energy services and so do not provide much useful information on the systematic risk (as captured by the equity beta) of firms supplying the regulated energy network services. The CCP16 has also noted that neither the international data nor the data on Australian infrastructure stocks make suitable comparator data due to differences between these stocks and the characteristics of the BEE.⁹⁶⁶

We note James Hancock stated that international and other Australian infrastructure firms should be included only if it can be shown that they provide useful information about the equity of Australian regulated network service providers.⁹⁶⁷ However, we have not received evidence to persuade us that these firms have similar degree of risk as a benchmark efficient entity with a similar degree of risk as a relevant service provider in in the provision of regulated energy network services.

International energy firms

Given the absence of substantively new information to justify the use of international energy firms, we remain of the view that they should not be included in our comparator set for the following reasons:

International energy firms deviate from our view of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services because they do not operate within Australia. Differences in regulation of businesses, the domestic economy, geography, business cycles and a number of different factors are likely to result in differences between equity beta estimates for similar businesses between countries.⁹⁶⁸ It is difficult to assign quantitative impacts to these qualitative factors. Partington and Satchell have also identified a range of difficulties in interpreting different betas from different countries including differing systems of utility regulation, different technologies and operating conditions, different energy market conditions, and differing leverage. Differing levels of leverage suggest fundamental differences in the nature of the businesses and/or the environment they operate in.⁹⁶⁹

⁹⁶⁵ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 44.

⁹⁶⁶ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68

⁹⁶⁷ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 45.

⁹⁶⁸ This is supported by Partington and Satchell. See Partington and Satchell, *Report to the AER: Cost of equity issues 2016 electricity and gas determinations*, April 2016, p. 11. They stated, 'Considerable caution in reaching conclusions about beta needs to be exercised when the comparators are drawn from overseas countries. This is because of differences in industry structure, technology, the nature of competition, the economic environment and regulatory and tax systems'.

⁹⁶⁹ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, 25 May 2018, p. 24

- We discuss equity beta estimates in the context of our foundation model, which is the domestic Sharpe-Lintner CAPM.⁹⁷⁰ This provides a strong rationale for estimating the equity beta using Australian data. If we included international energy firms in our comparator set, it may be more appropriate to use an international or global CAPM.⁹⁷¹
- Equity beta estimates from international comparators are measured with respect to the market portfolio of their home market.⁹⁷² This means the equity beta estimates from international comparators are not a measurement of the firm's systematic risk relative to the Australian domestic market portfolio.⁹⁷³
- They may not have the same structure as Australian energy network firms. For example, a number of US comparator businesses identified by the Competition Economists Group (CEG) are vertically integrated.⁹⁷⁴ They engage in energy generation, wholesale and retail supply of energy, as well as other activities distinct from energy distribution and transmission. Some of the firms even engage in telecommunications, real estate development and manufacturing activities.⁹⁷⁵ These activities are very different from our definition of a benchmark efficient entity, which is a pure play energy network business (operating within Australia) with a similar degree of risk as a provider of regulated energy services. As noted in the Guideline, we consider vertically integrated firms tend to have higher equity beta estimates than pure play energy network firms.⁹⁷⁶
- Partington and Satchell have also advised against the inclusion of international energy firms.⁹⁷⁷

⁹⁷⁰ We implement the Sharpe-Lintner CAPM under the assumption of a domestic market, but with a presence of foreign investors. This allows us to recognise that foreign investors cannot utilise imputation credits. However, the benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services operates in the Australian market by definition, and we estimate the MRP in the context of the Australian market portfolio.

⁹⁷¹ See Handley, *Advice on the return on equity*, October 2014, p. 24; Partington and Satchell, *Report to the AER:* Cost of equity issues 2016 electricity and gas determinations, April 2016, p. 16.

⁹⁷² This is the case unless the equity betas are estimated using an international CAPM framework.

⁹⁷³ This is supported by Handley and Partington and Satchell. See Handley, Advice on the return on equity, October 2014, pp. 23–24; Partington and Satchell, Report to the AER: Cost of equity issues 2016 electricity and gas determinations, April 2016, p. 16. In his May 2015 report, Handley concluded that he does not consider it necessary to change any of the findings in his earlier (2014) report. See: Handley, Advice on the rate of return for the 2015 AER energy network determination for Jemena Gas Networks, 20 May 2015, p. 28.

⁹⁷⁴ CEG describes vertically integrated US energy utility firms as 'common among [its] sample'. See: CEG, *Information on equity beta from US companies*, June 2013, p. 20.

⁹⁷⁵ CEG, Information on equity beta from US companies, June 2013, pp. 47–68.

⁹⁷⁶ In the rate of return guideline, we found the average equity beta of 56 US energy utilities (identified by CEG) was greater than the average equity beta of 18 US utilities identified by ACG as 'almost exclusively electricity and/or gas distribution and transmission businesses'. See: AER, *Explanatory statement to the rate of return guideline (appendices)*, December 2013, pp. 62–63. Also see: ACG, *Beta for regulated electricity transmission and distribution: Report to Energy Network Association, Grid Australia and APIA*, September 2008, p. 18; CEG, *Information on equity beta from US companies*, June 2013; SFG, *Regression-based estimates of risk parameters*, June 2013, p. 19.

⁹⁷⁷ Partington and Satchell, Report to the AER: discussion of comparator firms for estimating beta, June 2016, p. 15.

Stakeholder submissions have also acknowledged the limitations of using international data for informing the equity beta:

- APA acknowledged that the use of international data and other Australian infrastructure firms may increase statistical precision, but at the cost of biased estimates of equity beta.⁹⁷⁸ It noted that there is no simple mathematical adjustment which might be applied to those international data to make them comparable with data from the extant domestic comparators.⁹⁷⁹
- The CCP16 noted that there are considerable difficulties in 'normalising' these firms for better comparability with the benchmark efficient entity.⁹⁸⁰
- The ENA has acknowledged international firms operate in different markets compared to the benchmark efficient entity we regulate.⁹⁸¹

Further, experts have acknowledged difficulties with using international firms to estimate equity beta:

- Experts noted that using international energy firms would be problematic (for example due to different regulatory frameworks and systematic risks)⁹⁸² and foremost weight should be placed on domestic comparators because they are the most relevant.⁹⁸³
- Ilan Sadeh noted that market practitioners use international estimates as a cross check. ⁹⁸⁴
- Experts have agreed that no simple mathematical adjustment exists to allow appropriate consideration of international data.⁹⁸⁵

The above information does not imply that the empirical evidence based on international energy network firms should be discarded completely. Rather, we consider that such evidence may have some use in informing the equity beta point estimate in a cross-checking role which is discussed in section 8.3.4.

⁹⁷⁸ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, pp. 18–19.

⁹⁷⁹ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 19

⁹⁸⁰ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

⁹⁸¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62

⁹⁸² AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 33, 29, 28

⁹⁸³ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 28, 33, 35

⁹⁸⁴ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 28

⁹⁸⁵ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 46.

Other Australian infrastructure firms

We have considered the divergent submissions on including other Australian infrastructure firms in the comparator set. Material submitted to us proposing the use of these firms was generally in the context of the reduction in the existing comparator firms.

We note Stephen Satchell stated that the use of other Australian infrastructure firms depends on whether these companies are fundamentally similar.⁹⁸⁶

Our assessment is that the risk characteristics of these businesses are different to those of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services (for example, due to demand risk, different (or no) regulatory framework, etc.) in the provision of regulated energy network services.

Others have made similar observations:

- The CCP observed Ilan Sadeh noted that domestic infrastructure businesses are very poor comparators to include in the estimation of the equity beta for the BEE, because of their different approaches to debt, other funding arrangements and private ownership (among other things). ⁹⁸⁷
- The NSG observed that domestic infrastructure firms from other sectors are of very limited value due to different regulatory environments and capital requirements. ⁹⁸⁸
- The CCP16 stated that there are considerable difficulties in normalising domestic infrastructure equity betas to achieve better comparability with the BEE. ⁹⁸⁹

Further, we have not been provided with sufficient evidence to persuade us that other Australian infrastructure firms carry a sufficiently similar degree of risk as a benchmark efficient entity. Ilan Sadeh also appears to caution that it would be 'most [dangerous]' to look at other Australian infrastructure firms.⁹⁹⁰

Therefore, we do not consider that other Australian infrastructure firms should be included in our comparator set.

We disagree with APA's statement that domestic infrastructure firms operate in the Australian market in an industry that is close to, but not exactly equivalent to, the provision of energy network services.⁹⁹¹

⁹⁸⁶ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 26

⁹⁸⁷ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 82

⁹⁸⁸ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.

⁹⁸⁹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 81.

⁹⁹⁰ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 33.

⁹⁹¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62

Use of de-listed firms

We note CKI and the ENA have proposed excluding de-listed firms from the comparator set because they do not reflect prevailing market conditions.^{992 993 994}

However, experts agreed that de-listed firms should be included in the comparator set but held divergent views on the weight given to them:

Jim Hancock noted that de-listed firms should not be dismissed because, to the extent the betas changed, they are probably cycling.⁹⁹⁵ He added that we should be hesitant to put less weight on de-listed firms as that increases weight on others which raise the question if there are better alternatives. Further, if estimates cycle up and down then historical can still give a reasonable estimate of long run average⁹⁹⁶

Stephen Satchell advised that to disregard information that at least is historically reliable could only be justified if we could find something better and there does not seem to be better information.⁹⁹⁷

Stephen Gray stated that de-listed firms have their beta frozen in time and averaging across estimates will be misleading if most of the firms are de-listed and the remaining firms suggest a material increase recently⁹⁹⁸

We have considered the proposal to exclude de-listed firms. We recognise that they may not necessarily provide the most up-to-date information about the equity beta. However, they still provide (historically) reliable and accurate information on the systematic risk of a BEE with a similar degree of risk as a relevant service provider in the provision of regulated energy services. Neither could be said with certainty for international energy firms and other Australian infrastructure firms from discussions in the above sections. Any adjustments to these firms also appears problematic due to issues quantifying differences with a supplier of the regulated energy services.

Further, Partington and Satchell continue to support the use of de-listed firms in their latest advice as 'beta has been stable through time and therefore historic estimates of beta, including from companies that are now delisted, can be used to inform current estimates'.⁹⁹⁹

⁹⁹² Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, p. 5.

⁹⁹³ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31.

⁹⁹⁴ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 62.

⁹⁹⁵ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 24

⁹⁹⁶ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 25

⁹⁹⁷ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 25

⁹⁹⁸ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 24–25

⁹⁹⁹ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, 25 May 2018, p. 25; Partington and Satchell, Report to the AER: discussion of comparator firms for estimating beta, June 2016, p. 9..

Experts at the concurrent expert evidence session also agreed that equity beta is relatively stable because the true systematic risk is likely to be stable.¹⁰⁰⁰ We consider this supports the inclusion of de-listed firms which should still provide useful information for informing the equity beta parameter.

8.3.4 International comparators

In the 2013 Guidelines, we considered that international estimates may be used to inform a point estimate within our empirical range. This is because we considered that international comparators are less aligned with a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services compared to Australian comparators.¹⁰⁰¹

Submissions and information from this review process has pushed us to further consider the weight to international comparators:

- Stephen Satchell has advised that 'it is not clear that a cross section of betas in one market is directly comparable with those from another market'.¹⁰⁰²
- Ilan Sadeh noted that assessment would be more qualitative if the evidence is not domestic Australian firms.¹⁰⁰³
- Stephen Gray noted that other regulators' estimates can provide information on how they have made inferences in the context of limited listed comparators¹⁰⁰⁴
- Service providers and networks associations have pushed for more weight to international comparators as noted in section 0.

We remain of the view that it would be difficult to use international comparators in accordance with good practice for estimating the equity beta parameter.¹⁰⁰⁵ The multitude of differences observed in section 0 with a supplier of the regulated energy services also means that we cannot (reliably) quantify and adjust international estimates to make them comparable to domestic estimates which are the most suitable comparators.

If we cannot reliably quantify and adjust for these differences, then it would not be appropriate to use this material inform a point estimate from the empirical range. Therefore, we do not consider it appropriate to retain international comparators in its current role.

¹⁰⁰⁰ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 51.

¹⁰⁰¹ AER, Better Regulation Explanatory Statement Rate of Return Guideline, December 2013, p. 85.

¹⁰⁰² AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 26

¹⁰⁰³ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 33

¹⁰⁰⁴ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 45.

¹⁰⁰⁵ AER, Better Regulation Explanatory Statement Rate of Return Guideline, December 2013, p. 85.

However, we consider international comparators can still provide some information on the systematic risk of a firm. This will necessarily be in a qualitative role, similar to conceptual analysis, because both are unable to provide information on a supplier of the regulated energy services that can be (reliably) quantified.

Empirical international estimates

The most recent empirical study we received on international energy firms was from a 2016 Frontier report.¹⁰⁰⁶ It estimated equity beta for 56 US-listed energy network companies over a 20 year period from December 1995 to December 2015. The sample was originally compiled by CEG in 2013 and was based on firms where at least 50 per cent of the revenue was regulated.¹⁰⁰⁷

We have used the set of firms from the 2016 Frontier report to inform international empirical estimates of equity beta while recognising differences with the firms we regulate in section 0.

The figure below summarises the range of results of our OLS results using quartile estimates.¹⁰⁰⁸ The figure uses a box and whiskers chart to present the minimum, 1st quartile, 3rd quartile and maximum from the results.

We make the following observations:

- Estimates, across all estimation periods, cluster below 1.0.
- Estimates for the longest period (which we give most weight to when considering the empirical range) cluster below 1.0.

¹⁰⁰⁶ Frontier, Estimating the equity beta for the benchmark efficient entity, January 2016.

¹⁰⁰⁷ SFG, Regression-based estimates of risk parameters, June 2013, pp. 15, 19; CEG, Information on equity beta from US companies, June 2013.

¹⁰⁰⁸ Quartiles are the values that divides a list of numbers into quarters. The first quartile is the data point in a data set that separates the bottom 25 per cent of data points from the top 75 per cent. The second quartile is the data point in a data set that separates the data in half. The third quartile is the data point in a data set that separates the bottom 75 per cent of data points from the top 25 per cent.





Source: AER analysis; Bloomberg

Note: This figure shows the quartile distribution of estimates by charting the minimum, first quartile, third quartile and maximum of the relevant estimates. The top of the top line indicate the maximum and bottom of the bottom line indicate the minimum. The bottom of the rectangle represents the first quartile. The top of the rectangle represents the third quartile.

PTEG is Scenario 2 which is post tech boom excluding GFC.

Service providers and network associations have noted the NZCC's estimation of international estimates in 2016. We note there is significant overlap between the comparator sets from SFG and the NZCC. However the NZCC's selection methodology may lead to firms with minority regulated operations being included in its comparator set.¹⁰¹⁰ This is because selection appears to be based on Bloomberg's description of businesses' operations as 'electricity', 'gas distribution', 'pipelines' and 'multiutilities'. Therefore, we use the sample of 56 US energy firms.

¹⁰⁰⁹ This figure shows the quartile distribution of estimates. The top of the top line indicate the maximum and bottom of the bottom line indicate the minimum. The bottom of the rectangle represents the first quartile. The top of the rectangle is represents the third quartile.

¹⁰¹⁰ NZCC, Input methodologies review decisions Topic paper 4: Cost of capital issues, 20 December 2016, p. 63.

8.3.5 Black CAPM and low beta bias

In the 2013 Guidelines and subsequent regulatory decisions, we used the theory of the Black CAPM (to account for potential market imperfections that may cause actual returns to diverge from expected returns) to select a point estimate towards the upper end of our empirical range.

This was because there are a range of limitations with the model, such as empirical instability and implementation,¹⁰¹¹ and while the direction of its effect may be known, the magnitude is much more difficult to ascertain.

However, we acknowledged that it can provide some information in selecting the equity beta point estimate towards the upper end of our empirical range.¹⁰¹² Therefore, the theory of the Black CAPM was used (to account for potential market imperfections) to select a point estimate towards the upper bound of our empirical range in the 2013 Guidelines and subsequent regulatory decisions.¹⁰¹³

Low beta bias

We first note that the Black CAPM and the low beta bias are two different concepts and need to be distinguished:

• The Black CAPM is an alternative model to the Sharpe-Lintner CAPM. The key theoretical difference between the Black CAPM and the Sharpe-Lintner CAPM relates to borrowing and lending assumptions.¹⁰¹⁴ As a result of slightly different starting assumptions, the Black CAPM predicts a slope of estimated returns that can be flatter than for the Sharpe-Lintner CAPM.¹⁰¹⁵

¹⁰¹¹ AER, Better Regulation Explanatory Statement Rate of Return Guideline (appendices), December 2013, pp. 71-72.

¹⁰¹² In the Guideline we performed a rough assessment of the reasonableness of the option to select a point estimate towards the upper end of the equity beta range (to reflect the differing predictions of the Black CAPM relative to the SLCAPM). We noted for clarity that we do not consider the possible zero beta premiums presented in table C.11 of the explanatory statement to the Guideline are accurate or reliable as empirical estimates because we do not consider that there is any reliable empirical estimate for this parameter. However, in light of the available evidence, if the Black CAPM captured the 'true' state of the world better than any other asset pricing model (although we are not implying that it does), selecting a point estimate towards the upper end of the equity beta range appeared open to us. See: AER, *Explanatory statement to the rate of return guideline (appendices)*, December 2013, pp. 70–71.

¹⁰¹³ For example, see: AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, pp. 64–73;

¹⁰¹⁴ The Sharpe-Lintner CAPM assumes that investors can access unlimited borrowing and lending at the risk free rate. The Black CAPM relaxes this assumption, and instead assumes that investors can access unlimited short selling of stocks, with the proceeds immediately available for investment. Either of these assumptions might correctly be criticised as being unrealistic, and it is not clear which assumption is preferable.

¹⁰¹⁵ Fischer Black's 1972 paper on the Black CAPM develops two model specifications. The base specification assumes no risk free asset exists (no risk free borrowing or lending). The second specification assumes that the representative investor can lend but not borrow at the risk free rate. In the base specification, the return on the zero beta portfolio can be above the risk free rate. In the second specification, the return on the zero beta portfolio must be above the risk free rate. See: Black, *Capital market equilibrium with restricted borrowing*, Journal of Business 45(3), July 1972, pp. 452–454.

• The low beta bias is an observation that ex-post returns from low beta stocks tend to outperform expected returns.

Experts also appear to recognise the difference as they do not confuse the low beta bias with the Black CAPM.¹⁰¹⁶

Submissions from network businesses, investors and networks associations propose adjusting the return on equity for the low beta bias.¹⁰¹⁷ ¹⁰¹⁸ ¹⁰¹⁹ ¹⁰²⁰ Some of these submissions supported more (and explicit) adjustments for the low beta bias.¹⁰²¹ The CCP16 did not consider the low beta bias to be particularly suitable for estimating exante the equity beta or for 'adjustment' to the empirical data.¹⁰²² It noted that it is based on ex-post empirical assessment of actual outturns which is not an unbiased estimate of ex-ante expectations

We note submissions in support of adjusting the return on equity for low beta bias do not provide substantively new information from those considered in our 2013 Guidelines and subsequent regulatory decisions. That is, they revolve around observations of the low beta bias on an ex-post basis¹⁰²³ which needs to be factored into the ex-ante rate of return through an adjustment of some sort. ¹⁰²⁴ ¹⁰²⁵ ¹⁰²⁶ Stephen Gray also stated that the adjustment should offset the low beta bias.¹⁰²⁷

¹⁰¹⁶ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 53.

¹⁰¹⁷ Australian Pipeline and Gas Association, Submission to the Issues Paper: AER review of the rate of return guideline, 12 December 2017, p. 9; Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, p. 5; Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31; Ergon Energy and Energex, AER Issues paper review of the rate of return guidelines Ergon Energy and Energex submission, 12 December 2017, p. 6; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 59; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 58.

¹⁰¹⁸ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9.

¹⁰¹⁹ SAPN, Victorian Power Networks, Australian Gas Infrastructure Group, AER Discussion Papers – Review of the Rate of Return Guideline, 4 May 2018, p. 6.

¹⁰²⁰ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.

¹⁰²¹ Evoenergy, Review of rate of return guideline –evidence sessions, 4 May 2018, p. 3; APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9.

¹⁰²² Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹⁰²³ By comparing actual realised returns against expected returns.

¹⁰²⁴ For example, APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.

¹⁰²⁵ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 27.

¹⁰²⁶ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 52.

¹⁰²⁷ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 53.

We acknowledge that ex-post return data can indicate that actual returns exceed expected returns for low beta stocks.

However, given the lack of substantively new material, our considerations for giving no weight to low beta bias in our 2013 Guidelines and subsequent regulatory decisions also remain relevant:

- Many of the tests and exercises which indicate low beta bias are themselves the subject of ongoing academic debate and carry limitations which throw doubt on their results and suitability for our regulatory task¹⁰²⁸
- There are a number of explanations (for example, economic conditions) that do not imply a bias in equity beta.¹⁰²⁹ For example, Partington and Satchell have previously observed that beta for a given portfolio remains remarkably constant which suggest that it may not be bias in beta that explains non-zero alphas, but that it has more to do with economic conditions.¹⁰³⁰

It is also not clear that the low beta bias exists on an ex-ante basis or is accounted for by investors and market practitioners on the same ex-ante basis:

- It cannot be proven or quantified on an ex ante basis.¹⁰³¹ Partington and Satchell have previously advised that a myriad of factors can contribute to actual returns differing from expected returns.¹⁰³²
- Our analysis of broker reports and expert valuation reports shows that very few reports (if any) adjust the rate of return for the low beta bias. This indicates that market practitioners and investors do not appear to account for the low beta bias when estimating the required rate of return.
- Partington and Satchell have advised that the empirical ex-post results does not necessarily imply low beta bias or that the bias should warrant increasing the allowed rate of return.¹⁰³³ This is because an increase in beta estimates may be driven by factors that may not necessarily reflect a change in the systematic risk of supplying the regulated energy services. Partington and Satchell have noted that interest rate movements, over pricing of high beta stocks and a low beta¹⁰³⁴ can all drive increase in estimates.¹⁰³⁵ 1036

¹⁰³⁴ Partington and Satchell observed that a low beta is correlated with high alpha

¹⁰²⁸ AER, 2013 Guidelines appendices, December 2013, pp. 11–12; For example, AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463, p. 288.

¹⁰²⁹ For example, AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463, p. 285

¹⁰³⁰ Partington and Satchell, *Report to the AER: Return of equity and comment on submissions in relation to JGN*, May 2015, p. 16

¹⁰³¹ AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463, p. 288.

¹⁰³² Partington and Satchell, *Report to the AER: Discussion of submissions on the cost of equity*, 8 June 2017, p. 30.

¹⁰³³ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, 25 May 2018, p. 26–28.

¹⁰³⁵ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, 25 May 2018, p. 26–28.

¹⁰³⁶ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 53.

- APA also acknowledged that the observed low beta bias may be a consequence of the model correctly estimating expected returns which are then being compared against – different – realized returns
- Experts and submissions in support of including the low beta bias generally noted the low beta bias is observed in ex-post data, textbooks and academic research.¹⁰³⁷ However, they did not advance evidence that the low beta bias is factored in or that investors and market practitioners account for it on an ex-ante basis.
- Ilan Sadeh stated that it is hard to explain the low beta bias and noted actual returns may include an 'alpha' for expected outperformance on items such as incentive schemes because empirically listed network stocks have benchmarked more efficient than average.¹⁰³⁸ He also noted that the rate of return is for the return on RAB and returns higher than that set by the AER is attributable to the (extra) risk borne on items such as opex allowance.¹⁰³⁹
- The CCP16 noted the low beta bias is based on ex-post empirical assessment of actual outturns which is not an unbiased estimate of ex-ante expectations. ¹⁰⁴⁰

Based on the considerations above, our draft decision is to continue give no weight to the low beta bias. The APA has also acknowledged that estimates of beta are not, themselves, biased.¹⁰⁴¹

We note two submissions appear to state that the AER sought to correct the low beta bias by giving weight to the Black CAPM.¹⁰⁴² ¹⁰⁴³ ¹⁰⁴⁴ We disagree. We gave a role to the theory of the Black CAPM in the 2013 Guidelines to capture possible market imperfections that may lead to actual returns to differ from expected returns. We did not, and do not, give weight to low beta bias.

¹⁰³⁷ For example, see: APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p.9.; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 58.

¹⁰³⁸ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 53.

¹⁰³⁹ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 46.

¹⁰⁴⁰ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70.

¹⁰⁴¹ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 24.

¹⁰⁴² Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 59

¹⁰⁴³ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 25.

¹⁰⁴⁴ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 26.

In its April 2018 report for AusGrid, Frontier submitted that the AER's adjustment does not fully correct the low beta bias.¹⁰⁴⁵ After reviewing this material, we consider that it is substantively the same as a January 2017 Frontier report titled 'low-beta bias'.¹⁰⁴⁶

Both reviewed selected academic papers and previous consultant reports on the Sharpe-Lintner CAPM's low beta bias to conclude that the AER should set a beta higher than 0.7 to 'better' correct this bias. Both also raised issues we have considered in previous regulatory decisions and our (previous) consideration of these issues remain relevant for informing this decision. As a result, we consider our reasons for rejecting the January 2017 report appropriate for informing our rejection of the April 2018 material:

- The reviewed academic papers and consultant reports generally conduct or refer to
 empirical tests of asset model performance to test for the bias.¹⁰⁴⁷ We have
 consistently noted a range of issues with these tests which cast doubt on this
 source of material and its suitability for informing the required return on equity.¹⁰⁴⁸
 For example, results of asset model tests can depend on how the tests are
 designed and has been observed to indicate 'more about the shocks to the
 expected returns (volatility) rather than the equilibrium expected returns'.¹⁰⁴⁹
- Frontier noted that the Black CAPM or an equivalent equity beta adjustment can be used to correct the low beta bias.¹⁰⁵⁰ We note Frontier's equity beta adjustment hinges on output from the Black CAPM.¹⁰⁵¹ We have consistently noted that there are a range of issues with implementing the Black CAPM¹⁰⁵² and there is little evidence that other regulators, academics or market practitioners use the Black CAPM to estimate the return on equity.
- Frontier has mischaracterised our approach for equity beta.¹⁰⁵³ We did not uplift the equity beta to 0.7 (from 0.5) to account for the low beta bias.¹⁰⁵⁴ We selected 0.7 (from a range of 0.4–0.7) to account for the theory of the Black CAPM (to account for potential market imperfections) and other relevant information. ¹⁰⁵⁵

The APGA has submitted extensively on the topic of low beta bias which we respond to in the table below.

¹⁰⁴⁵ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 37–64.

¹⁰⁴⁶ Frontier, *Low beta bias*, January 2017.

¹⁰⁴⁷ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 41–49, 59–63.

¹⁰⁴⁸ AER, *Draft decision Multinet Gas Access Arrangement 2018–2022 Attachment 3–Rate of return*, July 2017, pp. 179–187.

 ¹⁰⁴⁹ AER, Draft decision Multinet Gas Access Arrangement 2018–2022 Attachment 3–Rate of return, July 2017, pp. 180.

¹⁰⁵⁰ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 38.

¹⁰⁵¹ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 51–54, 64.

¹⁰⁵² For example, the zero-beta return is unobservable and there is no apparent consensus on methods for estimating this return. AER, *Draft decision Multinet Gas Access Arrangement 2018–2022 Attachment 3–Rate of return*, July 2017, pp. 188–201.

¹⁰⁵³ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 55–56.

¹⁰⁵⁴ AER, Better Regulation Explanatory Statement Rate of Return Guideline, December 2013, p. 86.

¹⁰⁵⁵ AER, Better Regulation Explanatory Statement Rate of Return Guideline, December 2013, p. 86.

Table 37 Consideration of APGA submission on low beta bias

Submission	Consideration
Experts were in unanimous	We first note that we do not use the theory of the Black CAPM to account for the low beta bias. We did not, and do not, give weight to the low beta bias.
beta bias and 'ignoring' it due to concerns with the empirical reliability of the Black CAPM would not be prudent. ¹⁰⁵⁶	Second, we have outlined a range of concerns with the Black CAPM such as lack of real-world use, empirical implementation, divergent results etc. ¹⁰⁵⁷ We consider that if a model cannot meet our assessment criteria well then we have reservations regarding its use. Further consideration of the Black CAPM in this review has reinforced these concerns.
AER risk unbalanced treatment of evidence if it adopts Stephen Satchell's advice on interest rate movements when it rejected the Fama French Model. ¹⁰⁵⁸	We consider all submissions, materials and evidence on their strengths, weaknesses and suitability for our regulatory task.
Ignoring actual returns entirely because actual returns are different to expectations does not appear to be an adequate response. ¹⁰⁵⁹ It noted that the 'usual' way to address the problem the AER faces in respect of the low beta bias is to use actual returns to adjust models like the CAPM. ¹⁰⁶⁰	We need to consider what is causing actual returns to differ from expectations and if these factors are priced by investors on an ex-ante basis. Partington and Satchell have noted that this could be due to economic shocks, business cycle and other factors in their previous advice. ¹⁰⁶¹ Broker and valuation reports also do not appear to adjust for the low beta bias. Therefore, it is not clear to us that the historical outperformance necessarily warrant adjustment to the expected rate of return.
The AER should consider the empirical results of low beta bias (even if they can be wide ranging) instead of using judgement or theory. ¹⁰⁶² The APGA noted that best approach we have seen in the Australian regulatory context is that undertaken by HoustonKemp for Multinet in its most recent proposal. ¹⁰⁶³	As noted above, it is not clear that the low beta bias exists on an ex-ante basis or is accounted for by investors and market practitioners on the same ex-ante basis. Further, we disagree with the use of ex-post data for a number of reasons including interest rate movements and economic shocks can all lead to ex-post returns diverging from expected returns.
Disregarding the low beta bias would decrease investor returns which may deter investment and would not be in the long run interests of consumers. ¹⁰⁶⁴	As noted above, investors do not appear to factor in the low beta bias when forming their investment decisions.

- ¹⁰⁵⁶ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 8.
- ¹⁰⁵⁷ For example, see AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463, p. 79, 309–315.
- ¹⁰⁵⁸ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹⁰⁵⁹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹⁰⁶⁰ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹⁰⁶¹ Partington and Satchell, Report to the AER: Discussion of submissions on the cost of equity, 8 June 2017, pp. 29– 30.
- ¹⁰⁶² APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹⁰⁶³ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹⁰⁶⁴ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 7.

Submission	Consideration
The Guideline should examine and account for the low beta bias as it has been observed in practice and academic research. ¹⁰⁶⁵	We recognise that there is evidence of ex-post returns of low beta stocks outperforming expected returns. However, we have to consider this in the context of setting a forward looking, ex-ante rate of return that compensates for the efficient financing cost of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services. As noted, investors and market practitioners do not appear to account for the low beta bias on an ex-ante basis. Actual returns can deviate from expected returns for a range of reasons including shocks to the economy. This was also noted by Partington and Satchell in their previous advice. ¹⁰⁶⁶

Black CAPM

We note that we gave a role to the theory of the Black CAPM in the 2013 Guidelines to capture potential imperfections in the market that may cause actual returns to differ from expected returns.¹⁰⁶⁷

There were divergent submissions on the Black CAPM.

Submissions from network businesses, investors and networks associations propose the continued use of the Black CAPM¹⁰⁶⁸. Some of these submissions supported maintaining the weight given to the Black CAPM.¹⁰⁶⁹

Submissions from consumer groups generally opposed having regard to the Black CAPM.¹⁰⁷⁰ The CCP16 did not consider the Black CAPM to be particularly suitable for estimating ex-ante the equity beta or for 'adjustment' to the empirical data.¹⁰⁷¹ It noted that it based on ex-post empirical assessment of actual outturns which is not an unbiased estimate of ex-ante expectations. Moreover, it noted significant variability in the zero beta estimates, a variability that is consistent with the fact that neither the market practitioners nor regulators generally use this estimate.

We have further considered the Black CAPM.

We note submissions have not raised substantively new material to those considered in the 2013 Guidelines and subsequent regulatory decisions. As a result, we continue

¹⁰⁶⁵ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.

¹⁰⁶⁶ Partington and Satchell, Report to the AER: Discussion of submissions on the cost of equity, 8 June 2017, pp. 29– 32.

¹⁰⁶⁷ AER, Better Regulation Explanatory Statement Rate of Return Guideline, December 2013, p. 86.

¹⁰⁶⁸ ATCO gas, Re: Review of rate of return guideline-issues paper, 12 December 2017, p. 8–9; Network Shareholder Group, Re: Response to issues paper on the review of the rate of return guideline, 12 December 2017, pp. 9-10; APA, Review of the rate of return guidelines APA submission responding to AER issues paper, 12 December 2017, p. 11.

¹⁰⁶⁹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9.

¹⁰⁷⁰ Energy Consumers Australia, Review of the rate of return guideline response to the AER Issues Paper, December 2017, p. 23; Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 85, 89;

¹⁰⁷¹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

to note the shortcomings of the Black CAPM identified in the 2013 Guidelines. For example:

- The model is not empirically reliable.¹⁰⁷²
- The model is not widely used to estimate the return on equity by equity investors, academics or regulators.¹⁰⁷³
- The model does not meet our assessment criteria well¹⁰⁷⁴

These shortcomings have been reinforced by material and evidence received since the 2013 Guidelines which raise questions about use of the (theory of the) model for selecting a point estimate towards the top of the observed empirical range:

- Market practitioners, investors and regulators do not make use of the Black CAPM :
 - Our analysis of broker reports and expert valuation reports shows that very few reports (if any) use the Black CAPM. This indicates that market practitioners and investors do not appear to use the Black CAPM when estimating the required rate of return
 - Experts at the expert concurrent evidence session did not provide evidence of the Black CAPM's use in practice.
 - Other Australian regulators do not give consideration to the Black CAPM
- We have less confidence in the information provided by the model as it is empirically unstable, sensitive to the choice inputs and lacks consensus:
 - APA has acknowledged the considerable difficulties associated with obtaining reliable estimates of the return on the zero-beta portfolio. ¹⁰⁷⁵
 - Its assumptions are no more realistic than those of the Sharpe-Lintner CAPM¹⁰⁷⁶
 - The Black CAPM can produce counter-initiative and non-sensical results. For example, we previously noted that it can produce a zero-beta premium greater than the market risk premium and a negative relationship between

¹⁰⁷² AER, Rate of return guidelines, explanatory statement - appendices, December 2013, pp. 68-71.

¹⁰⁷³ AER, Final decision on SA Power Networks' 2015-20 electricity distribution determination, Attachment 3 - Rate of Return, October 2015, pp. 311-312.

¹⁰⁷⁴ In the 2013 Guidelines we considered a set of criteria for assessing the relevance and quality of relevant evidence and how they may be used for informing our estimates. The criteria are: reflective of economic and finance principles and market information where applicable, fit for purpose, implemented in accordance with good practice, models based on quantitative modelling and market data and other information. Our conclusion was that the Black CAPM only met some of the criteria and should be used to select a point estimate towards the upper half of the range. See: AER, *Explanatory statement rate of return guideline (appendices)*, 17 December 2013, pp. 16–18; AER, SA Power Networks final decision, October 2015, pp. 75–79

¹⁰⁷⁵ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 25

¹⁰⁷⁶ AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, SAPN final decision, p. 78

returns and beta–which is not consistent with the theory underpinning the Sharpe-Lintner CAPM or the Black CAPM.¹⁰⁷⁷

- Partington and Satchell have consistently advised against use of the Black CAPM and continue to do so in their latest advice for the following reasons:¹⁰⁷⁸
 - It contains a number of unrealistic assumptions that Black himself has agreed was not plausible.
 - There are major implementation problems and the robustness of estimates was poor (the estimates are volatile and unreliable).
 - o It is not used for estimating the cost of capital in practice.

Our concerns are shared by the CCP16 who has noted that the model is not particularly suitable for estimating ex-ante the equity beta or for 'adjustment' to the empirical data:¹⁰⁷⁹

- There is no consensus on the ex-ante estimate of the zero beta premium.¹⁰⁸⁰
- Estimates of the return on the zero-beta portfolio are uncertain and unstable.¹⁰⁸¹ 1082
- The significant variability in the proposed zero beta estimates, a variability that is consistent with the fact that neither the market nor regulators generally rely on the estimate.¹⁰⁸³
- There is too much uncertainty around the empirical analysis of the Black CAPM theory for it to play a substantive role in the AER's decision, and it is not generally applied by market practitioners or regulators.¹⁰⁸⁴

Given the above concerns, we note that the 2013 Guidelines may have overstated the appropriate consideration to give to the theory of the Black CAPM. Our further review of the relevant evidence supports not moving our point estimate (towards the top of the observed range) for the theory of the Black CAPM for the following reasons:

• It is not used in practice to set an ex-ante return on equity

¹⁰⁷⁷ For example, AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463, p. 284

¹⁰⁷⁸ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 16–17.

¹⁰⁷⁹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹⁰⁸⁰ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 73.

¹⁰⁸¹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 85.

¹⁰⁸² Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 85.

¹⁰⁸³ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹⁰⁸⁴ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68

- Issues with implementation and output have reduced our confidence in the model. If its effect cannot be reliably and robustly quantified with consistent results, we do not consider it appropriate to use the (theory of the) Black CAPM when selecting our estimates.
- The model does not meet our assessment criteria well¹⁰⁸⁵

This is supported by the CRG who submitted that it would be an incorrect assessment of the efficient financing costs of a benchmark efficient entity to place any weight on the Black CAPM.¹⁰⁸⁶

We observe some submissions seem to mischaracterise the role of the Black CAPM stated in the 2013 Guidelines and we respond in the table below.

Table 38 Response to mischaracterisation of consideration to BlackCAPM in 2013 Guidelines

Submission	Consideration
The 2013 Guidelines sets out an estimate of the zero-beta rate that is 2.8 per cent higher than the risk free rate, which is an extremely high value and indicates that little to no weight should be given to the Black CAPM. ¹⁰⁸⁷	We note the 2013 Guidelines did not use the Black CAPM in the manner described by the CRG. The theory of the Black CAPM was used to select a point estimate towards the upper bound of the range. ¹⁰⁸⁸
The AER has not specified what adjustment it has made in relation to the Black CAPM evidence. ¹⁰⁸⁹	We have transparently set out the consideration given to the theory of the Black CAPM in the 2013 Guidelines and subsequent regulatory decisions. It is used to select a point estimate towards the upper end of the empirical range.

Equilibrium framework

In proposing the use of expected returns to account for the low beta bias issue, APGA submitted that there are issues with the "expected equilibrium" framework espoused by Partington and Satchell as:¹⁰⁹⁰

- It ignores actual returns
- It is not clear that the AER assumes that investors form equilibrium expectations.

¹⁰⁸⁷ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 50

¹⁰⁸⁵ Given the absence of substantively new material on the Black CAPM, we consider that our views on the assessment criteria from the 2013 Guidelines remain relevant. Our conclusion was that the Black CAPM only met some of the criteria. See: AER, *Explanatory statement rate of return guideline (appendices)*, 17 December 2013, pp. 16–18; AER, SA Power Networks final decision, October 2015, pp. 75–79

¹⁰⁸⁶ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 49.

¹⁰⁸⁸ AER, Rate of Return Guidelines Explanatory Statement, December 2013, p. 86.

¹⁰⁸⁹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 60

¹⁰⁹⁰ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, pp. 24

• If the AER adopts an expected equilibrium framework approach there is nothing to suggest that it is the equilibrium suggested by the CAPM.¹⁰⁹¹

We consider that the substance of the APGA's submission is that the AER should use actual returns when estimating the return on equity and the AER's use of the Sharpe-Lintner CAPM (when the market is not in equilibrium) is incorrect.

We have consistently and transparently assessed a range of equity models based on their merits and suitability for our regulatory task when developing our Guideline and in subsequent regulatory decisions. We determined that the Sharpe-Lintner CAPM should be used as the foundation model for estimating the return on equity because it will contribute to the achievement of the ARORO. The Sharpe-Lintner CAPM:

- Has stood the test of time
- Is widely used by market practitioners
- Transparently presents the relationship between risk-reward

The Tribunal has upheld our use of the Sharpe-Lintner CAPM. Expert advice from John Handley and Graham Partington supports our use of the Sharpe-Lintner CAPM.¹⁰⁹² Service providers have also adopted the Sharpe-Lintner CAPM as the foundation model for estimating the return on equity.¹⁰⁹³

Further, there are a range of issues with using data to test and adjust the Sharpe-Lintner CAPM, including but not limited to:¹⁰⁹⁴

- Testing of an asset pricing model involves how well it describes ex-ante expected returns when security prices are in equilibrium. Empirical work attempts to examine how well the asset pricing model explains ex-post realised returns which 'may not be a particularly good test'.¹⁰⁹⁵
- The results are dependent on the method used to conduct the test (for example the characteristics used in sorting stocks into portfolios when testing model performance), was also noted by Kan, Robotti and Shanken. ¹⁰⁹⁶

Our regulatory task is to estimate the required return on equity for a benchmark efficient entity with a similar degree of risk as a relevant service provider in supplying the regulated services. This must provide ex-ante compensation for efficient financing costs as our regulatory regime is an ex-ante (forward looking) regime.¹⁰⁹⁷

¹⁰⁹¹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, pp. 24

 ¹⁰⁹² Australian Competition Tribunal, Applications by Public Interest Advocacy Centre Ltd and Ausgrid [2016] ACompT
 1, 26 February 2016, paragraph 735.

¹⁰⁹³ For example APTPPL, APA, Multinet, AusNet, AGN.

¹⁰⁹⁴ AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463, pp. 179–180.

¹⁰⁹⁵ Partington & Satchell, *Report to the AER: Analysis of criticism of 2015 determinations,* October 2015, p. 20.

¹⁰⁹⁶ Partington & Satchell, Report to the AER: Analysis of criticism of 2015 determinations, October 2015, pp. 23–24.

¹⁰⁹⁷ The AEMC describes, 'allowed revenues for network businesses are now set using the expenditure required by prudent, efficient operators as a benchmark. Companies have incentives to beat the benchmarks so they can keep some of their savings and pass the rest on to customers'. See AEMC, *Overview 2014–15*.

We note the use of ex-post data to compute adjustments is not dissimilar to previous regulatory processes where service providers submitted on the empirical performance of the Sharpe-Lintner CAPM and proposed use (or elements) of the Black CAPM and Fama-French model for estimating the return on equity.¹⁰⁹⁸

We do not agree with the use of ex-post data to adjust the Sharpe-Lintner CAPM estimates. As noted above, expected returns can diverge from realised returns over a persistent period of time, markets can be in disequilibrium and expectations are not always realised even on average.¹⁰⁹⁹

We note that it has previously been acknowledged that one could not truly test if the market is out of equilibrium.¹¹⁰⁰

The APGA's concern with the use of the SLCAPM (when the market is not in equilibrium) would also apply to every model. The issue can be resolved by assuming that historical returns are equivalent to equilibrium expected returns. However, there are a range of issues with this assumptions and it has been previously acknowledged that the two are not equivalent.¹¹⁰¹

In terms of the equity beta, our approach of empirically estimating beta using historical data is widely used both in practice and in academic work. The fact that realised returns can diverge from expected returns for a long period of time do not invalidate the equity beta data during that period. That is, it still measures the sensitivity of an asset or business's returns to movements in the overall market returns at a given period/time.

APGA submitted that there are alternatives to the SLCAPM such as the international CAPM, the Black CAPM and the Hong and Sraer (2016) model.¹¹⁰² As noted above and in section 2.4.1, we determined that the Sharpe-Lintner CAPM should be used as the foundation model for estimating the return on equity for a number of reasons including standing the test of time and widely used by market practitioners. The other models suggested by the APGA contain a number of shortcomings which makes them unsuitable for our regulatory task. For example:

 To our knowledge, the Black CAPM and the Hong and Sraer model are not used in practice by market practitioners to set ex-ante return on equity based on our review of broker reports and valuation reports

¹⁰⁹⁸ See for example Frontier, *The required return on equity under a foundation model approach*, January 2016, p. 55. Both HoustonKemp and Frontier use a return on equity that is deemed absent of low-beta bias to estimate an adjustment to the equity beta in the Sharpe-Lintner CAPM. HoustonKemp appears to use ex-post return on equity. Frontier uses a return on equity from its Black CAPM (which is derived using ex-post data). HoustonKemp also uses ex-post return on equity to estimate an 'alpha' term to include in the Sharpe-Lintner CAPM.

¹⁰⁹⁹ Partington and Satchell, Report to the AER: Discussion of submissions on the cost of equity, 8 June 2017, p. 30.

¹¹⁰⁰ Multinet Gas, *Rate of Return Overview*, 16 December 2016, p. 21.

¹¹⁰¹ Multinet Gas, *Rate of Return Overview*, 16 December 2016, p. 16.

¹¹⁰² APGA, Submission to the AER review of rate of return guideline, 4 May 2018, pp. 24

- The Black CAPM contains a range of shortcomings (for example, empirical instability, potential for non-sensical results, etc.) that we discuss in detail in section 8.3.5 and we are not persuaded to use this material when selecting a point estimate in this review.
- The Hong and Sraer model appears to be a (relatively) new model and is tested using US stock returns.¹¹⁰³ It is not clear that the model applies in the Australian context. The model's empirical analysis appears to be based on certain assumptions that raise questions about its practical use. For example, it excluded stocks with a share price below \$5 and micro caps,¹¹⁰⁴ and is based on long term analyst EPS forecasts. Long term EPS forecasts has been shown to be 'extremely' inaccurate¹¹⁰⁵ and 'poor predictor of realised earnings growth'.¹¹⁰⁶
- Submissions to the review process have generally supported the use of the SLCAPM as the foundation model for estimating the return on equity

8.3.6 Empirical methodology

Estimation technique

The 2013 Guidelines relied on Henry's empirical study which used Ordinary Least Squares (OLS) (and Least Absolute Deviation (LAD) estimators as a robustness check for outliers in the underlying data) to inform the equity beta parameter.

APA submitted that only the OLS method should be used because an LAD estimator does not have correspondence with the economic meaning of equity beta, ($\beta i = cov(ri, rM)/var(rM)$).¹¹⁰⁷

Our view is to maintain the use of LAD estimators. This is because there may be some concerns about the validity of OLS estimators in the presence of outliers.¹¹⁰⁸ The LAD estimator is less affected by the presence of outlier observations than OLS estimators.

Firm and portfolio estimates

The 2013 Guidelines used both firm and portfolio estimates from Henry's study to inform the range and point estimate for equity beta.

¹¹⁰³ Hong, Sraer, Harrison, David, Speculative betas, 15 June 2015, p. 19.

¹¹⁰⁴ Defined as stocks in the bottom 2 deciles of the monthly market capitalisation distribution using NYSE breakpoints.

¹¹⁰⁵ Harris, Richard D.F., The Accuracy, bias and efficiency of analysts' long run earnings growth forecasts, Journal of Business Finance & Accounting, June/July 1999, p. 726

¹¹⁰⁶ Da, Warachka, Zhi, Mitch, The disparity between long-term and short-term forecasted earnings growth, Journal of financial economics 100, 2011, p. 426.

¹¹⁰⁷ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 22

¹¹⁰⁸ Henry Olan, Estimating beta: an update, p. 8.

APA cautioned against the use of portfolio estimates for the return on equity of electricity service providers (SKI and AST) or pipeline service providers (APA) due to a difference in beta estimates.¹¹⁰⁹

We consider that because no one comparator firm is perfectly reflective of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services, we use portfolios and averages of individual firm estimates to determine the equity beta range. We consider taking an average over the individual equity beta estimates is likely to produce an equity beta estimate that is more reflective of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services than considering individual firm estimates in isolation.¹¹¹⁰ In this respect, we also consider equity beta estimates and averaged individual firm estimates combine information from multiple comparator firms, instead of considering single firms in isolation.

We do observe a difference between estimates. However, this is not necessarily indicative of different systematic risk in the supply of regulated energy network services for gas and electricity providers. We observe that APA has undertaken a range of transactions that would increase their exposure to systematic risk from unregulated assets and/or assets that are different from the risk of providing services with a similar degree of risk as the regulated energy network services.¹¹¹¹

We acknowledge that APA has a high proportion of unregulated operations. However, we still include APA in our comparator set because we recognise service providers' concern with a small number of comparators.

Length of estimation periods

In the 2013 Guidelines we recognised that there is generally a trade–off in determining the length of the estimation period.¹¹¹² On one hand, older data might be considered less reflective of current systematic risk assessments (which would suggest a shorter period). On the other hand, in order to obtain a robust and statistically reliable equity beta estimate we need to have sufficient number of observations (which would suggest a longer period).

On balance, we considered it was reasonable to use an estimation period of at least five years and to have regard to the longest period available. We proposed to consider regressions using three permutations of the estimation period:

¹¹⁰⁹ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 23

¹¹¹⁰ For example, AER, Final decision SA Power Networks determination 2015–16 to 2019–20, October 2015, pp. 451–463, p. 469

¹¹¹¹ <u>https://www.apa.com.au/about-apa/our-history/; http://www.duet.net.au/getattachment/ASX-releases/2015/DUET-Completes-Acquisition-of-Energy-Developments/DUET-Completes-Acquisition-of-Energy-Developments-L/DUET-Completes-Acquisition-of-Energy-Developments-Limited.pdf.aspx</u>

¹¹¹² AER, Better Regulation Explanatory Statement Rate of Return Guideline (Appendices), December 2013, p. 49.
- The longest period available
- The period after the 'technology bubble' and before the global financial crisis (GFC) and the period after GFC
- The last five years of available data

We remain of the view that a variety of sampling periods should be used in our empirical study to estimate equity betas for firms we consider most comparable to the firms/assets we regulate. Hence we maintain the use of these 3 estimation periods as they provide long term and more recent estimates of equity beta.

This is supported by thee expert joint report which noted that both long and short term estimates should be considered. Some experts noted that a long data series are likely to produce the most statistically reliable estimates while others noted that a shorter period (e.g. 5 years) may reflect more recent movements.¹¹¹³

APA submitted that variation over time cannot be ignored and older data become less relevant if estimates are to reflect prevailing financial market conditions. It added that estimation using five years of monthly data should allow closer examination of changes, if any, over time in the equity beta.¹¹¹⁴

We consider this is consistent with our use of last 5 years of available data in our empirical study.

Gearing adjustment

We de-lever a firm's raw equity beta estimates (using a firm's actual gearing).

In the 2013 Guidelines, for firms that hold a minority interest (less than 50 per cent ownership) in an asset or company, their debt and hence gearing may be understated. This is because the investments may be reported using the equity accounting method, which does not require firms to report debt held by those assets on its balance sheet.¹¹¹⁵ We would need to make adjustments to ensure a firm's gearing appropriately reflects the level of debt held by itself and its assets.

Such an instance rose with Spark Infrastructure (SKI) which holds a minority interest in regulated energy networks (SAPN, CitiPower, Powercor and TransGrid). Its share of

¹¹¹³ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 50.

¹¹¹⁴ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 24

¹¹¹⁵ The equity method is used for reporting investments in an associate or a joint venture where the investments is initially recognised at cost and the carrying amount is increased or decreased to recognise the investor's share of the profit or loss of the investee. Essentially firms report their share of profit and loss from investments but is not required to report its share of debt held by its assets. See: AASB Board Standard, Investments in associates and joint ventures, para 10–15. Available at: http://www.aasb.gov.au/admin/file/content105/c9/AASB128_08-15_COMPdec15_01-18.pdf

debt held by those networks was not reported in its public financial reports due to the use of the equity accounting method.

As a result, gearing information computed from its firm-level estimates was likely under-stated. SKI also provided related party lending to those same networks.

To ensure gearing appropriately reflects all debt attributable to SKI, the look through method was used to incorporate the related party lending and SKI's share of the asset-level debt. This method is summarised in the following steps:

- Estimate SKI's share of its assets' borrowings (net of related party borrowings);
- This estimate is then combined with related party borrowings and other borrowings to arrive at total debt, which is then used to derive overall gearing in combination with market value of equity.¹¹¹⁶

We note that SKI continues to hold minority interest in regulated energy networks:1117

- 49 per cent interest in CitiPower and Powercor
- 49 per cent interest in SA Power Networks
- 15.01 per cent interest in TransGrid

It also continues to provide related party lending to those networks.¹¹¹⁸

We therefore consider it appropriate to maintain the use of the look through adjustment to ensure gearing appropriately reflects all debt attributable to SKI.

Use of other information to estimate equity beta

Mr David Johnstone raised the prospect of using regulated cashflows to estimate the equity beta.¹¹¹⁹ However other experts doubted this proposal due to limitations such as a high degree of subjectivity, potential for data manipulation, departure from the current approach and insufficient data frequency.¹¹²⁰

The Consumer Reference Group (CRG) also raised the following questions regarding the use of market data for estimating equity beta:

 How to account for volatility that is driven by issues totally unrelated to the risks the firm takes with its investment in assets and risk fundamentals of defensive stocks.¹¹²¹

¹¹¹⁶ This is consistent with the approach in Henry's study where debt and equity are used to estimate a firm's gearing level.

¹¹¹⁷ SKI, Annual Report 2017, p. 90.

¹¹¹⁸ For example, see SKI, Annual Report 2017, p. 93.

¹¹¹⁹ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 48.

¹¹²⁰ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 48.

¹¹²¹ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 44, 48.

- Market data may not appropriately reflect information for a firm. The market is not fully aware of all aspects of a firm's operations and many investors have little knowledge of a firm's specific risk factors.¹¹²²
- The return on equity is seen more as a resetting bond with its defined yield and certain maturity date rather than an investment in the stock market.¹¹²³
- Firms like the benchmark efficient entity have mechanisms available to them to manage exposure to systematic risk which are not available to other firms. The market data used to generate the parameters in the CAPM include full exposure to all these risks.¹¹²⁴ As a result, the equity beta are overstated.¹¹²⁵

We consider the CRG's substantive issue is that the relatively high volatilities (in share price) for APA, SKI and AST appear inconsistent with the (relatively) low risk of supplying the regulated energy services and the overall low volatility in the Australian market.

As noted in section 2.4.1, we consider that the use of market data is appropriate for estimating equity beta. The rate of return needs to be consistent with the prevailing cost of equity¹¹²⁶ and that is best measured through market data. The rate of return also needs to reflect the efficient cost of finance. We consider that efficient financing costs are more likely to be reflected in the prevailing market cost of capital.

Further, experts also agreed that beta should be estimated from stock market data.¹¹²⁷ Partington and Satchell have cautioned against using non-market data (such as cashflows) for estimating equity beta. They noted there is a paucity of data (due to cashflows being reported on annual basis) and there would be many difficulties to overcome.¹¹²⁸

De/re-levering

The raw equity beta estimates of comparator businesses will reflect varying levels of actual financial leverage. These raw estimates can be de-levered to obtain the asset beta of the business. The result of de-levering reflects the beta of the asset if the asset was financed 100 per cent by equity, with zero debt. These asset betas can then be relevered to match the level of gearing associated with a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services (as adopted by the regulator).

¹¹²² CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 44.

¹¹²³ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 48.

¹¹²⁴ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 45

¹¹²⁵ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 48

¹¹²⁶ NER; NGR

¹¹²⁷ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 42.

¹¹²⁸ Partington and Satchell, Report to the AER: Allowed rate of return 2018 guideline review, 25 May 2018, p. 14–15.

In the 2013 Guidelines we estimated a benchmark gearing ratio of 60 per cent and we used the Brealey–Myers formula (assuming a debt beta of zero) to de-lever and relever the comparable businesses' equity beta estimates.¹¹²⁹ That is:

$$\beta_e = \beta_a \left(1 + \frac{D}{E} \right)$$

where:

- \circ β_e is the equity beta
- \circ β_a is the un-levered asset beta, and
- $\circ \frac{D}{F}$ is the debt to equity ratio.

We note there are views both for and against de-levering and re-levering equity beta estimates. On one hand, the resulting estimates will be more aligned with our benchmark. On the other hand however, the relationship between equity beta, financial leverage and financial risk is complex and uncertain.

There were divergent views on the de/re-levering process:

- The ENA, NSG and APA supported the current approach.¹¹³⁰ ¹¹³¹ ¹¹³² The ENA noted that equity beta not re-levered in this way cannot be compared on a like-for-like basis.
- Stephen Gray, Simon Wheatley and Ilan Sadeh agreed that equity betas need to be re-geared to 60 per cent.¹¹³³
- Graham Partington expressed concern about the de/re-leveraging process due to concerns such as the measurement of leverage, choice of data¹¹³⁴ He added that the use of equity beta estimates (without re-levering) may be justified given the AER's use of the plain vanilla WACC or if any difference between firm gearing and benchmark gearing is not material¹¹³⁵

¹¹²⁹ AER, Better Regulation Explanatory Statement Rate of Return Guideline (Appendices), December 2013, pp. 50– 51

¹¹³⁰ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 21.

¹¹³¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9, 49.

¹¹³² Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 14.

¹¹³³ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 11, 12, 13.

¹¹³⁴ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 39–40.

¹¹³⁵ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 10

- CCP16 acknowledged that the de/re-levering process is standard practice to 'normalise' equity beta for different gearing levels. However, it also saw merit in Partington and Satchell's view for using raw equity beta estimates. ¹¹³⁶ ¹¹³⁷
- We maintain the decision to de-lever and then re-lever equity beta estimates to 60 per cent gearing for our empirical analysis.

We note that a firm's gearing level affects its equity beta estimate and our comparator firms have gearing levels that deviate from the benchmark level of 60 per cent. This requires de/re-levering to ensure they are on a like-for-like basis with the benchmark gearing to inform the systematic risk of a benchmark efficient entity with a similar degree of risk as a relevant service provider in respect of the provision of regulated services.

Further, we note the choice of whether or not to de-lever and re-lever is unlikely to be material on the average of individual firm estimates and portfolio estimates. This is because the industry average gearing and the benchmark gearing are very similar. However, the difference between raw and re-levered equity beta estimates for individual firms may be greater because some firms have higher or lower gearing than a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services.

Brealey–Myers formula

There appeared to be confusion regarding the formula we use for de/re-levering during the process to date.

This was noted by Professor Graham Partington at the concurrent evidence session.¹¹³⁸

He observed that some said the AER uses the Miles and Ezzell formula while the AER says it uses the Brealey–Myers formula.¹¹³⁹ He noted that the Brealey–Myers formula assumes a target leverage and continuous rebalancing to never deviate from target.¹¹⁴⁰ He added that the AER does not use the Miles and Ezzell formula which assumes annual rebalancing for gearing.¹¹⁴¹

¹¹³⁶ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68

¹¹³⁷ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 80.

¹¹³⁸ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 8

¹¹³⁹ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 8

¹¹⁴⁰ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 8

¹¹⁴¹ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 19

However, Stephen Gray, Simon Wheatley and Ilan Sadeh noted that the AER uses the Miles and Ezzell formula.¹¹⁴² This was also in the ENA's submission which stated that the Miles-Ezzell formula should be used.¹¹⁴³

The Miles and Ezzell formula is reproduced from the joint expert report:¹¹⁴⁴

$$\beta_e = \beta_a \left(1 + \frac{D}{E} \right) \left(\frac{1 + r_d \left(1 - T_c \frac{D}{V} \right)}{\chi \quad 1 + r_d} \right)$$

We note that this is clearly different from the Brealey-Myers formula we use so it is not clear why some note we use the Miles and Ezzell formula. We reiterate that we use the Brealey–Myers formula for de/re-levering equity beta estimates.

Debt beta

There were divergent submissions on assuming a debt beta of zero:

- Graham Partington noted that 'assuming that the debt beta is zero...results in an upward biased estimate of the equity beta' ¹¹⁴⁵
- Stephen Gray noted that debt beta does not carry a material impact when estimating equity beta so long as it is used consistently when de/re-levering.¹¹⁴⁶ This had agreement from Glen Wheatley and Ilan Sadeh.
- Ilan Sadeh stated that independent valuers rarely use debt beta.¹¹⁴⁷ He added that it would be inconsistent to use market-based evidence to estimate equity beta and then adjust it using a debt beta that 'nobody knows where it came from'.
- The ENA supported a debt beta of 0 because a reasonable estimate of debt beta has no material impact on the final estimate of equity beta and because of measurement issues.¹¹⁴⁸ It added that in practice it is common to use a debt beta of zero because debt beta estimates tend to be very small¹¹⁴⁹

¹¹⁴² AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 12–13

¹¹⁴³ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46, 50

¹¹⁴⁴ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 40.

¹¹⁴⁵ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 1, 15 March 2018, p. 41

¹¹⁴⁶ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 12

¹¹⁴⁷ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 13

¹¹⁴⁸ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46.

¹¹⁴⁹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 50.

Our view is that the debt beta should continue to be zero. It appears market practice to assume a debt beta of zero¹¹⁵⁰ and Damodaran has made the same assumption in his works.¹¹⁵¹ Further, the debt beta of businesses in the provision of regulated energy network services are likely to be low due to the relatively low risk of supplying regulated energy network services.

8.3.7 Industry analysis

In the Equity Beta Discussion Paper,¹¹⁵² we estimated the industry/sector beta for 11 Australian industries as classified by Bloomberg. We noted that the composition of the industry indices suggests that utilities would be the most comparable industry group. This is because the utilities index includes our comparator firms¹¹⁵³ which make up 3 of the top 5 firms in this index.¹¹⁵⁴

Submissions and experts have supported the use of industry indices:

- The CCP16 considered that using industry/sector indices (as defined by Bloomberg) to improve the estimate of the equity beta for the BEE appears to have some potential and should certainly form part of the AER's considerations for the new instrument.¹¹⁵⁵ It added that there is considerable benefit in the AER further pursuing the option to use the Bloomberg Utilities index as part of the information contributing to the estimation of the equity beta for the BEE. ¹¹⁵⁶ ¹¹⁵⁷
- Partington and Satchell noted the utilities industry would probably be the industry that most closely resembles the energy network businesses as the industry classification includes these businesses.¹¹⁵⁸
- The expert joint report noted that industry and sector indices are likely to be too broad and with different risks reflects than those of the BEE.¹¹⁵⁹ However, some experts noted that industry indices might be relevant and would be open to its use.¹¹⁶⁰

¹¹⁵⁰ Based on our analysis of broker reports and expert valuation reports.

¹¹⁵¹ Damodaran, A. (2002). Investment Valuation – Tools and Techniques for Determining the Value of any Asset, 2nd edition, New York, John Wiley & Sons, p. 194.

¹¹⁵² AER, Discussion paper equity beta, March 2018.

¹¹⁵³ The Bloomberg utilities index includes our comparator firms that remain listed (APA, AST and SKI).

¹¹⁵⁴ The top 5 firms in the utilities index make up the majority of the index: 3 of these firms are part of our comparator set (APA, AST and SKI).

¹¹⁵⁵ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 82.

¹¹⁵⁶ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 82.

¹¹⁵⁷ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹¹⁵⁸ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, May 2018, p. 15.

¹¹⁵⁹ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 47.

¹¹⁶⁰ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 47.

We have updated this analysis to March 2018 using OLS and LAD estimators and 3 scenarios.¹¹⁶¹ Figure 28 shows that OLS estimates range between 0.55–0.67 and LAD estimates range from 0.53–0.64.



Figure 28 Utilities index beta estimates

Source: AER analysis, Bloomberg Note: PTEG is Scenario 2 which is post tech boom excluding GFC

8.3.8 Other information informing range and point estimate

We note there is other information available to inform our selection of the range and point estimate:

- Our analysis of the impact of regulation in section 8.3.2 suggests that regulation reduces the systematic risk. Our comparator set contains firms with varying levels of regulated operations and firms with majority regulated operations typically have lower equity beta estimates. We do not exclude firms with fewer regulated operations from our comparator set. However, estimates for the longest estimation period derived from firms with a high proportion of regulated operations are clustered in the bottom half of the empirical range.
- There has been a material decrease in the debt risk premiums in the market since the 2013 Guidelines. Further, the spread between the prevailing debt risk premium and the 2013 Guideline's estimated equity risk premium has widened from approximately 1.1 per cent (in 2013) to 2.8 per cent. This suggests that required

¹¹⁶¹ The longest estimation period, post tech-boom excluding GFC, last 5 years.

risk margins on capital have decreased since 2013. This information is supportive of choosing a lower equity risk premium than in 2013.

- The regulatory framework mitigates systematic risk in the supply of regulated energy network services. This is especially the case where our use of a trailing average cost of debt provides a natural hedge against movements in interest rates and our method for accounting for inflation provides compensation for outturn inflation. This provides support for a point estimate towards the bottom half of the range.
- Certainty and stability are desirable to attract and retain funds Consumers, investors and regulated entities value stability. This factor supports choosing a beta value consistent with our current approach.
- The Bloomberg Australian utilities index¹¹⁶² includes the 3 still-listed comparator firms which make up 3 of the top 5 firms in this index. We give most weight to the longest estimation period which yields OLS estimate of 0.55 and LAD estimate of 0.53.

8.3.9 Range and point estimate

We choose an empirical range of 0.4–0.8 based on our update of Henry's study in section 8.3.2. We consider the equity beta of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services is in this range for the following reasons:

- Our conceptual analysis supports that the equity beta of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services would be low and below 1.0 (see section 8.3.1).
- International empirical estimates supports an equity beta estimate below 1.0 (see section 8.3.4).
- We consider the systematic risks of the networks we regulate are sufficiently similar between the different types of network we regulate to warrant one estimate (see section 2.4.2).

In selecting a point estimate, we continue to give most weight to empirical estimates of the firms we regulate. Our conclusions from examining these estimates are:

• Estimates across all estimation periods and techniques cluster around the 0.5–0.6 range. The average of OLS estimates across all periods is 0.57.

Empirical estimates have increased since 2013. However, the increase has been small, particularly over the longest estimation period which we given the most weigh to (as discussed in section 8.3.2). Further most estimates continue to cluster around the 0.5–0.6 range as observed in Figure 25 and

¹¹⁶² The composition of the industry indices suggests that utilities would be the most comparable industry group.

- Table 31.
- We give most weight to the longest estimation period. This supports an empirical estimate of 0.51¹¹⁶³ and is consistent with the clustering observed in the previous dot point.

In the 2013 Guidelines we estimated a beta point estimate that was some distance above the best empirical estimates for our comparator firms. We did so to promote stability and caution. The point estimate in the 2013 Guidelines was a decrease from the beta value of 0.8 used in previous regulatory determinations and we did not want to move a large increment. We also took into account the theory of the Black CAPM.

On the balance of evidence, we consider that an equity point estimate of 0.6 is reasonable as it reflects:

- The result of our empirical analysis which we give most weight. This supports an equity beta estimate of 0.57 across OLS estimates for all estimation periods and a value of 0.51 for the longest estimation period
- The other information we have considered as discussed in section 8.3.8
- A clear and transparent manner with which to set the equity beta estimate.
- Promoting stability by not departing substantially from our previous value and leaves some scope to account for concerns around market imperfections affecting the SLCAPM while recognising the other factors we have identified as relevant.

In support of an equity beta of 0.7 or higher, service providers, investors and network associations have submitted that beta estimates have increased since the 2013 Guidelines.¹¹⁶⁴ ¹¹⁶⁵ ¹¹⁶⁶ ¹¹⁶⁷ Some experts noted that the AER's own estimates shows this movement. ¹¹⁶⁸ In its April 2018 report for AusGrid, Frontier has also submitted that the AER's own evidence shows an increase since the 2014 which has been muted due to the de-listed firms.¹¹⁶⁹

We first note these submissions appear similar to Prof. Gray's suggestion of starting from the 2013 value of 0.7 and letting market movements dictate if a change is warranted.¹¹⁷⁰ However, this ignores considerations in the 2013 Guidelines, which

¹¹⁶³ Simple average of portfolio-level estimates and the average of firm-level estimates.

¹¹⁶⁴ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 3.

¹¹⁶⁵ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46.

¹¹⁶⁶ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 55.

¹¹⁶⁷ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.

¹¹⁶⁸ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 52.

¹¹⁶⁹ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 33.

¹¹⁷⁰ AER, Transcript of proceedings Australian Energy Regulator office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 39.

supported an empirical point estimates of 0.5. The (eventual) point estimate of 0.7 was selected due to the theory of the Black CAPM, international estimates, certainty and stability. In this review we are not persuaded to select an equity beta towards the upper end of our observed empirical range based on our further consideration of the Black CAPM and international estimates.

It is also not clear to us that the movements since 2013 are supportive of an increase in the systematic risk in the supply of regulated energy services and an equity beta above 0.7:

- Submissions in support of increased risk generally focused on the effect of technological risks, natural disasters and policy risk. We consider these risks are non-systematic and do not warrant compensation through the rate of return in section 2.4.1.
- Partington and Satchell have advised that a number of reasons can contribute to observations of an increase in beta including the de/re-leveraging process and changes in the individual characteristics of one firm.¹¹⁷¹
- Consultants retained by the service providers have previously acknowledged that the increase is driven by gearing movements.¹¹⁷²
- Some firms have undertaken a range of transactions that would increase their exposure to systematic risk from unregulated assets and/or assets that are different from the risk of providing the regulated energy network services.¹¹⁷³

This view is shared by the CCP16 who noted that the evidence to date does not support an increase in the estimate of the equity beta as they are based on short term estimates and many features of the post 2013 regulatory regime would reduce systematic risk.¹¹⁷⁴

We note one key factor for service providers' observation may be falling interest rates since the 2013 Guidelines. Partington and Satchell have noted that low beta stocks such as our comparator firms can be considered bond proxies and have been noted as such by financial institutions and practitioners.¹¹⁷⁵

They explained that 'bond proxies are likely to respond to interest movements in a fashion that is to some extent similar to bonds. That is, there is likely to be an inverse

¹¹⁷¹ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 17.

¹¹⁷² For example, CEG, Replication and extension of Henry's beta analysis, 21 September 2016; CEG, Replication and extension of Henry's beta analysis, November 2016

¹¹⁷³ <u>https://www.apa.com.au/about-apa/our-history/; http://www.duet.net.au/getattachment/ASX-releases/2015/DUET-Completes-Acquisition-of-Energy-Developments-L/DUET-Completes-Acquisition-of-Energy-Developments-Limited.pdf.aspx</u>

¹¹⁷⁴ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 73

¹¹⁷⁵ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 18; DJ Carmichael, Increasing Focus on Global Bond Yields, 20 July 2017; Motley Fool, Citigroup thinks this income could deliver a fatter-than-expected dividend, February 2018, available at: https://www.fool.com.au/2018/02/28/citigroupthinks-this-income-stock-could-deliver-a-fatter-than-expected-dividend/

relation between prices and interest rates.' ¹¹⁷⁶ As a result, we consider that they would tend to outperform the market during times of interest rate decreases.¹¹⁷⁷ This view is shared by Partington and Satchell whom noted that bond proxies '[does] particularly well with substantial price appreciation in the period of low interest rates'. ¹¹⁷⁸

The associated outperformance would drive an increase in equity beta estimates¹¹⁷⁹ even though the true systematic risk of supplying regulated energy network services may remain (relatively) unchanged. This is because the revenue earnt from regulated energy services are determined during regulatory resets and (generally) guaranteed over a regulatory period. Regulation of energy network services also mitigates exposure to systematic or compensable risk as discussed in section 2.4.3.

The outperformance would be magnified by the low risk from the market (as indicated by the VIX index in section 7.3.5) which made bond proxies even more bond like.¹¹⁸⁰

Given the above concerns with beta movements since the 2013 Guidelines and recent interest rate movements, we consider they lend more support for giving most weight to empirical estimates from the longest estimation period.

In section 8.3.2, our comparison of still-listed majority regulated firms indicated potential for a value less than that from the whole comparator set. However, we note that the 2013 Guidelines estimated a lower equity beta than previous determinations (from 0.8 to 0.7) even though longer term estimates were clustered materially below 0.7. In part, we took into account the need to promote stability and predictability and therefore decided not to make a larger change. We adopt similar considerations in this decision and consider a further reduction at this time will provide an estimate that is more in line with empirical estimates and so we have reduced the equity beta by 0.1.

We note Prof. Gray and Dr. Wheatley stated that our estimates of equity beta in the 2013 Guidelines were at a 'low point in the cycle' and have increased since that time.¹¹⁸¹

We consider that, if there is a cyclical aspect, this would warrant placing more emphasis on the use of long term data series for estimating parameters and avoid over-reliance on short term estimates which may capture a section of the cycle and unduly effected by interest rate movements, volatilities and 'one-off' events.

Material informing range and point estimate

 ¹¹⁷⁶ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 21.
 ¹¹⁷⁷ <u>https://www.clime.com.au/investing-report-archive/time-sell-bond-proxies/;</u>

https://www.commbank.com.au/guidance/retirement/what-you-need-to-know-about-bond-proxies-201610.html;

Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 18.
 Equity beta measures the 'riskiness' of a firm's return compared with that of the market. Both negative and positive outperformance compared with that of the market would drive increase in the equity beta estimate.

¹¹⁸⁰ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 19.

¹¹⁸¹ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 24.

Stakeholders submitted divergent views on the weight given to particular piece of evidence for informing the equity beta parameter.

Consumer groups supported an approach that focused on the use of market practice, advice from Partington and Satchell and reducing the weight given to the Black CAPM.¹¹⁸² Service providers and network associations proposed for more and explicit weight for the Black CAPM and low beta bias (as noted and discussed in section 8.3.5). They also supported expanding the comparator set to include international energy firms and other Australian infrastructure firms (as noted and discussed in section 0).

We have assessed the strengths, weaknesses and suitability of each relevant piece of evidence.

We concluded that international empirical estimates are not suitable for inclusion in the comparator set in section 0. Our view is that they are better used in a qualitative manner as a cross check for our empirical range. Differences with a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services means that international estimates cannot be used to meaningfully and robustly inform the equity beta estimate.

We have considered the Black CAPM and low beta bias. Our view is to not use the (theory of the) the Black CAPM or the low beta bias when selecting our point estimate. Our assessment, in section 8.3.5, concluded that they are not suitable for our regulatory task which is to estimate the ex-ante rate of return required to efficiently compensate for a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy network services.

The ENA submitted that the international evidence considered by the AER and the evidence from other domestic infrastructure indicates an equity beta above 0.7.¹¹⁸³ Our assessment of international energy estimates in section 0 and 8.3.4 shows that they should be used in a cross checking role. We consider no weight should be given to other Australian infrastructure firms in section 0.

We note Energy Australia recommended we focus on ensuring the best estimate from the Sharpe-Linter CAPM is applied, as it is a model widely used to ascertain an appropriate weighted average cost of capital.¹¹⁸⁴

Submissions on selection of range and point estimate

We received a number of submissions on the selection of range and point estimate. We respond to them in the table below.

¹¹⁸² Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 89; Energy Consumers Australia, Review of the rate of return guideline response to the AER Issues Paper, December 2017, p. 23, 27; EUAA, EUAA submission – AER rate of return review issues paper October 2017, 18 December 2017, p. 9

¹¹⁸³ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

¹¹⁸⁴ Energy Australia, AER – Review of the rate of return guideline –evidence sessions, 4 May 2018

Table 39 Response to submission on selection of range and pointestimate

Submission	Consideration
Submissions to the review have submitted for more transparency in the selection of range and point estimate ¹¹⁸⁵	We have transparently set out our assessment of and the weight given to each piece of relevant evidence in accordance with step 2 of our foundational model approach. We have done this in the 2013 Guidelines,
Professor Stephen Gray also noted that that the AER's current approach does not make sense because the empirical range should not bound information on estimates from other sources of evidence. He noted that it would be better to set out and consider all evidence on equal footing	subsequent regulatory decisions, and this decision. ¹¹⁰ We have set out all the relevant evidence equally in step 2 of our decision and the weight given to a piece of evidence is based on its strengths, weaknesses and suitability for our regulatory task. To give equal weight to all relevant evidence ignores the fact that some material may be better suited in a qualitative role.
We received divergent submissions on the point estimate for equity beta. Consumer groups submitted for a point estimate of 0.6 and below. ¹¹⁸⁷ Service providers, investors and network associations espoused an equity beta of 0.7 and above. ¹¹⁸⁸ ¹¹⁸⁹ ¹¹⁹⁰	We set out how we arrive at our point estimate in section 8.3.9. We have had regard to all the relevant evidence in accordance with the weight we give them after assessing their strengths, weaknesses and suitability for our regulatory task.
We note Prof. Gray's view that we should start from the 2013 Guidelines	We note our regulatory task is to estimate an efficient rate of return to compensate for a benchmark efficient entity with a similar degree of risk as a relevant service provider in supplying regulated energy network services. Simply starting from a value of 0.7 would discount a range of relevant considerations, including:
estimate of 0.7 and let evidence indicate any changes to this point estimate. ¹¹⁹¹	• those from the 2013 Guidelines that indicated a range of 0.4–0.7 and supported a (possible) point estimate towards the lower value within this range
	 information we have received since the 2013 Guidelines on relevant evidence that would change the weight given to them. The 2013

- ¹¹⁸⁵ Ergon Energy and Energex, AER Issues paper review of the rate of return guidelines Ergon Energy and Energex submission, 12 December 2017, p. 2; Network Shareholder Group, Re: Response to issues paper on the review of the rate of return guideline, 12 December 2017, pp. 9-10; Evoenergy, Review of rate of return guideline –evidence sessions, 4 May 2018, p. 3; APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 3.
- ¹¹⁸⁶ For example, see: AER, Final decision SA Power Networks determination 2015–16 to 2019–20 Attachment 3 Rate of return, October 2015, p. 52–55, 93– 95
- ¹¹⁸⁷ Energy Consumers Australia, Review of the rate of return guideline response to the AER Issues Paper, December 2017, p. 27; CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 51, 68; Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68.
- ¹¹⁸⁸ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 60, 63; Evoenergy, Review of rate of return guideline – evidence sessions, 4 May 2018, p. 3; APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 5.
- ¹¹⁸⁹ CKI, AER discussion papers-review of the rate of return guideline, 4 May 2018, p. 6.
- ¹¹⁹⁰ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 14
- ¹¹⁹¹ AER, Transcript of proceedings Australian Energy Regulator office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 39.

Submission	Consideration
	Guidelines estimated an equity beta point estimate of 0.7 having regard to the theory of the Black CAPM and international estimates. As discussed in section 8.3.5 and 8.3.4, we are not persuaded to move our point estimate (towards the upper end of the observed empirical range) for the Black CAPM and we use international estimates qualitatively in a cross-checking role.
	• Professor Satchell also disagreed with Prof. Gray's proposal as that would discard information that supported lower values and may misconstrue the 0.7 as the mid-point of a range. ¹¹⁹²
The APGA submitted that the remaining listed firms should be used to form an estimate and confidence interval with other information subsequently used to adjust the confidence interval to arrive at a point estimate. ¹¹⁹³ At the expert concurrent evidence session, experts discussed setting confidence interval (using standard deviations) and a point estimate for individual parameters which is then used to set a point estimate and a confidence interval for the WACC. ¹¹⁹⁴ Other (qualitative) information can be used to select the end WACC from the confidence interval ¹¹⁹⁵	 We disagree with the use of confidence intervals in this way for reasons we have previously outlined.¹¹⁹⁶ These reasons include: The presence of outliers can affect point estimates and their associated confidence intervals. The presence of autocorrelation and heteroskedasticity creates difficulties in discerning whether confidence intervals overstate or understate the upper bound estimate. Further, we consider the intent of the method discussed by experts is not dissimilar to the foundational model approach. The focus is on using other information to triangulate and perform a reasonableness check on the end estimate. Step 5 of our foundational model approach compares our return on equity against information from other relevant evidence to assess the reasonableness of our estimate.
The CCP16 submitted that most weight should be placed on longer estimation periods. ¹¹⁹⁷ It added that it is essential that a long-term data series is used, as on balance and noting the risks of this, the task of the AER is to estimate ex- ante the long-term beta and to see through the ups and downs of annual data and short-term trends ¹¹⁹⁸	This supports our view to give most weight to the longest estimation period and the view that longer estimation period is likely to provide more robust equity beta estimates. ¹¹⁹⁹
David Johnstone submitted that there is a risk of circularity with decisions on the	It is not clear that such circularity exists as we have not been provided with evidence to allow assessment of this hypothesis. We also note that it

¹¹⁹² AER, Transcript of proceedings Australian Energy Regulator office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 44

¹¹⁹³ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 6.

¹¹⁹⁴ AER, Transcript of proceedings Australian Energy Regulator office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 131–132

¹¹⁹⁵ AER, Transcript of proceedings Australian Energy Regulator office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 131–133

¹¹⁹⁶ AER, *Final decision: WACC review*, May 2009, pp. 286-290.

¹¹⁹⁷ Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 83.

¹¹⁹⁸ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹¹⁹⁹ AER, Final decision: WACC review, May 2009, p. 329.

Submission	Consideration
rate of return parameters influencing future estimates. ^{1200 1201 1202}	would be difficult for an entity to manipulate its own price to distort its beta estimate. ¹²⁰³ Further, this did not appear to be a substantive concern as the CCP16 acknowledged that the 'decision on the level of betadoes not affect the future empirical estimates of beta'. ¹²⁰⁴
The ENA submitted the only evidence that supports maintenance of the same equity beta as in 2013 is the evidence from delisted firms, whose beta estimates are frozen in time forever. ¹²⁰⁵	We disagree. In the 2013 Guidelines we chose to estimate a beta value that was some distance above the best empirical estimates for our comparator firms. We did so to promote stability and caution. The beta estimate in the 2013 Guidelines was a decrease from the beta value of 0.8 in previous determinations, and we did not want to move a large increment. We also took into account of the theory of the Black CAPM and international estimates.
	Our further consideration of the relevant evidence provide a range of 0.4– 0.8 and a point estimate of 0.6. We consider that international estimates should be used qualitatively and we are not persuaded to select an equity beta towards the top of the observed empirical estimate for the (theory of the) Black CAPM
The CCP16 submitted that it does not accept that the AER should select a figure at the top of the empirical range. It recommended that the AER adopt a more balanced view on each of the parameters including the equity beta. ¹²⁰⁶	As discussed in section 8.3.9, on the balance of the evidence, our draft decision is to adopt a point estimate of 0.6 which is towards the middle of the empirical range.
Both the CCP16 and ATCO gas appeared to support an approach that results in stability ^{1207 1208}	In forming the range and point estimate we have had regard to stability as part of our consideration of other information in section 8.3.8 and 8.3.9.
We suggest the AER take a more balanced approach, by selecting values for each parameter that are more towards the mid-point of the calculated range. This would ensure that the allowed rate of return for a network business is commensurate with the efficient financing costs of benchmark	As discussed in section 8.3.9, on the balance of the evidence, our draft decision is to adopt a point estimate of 0.6 which is towards the middle of the empirical range. We do not simply set the point estimate as the middle-point of the empirical range.

- ¹²⁰¹ Cambridge Economic Policy Associates, Rate of Return Guideline Review Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 41–42.
- ¹²⁰² Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 80.
- ¹²⁰³ Cambridge Economic Policy Associates, Rate of Return Guideline Review Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 42.
- ¹²⁰⁴ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 42.
- ¹²⁰⁵ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9
- ¹²⁰⁶ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 69
- ¹²⁰⁷ Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 89.
- ¹²⁰⁸ ATCO gas, Re: Review of rate of return guideline-issues paper, 12 December 2017, p. 7–8.

¹²⁰⁰ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 80.

Submission	Consideration
efficient entity with a similar degree of risk. It would also be more consistent with meeting the National Electricity Objective and the National Gas Objective. ¹²⁰⁹	
Investors also noted that there should be a high bar to change from the current approach and there is no compelling reason to make any adjustment to the approach. ¹²¹⁰	As discussed in section 8.3.4 and 8.3.5, divergent submissions have led us to further consider the weight given to certain relevant evidence. We consider information and advice that we have received since the 2013 Guidelines has reinforced existing and highlighted new limitations of the Black CAPM and international estimates. Therefore, our draft decision is that international estimates should be used qualitatively and we are not persuaded to select a point estimate towards the upper end of our observed empirical range for the (theory of the) Black CAPM.
APA submitted that current estimate shows the systematic risk faced by APA is above that for which the benchmark is compensated. It added that there is nothing in the beta estimate of APA that tells us that the current estimate of 0.7 over-compensates gas businesses in particular. ¹²¹¹	We note that current estimates for APA does not necessarily reflect the risk of supplying regulated energy network services as it has undertaken a range of transactions that would increase exposure to systematic risk from unregulated assets and/or assets that are different from the risk of providing services with a similar degree of risk as the reference services. ¹²¹² This should not affect their systematic risk of supplying regulated energy network services. However, we have not excluded APA from our comparator set due to service providers' concerns with a small comparator set.
Major Energy Users submitted that a revenue cap is less risky than a price cap and would lead to an overestimated beta. ¹²¹³	This issue was previously considered in the 2009 WACC review. We considered that there was no compelling evidence to suggest that the equity beta should differ based on the form of control (revenue cap vs. price cap). ¹²¹⁴ The MEU acknowledged that there was only marginal difference between price and revenue caps on exposure to systematic risk and did not set propose to set a different equity beta based on the form on control.

AusGrid submitted an April 2018 Frontier report titled 'Estimation of certain aspects of the allowed rate of return' as part of its regulatory proposal.¹²¹⁵ The report submitted that empirical estimates have increased since the 2013 Guidelines, which warrants an equity beta of at least 0.7 and other ASX-listed infrastructure firms support an equity beta materially higher than 0.7.¹²¹⁶

We first note that Frontier appears to regard the 2013 Guidelines' point estimate of 0.7 as a starting point for this review. As discussed in Table 39, we disagree with this view because to estimate an efficient rate of return to compensate for a benchmark efficient entity with a similar degree of risk as a relevant service provider in supplying regulated

¹²⁰⁹ Energy Australia, AER – Review of the rate of return guideline –evidence sessions, 4 May 2018

¹²¹⁰ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 14.

¹²¹¹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 4.

¹²¹² For example: https://www.apa.com.au/about-apa/our-history/; http://www.duet.net.au/getattachment/ASXreleases/2015/DUET-Completes-Acquisition-of-Energy-Developments/DUET-Completes-Acquistion-of-Energy-Developments-L/DUET-Completes-Acquistion-of-Energy-Developments-Limited.pdf.aspx

¹²¹³ Major Energy Users, Review of the rate of return guidelines, December 2017, p 7–8.

¹²¹⁴ AER, Final decision: WACC review, May 2009,, p. 251–252, 341.

¹²¹⁵ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018.

¹²¹⁶ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, pp. 14–27.

energy network services. Simply starting from 0.7 would discount a range of relevant considerations.

We have reviewed this report and note that this is substantively the same as the August 2017 Frontier report considered in our November 2017 final decision for APA VTS.¹²¹⁷ Similar to the August 2017 report, the April 2018 report relies on the following material:

- Comparison of 5 year estimates to 10 year estimates¹²¹⁸
- Rolling 5 year beta estimates¹²¹⁹
- ASX-listed infrastructure firms¹²²⁰

As a result, our observations of the August 2017 report remains appropriate for informing our view on the April 2018 report:¹²²¹

- Frontier's observations of increases continue to be driven by shorter term (5 year) estimates of equity beta such as 5 year weekly estimates.¹²²² Frontier itself supported the use of longer term data and noted that 'five years of data is insufficient to provide statistically reliable estimates of beta'.¹²²³ This is consistent with our approach because longer-term data is less vulnerable to interest rate movements, market volatility and one-off events (for example, the GFC) which may unduly affect the 'true' beta of supplying the regulated energy services.
- To the extent we have regard to Frontier's 10-year estimates, they are consistent with our empirical range and do not support an increase to our range and point estimate:
 - Frontier's average of 10-year weekly firm-level estimates is 0.51 which is consistent with our empirical estimates which cluster in the 0.5–0.6 range.¹²²⁴
 - Frontier's 10-year weekly portfolio-level estimates are consistent with our portfolio estimates.¹²²⁵
- We do not consider that other ASX-listed infrastructure firms can be used to inform the equity beta of a benchmark efficient entity with a similar level of risk as a relevant service provider in providing reference services. As noted in section 0, this

¹²¹⁷ Frontier, updated rate of return parameter estimates, August 2017; AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, p. 68

¹²¹⁸ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 20.

¹²¹⁹ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 21-22.

¹²²⁰ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 25-27.

¹²²¹ AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, , p. 71.

¹²²² Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 20.

¹²²³ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 19.

¹²²⁴ Frontier, Estimation of certain aspects of the allowed rate of return, April 2018, p. 20.

¹²²⁵ Frontier estimated fixed-weight portfolio beta of 0.55 and value-weighted portfolio beta of 0.59. Henry's study supported a range of 0.38–0.71 for portfolio-level estimates with a range of 0.38–0.52 for fixed weight portfolios and 0.38–0.71 for value-weighted portfolios.

is because the risk characteristics other Australian infrastructure businesses would be very different to a firm supplying the regulated energy services.

We recognise that we rejected the August 2017 report and maintained an equity beta of 0.7 in the APA VTS decision.¹²²⁶ However, this was in the context of rejecting APA VTS's proposal a beta of 0.8, and occurred prior to this review. We have concluded that a point estimate of 0.6 is appropriate following a further consideration of the relevant evidence in this review. Our consideration of the August 2017 report is consistent with the views in this decision.

We note CEG's November 2017 report for Essential Energy proposed an equity beta of 0.8–0.9 based on the following key arguments:¹²²⁷

- Increase in empirical estimates¹²²⁸
 - The average 5 year re-levered beta of four firms ¹²²⁹ has increased to 0.81 and would lead to 0.94 if DUET is excluded.
 - Portfolio estimates consisting of the four firms have increased over time particularly for 3 year and 5 year estimates which have increased to above 0.9.
- The Black CAPM should be used to address the low beta bias associated with the SLCAPM and the AER has acknowledged in the 2013 Guidelines the bias as a reason for selecting a top of the range point estimate.¹²³⁰ Adjusting for the low beta bias (using a range of zero beta premiums)¹²³¹ supports a beta of above 0.7 and at least 0.8.¹²³²

We caution over reliance on short term estimates for informing the equity beta. This is because they can be unduly effected by factors such as market volatilities, one-off events and interest rate movements which can mask the systematic risk of a firm supplying regulated energy services. As noted in 8.3.2, our empirical study, which is based on a variety of estimation periods, supports an empirical range of 0.4–0.8 and a point estimate of 0.6.

In section 8.3.5, we discussed that the low beta bias and Black CAPM are different concepts. We did not, and do not, give weight to low beta bias either in the 2013 Guidelines or in this review. We have also further considered the Black CAPM and are not persuaded to select an equity beta towards the upper end of the observed empirical range due to less confidence in the model, empirical issues and lack of use in

¹²²⁶ AER, Final decision APA VTS gas access arrangement 2018 to 2022 Attachment 3–Rate of return, November 2017, , p. 71.

¹²²⁷ CEG, WACC parameter estimates for Essential Energy, November 2017, p. 25.

¹²²⁸ CEG, WACC parameter estimates for Essential Energy, November 2017, p. 25.

¹²²⁹ APA, DUET, Spark Infrastructure (SKI), AusNet Services (AST)

¹²³⁰ CEG, WACC parameter estimates for Essential Energy, November 2017, p. 30.

¹²³¹ Zero beta premiums are estimated as part of implementing the Black CAPM. This is added to the risk free rate to form the zero beta return which is the intercept in the Black CAPM.

¹²³² CEG, WACC parameter estimates for Essential Energy, November 2017, p. 33.

practice. CEG's table of various zbp/MRP estimates range from 0.5 to 5.57 reinforces our view on the model's shortcomings such as empirical instability and sensitivity to the choice of inputs.

8.4 Summary of submissions and responses

Key point	Submission	Stakeholder	AER response
Different betas should be set for gas and electricity businesses	Gas and electricity businesses face different risks.	APGA ¹²³³ , Stephen Gray and Greg Houston	We consider systematic risk are sufficiently similar between businesses to warrant one benchmark and beta. This is because of their intrinsic business risk (supply of monopoly service) and similar regulatory
	There are no strong theoretical reasons for believing that the asset β s of regulated electricity and gas businesses should be the same, or that β s are the same across transmission and distribution. Those β s should be estimated from stock market data, but that will not be without its challenges.	APA ¹²³⁴	risk).Partington and Satchell also noted difficulties in quantifying any difference.
	The NZCC has set a higher beta for gas businesses	APGA ¹²³⁵	The NZCC noted that none of its reasons justified an uplift alone and its reasons appears to be NZ- specific and not relevant to a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services.
Empirical beta estimates have increased the 2013 Guidelines	The AER's update indicate increase in empirical estimates.	APGA ¹²³⁶ ENA ¹²³⁷ ¹²³⁸ NSG ¹²³⁹	We observe some increase in empirical estimates since the 2013 Guidelines. However, we give most weight to the longest estimation period which indicates marginal
	CCP16 is sceptical of finding	CCP16 ¹²⁴⁰	movement from 2013. It is also not clear that this represents an increase in the systematic risk of

Table 40 Summary of submissions and responses on equity beta

¹²³³ Australian Pipeline and Gas Association, Submission to the Issues Paper: AER review of the rate of return guideline, 12 December 2017, p. 4

¹²³⁴ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 17

¹²³⁵ Australian Pipeline and Gas Association, Submission to the Issues Paper: AER review of the rate of return guideline, 12 December 2017, p. 4

¹²³⁶ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 3.

¹²³⁷ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46.

¹²³⁸ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 55.

¹²³⁹ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.

¹²⁴⁰ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

Key point	Submission	Stakeholder	AER response
	from the networks' consultants that equity betas have increased in recent years		 supplying regulated energy network services: Consultants retained by service providers previously acknowledged that the increase is driven by gearing movements. Interest rate movements may drive changes due to the comparator firms being bond proxies Partington and Satchell also advised that the de/re-leveraging process and changes in individual characteristics of a firm may drive movements
The comparator set is not representative	The comparator set do not represent the BEE as some firms contain substantial unregulated businesses	CRG ¹²⁴¹ MEU ¹²⁴²	We consider, ideally, firms that share all or most of the key characteristics of a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy
	The comparator set is not representative of gas businesses with only 1 firm left	APA, ENA ¹²⁴³	services would be used when conducting our empirical analysis to estimate the equity beta. However, in practice, few firms would fully reflect this benchmark. Therefore we use market data for domestic businesses that are considered to be
	The Beta should not be based on unregulated firms and that it has been at the higher end of its potential range.	CRG ¹²⁴⁴	reasonable comparators to the benchmark efficient entity to inform the equity beta estimate.
			We disagree.
The comparator set should be expanded	It is too small and needs to be bolstered by international and other Australian infrastructure	ENA ¹²⁴⁵ CKI ¹²⁴⁶ Jemena ¹²⁴⁷ ENA ¹²⁴⁸	Our view is that the comparator firms need to reflect information from firms that are most comparable to a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services we regulate. The most relevant data for estimating equity beta is domestic energy network firms.
firms		Stepnen Gray	A number of differences with a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services (regulatory framework, business risk, etc.) means that international firms and other Australian infrastructure firms should not be included in the

¹²⁴¹ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 26.

¹²⁴² Major Energy Users Inc, Review of the rate of return guidelines issues paper submission by the Major Energy User Inc, December 2017, p. 12.

¹²⁴³ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 18, ENA, Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

 $^{^{1244}\;}$ AER, Summary of 11 December 2017 Session with CRG, 11 December 2017,

¹²⁴⁵ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31,

¹²⁴⁶ Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, p. 5.

¹²⁴⁷ Jemena, Submission on concurrent expert sessions and discussion papers, 4 May 2018, p. 3.

¹²⁴⁸ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46, 62.

Key point	Submission	Stakeholder	AER response
			comparator set.
	Caution the inclusion of international estimates and other Australian infrastructure firms if the new data is not particularly relevant to the benchmark efficient entity	CCP16 ¹²⁴⁹	We agree.
	Narrow set of firms is of itself an [insufficient] rationale to include additional comparators'	NSG ¹²⁵⁰	We agree.
	The AER's sample of domestic comparators has further reduced and now numbers only three. Logically, as the sample of close domestic comparators reduces, relatively more weight must be given to the other relevant evidence.	ENA ¹²⁵¹	If we were to expand comparator set, this must be after satisfying that the additional firms are appropriately similar to a benchmark efficient entity with a similar degree of risk to the relevant service provider in the provision of regulated energy services. We consider neither international firms nor other Australian infrastructure firms meet this.
	It is necessary to weigh up the high statistical reliability of that evidence (because the sample size is so much larger than the three domestic comparators) against the fact that the evidence is not as directly applicable to the BEE, which operates exclusively in the Australian market	ENA ¹²⁵²	Our assessment of international and other
	The international comparators have the benefit of being energy network businesses and providing an energy network service, but the disadvantage of operating in a different market. This leads to another set of comparators – other domestic infrastructure firms. These firms operate in the Australian market in an industry that is close to, but	ENA ¹²⁵³	are not comparable to a BEE with a similar degree of risk to the relevant service provider in the provision of regulated energy services.

¹²⁴⁹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018

¹²⁵⁰ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.

¹²⁵¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

¹²⁵² Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62

¹²⁵³ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62

Key point	Submission	Stakeholder	AER response
	not exactly equivalent to, the provision of energy network services		
	There is no simple mathematical adjustment which might be applied to those international data to make them comparable with data from the extant domestic comparators	APA ¹²⁵⁴	We agree.
	the use of international data may increase statistical precision, but at the cost of biased estimates of β	APA ¹²⁵⁵	We agree.
	If the set of comparable businesses is to be expanded, then careful consideration should be given to other Australian infrastructure businesses before turning to possible international comparators.	APA ¹²⁵⁶	We disagree with expanding the comparator set to
	Neither the international data nor the data on Australian 'infrastructure stocks' make suitable comparator data for informing the point estimate of the differences between these stocks and the characteristics of the BEE.	CCP16 ¹²⁵⁷	include other Australian infrastructure firms as a range of differences with a benchmark efficient entity with a similar degree of risk as a relevant service provider in the provision of regulated energy services makes them unsuitable for informing the equity beta parameter.
	It is very difficult to effectively 'normalise' international data, and it should be used only for information.	CCP16 ¹²⁵⁸	
	The views of the investors' experts lead CCP16 to have concerns with the use of Australian infrastructure	CCP16 ¹²⁵⁹	

¹²⁵⁴ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 19

¹²⁵⁵ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 19

¹²⁵⁶ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 19

¹²⁵⁷ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68

¹²⁵⁸ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹²⁵⁹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

Key point	Submission	Stakeholder	AER response
	businesses given, inter alia, many are privately owned in complex structures.		
	There is value in considering the broader Bloomberg Utilities Index which includes but goes beyond the three remaining listed networks.	CCP16 ¹²⁶⁰	We agree.
	It is important to investigate the characteristics of any comparator firms as well as the changes in the structures of the existing network firms.	CCP16 ¹²⁶¹	We agree.
	The weight consideration given to the comparators should be reduced as their congruence with characteristics of the BEE reduces. For example:		
	• Delisted Australian network firms could be considered if compared to equity betas of the three comparator firms up to the time that the firm de-listed. Once a firm is delisted, the equity beta will not change. If this does not occur, emerging risks and market cycles are unlikely to be properly accounted for.	NSG ¹²⁶²	We consider de-listed firms can still provide (historically) accurate and reliable information. We consider International estimates carry a range of differences with the BEE with a similar degree of risk to the relevant service provider in the provision of regulated energy services so we only use it as a cross checking tool.
	 International energy network firms should only be used if considered in the context of varying regulatory, economic and political environments between geographies and should be calibrated to the three Australian comparators over time. 		
	Reliance on Australian	QCOSS ¹²⁶³	We disagree. We consider empirical estimates from

¹²⁶⁰ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹²⁶¹ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹²⁶² Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15

Key point	Submission	Stakeholder	AER response
	regulated energy utility information makes the calculated WACC inherently less reliable.		firms that mostly close resemble a BEE with a similar degree of risk to the relevant service provider in the provision of regulated energy services should be main determinant in informing
	The broader problem is that the equity beta is set based on a narrow comparator group and can easily be distorted by a few data points. In our view it is it is more appropriate to assess the reasonableness of the equity beta and the overall rate of return based on observations of whether networks are over- investing than on the basis of a very restricted range of Australian energy utility firms. The pattern of investment of networks in the market is a supporting reliable guide to setting the equity beta and rate of return than observations of a restricted group of near and not-so-near comparators.	QCOSS ¹²⁶⁴	The equity beta.
De-listed firms should be excluded from comparator set	They do not reflect prevailing market conditions	CKI ¹²⁶⁵ ENA ¹²⁶⁶ 1267	We consider that they still provide (more) reliable and (historically) accurate information on the systematic risk of a BEE with a similar degree of
	The evidence presented in the AER's Equity Beta Discussion Paper is that the beta estimates for all live firms have increased since 2013, but (of course) the beta estimates for the delisted firms have remained frozen. When both sets are included in the sample, the 'dead' firms dampen the increase that the AER has documented for the 'live' firms.	ENA ¹²⁶⁹	risk as a relevant service provider in the provision of regulated energy services. Neither could be said with certainty for international energy firms and other Australian infrastructure firms. Experts at the concurrent expert evidence session agreed that equity beta for is relatively stable because the true systematic risk is likely to be stable. ¹²⁶⁸ To the extent systematic risk remains stable, de-listed firms should provide useful information for informing the equity beta parameter. Further, we estimate a long term WACC, this supports the use of long term data.
Return on equity should be adjusted	Actual returns for low beta stocks tend to exceed	APGA, CKI, ENA, Ergon,	We acknowledge that ex-post return data can indicate that actual returns exceed expected returns

¹²⁶³ Queensland Council of Social Service, Submission on review of Rate of Return Guideline, May 2018, p. 18

¹²⁶⁴ Queensland Council of Social Service, Submission on review of Rate of Return Guideline, May 2018, p. 18

¹²⁶⁵ Cheung Kong Infrastructure, AER Issues Paper – Review of the rate of return guideline, 12 December 2017, p. 5.

¹²⁶⁶ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31.

¹²⁶⁷ Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 62.

¹²⁶⁸ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 51.

¹²⁶⁹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 62

Key point	Submission	Stakeholder	AER response
for low beta bias	expected returns	Energex, NSG, SAPN, Vic Power Networks, AGN ¹²⁷⁰ 1271 1272 1273	 for low beta stocks. However, given the lack of substantively new material, our considerations for giving no weight to low beta bias in our 2013 Guidelines and subsequent regulatory decisions also remain relevant: Many of the tests and exercises which indicate low beta bias are themselves the subject of ongoing academic debate and carry limitations which throws doubt on their results and suitability for our regulatory task¹²⁷⁴ There are a number of explanations (for example, economic conditions) that do not imply a bias in equity beta¹²⁷⁵ It is also not clear that the low beta bias exists on an ex-ante basis or accounted for by investors and market practitioners on the same
	(More explicit) adjustment should be made for the low beta bias	Evoenergy. APA, ENA ¹²⁷⁶	We did not adjust for low beta bias in the 2013 Guidelines and we maintain that approach in this review.
	Experts were in unanimous agreement on the existence of low beta bias and 'ignoring' it due to concerns with the empirical reliability of the Black CAPM would not be prudent.	APGA ¹²⁷⁷	We note that we set an ex-ante rate of return. It is not clear that the low beta bias exists on an ex-ante basis or accounted for by investors and market practitioners on the same basis. There are also issues with ex-post tests of performance. The Black CAPM and low beta bias are two different concepts. We did not and do not give

- ¹²⁷⁰ Australian Pipeline and Gas Association, Submission to the Issues Paper: AER review of the rate of return guideline, 12 December 2017, p. 9; Cheung Kong Infrastructure, AER Issues Paper Review of the rate of return guideline, 12 December 2017, p. 5; Energy Network Australia, AER Rate of return guidelines response to issues paper, 12 December 2018, p. 31; Ergon Energy and Energex, AER Issues paper review of the rate of return guidelines Ergon Energy and Energex submission, 12 December 2017, p. 6; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 59; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 58.
- ¹²⁷¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9.
- ¹²⁷² SAPN, Victorian Power Networks, Australian Gas Infrastructure Group, AER Discussion Papers Review of the Rate of Return Guideline, 4 May 2018, p. 6.
- ¹²⁷³ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 15.
- ¹²⁷⁴ AER, Rate of Return Guidelines Explanatory Statement Appendices, December 2013, pp. 11–12; For example, SAPN final decision, p. 288.
- ¹²⁷⁵ For example, SAPN final decision, p. 285
- ¹²⁷⁶ Evoenergy, Review of rate of return guideline –evidence sessions, 4 May 2018, p. 3; APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018; Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9.
- ¹²⁷⁷ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 8.

AER risk unbalanced AFR risk unbalanced We consider all submissions, materials and evidence on their strengths, weaknesses and suitability for our regulatory task. This necessarily entail giving the relevant weight to evidence that we consider are relevant for our rads. Image: the fama French Model. APGA ¹²⁷⁸ We need to consider all submissions, materials and evidence on their strengths, weaknesses and suitability for our regulatory task. This necessarily entail giving the relevant weight to evidence that we consider are relevant for our task. Image: the fama French Model. APGA ¹²⁷⁸ We need to consider what is causing actual returns to differ from expectations and if these factors are priced by investors on an ex-ante basis. Partington and Satchell has noted that this could be due to aconomic shocks, business cycle and other factors in respect of the low beta bias. Image: the problem the AER faces in respect of the low beta bias is to use actual returns to adjust models like the CAPM. ¹²⁸⁰ APGA ¹²⁸¹ The AER should consider the empirical results of low beta bias (even if they can be wide ranging) instead of using judgement or theory. ¹²⁸⁴ The APG Area to the expected rate of return. APGA ¹²⁸⁶ APGA for an example active theory is most recent proposal. ¹²⁸⁶ APGA ¹²⁸⁶ It is not clear that the low beta bias exists on an example active and returns to the expected rate of return. Image: the problem the AER faces in response. ¹²⁹⁶ It is not clear that historical outperformance warrant adjust models like the CAPM. ¹²⁸⁶ APGA ¹²⁸⁶ Image: the problem theory inde exec set	Key point	Submission	Stakeholder	AER response
AER risk unbalanced treatment of evidence if it adopts Stephen Satchell's advice on interest rate movements when it rejected the Fama French Model.APGA ¹²²³ We consider all submissions, materials and evidence on their strengths, weaknesses and suitability for our regulatory task. This necessarily entail giving the relevant weight to evidence that we consider are relevant for our task.Image: Image: Ima				consideration to the Black CAPM to account for the low beta bias.
Ignoring actual returns entirely because actual returns are different to expectations does not appear to be an adequate response. ¹²⁸⁹ It noted that the 'usual' way to address the problem the AER faces in respect of the low beta bias is to use actual returns to adjust models like the CAPM. ¹²⁸⁰ APGA ¹²⁸¹ We need to consider what is causing actual returns to differ from expectations and if these factors are priced by investors on an ex-ante basis. Partington and Satchell has noted that this could be due to economic shocks, business cycle and other factors in their previous advice. ¹²⁸² Mr Ilan Sadeh noted this could be due to outperformance on items associated with non-systematic risks such as operating expenditure. ¹²⁸³ As a result, it is not clear to adjust models like the CAPM. ¹²⁸⁰ The AER should consider the empirical results of low beta bias (even if they can be wide ranging) instead of using judgement or theory. ¹²⁸² Mr Ilan Sadeh noted that sapproach we have seen in the AVEGA noted that best approach we have seen in the Australian regulatory context is that undertaken by HoustonKemp for Multinet in its most recent proposal. ¹²⁹⁸ APGA ¹²⁸⁶ Disregarding the low beta bias would decrease investor returns which may deter investment and would not be to sams which may deter investment and would not prevent and would not prevent and continue to expected returns.Disregarding the low beta bias would decreases investor returns which may deter investment and would not press of consumers. ¹²⁹⁷ APGA ¹²⁸⁶ APGA ¹²⁸¹ APGA ¹²⁸⁶ Disregarding the low beta bias would decreases of investment and would not be in the long run interests of consumers. ¹²⁹⁷ Disregarding the lo		AER risk unbalanced treatment of evidence if it adopts Stephen Satchell's advice on interest rate movements when it rejected the Fama French Model.	APGA ¹²⁷⁸	We consider all submissions, materials and evidence on their strengths, weaknesses and suitability for our regulatory task. This necessarily entail giving the relevant weight to evidence that we consider are relevant for our task.
The AER should consider the empirical results of low beta bias (even if they can be wide ranging) instead of using judgement or theory. ¹²⁸⁴ The APGA noted that best approach we have seen in the Australian regulatory context is that undertaken by HoustonKemp for Multinet in its most recent proposal. ¹²⁸⁵ APGA ¹²⁸⁶ It is not clear that the low beta bias exists on an ex- ante basis or is accounted for by investors and market practitioners on the same ex-ante basis. Our analysis of broker and valuation reports support this.Disregarding the low beta bias would decrease investor returns which may deter in the long run interests of consumers. ¹²⁸⁷ APGA ¹²⁸⁶ We disagree with the use of ex-post data for a number of reasons including interest rate movements and economic shocks can all lead to ex-post returns diverging from expected returns. We have responded to the HoustonKemp report in the draft decision for Multinet and continue to disagree for similar reasons.The Guideline shouldAPGA ¹²⁸⁰		Ignoring actual returns entirely because actual returns are different to expectations does not appear to be an adequate response. ¹²⁷⁹ It noted that the 'usual' way to address the problem the AER faces in respect of the low beta bias is to use actual returns to adjust models like the CAPM. ¹²⁸⁰	APGA ¹²⁸¹	We need to consider what is causing actual returns to differ from expectations and if these factors are priced by investors on an ex-ante basis. Partington and Satchell has noted that this could be due to economic shocks, business cycle and other factors in their previous advice. ¹²⁸² Mr Ilan Sadeh noted this could be due to outperformance on items associated with non-systematic risks such as operating expenditure. ¹²⁸³ As a result, it is not clear to us that historical outperformance warrant adjustment to the expected rate of return.
Disregarding the low beta bias would decrease investor returns which may deter investment and would not be in the long run interests of consumers.1287APGA1288We disagree with the use of ex-post data for a number of reasons including interest rate movements and economic shocks can all lead to ex-post returns diverging from expected returns. We have responded to the HoustonKemp report in the draft decision for Multinet and continue to disagree for similar reasons.The Guideline shouldAPGA1290		The AER should consider the empirical results of low beta bias (even if they can be wide ranging) instead of using judgement or theory. ¹²⁸⁴ The APGA noted that best approach we have seen in the Australian regulatory context is that undertaken by HoustonKemp for Multinet in its most recent proposal. ¹²⁸⁵	APGA ¹²⁸⁶	It is not clear that the low beta bias exists on an ex- ante basis or is accounted for by investors and market practitioners on the same ex-ante basis. Our analysis of broker and valuation reports support this. Further the low beta bias could be attributed to a number for factors (economic shock, outperformance on opex) that do not necessarily warrant compensation through the rate of return.
The Guideline should APGA ¹²⁹⁰		Disregarding the low beta bias would decrease investor returns which may deter investment and would not be in the long run interests of consumers. ¹²⁸⁷	APGA ¹²⁸⁸	We disagree with the use of ex-post data for a number of reasons including interest rate movements and economic shocks can all lead to ex-post returns diverging from expected returns. We have responded to the HoustonKemp report in the draft decision for Multinet and continue to disagree for similar reasons
		The Guideline should	APGA ¹²⁹⁰	

- ¹²⁷⁹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹²⁸⁰ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹²⁸¹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- Partington and Satchell, Report to the AER: Discussion of submissions on the cost of equity, 8 June 2017, pp. 29– 30.
- ¹²⁸³ Cambridge Economic Policy Associates, Rate of Return Guideline Review Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 53.
- ¹²⁸⁴ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹²⁸⁵ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹²⁸⁶ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹²⁸⁷ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 7.
- ¹²⁸⁸ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 7.

¹²⁷⁸ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.

Key point	Submission	Stakeholder	AER response
	examine and account for the low beta bias as it has been observed in practice and academic research. ¹²⁸⁹		
			The substance of the APGA's statement is that the AER should use actual returns when estimating the return on equity and the AER's use of the Sharpe-Lintner CAPM when the market is not in equilibrium is incorrect.
	There are issues with the "expected equilibrium" framework espoused by Partington and Satchell because it ignores actual returns, investors may not form equilibrium expectations, and it may not be the equilibrium indicated by the SLCAPM.	APGA ¹²⁹¹	We have determined that the Sharpe-Lintner CAPM should be used as the foundation model for estimating the return on equity because it will contribute to the achievement of the ARORO. This has been upheld by the Tribunal.
			We do not agree with the use of ex-post data to adjust the Sharpe-Lintner CAPM estimates. Such a method assumes that 'markets are efficient and in equilibrium, hence realised returns are an appropriate benchmark'. ¹²⁹² However, as noted above, realised returns can diverge from expected returns for a number of reasons (such as economic shocks, interest rate movements, etc.) which do not necessarily warrant an adjustment on an ex-ante basis.
	The AER has recognised the existence of low-beta bias and has stated that it will use the Black CAPM to inform the equity beta estimate so as to "mitigate possible low beta bias	ENA ¹²⁹³	We disagree. In the 2013 Guidelines we gave a role to the theory of the Black CAPM to capture possible market imperfections that may lead to actual returns to differ from expected returns. We did not, and do not, give weight to low beta bias.
	This observation has been referred by as "low beta bias" in the CAPM, although estimates of β are not, themselves, biased.	APA ¹²⁹⁴	We agree.
	When developing the current Rate of Return Guideline, the AER sought to "correct" this low β bias by appeal to the theory of the Black CAPM.	APA ¹²⁹⁵	We did not give weight to low beta bias in the 2013 Guidelines.

- $^{1290}\;$ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹²⁸⁹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 9.
- ¹²⁹¹ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, pp. 24
- ¹²⁹² Partington and Satchell, Report to the AER: Discussion of submissions on the cost of equity, 8 June 2017, p. 17.
- ¹²⁹³ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 59
- ¹²⁹⁴ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 24
- ¹²⁹⁵ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 25

Key point	Submission	Stakeholder	AER response
	Instead of using the Black CAPM, the AER has opted for a qualitative method of recognising the effect of low beta bias associated with use of the CAPM.	APA ¹²⁹⁶	
	Simon Wheatley proposed that the AER adjust the β for the benchmark efficient entity for the higher returns on low β companies. The adjustment should rely on empirical evidence, and not the theory underlying the Black CAPM. The method of adjustment proposed by Multinet Gas in its last access arrangement revisions proposal (which, we understand, was in part developed by Dr Wheatley) should be reconsidered by the AER.	APA ¹²⁹⁷	It is not clear that the low beta bias exists on an ex- ante basis or is accounted for by investors and market practitioners on the same ex-ante basis. Our analysis of broker and valuation reports support this. Further the low beta bias could be attributed to a number for factors (economic shock, outperformance on opex) that do not necessarily warrant compensation through the rate of return.
The Black CAPM	The AER should have continued regard to the Black CAPM.	ATCO gas, Spark, APA ¹²⁹⁸ Shortcomings identified in the 2013 G continue to indicate that the Black CA unsuitable for our regulatory task incl	Shortcomings identified in the 2013 Guidelines continue to indicate that the Black CAPM is unsuitable for our regulatory task including unreliability, not widely used in practice and
	The AER should maintain the weight to the Black CAPM.	ENA ¹²⁹⁹	unrealistic assumptions. These shortcomings have been reinforced by material and evidence that we have received since the 2013 Guidelines. Partington and Satchell have consistently advised against the Black CAPM since the Guideline. If we cannot have confidence in the model's results and it is not used by practitioners and investors, then we do not consider it appropriate to select an equity beta towards the top of the observed empirical estimate for this model.
	The AER is assuming that the "zero beta" rate is 2.8 per cent higher than the risk free rate.	CRG ¹³⁰⁰	We used the theory of the Black CAPM to select a point estimate towards upper end of the range in the 2013 Guidelines.

¹²⁹⁶ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 26.

¹²⁹⁷ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 27.

¹²⁹⁸ ATCO gas, Re: Review of rate of return guideline-issues paper, 12 December 2017, p. 8–9; Network Shareholder Group, Re: Response to issues paper on the review of the rate of return guideline, 12 December 2017, pp. 9-10; APA, Review of the rate of return guidelines APA submission responding to AER issues paper, 12 December 2017, p. 11.

¹²⁹⁹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9.

¹³⁰⁰ CRG, CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 50.

Key point	Submission	Stakeholder	AER response
	That is an extremely high value — it not only assumes the investor can't access funds at the risk free rate, but that the spread they face is about that of a BB rated entity.		
	Accordingly the CRG believes that little to no weight should be given to the Black CAPM in determining the beta for the benchmark efficient entity.		
	To date, the AER has not specified what adjustment it has made in relation to the Black CAPM evidence, nor what slope it considers to be reasonable that if the same adjustment was made to a starting point beta estimate above 0.5, the final beta allowance would be above 0.	ENA ¹³⁰¹	We have transparently set out the considerations we gave to the theory of the Black CAPM in the 2013 Guidelines and subsequent regulatory decisions: it is used to select a point estimate towards the upper end of the empirical range.
	As the AER has pointed out, considerable difficulties associated with obtaining reliable estimates of the return on the zero-beta portfolio.	APA ¹³⁰²	We agree with APA that there are a range of difficulties with obtaining reliable estimates of the return on the zero-beta portfolio.
	There is too much uncertainty around the empirical analysis of the Black CAPM theory for it to play a substantive role in the AER's decision, and is not generally applied by market practitioners or regulators.	CCP16 ¹³⁰³	We agree.
	The theory of the Black CAPM/low beta bias: CCP16 notes the theory of the Black CAPM/low beta bias and its derivation based on ex-post empirical assessment of actual outturns. However, this is not an unbiased estimate of ex-ante expectations.	CCP16 ¹³⁰⁴	We note that the low beta bias and Black CAPM are two different concepts but both use ex-post return data to make adjustments to parameters. We agree that the Black CAPM and the low beta bias should not be relied on to select a point estimate the upper end of our observed empirical range.

¹³⁰¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 60

¹³⁰² APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 25

¹³⁰³ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68

¹³⁰⁴ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

Key point	Submission	Stakeholder	AER response
	Moreover, there is significant variability in the proposed zero beta estimates, a variability that is consistent with the fact that neither the market nor regulators generally rely on the estimate. We do not, therefore, consider it is particularly suitable for estimating ex-ante the equity beta or for 'adjustment' to the empirical data.		
	It is possible to derive almost any value for the ROE ex ante, and this decision could be overly influenced by prior assumptions on the value of ROE.	CCP16 ¹³⁰⁵	We disagree with CCP16's view that one could derive almost any value for the ROE ex-ante. We estimate the risk free rate using the prevailing yield on the 10-year CGS averaged over a pre-specified period. We give foremost weight to market data (empirical estimates for equity beta and historical excess returns for MRP) when estimating equity beta and MRP. We agree that we should not select an equity beta towards the upper end of our empirical observed range for the (theory of the) Black CAPM.
	CCP16 concluded, therefore, that the Black CAPM is not an unbiased estimator of expected returns and is not suitable for selecting either a range or point estimate of expected returns. CCP16 also noted that neither the theory of the Black CAPM nor the associated estimates of the zero beta premium have been used by investment practitioners.		
	The Black model is inconsistent with other models such as the foundation model used by the AER.	QCOSS ¹³⁰⁶	
Firm and portfolio estimates	The estimates indicate a difference between the β for the one (predominantly) pipeline service provider, APA Group, and (predominantly) electricity service providers Spark Infrastructure and AusNet Services. This apparent difference is a	APA ¹³⁰⁷	We note that the difference would likely be driven by APA undertaking a range of transactions that would increase its exposure to systematic risk from unregulated assets and/or assets that are different from the risk of providing services with a similar degree of risk as the reference services. ¹³⁰⁸ Evidence has not been provided to persuade to us that the systematic risk of supplying regulated gas and electricity services are sufficiently different to

- ¹³⁰⁵ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70
- ¹³⁰⁶ Queensland Council of Social Service, Submission on review of Rate of Return Guideline, May 2018, p. 18
- ¹³⁰⁷ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 23

¹³⁰⁸ <u>https://www.apa.com.au/about-apa/our-history/; http://www.duet.net.au/getattachment/ASX-releases/2015/DUET-Completes-Acquisition-of-Energy-Developments-L/DUET-Completes-Acquisition-of-Energy-Developments-Limited.pdf.aspx</u>

Key point	Submission	Stakeholder	AER response
	caution against the use of β estimates of portfolios of extant businesses for estimation of either the return on equity of electricity service providers or gas pipeline service providers.		warrant separate equity beta estimates.
De/re-levering	Supports the current approach of de/re-levering. ¹³⁰⁹	ENA, NSG and APA. ^{1310 1311 1312} Stephen Gray, Simon Wheatley and Ilan Sadeh ¹³¹³	We note that a firm's gearing level affects its equity beta estimate and our comparator firms have gearing that deviate from the benchmark level of 60 per cent. This requires de/re-levering to ensure they are on a like-for-like basis with the benchmark gearing to inform the systematic risk of a
	Concerns about the re- leveraging process due to the leverage ratio as the common practice of market value of equity and book value of debt is not always a reasonable approximation.	Graham Partington ¹³¹⁴	risk as a relevant service provider in respect of the provision of regulated services. As a result, we maintain the decision to de-lever and then re-lever equity beta estimates to 60 per cent gearing for our empirical analysis.
	The AER uses the Miles and Ezzell formula for de/re-levering.	Stephen Gray, Simon Wheatley, Ilan Sadeh and ENA ¹³¹⁵	We use the Brealey Myers formula as shown in the 2013 Guidelines. We observe the Miles and Ezzell formula in the joint expert report and it differs from the Brealey-Myers formula. It appears that the other experts' and ENA's submission may not take issue with the actual formula we use but rather the name.
	The AER uses the Brealey Myers formula.	Graham Partington	We reiterate that we use the Brealey–Myers formula for de/re-levering equity beta estimates.
	In APA's view, the issues raised by Associate Professor Partington with de/re-levering are all valid. How they might be addressed – if, indeed, they need to be – is less clear. The list of alternative	APA ¹³¹⁶	As discussed in section 8.3.6, we consider it appropriate to de/re-lever equity beta estimates to allow like-for-like comparison between firms with different levels of actual gearing. We also consider to use a debt beta of zero as it appears market practice.

¹³⁰⁹ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 14

¹³¹⁰ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 21.

¹³¹¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9, 49.

¹³¹² Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 14

¹³¹³ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 11, 12, 13.

¹³¹⁴ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 7.

¹³¹⁵ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46, 50

¹³¹⁶ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 21

Key point	Submission	Stakeholder	AER response
	models for delevering and relevering is long, the evidence on debt β s is much less extensive and less robust than the evidence on equity β s, and abandoning the assumption of constant leverage raises difficult questions about what should be assumed.		
	CCP16 questions whether the benefits outweigh the risks in practice of de/re-levering, given that: The actual gearing is relatively close to the BEE target of 60 per cent; and The actual leverage does not appear to affect the network' credit ratings (within a reasonable range of 60 per cent to 75 per cent leverage).	CCP16 ¹³¹⁷	We note that a firm's gearing level affects its equity beta estimate and our comparator firms have gearing that deviate from the benchmark level of 60 per cent. This requires de/re-levering to ensure they are on a like-for-like basis with the benchmark gearing to inform the systematic risk of a
	The use of leverage to 'normalise' the gearing to the regulatory 60 per cent raises many questions as indicated by the debate in the concurrent evidence session. CCP16 considers Partington and others have a good case to use the raw estimates of beta for the networks, although this raises other issues in the context of expanding the comparator set.	CCP16 ¹³¹⁸	benchmark efficient entity with a similar degree of risk as the relevant service provider in respect of the provision of regulated services. As a result, we maintain the decision to de-lever and then re-lever equity beta estimates to 60 per cent gearing for our empirical analysis.
Estimation period	The changes in β appear, to APA, to be dampened by the use of the longest available data series for estimation. The experts were agreed that long data series provide more precise estimates but, if beta were changing, the use of shorter series for estimation might be appropriate.	APA ¹³¹⁹	We consider it is helpful to use a variety of sampling periods and estimation techniques in our empirical study to estimate equity betas for firms we consider most comparable to the firms/assets we regulate. We have most weight to long term estimates to mitigate short term volatilities and one-off events that could obscure the true equity beta of a firm in the supply of regulated energy network services. We consider that the concurrent expert evidence

¹³¹⁷ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68

¹³¹⁸ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹³¹⁹ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 24

Key point	Submission	Stakeholder	AER response
	Weight might be given to the data from delisted firms to augment the size of the sample for β estimation. But β variation over time cannot be ignored, and older data become less relevant if estimates are to reflect prevailing financial market conditions. β estimation using five years of monthly data should allow closer examination of changes, if any, over time		session supported our estimation periods as experts noted that equity beta is relatively stable as true systematic risk is likely to be stable. ¹³²⁰ As a result we consider it appropriate to maintain the current estimation periods. ¹³²¹
	Submissions from non- network stakeholders have generally cautioned against reliance on short term estimates of equity beta.	CCP16, Ian McAuley, Canegrowers	
	Experts noted that a long data series are likely to produce the most statistically reliable estimates while others noted that a shorter period (e.g. 5 years) may reflect more recent movements.	Expert joint report ¹³²³	
Debt beta	Debt beta would have no material impact on the final estimate of equity beta	Stephen Gray, Ilan Sadeh, Simon Wheatley, ENA ¹³²⁴	The debt beta should continue to be zero. It appears market practice to assume a debt beta of zero and Damodaran has made the same assumption in his works. Further, the debt beta of
	It is common to use a debt beta of zero in practice	Sadeh, ENA ¹³²⁵	businesses in the provision of regulated energy network services are likely to be low due to the relatively low risk of provisioning regulated energy
	A debt beta of zero would lead to an upward biased beta	Graham Partington ¹³²⁶	network services.

¹³²⁰ Cambridge Economic Policy Associates, Expert Joint Report, 21 April 2018, p. 51.

¹³²¹ AER, Rate of Return Guidelines Explanatory Statement Appendices, December 2013, p. 49.

¹³²² Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 96; Ian McAuley, Submission to Australian Energy Regulator on rate of return guidelines, December 2017, p. 3; Canegrowers, Submission to AER review of the rate of return guideline, 19 December 2017, p. 4; Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 83.

¹³²³ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 50.

¹³²⁴ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46.

¹³²⁵ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 46.

¹³²⁶ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 1, 15 March 2018, p. 41

Key point	Submission	Stakeholder	AER response
	estimate.		
Use other information for equity beta	While firms like the BEE do face systematic risk, they have mechanisms available to them to manage this which are not available to other firms As all firms operating in the competitive market face these risks, the market data used to generate the parameters in the CAPM include full exposure to all these risks.	CRG ¹³²⁷	We consider that the use of market data is appropriate for estimating equity beta. The rate of return needs to have regard to the prevailing conditions in the market for equity funds ¹³²⁸ and that is best measured through market data. Further, the rate of return needs to reflect the efficient cost of finance. We consider that efficient
	David Johnstone raised using regulated cashflows to		financing costs are more likely to be reflected in the prevailing market cost of capital.
			stock market data. ¹³²⁹
	It is not clear why equity beta is an acceptable surrogate on which to base the fundamental risks faced by a network as share price volatility for defensive stocks such as networks is driven more by share traders seeking the maximise their profitability and minimise their losses and less by the risk fundamentals of defensive stocks.	CRG ¹³³¹	Partington and Satchell have also cautioned against using non-market data (such as cashflows) for estimating equity beta. They noted there is a paucity of data (due to cashflows being reported on annual basis) and there would be many difficulties to overcome. ¹³³⁰
	Equity betas used in developing the equity beta for the BEE are overstated as the equity beta does not account for the risks that network firms can pass to consumers through the rules.	CRG ¹³³²	We have considered the impact of regulation on equity beta as part of our empirical update and selection of point estimate.
Estimation technique	The OLS estimator of bi is the sample estimate of $\beta i = cov(ri, rM)/var(rM)$. The LAD estimator does not have this correspondence with the economic meaning of the parameter being estimated.	APA ¹³³³	We maintain the use of LAD as a robustness check.

¹³²⁷ Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 45

¹³²⁸ For example, see NER 6.5.2 (g)

¹³²⁹ Cambridge Economic Policy Associates, Rate of Return Guideline Review – Facilitation of Concurrent Expert Evidence: Expert Joint Report, 21 April 2018, p. 42.

¹³³⁰ Partington and Satchell, Report to the AER: Allowed rate of return 2018 Guideline review, 25 May 2018, p. 14.

¹³³¹ CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 48

¹³³² CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 48

¹³³³ APA, Review of the rate of return guidelines APA submission responding to discussion papers and expert evidence, 4 May 2018, p. 22

Key point	Submission	Stakeholder	AER response
	The LAD estimator does not provide an estimate of β .		
	β estimation should use the Ordinary Least Squares method		
Range and point estimate	Focus on the use of market practice, advice from Partington and Satchell and reducing the weight given to the Black CAPM.	CCP16, ECA, EUAA ¹³³⁴	We place most weight on empirical estimates (based on market data) for informing equity beta. After further considerations, we are not persuaded to select an equity beta towards the upper end of the observed empirical range for the (theory of the) Black CAPM.
	More transparency in the	Ergon and Energex, Spark,	We have transparently set out our approach in the selection of range and point estimate.
	selection of range and point estimate.	Evoenergy, APGA ¹³³⁵ and Stephen Gray	We give most weight to our empirical study and international estimates and conceptual analysis are used in a cross checking role.
	The AER's current approach does not make sense because the range should not bound the information from other evidence. He noted that it would be better to set out and consider all evidence on equal footing. ¹³³⁶	Stephen Gray	We have transparently set out our assessment of and weight given to each piece of relevant evidence in accordance with step 2 of our foundational model approach. We have done so in the 2013 Guidelines, subsequent regulatory decisions, and this decision. We have set out all the relevant evidence equally in step 2 of our decision and the weight given to a piece of evidence is based on its strengths, weaknesses and suitability for our regulatory task. To give equal weight to all relevant evidence ignores the fact that some material is better suited in a qualitative role.
	Consumer groups submitted for a point estimate of 0.6 and below.	ECA, CRG, CCP16 ¹³³⁷	We have had regard to all the relevant evidence in accordance with the weight we give them after
	Service providers, investors and network associations espoused an equity beta of	ENA, Evoenergy, NSG ^{1338 1339 1340}	suitability for our regulatory task. This leads us to set a point estimate of 0.6 from a range of 0.4–0.8.

¹³³⁴ Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 89; Energy Consumers Australia, Review of the rate of return guideline response to the AER Issues Paper, December 2017, p. 23, 27; EUAA, EUAA submission – AER rate of return review issues paper October 2017, 18 December 2017, p. 9.

¹³³⁵ Ergon Energy and Energex, AER Issues paper review of the rate of return guidelines Ergon Energy and Energex submission, 12 December 2017, p. 2; Network Shareholder Group, Re: Response to issues paper on the review of the rate of return guideline, 12 December 2017, pp. 9-10; Evoenergy, Review of rate of return guideline –evidence sessions, 4 May 2018, p. 3; APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 3.

¹³³⁶ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 43.

¹³³⁷ Energy Consumers Australia, Review of the rate of return guideline response to the AER Issues Paper, December 2017, p. 27; CRG, Submission to the Australian Energy Regulator Rate of Return Guideline Review, May 2018, p. 51, 68; Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 68.
Key point	Submission	Stakeholder	AER response				
	0.7 and above. The AER should start from the 2013 Guidelines' estimated value of 0.7 and let evidence indicate any changes to this point estimate.	Stephen Gray ¹³⁴¹	 Simply starting from a value of 0.7 would discount a range of relevant considerations, including: those from the 2013 Guidelines that indicated a range of 0.4–0.7 and supported a (possible) point estimate towards the lower value within this range information we have received since the 2013 Guidelines on relevant evidence. The 2013 Guidelines selected an equity beta point estimate of 0.7 having regard to the theory of the Black CAPM and international estimates. As discussed in section 8.3.5 and 8.3.4, we are not persuaded to select an equity beta towards the upper end of our observed empirical range for the (theory of the) Black CAPM and we use international estimates qualitatively in a cross-checking role. Professor Satchell also disagreed with Professor Gray's proposal as that would discard information that supported lower values and may misconstrue the 0.7 as the mid-point of a range.¹³⁴² (Satchell, 44, 35) 				
	The remaining listed firms should be used to form an estimate and confidence interval with other information subsequently used to adjust the confidence interval to arrive at a point estimate.	APGA ¹³⁴³	 We disagree with the use of confidence intervals for reasons we have previously outlined.¹³⁴⁴ These reasons include: The presence of outliers can affect point estimates and their associated confidence intervals. The presence of autocorrelation and heteroskedasticity creates difficulties in discerning whether confidence intervals overstate or understate the upper bound estimate.¹³⁴⁵ 				

- ¹³³⁸ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 60, 63; Evoenergy, Review of rate of return guideline – evidence sessions, 4 May 2018, p. 3; APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 5.
- ¹³³⁹ CKI, AER discussion papers-review of the rate of return guideline, 4 May 2018, p. 6.
- ¹³⁴⁰ Network shareholder group, Submission on the Rate of Return Guideline (RORG) review from the Network Shareholder Group (NSG), 4 May 2018, p. 14.
- ¹³⁴¹ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 44.
- ¹³⁴² AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 44.
- ¹³⁴³ APGA, Submission to the AER review of rate of return guideline, 4 May 2018, p. 6.
- ¹³⁴⁴ AER, *Final decision: WACC review*, May 2009, pp. 286-290.
- ¹³⁴⁵ Autocorrelation is present when the errors in the regression have a relationship or trend with errors in the past. Heteroskedasticity is where the variance in the errors is not constant (over time or as the values of the independent variables change).

Key point	Submission	Stakeholder	AER response
			 Confidence intervals are less likely to represent the 'true' equity beta point estimate, compared to the range of point estimates derived from different samples and sampling periods.
	Experts discussed setting confidence interval and a point estimate for individual parameters which is then used to set a point estimate and a confidence interval for the WACC. Other (qualitative) information can be used to select the end WACC from the confidence interval (p 133	Expert evidence session ¹³⁴⁶	We consider the intent of the method discussed by experts is not dissimilar to the foundational model approach. The focus is on using other information to triangulate and perform a reasonableness check on the end estimate. Step 5 of our foundational model approach compares our return on equity against information from other relevant evidence to assess the reasonableness of our estimate.
	Submissions from non- network stakeholders have generally cautioned against reliance on short term estimates of equity beta.	CCP16, Ian McAuley, Canegrowers ¹³⁴⁷	We agree. Hence we place most weight on estimates from the longest estimation period.
	RAB multiples and profitability metrics have no useful role to play in the estimation of equity beta.	ENA ¹³⁴⁸	We agree that these measures may not at the parameter level but may apply at the overall return on equity level.
	The international evidence considered by the AER all indicates an equity beta above 0.7.	ENA ¹³⁴⁹	We now use international evidence as a cross check for our empirical range following further consideration of this evidence. Our update indicate that international estimates cluster below 1.0 and support range of 0.4–0.8.
	Evidence from other domestic infrastructure firms all indicates an equity beta above 0.7.	ENA ¹³⁵⁰	Our consideration is to give no weight to this material for informing the equity beta estimate. This is because of a range of differences with firms supplying the regulated energy services.
	The only evidence that supports maintenance of the	ENA ¹³⁵¹	We have most regard to longest period estimate which support a value of 0.5–0.6. We also note that

¹³⁴⁶ AER, Transcript of proceedings Australian Energy Regulator Office: Review of rate of return guidelines concurrent expert evidence session 2, 5 April 2018, p. 126–127, 128–133.

¹³⁴⁷ Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 96; Ian McAuley, Submission to Australian Energy Regulator on rate of return guidelines, December 2017, p. 3; Canegrowers, Submission to AER review of the rate of return guideline, 19 December 2017, p. 4; Consumer Challenger Panel (sub-panel 16), Submission to the AER on its rate of return guideline issues paper, December 2017, p. 83.

¹³⁴⁸ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

¹³⁴⁹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

¹³⁵⁰ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

¹³⁵¹ Energy Networks Australia, AER review of the rate of return guideline response to Discussion Papers and concurrent expert evidence sessions 4 May 2018, p. 9

Key point	Submission	Stakeholder	r AER response				
	same equity beta as in 2013 is the evidence from delisted firms, whose beta estimates are frozen in time forever.		de-listed firms can provide useful information.				
	We do not accept that the AER should select a figure at the top of the empirical range. CCP16's analysis demonstrates that the AER should no longer err on the side of regulatory caution. The balance between investment risk and consumer price risk has shifted, as prices have risen and the market growth has stalled. CCP16 now strongly recommends that the AER adopt a more balanced view on each of the parameters including the equity beta.	CCP16 ¹³⁵²					
	When each of the AER's discussion paper estimates is considered independently, two thirds of the above estimates are less than 0.6. In addition, the mean and median values of the estimated betas across scenarios, methods and portfolios are also less than 0.6.		We have reviewed evidence and submissions as part of this review. Our updated empirical estimate support a range of 0.4–0.8. We now have regard to a range of information to inform the point estimate. Our international estimates and conceptual analysis support our range and point estimate. We are not persuaded to select a point towards the upper end of our observed empirical range for the (theory of the) Black CAPM.				
	The AER also very usefully considered betas at a sector/industry level using Bloomberg's industry data. The Utility index, which included the 3 remaining listed networks in the top 5, indicated an equity beta of around 0. 6.	CCP16 ¹³⁵³					
	CCP16 believes that the AER can no longer ignore the weight of evidence pointing to an equity beta of less than 0.7. However, we are aware of a range of other considerations for the AER and also posited by the networks and their						

¹³⁵² Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 69

¹³⁵³ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 16

Key point	Submission	Stakeholder	AER response				
	consultants. It is essential that a long-term data series is used, as on balance and noting the risks of this, the task of the AER is to estimate ex-ante the long- term beta and to see through the ups and downs of annual data and short-term trends	CCP16 ¹³⁵⁴	We agree. As a result we give most weight to empirical estimates from the longest estimation period. We consider that short term data is more likely to be unduly affected by one-off events, interest rate movements and volatilities which may obscure the true equity beta of a firm in the supply of regulated energy services.				
	The majority of the empirical estimates of the beta for both the individual and portfolio network firms sit around a median value of 0.5 to 0.6 (using the AER's 2017 analysis), and the Bloomberg Utility index also sits around a median value of 0.5 to 0.6.	CCP16 ¹³⁵⁵	We have updated our empirical analysis to 2018. Our results indicate that estimates cluster around the 0.5–0.6 range.				
	APA's results indicated that there might be a change in the last five years in the systematic risk. This result is not, however, surprising given the substantial change in APA's revenue sources and business plans. Indeed, it would be surprising if moving more into the energy sector would not increase the equity beta of APA given the considerably higher average beta of that sector compared to the utility sector AusNet Services (AST) has also shown increase in the equity beta which is difficult to explain on the basis of fundamentals of the company.	CCP16 ¹³⁵⁶	In previous regulatory decisions, we have noted that recent movements in equity beta estimates may not necessarily reflect a change in the systematic risk of supplying the regulated energy network services. ¹³⁵⁷ This because some of the of the still-listed firms have undertaken a range of transactions that would increase their exposure to systematic risk from unregulated assets and/or assets that are different from the risk of providing services with a similar degree of risk as the reference services. ¹³⁵⁸ Further we have noted that interest rate movements would likely drive the equity beta estimates due to the comparator firms being considered bond proxies. Due to these considerations, and our concerns with short term estimates from the longest estimation period.				

¹³⁵⁴ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 70

¹³⁵⁵ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 16

¹³⁵⁶ Consumer challenger panel 16, Submission to the AER on its rate of return guideline review concurrent evidence sessions, 4 May 2018, p. 85

¹³⁵⁷ AER, APA VTS final decision Attachment 3 Rate of Return, November 2017, p. 255.

¹³⁵⁸ <u>https://www.apa.com.au/about-apa/our-history/; http://www.duet.net.au/getattachment/ASX-releases/2015/DUET-Completes-Acquisition-of-Energy-Developments-L/DUET-Completes-Acquisition-of-Energy-Developments-Limited.pdf.aspx</u>

9 Return on debt approach

In this chapter we discuss our overall approach to debt and the implementation of our approach to debt.

Our draft decision is to continue to adopt key elements of our current approach for estimating the return on debt. In particular, we will continue to:

- Adopt a benchmarking approach to estimating the allowed return on debt, which manifests in a benchmark:
 - Term of debt. We discuss the choice of the benchmark term to maturity of ten years in section 10.3.
 - Credit rating. We discuss the choice of the benchmark credit rating of BBB+ in section 10.1.
- Adopt a 10-year trailing average return on debt (based on a 10 year term to maturity) and update return on debt estimation annually.
- Adopt a 10-year transition between the previous 'on-the-day' approach and the 10year trailing average. For clarity, our draft decision is to adopt a consistent transition approach across all networks we regulate. That is, where we have commenced the transition in a previous determination for a service provider, we will continue that transition. Where we have not yet commenced the transition, we will adopt the same form of transition as has been applied to the other networks.
- Estimate the return on debt by reference to published third-party yield curves.
- For each year of the 10-year trailing average, estimate the return on debt as the simple average of rates observed over a period of time nominated by the service provider to whom the allowed return on debt will apply.

Through their submissions and other correspondence to date, the majority of stakeholders have indicated support for these aspects of our approach. This is also consistent with our position, accepted by most stakeholders, that this review should be an incremental review. We consider that this provides the necessary certainty and predictability that stakeholders have said they value whilst allowing us to discharge our regulatory task in a manner that is most likely to contribute to the legislative objectives.

However, there is one aspect of our approach where some stakeholders have submitted an alternative, being our transition into the trailing average portfolio approach. Our consideration of these submissions is detailed in sections 9.1, 9.2 and 9.3 below.

We acknowledge that legislative amendments have been proposed that would make this Guideline binding. This would be unlike the 2013 Guidelines, for which the current legislative framework allows both service providers and ourselves the opportunity to depart from the 2013 Guidelines if the evidence justified that doing so would result in an outcome that better achieves the allowed rate of return objective. However, our current approach has been to adopt an annually updating return on debt approach. Under the current rules framework, an annually updating return on debt is required to be given effect by automatic application of a formula. For this reason, our view is that implementation of a binding rate of return instrument requires only incremental changes to our current approach. On this basis, we have considered refinements to our averaging period criteria and list of contingency actions compatible with a binding rate of return instrument. These are detailed in sections 10.7 and 10.8.

9.1 Submissions on transition to the trailing average

Through their submissions and other correspondence to date, the majority of stakeholders have indicated support for maintaining a continued application of the trailing average portfolio approach with the transitioning period.¹³⁵⁹ At the same time, Network Shareholder Group¹³⁶⁰ and ATCO Gas Australia¹³⁶¹ have expressed preferences for removing the transitioning arrangement.

Our draft decision is to maintain the current transitioning arrangement which is in place for the trailing average portfolio approach.

The trailing average portfolio approach was implemented with a transitioning period of 10 regulatory years for all service providers to allow a progressive change between two different approaches of setting the allowed return on debt - a transition from an 'on-theday' approach to a 'trailing average' approach. Therefore we are currently in the midst of the transitioning period which commenced from the start of each service provider's first regulatory period after May 2014.

As we set out in our issues paper, based on the information currently before us, changing this approach will not contribute to the achievement of the national gas and electricity objectives. The subsections below outline our reasoning for maintaining the current approach to transitioning to the trailing average, and our response to submissions on the transition.

A trailing average with a transition will achieve the objectives

In chapter 2 we discussed the requirements in the national gas and electricity objectives, and the allowed rate of return objective, for our rate of return to promote efficiency. We outlined how the allowed rate of return will promote efficiency when it

¹³⁵⁹ Australian Pipelines and Gas Association, Submission to the Issues Paper, December 2017, p4 – 6 Energy Networks Australia, AER Rate of Return Guideline, December 2017, p16-17, p19-20; Ergon Energy and Energex, Issues paper – review of the rate of return guidelines, December 2017, p4-5; Major Energy Users, Review of the rate of return guidelines, December 2017, p10-11, 15; APA, APA submission responding to AER issues paper, December 2017, p8; Cheung Kong Infrastructure, Submission on rate of return issues paper, December 2017, p3, Energy Users Association of Australia, EUAA submission – AER Rate of Return Review Issues Paper, October 2017, p8; AusNet Services, Review of Rate of Return Guideline – Issues Paper, December 2017, p1; Network Shareholder Group, Submission on the RoRG review, May 2018, p.11, Consumer Reference Group, Submission to the AER RoRG review, May 2018, p.41

¹³⁶⁰ The Network Shareholder Group refers to a consortium of network investors consisting of Spark Infrastructure, Hastings, AMP Capital, AustralianSuper, Macquarie Infrastructure and Real Assets, sand IFM Investors.

¹³⁶¹ ATCO Gas Australia, Review of rate of return guideline – issues paper, December 2017, p6-7

reflects market rates and provides for ex-ante efficient compensation given the risks of providing regulated services.

We consider that a revenue neutral transition between the on-the-day approach and trailing average approach is necessary to achieve the national gas and electricity objectives, and the allowed rate of return objective. Without a revenue neutral transition our change in approach could increase a benchmark efficient entity's value, then this would benefit its equity holders at the expense of consumers. Conversely, if such changes decreased a benchmark efficient entity's value, then this would cost its equity holders but provide a short term financial benefit to consumers. As such, this methodological change may also have a negative impact on the confidence in the predictability of the regulatory regime.

We consider ex-ante efficient compensation can hold under either the on-the-day approach or the trailing average approach (if a transition is applied). As such, both approaches are capable of being approximately equivalent over the term of the RAB (which will be multiple regulatory periods).

However, under the trailing average approach, for any given regulatory period, the present value of expected net operating cash flows over the regulatory period plus the closing RAB will not necessarily equal the opening RAB. That is, at the start of any given regulatory period, the present value of expected future cash flows will unlikely equal the RAB because the cash flows based on historical interest rates will either be too high or too low (relative to the prevailing cost of debt in the market). Given this, switching between regimes without a full transition would not satisfy the requirement to provide service providers with a reasonable opportunity to recover at least efficient costs over either the regulatory period or over the term of the RAB.

As either the on-the-day or trailing average approach would contribute to the achievement of the ARORO, a switch between regimes that is accompanied by a revenue neutral transition will also contribute to the achievement of the ARORO. For a more thorough explanation, we refer to our final decision on the 2018-22 access arrangement determination for APA's Victorian Transmission System.¹³⁶²

Previously, our approach to transition to the trailing average has been extensively contested in litigation. Most recently, our approach was upheld by:

- the Australian Competition Tribunal in its decisions on appeals from SAPN, the Victorian electricity distribution network service providers
- the Full Federal Court in its decision on SAPN's appeal of the Australian Competition Tribunal's decision.

These outcomes reinforce our view that a revenue neutral transition is necessary to advance the NEO and NGO.

¹³⁶² See AER, Final Decision APA VTS gas access arrangement 2018 to 2022 Attachment 3 – Rate of return, November 2017, p327 - 347

Submissions for an immediate transition to the trailing average

In its submission on our October 2017 issues paper, the Networks Shareholder Group submitted that:¹³⁶³

...there should be no transition and there should be an immediate adoption of a trailing average methodology. The AER has acknowledged that a 10-year staggered portfolio of debt is an efficient debt strategy... there is no merit in adopting a transition which, by definition, prolongs an inefficient approach until some future date...many networks already manage their debt portfolios in this more efficient manner and may be penalised by a transition. This reduces the incentives to adopt an efficient approach unless it exactly matches the regulatory assumptions.

....NSPs have a right to an opportunity to recover efficient costs, including financing costs. A focus on revenue neutrality when it is the efficient costs, or assessment of efficiency costs, that is changing is inconsistent with this important principle under the National electricity and gas rule. A transition must consider the implications for recovering efficient costs and the incentives to deliver efficient outcomes both of which are required to promote the long-term interests of consumers.

We consider that a trailing average portfolio approach promotes efficiency by providing ex-ante efficient compensation over the term of the RAB. We have decided to adopt the trailing average portfolio approach and continue the transition to the trailing average from the on-the-day approach that was previously applied. However, we do not consider that the on-the-day approach was an inefficient approach.

The Network Shareholder Group submission refers to a 10-year staggered portfolio of debt as an efficient debt strategy and that service providers should be able to recover efficient costs.

However, as noted above, we consider that an allowed rate of return that meets the objectives must provide ex-ante compensation for efficient financing and provide a service provider with a reasonable opportunity to recover at least its efficient financing costs (sometimes referred to as the NPV=0 principle). Either the on-the-day-approach or trailing average approach will provide ex-ante efficient compensation over the term of RAB. The actual debt management practice by a service provider may differ from our regulatory approach, but this does not mean that our allowed rate of return does not provide ex-ante efficient compensation. As noted in our APA VTS determination, to the extent that there is a difference in debt management approach, interest rate risk is captured in the estimation of beta thus the overall rate of return is likely to provide efficient compensation to a service provider.

¹³⁶³ Network Shareholder Group, Response to issues paper on the review of the Rate of Return Guideline, December 2017, p8-9

One of the key factors in our decision to change from the on-the-day to the trailing average approach was a consideration that the trailing average may more closely match actual efficient financing practices, thereby reducing interest rate risk. However, this does not mean that we considered that the on-the-day approach would not provide ex-ante efficient compensation. We note that actual efficient financing practices adopted by service providers may differ from each other and that they may also change from time to time in response to factors such as market conditions, management strategies and policies, terms and conditions of legacy debt instruments, etc. Overall we consider that both the on-the-day and trailing average approaches provide ex-ante efficient compensation to satisfy the NPV=0 concept given effect in the RPPs.

Further, as we are now in a transitioning period, the majority of the service providers have adopted a debt management strategy that takes account of the regulatory approach, including the transition. Changing the current arrangement may impact on confidence in the predictability of regulatory arrangements.

We also note that the Tribunal in its decisions for ActewAGL (Gas) Distribution and Jemena Electricity Networks Ltd concluded that an on-the-day approach could be lawfully applied in those cases and that an immediate trailing average would not achieve the objectives or reflect efficient financing costs.¹³⁶⁴ Through this decision, the Tribunal concurred with our view that a non-revenue neutral position which would be brought on by an immediate change to trailing average would not satisfy the objectives.

ATCO Gas Australia submitted that the AER should consider reasoning given by ERA in moving directly to hybrid trailing average approach¹³⁶⁵. In its final determination on the access arrangement for Mid-West and South-West Gas Distribution Systems in September 2015, the ERA determined that it was more appropriate to move directly to the hybrid trailing average approach without transition because:

...it recognises that there is no change required in hedging arrangements between the previous approach and the hybrid trailing average approach, as both involved a single estimate of the risk-free rate, set once at the start of the regulatory period. For the DRP, however it is likely that the benchmark efficient firm would have adopted a portfolio of debt with a ten-year average term, and that the firm would have been reasonably recompensated over the past three access arrangements, without being excessively compensated. However, a transition on the DRP would likely introduce a shortfall 'under' for the regulated firm over the AA4 period, which could then not be recovered as the full transition to the trailing average DRP occurred in the AA5 period¹³⁶⁶.

¹³⁶⁴ AER, Position paper: Remitted debt decisions for NSW/ACT 2014-19 electricity distribution determinations and Jemena Gas Networks 2015-20 (NSW) Access Arrangement, December 2017, p15

¹³⁶⁵ ATCO Gas Australia, *Review of rate of return guideline – issues paper*, December 2017, p6-7;

¹³⁶⁶ Economic Regulation Authority, Final decision on proposed revisions to the access arrangement for the Mid-West and South-West gas distribution systems, September 2015, p.353-354

The ERA's reasoning appears based on whether the change in approach from on-theday to trailing average requires adjustments to existing hedging arrangements. Our rationale for transitioning to the trailing average is not based on the actual debt management practices of service providers. As stated above, we do not consider the actual debt management practices of service providers alters the extent to which the trailing average achieves the legislative objectives or whether a revenue neutral transition is required to achieve the legislative objectives. Rather, to the extent it is systematic, interest rate risk from differences between actual debt management practices and regulatory allowances is likely to be reflected in our equity beta estimate.

In relation to the definition of 'efficient debt financing costs', QTC sought confirmation of whether the views expressed by us in the 2017 final decision documents for APA VTS were intended to apply to the trailing average approach following the conclusion of the transition. The views in question relate to our definition of efficient debt financing costs in the context of the ex-ante nature of the regulatory scheme rather than past financing practices, and that an allowed return on debt that reflects the prevailing market cost of debt promotes efficient investment decisions¹³⁶⁷. The interpretation of 'efficient financing costs' as an ex-ante concept was approved by the Tribunal in the ActewAGL (Gas) Distribution and Jemena Electricity Networks Ltd appeals of our May 2016 determinations¹³⁶⁸. We consider that a rate of return that meets the legislative objectives must provide ex-ante compensation for efficient financing costs and this return would give a service provider a reasonable opportunity to recover at least its efficient financing costs. This is a zero net present value (or NPV=0) investment condition¹³⁶⁹. We consider that this interpretation of efficient financing costs and the NPV=0 concept will continue to be relevant in transition as well as post-transition of the trailing average approach.

9.2 NT Power & Water's proposal for immediate transition

In its regulatory proposal for the 2019-24 period, NT Power & Water Corporation proposed a return on debt estimate based on a 10-year trailing average without any transition. NT Power & Water Corporation submitted that its current allowed rate of return is effectively already set as a 10-year trailing average and so an immediate adoption of the trailing average for its 2019-24 regulatory period would not result in a windfall gain or loss to consumers.

Our decision is to commence a 10-year transition to the trailing average in the first year of NT Power & Water Corporation's 2019-24 regulatory period, on the basis that:

¹³⁶⁷ Queensland Treasury Corporation, Rate of return guideline review issues paper, December 2017, p7

¹³⁶⁸ Australian Competition Tribunal, Application by ActewAGL Distribution [2017] ACompT 2, October 2017

¹³⁶⁹ SFG Consulting advice to the AEMC during the rule change process also supports the position that setting an allowed return that results in a zero NPV investment outcome is important to achieving efficient investment incentives. SFG Consulting, *Rule change proposals relating to the debt component of the regulatory rate of return*, 21 August 2012, p63-64

- NT Power & Water Corporation's current allowed rate of return is not set via an annually updating return on debt
- An immediate transition to the trailing average is unlikely to be revenue neutral.

NT Power & Water Corporation submitted that:

We agree that a trailing average approach best serves the long-term interests of consumers. We also accept that a DNSP should not receive a windfall gain when adopting that approach – and consumers should not be asked to (effectively) pay twice for the same high period in the interest rate cycle.

However, in our circumstances, we consider that adopting the trailing average approach immediately would not provide a windfall gain because unlike all other service providers regulated by the AER that we are aware of:

- the allowed return on debt reflected in our current tariffs (~4.21%) is significantly below an on-the-day rate and when averaged with the UC determined return on debt for the prior period (8.51%) gives a value (6.36%) that is consistent with the 10-year trailing average that we propose (6.37%), and
- adopting a trailing average approach would not include rates observed during the peak of the Global Financial Crisis over 2008 and early 2009 – as the averaging period used to apply that approach need only stretch back to July 2009.

NT Power & Water Corporation submitted that the average of the allowed rates of return for its past two regulatory periods (2009-10 to 2013-14 and 2014-15 to 2018-19) give a similar result to a trailing average.

For the reasons set out in this chapter, we do not accept this submission. A key feature of our transition to the trailing average is that, in each year during which we update the trailing average portfolio, we do so by adding an estimate of debt based on the prevailing cost of debt. It is this feature of our approach that provides for revenue neutrality and satisfies the NPV=0 principle. In our view, NT Power & Water Corporation's proposed approach would be backward looking and incorporate past estimates of the cost of debt. For the reasons set out in our decision on averaging periods, we consider that selection of historical averaging period can introduce bias into outcomes. As a result, it is most likely that such an approach would lead to windfall gains or losses which would not be consistent with the NEO and NGO.

9.3 TasNetworks' proposal to alter its transition path

In its regulatory proposal for the 2019-24 regulatory period, TasNetworks proposed to align the allowed return on debt for both its distribution and transmission networks in a manner that affects their transition to the trailing average portfolio approach. Our decision is to:

 maintain the current transition paths for TasNetworks' transmission and distribution networks, and address the revenue impacts of aligning the allowed return on debt through other mechanisms available in our regulatory determination and annual pricing processes.

Previously, TasNetworks' transmission and distribution networks were regulated separately on different timetables. Now the two network decisions have been aligned. The current cost of debt for TasNetworks' transmission and distribution networks are not aligned because we commenced the transition to the trailing average portfolio approach at different times. This is because we commenced the transition at the start of the current determinations for transmission and distribution, and these determinations were made at different times.

Our current determination for TasNetworks' transmission network was made for a 2014-19 regulatory period. Our current determination for TasNetworks' distribution network was made for a 2018-19 regulatory period. Our next determinations for TasNetworks' transmission and distribution networks will both be for an aligned 2019-24 regulatory period.

Therefore, at the start of the 2019-20 regulatory year, TasNeworks' transmission network will be five years into a ten year transition, while its distribution network will be two years into a ten year transition. TasNetworks proposed to align the return on debt for its transmission and distribution networks by adopting the lower of the two. This will be the distribution network return on debt. TasNetworks proposed this adjustment to its return on debt to address consumer affordability concerns.¹³⁷⁰

TasNetworks' proposal means that, in effect, TasNetworks' transmission network will have a transition to the trailing average that is:

- longer than ten years
- not the uniform, linear transition as set out in our 2013 Guidelines (there will effectively be a 'split' in the transition path occurring between the 2018-19 and 2019-20 regulatory years).

In section 9.1 above we set out why we consider a 10-year transition will be revenue neutral and therefore contribute to the achievement of the legislative objectives. For this reason, and for the benefits from predictability and maintaining less complicated guidelines, we consider that the guidelines should maintain the start and end dates for transition paths that we have already made in past regulatory determinations.

While our decision is to apply a consistent transition approach to all networks, we recognise that TasNetworks is proposing a revenue path below what it might otherwise be entitled to in seeking to promote the long term interests of its consumers. We will explore the use of other possible mechanisms to achieve this outcome with TasNetworks as part of our determination for their 2019-24 regulatory period.

¹³⁷⁰ TasNetworks, Tasmanian Transmission revenue and distribution regulatory proposal—Regulatory control period 1 July 2019 to 30 June 2024, January 2018, p. 166.

9.4 NSW and ACT determinations to be re-made

Where we have commenced the transition to the trailing average portfolio approach in a previous determination for a service provider, we will continue that transition.

For Ausgrid, Endeavour Energy, Essential Energy, Evoenergy (electricity distribution), and Jemena Gas Networks, we initially made determinations in 2015 that commenced this transition. However, these determinations were subsequently subject to a prolonged period of appeal and review, including review of our allowed return on debt approach. The Australian Competition Tribunal remitted these determinations back to us to be remade and directed us to reconsider our return on debt approach. These remitted determinations have not yet been remade. Consequently, it is not yet clear if these determinations will set out a start date for a transition to a trailing average.

The Essential Energy remittal determination has now been finalised, but the other remittal processes are ongoing.

For the avoidance of doubt, we will treat the first year of these determinations as the commencement of a transition to the trailing average for the purposes of applying a transition in subsequent regulatory periods.

We note that - for the purposes of their 2019-24 determinations - Ausgrid, Endeavour Energy, Essential Energy, and Evoenergy have all proposed to adopt a transition into the full trailing average. These service providers have proposed that the return on debt for the first year of their 2019-24 regulatory periods commence the transition at year 6 and then transition progressive each year through the full ten year transition period. We are yet to receive a regulatory proposal from Jemena Gas Networks to indicate their views.

The treatment of the transition for the purposes of the 2014-19 period is yet to be determined. Nonetheless, the proposals from the NSW and ACT service providers indicate that for the purposes of subsequent regulatory periods - which will be the periods subject to this rate of return guideline - the use of a transition and the start date for that transition is not contested.

9.5 Return on debt incentive scheme

The Major Energy Users Inc. submitted that we should consider implementing an incentive scheme for the return on debt, stating:¹³⁷¹

While the benefits of improved reliability, opex and capex all now have incentives built into them where the benefits are to accrue to consumers, this does not apply to the cost of debt.

As noted in our chapter 2 we will apply a benchmarking approach that provides incentives service providers to outperform our allowed rate of return. We consider that

¹³⁷¹ Major Energy Users, Submission on AER Rate of Return Issues Paper, December 2017, p. 10.

the ex-ante setting of an allowed rate of return provides incentives for service providers to undertake efficient financing practices. Benefits from this approach also accrue to consumers as we undertake regular reviews of our allowed rate of return and update our benchmarks accordingly. We have done this as part of this review, and have updated our empirical analysis of gearing ratios, credit ratings, service providers' actual debt costs, and equity beta estimates. We do not consider that an additional incentive scheme is required for the return on debt or any other aspect of the allowed rate of return.

9.6 Summary of submissions

This appendix details the submissions we have received from stakeholders as part of our review process to date, and notes how we have had regard to each submission.

Key Point	Submission	Stakeholder	AER Response
The transition and trailing average approach	Support the transition to a trailing average approach	APGA, ¹³⁷² ENA, ¹³⁷³ Ergon Energy & Energex, ¹³⁷⁴ APA, ¹³⁷⁵ CKI, ¹³⁷⁶ EUAA, ¹³⁷⁷ AusNet ¹³⁷⁸	Our draft decision is to maintain the 10-year trailing average and the approach to transition set out in the 2013 Guidelines. We consider that this provides the necessary certainty and predictability that stakeholders have said they value, whilst allowing us to discharge our regulatory task in a manner that is most likely to contribute to the legislative objectives.
	It is more appropriate to skip the transition and immediately move to the trailing average approach.	ATCO, ¹³⁷⁹ Network Shareholder Group ¹³⁸⁰	We consider changing this approach will not contribute to the achievement of the national gas and electricity objectives. Our current approach on transition to the trailing average return on debt has been considered extensively in Australian Competition Tribunal and Full Federal Court decisions.
	Support the AER's ongoing commitment to the trailing	QTC ¹³⁸¹	One of the key factors in our decision to change from the on the

Table 41 Summary of submissions on the return of debt approach

- ¹³⁷³ Energy Networks Australia, AER rate of return guideline, December 2017, p19-20
- ¹³⁷⁴ Ergon Energy and Energex, Issues paper review of the rate of return guidelines, December 2017, p4-5
- ¹³⁷⁵ APA, APA submission responding to AER issues paper, December 2017, p8
- ¹³⁷⁶ Cheung Kong Infrastructure, Submission on rate of return issues paper, December 2017, p3
- ¹³⁷⁷ Energy Users Association of Australia, EUAA submission AER Rate of return review issues paper, October 2017, p8
- ¹³⁷⁸ AusNet Services, Review of rate of return guideline issues paper, December 2017, p1
- ¹³⁷⁹ ATCO Gas Australia, Review of rate of return guideline issues paper, December 2017,p6-7
- ¹³⁸⁰ Network Shareholder Group, Response to issues paper on the review of the rate of return guideline, December 2017, p8-9
- ¹³⁸¹ Queensland Treasury Corporation, Rate of return guideline review issues paper, December 2017, p3

¹³⁷² Australian Pipelines and Gas Association, Submission to the Issues Paper, December 2017, p4-6

average approach

Submits that the 10-year trailing average is consistent with the most prudent way to manage refinancing risk.

Seeks confirmation that the AER's view that the ex-ante efficient financing cost with the NPV=0 concept, would apply both in transition and in post-transition of the trailing average approach.

day to the trailing average approach was a consideration that the trailing average may more closely match actual efficient financing practices, thereby reducing interest rate risk. However, this does not mean that we concluded that the on the day approach would not provide ex ante efficient compensation.

We consider that both the on-theday and trailing average approaches with transition provide ex-ante efficient compensation to satisfy the NPV=0 concept given effect in the RPPs

We confirm that the ex-ante efficient financing cost definition with the NPV=o concept applied in transition as well as in posttransition of the trailing average approach.

10 Implementation of our return on debt approach

In section 9 we set out our approach to estimating the allowed return on debt. In that section, and in our issues paper and discussion paper, we noted we would have regard to updated evidence to evaluate how we should implement our return on debt approach. In doing so we have:

- Updated credit rating and benchmark term data across the sector, detailed in sections 10.3 and 10.3 respectively.
- Assessed information on new third party data providers. This assessment is set out in section 10.4. Consideration of how to combine third party data to come to a benchmark return on debt is set out in sections 10.4 to 10.6
- Undertaken a 'sense check' of our current approach by reviewing actual return on debt instruments issued by service providers over 2013–17. As detailed in our discussion paper, our review of actual return on debt instruments raised issues relevant to our benchmark debt term and the implementation of our benchmark credit rating, and these are discussed in sections 10.3 and 10.5

We acknowledge that legislative amendments have been proposed that would make this Guideline binding. In section 9 we stated our decision is to set a formula for calculating the allowed return on debt, and that we currently use a formulaic approach to annually updating the return on debt. For this reason, our view is that implementation of a binding rate of return instrument requires only incremental changes to our current approach. On this basis, we have considered refinements to our averaging period criteria and list of contingency actions compatible with a binding rate of return instrument. These are detailed in sections 10.7 and 10.8 respectively.

10.1 Benchmark credit rating

Our draft decision is to adopt a benchmark credit rating of BBB+. We consider this is consistent with the available empirical evidence. Table 42 below shows the historical credit ratings for service providers from 2006 to 2018.

Issuer	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
APT Pipelines Ltd	NR	NR	BBB									
ATCO Gas Australia LP	NR	NR	NR	NR	BBB	BBB	A-	A-	A-	A-	BBB+	BBB+
DBNGP Trust	BBB	BBB	BBB-	BBB	BBB							
DBNGP Finance Co P/L	BBB	BBB	BBB-	BBB	BBB							
DUET Group	BBB-	BBB-	BBB-	BBB-	BBB-	BBB-	NR	NR	NR	NR	NR	NR
ElectraNet P/L	BBB+	BBB+	BBB	BBB	BBB	BBB	BBB	BBB+	BBB+	BBB+	BBB+	BBB+
Energy Partnership (Gas) P/L	BBB	BBB-	BBB+	BBB+								
Australian Gas Networks Ltd	BBB-	BBB-	BBB-	BBB-	BBB-	BBB-	BBB	BBB+	BBB+	BBB+	BBB+	BBB+
ETSA Utilities	A-	NR										
ETSA Utilities Finance P/L	A-											
Powercor Australia LLC	A-	A-	A-	A-	A-	A-	BBB+	BBB+	NR	NR	NR	NR
SP AusNet Services	A	A-	NR									
AusNet Services	NR	NR	NR	NR	NR	NR	A-	A-	A-	A-	A-	A-
AusNet Service Holdings P/L	A	A-										
AusNet Transmission Group P/L	A	A-										
SGSP (Australia)	NR	A-	A-	A-	A-	A-	BBB+	BBB+	BBB+	A-	A-	A-

Table 42 Historical credit ratings of service providers

Assets Pty Ltd												
The CitiPower Trust	A-	A-	A-	A-	A-	A-	BBB+	BBB+	NR	NR	NR	NR
United Energy Distribution P/L	BBB	BBB	BBB	BBB	BBB	BBB	BBB	BBB	BBB	BBB	A-	A-
Victoria Power Networks Pt/L	NR	NR	NR	NR	NR	NR	NR	NR	BBB+	BBB+	BBB+	BBB+
Victoria Power Networks (Finance) P/L	NR	NR	NR	NR	NR	NR	NR	NR	BBB+	A-	A-	A-
NSW Electricity Networks Finance P/ L	NR	NR	NR	NR	NR	NR	NR	NR	NR	BBB	BBB	BBB
Ausgrid Finance P/ L	NR	NR	NR	NR	NR	NR	NR	NR	NR	BBB+	BBB+	BBB
Network Finance Company P/L	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR	BBB+	BBB+
Industry median (yearly)	BBB+/ A-	A-	BBB+	BBB+	BBB	BBB	BBB+	BBB+	BBB+	BBB+	BBB+	BBB+

Source: Bloomberg (S&P Global, Moodys), AER analysis

Notes:

1. The 2018 data is as at 20 June 2018, while all other years are as at 31 December.

2. ATCO Gas Australia, DBNGP Trust and DBNGP Finance are not under AER regulation.

The above ratings for NSW Electricity Networks Finance and Network Finance Co are S&P Global equivalent of Moodys ratings as these corporates are rated by Moodys only (Baa2 and Baa1 respectively).
 APT Pipelines is rated Baa2 (LT issuer), DBNGP Finance is rated Baa3 (senior secured) and Ausgrid Finance is rated Baa1 (senior secured) by Moodys - however they are all rated BBB by S&P Global.
 For some of the service providers there is now more than one related entity listed in the table above, which may affect the calculation of the median. However overall and after considering this factor, we still consider BBB+ to be the appropriate benchmark.

In our view, this indicates that those service providers have collectively maintained stable credit ratings over an extended period, which includes the GFC. All debt issuers within the sample have maintained investment grade credit ratings (between BBB– and A–) over the period. Whilst Table 42 does show that the median credit rating has moved between BBB and A-, the six most recent years of data support a rating of BBB+. We consider that this recent concentration of ratings at BBB+ provides sufficient evidence that this is the appropriate benchmark credit rating. While some submissions

argued that a BBB+ credit rating would be conservative, most supported its adoption.¹³⁸²

Other stakeholders did not agree with this view. The Major Energy Users submitted that¹³⁸³:

the observed data reflects the reality of the BEE and its low risk profile as the revealed data includes for more than the BEE activities. The MEU considers that rather than taking an average of the revealed data, the AER should recognise that the revealed data is conservative and needs to be adjusted to a higher level of credit rating. In this regard, the MEU considers that the credit rating for the BEE should be A or A-, similar to that shown for ETSA and Ausnet which probably more closely reflect the BEE than other firms in the AER listing.

We do not agree that the A or A- rating would be appropriately reflective of the industry's credit risk profile because:

- Direct analysis of the credit ratings faced by energy networks indicates a median credit rating of BBB+. In our view, this is the best and most direct source of evidence on the benchmark credit rating.
- In our view there is no clear basis on which to conclude that ETSA and AusNet more clearly reflect the risks in providing regulated services than other networks in the sample.

However, as we discuss in detail in section 10.5, we have recognised that the current use of broad-BBB credit rating band in implementing the benchmark credit rating may not be appropriately reflecting the actual cost of debt and we are proposing a change to use a combined weighted average of broad-BBB and broad-A curves.

10.2 The role of the Chairmont report

We engaged Chairmont Group (Chairmont) to assist us in obtaining and analysing actual debt data from a total of 11 privately owned service providers, for comparison to the broader corporate debt market. We requested all debt instruments and financial hedging instruments issued between 1 January 2013 and 31 December 2017 as well as the issuer's debt portfolio outstanding as at 1 January 2013. Chairmont developed this data into an energy infrastructure credit spread index (EICSI) and provided us with: ¹³⁸⁴

- A report setting out its methodology, reasons for that methodology and high level conclusions.
- The data included in the EICSI series on which we could undertake further analysis.

¹³⁸² See appendix A for a detailed summary of submissions.

¹³⁸³ Major Energy Users, Estimating the allowed return on debt discussion paper, p7

¹³⁸⁴ Chairmont Consulting, *Aggregation of return on debt data report*, 28 April 2018

In engaging Chairmont, we requested assistance in aggregating actual debt issuance information in order to undertake a sense check of our approach compared to actual practices in the industry. Having considered Chairmont's report, some stakeholders recommended that this series should have a direct role in estimation of the return on debt. For example, the Consumer Reference Group submitted that:¹³⁸⁵

The Chairmont report has generated an Energy Infrastructure Credit Spread Index (EICSI) which from Graph 1 of the report ranges from a spread of just over 120 basis points to 160 basis points. This is compared to the current AER Index in Graph 2, with the AER index ranging from about 170 to 270 basis points.

...

This provides the possibility of an alternative approach which is to set the return on debt by using the EICSI result directly – to allow some space for overperformance a rate of return that allowed a spread of 160 basis points on the Bank Bill Swap Rate is the most accurate estimate of the efficient debt financing cost of a benchmark efficient entity.

At this time, we do not intend to rely directly on the EICSI or similar historical index directly to estimate the return on debt. We hold this view because:

- We consider the analysis is best used as a 'sense check' on our benchmark characteristics and how we implement them.
- In our view, the use of third party data series remains appropriate and a relatively transparent and testable way to estimate the return on debt. By checking our approach against the actual data, our view is that we can refine our selection of third party data sources over time to better reflect observed practices while retaining the benefits of using third party data series.
- Chairmont has adopted general principles for inclusion and exclusion of specific debt instruments within its sample (EICSI). Our view is that these principles are reasonable and result a fit-for-purpose series for a 'sense check'. Nonetheless, we recognise that:
 - Networks have, in submissions and in further discussions about their specific responses, raised a number of methodological concerns about the analysis in the Chairmont report and, in some limited cases, made recommendations about specific debt instruments that should be included in or excluded from the sample.
 - We address these in sections 10.3, 10.5 and appendix B. At this stage, we are not persuaded that it is necessary to change the selection of debt instruments within the EICSI sample. However, we do agree that some of these concerns require further consideration and analysis. As such, we are not currently of the view that the EICSI should be used determinatively.

¹³⁸⁵ Consumer reference group, Estimating the allowed return on debt—Response to discussion paper, May 2018, p 4

Nonetheless, we are of the view that the evidence suggests that spreads on issued debt appears to have been less volatile than our current approach. We intend to continue collecting data in line with this request so that in future guideline reviews we can evaluate whether this is a consistent outcome over time.

10.3 Benchmark term

We need to specify the benchmark debt term for a debt portfolio in order to estimate the allowed return on debt for a service provider. The benchmark term:

- establishes the period over which the trailing average is calculated
- determines the period of the transition to the trailing average
- is an input to obtaining yields to estimate the return on debt.

Our decision is to maintain the current benchmark debt term of 10 years. In our issues paper we proposed that we would not conduct an extensive review of our approach to setting the benchmark term but instead update the empirical elements of our current consideration of the benchmark term.

Most service providers and investors submitted that we should continue to rely on the 10-year benchmark term.

In response to discussion paper, networks and investors raised a series of reasons in support of maintaining the 10-year term, including:

- that the 10-year transition to the trailing average return on debt implies a level of regulatory commitment to the approach over a number of regulatory periods
- that there remains a conceptual basis to expect networks to issue longer term debt to match the lives of their assets
- a series of contextual factors such as implementation of the transition and the privatisation of networks within the sample suggesting that the past 5-year sample may not reflect 'business as usual' with respect to the term of debt issuance
- methodological concerns with the use of a simple average of term on issued debt over 2013–17 as an indicator on which to base conclusions about the appropriate term of debt.

Consumers and retailers submitted a range of different views:

CCP16 and the CRG pointed to evidence of shorter term issuance over the past 5 years but did not recommend a departure from a 10-year benchmark due to the uncertainty and complexity of changing the term while utilities are in the process of responding to the transition to a trailing average approach¹³⁸⁶.

¹³⁸⁶ Consumer Challenge Panel 16, Submission to the AER on its allowed rate of return on debt discussion paper, 30 May 2018, p25-26 ; Consumer reference group, Estimating the allowed return on debt—Response to discussion paper, May 2018, p3

- The MEU submitted that even a 7.5 year benchmark term to maturity may overstate the required return on debt¹³⁸⁷.
- The CRG and MEU recommended alternative approaches return on debt should be based on the average observed spreads of debt in our sample. This would not require an estimate of the benchmark term¹³⁸⁸.
- Energy Australia recommended the reduction of the benchmark term of debt to 7.5 years¹³⁸⁹.

Having regard to these submissions and the evidence available to us, we consider the benchmark 10-year term remains appropriate.

Our key reasons for this view are that:

- Conceptually, we expect service providers would seek to issue long term debt where possible to match the lives of their assets. In our view, this continues to support use of a 10-year benchmark term.
- Consideration of service providers' actual debt raising practices and relevant market circumstances over 2013–17 does not reveal clear conclusions.
- Over the period for which we have collected actual debt data (2013-17) we have implemented a transition to the trailing average return on debt approach. This was a material change to the return on debt approach, and we expect it would have impacted debt raising practices to some extent. Based on the data available to us, it is unclear whether or not the observed debt issuance patterns are a transient adjustment in response to our transition to a trailing average approach.
- A simple average estimate of terms at issuance (over 2013-17) within the sample of collected actual debt data suggests an average term of 7.4 years. However, we agree with the view expressed by service providers in their submissions¹³⁹⁰ that a simple average across instruments in the sample may understate the 'true' observed term of debt over 2013–17 where particular short term debt facilities have been refinanced numerous times over the period without growing in value.

As we detail below, we have conducted further analysis to ascertain whether the benchmark debt term of 10-years was still appropriate for estimation of allowed return on debt. We are satisfied that a 10-year term remains consistent with the evidence.

 ¹³⁸⁷ Major Energy Users, Submission on Estimating the allowed return on debt discussion paper, 28 May 2018, p5
 ¹³⁸⁸ Consumer reference group, Estimating the allowed return on debt—Response to discussion paper, May 2018, p4;

Major Energy Users, Submission on Estimating the allowed return on debt discussion paper, 28 May 2018, p5-6 ¹³⁸⁹ Energy Australia, Submission to AER review of rate of return guideline Return on Debt, 21 June 2018, p1-2

¹³⁹⁰ See section 10.9 for a detailed summary of submissions.

This view was supported by the majority of stakeholders through their submissions to our discussion paper on estimating the return on debt allowance.¹³⁹¹ In this section, we set out further analysis on:

- Conceptual expectations for the benchmark term of debt
- Implications of the EICSI for the benchmark term
- Contextual factors affecting the 2013–17 sample, including:
 - o The impact of transition to a trailing average return on debt
 - o Differences in term profiles between networks
- Interaction with the form of the trailing average.

Conceptual expectations

We consider that a business will, within the constraints of the market for corporate bonds, aim to match the length of the debt term to the asset life in order to minimise refinancing risk. However, we consider this is subject to consideration of the term premium of longer term issuance.

We maintain our view from the 2013 Guidelines in which we concluded that:¹³⁹²

A significant proportion of regulated energy assets are long-lived. We observe that electricity transmission lines and gas pipelines are depreciated for regulatory purposes over as long as 60 years.¹³⁹³ Accordingly, we consider that the entity will seek to fund the long-lived energy assets with longer debt tenors in order to manage refinancing and interest rate risk. By issuing longer term debt the entity reduces the frequency with which it must approach the market, thereby reducing the risk associated with not being able to secure funding at the time when it is required, or at rates that are higher or lower than those it currently pays. In approaching the market less frequently there is less risk associated with changing interest rates, which reduces the volatility in debt servicing costs and the likelihood of mismatch between the business' cash flows and its debt servicing obligations.

¹³⁹¹ AusNet, Response to the AER's discussion paper on estimating the allowed return on debt, 30 May 2018, p1-3; APA, Submission responding to return on debt discussion paper, p11; APGA, Submission to the AER discussion paper estimating the allowed return on debt, p8-10; Network Shareholder Group, Submission to the AER's discussion paper on estimating the allowed return on debt, 31 May 2018, p1, 4; Energy Networks Australia, Response to AER discussion paper estimating the allowed return on debt, p1, 7-13; SA Power Networks, Australian Gas Infrastructure Group, Citipower, United Energy & Powercor (the Businesses), AER discussion paper - estimating the allowed return on debt, p1-3; Consumer Challenge Panel 16, Submission to the AER on its allowed rate of return on debt discussion paper, p24-26; Queensland Treasury Corporation, Estimating the allowed return on debt discussion paper submission to the Australian Energy Regulator, May 2018, p1-3

¹³⁹² AER, Final rate of return guideline—Explanatory statements, December 2013, P. 136.

¹³⁹³ As indicated by PTRM models from the following determinations: AER, *Final decision: Envestra access arrangement Vic, Part 2: Attachments*, March 2013; AER, *Final decision: Aurora distribution determination*, April 2012; AER, *Final decision: SPI Networks (Gas) access arrangement*, March 2013

However, longer-term debt costs more than shorter-term debt in normally functioning markets, as debt holders require compensation for the risks associated with committing capital over a longer period of time. This will lead the entity to trade-off the increase in refinancing risk and the increase in transactions costs due to more frequent issuance associated with shorter-term debt against the increased cost of longer-term debt. The AOFM stated, 'a debt portfolio that reprices less frequently gives rise to less volatile debt servicing cost outcomes... Experience suggests that this risk reduction usually comes at appreciable cost.'

The outcome of this trade-off between refinancing risk and cost may vary over time. For this reason, we consider that the benchmark term of debt is also an empirical question. In the remainder of this section, we have considered evidence from actual debt raising by service providers over 2013–17.

Implications of the EICSI for benchmark term

With respect to benchmark term, Chairmont indicated that the average debt term across all instruments within the EICSI sample was 7.4 years.

In our discussion paper, we recognised that a simple average approach to averaging the term at issuance of instruments within the sample has potential limitations. Chairmont identified a series of possible factors by which this average could be weighted, including:¹³⁹⁴

- Size
- Term
- Credit rating
- Pricing date clustering.

In doing so, Chairmont identified that:1395

There are a range of adjustment methods that may be used to create an adjusted EICSI. Some of these are outlined below, however a weighted index would introduce significant model risk. There is no uncontroversial method to weight reported spreads for factors which influence the relative spread level. The danger is that any adjustment method could reduce the clarity and informational benefit of collecting and aggregating actual industry spread data. This is an area which AER and the industry may wish to further explore.

Chairmont's comments relate specifically to the aggregation of spread information. We have discussed in greater detail our analysis of spreads on issued debt in section 10.5. However, we consider the same is true with respect to developing a measure of the terms of debt at issuance over the 2013–17 sample period. Specifically, with respect to

¹³⁹⁴ Chairmont, Aggregation of return on debt data, April 2018, pp. 12–13

¹³⁹⁵ Chairmont, Aggregation of return on debt data, April 2018, p. 12

estimation of an appropriate benchmark term of debt, we consider that alternative weighting systems for an average term at issuance might address potential shortcomings of a simple average.

In our discussion paper, to test the sensitivity of the simple average to size of debt issuance, we also presented an average of term issued weighted by the size (face-value at issuance) of the debt issued as a proportion of the total face-value of debt in the sample. We found that there was no significant difference between unweighted and weighted average debt term of the sample with the weighted term being 7.5 years compared to 7.4 years for unweighted.

Similarly, the ENA and other network stakeholders submitted that the use of a simple average would over-represent short-term debt. For example, the ENA submitted that:¹³⁹⁶

[C]onsider a firm that behaves exactly in accordance with the AER's assessment of the benchmark efficient approach to debt financing. That firm will have a staggered-maturity portfolio of 10-year debt, of which 10% will be refinanced each year. It may also have one or more tranches of short-term debt for liquidity and/or working capital purposes – assume for this example that such debt is rolled over every 3 months. In this case, in any 12-month period, there will be:

- four observations of the short-term debt being refinanced;
- one observation of a 10-year bond being refinanced; and
- zero weight given to the nine 10-year bonds that were not refinanced during the period.

Thus, the short-term debt will be materially over-represented and the majority of the long-term debt will be omitted from the calculation entirely.

We agree that this is a potential limitation in relying on a simple average 'term at issuance' measure over a five-year period. Of the ENA's recommended alternative measures, we consider that calculation of a sector-wide portfolio over the five-year sample may avoid this potential limitation. This would require us to undertake a similar process to Chairmont in determining debt instruments in the portfolio at the commencement of 1 Jan 2013 should be included and excluded from the sample. This is a complex process requiring some amount of expert judgement and we have not been able to undertake this analysis in time for this draft guideline. Also, because the starting portfolio at the time of the 2013–17 sample would reflect debt issuance practices under the 'on-the-day' debt regime, we would still need to draw conclusions about the extent to which this was indicative of normal debt raising practices under the trailing average regime.

¹³⁹⁶ Energy Networks Association, Estimating the allowed return on debt—Response to the AER discussion paper, 31 May 2018, p11

For these reasons, we consider establishment of a portfolio return on debt is likely to be a useful exercise in future guideline reviews. We intend to continue collecting this data to develop a long-term time series to aid in this analysis.

Interaction with the form of the trailing average

In arriving at the proposed retention of the current benchmark term of 10 years, we have taken into consideration that the trailing average approach and the transition from an 'on-the-day' debt approach to the trailing average depend on the benchmark term. Under this approach¹³⁹⁷:

- We have initially adopted a 10-year transition path in which:
 - The first year is estimated as over a single averaging period with a 10 year term.
 - In subsequent years of the 10-year transition period, the portfolio estimate is updated to include 10 per cent of a further tranche of 10-year debt. The weighting of the first year estimate (initially weighted at 100 per cent) is reduced by 10 per cent per year each year.
- Once the 10-year transition is complete, the return on debt in any year will reflect annual estimates over the current year and preceding 9 years, weighted at 10 per cent per year.
- If we were to adopt a different benchmark term or change it during the transition period, it would be necessary to undertake a further transition between approaches or make adjustments to the trailing average calculation methods in order to achieve the NPV=0 principle which underpins the building block revenue framework. The implementation of this change would require a further complex transition from midway through the ongoing transition based on the 10-year term.

On this issue, CCP16 submitted that:1398

Clearly, there is a strong argument that the adoption of a benchmark term of 10 years contributes to a conservative (i.e. overestimate) ROD. The question of what to do about this is more complex and difficult. The primary difficulty is the uncertainty and complexity of changing the term while most utilities are in the process of adjusting their debt portfolios as part of transitioning to the trailing average of 10-years. Given the extensive analysis and debate (and administrative and judicial reviews) we would not propose that the AER change its current approach to better approximate a benchmark term based on the actual behaviour of the NSPs as part of the new Guideline – although we would not rule it out as a suitable approach beyond the transition period.

Similarly, we stated the following at the time of our previous guideline review:

¹³⁹⁷ AER, *Discussion paper – Estimating the allowed return on debt*, May 2018, p35

¹³⁹⁸ CCP sub-panel 16, Submission to the AER on its allowed rate of return on debt discussion paper, May 2018, p. 25.

...in moving to a trailing average approach we consider that we are committing to a debt term for the period nominated. To change the benchmark debt term in response to updated debt information would not be conducive to regulatory stability. In light of this, we propose to use a 10 year debt term for estimating the return on debt and for setting the period of the trailing average. It also means that a 10-year transition will apply.¹³⁹⁹

Whether the debt issuance patterns are temporary or typical

In reaching a conclusion on the benchmark term of debt, we seek to estimate an appropriate sector wide benchmark for the forward looking period to which the guideline will apply. For the reasons set out in this chapter, we consider that analysis of actual debt issuance practices is an important information source on which to base this conclusion. However, due to the timing of the relevant determinations, many of the service providers in the sample would only have had one or two years of revenue determinations under the current approach. In our view, this may be too limited a time series on which to base conclusions about longer term practices in response to the trailing average return on debt approach.

Complicating interpretation of the sample period further, many of the networks in the sample appealed aspects of our determinations to the Australian Competition Tribunal, specifically including the approach to estimate the return on debt. The process of resolving these appeals and finalising the relevant determinations has taken several years, and is yet to be finalised in some cases. It is therefore unclear when, if at all, any of the service providers affected by these determinations would adopt new debt raising strategies in response to the new approach.

In addition, the sample includes newly privatised networks for which privatisation processes took place during the period where we might expect the transition to the trailing average to be underway. Debt raising practices during the initial acquisition of an asset may not reflect 'business as usual' debt raising practices.

We also recognise that debt issuance from service providers or their parent companies are unlikely to respond only to the approach we adopt to estimating the return on debt. As identified in previous reports by Chairmont, there are a range of different strategies a service provider could adopt depending on its appetite for risk¹⁴⁰⁰.

Differences in debt term profile between service providers

In our discussion paper, we also estimated the weighted average (by face value at issuance) term at issuance for each issuer of debt to assess whether or not terms are common across the sector.

 ¹³⁹⁹ AER, *Better Regulation Explanatory Statement Rate of Return guideline*, December 2013, p137
 ¹⁴⁰⁰ Chairmont Consulting, *Financial practices under regulation: past and transitional*, October 2015, pp. 75-84

We found that some issuers had an average debt term at issuance at around 10 years including some with in excess of 10 years, while other issuers had an average debt term at issuance of at or around 5 years. After further analysis our view is that:

- The networks with average terms issued at or around 5 years appear as though they may be impacted by the tendency for a simple average of debt instruments issued over a relatively short sample to over-represent short-term debt
- We have had regard to this possibility of over-representation in reaching our view on the empirical evidence on benchmark term.

As we identified in developing the 2013 Guidelines,¹⁴⁰¹ the choice of term at issuance reflects a trade-off between:

- Refinancing risk— this is the risk that a firm would not be able to efficiently finance its debt at a given point in time. This may be because the debt instruments that it seeks are not available to it, or because they are expensive. Refinancing risk is often due to systematic factors, such as macroeconomic trends or changes in debt market liquidity. However, refinancing risk may also result from company specific matters. For example, if lenders knew that a company needed to refinance its debt at a certain time or risk bankruptcy, they might raise the interest rates that they demand from the company.
- Higher overall portfolio costs—the need to manage refinancing risk is balanced against the overall cost of the benchmark efficient entity's debt portfolio. For example, a longer average term of debt for a debt portfolio means lower refinancing risk. But it also means the total cost of the debt portfolio is higher. Hence, the efficient debt financing practices would address this trade–off.

Different average terms between the service providers could be a reflection of different appetites for refinancing risk across the sector. The nature of a benchmark term allows for the possibility that different networks might adopt strategies facing more or less risk and either benefit from or face the consequences of that risk. For this reason, neither the lower nor higher-risk approach necessarily reflects the most efficient approach.

10.4 Choice of third party data provider

In our 2013 Guidelines and subsequent regulatory determinations we decided to source debt data from the RBA and Bloomberg. Since then we have become aware of two additional data sources: Thomson Reuters and S&P Global. For our 2018 Guidelines our decision is to continue to source data from RBA and Bloomberg and to also source data from Thomson Reuters. Specifically, we will rely on:

- RBA estimates from its F3 data series
- Bloomberg estimates from its BVAL series (BVSCAB and BVSCAE)

¹⁴⁰¹ AER, Final rate of return guideline—Explanatory statement, December 2013, p 104

 Thomson Reuters estimates from its blended AUD corporate series (BBBAUDBMK and AAUDBMK).

In contrast, we will not use third party yield data from S&P Global at this point in time.

Stakeholder submissions generally supported continued reliance on the Bloomberg and RBA curves but there were mixed views on:

- Whether to include the Thomson Reuters or S&P Global curves—the CRG supported inclusion of the new curves.¹⁴⁰² CCP16 supported inclusion of the Thomson Reuters curve, and subject to further testing, the adoption of the S&P Global curve contingent on availability of a reliable longer historical series.¹⁴⁰³ Networks, investors and QTC all submitted that we should maintain the current approach of using BVAL and RBA curves only.¹⁴⁰⁴
- What weight to put on the curves—for example, CCP16¹⁴⁰⁵ and the CRG proposed we should give greater weight to the RBA curve compared to the other curves.

Having regard to the available evidence, we consider none of the RBA, BVAL or Thomson Reuters methodologies is clearly superior. Our view is that the combined use of the three data providers will contribute to achievement of the NEO and NGO to the greatest degree. Our key reasons for this view are:

- On the bond selection criteria (including approach for identifying outliers) and curve fitting (or averaging) methodologies, we consider that the approaches employed by the RBA, Bloomberg and Thomson Reuters have their unique strengths and weaknesses, but we are not satisfied that any curve is clearly superior.
- All of the curves from all three of the data providers require adjustment from their published form to make them fit for purpose. We are not satisfied that one can be more simply or reliably adjusted to estimate the annual return on debt than another.
- In our view, applying equal weight to each of the three data providers is simple and fit-for-purpose. The process of developing a more sophisticated weighting scheme would rely on contentious assumptions and we are not persuaded that the increase in complexity would result in an estimator we have greater confidence in. In our view, there is no persuasive evidence that the likely difference in average from different weighting schemes will be material over time.
- An average of the three data providers reduces the impact of shocks in any one of the individual curves. This will reduce potential volatility. Further, the use of three data providers incorporates a natural contingency in the event that one of the data providers ceases publication.

¹⁴⁰² CRG, Estimating the allowed return on debt—Response to discussion paper, May 2018, pp. 2–3

¹⁴⁰³ CCP16, Submission to the AER on its allowed rate of return on debt discussion paper, May 2018, p. 21.

¹⁴⁰⁴ See section 10.9 for a detailed summary of submissions.

¹⁴⁰⁵ CCP16, Submission to the AER on its allowed rate of return on debt discussion paper, May 2018, p. 21; CRG, Estimating the allowed return on debt—Response to discussion paper, May 2018, pp. 2–3





Source: Bloomberg, RBA, Thomson Reuters, S&P Global, AER analysis.

Our detailed analysis of the four curve providers is set out in full in our discussion paper. In the remainder of this section, we discuss:

- reasons for our decision to rely on Thomson Reuters curves
- reasons why we will not rely on the S&P Global curves at this time
- weighting of the data providers
- the RBA's recent updates to its historical data series.

Reasons for our decision to rely on the Thomson Reuters curve

Based on evaluation of available information on the curve methodologies, we hold the view that all four curves have strengths and weaknesses and none is clearly superior with respect to either the bond selection criteria or curve fitting methodology. Overall, there is a substantial overlap between the curves in terms of bond selection criteria, though each curve has distinctive characteristics. In our view none of the differences are clearly 'right' or 'wrong'.

Based on the factors set out in our analysis in the decision paper, we are satisfied that the Thomson Reuters data series are fit for purpose.

The analysis of technical characteristics in the RBA, Bloomberg and Thomson Reuters debt series have been addressed in prior reports by the ACCC's Regulatory Economics Unit and by Dr Lally.¹⁴⁰⁶. A comparison of the bond selection criteria and curve fitting methodologies of the four data providers is set out in our May 2018 return on debt discussion paper.

In contrast, the ENA submitted that the Thomson Reuters curve performs poorly against four criteria it sets out in its submission.¹⁴⁰⁷ We do not agree that either the BVAL or RBA curves are superior to the Thomson Reuters curve in overall fitness for purpose. We set out our views on the ENA's submissions in the table, below.

¹⁴⁰⁶ ACCC Regulatory Economics Unit, Thomson Reuters Credit Curve Methodology – Note for the AER, April 2017, Lally, Implementation issues for the cost of debt, November 2014; ACCC, Regulatory Economic Unit, Return on debt estimation: a review of the alternative third party data series – Report for the AER, August 2014.

¹⁴⁰⁷ Energy Networks Australia, Estimating the allowed return on debt, Response to AER Discussion Paper, May 2018, p. 15.

Table 43AER responses to ENA submissions on the Thomson Reuterscurve

ENA submission	AER comments
The main 'blended' Thomson Reuters curve places no restrictions on the ownership or country of risk	We agree that our data series would ideally include only debt issuers with Australia as a country of risk. However, all curves include material departures from our benchmark companies. For example, all of the possible curves include bonds which are not Australian regulated utilities. In our view, there is no clear basis to conclude that differences in sovereign risk characteristics will be more material than those departures from the benchmark. As such, we consider the benefits of the broader data series outweigh the costs of this difference compared to our service providers.
As the Discussion Paper notes, the Thomson Reuters curve is intermittent in its availability. If this intermittency is due to the underlying curve-fitting methodology used by Reuters, as suggested in the Issues Paper, then the availability of the Thomson Reuters curve may continue to be sporadic over time.	 As noted in the discussion paper, Thomson Reuters does not extrapolate its curve beyond the longest term bond in its sample. However: Thomson Reuters has published a 10 year estimate on approximately 75 per cent of business days since April 2015 Even where it does not publish 10 year estimates, Thomson Reuters has published 8 and 9 year estimates on 97 per cent of business days. This is consistent with normal practice for the RBA curve, which typically publishes its 10 year estimate with an 'effective term' of 8–9 years.¹⁴⁰⁸
The adoption of up to four different yield curves in order to determine the return on debt allowance would be overly cumbersome, complex and burdensome on stakeholders with no obvious offsetting benefit. ENA supports the relatively simple approach of using two reliable data sources—RBA and Bloomberg—which appears to have been working effectively for a number of years.	 We recognise the additional costs and administrative costs from the inclusion of additional curves in our sample. However, in our view, the benefits of a greater mix of curve providers outweigh those incremental costs. In particular: The Thomson Reuters curve requires fewer adjustments for use than (for example) the RBA curve. In our view, the incremental administrative burden of adding this curve provider to our methodology is low. Additional curves provide a greater level of mitigation against idiosyncratic outcomes from a particular curve or the temporary or ongoing cessation of publication of a curve.

Source: Energy Networks Australia, Estimating the allowed return on debt, Response to AER Discussion Paper, May 2018, pp. 15-17; AER analysis.

Reasons for our decision not to rely on the S&P Global curves at this time

In general, our view is that consideration of data providers' methodologies is the most important criteria on which to base our decision on the choice of data series. However,

¹⁴⁰⁸ The average 'effective tenor' over the full series is 8.79 years.

in this case, the S&P Global's Australian-dollar-denominated curves produce outcomes which are materially different to the other curve providers and to our expectations: ¹⁴⁰⁹

- Over the data series we have available, the S&P Global broad-A and broad-BBB curves produce very similar results where we would expect a more material difference. In contrast, the BVAL, RBA and Thomson Reuters curves as well as S&P Global's US-dollar-denominated curves exhibit a more material difference.
- For the majority of the period since December 2013, the S&P Global Australiandollar-denominated broad-BBB yield curve produce yields estimates below the 'A' rated curves from the other curve providers.

We recognise that there may be valid drivers of the differences between curve estimates. However, disaggregation of the drivers of these differences is complex due to the proprietary nature of curve estimation and we have not been able to reconcile the differences at this time.

Weightings for the data providers

We are satisfied that a simple average of the three curves will result in a return on debt that is fit for purpose. This is because:

- On the bond selection criteria (including approach for identifying outliers) and curve fitting (or averaging) methodologies, we consider that the approaches employed by RBA, Bloomberg and Thomson Reuters have their unique strengths and weaknesses, but we are not satisfied that any curve is clearly superior.
- All three curves require adjustments from their published form to make them fit-for purpose, and we are not satisfied that either can be more simply or reliably adjusted to estimate the return on debt than the other. We also note that selecting curves based on their required adjustments may come at a cost of the contingency value in using multiple curves.
- The three curves have regularly produced materially different results at particular points in time. The curves have their strengths and shortcomings, but it is not clear to us that one approach is clearly superior.
- A simple average of the three curves will reduce the likely price shock if either curve becomes unavailable or produces erroneous estimates during the period.
- The published curves are widely used and market respected.
- Queensland Transport Corporation (QTC) submitted that the current approach of using equally weighted curves such as RBA and Bloomberg is appropriate. The Australian Pipeline and Gas Association (APGA) submitted that we should consider the pros and cons of each curve when deciding on weightings.

¹⁴⁰⁹ To assist in our analysis, S&P Global has kindly provided us with a longer historical time series of monthly data than is currently publicly available. We have had regard to this analysis in reaching our decision.

Currently the RBA and BVAL curves are weighted equally. The weighting is based on findings from Lally. In particular, Lally derived formulae to demonstrate that, subject to some assumptions, the mean squared error (MSE) of the two variable estimator would be reduced with equal weighting.¹⁴¹⁰ In our previous decisions we mentioned that while we could make simplifying assumptions to include a new curve at equal weight, it is unclear if these would be reasonable without further analysis.

However, reliance on the MSE estimator in these circumstances relies on assumptions about the underlying bias of the series. That is a complex exercise which, in our view, exhibits diminishing returns on top of an already complex approach. In our view, a simple average of the curves:

- Is intuitively reasonable
- Gives equal weight to the strengths and weaknesses of the three curves, which is generally consistent with our evaluation of the curves
- Mitigates equally against price shocks in the event that any one curve temporarily or permanently ceases to be published.

The RBA's revisions to its historical data series

In its 5 June 2018 data release, the RBA made a series of historical revisions to its F3 data series.¹⁴¹¹ We rely on the RBA data sources in our return on debt approach. Following this update, we sought further information from the RBA about the specific nature of any changes.¹⁴¹² . We received a response from the RBA on 4 July and have published it on our website with this draft guideline.¹⁴¹³ Due to the timing of its receipt, we have not yet had an opportunity to consider its substance. Based on the information currently available to us, our draft decision is to continue to rely on the RBA's estimates because:

- In its published documentation for the curve, the RBA has not flagged material changes to its methodology as set out in the original bulletin article. Our analysis of the fitness for purpose of the RBA curve reflects the methodology set out in the bulletin. To the extent that the RBA has refined its implementation of that methodology, we remain satisfied that use of the RBA series within our curve-mix will contribute to achievement of the NEO and NGO.
- The RBA has previously updated its methodology, including historical data revisions, to better capture the conversion of US Dollar bonds into Australian dollar

¹⁴¹⁰ Lally, *Implementation issues for the cost of debt*, 20 November 2014, pp. 7–21

¹⁴¹¹ See RBA, Changes to statistical tables, 5 June 2018—Available at: http://rba.gov.au/statistics/tables/changes-totables.html

¹⁴¹² AER, Letter to the RBA—Revisions to statistical table F3, 26 June 2018.

¹⁴¹³ RBA, Letter to the AER—Revisions to statistical table F3, 4 July 2018.

equivalents.¹⁴¹⁴ Stakeholders have continued to support reliance on the RBA curve since those changes.

We also note that the revisions in the RBA curve:

- bring the estimated outcomes closer into line with the estimates from the BVAL curve
- reduces by a small margin the differences between our approach and the spreads on networks' actual debt instruments.

10.5 Implementation of the benchmark credit rating

In this section, we discuss the implementation of the broad-credit rating band we adopt for estimation. While our benchmark credit rating has been BBB+, curve providers typically offer Australian Dollar debt curves as broad-BBB (BBB-,BBB,BBB+) or broad-A (A-,A,A+). We are not aware of a curve that directly estimates AUD corporate BBB+ debt. As a result, we have to make a choice about which broad credit rating band or combination of bands best gives effect to our target credit rating, which has been BBB+.

Our draft guideline approach is to adopt a weighted average of the broad-BBB and broad-A curves offered by Bloomberg (BVAL), RBA and Thomson Reuters. Specifically, our estimates of the 10-year AUD corporate yield from each curve provider will:

- Be weighted 1/3 of the 10-year estimate derived from the broad-A curve for a provider
- Be weighted 2/3 of the 10-year estimate derived from the broad-BBB curve for a provider.

We consider that this weighted average will contribute to achievement of the NEO and NGO to the greatest extent because:

- For the reasons discussed in this chapter, we consider credit ratings are imperfect measures of debt risk. However, we consider the use of a 'broad-BBB' series alone will, other things held constant, overestimate the return on debt required for a BBB+ rated entity.
- Similarly, we consider sole reliance on a broad-A curve will overestimate the return on debt required for a BBB+ rated entity.
- Therefore, some combination of broad-BBB and broad-A curves should provide the best fit to a BBB+ benchmark rating. In our view, a 2/3 broad-BBB: 1/3 broad A rating is most likely to match a BBB+ benchmark credit rating.¹⁴¹⁵

¹⁴¹⁴ See RBA, Changes to statistical tables, 5 June 2017—Available at: http://rba.gov.au/statistics/tables/changes-totables.html; RBA, Letter to the AER— Questions regarding aggregate measures of Australian corporate bond spreads and yields (statistical table F3), July 2017.

- Our analysis of actual debt instruments raised by service providers compared to our current approach suggests that:
 - When term and date of issuance are controlled for, the use of broad-BBB curves has, over 2013–17, overestimated by approximately 27 basis points the spreads at which service providers have issued debt
 - When term and date of issuance are controlled for, a weighted average of 2/3 broad-BBB:1/3-broad A curves has, over 2013–17, overestimated by approximately 9 basis points the spreads at which service providers have issued debt

Why we consider a broad-BBB curve is conservative

For the reasons discussed in this chapter, we consider credit ratings are imperfect measures of debt risk. However, we consider the use of a 'broad-BBB' series alone will, other things held constant, overestimate the return on debt required for a BBB+ rated entity.

In regulatory determinations made after the 2013 Guidelines, we have acknowledged that reliance on a broad-BBB curve only is likely to overestimate the yield for a BBB+ benchmark.¹⁴¹⁶

This is because, to the extent that credit ratings are an informative measure of credit risk, we expect:

- reliance on a broad-BBB curve is likely to overestimate the level of credit risk (and ultimately the required yields) of a BBB+ benchmark credit rating- because the benchmark credit rating (BBB+) is the highest rating band amongst the constituents, the inclusion of any of the lower rated bonds in the sample (BBB or BBB-) would, other things held constant, overestimate the required return on debt for the benchmark credit rating
- reliance on a broad-A curve only would underestimate the level of credit risk (and ultimately the required yields) for a BBB+ benchmark credit rating because all constituents (A- ,A ,A+) are higher rated than the BBB+ benchmark credit rating
- some combination of broad-BBB and broad-A curves should therefore provide the best fit to a BBB+ benchmark credit rating. As a conceptual expectation, our view is that a 2/3 broad-BBB: 1/3 broad A rating is most likely to match a BBB+ benchmark credit rating.

¹⁴¹⁵ We explain our reasons for this view in more detail in appendix 0.

¹⁴¹⁶ See for example: AER, Final determination— AusNet Services transmission determination 2017-2022— Attachment 3: Rate of return, April 2017, p. 340
Spread comparison at matched terms – evidence that the current approach (broad-BBB curve only) is conservative

The term to maturity of debt issuance appears to be a material driver of the differences between the AER approach and EICSI at points in time during the observed sample (2013-17). However, for the reasons set out in section 10.3, our draft decision is to maintain a 10-year benchmark term. Nonetheless, in our view, analysis of the EICSI sample supports a conclusion that service providers have outperformed our current approach to a material extent even once we control for issuance of debt at shorter terms than the 10-year benchmark.

To assess this, we have undertaken a comparison of credit spreads for debt instruments within the EICSI against the AER approach at a matching term on the commencement date of the debt instrument.

Specifically, we have compared the spreads on issued debt against an average credit spread estimated using the BVAL and RBA broad-BBB curves at matched-terms.¹⁴¹⁷ For example, if a debt instrument was issued with 5 years' term of maturity on 1 Jan 2013, we have compared its credit spread against a simple average of the BVAL and RBA broad-BBB curve estimates also issued at a 5 year term on 1 January 2013.¹⁴¹⁸ Where there is a difference between these two credit spreads, this implies a difference caused by factors other than term or the timing of debt issuance.

Our analysis suggests that debt within the EICSI is raised at, on average, approximately 27 basis points less than equivalent debt estimated using the an average of the BVAL and RBA broad-BBB curves.¹⁴¹⁹ However, as set out in Figure 30, there is time-variation in these spread differences.

¹⁴¹⁷ In some cases, there is no corresponding BVAL or RBA estimate because the term of issued debt is longer term than the longest published term at issuance by either of the curve providers. We have not calculated a 'difference estimate' in these cases because they would require strong assumptions to extrapolate the curves. This excludes approximately 10 per cent of the sample. However, we have undertaken a sensitivity check using a conservative assumption (spreads held constant from longest published term) and it does not appear to materially change the result.

¹⁴¹⁸ We have interpolated the third party yield curves between their published terms using linear extrapolation. For the RBA curve, the shortest published term to maturity is 3 years. We have used the rate of change of the spread to swap between the 3 and 5 year terms to interpolate estimates at 1 and 2 year terms to maturity. Bloomberg typically publishes its BVAL estimates at a greater number of term points, resulting in less need for interpolation.

¹⁴¹⁹ This has been updated since our discussion paper to reflect the RBA's revisions to its historical F3 data series.

Figure 30 Comparison of spreads on debt instruments against the Bloomberg and RBA broad-BBB estimate at matching term to maturity



Source: AER analysis, Bloomberg, RBA.

Note: The figure presents annual averages of matched-term spread differences. It has been updated since the discussion paper to incorporate the RBA's June 2018 revisions to its F3 statistical table.

This time variation appears to be driven by a range of factors. In particular, there has been more variation in the published third party yield curves than in the spreads on issued debt.¹⁴²⁰

Spread comparison at matched terms- weighted average of broad-BBB and broad-A curve

In section 10.1, we discussed the service providers' benchmark credit rating.

To implement this benchmark credit rating, we would ideally select a third party yield curve based on bonds of a credit rating matching our benchmark. However, both providers that we currently rely on (Bloomberg and RBA) and the two additional providers whose curves we are now also considering (Thomson Reuters and S&P Global) publish curves based on broad credit-rating bands. This means that, rather than including only BBB+ rated bonds, the curves we have relied on to date are 'broad-BBB' curves and include BBB-, BBB and BBB+ rated debt.

We have also received expert advice in the past indicating that credit ratings are an informative but not perfect proxy for the risk of debt.¹⁴²¹ This is because:

¹⁴²⁰ On this point, we observe that the RBA curve has been more volatile than the BVAL curve.

- credit ratings are primarily an indicator of the risk of default, whereas required returns on debt also depend on other factors including the likely loss given default¹⁴²²
- credit ratings depend on the use of evaluative judgement by the credit rating agencies
- issues from different industries within the same credit rating band respond differently.

When considering how best to implement the benchmark credit rating, there are alternatives to the use of the broad-BBB curve alone. One possible alternative is combined use of both broad-BBB and broad-A (including A-, A and A+ rated debt). This could be in the form of either a simple or weighted average.

By comparing the outcomes of alternative approaches against actual return on debt information such as the EICSI, we may be able to better inform a view on the best broad-rating curves to implement our benchmark.

To illustrate the possible impacts of a change to our implementation of the benchmark credit rating, Figure 32 below illustrates the outcomes if we had adopted a weighted (2/3 broad-BBB, 1/3 broad-A) average of broad-BBB and broad-A curve estimates over 2013–17. This follows the same 'matched-term differences' approach underlying Figure 30. This weighting system should, on average, more closely match an average credit rating of BBB+ compared to the use of a broad-BBB curve alone.

The average difference across all issuances over the five year sample, captured in Figure 32, is approximately 9 basis points. This follows the same 'matched-term differences' approach underlying Figure 30. That is, service providers issued debt at spreads to swap 9 basis point below the weighted average calculated using the AER's approach to implementing the benchmark credit rating.

Figure 31, below, shows that, by using a weighted average of broad-A and broad-BBB yield curves rather than BBB-only, the industry index and AER series converge where the average term in the industry index comes closer to 10 years. Where the average term is shorter than 10 years, there remains a difference between the industry index and AER series. However, due to the effects of the use of broad-A and broad-BBB curves, this difference is narrower than it would be using our current approach.

¹⁴²¹ See for example: ACCC Regulatory Economic Unit, *Thomson Reuters Credit Curve Methodology – Note for the AER*, April 2017, p. 8-11;

¹⁴²² ACCC, Regulatory Economic Unit, Return on debt estimation: a review of the alternative third party data series – Report for the AER, August 2014, p. 23

Figure 31 Impact of using a 2/3:1/3 average of broad-A and broad-BBB yield curves— after adjustment for June 2018 RBA data revisions¹⁴²³



Source: AER analysis, Chairmont spreadsheet.

Figure 32, below, illustrates the differences between credit spreads at issuance once we have controlled for the effects of term. Where there is an average difference between credit spreads in Figure 32, this is likely to be driven by factors other than term. It does not imply that in 2013, for example, networks raised debt at higher spreads than the 2/3 broad-BBB: 1/3 broad-A average using a 10 year benchmark term. We note that there was a substantially higher number of individual debt issuances in 2016 and 2017. For this reason, the differences for those years in Figure 32 have been more influential in calculating an overall average difference across the five year sample.

¹⁴²³ Note: The 'spread' axis in this figure refers to the credit spread in basis points. Compared to the discussion paper, we have updated this figure to reflect the RBA's historical revisions to its F3 data series.

Figure 32Comparison of spreads on debt instruments against theBloomberg and RBA weighted average broad-BBB and broad-A estimateat matching term to maturity



Source: AER analysis, Bloomberg, RBA

Note: The figure presents annual averages of matched-term spread differences. It is based on a 2/3 broad-BBB 1/3 broad-A weighted average estimate. It has been updated since the discussion paper to incorporate the RBA's June 2018 revisions to its F3 statistical table.

While there is some time-variation in the sample, we are satisfied that the use of this weighted average of broad-A and broad-BBB curves would have resulted in a closer fit between the AER's approach and the actual spreads at which service providers raised debt over 2013–17. In particular:

- the average outperformance over the sample at matched terms is +9 basis points compared to +27 basis points using broad-BBB only
- average annual outperformance and under-performance appears to be roughly symmetrical—there are 2 years in the sample in which the average matched-term difference is positive and 3 years in which it is negative including one year where the negative average is below 10 basis points.

Submissions on spread comparisons at matched terms

Some stakeholders raised methodological concerns about the analysis of differences at matched terms.

In particular, the ENA submitted that these comparisons of spreads at matched terms unreasonably excluded debt at terms of greater than 10 years.¹⁴²⁴ Our analysis in this section is limited to instruments at 10 or fewer years because the RBA curve is never published at terms longer than 10 years. In order to compare spreads on actual debt issued against spreads using our current approach, we consider it is appropriate to limit any extrapolation beyond the longest term published by the curve providers we currently rely on.

However, we consider our sensitivity testing of instruments at or longer than 10 year terms supports our overall conclusions:

- Bloomberg has published BVAL estimates of up to 30 years since April 2015. We
 have tested the sensitivity of our conclusions on longer term debt by estimating the
 spread differences at matched terms using the BVAL curve only, which allows us to
 capture the majority of longer term debt in the sample.
- The average difference for debt instruments issued at terms of 10 years or longer compared against the use of broad-BBB Bloomberg data is approximately 22 basis points. That is, on bonds with terms at or greater than 10 years for which there is published BVAL data at a common or longer term, service providers have issued debt at, on average, 22 basis points below what is estimated under the BVAL curve.
- This analysis includes approximately 77 per cent of the debt instruments in the EICSI sample that have a term at issuance of ten years or longer. For the remaining 23 per cent there is neither BVAL nor RBA data published at a corresponding or longer term.

Beyond this, the ENA made a series of further submissions on general methodological issues relating to the EICSI and our analysis of it.¹⁴²⁵ These include submissions relating to:

- The inclusion or exclusion of callable debt
- Overweighting of short term instruments in a simple average
- Exclusion of subordinated debt
- Treatment of fees
- Exclusion of 2018 data.

These issues were outlined in further detail in a CEG memorandum provided to the AER as a late submission on 21 June 2018. We discuss our views on these issues in more detail in appendix B. Having had regard to these issues, we remain of the view that analysis of spread differences at matched terms suggests that:

¹⁴²⁴ ENA, Estimating the allowed return on debt—Response to the AER discussion paper, May 2018, p. 22.

¹⁴²⁵ ENA, Estimating the allowed return on debt—Response to the AER discussion paper, May 2018, pp. 18–23.

- When term and date of issuance are controlled for, the use of broad-BBB curves has, over 2013–17, overestimated by approximately 27 basis points the spreads at which service providers have issued debt
- When term and date of issuance are controlled for, a weighted average of 2/3 broad-BBB:1/3-broad A curves has, over 2013–17, overestimated by approximately 9 basis points the spreads at which service providers have issued debt.

10.6 Adjustments to published data

We rely on published third party yield curves in order to implement our return on debt approach. However, in some cases these published third party yield curves require minor adjustments to meet the requirements for our estimation process. Table 44, below, sets out the current features of published yield curves that may necessitate some adjustment. Presently, these required adjustments involve extrapolation, interpolation, and conversion to an effective annual rate. Our draft decision on extrapolation, interpolation, and conversion is to adopt a common approach to the published curves.

Extrapolation

Where the published curve has a maximum published effective term of less than the target term to maturity,¹⁴²⁶ we will extrapolate that term to our benchmark term of 10 years. Specifically:

- If we need to extrapolate a curve with a longest published estimate less than 10 years but greater than or equal to 7 years, we will linearly extrapolate the spread to CGS component of the published yield to 10 years using the two longest published estimates and will add this to a 10 year CGS
- If a curve ceases publishing a curve with a longest published term of greater than or equal to 7 years, we will not rely on that curve.

In the past, we have received advice from Lally indicating that linear extrapolation is reasonable where the extrapolation term range is relatively small.¹⁴²⁷ In decisions made since our 2013 Guidelines and having regard to this advice from Lally, we extrapolated the published 5 or 7 year BVAL estimate to 10 years using the corresponding margin in the RBA curve.

However, we propose now to adopt a consistent approach across all three curve providers. Specifically, we will rely consistently on linear extrapolation where a curve provider publishes a curve with a longest term between 7 and 10 years. We reach this view because:

¹⁴²⁶ We distinguish a published 'effective' term because the RBA, while publishing a 10 year yield estimate, identifies in the F3 table that the 'effective term' can differ from 10 years. We understand that, for example, a published 10-year RBA yield estimate is made up of a 10 year base rate and an approximately 10 year credit swap component. To correct for this, we extrapolate the credit spread component so that the effective term of the estimate is 10 years.

¹⁴²⁷ Lally, Implementation issues for the cost of debt, November 2014, pp. 38--44.

- In our view, all three curve providers have strengths and weaknesses. Further, there is no reason we are aware of to expect that any single curve provider is more likely than the other providers to continue publishing its yield curve estimates to 10 years. In principle, our view is therefore that our extrapolation approach for any one curve should not rely on the continued publication of another curve.
- We have designed our approach in the context of a binding methodology to be applied to determinations over a four year period to which we expect a binding rate of return instrument would apply. For this reason, we consider it is preferable to adopt an extrapolation approach which does not depend on the continued publication of other curves.
- Since 2015, all three curve providers have generally published yield curves at both broad A and broad BBB credit rating bands of at least 7 years. As such, recent evidence suggests that the need to extrapolate from a 7 year estimate may be infrequent. The addition of a further curve provider to our curve mix should diminish the materiality of the extrapolation approach applied to any one curve.
- The RBA curve publishes its 10 year estimates with an effective term that has consistently been less than 10 years and commonly less than 9.¹⁴²⁸ We have relied on linear extrapolation to convert this into a yield estimate with an effective term of 10 years. Similarly, the Thomson Reuters broad-A and broad-BBB curves are most commonly published to a 10 year term, but where they are not they are almost always published to 9 or 8 year maximum terms.¹⁴²⁹ As such, for consistency with our approach used to extrapolate the RBA curve we would otherwise rely on linear extrapolation from 8 or 9 years for the Thomson Reuters curve. In light of the practical benefits of a consistent and simple extrapolation approach across all three curve providers, we consider the extension of linear extrapolation back to 7 years to be an incremental change.

Interpolation

Where we need a value for which there is no published estimate but it lies between two published estimates. For example, the RBA only publishes its curve estimates for one day each month, but we require estimates for each business day. As a result, we interpolate the RBA month-end data across all business days in the month.¹⁴³⁰ This requires assumptions about the linearity of spread movements over the course of the

¹⁴²⁹ See

Table 43 in section 10.4.

¹⁴²⁸ See RBA, F3—Aggregate measures of Australian corporate bond spreads and yields, Available at: http://rba.gov.au/statistics/tables/#interest-rates

¹⁴³⁰ For the purposes of all return on debt calculations, 'business days' are those days on which the RBA publishes CGS data in its F16 data release—Indicative Mid-Rates of Australian Government securities.

month. We have discussed the potential effects of these assumptions in previous decisions.¹⁴³¹

Conversion to an effective annual rate

The effective annual rate is calculated by taking the nominal rate and adjusting it for the number of compounding periods in the year, as follows:

Effective annual rate =
$$\left(1 + \frac{r}{n}\right)^n - 1$$

Where:

- r is the stated yield
- n is the number of compounding periods in a year

Table 44 Necessary adjustments to published yield curves

Curve	Necessary adjustments
BVAL	Bloomberg typically publishes a daily 10 year BVAL estimate so the only necessary adjustment is conversion to an effective annual rate, which is a straightforward and small adjustment.
	The RBA only publishes data on one day per month. As a result, we are required to interpolate monthly spreads to Commonwealth Government Securities (CGS) to produce a daily yield series.
RBA	Also, as a consequence of the RBA's curve-fitting methodology, its published 10 year estimate typically has an 'effective term' of less than 10 years. We extrapolate the RBA curve from its 'published' 10 year term (effective term is closer to 9 years) to an 'actual' 10 year term using linear extrapolation from the published 7 and 10 year estimates.
	In addition, RBA estimates require conversion to an effective annual rate.
Thomson Reuters	Thomson Reuters typically publishes a daily 10 year estimate so the only necessary adjustment is conversion to an effective annual rate. However, Thomson Reuters does not extrapolate beyond the longest term in its bond sample and the availability of its 10 year estimate may vary.
S&P Global	Over the period its curve has been offered (since January 2017), S&P Global typically published a daily 10 year estimate so the only necessary adjustment is conversion to an effective annual rate.

Source: AER analysis, Bloomberg, RBA, Thomson Reuters, S&P Global.

10.7 Averaging periods

In our view, our current approach for determining averaging periods remains mostly appropriate for the purposes of implementing the return on debt approach. However, we consider some incremental changes are necessary to allow for the nomination and acceptance of averaging periods to occur without the exercise of any discretion. Table 45, below, sets out a set of requirements for return on debt averaging periods which are contained in clause 18 of the draft rate of return guidelines. We are satisfied the

¹⁴³¹ See for example: AER, *Final determination— AusNet Services transmission determination 2017-2022— Attachment 3: Rate of return*, April 2017, pp. 204–209.

provisions in clause 18 for return on debt averaging periods will contribute to achievement of the legislative objectives. We have introduced two new criteria beyond those specified in our current approach. As set out in clause 18, the averaging periods nominated by a service provider will be kept confidential, to continue the practice in our current approach.

Clause no.	Criteria	Previous criteria	Comments
Cl 18(a)	An averaging period for a regulatory year must finish no earlier than 12 months prior to the commencement of a regulatory year.	n/a	 This criterion, in combination with the other criterion below, establishes unambiguous limit on the period in which an averaging period can occur. By allowing for a 12 month window prior to the averaging period, this should: avoid service providers being forced to raise debt in some months during which some participants choose to stay out of the market allow service providers which raise debt as part of a corporate group to select averaging periods which overlap
Cl 18(b)	An averaging period for a regulatory year must finish no later than 3 months prior to the commencement of a regulatory year.	n/a	This criterion, in combination with the other criterion below, establishes unambiguous limit on the period in which an averaging period can occur. In particular, the 3 months minimum between end of an averaging period and commencement of a regulatory year is necessary for the calculation of annual updates to apply during annual pricing processes.
Cl 18(c)	An averaging period for a regulatory year must be observed over a period of 10 or more consecutive business days up to a maximum of 12 months.	Observed over a period of 10 or more consecutive business days up to a maximum of 12 months	Averaging daily estimates over a number of days smooths out short term volatility in the annually updated return on debt allowance.
Cl 18(d)	An averaging period for a regulatory year must be specified prior to the commencement of the regulatory control period.	It should be specified prior to the commencement of the regulatory control period.	This allows us to substantively assess the service provider's proposal. This avoids the practical difficulties with either (1) creating a new process for approving averaging period proposals or (2) assessing averaging period proposals during the annual pricing process, which is meant to be a compliance check that takes place over a short time frame.
Cl 18(e)	An averaging period for a regulatory year must take place in the future.	At the time it is nominated, all dates in the averaging period must take place in the future.	If a regulated service provider can select an averaging period by looking at historical yields, it may introduce an upward bias.

Table 45 Return on debt averaging period criteria - Clause 18 of draft rate of return guideline

Cl 18(f)	An averaging period for a regulatory year must be specified for each regulatory year within the regulatory control period.	An averaging period needs to be specified for each regulatory year within a regulatory control period.	This allows for the annual debt update. The annual debt update reduces the potential for a mismatch between the allowed and actual return on debt for the benchmark efficient entity.
Cl 18(g)	An averaging period for a regulatory year must not overlap for different regulatory years, although the averaging period is not required to be identical for each regulatory year.	The proposed averaging periods for different regulatory years are not required to be identical but should not overlap.	This avoids double counting averaging periods. This would detract from our specification of the trailing average, which weights periods equally. Not requiring periods to be identical helps preserve confidentiality and provide service providers with a degree of flexibility.
Cl 18 Note 19:	If the start date of any averaging period, nominated in accordance with clause 18 does not fall on a business day, then the start date of the averaging period must be deemed to be the next business day. If the end date of any averaging period, nominated in accordance with clause 18 does not fall on a business day, then the end date of the averaging period must be deemed to be the previous business day. This is to ensure that clause 18(a) and clause 18(b).	n/a	This sets out an adjustment mechanism for the start date and/or the end date of a nominated averaging period in cases where a nominated date falls on a non- business day (whether due to oversight or changes to public holidays subsequent to a regulatory proposal submission). Where an adjustment under Note 19 does not satisfy clause 18(c), the regulatory proposal will be deemed non- compliant and must be re-submitted, as per clause 18(a).

Source: AER analysis.

For the reasons set out in the table, we are satisfied that averaging periods based on these criteria will contribute to a return on debt approach that will promote achievement of the NEO, NGO and RPPs.

Averaging periods which do not meet these criteria

Based on the current draft legislation to implement a binding rate of return instrument, our expectation is that:

- service providers will be required to give effect to the binding rate of return instrument in making regulatory proposals
- we will be required to give effect to the binding rate of return instrument in making determinations.

In practice there is a possibility that, due to misunderstanding or oversight, a service provider may submit in its proposal averaging periods which do not give effect to the above criteria. If that is the case, our expectation is that the regulatory proposal would not be compliant with the rules requirements and would need to be re-submitted.

10.8 Contingencies

As discussed in section 9, our draft decision is to adopt an annually updating return on debt approach.

As a result, our decision on how to apply these third party data sources must be fully specified upfront in each determination, and must be capable of application over the regulatory control period without the use of subsequent judgement or discretion. This principle applies equally if this guideline is binding.

For this reason, we have described series of contingencies below. These contingencies are set out formally in clause 19 of the draft rate of return instrument. They set out how we propose to estimate the annual return on debt in the event of revisions in the RBA's, Thomson Reuters' or Bloomberg's methodologies or other changes to data availability. The underlying principles for these contingencies are largely consistent with our current approach. However, we have updated and generalised aspects of our approach as a consequence of adding an additional curve provider and relying on two broad credit rating bands for each provider. Our overall principles are that the contingencies should:

- Be clear and unambiguous—the rules require the automatic application of a formula to update the trailing average portfolio return on debt. As a result, we will be unable to analyse changes to the approaches or new approaches during the regulatory control or access arrangement period. Therefore, it is important that any contingency be clear and easily implementable.
- Use curves in a form as close as possible to their published form—for example, in April 2015 Bloomberg commenced publication of a 10 year BVAL curve. Accordingly, for averaging periods where the 10 year estimate is available, we will adopt this estimate rather than extrapolating a shorter curve estimate.
- Where necessary, rely on the independent expert judgement of the RBA, Bloomberg and Thomson Reuters—In particular, where Thomson Reuters, the RBA or Bloomberg makes changes to its methodology, we would prefer to evaluate these changes before concluding we are satisfied the curve still meets the criteria set out in the Guideline. However, this is not possible during the regulatory control or access arrangement period. In these circumstances, we therefore are faced with the two alternatives of ceasing to rely on the updated curve, or temporarily relying on the updated curve on the basis that we have assessed the data provider as credible. As we are satisfied that Thomson Reuters, the RBA and Bloomberg are credible and independent, but not that any curve is clearly superior, we consider it is preferable that we adopt the updated curve to limit stakeholders' exposure to the distinct characteristics of a single curve. This is consistent with our position of placing weight on both curves to minimise the mean squared error.
- Preserve the use of as many data sources as possible—where a curve provider shortens its longest published term below 10 years but greater than or equal to 7 years, we will use linear extrapolation to allow for a 10 year estimate for that curve.
- Favour up-to-date data— where we cannot source data for one or two of the three yield curve providers on a particular day, we will rely only on the remaining curve providers. While this results in a smaller data set, it reflects up-to-date data. Only where all third party yield curve providers cease publication will we rely on historical data.

Table 46 Contingencies for implementing the return on debt approach -Clause 19 of draft rate of return guideline

Clause No.	Event	Contingency approach
Cl 19 (a)	A curve provider on day i publishes either a broad A-rated or broad BBB- rated yield estimate with a term less than 10 years but greater than or equal to 7 years	The relevant yield estimate for day i must be linearly extrapolated to an exact term of 10 years in accordance with formulae set out in the (draft) rate of return guideline. For the avoidance of doubt, clause 19(a) also applies to all contingencies in clause 19, if it is necessary to extrapolate any yield estimates to an exact term of 10 years.
Cl 19(b)	A curve provider on day i does not publish both a broad A-rated and broad BBB-rated yield estimate with term greater than or equal to 7 years but less than or equal to an exact term of 10 years.	The yield for day i must be calculated in accordance with relevant clause in the (draft) rate of return guideline using the data from the remaining curve providers, subject to cl 19(c). For the avoidance of doubt, equal weighting must be adopted for the remaining curve providers. For example, if data is available only from two curve providers on day i, they will each have 50 per cent weighting.
Cl 19 (c)	All curve providers on day i do not publish both a broad A-rated and a broad BBB-rated yield estimate with term greater or equal to 7 years but less than or equal to an exact term of 10 years.	 The applicable yield for day i will be calculated by: calculating a simple average of the spread to10-year CGS over the preceding 100 business days for each broad A-rated and BBB-rated curve for all curve providers, which is added to: the daily 10-year CGS yield estimates, to determine each curve provider's broad A-rated and broad-BBB rated yield estimates. For the avoidance of doubt, when calculating the average spread to 10-year CGS over the preceding 100 business days, all available daily yield estimates, for all curve providers must be used, as long as they have not been previously excluded to calculate the yield estimates due to clause 19(b). The daily 10-year CGS yield estimates must be calculated as set out in the (draft) rate of return guideline.
Cl 19 (d)	Any curve provider substitutes its current methodology for a revised or updated methodology	The revised or updated methodology must be used to calculate the yield for day i as long as the yield estimates are obtained from the data sources as identified in the (draft) rate of return guideline. For the avoidance of doubt, if a curve provider ceases publication of its existing curves and publishes a new curve product, data sources other than those identified in the rate of return guideline, then we will not rely on the new curve product. The data provider will be taken to have ceased publication of the relevant curve.
Cl 19 (e)	Any curve provider revises or updates its historical yield estimates,	The revised or updated historical yield estimates must not be used to recalculate the allowed return on debt that has been

		intended for any regulatory year.
Cl 19 (f)	The RBA replaces its monthly publication of yield estimates.	Linear interpolation is no longer necessary and the published daily estimates must be used to calculate the yield for day i and must be extrapolated to an exact term of 10 years if necessary in accordance with the rate of return guideline.
Cl 19 (g)	Either Thomson-Reuters or Bloomberg replaces their publication with another frequency of publication of yield estimates (for example, monthly).	The new yield estimates must be converted into daily yield estimates in accordance with relevant clauses in the rate of return guideline.

finalised for any regulatory year

Source: AER analysis.

As addressed in section 10.4, the RBA has recently made material revisions to its historical data series on which our approach is based. In the next section, we discuss how this type of data change will be treated as a broader principle under this guideline.

How we will respond to retrospective changes in the data series we rely on

At the time we make determinations we use up-to-date information. For example, we use the most recent inflation estimates released prior to our final decisions. We follow this principle because:

- it contributes to estimates that are most reflective of efficient costs and most likely to promote achievement of the NEO/NGO and RPPs (and ARORO in the case of RoR)
- it is unbiased (changes can be positive or negative)
- each decision reflects market expectations, which should include the most up-todate information, at the time of a decision. That is, the decision is appropriate in expectation.

Generally this principle is straightforward to apply, because decisions on parameter values within the rate of return typically apply only for one regulatory period. As a result, at each new determination, we consider afresh the estimate of a parameter or value.

In contrast, the adoption of a 10 year trailing average return on debt implies a methodology that spans multiple regulatory periods.

For most service providers this guideline will apply from year six of the transition to the trailing average return on debt. The portfolio estimates for year six (and onwards) in the transition path can be expressed as follows:

$$\begin{split} R_6 &= (A_1 \times 50\%) + (A_2 \times 10\%) + (A_3 \times 10\%) + (A_4 \times 10\%) + (A_5 \times 10\%) \\ &+ (A_6 \times 10\%) \end{split}$$

$$\begin{aligned} R_7 &= (A_1 \times 40\%) + (A_2 \times 10\%) + (A_3 \times 10\%) + (A_4 \times 10\%) + (A_5 \times 10\%) \\ &+ (A_6 \times 10\%) + (A_7 \times 10\%) \end{aligned}$$

$$R_8 = (A_1 \times 30\%) + (A_2 \times 10\%) + (A_3 \times 10\%) + (A_4 \times 10\%) + (A_5 \times 10\%) + (A_6 \times 10\%) + (A_7 \times 10\%)\%) + (A_8 \times 10\%) R_9 = (A_1 \times 20\%) + (A_2 \times 10\%) + (A_3 \times 10\%) + (A_4 \times 10\%) + (A_5 \times 10\%) + (A_6 \times 10\%) + (A_7 \times 10\%)\%) + (A_8 \times 10\%) + (A_9 \times 10\%)$$

$$\begin{aligned} R_{10} &= (A_1 \times 10\%) + (A_2 \times 10\%) + (A_3 \times 10\%) + (A_4 \times 10\%) + (A_5 \times 10\%) \\ &+ (A_6 \times 10\%) + (A_7 \times 10\%)\%) + (A_8 \times 10\%) + (A_9 \\ &\times 10\%) + (A_{10} \times 10\%) \end{aligned}$$

Where---

- Rn is the portfolio estimate in year n
- An is the annual estimate in year n

This illustrates that the portfolio estimates for year 6 and beyond depend in part on annual estimates for years 1 to 5 of the transition.

In line with determinations made since the 2013 Guidelines,¹⁴³² our view is that:

- we will determine each annual estimate using the best information at the time
- once we have determined an annual estimate, we will continue to use that annual estimate for as long as it enters into a portfolio estimate.

Our key reasons for this view are that:

- This is consistent with the approach we've adopted to date, including for the Transgrid 2018-23 determination. This is the first instance of a service provider entering its second regulatory period under the trailing average approach.
- It is unbiased— in this instance, the changes would result in a downward revision to estimates. However, we consider there is an equal likelihood of any future revisions being upward or downward. During the current cycle of determinations we have not updated numbers in response to RBA data revisions where doing so would have increased annual estimates.
- It reduces regulatory uncertainty— in seeking to promote the NEO and NGO to the greatest degree, we weigh the consequences of any increase to regulatory uncertainty against the incremental improvement in estimates from the change in approach. With the passage of time and the benefit of hindsight, there could be a range of reasons why we now consider there are better estimates of numbers employed previously. In this case, it is probable that the updated RBA estimates are an improvement on its previous estimates. However, under CoAG's proposed legislative amendments for a binding rate of return instrument we must make a guideline that will be applied automatically. Therefore, we must account for any

¹⁴³² See, for example, AER, Final decision: AusNet Services transmission determination—Attachment 3: rate of return, April 2017, p. 145.

unknown future changes. In any case, customers, networks and shareholders are likely to have made decisions based on the expectation that the previous estimates would continue to be employed.

Therefore, we consider the better view is to continue to use estimates that we previously set out in determinations even if there are reasons why we might now employ different estimates. Of course going forward, we will use the most up to date information to inform our decisions when developing new estimates. We are open to receiving submissions on this issue as we finalise the guideline.

10.9 Summary of submissions

This appendix details the submissions we have received from stakeholders as part of our review process to date, and notes how we have had regard to each submission.

Key Point	Submission	Stakeholder	AER Response
Choice of curves	Support the review of third- party debt data series	Ergon Energy & Energex, ¹⁴³³ CKI, ¹⁴³⁴ Network Shareholder Group, ¹⁴³⁵ ATCO, ¹⁴³⁶ EUAA ¹⁴³⁷	As flagged in our issues paper, we have evaluated the available third party data series. Our analysis is set out in greater detail in the return on debt discussion paper and in section 10.4
	The RBA and Bloomberg		Our draft decision is to:
	third party data sources are well accepted and working		 Maintain use of the BVAL and RBA curves; and
well and should be maintained. Has a number of significant concerns in relation to the	well and should be maintained.		Add the Thomson Reuters blended AUD corporate curve
	ENA ¹⁴³⁸	to our curve mix.	
	Reuters and S&P Global curves and considers that they should not be used at this stage		Our reasons for adopting the Thomson Reuters curve and not
			adopting the S&P Global curve are discussed in greater detail in section 10.4.
	Supports continued use of	SA Power Networks,	Our draft decision is to:
	the current two providers. TR and S&P Global are not fit for purpose or	Australian Gas Infrastructure Group,	 Maintain use of the BVAL and RBA curves; and
	nt for purpose or appropriate. Further review of the new curves should be	CitiPower, United Energy and Powercor ¹⁴³⁹	Add the Thomson Reuters blended AUD corporate curve

Table 47 Summary of submissions on the return of debt

- ¹⁴³³ Ergon Energy and Energex, Submission on rate of return issues paper, December 2017, p5
- ¹⁴³⁴ Cheung Kong Infrastructure, Submission on rate of return issues paper, December 2017, p3
- ¹⁴³⁵ Network Shareholder Group, Response to issues paper on the review of the rate of return guideline, December 2017, p9
- ¹⁴³⁶ ATCO Gas Australia, Submission on rate of return issues paper, December 2017, p6-7
- ¹⁴³⁷ EUAA, submission Submission on rate of return issues paper, December 2017, p8
- ¹⁴³⁸ Energy Networks Australia, Submission on Debt Paper, May 2018, p4
- ¹⁴³⁹ SAPN-CitiPower Powercor United Energy AGIG, Submission on Debt Paper, May 2018, p2

Clobal series unsuitable for	 done at the next guideline review. The current approach of giving equal weight to the corporate yield estimates from the RBA and Bloomberg is appropriate Both estimates broadly move together over time and there is no evidence to suggest that either series is systematically biased upwards or downwards over the long-term The yield estimates produced by S&P Global have been materially and consistently lower than the simple average of the RBA and Bloomberg estimates. This may indicate a systematic downward bias, which makes the S&P 	QTC ¹⁴⁴⁰	to our curve mix. Our reasons for adopting the Thomson Reuters curve and not adopting the S&P Global curve are discussed in greater detail in section 10.4. Our draft decision is to: Maintain use of the BVAL and RBA curves; and Add the Thomson Reuters blended AUD corporate curve to our curve mix. Our reasons for adopting the Thomson Reuters curve and not adopting the S&P Global curve are discussed in greater detail in section 10.4.
	Supports continued use of the RBA and BVAL curves. Advice in the discussion paper does not give us confidence that TR and S&P Global are fit for purpose. Notes that TR estimates have been available since 2015 but not continuously	APGA ¹⁴⁴¹	 Maintain use of the BVAL and RBA curves; and Add the Thomson Reuters blended AUD corporate curve to our curve mix. Our reasons for adopting the Thomson Reuters curve and not adopting the S&P Global curve are discussed in greater detail in section 10.4. We note that RBA's published 10 year estimate typically has an 'effective term' of 8 to 9 years. While TR does not extrapolate its curve and their 8 and 9 year estimates have been consistently available. We discuss this in greater detail in section 10.4
 Maintain use of the BVAL and RBA curves; and Add the Thomson Reuters blended AUD corporate curve to our curve mix. Add the Thomson Reuters blended AUD corporate curve to our curve mix. Our reasons for adopting the Thomson Reuters curve and not adopting the S&P Global curve are discussed in greater detail in section 10.4. We note that RBA's published 10 year estimate typically has an 'effective term' of 8 to 9 years. While TR does not extrapolate its curve and their 8 and 9 year estimates have been consistently available. We discuss this in greater detail in section 10.4 	No question regarding the market expertise of the four	APA ¹⁴⁴²	We agree that the RBA and BVAL curves have longer backcast time-

¹⁴⁴⁰ Queensland Treasury Corporation, Submission on Debt Paper, June 2018, pp3-4

¹⁴⁴¹ Australian Pipelines and Gas Association, Submission on Debt Paper, May 2018, pp 5-6

¹⁴⁴² APA, Submission on Debt Paper, May 2018, pp 4-6.

providers. For overall fitness for purpose, the RBA and BVAL curves are superior as they have been subject to independent scrutiny. S&P Global and TR have relatively short series only.		series, and that a longer series is preferable where possible. However, we note that daily estimates of the Thomson Reuters AUD corporate curve has been published consistently since early 2015.
Does not consider benefits in adding the two new curves, would not justify the additional administrative costs. Especially concerned with the S&P Global curve – not clear why it is such an outlier.	AusNet ¹⁴⁴³	In our view, the benefits of adding an additional data source, such as having an expanded curve mix to mitigate any shocks to outcomes in the event that a provider ceases publishing, outweigh the incremental administrative costs.
Support the current use of the RBA and BVAL curves, and consider the other curves should be incorporated as back-up. RBA is useful as it publishes an A- curve. Availability of long time series is important.	Energy Australia ¹⁴⁴⁴	 Our draft decision is to: Maintain use of the BVAL and RBA curves; and Add the Thomson Reuters blended AUD corporate curve to our curve mix. Our reasons for adopting the Thomson Reuters curve and not
Chairmont analysis is clear evidence that use of current data sources gives significantly higher return on debt allowance than the actual. MEU is concerned that the various sources of data is not transparent leading to no certainty that the data is appropriate or biased. The four series examined have failings to a varied extent. This implies that a combination of the available data series is therefore necessary.	Major Energy Users ¹⁴⁴⁵	 Inornson Reuters curve and not adopting the S&P Global curve are discussed in greater detail in section 10.4. We recognise that all of the curves have strengths and weaknesses, however: All curve providers have made substantial information available regarding their bond selection criteria There is a varying degree of public information available on the curve fitting methodologies for the four curve options, but in our view the providers have made a substantial amount of information available for public consultation. We set out this information in greater detail in the return on debt discussion paper. We note that all of the third party yield curve providers have broad-BBB and broad-A rated curves available.
CCP considers the TR	CCP16 ¹⁴⁴⁶	In our view, there is substantial

¹⁴⁴³ AusNet Services, Submission on Debt Paper, May 2018, p3

¹⁴⁴⁴ Energy Australia, Submission on Debt Paper, May 2018, p2

¹⁴⁴⁵ Major Energy Users, Submission on Estimating the allowed return on debt Discussion paper, May 2018, pp 3-4

¹⁴⁴⁶ Consumer Challenge Panel 16, Submission to the AER on its allowed rate of return on debt discussion paper, 30 May 2018, pp 6,12

	series replicates the BVAL series on important bond selection criteria and to include TR with equal weighting would underweight the RBA series Add the TR series with a weighting of RBA 50 per cent, BVAL 25 per cent and TR 25 per cent. If S&P Global can provide a reliable longer time series, add S&P Global as well: RBA 50 per cent, BVAL, TR and S&P Global with 16.6 per cent each.		overlap between all four of the curve providers, but each have distinct characteristics. Further, we note that each provider has independently approached the question of how best to estimate AUD corporate bond yields. To the extent that BVAL and Thomson Reuters have adopted similar bond selection criteria, this might suggest that a majority of reliable providers have reached consensus. We consider there are strength and weakness in each of the providers curves but none are either superior or inferior to the other. Our draft decision is to apply equal (33 per cent) weighting to data from each of Bloomberg, the RBA and Thomson Reuters. Our reasons for this decision are set out in greater detail in section 10.4.
Weighting of the curve	Would not necessarily place the same weights we have placed on our findings, but acknowledge that Dr Lally and the Regulatory Economic Unit carried out comprehensive assessments of the RBA and BVAL curves	APGA ¹⁴⁴⁷	Our draft decision is to apply equal (33 per cent) weighting to data from each of Bloomberg, the RBA and Thomson Reuters. Our reasons for this decision are set out in greater detail in section 10.4.
	The benchmark debt yield should continue to be estimated by giving equal weight to corporate yield estimates from RBA and Bloomberg	QTC ¹⁴⁴⁸	 Our draft decision is to: Maintain use of the BVAL and RBA curves; and Add the Thomson Reuters blended AUD corporate curve to our curve mix. Our reasons for adopting the Thomson Reuters curve and not adopting the S&P Global curve are discussed in greater detail in section 10.4. Our draft decision is to apply equal (33 per cent) weighting to data from
	25 per cent BVAL, 12.5 per cent each TR and S&P Global. Each series should be separately weighted	CRG ¹⁴⁴⁹	each of Bloomberg, the RBA and Thomson Reuters. Our reasons for this decision are set out in greater detail in section 10.4.
Benchmark Term	The 10 year benchmark	EE&E ¹⁴⁵⁰ / CKI ¹⁴⁵¹	Our draft decision is to maintain the

¹⁴⁴⁷ APGA, Submission on Debt Paper, May 2018, p4

 $^{\rm 1449}\,$ CRG, Submission on Debt Paper, May 2018, pp 2-3 $\,$

 $^{^{\}rm 1448}~$ QTC, Submission on Debt Paper, May 2018, p1

term should be maintained

current benchmark term of 10 years. Our reasons are set out in more detail in section 10.3

If the actual cost of debt is relatively constant over time, the MEU considers that the term of debt should not be 'the driving force' in setting the allowed cost of debt. The MEU also recommends that the AER should undertake analysis where the term of debt is varied to see if this results in an outcome which more closely matches the observed cost of debt	Major Energy Users ¹⁴⁵²	We remain satisfied that a10 year benchmark term is appropriate for the reasons set out in 10.3. Having done so, we undertook analysis of spread differences at matched terms to test whether evidence of outperformance remained having taken term into account. We discuss this in greater detail in section 10.5
Evidence does not support a change from 10 years. Chairmont report – understands the reasons for the absence of transparency in the analysis (but EISCI was about half of APA's credit spreads of past 6 years). Additional work is required before definitive conclusions can be drawn from the analysis.	APA ¹⁴⁵³	Based on APA's submission, it is unclear how APA's 'average interest rate' shown in their submission was calculated therefore unable to verify the comparison of APA's credit spreads to EISCI. To the extent APA's spreads to swap are high compared to the rest of the sample, this may reflect factors specific to APA.
Does not agree 7.5 years should be a new benchmark term - AusNet is transitioning and longer term debt have been issued since 2013. AusNet debt maturity profile shows lengthening of debt term issued post 2013. Privatisation must have had an impact on the analysis. AER should present to stakeholders the materiality of the impact of privatisation. Consider inclusion of bank debt in the	AusNet ¹⁴⁵⁴	Our draft decision is to maintain the current benchmark term of 10 years. Our reasons are set out in more detail in section 10.3.

- ¹⁴⁵⁰ Ergon Energy and Energex, Issues paper review of the rate of return guidelines, December 2017, p4-5
- ¹⁴⁵¹ Cheung Kong Infrastructure, Submission on rate of return issues paper, December 2017, p3
- ¹⁴⁵² Major Energy Users, Submission on Estimating the allowed return on debt discussion paper, 28 May 2018, p4-5
- ¹⁴⁵³ APA, Review of the rate of return guidelines APA submission responding to return on debt discussion paper, May 2018, p10
- ¹⁴⁵⁴ AusNet, Response to the AER's discussion paper on estimating the allowed return on debt, 30 May 2018, p1-2

sample may need to be reviewed - AusNet only use these debt as an offset facility and usually undrawn.

Against lowering the term from 10 years. Evidence does not support a change from 10 years. APGA supports the AER working with stakeholders to develop a better understanding of the data and its implications.

APGA1455

Our draft decision is to maintain the current benchmark term of 10 years. Our reasons are set out in more detail in section 10.3

Given the latest analysis, EA recommends reducing the tenor from 10 to 7.5 years - the AER adopted a 10 year term for the current RoRG when the average term at issuance by a Our draft decision is to maintain the representative sample of networks was 8.7 years. current benchmark term of 10 Energy Australia¹⁴⁵⁶ They do not consider years. Our reasons are set out in additional regulatory more detail in section 10.3 uncertainty arises if a consistent principle is applied. Further transitioning may be required if this were to result in material differences to the current cost of debt estimate. The MEU considers that the AER assessment of the way each network addresses its debt is flawed in that there is an assumption that every In our view, networks have long network addresses its debt in the most efficient manner which will be in the long term interests of consumers. In fact, each network will address its debt Major Energy Users¹⁴⁵⁷ management in the interests of its shareholders and therefore will be unique to

MEU recommends that the AER should model more scenarios that simply the 10 years debt terms, e.g. modelling of a mix of credit ratings, various term of debt, to identify closer 'match to

each network

term incentives to manage refinancing risks subject to cost and constraints of the markets. By periodically reviewing our benchmarks and implementation of those benchmarks having regard to actual debt raising practices, these behaviours can ultimately be reflected in forward looking regulatory allowances, promoting the long-term interests of consumers.

¹⁴⁵⁵ Australian Pipelines and Gas Association, Submission to the AER discussion paper estimating the allowed return on debt, 30 May 2018, p2, 8-9

¹⁴⁵⁶ Energy Australia, Submission to AER review of rate of return guideline Return on Debt, 21 June 2018, p1-2

¹⁴⁵⁷ Major Energy Users, Submission on Estimating the allowed return on debt discussion paper, 28 May 2018, p4

	the actual cost".		
	Supports the continued use of a 10-year tenor. The analysis by Chairmont does not support changing the current 10-year benchmark term	QTC ¹⁴⁵⁸	Our draft decision is to maintain the current benchmark term of 10 years. Our reasons are set out in more detail in section 10.3
Averaging Periods	It is important for service providers to set their own return on debt averaging periods in order to issue (or price) their debt to reasonably match their allocated allowance	ENA ¹⁴⁵⁹	We agree with this submission. Under our draft guidelines, service providers are able to nominate their averaging periods subject to a range of criteria that are unambiguous but allow substantial flexibility in terms of timing and length.
	Supports the current approach to return on debt averaging periods but submits that the further conditions set out in revenue determinations (no later than 25 business days before a service provider submits a pricing proposal and commence no earlier than 12 months and 25 business days) should be specified	EE&E ¹⁴⁶⁰	We agree with this submission and have implemented similar criteria although our draft guideline criteria are specified relative to commencement of a regulatory year. This is because not all service providers submit pricing proposals.
	Averaging periods should be set by the AER and be consistent	Major Energy Users ¹⁴⁶¹	We agree that our current approach for estimating averaging periods remains mostly appropriate for the purposes of implementing the return on debt approach.
Credit Rating Band	Even using a 2:1 ratio to weight the A and BBB series only slightly closes the gap between the AER series and the industry index.	CRG ¹⁴⁶²	We propose to adopt a 2/3:1/3 weighting of broad BBB to broad A curves. We consider a broad-BBB yield curve is likely to overstate the yield required for a BBB+ benchmark. For the reasons set out in this chapter, we consider this is supported on both a conceptual and empirical basis.
	Should continue to use broad-BBB. AER suggested combination of 2/3 broad-	APA ¹⁴⁶³	We propose to adopt a 2/3 broad- BBB 1/3 broad-A weighting. We consider a broad-BBB yield curve is

- $^{\rm 1459}\,$ ENA, Submission on rate of return issues paper, December 2017, p17 $\,$
- $^{\rm 1460}\,$ EE&E, Submission on rate of return issues paper, December 2017, p4
- ¹⁴⁶¹ Major Energy Users, Submission on rate of return issues paper, December 2017, p15
- ¹⁴⁶² Consumer Reference Group, Estimating the allowed return on debt response to discussion paper, 29 May 2018, p3
- ¹⁴⁶³ APA, Submission responding to return on debt discussion paper, 31 May 2018, p11

¹⁴⁵⁸ Queensland Treasury Corporation, Submission to the AER estimating the allowed return on debt discussion paper May 2018, p1-3

	BBB and 1/3 broad-A is not supported		likely to overstate the yield required for a BBB+ benchmark. For the reasons set out in this chapter, we consider this is supported on both a conceptual and empirical basis.
	Should continue to use broad-BBB. AER suggested combination of 2/3 broad- BBB and 1/3 broad-A is not supported.	APGA ¹⁴⁶⁴	We propose to adopt a 2/3:1/3 weighting of broad BBB to broad A curves. We consider a broad-BBB yield curve is likely to overstate the yield required for a BBB+ benchmark. For the reasons set out in this chapter, we consider this is supported on both a conceptual and empirical basis.
	While Chairmont's results are not definitive, clearly confirm the view of consumers and CCP16 that the AER's current approach on RoD is conservative. Suggests we should focus on developing our own debt database. Recommends adopting a weighted average of the broad-A and broad-BBB series, the actual weighting is still an open question	CCP16 ¹⁴⁶⁵	We consider the EICSI and our analysis of spread differences are matched term support a conceptual expectation that the use of a broad- BBB credit rating is likely to be conservative. We agree and intend to continue collecting data in line with the current request and further refinement
Benchmark credit rating	EA estimates the current approach of using a broad- BBB adds about 10-15BP to the cost of debt. "A tiered credit rating approach using a credit rating of A- and then BBB might provide a more accurate credit rating profile across the industry."	Energy Australia ¹⁴⁶⁶	We do not agree. We consider the empirical evidence indicates the industry (median) credit rating is BBB+. However for implementation methodology, we are proposing a weighted average of broad-BBB and broad-A which we consider better reflect the BBB+ level credit based on our analysis.
	MEU considers that AER has to implement an approach &/or implementation which reasonably better matches the actual costs of debt as	Major Energy Users ¹⁴⁶⁷	For the reasons set out in this chapter, we consider the empirical evidence supports our proposal to maintain the benchmark credit rating of BBB+. We do not agree. We consider the empirical evidence indicates the industry (median)

¹⁴⁶⁴ Australian Pipelines and Gas Association, Submission to the AER discussion paper estimating the allowed return on debt, 30 May 2018, p4

¹⁴⁶⁵ Consumer Challenge Panel 16, Submission to the AER on its allowed rate of return on debt discussion paper, 31 May 2018, p4, 9 & 28

¹⁴⁶⁶ Energy Australia, Response to AER discussion paper estimating the allowed return on debt, 30 May 2018, p1

¹⁴⁶⁷ Major Energy Users, Submission on Estimating the allowed return on debt discussion paper, 28 May 2018, p3

	seen in Chairmont's report. 'The credit rating for the BEE should be A or A-, similar to that shown for ETSA and AusNet which probably more closely reflect the BEE than other firms in the AER listing': MEU does not consider 'the observed data reflects the reality of the BEE and its lower risk profile as the data includes more than BEE activities'.		credit rating is BBB+. However for implementation methodology, we are proposing a weighted average of broad-BBB and broad-A which we consider better reflect the BBB+ level credit based on our analysis.
	APGA states that the majority of credit ratings from companies in our sample are BBB or BBB+.	APGA ¹⁴⁶⁸	The proportion of companies in our sample in our discussion paper ¹⁴⁶⁹ rated BBB+ was 50 per cent, 71 per cent were rated BBB or BBB+ which was less than those rated BBB+ or A- which counted for 78 per cent. We have consistently estimated our benchmark credit rating using the median which in our view, best estimates the central tendency of the sample.
	Suggests a BBB+ benchmark credit rating would be conservative.	ENA ¹⁴⁷⁰	The proportion of companies in our sample rated BBB+ was 50 per cent, 71 per cent were rated BBB or BBB+ which was less than those rated BBB+ or A- which counted for 78 per cent. We have consistently estimated our benchmark credit rating using the median which in our view, best estimates the central tendency of the sample.
Overall approach and implementation	Possible alternative approach for the rate of return - a defined margin (160bp) over BBSW	CRG ¹⁴⁷¹	Based on the evidence before us, the spreads at which networks raise debt appear to be more stable on average than the third party curves. In principle, we agree that this may support some reliance on a fixed average margin, possibly in addition to a variable component reflecting third party yield curves. However we do not proposed to adopt this approach at this stage as, in our view, it is too material a change in approach to be made on

¹⁴⁶⁸ Australian Pipelines and Gas Association, Submission to the AER discussion paper Estimating the allowed return on debt, 30 May 2018, p4

¹⁴⁶⁹ AER, Discussion paper - Review of rate of return guideline - Estimating the allowed return on debt, May 2018, p15

¹⁴⁷⁰ Energy Networks Association, Estimating the allowed return on debt - Response to AER discussion paper , 31 May 2018, p6

¹⁴⁷¹ CRG, Estimating the allowed return on debt - response to discussion paper, 29 May 2018, p2

empirical grounds over a five year sample.

Based on the evidence before us, the spreads at which networks raise debt appear to be markedly

based on the Chairmont analysis of 150BP. Debt allowance for 2019 should be 90day BBSW + identified debt allowance Analyse the 2018 actual debt data and incorporate it into the debt cost series and refine the DRP to be applied for 2020.	Major Energy Users ¹⁴⁷²	more stable on average than the third party curves. In principle, we agree that this may support some reliance on a fixed average margin, possibly in addition to a variable component reflecting third party yield curves. However we do not proposed to adopt this approach at this stage as, in our view, it is too material a change in approach to be made on empirical grounds over a five year sample.
Submit that we should maintain the current RoD approach and implementation methodologies. Any changes should meet a high threshold. Do not consider that the information presented by the AER in the paper supports a change to the benchmark term or credit rating. Indicates that analysis of networks' actual debt cost is inconsistent with an incremental review Submits that the EICSI is inappropriate because, among other reasons, there are no pure play NSPs in the sample	Network Shareholder Group ¹⁴⁷³	We do not propose to change the approach as we consider there was not sufficient evidence to support it. However we are proposing a change to the mix of credit rating band in implementation methodology as based on our analysis, it will better reflect the benchmark credit rating which is maintained at BBB+ We do not agree. Like all other aspects of the benchmark regulatory framework, examination of outturn performance is a useful source of evidence for considering the appropriate forward looking performance.
Considers Chairmont's analysis has material limitations. Want to work further with AER to fully understand the analysis. As it stands, it does not provide sufficient evidence to warrant any change to the RoD approach	SA Power Networks, Australian Gas Infrastructure Group, CitiPower, United Energy and Powercor ¹⁴⁷⁴	and potential loss given default to be low for regulated NSPs compared to unregulated networks. As such, we expect the EISCI may, if anything, understate the difference between a pure-play service provider and benchmark spreads. Broadly, we consider Chairmont's approach in aggregating the EICSI

- ¹⁴⁷² Major Energy Users, Submission on Estimating the allowed return on debt discussion paper, 28 May 2018, p6
- ¹⁴⁷³ Network Shareholder Group, Submission to the AER's discussion paper on estimating the allowed return on debt, 31 May 2018, p1 & 4
- ¹⁴⁷⁴ SA Power Networks, Australian Gas Infrastructure Group, Citipower, United Energy & Powercor (the Businesses), AER discussion paper - estimating the allowed return on debt, 31 May 2018, p2

AER should estimate a DRP

is reasonable and fit-for-purpose. We recognise that there may be some further refinement to take place, however we have sensitivity tested our conclusions for concerns raised by the networks to the extent we agree they are material. Our analysis is set out in this chapter.

The ENA mode a cories of	Supports the position put forward by the CRG that the rate of return settings have failed to achieve consumer outcomes. Particularly that: The reduction in risk faced by the service providers afforded by the rules are not reflected in lower returns through the rate of return settings; and the lack of data on actual returns against modelled returns, significantly reduces confidence in the AER's ability to make informed decision on the guidelines settings	Agriculture Industries Energy Taskforce ¹⁴⁷⁵	We recognise that achieving outcomes that promote the long- term interests of consumers require us to periodically review our approach to test it remains appropriate and reflective of efficient financing practices and a level of risk commensurate with the risk involved in providing network services. We have sought to achieve this outcome by re- evaluating both our benchmarks
 The ENA finition of a series of further submissions on general methodological issues relating to the EICS1 and our analysis of it. These include submissions relating to: Comparison of 12 month trailing averages The inclusion or exclusion of callable debt Overweighting of short term instruments in a simple average Exclusion of callable and subordinated debt Treatment of fees Exclusion of 2018 data. 	 The ENA made a series of further submissions on general methodological issues relating to the EICSI and our analysis of it. These include submissions relating to: Comparison of 12 month trailing averages The inclusion or exclusion of callable debt Overweighting of short term instruments in a simple average Exclusion of callable and subordinated debt Treatment of fees Exclusion of 2018 data. 	Energy Networks Australia ¹⁴⁷⁶	We agree to further engage with networks on Chairmont's analysis so that it is fully understood by the stakeholders. This was necessary to protect the confidential nature of the information. Our response to these issues in in appendix B. The comparison of spreads at matched term to maturity provided empirical evidence that the networks have materially and consistently outperformed our current approach.
On Chairmont's EICSI time APGA ¹⁴⁷⁷ We acknowledge that market may move over time and regu	 On Chairmont's EICSI time series, APGA points that a	APGA ¹⁴⁷⁷	We acknowledge that market rates may move over time and regulatory

¹⁴⁷⁵ Agriculture Industries Energy Taskforce, Submission on Debt Paper, 1 June 2018, p2

¹⁴⁷⁶ Energy Networks Australia, Response to AER discussion paper estimating the allowed return on debt, 31 May 2018, p3, 21-22

¹⁴⁷⁷ Australian Pipelines and Gas Association, Submission to the AER discussion paper Estimating the allowed return on debt, 30 May 2018, p7

Submits that networks have considerable flexibility in tenor and volume of debt they issue.

CCP16¹⁴⁷⁸

QTC¹⁴⁷⁹

determinations may be made when these rates are relatively high or low. Past patterns in this regard may not continue in the future.

Our discussions with the networks and our analysis on the data suggest that the networks have some flexibility but are constrained to an extent by refinancing and liquidity management within treasury policies which also involve rating agency compliance.

Submits that the AER's sample considers only regulated firms. Suggests the sample should include broadly comparable unregulated firms

Submits that the actual debt cost analysis should not include NSW privatisations

Argues that low refinancing risk is necessary to maintain BBB+ credit rating at high gearing levels. To address refinancing risk, QTC indicates a service provider should refinance early and reinvest the proceeds in a risk free asset to offset the cost

The current transition should be left to run its course

We do not agree that the EICSI should exclude newly privatised networks. We note that, as part of its methodology, Chairmont has excluded special purpose debt such as bridging loans. To the extent that the acquisition phase in network privatization is not reflect of 'business as usual' debt raising, we consider Chairmont's criteria appropriately control for this. We are not persuaded that debt issued by newly privatized networks after the initial acquisition phase should be excluded from the sample.

¹⁴⁷⁸ Consumer Challenge Panel 16, Submission to the AER on its allowed rate of return on debt discussion paper, 31 May 2018, p5

¹⁴⁷⁹ Queensland Treasury Corporation, Estimating the allowed return on debt discussion paper submission to the Australian Energy Regulator, 4 June 2018, p1-2

11 Imputation credits

Our draft decision is to set a value for imputation credits (or gamma value) of 0.5 from a range of 0.3 to 0.6.¹⁴⁸⁰

Our estimate of 0.5 is rounded to one decimal place from an estimate of 0.53 based on the product of an estimated utilisation rate of 0.6 and an estimated payout ratio (or distribution rate) of 0.88. Given the precision of the underlying data we consider rounding the value of imputation credits to one decimal place is appropriate. This is consistent with our existing approach set out in the 2013 Guidelines.

We have used a payout ratio of 0.83 in this draft instrument to be internally consistent with our rounded gamma value of 0.5 and our utilisation rate of 0.60.

We have continued to apply a 'utilisation' approach to estimating the value of imputation credits, an approach found to be open to us by the Full Federal Court in May 2017.¹⁴⁸¹ In coming to this decision we have reviewed the relevant evidence and weight to put on this evidence.

The change in our estimated value of the imputation credits to 0.5 relative to our most recent TransGrid determination released in May 2018 (that applied 0.4) reflects further consideration of the relevant material in this review process. This is driven by:

Placing primary weight on Lally's updated distribution rate estimate from the financial reports of the top 20 ASX listed firms. This suggests a distribution rate of at least 0.88.¹⁴⁸² Previously, we placed most reliance on the Australian Taxation Office (ATO) franking account balance (FAB) data that suggested a distribution rate of 0.7 for all equity and 0.75 for listed equity.¹⁴⁸³ However, the ATO in a recent note and subsequent meeting advised the AER that the ATO FAB data should not be used for detailed time series analysis of Australia's imputation system.¹⁴⁸⁴ In light of the ATO's advice and having considered all the other information before us, we now place no reliance on the ATO FAB data. We also no longer consider a market wide distribution rate is appropriate for the benchmark efficient entity.

¹⁴⁸⁰ In this document we use 'value of imputation credits' and 'gamma' interchangeably. It is common to refer to the value of imputation credits as gamma.

¹⁴⁸¹ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, p. 216.

¹⁴⁸² We note a distribution rate of 0.88 and a utilisation rate of 0.6 give a value of imputation credits of 0.53. This is rounded to 0.5 if we keep one decimal place for the value of imputation credits. To ensure consistency between rate of return parameters given we have used an overall value of imputation credits of 0.5 and a utilisation value of 0.6 (for MRP), this implies a distribution rate of 0.83.

¹⁴⁸³ See the AER's draft decision for TransGrid or ElectraNet for detail. The AER, draft decision: TransGrid transmission determination 2018 to 2023- Attachment 4: Value of imputation credits, September 2017; The AER, Draft decision for ElectraNet transmission determination 2018 to 2023, Attachment 4- Value of imputation credits, October 2017

¹⁴⁸⁴ ATO, ATO Note – Franking account balance – tax of time series data from Taxation Statistics, 9 May 2018; AER, AER minute of 21 June 2018 ATO meeting with ATO staff and comments on ENA summary, 5 July 2018.

 An increase in the utilisation rate estimates from the equity ownership approach. Our most recent updated estimates based on Australian Bureau of Statistics (ABS) data for all equity suggests a range for the utilisation rate of 0.6 to 0.7, whereas the utilisation rate for all equity and listed equity in our most recent ElectraNet draft decision suggested a range of 0.57 to 0.68 and 0.38 to 0.55.¹⁴⁸⁵ The most recent quarters' ABS equity ownership data supports a utilisation rate based on all equity of around 0.65.¹⁴⁸⁶

This document does not attempt to cover in detail the theoretical, empirical and legal debate that has occurred since 2013. Rather, it focuses on the incremental change in information since 2013, our understanding of the law as clarified by the Full Federal Court, and the material put before us in this process. For parties interested in more detail on the theoretical, legal and empirical debates over this period, we recommend reading the 2013 Guidelines explanatory documents, the Full Federal Court decisions for Ausgrid and South Australia Power Networks (SAPN), the Australian Competition Tribunal decisions for SAPN and ActewAGL Gas Distribution, and our most recent decision on the value of imputation credits as set out in the attachment 4 for the ElectraNet Draft Decision.¹⁴⁸⁷

While there is support for a slightly higher value of imputation credits of around 0.6 based on the most recent all equity Australian equity ownership rate of 65 per cent and a distribution rate of at least 0.88 based on Lally's updated work, we consider an incremental move upwards to 0.5 appropriate at this time. The use of 0.60 for the utilisation rate is also consistent with Lally's most recent advice.¹⁴⁸⁸

We also note for the purposes of consistently estimating the post company tax return on equity, we have applied a value for the utilisation rate of 0.6. We consider a value of imputation credits of 0.5 when used with a consistently estimated vanilla rate of return based on a utilisation rate of 0.6 will best achieve the National Electricity Objective / National Gas Objective (NEO/NGO) in light of the evidence currently before us.

11.1 Introduction

Imputation credits are valuable to investors and are therefore a benefit in addition to any cash dividend or capital gains they receive from owning shares. Under the

¹⁴⁸⁵ The AER, Draft decision for ElectraNet transmission determination 2018 to 2023, Attachment 4- Value of imputation credits, October 2017

¹⁴⁸⁶ The most recent December 2017 ABS data release indicates the point estimate of the domestic equity ownership of the Australian equity market at December 2017 is 0.65.

¹⁴⁸⁷ The AER, Explanatory statement- Rate of return guideline, December 2013; The AER, Explanatory statementappendices- Rate of return guideline, December 2013; Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017; Federal Court of Australia, SA Power Networks v Australian Competition Tribunal (No 2) [2018] FCAFC 3, Jan 2018, para. 56; Australian Competition Tribunal, Application by SA Power Networks [2016] ACompT 11, 28 October 2016; Australian Competition Tribunal, Application by ActewAGL Distribution [2017] ACompT 2, 17 October 2017; The AER, Draft decision for ElectraNet transmission determination 2018 to 2023, Attachment 4- Value of imputation credits, October 2017

¹⁴⁸⁸ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 18.

Australian imputation tax system, investors can receive an imputation credit for income tax paid at the company level.¹⁴⁸⁹ For eligible investors, this credit offsets their Australian income tax liabilities. If the amount of imputation credits received exceeds an investor's tax liability, that investor can receive a cash refund for the balance.

The National Electricity Rules and National Gas Rules (NER/NGR) recognise that a service provider's allowed revenue does not need to include the value of imputation credits. Under the NER/NGR, service providers are to recover revenue that compensates them for their efficient costs in providing network services. This includes, among other things, a return to be provided to equity investors (return on equity) that is required to promote efficient levels of investment. The more that imputation credits are valuable, the less return that equity investors require from dividends and capital gains. However, the estimation of the return on equity is not adjusted directly to account for imputation credits.¹⁴⁹⁰ Therefore, an adjustment for the value of imputation credits is required. This adjustment could take the form of a decrease in the estimated return on equity itself.

An alternative but equivalent form of adjustment, which is employed by the NER/NGR, is via the revenue granted to a service provider to cover its expected tax liability. Specifically, the NER/NGR require that the estimated cost of corporate income tax be determined in accordance with a formula that reduces the estimated cost of corporate tax by the 'value of imputation credits' (represented by the Greek letter, γ , 'gamma').¹⁴⁹¹ This form of adjustment recognises that it is the payment of corporate tax which is the source of the imputation credit return to investors.

11.1.1 Approach to date

To estimate gamma, we have used the Monkhouse (1996) formula. The Monkhouse formula is a common way to model the value of imputation credits to investors. It implies that the value of imputation credits is the product of a payout ratio and a utilisation rate. However, both of these parameters are conceptually complex and difficult to estimate. Experts on the value of imputation credits have advocated a range of values that investors place on generated imputation tax credits. These span from zero, meaning no value, to one, meaning full value, and different points within this range.

In 2009, the AER conducted its own review of the value of imputation credits as part of the 2009 WACC review. In that review, we adopted 0.65 as the value for imputation credits, made up of:

• a payout ratio of 1

¹⁴⁸⁹ Income Tax Assessment Act 1997, parts 3-6.

¹⁴⁹⁰ While the return on equity is not reduced to take into account the value of imputation credits, we note our estimate of the market risk premium (MRP) does consider the value we use for imputation credits to ensure it reflects the value to investors in the domestic Australian market inclusive of credits.

¹⁴⁹¹ NER, cll. 6.4.3(a)(4), 6.4.3(b)(4), 6.5.3, 6A.5.4(a)(4), 6A.5.4(b)(4), 6A.6.4; NGR, rr. 76(c), 87A.

 a utilisation rate of 0.65—calculated as an average of the Beggs and Skeels dividend drop off study (0.57)¹⁴⁹² and the Handley and Maheswaran tax statistic study (0.74).¹⁴⁹³

We then applied a value of imputation credits of 0.65 in the Queensland and South Australian electricity distribution determinations. Energex and Ergon successfully appealed this decision to the Tribunal. The Tribunal set the payout ratio to 0.7 and commissioned a dividend drop off study from SFG.¹⁴⁹⁴ The Tribunal adopted SFG's recommendation that the utilisation rate (or theta) be set at 0.35.

In 2011, we adopted a value of imputation credits of 0.25, which was the product of:

- A payout ratio (F)-0.7
- A utilisation rate (θ)—0.35.

This was on the basis of the Australian Competition Tribunal adopting these values.¹⁴⁹⁵

In the development of the 2013 Guidelines the AER conducted a further review of the value of imputation credits. In that review, we proposed that the value of imputation credits should be set with regard to a benchmark efficient entity informed by market wide behaviour rather than with regard to industry or firm specific values.¹⁴⁹⁶ Applying this approach, we adopted 0.5 as the value of imputation credits in the AER Rate of Return Guideline published in Dec 2013. This was the product of:

- A payout ratio of 0.7
- A utilisation rate of 0.7

Since 2015, we have adopted a value for imputation credits of 0.4 for the final decisions released in 2015, 2016, 2017 and 2018, from within a range of 0.3 to 0.5.¹⁴⁹⁷ This was a departure from the value of 0.5 estimated in our 2013 Guidelines, which we made after re-examining the relevant evidence and estimates.

We departed from the 2013 Guidelines reasoning by not relying upon the 'conceptual goalposts approach'. This was based on advice from Handley indicating this approach is not a reasonable approach to estimating the utilisation rate.¹⁴⁹⁸ In addition, in the 2013 Guidelines we considered that the equity ownership approach supported a

D. Beggs and C.L. Skeels, 'Market arbitrage of cash dividends and franking credits', The economic record, Vol. 82, No. 258, September 2006, p. 247.

¹⁴⁹³ J.C. Handley and K. Maheswaran, 'A measure of the efficacy of the Australian imputation tax system', The economic record, Vol. 84, No. 264, March 2008, p. 90.

 ¹⁴⁹⁴ Australian Competition Tribunal, Application by Energex Limited (No 2) [2010] ACompT 7, October 2010, para.
 147.

¹⁴⁹⁵ Australian Competition Tribunal, Application by Energex Limited (Gamma)(No 5) [2011] ACompT 9, May 2011, para. 42.

¹⁴⁹⁶ AER, *Rate of return guideline*, December 2013, p. 23.

¹⁴⁹⁷ We first adopted this value on our November 2014 draft decisions for Ausgrid and others.

¹⁴⁹⁸ J. Handley, *Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits*, September 2014, p. 31.

utilisation rate of between 0.7 and 0.8. After the publication of the 2013 Guidelines we re-examined the relevant data from the national accounts. This resulted in us updating and refining our estimates. The updated estimates from the equity ownership approach indicated a lower utilisation rate.¹⁴⁹⁹

We have also had regard to the distribution rate for listed equity estimated from ATO data and estimated from the 20 largest ASX-listed firms in estimating the value of imputation credits. However, we have placed no weight on the estimate from ATO franking account balance data in light of recent ATO advice. We agree with Lally's advice that the distribution rate is a firm-specific parameter and with his recommendation that a distribution rate estimated from listed equity is a more appropriate benchmark than an estimate based on all equity data.¹⁵⁰⁰ We noted in the 2013 Guidelines we only considered the distribution rate across all equity.

We now consider that:

- It is open to us to have regard to evidence from all equity and/or listed equity only
- It is not necessary to combine estimates of the distribution rate and utilisation rate from the same dataset.

11.2 AER's assessment approach

In this section we set out the approach we have taken to assessing proposals on the value of imputation credits. This approach includes consideration of:

- the requirements of the NEL/NGL and NER/NGR
- legal processes
- our definition of the benchmark efficient entity
- interrelationships
- issues discussed in the concurrent evidence session and submissions

11.2.1 Regulatory requirements

The objective of the adjustment for the value of imputation credits is to reduce the cost of corporate income tax such that only the proportion of company tax which is expected to be retained by the government is reflected in the corporate income tax building block. That is, the adjustment is an estimate of the company tax paid which the government subsequently transfers to investors when they utilise imputation credits.

¹⁴⁹⁹ AER, Draft decision: Ausgrid distribution determination 2015-16 to 2018-19, Attachment 4: Value of imputation credits, November 2014, pp. 14-19.

¹⁵⁰⁰ Lally's view on this issue appears consistent with the views of Gray. See Frontier Economics, *An appropriate regulatory estimate of gamma*, June 2015, pp. 12–13; M. Lally, Gamma and the ACT decision, 23 May 2016, p. 5.

Clauses 6.5.3 and 6A.6.4 of the NER and rule 87A of the NGR set out the cost of corporate income tax rule. This includes an adjustment for the value of imputation credits as follows:

The estimated cost of corporate income tax of a Distribution/Transmission Network Service Provider for each regulatory year (ETC_t) must be calculated in accordance with the following formula:

 $\mathsf{ETC}_t = (\mathsf{ETI}_t \ge r_t) \ (1 - \gamma)$

Where:

ETIt is an estimate of the taxable income for that regulatory year that would be earned by a benchmark efficient entity as a result of the provision of standard control/prescribed transmission services if such an entity, rather than the Distribution/Transmission Network Service Provider, operated the business of the Distribution/Transmission Network Service Provider, such estimate being determined in accordance with the post-tax revenue model.

 $r_{\rm t}$ is the expected statutory income tax rate for that regulatory year as determined by the AER; and

 $\boldsymbol{\gamma}$ is the value of imputation credits

11.2.2 Legal processes

Since the 2013 Guidelines review the key issue that has been in dispute over the past few years is the meaning of "value" in the statutory context. Most of the businesses have taken the word "value" as the market value of imputation credits to investors as reflected in market prices and have proposed that the value of imputation credits should be examined principally through implied market value studies.

By contrast, we have taken the view that the value of imputation credits is a post company tax (post-tax) value before the impact of personal taxes and transaction costs.¹⁵⁰¹ Importantly, the concept of 'value' must be considered in this context. As such, we view the value of imputation credits as the proportion of company tax expected to be returned to investors through the utilisation of imputation credits (a 'utilisation' approach to theta and to the value of imputation credits).

The issue has been considered in a number legal processes and we consider is largely settled after the Full Federal Court's decisions for Ausgrid and SAPN.¹⁵⁰² The Court in its May 2017 and January 2018 decisions upheld the AER's decision on the value of imputation credits and considered it was not an error of construction for the AER to

¹⁵⁰¹ Post-tax refers to after company tax and before personal tax.

 ¹⁵⁰² Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC
 79, May 2017; Federal Court of Australia, SA Power Networks v Australian Competition Tribunal (No 2) [2018]
 FCAFC 3, Jan 2018.

focus on utilisation rather than on implied market value.¹⁵⁰³ In coming to the decision for this draft instrument, we have taken into account the outcome of the recent litigation. We consider our interpretation of the value of imputation credits is consistent with the result of the recent litigation and will contribute to the achievement of the NGO and NEO. We discuss the outcome of the recent litigation in detail in Appendix A.

11.2.3 Definition of a benchmark efficient entity

The NER/NGR refer to a 'benchmark efficient entity'. We consider that the benchmark efficient entity (BEE) is to be taken as having 'a similar degree of risk' as that which applies to the particular service provider in providing its 'regulated' network services. ¹⁵⁰⁴ This is consistent with the Full Federal Court's decision that was handed down in May 2017.¹⁵⁰⁵ The definition of a benchmark efficient entity we use for determining the rate of return is identical to the definition we use for determining the value of imputation credits in this draft instrument. We also note that as the NER and NGR are drafted, the benchmark firm pays tax at the relevant Australian corporate tax rate.

We consider one important element of the definition of a benchmark efficient entity is 'operating within Australia'. This is because the location of a business determines the conditions under which the business operates. This includes the regulatory regime, tax laws, industry structure and broader economic environment that impact the risks faced by the service provider in the provision of its regulatory services. An additional consideration that is particularly relevant to the value of imputation credits is that we recognise that both domestic and foreign investors participate in the Australian market. That is, we consider that the defined market is an Australian domestic market that recognises the presence of foreign investors to the extent that they invest in the Australian market. This is important for determining a value of imputation credits while foreign investors are not.

11.2.4 Interrelationships

The NER/NGR recognise that a service provider's allowed revenue does not need to include the value of imputation credits. The NER/NGR adjust for the value of imputation credits via the revenue granted to a service provider to cover its expected tax liability. This form of adjustment recognises that it is the payment of corporate tax which is the source of the imputation credit returned to investors.

Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC
 79, May 2017, para. 756.

Federal Court of Australia, SA Power Networks v Australian Competition Tribunal (No 2) [2018] FCAFC 3, Jan 2018, para. 56.

¹⁵⁰⁴ By regulated services we mean services that are subject to 'full' revenue cap or price cap regulation. This will be firms we regulate that provide standard control services and prescribed transmission services

¹⁵⁰⁵ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017. Para. 537.

The value of imputation credits is also interrelated with the market risk premium (MRP). The definition of the MRP in the Sharpe-Lintner capital asset pricing model (CAPM) should account for the capitalised value of imputation credits. Accordingly, in our determination of the return on equity in the draft instrument we adjust estimates of the MRP in a manner consistent with our determination of the value of imputation credits. This is also required by the NER/NGR.¹⁵⁰⁶

11.2.5 Issues from the concurrent evidence session and submissions

We released an issues paper in October 2017 outlining key areas of the Guideline review process. In March 2018, we released a discussion paper on imputation credits to provide background on the matters to be discussed at the concurrent evidence session and questions to frame that discussion.¹⁵⁰⁷ We have received a number of submissions from stakeholders in response to the papers we released on the value of imputation credits and in relation to issues discussed in the concurrent evidence session. In reaching our decision for this draft instrument, we have taken into account the issues discussed in the stakeholder submissions as well as the issues raised by the experts in the concurrent evidence session. We respond to these issues in detail in Appendix A.

The key issues on the value of imputation credits raised by the stakeholders include:

- whether a direct estimate of the value of imputation credits for the benchmark efficient entity can be obtained from the ATO tax statistics¹⁵⁰⁸
- the suitability of adopting the distribution rate of firms in the 20 largest ASX listed firms or the market as a whole as a proxy for the benchmark efficient entity¹⁵⁰⁹
- whether a distribution rate can be estimated from a set of comparators¹⁵¹⁰

¹⁵⁰⁶ NER, cll. 6.5.2(d)(2), 6A.6.2(d)(2); NGR, r. 87(4)(b).

¹⁵⁰⁷ AER, *Discussion paper gamma*, March 2018

¹⁵⁰⁸ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 12; NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 3; CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence, 21 April 2018, p. 74; Joint Energy Networks, submission to the AER, 4 May 2018, pp. 6-7; Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December 2017, pp. 5-6; ATCO Gas, Submission to the AER: Review of rate of return guideline-evidence sessions, 4 May 2018

¹⁵⁰⁹ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 11; NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18; APGA, Submission to the Issues Paper, 12 December 2017, p. 10; Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December 2017, pp. 5-6; CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence, 21 April 2018, pp. 77-78

¹⁵¹⁰ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16. CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131.

- whether a utilisation of 1 may be appropriate given that the model assumes the national equity markets are segmented and hence all the assets in the equity markets are owned by domestic investors and there is no foreign investment¹⁵¹¹
- issues with using the ABS equity ownership data for estimating the utilisation rate¹⁵¹²

11.3 Our draft approach

Having considered all the information before us, consistent with our approach since the 2013 Guidelines, we propose that the value of imputation credits within the building block revenue framework is an estimate of the expected proportion of company tax which is returned to investors through utilisation of imputation credits. This 'utilisation' approach considers that the value of imputation credits is a post-tax value before the impact of personal taxes and transaction costs.¹⁵¹³ This is consistent with the Officer framework, which models the value of imputation credits via the parameter gamma (usually labelled using the Greek letter, γ):¹⁵¹⁴

 γ [gamma] is the proportion of tax collected from the company which gives rise to the tax credit associated with a franked dividend.

This 'utilisation' approach to the value of distributed imputation credits was confirmed as legally open to us by the Full Federal Court in May 2017 where the Court found it was not an error of construction for the AER to focus on utilisation rather than on implied market value.¹⁵¹⁵

Further, and consistent with the Monkhouse formula and our approach since the 2013 Guidelines, we propose to estimate gamma as the product of two parameters:¹⁵¹⁶

The payout ratio, which is the proportion of imputation credits generated by the benchmark efficient entity that are distributed to investors.¹⁵¹⁷

¹⁵¹¹ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, pp. 106-108; CRG, *Submission to the AER on its Rate of Return Guideline Review*, May 2018, p. 60.

 ¹⁵¹² ENA, Response to AER Issues Paper, 12 December 2017, p. 37;
 Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December 2017, pp. 5-6;
 APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16;
 ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 91.

¹⁵¹³ Post-tax refers to after company tax and before personal tax.

¹⁵¹⁴ R. Officer, 'The cost of capital of a company under an imputation tax system', Accounting and finance, May 1994, vol. 34(1), p. 4.

¹⁵¹⁵ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, para. 756.

¹⁵¹⁶ See P. Monkhouse, 'The Valuation of Projects Under the Dividend Imputation Tax System', Accounting and finance, 1996, vol. 36(2), pp. 185–212.

¹⁵¹⁷ The imputation credit payout ratio is distinct from the dividend payout ratio, which is the proportion of available firm free cash flow distributed to equity holders via dividends. This choice of terminology is consistent with the draft instrument and most submissions on this issue. It is sometimes called the distribution rate or the access fraction, and in equations is sometimes referred to using the symbol F.
The utilisation rate, which is the extent to which investors can use the imputation credits they receive to reduce their personal tax.¹⁵¹⁸

We propose that the value of imputation credits, in particular the distribution rate, be set with regard to a benchmark efficient entity using appropriate benchmark estimates rather than with regard to market wide behaviour. We do not consider the observed distribution rates of private firms under the Australian imputation system are reflective of an appropriate benchmark distribution rate. Whereas, we consider the utilisation rate is appropriately set as a market wide parameter.

Applying this approach, overall, the evidence suggests a range of estimates for the value of imputation credits might be reasonable. We propose to choose a value of imputation credits of 0.5 from within a range of 0.3 to 0.6.

11.3.1 Payout ratio

In estimating the payout ratio, we propose to continue to use the cumulative payout ratio approach we have adopted since the 2013 Guidelines. This approach calculates the proportion of imputation credits generated (via tax payments) that have been distributed by companies over a certain period of time. We have considered the strengths and limitations of the relevant evidence. The body of evidence includes:

- An estimate by Lally based on the top 20 ASX-listed firms' financial reports
- ATO tax statistic based estimates
- estimates of potential comparators' distribution rates based on the data from the financial reports

Our estimate of 0.83 for the distribution rate is based on:

- giving primary weight on the updated estimate from Lally's top 20 ASX-listed firms' financial reports, which suggests an estimate of at least 0.88 after adjusting for the credits recycled
- having regard to ATO tax statistics estimate (dividend data), which suggests an estimate of 0.57 for all equity
- having regard to the distribution rate of the comparators from the same industry as the BEE, which suggests an estimate of 1 based on Lally's analysis
- ensuring consistency between parameters given we have used an overall value of imputation credits of 0.5 and a utilisation value of 0.6.

Our previous approach was to place primary weight on the ATO tax statistics for estimating the distribution rate. However, based on evidence and advice since the

¹⁵¹⁸ More formally, the utilisation rate is the complex weighted average (by value and risk aversion) of individual investors' utilisation rates. In turn, these reflect each investor's expected ability to use imputation credits to reduce their tax (or get a refund).

2013 Guidelines, we have reconsidered the merits of the various indicators. We propose to place primary weight on the estimate by Lally based on the imputation credit distributions of the 20 largest ASX-listed firms from 2001 to 2017 for this draft instrument. This is because:

- the data for top 20 ASX firms is of high quality given it is audited and subject to scrutiny in financial markets¹⁵¹⁹
- the ATO in its note to us recommends the AER not use the tax statistics as the basis of a detailed macro analysis of Australia's imputation system¹⁵²⁰
- the distribution rate for listed equity from the alternative approach (ATO FAB tax statistics) contains significant unresolved discrepancies and we now consider this data unsuitable for time series analysis¹⁵²¹

In submissions to the AER, some businesses propose a direct estimate of the value of imputation credits from the ATO tax statistics.¹⁵²² This approach uses the estimated imputation credits redeemed and imputation credits created based on ATO tax statistics to give a direct estimate for gamma. We consider this approach is subject to a number of drawbacks. In particular, this approach implies a market wide distribution rate. For this draft instrument, we no longer consider a market wide distribution rate is appropriate for a BEE given that:

- Lally considers most of the regulated firms are either listed or owned by listed firms. Therefore, a distribution rate estimated from listed equity would be appropriate for the BEE.¹⁵²³
- many unlisted firms are owned by individuals who have an incentive to reduce dividends to limit the amount of tax paid at higher marginal personal rates. Therefore, the dividend policy of these firms would be different from the BEE and a distribution rate from all equity will result in an overcompensation for the BEE.

A national gamma from the ATO tax statistics includes firms that make losses and disappear from the tax system without distributing their accumulated credits. The implied market wide distribution rate under this approach appears to have limited

¹⁵¹⁹ M. Lally, Gamma and the ACT Decision, May 2016, p. 26.

¹⁵²⁰ The ATO, ATO note to the AER, 9 May 2018

¹⁵²¹ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 19.

¹⁵²² ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 12; NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 3; CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence, 21 April 2018, p. 74; Joint Energy Networks, submission to the AER, 4 May 2018, pp. 6-7; Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December 2017, pp. 5-6; ATCO Gas, Submission to the AER: Review of rate of return guideline-evidence sessions, 4 May 2018; ENA; Response to additional AER and Australian taxation Office materials- estimating the value of imputation credits, 29 June 2018; N. Hathaway (Capital Research), Memorandum - Response to three questions asked by the ENA, 28 June 2018.

¹⁵²³ M. Lally, *The estimation of gamma,* November 2013, pp. 10-11; M. Lally, *Review of the AER's views on gearing and gamma,* 7 May 2018, p. 19.

connection with the BEE. Rather, it will underestimate the efficient distribution rate of the BEE.

APA and APGA propose that the ATO tax statistics could give a reliable gamma estimate for the BEE once the market wide distribution rate is adjusted based on Wheatley's advice.¹⁵²⁴ We note the adjusted gamma estimate from the ATO tax statistics supports a value of imputation credits of at least 0.5. The detailed discussion on Wheatley's proposed approach is set out in Section A.4.3. Moreover, APGA and CCP16 propose that the AER should examine the distribution rate of the BEE by referring to the distribution rate of comparator companies.¹⁵²⁵ We note the distribution rate of the comparator firms from the same industry as the BEE also supports a higher distribution rate. Our detailed reasoning for this draft decision on the payout ratio including our detailed response to the businesses' submissions is set out in Appendix A.

11.3.2 Utilisation rate

In estimating the utilisation rate, we have considered the strengths and limitations of different sources of information. This includes:

- the equity ownership approach that uses the ABS data
- estimates based on ATO tax statistics
- estimates from implied market value studies

We consider the current evidence suggests a utilisation rate of approximately 0.6 is appropriate. In coming to a utilisation rate of 0.6, we have updated the data to the latest releases and we place:

- significant reliance upon the equity ownership approach, which suggests a range for the estimate of 0.6 to 0.7 for all equity
- some reliance upon tax statistics, which suggests an estimate of 0.61 based on dividend data for all equity
- limited reliance upon implied market value studies, which suggest a range for the estimate of 0 to 0.5. In particular, the adjusted estimate from SFG's dividend drop off study suggests a utilisation rate of 0.4¹⁵²⁶

¹⁵²⁴ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 116; APA, Submission responding to discussion papers and expert evidence, 4 May 2018, pp. 11-12; APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 18.

¹⁵²⁵ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16; CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131.

¹⁵²⁶ Since the 2013 Guidelines we have considered that implied market value studies support an estimate of the utilisation rate between 0 and 0.5. The SFG dividend drop off study is one common type of implied market value studies that was adopted by most businesses. The businesses previously proposed a utilisation of 0.35 from SFG's study. We consider implied market value studies, if they are to be used at all, need to be adjusted for the incorrect estimates of the post company pre-personal tax value of cash dividends which would expect to also result in an incorrect estimate of the value of imputation credits. Based on Handley and Lally's advice, we consider the

- We note our estimate of the utilisation rate in our recent determinations was based on the data for all equity and listed equity. However, for this draft instrument, we have primarily relied on a utilisation rate estimated from data on all equity. In coming to this decision, we have had regard to the experts' advice in the concurrent evidence session and also stakeholder submissions. We agree with Lally that the utilisation rate is a market wide parameter and therefore is appropriately estimated based on data for all equity. ¹⁵²⁷
- We no longer consider a utilisation rate based on listed equity would give an appropriate utilisation estimate for the BEE, although we note that the current estimate of Australian ownership of listed equity is 58 per cent and combining this with a distribution rate of 88 per cent would still support a value for gamma of 0.5.

Another key issue raised by the stakeholders and the experts on the utilisation rate is that the Officer model assumes the equity markets are segmented.¹⁵²⁸ This implies a utilisation rate of 1 as it assumes all the assets in the equity markets are owned by domestic investors and there is no foreign investment. However, we consider the assumption of no foreign investment and no foreign investors would not reflect the empirical reality (of foreign investment in the Australian domestic market). In light of this we consider a more appropriate way for estimating the utilisation rate is to recognize the existence of foreign investors and therefore we interpret the utilisation rate as a weighted average over the utilisation rates of all investors in the Australian market.

The ENA submitted that an estimated distribution rate from the top 20 ASX firms suffers from the same issues as identified with ATO Franking account balance data.¹⁵²⁹ We do not agree with this submission. First, the top 20 ASX firms used to estimate the distribution rate are stable through time and therefore this analysis does not suffer the material entry and exit problems inherent in the use of aggregate ATO FAB data. For example, in the ATO FAB data set some firms will disappear over time due to liquidation causing leakage. Second, the data used for analysis of the distribution rate of the top 20 ASX firms is audited financial data and therefore is expected to be more reliable than the ATO FAB data which is based informational data on tax filings.

Our detailed reasoning for this draft instrument on the utilisation rate is set out in Appendix A.

estimate from SFG's dividend drop off study should be interpreted as an estimate of around 0.4. Our detailed discussion on implied market value studies is set out in the attachment 4 to our determination for ElectraNet. The AER, *Draft decision for ElectraNet transmission determination 2018 to 2023, Attachment 4- Value of imputation credits*, October 2017

¹⁵²⁷ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 95.

¹⁵²⁸ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, pp. 106-108; CRG, *Submission to the AER on its Rate of Return Guideline Review*, May 2018, p. 60.

¹⁵²⁹ ENA; Response to additional AER and Australian taxation Office materials- estimating the value of imputation credits, 29 June 2018

A Value of imputation credits: detailed analysis

In this appendix, we set out further material for our draft decision on the value of imputation credits. We respond in detail to the issues discussed in the concurrent evidence session and also issues raised in stakeholder submissions. In this draft instrument, we have adopted a value of imputation credits of 0.5 having considered all the material before us. This appendix is structured under the following headings:

- rule requirements
- · recent litigation on the value of imputation credits
- issues discussed in the concurrent evidence session and businesses' submissions
- reasons for approach

A.1 Rule requirements

Unlike many other aspects of the NER/NGR, there is no specific objective we must achieve for the value of imputation credits and no specific factors we must take into account in estimating it. The allowed rate of return objective does not specifically apply to the value of imputation credits. However, the rate of return must be determined on a nominal vanilla basis that is consistent with our estimate of the value of imputation credits.¹⁵³⁰

In this context, the conceptual rate of return framework developed by Officer in a 1994 paper informs our approach to interpreting and estimating the value of imputation credits.¹⁵³¹ This is because:

- The NER/NGR's cost of corporate income tax formula (shown above) mirrors Officer's framework for the treatment of imputation credits, including through the use of the parameter denoted by the Greek letter 'gamma'.¹⁵³²
- We have received expert advice that Officer's definition of the nominal vanilla rate of return provides the basis for the rate of return framework in the NER/NGR.¹⁵³³ Previous statements by the consultant for the majority of the service providers', Gray, and their industry association appear to support this consideration:
 - During the AEMC's 2012 rule change process, Gray advised the AEMC that
 '...there are a number of different WACC formulas that can all be identified

¹⁵³⁰ NER, cll. 6.5.2(d)(2), 6A.6.2(d)(2); NGR, r. 87(4)(b).

¹⁵³¹ R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, pp. 1–17.

¹⁵³² R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, equation 2.

¹⁵³³ J. Handley, Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits, 29 September 2014, pp. 7–8.

as post-tax nominal definitions of WACC. Officer (1994), in the paper that forms the basis for the regulatory rate of return framework, sets out four such definitions...¹⁵³⁴

 During the development of the 2013 Guidelines, the Energy Networks Association (ENA) submitted '[t]he fundamental economic framework in relation to dividend imputation was set out by Officer (1994)...¹⁵³⁵

The NER/NGR require that we determine the rate of return on a nominal vanilla basis that is consistent with our estimate of the value of imputation credits.¹⁵³⁶ The Officer framework provides a means for doing this. It provides a consistent framework for determining the rate of return for a business, which takes into account the value that investors expect to receive from utilising imputation credits.¹⁵³⁷ An important implication of this is that the value of imputation credits (or gamma) is not a standalone concept or parameter. It is part of a broader framework, and should be interpreted and estimated accordingly.

Consistent with the expert advice we have received, we consider that the Officer framework provides the basis for the rate of return framework in the NER/NGR. We therefore also consider that estimating the value of imputation credits consistent with the Officer framework will best promote the NEO/NGO ¹⁵³⁸ and other requirements of the NER/NGR.

To this end, we have had regard to the differing expert opinions on the proper interpretation of the value of imputation credits parameter in the Officer framework. As discussed in section A.4.2, we accept Handley's expert advice on the Officer framework. An important aspect of this advice is that the framework is on a 'before-personal-tax and before-personal-costs' basis.¹⁵³⁹ That is, 'the per dollar value of an imputation credit γ gamma should be measured prior to any personal tax on the credit and prior to any personal costs associated with the receipt of the credit'.¹⁵⁴⁰ A detailed consideration of Handley's advice on the Officer framework is set out in section A.7.3 of attachment 4 to our determination for ElectraNet.¹⁵⁴¹

By determining a value of imputation credits in a manner consistent with the Officer framework, we consider that we are making our decision in a manner that will or is

¹⁵³⁴ SFG, Response to submissions on rule change proposals, Report for the AEMC, 5 November 2012, para. 2.

ENA, Response to the Draft Rate of Return Guideline of the Australian Energy Regulator, 11 October 2013, p. 49.
 NER, cll. 6.5.2, 6A.6.2; NGR, r. 87.

¹⁵³⁷ For a detailed discussion of the Officer framework, see: J. Handley, *Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits*, 29 September 2014, pp. 7–12.

¹⁵³⁸ NEL, s. 16(1)(a); NGL, s. 28(1)(a).

¹⁵³⁹ Although the term 'personal' is used, we note that classes of investors other than individual persons can value imputation credits (for example, superannuation funds and charities). Therefore, an alternative characterisation might be 'before-investor-tax' and 'before-investor-costs'.

¹⁵⁴⁰ J. Handley, *Report prepared for the Australian Energy Regulator: Further advice on the value of imputation credits*, 16 April 2015, p. 5.

¹⁵⁴¹ The AER, Draft decision: ElectraNet transmission determination 2018 to 2013, Attachment4- Value of imputation credits, October 2017,

likely to contribute to the achievement of the NEO/NGO.¹⁵⁴² In taking this approach we have applied a 'utilisation' approach to estimating the value of imputation credits. We consider a 'utilisation' approach to the value of imputation credits consistent with the rate of return framework in the NER/NGR.

Further, when exercising our discretion in making the relevant parts of a decision, we must take into account the revenue and pricing principles (RPP).¹⁵⁴³ The RPP provide, amongst other things, that:¹⁵⁴⁴

- a service provider should be provided with a reasonable opportunity to recover at least the efficient costs the operator incurs providing network services and complying with regulatory obligations
- a service provider should be provided with effective incentives in order to promote economic efficiency with respect to the network services it provides, and
- a price, charge or tariff for the provision of a regulated service should allow for a return commensurate with the regulatory and commercial risks involved in providing the network services.

Therefore, the value of imputation credits we adopt must ultimately promote the achievement of the NEO/NGO (via its application in the estimated cost of corporate income tax building block) and must take into account the RPP.

With reference to the language of the RPP, this requires the exercise of our discretion in determining a tax building block (including the exercise of our discretion in determining the adjustment for the value of imputation credits) that is:

- not too low, in that it contributes to providing a reasonable opportunity to recover at least efficient corporate tax costs
- not too high, in that it contributes to a return that is not excessive and is commensurate with the relevant risks.

We consider that finding the right balance is best served by having regard to the merits of the full range of relevant evidence. We explain our consideration of, and reliance upon, the range of relevant evidence in this draft instrument. We determine a value of imputation credits that in combination with our allowed return on equity that is estimated consistently, we are satisfied achieves a balance between the opportunities for service providers to recover at least efficient costs but that is commensurate with relevant risks.

¹⁵⁴² NEL, s. 16(1)(a); NGL, s. 28(1)(a).

¹⁵⁴³ NEL, s. 16(2)(a)(i); NGL, s. 28(2)(a)(i).

¹⁵⁴⁴ NEL, ss. 7A(2)–(7); NGL, ss. 24(2)–(7).

A.2 Recent litigation on the value of imputation credits

The issue that has been in dispute over the past few years is the interpretation of "value" in the statutory context. It was heard by the Ausgrid Tribunal and the decision was handed down in February 2016.¹⁵⁴⁵ The Ausgrid Tribunal considered the AER had erred in adopting a conceptual approach to gamma that defines it as the value of imputation credits that are available for redemption.¹⁵⁴⁶ The Ausgrid Tribunal found this due to us placing most reliance on the equity ownership approach and effectively defining the utilisation value as the proportion of distributed imputation credits that are available for redemption.¹⁵⁴⁷ The Ausgrid Tribunal considered that such an approach:¹⁵⁴⁸

'is inconsistent with the concept of gamma in the Officer Framework for the WACC which underlies the rules and the objective of ensuring a market rate of return on equity by making an adjustment to the revenue allowance for taxation to take account for imputation credits.'

We sought judicial review of the Ausgrid Tribunal's decision in the Full Federal Court and our applications were heard in October 2016. The decision of the Full Federal Court was handed down on 24 May 2017.¹⁵⁴⁹ The Court found it was not an error of construction for the AER to focus on utilisation rather than on implied market value.¹⁵⁵⁰ It accepted the AER's submission that the Rules require consistency in the way the relevant building blocks interact, that is, after company tax but before personal tax and personal costs.¹⁵⁵¹ The Full Federal Court found the Tribunal erred in concluding that the value of imputation credits is (only) the value claimed or utilised as demonstrated by the behaviour of the shareholder recipients of the imputation credits.¹⁵⁵² The Full

¹⁵⁴⁵ Australian Competition Tribunal, *Application by Public Interest Advocacy Centre Ltd and Ausgrid* [2016] ACompT 1, 26 February 2016.

¹⁵⁴⁶ Australian Competition Tribunal, Application by Public Interest Advocacy Centre Ltd and Ausgrid [2016] ACompT 1, 26 February 2016, para. 1100; United Energy, Submission on AER preliminary determination - Submission on gamma, 26 April 2016; CitiPower/Powercor, Submission on implications of recent Australian Competition Tribunal Decision, 18 April 2016; ActewAGL, Implication of recent Tribunal decisions for final decision and updates to the allowed rate of return and forecast inflation estimate, 12 May 2016.

¹⁵⁴⁷ Australian Competition Tribunal, *Application by Public Interest Advocacy Centre Ltd and Ausgrid* [2016] ACompT 1, February 2016, para. 1100.

¹⁵⁴⁸ United Energy, Submission on AER preliminary determination - Submission on gamma, 26 April 2016; CitiPower/Powercor, Submission on implications of recent Australian Competition Tribunal Decision, 18 April 2016; ActewAGL, Implication of recent Tribunal decisions for final decision and updates to the allowed rate of return and forecast inflation estimate, 12 May 2016; Australian Competition Tribunal, Application by Public Interest Advocacy Centre Ltd and Ausgrid [2016] ACompT 1, 26 February 2016, para. 1100.

 ¹⁵⁴⁹ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC
 79, May 2017.

¹⁵⁵⁰ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, para. 756.

¹⁵⁵¹ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, para. 752.

 ¹⁵⁵² Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC
 79, May 2017, para. 754.

Federal Courts decision on this point is consistent with the approach we have taken in all regulatory decisions released since November 2014.

The Australian Competition Tribunal's decision for ActewAGL Distribution in October 2017 further upheld the AER's decision on the value of imputation credits.¹⁵⁵³ The Tribunal considered that it was open to the AER to consider both listed and all equity consistent with matching the distribution and utilisation rate estimates for each of those categories.¹⁵⁵⁴ It accepted that the AER was entitled to come to the view that the Rules require that the value of imputation credits be measured before investor taxes and costs.¹⁵⁵⁵ It accepted the expert advice that "before investor taxes and costs" means the value of imputation credits is to be estimated before allowing for the impact of those taxes and costs and that market studies, in particular, the dividend drop-off study, which takes into account of the investor's costs do not meet that requirement.¹⁵⁵⁶ The Tribunal further came to the following conclusions:¹⁵⁵⁷

- The AER made no relevant error in its choice of the period over which it considered equity ownership data. That approach was open to it, and as explained by it, reasonable.
- The AER made no relevant error in choosing the estimate that it did from within the equity ownership range. Its manner of choice, as described in its Final Decisions, was clearly open to it.
- The reliability of the tax statistics is unclear. The AER did not err in giving some reduced weight to tax statistics; nor did it err in the manner in which it took them into account, as set out in detail in its Final Decisions.

The Full Federal Court's January 2018 decision for SAPN also affirmed the AER's interpretation of the value of imputation credits.¹⁵⁵⁸ We consider this issue has been largely settled following the two Full Federal Court's decisions. The Full Federal Court found that it is not an error of construction for the AER to focus on utilisation rather than on implied market value.¹⁵⁵⁹

¹⁵⁵³ Australian Competition Tribunal, *Tribunal decision in ActewAGL – re Application by ActewAGL*[2017] ACompT 2, October 2017

¹⁵⁵⁴ Australian Competition Tribunal, *Tribunal decision in ActewAGL – re Application by ActewAGL[2017] ACompT 2*, October 2017, para. 299.

¹⁵⁵⁵ Australian Competition Tribunal, *Tribunal decision in ActewAGL – re Application by ActewAGL*[2017] ACompT 2, October 2017, para. 337.

¹⁵⁵⁶ Australian Competition Tribunal, *Tribunal decision in ActewAGL – re Application by ActewAGL[2017] ACompT 2*, October 2017, para. 338.

¹⁵⁵⁷ Australian Competition Tribunal, *Tribunal decision in ActewAGL – re Application by ActewAGL*[2017] ACompT 2, October 2017, para. 346, 347, 348,351.

¹⁵⁵⁸ Federal Court of Australia, *SA Power Networks v Australian Competition Tribunal (No 2) [2018] FCAFC 3*, Jan 2018, para. 56.

¹⁵⁵⁹ Federal Court of Australia, SA Power Networks v Australian Competition Tribunal (No 2) [2018] FCAFC 3, Jan 2018, para. 56; Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, para. 756.

A.3 Issues from the concurrent evidence session and submissions

We released an issues paper in October 2017 outlining key areas of the Guideline review process.¹⁵⁶⁰ In March 2018, we released a discussion paper on imputation credits to provide background on the matters to be discussed at the concurrent evidence session and questions to frame that discussion.¹⁵⁶¹

At the concurrent evidence session, the experts retained by relevant stakeholders expressed their views on various issues in estimating the value of imputation credits. The key issues discussed at the concurrent evidence session included:

- the AER's conceptual approach in estimating the value of imputation credits
- the definition of a benchmark entity
- the appropriate comparator companies
- the strengths and weaknesses of each source of evidence in estimating the gamma parameters

A number of general agreements were reached over the following issues:

For the purpose of estimating the distribution rate, a benchmark efficient entity should be defined as a corporate entity who pays tax at the full Australian statutory rate and does not have any foreign operations ¹⁵⁶²

The Officer model assumes the national equity markets are segmented. In a segmented market, all the equities are held by Australian investors and there is no foreign investment. Therefore, the appropriate estimate of the utilisation rate implied by the Officer model would be 1.¹⁵⁶³

We have considered the issues raised by the experts in the concurrent evidence session and also the issues proposed by the stakeholders in their submissions in coming to our draft decision on the value of imputation credits. Table 48 sets out the key issues raised by stakeholders and the experts. We also provide a summary of our response. Our detailed response to the issues raised by the experts and stakeholders is covered in sections below.

¹⁵⁶⁰ The AER, Issues paper- Review of the rate of return guidelines, October 2017

¹⁵⁶¹ The AER, Gamma discussion paper, March 2018

¹⁵⁶² CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 72.

¹⁵⁶³ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 69.

Table 48 issues on the value of imputation credits

Key point	Submission	Stakeholder	AER response
Overall approach	Limiting the review to the updating of earlier empirical analysis would not be in accordance with the Rule. The AER should re-examine the implementation of its approach to estimating its 'cash flow' interpretation of gamma.	APA ¹⁵⁶⁴ , ENA ¹⁵⁶⁵ , Spark Infrastructure ¹⁵⁶⁶	We have re-examined our conceptual framework of estimating the value of imputation credits, as well as the approach we have adopted to estimate the parameter in this draft instrument. We consider our current 'utilisation' approach is consistent with the Rules and will contributing to the achievement of the NEO and NGO.
	The value of imputation credits should be interpreted as the market value. Theta should be far below the AER's redemption rate.	Gray ¹⁵⁶⁷ , Wheatley ¹⁵⁶⁸ , Sadeh ¹⁵⁶⁹	We consider our interpretation of the value of the imputation credits is consistent with the Officer framework that the parameters need to be estimated on a pre- personal tax and cost basis. We do not consider the market value interpretation of the parameter is consistent with the framework.
	The AER's current 'utilisation' approach is consistent with the Full Federal Court's decision that is before investors' taxes and costs. The issue is largely settled.	Lally ¹⁵⁷⁰ , Partington ¹⁵⁷¹ , Ergon Energy and Energex ¹⁵⁷² , Public Interest Advocacy Centre ¹⁵⁷³ , EUAA ¹⁵⁷⁴	We agree.
	A gamma of 0.5 will result in an effective adjusted tax allowance of 15 per cent which is closer to the reality of the actual taxation	CCP16 ¹⁵⁷⁵	Based on the most recently updated empirical evidence, we consider a value of imputation credits of 0.5 would result in an

- ¹⁵⁶⁴ APA, APA submission responding to AER issues paper, 12 December 2017, p. 12.
- ¹⁵⁶⁵ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 3.
- ¹⁵⁶⁶ Spark Infrastructure, *Response to issues paper on the review of the Rate of Return Guideline*, December 2017, p.
 10.
- ¹⁵⁶⁷ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 68.
- ¹⁵⁶⁸ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 68.
- ¹⁵⁶⁹ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 68.
- ¹⁵⁷⁰ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 68.
- ¹⁵⁷¹ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 68.
- ¹⁵⁷² Ergon Energy and Energex, *Ergon Energy and Energex submission on AER Issues Paper*, 12 December 2017, p.
 7.
- ¹⁵⁷³ PIAC, *PICA letter to the AER*, December 2017, p. 2.
- ¹⁵⁷⁴ EUAA, EUAA submission to AER Rate of Return Review issues paper, December 2017, pp. 9-10
- ¹⁵⁷⁵ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 132.

overall return that is (at least) sufficient to compensate the investors of the BEE.		paid by the networks but will still overcompensate the firms given the imputation benefits.
We have re-examined our approach for estimating the value of imputation credits and updated the empirical evidence. Having considered all the information before us, we propose to increase the estimate from 0.4 to 0.5 for this draft instrument.	CCP16 ¹⁵⁷⁶ , Major Energy Users ¹⁵⁷⁷ , AIET ¹⁵⁷⁸	The actual tax paid by networks is far under the amount of tax assumed by the AER and has reservations about the assessment of utilisation and distribution rates that are used. A value of imputation credits of 0.4 is much too conservative and needs to be increased.
We have considered the composition of the empirical evidence used to update the value of imputation credits in this draft instrument. We have considered the relative strength and weaknesses of each piece of empirical evidence in estimating the value of imputation credits. We also define the benchmark efficient entity. However, we consider it is difficult to identify a set of companies that are most consistent with the BEE. We consider our approach for estimating the value of imputation credits for the benchmark efficient entity is consistent with the approach we have taken in estimating the other parameters in the rate of return.	ATCO Gas Australia ¹⁵⁷⁹ , Cheung Kong Infrastructure ¹⁵⁸⁰	The AER should consider the composition of the empirical evidence used to update the value of imputation credits as well as that of the benchmark efficient entity. This includes consideration of the suitability of adopting the distribution rate of firms in the ASX 20 or the market as a whole as a proxy for the benchmark efficient entity given that various business models exist among listed businesses. There should be consistency across the empirical evidence used to estimate the rate of return. The AER should weaknesses of each empirical evidence.
We have transparently set out how we have assessed the empirical evidence for estimating the parameters of the value of imputation credits and how we have chosen the point estimate for the vale of imputation credits.	Cheung Kong Infrastructure ¹⁵⁸¹ , Energy Networks Association ¹⁵⁸² , AusNet Services ¹⁵⁸³ , Spark Infrastructure ¹⁵⁸⁴	The AER should set out clearly how the updated empirical analysis has been used and distilled into a point estimate

¹⁵⁷⁶ CCP (subpanel 16), CCP Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p. 113.

 ¹⁵⁷⁷ Major Energy Users, Submission by the MEU to the review of the rate of return guideline, 18 December 2017, p.
 17.

¹⁵⁷⁸ Agriculture industries energy taskforce, *submission to AER discussion paper profitability measures for regulated gas and electricity*, December 2017, p. 12.

¹⁵⁷⁹ ATCO Gas Australia, Response to Revie of Rate of Return Guideline – Issues Paper, 12 December 2017, p. 9.

¹⁵⁸⁰ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, p. 6.

¹⁵⁸¹ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6

¹⁵⁸² ENA, Response to AER Issues Paper, 12 December 2017, pp. 35-38

¹⁵⁸³ AusNet Services, Submission to AER issues paper, December 2017

¹⁵⁸⁴ Spark Infrastructure, *Response to issues paper on the review of the Rate of Return Guideline*, December 2017, p. 10.

	Any change in the approach or parameter estimate should only occur if there is robust and reliable evidence that the parameter estimate is no longer the best estimate and will not give rise to rate of return or gamma that contributes to the achievement of the NEO/NGO to the greatest degree.	Joint Energy Networks ¹⁵⁸⁵	We have given our reasoning for the departures from some specific aspects of the 2013 Guidelines in this draft decision. In coming to the draft decision on the value of imputation credits, we have taken all the available information into account and we consider our current estimate for gamma is most likely to contribute to the achievement of the NEO/NGO.
	Theta is an economy wide measure and the distribution rate is firm specific and based on dividend distribution practice identified with the BEE.	APGA ¹⁵⁸⁶	We agree that theta (or the utilisation rate) is market wide parameter while the distribution rate is firm specific and should be determined based on the expected practice of the BEE.
	The AER's current approach values franking credits retained at zero, which is likely to downward bias the estimate for the value of imputation credits.	Partington ¹⁵⁸⁷	We agree. However, we consider assuming retained imputation credits have no value might be a reasonable assumption to the extent that imputation distribution rates are expected to remain constant overtime.
Benchmark efficient entity	The BEE pays at the statutory corporate tax rate and operates in Australia.	Agreed by the experts in the concurrent evidence session ¹⁵⁸⁸	We define a benchmark efficient entity to be taken as a domestic entity that operates within Australia and have 'a similar degree of risk' as that which applies to the particular service provider in providing its 'regulated' services. We note that as the NER and NGR are drafted, the benchmark firm pays tax at the relevant Australian corporate tax rate.
The distribution rate	Most of Lally's top 20 listed firms are banks or multinationals. The distribution rate estimate from Lally's 20 largest listed firms	ENA ¹⁵⁸⁹ , NSG ¹⁵⁹⁰ , APGA ¹⁵⁹¹ , Cheung Kong Infrastructure ¹⁵⁹² ,	Given that the ATO tax statistics contains significant unresolved discrepancies, we depart from 2013 Guidelines and consider the data

- ¹⁵⁸⁵ Joint Energy Networks, Submission on Rate of Return Guideline Review, 4 May 2018, p. 6.
- ¹⁵⁸⁶ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 15.
- ¹⁵⁸⁷ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 74.
- ¹⁵⁸⁸ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 71.
- ¹⁵⁸⁹ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 11.
- $^{\rm 1590}\,$ NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18
- ¹⁵⁹¹ APGA, Submission to the Issues Paper, 12 December 2017, p. 10
- ¹⁵⁹² Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6

does not give an accurate reflection for the BEE.	Sadeh ¹⁵⁹³ , Wheatley ¹⁵⁹⁴	from the financial statements for the top 20 listed firms gives a distribution rate that is likely to best reflect the expected distribution rate of a BEE based on Lally's advice.
The AER should be explicit in how it is representing the BEE when it determines the distribution rate estimate.	CCP16 ¹⁵⁹⁵ , APGA ¹⁵⁹⁶	We have had regard to the estimates of the distribution rate from varies empirical evidence. Our estimate of the distribution rate of 0.83 is based on our view an estimate of 0.88 is appropriate and then using a lower estimate of 0.83 that is internally consistent with a gamma value of 0.5 and utilisation rate of 0.6 We expect our current approach for estimating the distribution rate would give a best estimate for the parameter.
The service providers would have a lower distribution rate than the average entity because they need retain large proportion of profit to reinvest it into the businesses.	Sadeh ¹⁵⁹⁷	The evidence before us does not support this. The average real growth rate in RAB between 2013- 14 and 2016-17 was approximately 1.9 per cent per annum. Moreover, the service providers are compensated for the depreciation on their regulated asset base through the allowed revenue they get from the regulator. We consider the BEE can also retain funds while distributing credits through the use of dividend reinvestment programs, or raise new equity capital in other ways.
The distribution rate could be estimated from the equation below: $1 - \frac{I - D}{RAB * WACC}$	CRG ¹⁵⁹⁸	This approach is based on the assumption that apart from the revenue compensation it gets from the regulator for asset depreciation, a BEE could only fund its investment through retained earnings. However, firms have access to different sources of finance to fund their investment.
Lally's top 20 firms have material foreign income and it would not	Gray ¹⁵⁹⁹ , Cheung Kong	Lally considers the effect of including the firms with foreign

- ¹⁵⁹⁵ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 10.
- ¹⁵⁹⁶ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 15.
- ¹⁵⁹⁷ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 100.
- ¹⁵⁹⁸ CRG, Submission to the AER on its Rate of Return Guideline Review, May 2018, p. 59.
- ¹⁵⁹⁹ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 76.

¹⁵⁹³ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 77.

¹⁵⁹⁴ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 78.

accord with the definition of the BEE. An estimate based on Lally's analysis will overestimate the distribution rate. Infrastructure ¹⁴⁰⁰ , ENA ¹⁵⁰¹ , APGA ¹⁵⁰² operations is to underestimate the distribution is to underestimate the distribution rate for the benchmark firm. Lally's approach for estimating the distribution rate form financial statements is subject. Gray ¹⁶⁰⁴ , ENA ¹⁰⁰⁵ We consider the evidence supports that the benchmark firm. The distribution rate form financial statements is subject. Gray ¹⁶⁰⁴ , ENA ¹⁰⁰⁵ We consider the data from the financial statements is audited and should produce a significantly more reliable estimate for the distribution rate for the ATO tax data. The distribution rate should be based on all equity and the estimate for the listed equity has to be the upper bound. Sadeh ¹⁶⁰⁰⁵ We consider the distribution rate is a firm- isection rate or the benchmark efficient entity. We do not consider the estimate for the listed equity is an upper bound. The AER should identify the comparator companies that are most consistent with the BEE in estimating the distribution rate. APGA ¹⁶⁰⁷ , CCP16 ¹⁶⁶⁸ In practice, it is difficult to construct a data set for such companies. The AER should look at relevant data an taxation and imputation policies of the relevant businesses. CCP16 ¹⁶⁶⁹ In practice, it is difficult to distribution rate of tive companies from the same industry as the BEE and lound a distribution of 10 for those firms for which the data is available. ¹⁶⁰ We have also examined the distribution rate of the tive companies from the is cosee to Lally's estimates. The dis			
Lally's approach for estimating the distribution rate from financial statements is subject to various difficulties requiring adjustment.We consider the data from the financial statements is audited and shuld produce a significantly more reliable estimate for the distribution rate compared to the ATO tax data.The distribution rate should be based on all equity and the estimate for the listed equity has to be the upper bound.We consider the distribution rate is a firm- specific rather than a market-wide parameter. We agree that listed equity is likely to better reflect the distribution rate for the listed equity is to be the upper bound.The AER should identify the comparator companies that are most consistent with the BEE in estimating the distribution rate.APGA1607, CCP161000In practice, it is difficult to construct a data set for such companies.The AER should look at relevant data on taxation and imputation policies of the relevant businesses.CCP161000Lally examined the distribution rate of five companies from the same industry as the BEE and found a distribution rate of the distribution rate of the distribution rate.The distribution rate could beCCP161000Lally examined the distribution rate of the companies from the same industry as the BEE and found a distribution rate of the distribution rate of the distribution rate.	operations is to underestimate the distribution rate for the benchmark firm. ¹⁶⁰³ We consider the evidence supports that the benchmark firm with no foreign income should be able to distribute at least 88 per cent of its imputation credits.	Infrastructure ¹⁶⁰⁰ , ENA ¹⁶⁰¹ , APGA ¹⁶⁰²	accord with the definition of the BEE. An estimate based on Lally's analysis will overestimate the distribution rate.
The distribution rate should be based on all equity and the estimate for the listed equity has to be the upper bound.Sadeh ¹⁶⁰⁶ We consider the distribution rate is a firm- specific rather than a market-wide parameter. We agree that listed equity is likely to better reflect the distribution rate for the benchmark efficient entity. We do not consider the estimate for the listed equity the comparator companies that are most consistent with the BEE in estimating the distribution rate.APGA ¹⁶⁰⁷ CCP16 ¹⁶⁰⁰ In practice, it is difficult to construct a data set for such companies.The AER should look at relevant data on taxation and imputation policies of the relevant businesses.APGA ¹⁶⁰⁷ CCP16 ¹⁶⁰⁰ In practice, it is available. ¹⁶¹⁰ We have also examined the distribution rate of five companies from the same industry as the BEE and found a distribution of 1 for those firms for which the data is available. ¹⁶¹⁰ We have also examined the distribution rate of the five companies from their inacial reports. Our number is close to Lally's estimates.The distribution rate could beCCP16 ¹⁶⁰¹ A key issue with using the industry	We consider the data from the financial statements is audited and should produce a significantly more reliable estimate for the distribution rate compared to the ATO tax data.	Gray ¹⁶⁰⁴ , ENA ¹⁶⁰⁵	Lally's approach for estimating the distribution rate from financial statements is subject to various difficulties requiring adjustment.
The AER should identify the comparator companies that are most consistent with the BEE in estimating the distribution rate.APGA1607, CCP161608In practice, it is difficult to construct a data set for such companies.The AER should look at relevant data on taxation and imputation policies of the relevant businesses.Lally examined the distribution rate of five companies from the same industry as the BEE and found a distribution of 1 for those firms for which the data is available. ¹⁶¹⁰ We have also examined the distribution rate of the five companies from their financial reports. Our number is close to Lally's estimates.The distribution rate could beCCP161611A key issue with using the industry	We consider the distribution rate for the BEE should be estimated from listed equity given Lally's advice that the distribution rate is a firm- specific rather than a market-wide parameter. We agree that listed equity is likely to better reflect the distribution rate for the benchmark efficient entity. We do not consider the estimate for the listed equity is an upper bound.	Sadeh ¹⁶⁰⁶	The distribution rate should be based on all equity and the estimate for the listed equity has to be the upper bound.
The AER should look at relevant data on taxation and imputation policies of the relevant businesses.CCP16 ¹⁶⁰⁹ Lally examined the distribution rate of five companies from the same industry as the BEE and found a distribution of 1 for those firms for which the data is available. ¹⁶¹⁰ We have also examined the distribution rate of the five companies from their financial reports. Our number is close to Lally's estimates.The distribution rate could beCCP16 ¹⁶¹¹ A key issue with using the industry	In practice, it is difficult to construct a data set for such companies.	APGA ¹⁶⁰⁷ , CCP16 ¹⁶⁰⁸	The AER should identify the comparator companies that are most consistent with the BEE in estimating the distribution rate.
The distribution rate could be CCP16 ¹⁶¹¹ A key issue with using the industry	Lally examined the distribution rate of five companies from the same industry as the BEE and found a distribution of 1 for those firms for which the data is available. ¹⁶¹⁰ We have also examined the distribution rate of the five companies from their financial reports. Our number is close to Lally's estimates.	CCP16 ¹⁶⁰⁹	The AER should look at relevant data on taxation and imputation policies of the relevant businesses.
	A key issue with using the industry	CCP16 ¹⁶¹¹	The distribution rate could be

- ¹⁶⁰⁵ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 11.
- ¹⁶⁰⁶ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 100.
- ¹⁶⁰⁷ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16.
- ¹⁶⁰⁸ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131.
- ¹⁶⁰⁹ CCP (subpanel 16), CCP Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, pp. 9-10
- $^{1610}\,$ M. Lally, Review of the AER's views on gearing and gamma, 7 May 2018, p. 20.
- ¹⁶¹¹ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131.

¹⁶⁰⁰ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6

¹⁶⁰¹ ENA, *Response to AER Issues Paper*, 12 December 2017, p. 37.

¹⁶⁰² APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16.

¹⁶⁰³ M. Lally, *Issues in the estimation of gamma*, April 2017, pp. 9-12.

¹⁶⁰⁴ CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence, 21 April 2018, p. 77.

	calculated using actual behaviour of the individual network companies or a relevant group of companies that fit within the AER's definition of the BEE.		data is the set of firms is not large.
	The distribution rate for the BEE could be implied from the dividend payout rates of some suitable comparators.	Gray ¹⁶¹²	We do not consider the imputation credits distribution rate could be implied from the dividend payout rate. They are two fundamentally different concepts.
ATO tax statistics	The AER should place limited weight on the estimates of the parameters from the ATO tax statistics.	CCP16 ¹⁶¹³	We agree.
	The AER should continue to seek refinement of the tax return data given it should be useful for estimating the payout rate and the utilisation rate.	CCP16 ¹⁶¹⁴	We have attempted to refine the ATO tax data. Consistent with Hathaway's findings we also find that there is a significant discrepancy associated with the tracking of imputation credits in the ATO data.
	The ATO tax statistics gives a direct estimate for the value of imputation credits.	ENA ¹⁶¹⁵ , NSG ¹⁶¹⁶ , CEPA ¹⁶¹⁷ , Cheung Kong Infrastructure ¹⁶¹⁸ , Gray ¹⁶¹⁹ , Joint Energy Networks ¹⁶²⁰ , ATCO ¹⁶²¹	We consider this approach for estimating gamma involves a number of potential issues and would not give a reliable estimate for the value of imputation credits for the benchmark efficient entity.
	The ATO tax statistics gives an estimate of an upper-bound for a 'utilisation' gamma.	ATCO ¹⁶²²	We do not consider the ATO tax statistics would give either a reliable estimate or an upper bound estimate for gamma given it is an estimate and given the reliability concerns with the ATO data.

- ¹⁶¹² The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, pp. 112-114.
- ¹⁶¹³ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 10.
- ¹⁶¹⁴ CCP (subpanel 16), *CCP Submission to the AER on its Rate of Return Guideline Issues Paper*, December 2017, pp. 9-10
 - CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 133.
- ¹⁶¹⁵ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 12. ENA, Submission on rate of return issues paper, attachment C- Letter-Tax statistics, 12 December 2017
- ¹⁶¹⁶ NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 3.
- ¹⁶¹⁷ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 74
- ¹⁶¹⁸ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6
- ¹⁶¹⁹ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 74.
- ¹⁶²⁰ Joint Energy Networks, Submission on Rate of Return Guideline Review, 4 May 2018, pp. 6-7.
- ¹⁶²¹ ACTO Gas, Submission on rate of return guideline review, 4 May 2018
- ¹⁶²² ATCO Gas, Submission on rate of return guideline review, 4 May 2018

	The ATO tax statistics could give a reliable gamma estimate for the BEE once the market wide distribution rate is adjusted.	Wheatley ¹⁶²³ , APA ¹⁶²⁴ , APGA ¹⁶²⁵	The adjusted estimate of value of imputation credits from the ATO tax statistics indicates a value of slightly over 0.5. This is consistent with the value of imputation credits in the draft instrument.
	The distribution rate for the BEE could be inferred from the service providers' actual imputation credits distribution rate	Lally ¹⁶²⁶	We consider this approach may promote the businesses to adopt a distribution policy that would maximise their regulatory revenue. This may not necessarily be in the long term interest of the consumers, or consistent with achieving the NEO/NGO.
The utilisation rate	The model assumes the national equity markets are segmented. This implies a utilisation rate of 1as all the assets in the equity markets are owned by domestic investors and there is no foreign investment.	Lally, Gray ¹⁶²⁷ , CRG ¹⁶²⁸	We agree. However, we consider the assumption of no foreign investment and no foreign investors does not reflect the empirical reality.
	If one assumes that the presence of foreign investors must be reflected in the estimate of the utilisation rate, the ABS equity ownership data provides the natural estimate of the parameter.	Lally ¹⁶²⁹	We agree.
	The utilisation rate of the BEE should be estimated from the ABS data for all equity	CCP16 ¹⁶³⁰	We agree.
	The AER should examine the composition of the shareholder base of the BEE in estimating the utilisation rate	ENA ¹⁶³¹	We have considered the shareholder base of the BEE. The Officer CAPM assumes all the assets in the equity market are owned by domestic investors. This implies the BEE is owned by domestic investors and hence it suggests a utilisation of 1. However, we consider a utilisation rate of 1 would not reflect the reality.

- ¹⁶²³ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 116.
- ¹⁶²⁴ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, pp. 11-12.
- ¹⁶²⁵ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 18.
- ¹⁶²⁶ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 95.
- ¹⁶²⁷ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, pp. 106-108.
- ¹⁶²⁸ CRG, Submission to the AER on its Rate of Return Guideline Review, May 2018, p. 60.
- ¹⁶²⁹ CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence, 21 April 2018, p. 76.
- ¹⁶³⁰ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 132.
- ¹⁶³¹ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 89.

Such matters are only significant to the extent that such non redemptions are significant and there is no evidence showing that. We also note we have used a utilisation rate at the bottom of our suggested range from ABS data for all equity and below the current point estimate from this data.	ENA ¹⁶³² , Cheung Kong Infrastructure ¹⁶³³ , Wheatley ¹⁶³⁴	The utilisation rate estimate from the equity ownership approach does not take into account investors who receive imputation credits but do not redeem them.	
We consider such data issues would also apply to the alternative approaches. For example the estimates based on ATO data also require filtering that may be subject to sampling error	ENA1635, Cheung Kong Infrastructure ¹⁶³⁶ , APGA ¹⁶³⁷	The equity ownership estimates of the utilization rate based on the ABS data require filtering and adjustment, and may be subject to sampling error. The AER should seek further clarity to the issue on ABS data.	The equity ownership approach
The updated ABS data is considered to be more reliable than the data released earlier by the ABS, given that the purpose for the ABS data revision is to improve the data quality. As a result, we consider the updated data would produce a more reliable estimate of the utilisation rate than the utilisation rate estimated from the earlier ABS release.	ENA ¹⁶³⁸	The recent data released by the ABS raises more questions about the reliability of the equity ownership estimates than were apparent at the time of the 2013 Guidelines.	
The data from the most recent December 2017 release which is not based on 'backcasting' exercise suggests a utilisation rate of 0.65 for all equity. This is within the range of 0.6 and 0.7 of the utilisation rate.	ENA ¹⁶³⁹	The revision to the ABS data is based on a 'backcasting' exercise that could be less reliable than the estimates that it made at the time the historical data was collected	
We consider the utilisation rate consistent with the Officer model is 1. However, we do not consider this reflects the empirical reality.	Gray ¹⁶⁴⁰ , NSG ¹⁶⁴¹ , APGA ¹⁶⁴² , Partington ¹⁶⁴³	The model assumes the equity market is either complete segregation or complete integration, the equity ownership approach in estimating the utilisation rate is not consistent	

- ¹⁶³⁵ ENA, *Response to AER Issues Paper*, 12 December 2017, p. 37.
- ¹⁶³⁶ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6
- ¹⁶³⁷ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16
- ¹⁶³⁸ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 91, p. 11.
- ¹⁶³⁹ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 91.
- ¹⁶⁴⁰ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 108.
- $^{\rm 1641}\,$ NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18.
- ¹⁶⁴² APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 15.
- ¹⁶⁴³ EPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 69.

¹⁶³² ENA, *Response to AER Issues Paper*, 12 December 2017, p. 36.

¹⁶³³ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6

 ¹⁶³⁴ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 76.

		with any equilibrium models	
We consider the 45 day rule does not have a material impact on the utilisation rate. This is supported by Lally.1647 We have not been presented with any data that would suggest the 45 day rule is having any impact on the utilisation of imputation credits.	Gray ¹⁶⁴⁴ , APGA ¹⁶⁴⁵ , NSG ¹⁶⁴⁶	The AER should have some reliance on the equity ownership data while recognising it is an upper bounds to the extent that it does not factor in the 45-day rule. The AER should quantify the impact of this rule.	
We agree.	CCP16 ¹⁶⁴⁸	The ABS data gives an appropriate reference point for estimating the market wide parameter of the utilisation of imputation credits.	
In light of the Full Federal Court's decision, we remain of the view our interpretation of the value of the imputation credits is open to us and consistent with the Officer framework under which the parameters need to be estimated on a pre-personal tax and cost basis. The estimates from the market value studies, which is on a post personal tax and personal basis basis, is not consistent with the allowed rate of return parameters and the post (corporate) tax framework in the NER/NGR.	NSG ¹⁶⁴⁹	The value of imputation credits should be interpreted as the market value and the utilisation rate should be estimated from market value studies.	
We agree.	CCP16 ¹⁶⁵⁰	The role for the market based studies of the 'value of imputation credits' in the context of the building block regulatory regime is limited.	Implied market value estimates

- ¹⁶⁴⁴ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 115.
- ¹⁶⁴⁵ APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, pp. 16-17
- ¹⁶⁴⁶ NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18.
- ¹⁶⁴⁷ M. Lally, *Gamma and the ACT Decision*, May 2016, pp. 20-21.

- ¹⁶⁴⁹ NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18.
- ¹⁶⁵⁰ CCP (subpanel 16), CCP Submission to the AER on its Rate of Return Guideline Issues Paper, December 2017, p. 113.

¹⁶⁴⁸ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131.

CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 130.

A.4 Reasons for approach

In this section, we discuss the reasons for reaching the draft instrument value of 0.5. In particular, we provide detailed analysis to the issues raised in the stakeholder submissions and also the issues raised by the experts in the concurrent evidence session. During the development of the Guideline, we also commissioned expert advice on the value of imputation credits from Dr Martin Lally.

A.4.1 The building block framework

Under the NER/NGR, we employ a building block framework to estimate revenue for service providers. The building block framework sets out how to estimate the various components (that is, 'building blocks') that make up a total revenue allowance.¹⁶⁵¹ The function of this building block revenue estimate is to determine the revenue that a service provider requires to:

- fund its operating expenses.
- achieve adequate returns to raise debt and equity in order to finance its capital investments. This is made up of a rate of return on capital to compensate investors for the risks of investment. It also includes a return of capital (depreciation), which gradually returns the initial principal of the investment, and subsequent investments, back to investors.
- pay its tax liability.
- reflect any revenue increments or decrements from incentive mechanisms in the design of the regulatory regime.

Importantly, the building block framework is intended to compensate the service provider (and its investors) only for costs incurred by the service provider and not by its investors; that is, the framework is on a post-company before-personal-tax and before-personal-costs basis. Handley described this consideration as follows:¹⁶⁵²

The post-tax basis of the regulatory framework can be more fully described as an after-company-before-personal-tax framework. In other words, cash flows and returns are to be measured after company taxes but before personal taxes. By definition, this means that allowed revenues should include compensation for corporate taxes incurred by the regulated firm but not for personal taxes incurred by the firm's shareholders. Similarly, allowed revenues should include compensation for prudent, efficient costs incurred by the regulated firm but not for costs (including personal transactions costs) incurred at the shareholder level. Note, this does not mean that personal taxes and costs are being ignored or assumed not to exist – rather there is no need to explicitly include them in the modelling framework.

¹⁶⁵¹ NER, cll. 6.4.3, 6A.5.4; NGR, r. 76.

¹⁶⁵² J. Handley, Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits, 29 September 2014, pp. 4–5 and footnote 2.

The regulatory WACC framework is an after-company-before-personal-tax framework which requires explicit modelling of cash flows and returns after allowing for company tax but avoids most of the complications associated with having to model personal taxes - one complication which remains of course, is gamma. If one wanted to explicitly model personal taxes then an after-company-after-personal-tax WACC framework could be used instead.

...

In particular, Handley advised that the Officer framework provides the basis for the building block framework in the NER/NGR, and that the before-personal-tax and before-personal-costs interpretation is consistent with Officer.

There is no consensus among experts or regulators on the value of imputation credits or the techniques to use to estimate it. In the second current evidence session, the experts expressed their views on the issues relating to gamma estimation but without reaching a precise view on how they should be resolved. In particular, the experts retained by the businesses, Gray and Sadeh, expressed their different views on the AER's interpretation of the value of imputation credits proceeding on the basis of a 'utilisation' approach, but failed to provide any other "materially preferable" method to estimating gamma.1653 Some stakeholders in their submissions propose that the AER should examine its current interpretation of gamma in the review of the rate of return guideline.1654

For this draft instrument, we have re-examined the conceptual task of estimating the value of imputation credits. In response to a number of service providers' and experts' view that we prefer our conceptual valuation relative to a market mechanism derived value, and their concerns around our definition of gamma, we note:

- There is no market for imputation credits and therefore there is no directly observable market price.¹⁶⁵⁵
- The value of imputation credits as estimated through a dividend drop off study:
 - is not a post company tax value before personal taxes and personal transaction costs, although Lally in his recent report to the AER argues this approach could give an estimate that is consistent with the model as long as it is appropriately adjusted¹⁶⁵⁶

 ¹⁶⁵³ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 68.

¹⁶⁵⁴ ENA, *Response to discussion papers and concurrent expert evidence sessions*, 4 May 2018, p. 3; APA, *APA submission responding to AER issues paper*, 12 December 2017, p. 12; Spark Infrastructure, *Response to issues paper on the review of the Rate of Return Guideline*, December 2017, p. 10.

¹⁶⁵⁵ We note that if there was a post-tax market for imputation credits they should trade for approximately their face value as this is the amount of post-tax cash flow from the government they entitle the claimant to receive from the government.

¹⁶⁵⁶ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 25.

- is not clearly measuring a value to long term investors supplying capital to the benchmark efficient entity (even when adjusted for differential tax impacts)
- o is subject to significant uncertainty
- A number of limitations of these market value studies are discussed further in sections A.4.3
- Our definition of the utilisation value is consistent with Officer Framework underpinning the Rules that uses a post-tax Vanilla WACC, is consistent with the way we estimate the required return on equity, and is supported by Handley. Lally also considers our approach to imputation credits is consistent with the post-tax framework in the NER/NGR.¹⁶⁵⁷
- The explicit recognition that this is a pre personal taxes and pre personal transaction costs value is simply explicit recognition of something implicit in the Officer Vanilla WACC framework. This framework determines the required return to the company (i.e. before personal taxes and personal transaction costs are incurred) consistent with the definition of the after-tax net cash flows set out in Officer.¹⁶⁵⁸
- Our definition is consistent with the 'value' concept in the Rules given the Rules are requiring the AER to estimate a post-tax nominal Vanilla WACC consistent with Officer formula (12) in his 1994 paper, which as stated by Officer is one particular definition of "a company's after-tax cost of capital (WACC)" and "is determined by the definition used of after-tax operating income or really after-tax net cash flows".¹⁶⁵⁹ As these after-tax net cash flows are at the company level they are before personal taxes and personal transaction costs. This is supported by the advice of Handley and Lally.¹⁶⁶⁰
- The Full Federal recently found:¹⁶⁶¹

the expression "the value of imputation credits" is to be construed as a whole, in its context and having regard to the subject matter of the exercise. It would be an error to limit attention to the word "value" and give it a meaning in isolation.

••••

¹⁶⁵⁷ M. Lally, *Gamma and the ACT Decision*, May 2016, pp. 11-12.

¹⁶⁵⁸ R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, p. 7.

¹⁶⁵⁹ R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, p. 6.

J. Handley, Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits, 29 September 2014, pp. 4–5 and footnote 2; M. Lally, Gamma and the ACT Decision, May 2016, pp. 7, 11, 12.

¹⁶⁶¹ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, p. 215 - 216.

We accept the AER's submission that the context is the determination of a regulated return using a post-tax revenue model based on a nominal WACC. We accept the AER's submission that the Rules require consistency in the way the relevant building blocks interact, that is, a post-company tax and prepersonal tax and personal costs basis.

....

it is not an error of construction for the AER to focus on utilisation rather than on implied market value.

Contrary to the Full Federal Court's finding, Ausgrid submitted a report from Frontier Economics that argues the value of imputation credits should be interpreted as the market value and the utilisation rate should be estimated from market value studies.¹⁶⁶² We note the material on the value of imputation credits in the report submitted by Ausgrid is substantively similar to the Frontier report submitted by TransGrid in August 2017.¹⁶⁶³ Our response to the Frontier's report is set out in detail in attachment 4 of the AER's draft determination for TransGrid.¹⁶⁶⁴ The arguments in the report do not change our view on the conceptual framework for the estimation of the value of imputation credits.

We note that while our approach here to updating the estimates for theta (the utilisation rate) and the distribution rate is consistent with an incremental review, we did consider if departure from an incremental review was required in order to meet our legal requirements. We have determined it is not as we consider our current 'utilisation' approach is consistent with the Rules and will contributing to the achievement of the NEO and NGO.

A.4.2 The conceptual framework for the value of imputation credits

In this instrument, we view the value of imputation credits as the proportion of company tax returned to investors through the utilisation of imputation credits. Moreover, as noted above, it is the post-company pre-personal tax value of imputation credits to those investors that we seek to estimate. Our approach to interpreting and estimating the value of imputation credits is guided in the first instance by the conceptual framework developed by Officer.¹⁶⁶⁵ This is because:

¹⁶⁶² Frontier Economics, *Estimation of certain aspects of the allowed rate of return*, April 2018, p. 182.

¹⁶⁶³ Frontier, *Estimating gamma within the regulatory context*, August 2017,

¹⁶⁶⁴ The AER, Draft decision: TransGrid transmission determination 2018 to 2023- Attachment 4: Value of imputation credits, September 2017

¹⁶⁶⁵ R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, pp. 1–17.

- The construction of the tax building block mirrors the treatment of imputation credits in the framework developed by Officer, including through use of the parameter denoted by the Greek letter 'gamma'.¹⁶⁶⁶
- Handley advised that Officer's definition of the nominal vanilla rate of return provides the basis for the rate of return framework in the NER/NGR.¹⁶⁶⁷
- The NER/NGR require that we determine the rate of return on a nominal vanilla basis that is consistent with our estimate of the value of imputation credits.¹⁶⁶⁸

We do not contend that the Officer Paper is a "statute or a code". However, as the Officer Paper underpins the inclusion of gamma in the corporate income tax formula in NER 6.5.3 and NGR 87A, it is fundamental to a coherent understanding of the role of gamma in the regulatory scheme.

The Officer Paper specifically identified gamma in its WACC formulae to be the "proportion of tax collected from the company which gives rise to the tax credit associated with a franked dividend": It directly supports an interpretation of gamma which is focused on the utilisation or redemption of imputation credits, and an approach to theta which seeks to identify the proportion of investors that are eligible to utilise distributed imputation credits. So much is confirmed by Handley, who states:¹⁶⁶⁹

It is clear from Monkhouse (1996) that the second parameter refers to the utilisation value of a distributed imputation credit. This parameter is commonly denoted and called theta. It is also clear from the post-tax basis of the regulatory framework (and the Officer and Monkhouse WACC frameworks) that the item of interest is more precisely described as the after-company-before-personal-tax utilisation value of a distributed imputation credit.

The Officer Paper makes clear that gamma is:

- (a) the proportion of tax collected from the company which gives rise to the tax credit associated with a franked dividend; *which is*
- (b) the value of a dollar of tax credit to the shareholder; with the result that
- (c) if the shareholder can fully utilise the imputation tax credits then the value of gamma =1. 1670

Our approach to gamma, as drawing upon the Officer Paper, advances the NEO and the NGO and does not detract from those objectives. The purpose of including the

¹⁶⁶⁶ R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, equation 2.

¹⁶⁶⁷ J. Handley, *Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits*, 29 September 2014, pp. 7–8.

¹⁶⁶⁸ NER, cll. 6.5.2, 6A.6.2; NGR, r. 87.

¹⁶⁶⁹ Handley, *Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits*, 29 September 2014, pp. 9, 17.

¹⁶⁷⁰ R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, p. 4.

gamma parameter in the corporate income tax formula is to ensure that investors are appropriately compensated having regard to the combined value of dividends, capital gains and imputation credits they receive.

The Officer Paper included the following explanation of gamma at pp. 2-4:

Under an imputation system, credit is given to shareholders for the company tax implicitly levied on their dividend receipts, i.e. dividends paid after company tax has been levied which implies that the dividends have been taxed at the company level. Under a full imputation tax system, tax that is implicitly being levied on the dividends can be credited against any further tax liabilities of the shareholder (the recipient of the dividend).

The proportion of company tax that can be fully rebated against personal tax liabilities is best viewed as personal income tax collected at the company level. In effect, the tax collected at the company level is a mixture of personal tax and company tax, the company tax being that proportion of the tax collected which is not credited (rebated) against personal tax. If all the collection of tax from a company is rebateable (in the Australian terminology if all the franking credits can be used against personal tax liabilities), then for that company's shareholders company tax is effectively eliminated. The tax the company pays is simply the shareholders' personal income tax being collected at the company level.

The amount of tax collected from the company by the government is found by applying the effective tax rate (T) to the operating income less interest, i.e. $X_{o} - X_{D}$.

This amount, i.e. $T(X_o - X_o)$ represents the amount of tax collected from the company but not all of this is company tax. A proportion (γ) of the tax collected from the company will be rebated against personal tax and, therefore, is not really company tax but rather is a collection of personal tax at the company level. Therefore, if we wish to define the effective company tax collection, we need to reduce T by the proportion γ .

In these circumstances, the effective level of company tax paid is defined by:

 $XG = T(X_{\scriptscriptstyle O} - X_{\scriptscriptstyle D}) \text{ - } \gamma \ . \ T(X_{\scriptscriptstyle O} - X_{\scriptscriptstyle D})$

 $= T (X_{o} - X_{D}) (1-\gamma) (2)$

where

T is the tax rate effective for the definition of assessable income as defined in (2), it is the effective rate which is levied at the company level and it is a mixture of company tax $T(1-\gamma)$ and personal tax, T. γ , i.e. $T = T(1-\gamma) + T\gamma$. Thus γ is the proportion of tax collected from the company which gives rise to the tax credit associated with a franked dividend. This franking credit can be utilized as tax

credit against the personal tax liabilities of the shareholder. γ can be interpreted as the value of a dollar of tax credit to the shareholder.

The Officer Paper set out the following example of its interpretation of gamma:¹⁶⁷¹

For example, if the shareholder can fully utilize the imputation tax credits then ("value") $\gamma = 1$, e.g. a superfund or an Australian resident personal taxpayer. On the other hand a tax exempt or an offshore taxpayer who cannot utilize or otherwise access the value in the tax credit will set $\gamma = 0$.

While the above passages of the Officer Paper describe gamma using slightly different permutations of language, it is clear that, at its core, gamma is conceptualised by Officer as being concerned with investors' utilisation of tax credits.

Officer describes gamma in different ways, and this is a potential source of ambiguity regarding what the parameter represents and therefore how one might estimate it in practice. Whilst Handley acknowledged that Officer describes gamma in seemingly different ways, he advised that, when examined closely, there is no ambiguity in the meaning of Officer. Handley advised:¹⁶⁷²

Similarly, Officer has described gamma in seemingly different ways. For example he refers to:

"A proportion (γ) of the tax collected from the company will be rebated against personal tax."

and shortly thereafter:

" γ can be interpreted as the value of a dollar of tax credit to the shareholder."

But again, there is no ambiguity. These terms can and have been used interchangeably because the underlying source of value of an imputation credit to shareholders is the consequent reduction in personal taxes in recognition of taxes that were previously paid at the corporate level. In other words, within the Officer framework, it is clear that gamma represents the utilisation or redemption value of imputation credits and this value corresponds to the proportion of company tax which is in effect a prepayment of personal tax by the company on behalf of its shareholders. It is this identification of the personal tax component of the company tax paid which is the central idea of the paper.

In other words, gamma in the Officer framework represents the proportion of company tax that is returned to investors through the utilisation of imputation credits and this is the value of imputation credits to investors. This is the interpretation of the value of imputation credits we adopted in the 2013 Guidelines.

¹⁶⁷¹ R. Officer, 'The cost of capital of a company under an imputation system', *Accounting and finance*, vol. 34(1), May 1994, p. 4.

¹⁶⁷² J. Handley, *Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits*, 29 September 2014, p. 9.

For this draft instrument, we have reconsidered all the information before us. We propose to continue to adopt the interpretation of gamma as the proportion of company tax that is returned to investors through the utilisation of imputation credits. This interpretation is consistent with the Officer framework and is supported by advice from Handley. We consider our approach is consistent with the Officer framework and our estimation of the required return on equity. We consider what matters from a value to investor's perspective is the face value of imputation credits expected to be utilised as this reflects both the cost of these imputation credits to the service providers and the benefit that investors receive from these credits when utilised to reduce their tax payable (or to receive a refund from the government). The post company (prepersonal) tax value of an imputation credit when utilised is approximately¹⁶⁷³ its face value.

The Officer framework assumes all free cash flows (including imputation credits) are fully paid out each period. That is, the Officer framework is a 'perpetuity' framework. However, in reality not all imputation credits are necessarily paid out each period, nor are all other free cash flows necessarily paid out.¹⁶⁷⁴ For example, it is typical for a company to retain some earnings from a previous year to fund part of its future investment, rather than pay out all earnings as dividends and fully raise the funding of future investment from external sources. Work by Monkhouse (and others) extends the Officer framework by allowing for less than a full payout of cash flows and imputation credits each period. Handley advised that Monkhouse effectively shows that:¹⁶⁷⁵

$$\gamma = F\theta + (1 - F)\psi$$

where:

- *F* is the proportion of imputation credits generated that are distributed in a period (the 'distribution rate').
- θ (theta) is the utilisation value to investors in the market per dollar of imputation credits distributed (the 'utilisation rate').
- ψ (psi) is the utilisation value of a retained credit to investors in the market.

From the expression for γ above, we have not explicitly included the value of retained credits, ψ , when determining the value of imputation credits. This is mainly because we recognise that investors can only use imputation credits to reduce tax or receive a refund once the credits have been distributed. There is also the practical problem of

¹⁶⁷³ It is approximately equal due to the time value of money that impacts the present valuation of distributed imputation credits. However, we consider any discounting for this would be immaterial for the reasons discussed in section A.8.4 of attachment 4 to our determination for ElectraNet. The AER, *Draft decision for ElectraNet transmission determination 2018 to 2023, Attachment 4- Value of imputation credits*, October 2017

¹⁶⁷⁴ This is evident in companies having positive franking account balances in aggregate.

¹⁶⁷⁵ Handley considered that, although Monkhouse does not use the term gamma, the interpretation is clear: J. Handley, *Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits*, 29 September 2014, p. 11 and footnote 12.

how to quantify the value of retained credits. Handley acknowledged the potential for retained credits to have value, but also the difficulty in quantifying it:¹⁶⁷⁶

Retained imputation credits can be worth no less than zero but may be worth more than zero. Estimates of gamma using the traditional approach will therefore be downward biased to the extent that retained imputation credits have value. Although it is not possible to reasonably estimate the magnitude of the bias, its direction is clear.

We agree with Handley and consider assuming retained imputation credits have no value is a conservative assumption. We consider retained imputation credits have a positive value but it is difficult to quantify this value. There are many ways retained imputation credits could potentially benefit investors. For example, retained imputation credits may allow firms to conduct off market buy backs of their own stocks at a discount to prevailing market values. Off market buybacks can be structured in such a way that the purchase price is derived from both fully franked dividends and capital. Investors are prepared to sell back their shares at a discount as they derive value from imputation credits distributed and the capital gains loss that outweighs the capital loss they incur from selling at a discount.¹⁶⁷⁷ Shareholders that do not participate in off market buybacks benefit from capital appreciation of their shares as the firm's capital is brought back at a discount to the prevailing market prices. An example of a recently completed off market buy back completed a discount that likely was used to stream imputation credits to those who could utilise them most highly was by Caltex Australia.¹⁶⁷⁸

However, we consider assuming retained imputation credits have no value might be a reasonable assumption to the extent that imputation distributions rates are expected to remain constant overtime. To the extent firms maintain in perpetuity a distribution rate less than a 100 per cent, the proportion of imputation credits that are not distributed are worthless. As we assume the historical cumulative payout ratios (for listed equity) are reasonable estimates of the future expected payout ratios, we consider our approach is appropriate.

A.4.3 Evidence underlying our estimate

This section discusses the distribution rate and utilisation rate in more detail. More specifically, it describes our approach to estimating these sub-parameters of the value of imputation credits. This includes reasons for the relative levels of reliance we place on the underlying sources of evidence.

¹⁶⁷⁶ J. Handley, *Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits*, 29 September 2014, p. 14.

¹⁶⁷⁷ L. Gitman, R. Juchau and J. Flanagan, *Principles of Managerial Finance*, 6th Edition, 2011, p. 475.

¹⁶⁷⁸ Caltex Australia Limited, *Off-market buy-back booklet*, 26 February 2016; Caltex Australia Limited, *Caltex successfully completes* \$270*m off-market share buy-back*, 11 April 2016.

The distribution rate

The 'distribution rate' (or payout ratio), represents the proportion of imputation credits generated by a benchmark efficient entity that is expected to be distributed to investors.

For this draft instrument, we have updated the empirical evidence to the latest release. We propose to adopt a payout ratio of 0.83. When combined with a utilisation rate of 0.6, this gives an estimated value for imputation credits of 0.5. In coming to a distribution rate of 0.83, we place primary weight on the distribution rate estimated from:

• the financial reports for the 20 largest ASX-listed firms

We also have regard to the distribution rate estimated from:

- the financial reports for the firms within the same industry as the BEE
- ATO tax statistics

Lally's 20 largest ASX-listed firms

Lally in his reports to the AER considers an appropriate estimate for the distribution rate of the benchmark efficient entity is estimated from the financial statements for the 20 largest ASX-listed firms.¹⁶⁷⁹ This is because he considers:

- the data from financial statements is of high quality given it is audited and subject to scrutiny in financial markets¹⁶⁸⁰
- the data from the financial statements for the top 20 listed firms gives an estimated distribution rate for listed equity that is likely to best reflect the distribution rate of a BEE¹⁶⁸¹
- the listed equity distribution rate from the alternative approach (based on ATO tax statistics) contains significant unresolved discrepancies and the data is unlikely to be suitable for time series analysis¹⁶⁸²

For this draft instrument, we propose to place primary weight on the distribution rate estimated from the financial statements of the top 20 ASX-listed firms having considered Lally's advice. We agree with Lally that the information from the financial statements is audited and hence it is less likely to produce a biased distribution rate estimate compare to the ATO tax data. Moreover, the ATO in its note recommends the

 ¹⁶⁷⁹ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 19.
 M. Lally, *Gamma and the ACT Decision*, May 2016, p. 29.

M. Lally, Gamma and the ACT Decision, May 2016, p. 25.
 M. Lally, Gamma and the ACT Decision, May 2016, p. 26.

¹⁶⁸¹ M. Lally, *Gamma and the ACT Decision*, May 2016, pp. 4–6, 18, 25. In making this choice, Lally considered there is a trade-off between statistical reliability (which is greater if a market-wide estimate is used) versus potential bias (worse from a sector-wide estimate). Lally discussed various issues with using firm-specific data, industry averages and market-wide data to estimate the distribution rate.

¹⁶⁸² M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 19.

AER to not use the tax statistics as the basis of a detailed macro analysis of Australia's imputation system. 1683 ATO staff also indicated in a meeting with AER staff and representatives from the ENA some limitations with the use of ATO data.¹⁶⁸⁴ In light of the ATO's note and subsequent discussions with ATO staff, and having had regard to the merits of the evidence before us, we consider a distribution rate estimated from the top 20 ASX-listed firms' financial statements would give us a best estimate for the distribution rate of the BEE.

Lally defines the distribution rate F as the imputation credits distributed divided by the company tax payments to the ATO:¹⁶⁸⁵

$$F = \frac{DIST}{TAX}$$

Where the credits distributed is deduced from the fully franked dividends and the corporate tax rate over that period using the following formula:

$$DIST = DIV(\frac{T_c}{1 - T_c})$$

The dividend payments and the part that is fully franked are obtained from the "Dividends" note to the financial statements.¹⁶⁸⁶

Lally considers that the tax payments to the ATO are less obvious because the tax payments shown in the "Cash Flow Statement" will include payments to foreign tax authorities and separate identification of the payments to the ATO is not generally made in financial statements. However, the franking balance of an entity will have changed due to tax payments to the ATO and the distributions of credits via dividends over the period t to t+k:

$$B_{t+k} = B_{t-1} + TAX - DIST$$

Which could be written as:

$$TAX = DIST + B_{t+k} - B_{t-1}$$

Where

F denotes the imputation credits distribution rate of a firm over the period *t* to t + k, *DIST* the imputation credits distributed over the period, *TAX* the tax paid to ATO over the period, *DIV* fully franked dividends paid out over the period, B_{t-1} the franking account balance at time t - 1, B_{t+k} the franking account balance at time t + k, T_c the corporate tax rate over that period.

¹⁶⁸³ The ATO, ATO note to the AER, 9 May 2018

¹⁶⁸⁴ AER, AER minute of 21 June 2018 ATO meeting with ATO staff and comments on ENA summary, 5 July 2018

¹⁶⁸⁵ M. Lally, *Estimating the distribution rate for imputation credits*, June 2018, p. 2.

¹⁶⁸⁶ This data is drawn from the "Dividends" note to the Financial Statements for each year rather than the "Statement of Cash Flows", because the latter will not include dividends that are subject to a Dividend Reinvestment Plan.

Lally has now looked at the distribution rates for the period 2001 to 2017 of the top 20 ASX listed businesses.¹⁶⁸⁷ He considers more data would yield a more precise estimate but it also would raise the risk of bias arising from data that is not recent being unrepresentative of the current situation.¹⁶⁸⁸ Furthermore, the availability of financial statement data tails off from before 2000. Therefore, Lally considers it is appropriate to use the data since 2000.¹⁶⁸⁹

Based on his approach, the most recent data gives a distribution rate of 0.88 after adjusting for the credits recycled within the companies over the period 2001 to 2017.¹⁶⁹⁰ We note a distribution rate of 0.88 and a utilisation rate of 0.6 give a value of imputation credits of 0.53, which we have rounded down to 0.5. To ensure consistency between the rate of return parameters and given we have used an overall (rounded) value of imputation credits of 0.5 and a utilisation value of 0.6 (used for estimating historically realised excess market returns), this implies a distribution rate of 0.83.

We consider a gamma value of 0.5 based on a utilisation rate of 0.6 and a distribution rate of 0.83 will result in equity investors in the benchmark efficient entity receiving an ex ante total return on equity commensurate with the efficient equity financing costs of a BEE. In coming to this decision we have also taken into account the submissions and the issues discussed in the concurrent evidence session on the distribution rate of a BEE.

Some businesses in their submissions argue that Lally's distribution rate derived from the financial statement of the top 20 listed firms does not give an accurate reflection for the benchmark efficient entity being a highly levered and capital intensive firm.¹⁶⁹¹ Sadeh in the concurrent evidence session argued that the service providers retain large proportion of its profit to constantly reinvest it into the businesses and thus it would have a lower distribution rate than the average entity.¹⁶⁹²

¹⁶⁸⁷ We note for a few firms the first few years financial reports in the 2001 to 2017 period were not available. Where this is the case Dr Lally has looked at the data from when it was available to the end of 2017.

¹⁶⁸⁸ M. Lally, Estimating the distribution rate for imputation credits, June 2018, p. 3.

¹⁶⁸⁹ Lally calculated the distribution rate for the period 2001 to 2017. However, the calculation for the distribution rate for that period requires the data from the 2000 financial statements.

M. Lally, Estimating the distribution rate for imputation credits, June 2018, p. 3.

¹⁶⁹⁰ The credit recycled refers to the credits flow within companies, eg company A pays company B franked dividends. This would lead to double-counting of distributed credits when estimating the distribution rate if it is not appropriately adjusted. Lally considers the distribution rate adjusted for credits recycled should be determined as follows: d=(DIST-DR)/(DIST+ Δ FAB), where DIST denote the credits distributed by a firm over a period of time, DR is the credits received within that period and Δ FAB is the change in the franking account balance over that period. M. Lally, *Estimating the distribution rate for imputation credits*, June 2018, p. 8.

¹⁶⁹¹ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 11; NSG,
Submission on the Rate of Return Guideline review, 4 May 2018, p. 18; APGA, Submission to the Issues Paper,
12 December 2017, p. 10; Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December
2017, pp. 5-6; CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence,
21 April 2018, pp. 77-78.

¹⁶⁹² The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 100.

By contrast, CRG considers the electricity and gas network industry is in a mature market with largely mature technologies. Whereas they consider most firms listed on the stock exchange are in growth industries which can be expected to be withholding dividends to finance an expanding capital base.¹⁶⁹³

Given the arguments above, we have examined the actual growth in the regulated asset base of the network businesses to get a better understanding of the BEE. In the submission to the AER, ENA submits that there has been little growth in RAB since the 2013 Guidelines. It submitted that the average real rate of growth in RAB between 2013-14 and 2016-17 was approximately 1.9 per cent per annum for the electricity networks across the national electricity market.¹⁶⁹⁴

We do not consider the growth figure of 1.9 per cent supports the businesses and Sadeh's argument that the service providers would have a lower distribution rate than the average entity or Lally's top 20 ASX listed firms because they retain large proportion of its profit to constantly reinvest it into the businesses.¹⁶⁹⁵

Moreover, we consider even if a service provider required a relatively large amount of equity capital to be invested into its asset base each year on top of the allowed revenue for the depreciation they get from the regulator, there are number of ways that firms could fund their growth. This includes through the use of dividend reinvestment plans and secondary equity raisings. In particular, dividend reinvestment plans allow companies to retain their earnings while still distributing a high proportion of their imputation credits generated to shareholders. Therefore, for a BEE that operates efficiently, we do not agree with the businesses that it is necessarily true that the BEE would retain a larger proportion of its profit to fund its growth and thus have a lower distribution rate than the average entity or Lally's top 20 firms.

We note of the firms within the top 20 ASX firms Lally examined, 17 have used dividend reinvestment plans between 2001 and 2017, with only BHP, CSL and Westfield appearing to have not run dividend reinvestment programs over the period.¹⁶⁹⁶

We note the CRG proposes that the distribution rate could be estimated from the equation below:¹⁶⁹⁷

$$1 - \frac{I - D}{RAB * WACC}$$

¹⁶⁹³ CRG, Submission to the AER on its Rate of Return Guideline Review, May 2018, p. 60.

¹⁶⁹⁴ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p .25. RAB data collected from RFMs published by the AER as part of determinations for individual service providers, and from annual Economic Benchmarking RIN responses.

¹⁶⁹⁵ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 100.

¹⁶⁹⁶ Annual reports of the top 20 ASX listed firms examined by the AER. We note CSL annual reports were not readily available to AER staff for reporting years 2001 through 2004 and it is possible CSL ran a dividend investment program at some stage in this period.

¹⁶⁹⁷ CRG, Submission to the AER on its Rate of Return Guideline Review, May 2018, p. 59.

Where *I* denote the value of new and replacement asset, *D* the value of depreciation.

We have considered the CRG's proposed approach. We consider this approach is based on the assumption that apart from the revenue compensation it gets from the regulator for asset depreciation, a BEE could only fund its investment through retained earnings. However, as explained above, firms have access to different sources of finance to fund their investment, and may run dividend reinvestment programs to retain earnings while still maintaining a high franking credit distribution rate. Therefore, we do not consider the equation above based on the assumption would give a reliable estimate of the distribution rate.

Gray in the concurrent evidence session also argued that most of Lally's top 20 firms have material foreign income and it would not accord with the definition of the BEE.¹⁶⁹⁸ Lally recognizes that amongst the sample of 20 firms he looked at, many firms have foreign operations, which he considers are irrelevant to a BEE and these foreign operations could affect their distribution rate.¹⁶⁹⁹ However, he considers the effect of including the firms with foreign operations is to underestimate the distribution rate for the benchmark firm.¹⁷⁰⁰ This is based on his earlier analysis on the seven largest firms listed on the ASX which demonstrated that firms with more foreign operations are deleted at 25 per cent significant level, then the two principle candidates for deletion are BHP and Rio Tinto. This raises the estimated distribution rate from 0.88 to 0.95.¹⁷⁰² Taking this into account, Lally considers the appropriate estimate for the distribution rate of the benchmark firm (which has no foreign operations) is at least 0.88.¹⁷⁰³

The AER agrees that the lower distribution rates of BHP and RIO are materially reducing Lally's estimate of the distribution rate and exclusion of these would increase the estimate. We also note that the four lowest distribution rates in Lally's sample are Rio Tinto (Group) at 0.76, Woodside (Group) at 0.74, BHP (Group) at 0.70 and Woolworth (Group) at 0.78. The first three are relatively high risk mining companies and may not be particularly representative of the BEE that is relatively lower risk.

For the reasons outlined above, we do not consider that a distribution rate based on Lally's top 20 listed firms of 0.88 would be expected to overestimate the efficient distribution rate for the BEE. We consider the analysis and data supports that the BEE would be expected to be able to distribute at least 88 per cent of its credits while funding any required capital investment.

 ¹⁶⁹⁸ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 76.

APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16.

¹⁶⁹⁹ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 6.

¹⁷⁰⁰ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 6.

¹⁷⁰¹ M. Lally, *Issues in the estimation of gamma*, April 2017, pp. 9-12.

¹⁷⁰² The distribution rate of 0.95 is calculated by AER staff based the figures in table 1 of Lally's report :

M. Lally, Estimating the distribution rate for imputation credits, June 2018, p.10.

¹⁷⁰³ M. Lally, *Estimating the distribution rate for imputation credits*, June 2018, p. 9.

Lally has also examined the distribution rate of five firms from the same industry as the BEE over the last 10 years.¹⁷⁰⁴ Of the three firms for which the distribution rate can be estimated, Lally found a distribution rate of 1.¹⁷⁰⁵ This is higher than the distribution rate of 0.88 from Lally's updated analysis of the top 20 listed firms. Lally prior to his update of the distribution rates to 2017 considered this would support his earlier conclusion that the appropriate estimate for the distribution rate of the benchmark firm is at least 0.83.¹⁷⁰⁶ We consider this also supports his updated estimate based on data to 2017 of at least 0.88.¹⁷⁰⁷ We discuss this further below.

Gray also argued that Lally's approach for estimating the distribution rate from financial statements is subject to various difficulties requiring adjustment.¹⁷⁰⁸ One example involves BHP. Gray argued that BHP Ltd has distributed over \$1 billion of credits in BHP Plc to UK shareholders as part of its 'dividend equalisation scheme' which are completely wasted.¹⁷⁰⁹ However, Lally considers this issue involves the utilisation rate rather than the distribution rate and hence it will not affect our estimates for gamma.¹⁷¹⁰ We agree this is a utilisation rate issue.

Another example presented by Gray relates to AGL, which has had a \$300 million tax liability overturned. Gray argued that this refund would affect the estimated distribution rate if the period used to calculate it included the refund but not the original tax payments.¹⁷¹¹ Lally considers the same issue would also affect the ATO data and this is not a good reason for not using financial statement data.¹⁷¹²

We note that in the 2013 Guidelines we considered the distribution rate should be estimated on a market wide basis given the challenges of estimating it on either a firm-specific or industry specific basis.¹⁷¹³ Sadeh in the concurrent evidence session argued that the distribution rate should be based on all equity and the estimate for the listed equity has to be the upper bound.¹⁷¹⁴ For this draft instrument, we departed from our previous approach and consider the distribution rate for the BEE should be estimated from listed equity. This is based on Lally's advice, including advice at the time of the 2013 Guidelines, that the distribution rate is a firm-specific rather than a market-wide parameter and the more recent Lally advice that an estimate from the top 20 ASX firms

¹⁷⁰⁴ This includes APA Group, AusNet Services, DUET Group, Envestra and Spark Infrastructure.

¹⁷⁰⁵ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 20.

¹⁷⁰⁶ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 20.

¹⁷⁰⁷ M. Lally, *Estimating the distribution rate for imputation credits*, June 2018, p. 9.

¹⁷⁰⁸ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 77.

¹⁷⁰⁹ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 77.

¹⁷¹⁰ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 31.

¹⁷¹¹ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 77.

¹⁷¹² M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 31.

¹⁷¹³ AER, AER Explanatory statement – rate of return guideline, Dec 2013, p. 164.

¹⁷¹⁴ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 100.

supports an estimate of at least 0.88.¹⁷¹⁵ Lally considers that most of the regulated firms are either listed or owned by listed firms. Therefore, a distribution rate estimate from listed equity would be appropriate for the BEE.¹⁷¹⁶ Having considered the distribution rates achieved by the top 20 ASX firms from 2001 to 2017, we consider the BEE can be expected to distribute at least 88 per cent of its generated imputation credits. We also note a distribution rate estimate based on listed ASX firms is consistent with our primarily reliance on equity beta estimates from ASX listed firms for estimating the benchmark efficient firm's equity beta.

Further, we consider that many unlisted firms are owned by individuals who have an incentive to reduce dividends to limit the amount of tax paid at higher marginal personal rates. Hence, the dividend policy of these firms would be different from the BEE and a distribution rate from all equity will result in overcompensation for the BEE. Moreover, we do not consider the distribution rate based on listed equity has to be the upper bound.

The ENA submitted that an estimated distribution rate from the top 20 ASX firms suffers from the same issues as identified with ATO Franking account balance data.¹⁷¹⁷ We do not agree with this submission. First, the top 20 ASX firms used to estimate the distribution rate are stable through time and therefore this analysis does not suffer the material entry and exit problems inherent in the use of aggregate ATO FAB data. For example, in the ATO FAB data set some firms will disappear over time due to liquidation causing leakage. Second, the data used for analysis of the distribution rate of the top 20 ASX firms is audited financial data and therefore is expected to be more reliable than the ATO FAB data which is based informational data on tax filings.

Having considered all submissions and the experts' advice in the concurrent evidence session, we consider an estimate of the distribution rate from the top 20 listed firms' financial reports would give us a better estimate of the distribution rate for the BEE than the distribution rate from the ATO FAB data on listed equity. In coming to this view we have had regard to the fact the ATO data on the distribution rate contains significant unresolved discrepancies and we consider is likely to result in a materially less reliable estimate of the distribution rate for the BEE. The ATO has recently advised the AER that this may be caused by the dynamic nature of the tax system meaning the FAB data is not suitable for time series analysis, although the dividend data may be somewhat more reliable for estimating a market wide distribution rate. We note the national value of imputation credits from the ATO tax statistics after adjusting for the BEE's distribution rate also supports a gamma of at least 0.5.

M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 18.
 M. Lally, *Estimating the distribution rate for imputation credits*, June 2018, p. 9.

¹⁷¹⁶ M. Lally, *The estimation of gamma*, November 2013, pp. 10-11; M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 19.

¹⁷¹⁷ ENA; Response to additional AER and Australian taxation Office materials- estimating the value of imputation credits, 29 June 2018

ATO tax statistics estimates

The ATO publishes aggregate statistics on the tax returns submitted by individuals, superannuation funds and companies, as well as on the imputation credits refunded to certain income tax exempt entities (for example, charities).¹⁷¹⁸ It also publishes aggregate statistics on tax paid and franked dividends distributed by companies.

In the 2013 Guidelines we applied the cumulative payout ratio approach using ATO franking account balance data to estimate the distribution rate. This approach calculates the proportion of imputation credits generated (via tax payments) that have been distributed by companies over a certain period of time. For this draft instrument, we continue to adopt this approach. However, instead of placing primary weight on the data from the ATO on the accounts used by companies to track their stocks of imputation credits ('franking account balance') for calculating the distribution rate, we propose to place more weight on the data from the financial statements of the top 20 ASX firms.

In coming to this decision, we have considered all the relevant information before us including a report by Hathaway, which identifies a significant discrepancy associated with the tracking of imputation credits in the ATO data.¹⁷¹⁹ Hathaway found the franking account balance (FAB) data and dividend payment data from ATO tax statistics gives two different estimates of the distribution rate. Given the underlying issues with the tax statistics, Hathaway suggests to 'urge all caution in using ATO statistics for any estimates of parameters concerned with imputation credits'.¹⁷²⁰

We have updated the empirical evidence for this guideline. The FAB data suggests a distribution rate of 76 per cent for listed equity and 68 per cent for all equity.¹⁷²¹ Whereas the dividend data suggests a distribution rate of 57 per cent for all equity. We consider the limitations of a distribution rate or utilisation rate from the ATO tax statistics include:

- It gives two estimates for each gamma sub-parameter.
- It is not clear which estimates should be used (although it now appears likely the FAB data is likely to be the less reliable).

¹⁷¹⁸ These statistics are available at: https://www.ato.gov.au/About-ATO/Research-and-statistics/Taxation-statistics/. Accessed 9 April 2015.

¹⁷¹⁹ N. Hathaway, *Imputation Credit Redemption ATO data 1988–2011 Where have all the credits gone?*, September 2013,

¹⁷²⁰ N. Hathaway, *Imputation credit redemption ATO data 1988–2011: Where have all the credits gone?*, September 2013, p. 5.

¹⁷²¹ The cumulative payout ratio is for the period 2004 to 2016 based on Hathaway's advice that the ATO statistics are subject to a number of issues prior to 2004.

N. Hathaway, *Imputation credit redemption ATO data 1988–2011: Where have all the credits gone?*, September 2013, para. 32.
• The note based on the information from the ATO indicates potential issues with using the tax statistics to estimate the gamma parameters.¹⁷²²

In addition to Hathaway's report, we have also taken into account a recent ATO note to the AER, a meeting with ATO staff to discuss the limitations with ATO data, a recent submission by the ENA, and a further note by Hathaway in reaching our draft instrument decision. The ATO in the note recommends the AER to not use the taxation statistics data as the basis of a detailed macro analysis of Australia's imputation system.¹⁷²³ In particular, the ATO has indicated a key concern is with the dynamic nature of the tax system meaning the FAB data is not suitable for time series analysis. This was confirmed in a subsequent meeting with ATO staff. Given the ATO has indicated the unresolved discrepancies in Hathaway's paper are likely driven by the issues with using the FAB data as the basis for a macro-economic analysis of the Australian imputation system, we have placed greater reliance on the other data. This means we now have somewhat more confidence in estimates from ATO data of credits created, credits redeemed and imputation credits distributed. We consider greater reliance is also justified given the combined ATO data (excluding the ATO FAB data) produces an estimated utilisation rate of around 0.6 which is close to the estimated utilisation rate based on the equity ownership approach based on ABS data. As noted above, in coming to this view we have considered discussions held between AER and ATO staff and ENA representatives on 21 June 2018. The background to these discussions is contained in an AER staff minute of the discussions.¹⁷²⁴

Lally in his overview report expresses his concern with using the ATO data for estimating gamma parameters and suggests the use of financial statements for estimating the distribution rate.¹⁷²⁵ The CCP also proposes to place limited weight on the tax statistics, although some experts have suggested various ways in which these statistics might be used.¹⁷²⁶ It considers all the suggested methodologies face the same underlying issues.¹⁷²⁷

The ENA and several service providers argue that the best estimate for gamma is to use credits redeemed to company tax paid from the ATO taxation statistics, which suggest a gamma of 0.34.¹⁷²⁸ This is supported by a note and subsequent

¹⁷²² The AER, Note on ATO staff response to AER staff inquiries about Hathaway's 2013 report on imputation credit redemption, 29 March 2018

¹⁷²³ The ATO, ATO note to the AER, 9 May 2018

¹⁷²⁴ AER, AER minute of 21 June 2018 ATO meeting with ATO staff and comments on ENA summary, 5 July 2018.

¹⁷²⁵ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 19.

¹⁷²⁶ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 10.

¹⁷²⁷ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 10.

¹⁷²⁸ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 12; NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 3; CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence, 21 April 2018, p. 74; Joint Energy Networks, submission to the AER, 4 May 2018, pp. 6-7; Cheung Kong Infrastructure, Review of the Rate of Return Guideline, 12 December 2017, pp. 5-6; ATCO Gas, Submission to the AER: Review of rate of return guideline-evidence

memorandum from Hathaway that suggest the ATO data could provide a reliable estimate of gamma.¹⁷²⁹ Gray in the concurrent evidence session considered that the company tax paid and credits redeemed data from the ATO to be reliable and would provide a direct estimate of gamma across the economy.¹⁷³⁰

However, Lally considers this approach suffers from four drawbacks. He states:¹⁷³¹

"Firstly, even if gamma were correctly estimated, the Officer model also requires an estimate of the utilization rate in order to estimate the MRP, that estimate would presumably have to use the ATO data if gamma were estimated from the ATO data, and the unreliability of the ATO data in estimating the credits distributed (and hence the utilization rate) would then be problematic. Secondly, the ATO gamma data uses all companies, this implies that the credit distribution rate is estimated for all firms, this is inappropriate for the regulated businesses, and would underestimate their distribution rate. Thirdly, Hathaway's belief that the ATO data used to estimate gamma are "100% reliable" is contradicted by the ATO (as reported by the AER), in claiming that Hathaway has used the wrong figure for company tax. Fourthly, the fact that the ATO data offers two conflicting estimates of the credits distributed and neither Hathaway nor the ATO can reconcile this discrepancy ought to make any observer sceptical about anything drawn from the ATO database."

We agree with Lally that this approach for estimating gamma involves a number of potential issues. In particular, we consider that even if the credits redeemed and company tax paid calculated from the ATO taxation statistics were reliable, a market wide value of imputation credits which implies a distribution rate estimate from all equity is not representative of the value of imputation credits to the BEE. We consider the distribution rate of a BEE should be firm specific and should be estimated from listed firms for the reasons discussed in the previous section. We do not consider an estimate of the distribution rate from all equity data an appropriate estimate.

In the concurrent evidence session, Wheatley argued that this apparent pitfall from using the ATO data on all firms to estimate gamma, which involves using a market wide distribution rate being a poor proxy for the BEE, can be overcome.¹⁷³² This can be done by adjusting the gamma estimate for any difference between the distribution rates at which the representative company distributes credits and the rate at which a benchmark entity distribute credits.¹⁷³³ This approach is adopted by APA and

sessions, 4 May 2018; Response to additional AER and Australian taxation Office materials- estimating the value of imputation credits, 29 June 2018.

¹⁷²⁹ ENA, Submission on rate of return issues paper- Attachment C- Letter- Tax statistics, 12 December 2017; N. Hathaway (Capital Research), Memorandum - Response to three questions asked by the ENA, 28 June 2018.

¹⁷³⁰ CEPA, *Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence*, 21 April 2018, p. 74.

¹⁷³¹ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 37.

¹⁷³² The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 116.

¹⁷³³ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 116.

AGPA.¹⁷³⁴ We note the most recent ATO data suggests a gamma of 0.35. If we use Wheatley's suggested approach and Lally's suggested distribution rate of at least 0.88 for the BEE, an adjusted estimate of gamma will be at least:

$$0.35 \times \frac{0.88}{(0.68 + 0.57) \div 2} = 0.49$$

This supports our estimate of the value of imputation credits of 0.5. It implies a utilization rate of 0.55 which is slightly lower than the estimate of 0.6 from the ABS equity ownership data. For completeness, we also note that the use of a distribution rate based on listed equity only from ATO tax statistics when multiplied by a utilization rate of 0.6 would also produce a value of imputation credits of 0.5 when rounded.

We note the ATO has now advised us that the distribution rate estimated from the FAB data appears to be of low reliability due to the dynamic nature of the tax system.¹⁷³⁵ It appears that the ATO considers the discrepancy in Hathaway's analysis between the FAB and dividend data is potentially due to the issues with using the FAB data as the basis for a macro-economic analysis of the Australian imputation system. We note the dividend data may not have the time series issues the FAB data has as it should reflect a summation of dividends distributed in given years. On the basis a national gamma value can be calculated from ATO data, and assuming the dividend data provides an estimate of the national distribution rate consistent with this figure, we consider an estimate of gamma can be calculated that is adjusted for the estimated distribution rate of the BEE as follows:

$$0.35 \times \frac{0.88}{0.57} = 0.54$$

Lally expressed his concern about using Wheatley's approach. He considers it is possible that both of the distribution rate estimates from the ATO data are wrong and this possibility undercuts the use of any such adjustment to the gamma estimate of 0.35 to reflect the distribution rate for the BEE.¹⁷³⁶ However, following the most recent ATO advice we are now somewhat less concerned with the use of the dividend data for the above adjustment. We also note the ATO dividend data implies an estimated utilisation rate as follows

$$\frac{0.35}{0.57} = 0.61$$

This is largely consistent with the ABS equity ownership data and also consistent with our estimated imputation utilization rate of 60 per cent. As noted earlier, this

¹⁷³⁴ APA, Submission responding to discussion papers and expert evidence, 4 May 2018, pp. 11-12. APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 18.

¹⁷³⁵ ATO, ATO Note – Franking account balance – tax of time series data from Taxation Statistics, 9 May 2018; AER, AER minute of 21 June 2018 ATO meeting with ATO staff and comments on ENA summary, 5 July 2018.

¹⁷³⁶ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, pp. 33-34.

consistency gives us comfort in placing somewhat more reliance on the ATO dividend data.

Having considered all the information before us, we consider a value of imputation credits of 0.5 will provide for a post company tax return on equity inclusive of imputation credits at least sufficient to contribute to a rate of return that meets the ARORO.

The AER intends to continue to work with the ATO to better understand the ATO data sets, and the reliance that should be place on the different data sets for estimating gamma, in advance of the final Guideline being published in Dec 2018.

Comparators from the same industry

Our definition of the benchmark efficient entity is an entity that operates in Australia and has a similar degree of risk as that which applies to the particular service provider. To estimate the distribution rate for the BEE, we consider an appropriate approach is to use data from a broader range of companies that are comparable to the BEE in a relevant way. Consistent with our view, the CCP and APGA proposes that the AER should define the BEE and identify comparator companies that are most consistent with the BEE in estimating the distribution rate.¹⁷³⁷

In practice, it is difficult to construct a data set for such companies. Lally suggests one option is to pick a collection of companies within the same industry as the BEE.¹⁷³⁸ He considers that the five listed energy network businesses that examined by the AER for the purpose of estimating gearing would seem to be the appropriate comparators.¹⁷³⁹ This includes APA Group, AusNet Services, DUET Group, Envestra (now Australian Gas Networks), and Spark Infrastructure. However, only three of the firms that Lally looked at have the data available. The data suggests a distribution rate of 1 over the last ten years in all three cases.¹⁷⁴⁰ This supports Lally's updated estimate of the distribution rate for the benchmark firm of at least 0.88.

The CCP agrees with Lally that it considers the distribution rate can be best calculated from the actual behaviour of individual network companies or a relevant group of companies that fit within the AER's definition of the BEE, including the existing networks.¹⁷⁴¹

However, the main issue of using the industry data is the set of firms is not large. Lally considers the choice of whether or not to include certain marginal cases is likely to

M. Lally, Review of the AER's views on gearing and gamma, 7 May 2018, p. 18.

¹⁷³⁷ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131; APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16.

¹⁷³⁸ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 95.

¹⁷³⁹ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 95.

¹⁷⁴⁰ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 20.

¹⁷⁴¹ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131.

have a material impact on the resulting estimate.¹⁷⁴² We agree with Lally that the distribution rate of the comparators could give us guidance on the distribution rate of the BEE. However, given the sample size is relatively small we consider we should place less weight on it.

Further issues

In the concurrent evidence session, Gray suggested that the distribution rate for the BEE could be implied from the dividend payout rates of some suitable comparators.¹⁷⁴³ He considers if one characteristic of the BEE is a firm who operates within Australia, the dividend distribution rate and the credits distribution rate would be the same.¹⁷⁴⁴

Lally considers that this approach could only work if you assume the denominator in the dividend payout rate is the taxable income for which tax payments are made to the ATO less the tax payment, rather than accounting profit or net cash flow after tax.¹⁷⁴⁵ We agree with Lally that accounting profits and taxable income are two fundamentally different concepts. The accounting profit reported in the financial statements could be quite different from the actual taxable income firms reported to the ATO for tax purpose. Firms may have lower taxable incomes than the economic profits recorded in their statutory accounts or pay no tax for a range of reasons. Some of the major reasons include:¹⁷⁴⁶

- Businesses losses: The tax law recognizes companies can and do incur business losses. It allows these losses to be carried forward and recouped, for tax purposes, against subsequent profits. The same business and continuity of ownership tests provide integrity to the loss rules. Taxable income can be reduced by losses incurred in previous years, reducing the company's taxable income below its accounting profit. Over the past 10 years, 20 per cent to 30 per cent of Australian Securities Exchange (ASX) 500 companies have reported a net accounting loss in any given year.
- Special tax rules for trusts: Trusts are widely used for investment and business
 purposes by large corporate groups. Trusts are treated as taxpayer entities for tax
 administration purposes. The trustee is responsible for managing the trust's tax
 affairs, including paying some tax liabilities. When shares in some companies are
 sold together with units in an associated trust, they are said to be 'stapled'
 together. Income from the trust is returned by the unit holder in their return rather
 than by the company. This results in company taxable income returned being much

¹⁷⁴² M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 19.

¹⁷⁴³ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, pp. 112-114.

¹⁷⁴⁴ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, pp. 112-114. Dividend distribution rate is calculated as dividend distributed/ net profits or dividend distributed/operating cash flow after tax. These number can be found in financial statements. Whereas the distribution rate for imputation credits is calculated as imputation credits distributed/ tax paid to the ATO. Firms do not report their taxable income lodged to the ATO or tax paid to the ATO in the financial statements.

¹⁷⁴⁵ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 31.

¹⁷⁴⁶ See: https://www.ato.gov.au/general/tax-and-corporate-australia/tax-is-not-simply-30--of-profit/

less than total business profits, but is offset by the tax payable at the unitholder level.

- Tax concessions: Some features of tax law are designed to stimulate investment and economic growth. These various exemptions and concessions may also explain, in part, why some corporate groups appear to pay tax at a rate less than 30 per cent of the company's accounting profit (and indeed less than 30 per cent of their taxable income). Tax concessions include:
 - research and development expenditure to promote innovation and the social and economic benefits innovation brings
 - capital allowances to encourage business investment through shorter effective lives of assets for tax purposes than for accounting purposes, with particular policy concessions for certain exploration expenditure and capped effective lives for certain depreciating assets. By deferring tax to the later years of an asset's useful life, capital allowances give rise to earlier positive cash flows.
- Offshore companies investing in Australia: Overlaying our Australian tax rules, we
 have a network of tax treaties to assist international trade and fair taxation. As tax
 treaties assign taxing rights, they also impact on our domestic tax payments.
 Guidelines agreed by Australia and other countries at the Organisation for
 Economic Cooperation and Development (OECD) set the correct way to resolve
 taxing disputes.
- Investing in Australian companies: Under the imputation system, a share of corporate tax paid is imputed to shareholders. The shareholder declares both the dividend they receive and an imputed amount of corporate tax. The imputation or franking credit offsets the shareholder's tax liabilities. An Australian company that has a stake in another Australian company will not pay tax on a dividend twice. If the other company pays a franked dividend, it will not be taxable again in the hands of the shareholding company. This is even though it may be included in the accounting profits of the shareholding company.

For the reasons outlined above, we consider it is likely that firms would report a higher profit in the financial reports to the shareholders than the profit reported to the ATO. As a result, we consider a distribution rate estimated from the dividend payout rate with the denominator of accounting profit or net cash flow is likely to materially underestimate the actual distribution rate of a BEE.

A utilisation rate based on the dividend payout rate would therefore be expected to overcompensate the firms and their investors for the efficient costs that the firms incur in providing their network services. We do not consider this will meet the requirement

of the NEL/NGL. We note the CCP also expressed its concern of using Gray's proposed approach in estimating the distribution rate.¹⁷⁴⁷

Lally considered another possible way of estimating the distribution rate for the BEE is by referring to the service providers' actual imputation credits distribution rate.¹⁷⁴⁸ However, he considered the problem with that is the company would have a strong incentive to manipulate its distribution rate.¹⁷⁴⁹ We do not consider this approach will necessarily create an incentive for the service provider to operate efficiently knowing that their actual imputation credits distribution rate will affect their regulatory revenue. Rather, we consider it is likely to promote the businesses to adopt a distribution policy that would maximise their regulatory revenue. This may not necessarily be in the long term interest of the consumers, or consistent with achieving the NEO/NGO.

The utilisation rate

The 'utilisation rate' is the value to investors of utilising imputation credits per dollar of imputation credits distributed.¹⁷⁵⁰ In the Monkhouse framework, the utilisation rate is equal to the weighted average, by wealth and risk aversion, of the utilisation rates of individual investors. For an 'eligible' investor, each dollar of imputation credit received can be fully returned to the investor in the form of a reduction in tax payable or a refund.¹⁷⁵¹

For this guideline, we consider the utilisation rate should be based on the body of utilisation rate estimates with regard to its strengths and weaknesses. This includes the equity ownership approach, ATO tax statistics and implied market value studies. With current evidence, we consider this suggest an estimated utilisation rate of approximately 0.6 is reasonable. In coming to a utilisation rate of 0.6, we have updated the data to the latest release and we place:

- significant reliance upon the equity ownership approach
- limited reliance upon tax statistics
- limited reliance upon implied market value studies.

In reaching our view, we have re-examined:

- The representative investor in the Officer model
- The empirical evidence of sources for the estimate including the equity ownership approach, tax statistics estimates and various implied market value estimates

¹⁷⁴⁷ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 131.

¹⁷⁴⁸ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 95.

¹⁷⁴⁹ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 95.

¹⁷⁵⁰ In this decision we use the terms theta, utilisation value and utilisation rate interchangeably to mean the same thing.

¹⁷⁵¹ This is the return to eligible investors before administrative costs, personal taxes and diversification costs. Handley advises that this is the desired basis for the utilisation rate.

The representative investor in the Officer model

Before reaching our view on the best approach for estimating the utilisation rate, we consider the definition of the representative investor.

To answer the question of the appropriate representative investor, we have considered:

- the Sharpe–Lintner CAPM framework under imputation as derived in Officer, Monkhouse, Lally and Van Zijl, and Lally¹⁷⁵²
- the construction of the corporate tax building block in the rules and how this interacts with the Officer framework used within the rate of return
- the experts' opinion in the concurrent evidence session
- analysis of the framework by academic expert Lally

The Rule requires that the estimate of the value of imputation credits to be consistent with each interrelated element of the regulatory scheme. The Capital asset pricing model (CAPM) we use for the return on equity estimation assumes the national equity markets are segmented. This implies all the assets in the equity markets are owned by domestic investors and there is no foreign investment. Under this assumption, all the investors are eligible to redeem the imputation credits and therefore have a utilisation rate of 1. This is supported by Lally and Gray.¹⁷⁵³

However, we consider the assumption of no foreign investment and no foreign investors would not reflect the empirical reality. In light of this we consider a more appropriate way to estimate the utilisation rate is to recognize the existence of foreign investors. In particular, we consider that domestic investors are eligible to utilise imputation credits and would have a utilisation rate of 1. Conversely, foreign investors cannot utilise imputation credits and would have a utilisation rate of 0. It follows that the utilisation rate reflects the extent to which investors can utilise the imputation credits they receive to reduce their tax or obtain a refund.

This approach was reviewed by Lally. He concurs with the AER's view that the Officer CAPM assumes a segmented market where the utilisation rate is 1. However, he considers it is not necessarily wrong for the AER to include foreign investors in estimating the utilization rate in the belief that this produces more realistic results.¹⁷⁵⁴ If foreign investors are taken to be incorporated into the model, Lally suggests the utilisation rate should be defined as a weighted average of the utilisation rates of

¹⁷⁵² R. Officer, 'The cost of capital of a company under an imputation tax system', *Accounting and finance*, May 1994, vol. 34(1), pp. 1–17; P. Monkhouse, 'The cost of equity under the Australian dividend imputation system', *Accounting and finance*, November 1993, vol. 33(2), pp. 1–18; M. Lally and T. van Zijl, 'Capital gains tax and the capital asset pricing model', *Accounting and finance*, July 2003, vol. 43(2), pp. 187–210; and M. Lally, 'The CAPM under dividend imputation', *Pacific accounting review*, December 1992, vol. 4(1), pp. 31–44.

¹⁷⁵³ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, pp. 106-108.

¹⁷⁵⁴ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 4.

individual investors.¹⁷⁵⁵ This is consistent with the AER's current interpretation of the utilisation rate.

We conclude that the representative investor:

- is a domestic investor who is eligible to redeem all the imputation credits available under the construction of the domestic CAPM
- or, is the weighted average of investors within a defined market with the presence of both domestic and foreign investors, where the weightings reflect wealth and risk aversion.

We consider both definitions of the representative investor could meet the ARORO. However, for this draft instrument we continue to recognise the existence of foreign investors in the equity market and consider that a utilisation rate estimated from a weighted average over the utilisation rates of all investors in the Australian market remains appropriate. Having reached this view, we consider it has important implications for the practical task of estimating the value of imputation credits.

Arriving at an estimate of the utilisation rate

We propose to estimate the utilisation rate using the body of relevant evidence with regards to its strengths and weaknesses. In particular, we consider the concurrent evidence session on imputation credits was very useful in assisting us to reach a view on whether we should continue to have regard to datasets for both all equity and listed equity only when determining a utilisation rate.

Lally also consider the utilisation rate should be based on all equity given it is a market wide parameter.¹⁷⁵⁶ We agree on this point, although we note the ENA proposes that the AER should assess the assumed composition of the shareholder base of the BEE in estimating the utilisation rate.¹⁷⁵⁷

We agree that the utilisation rate should be seen as a market wide parameter and therefore consider that it is appropriate to refer to market wide estimates to assess the utilisation rate. Following from this, we therefore are no longer placing reliance upon equity ownership data for listed equity only when determining the utilisation rate. We consider it is not necessary to estimate the distribution rates and the utilisation rates from the same dataset based on the theory indicating the distribution rate is firm specific while the utilisation rate is market wide. This is supported by Lally, APGA and the CCP.¹⁷⁵⁸ Nevertheless, we note the current utilisation rate estimated from ABS

¹⁷⁵⁸ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 95.

¹⁷⁵⁵ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, pp. 106-107.

¹⁷⁵⁶ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 95.

¹⁷⁵⁷ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 83.

Lally's view on this issue appears consistent with the views of Gray. See Frontier Economics, *An appropriate regulatory estimate of gamma*, June 2015, pp. 12–13.

CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 10.

listed equity data when combined with a distribution rate of 0.88 still supports a gamma value of 0.5.

This section addresses:

- the equity ownership approach—on current evidence, this suggests an estimate between around 0.6 and 0.7
- tax statistic estimates—on current evidence, these suggest an estimate of 0.51 based on the FAB data and an estimated of 0.61 based on the dividend data
- implied market value studies—on current evidence, these suggest an estimate between 0 and 0.5. In particular, the adjusted estimate from SFG dividend's drop off study suggests a utilisation rate of 0.4.

The equity ownership approach

The equity ownership approach estimates the value-weighted proportion of domestic investors in the Australian equity market. This reflects that generally, domestic investors who are eligible to utilise imputation credits would have a utilisation rate of 1 whereas foreign investors would have a utilisation rate of 0. We use data from the National Accounts of the Australian Bureau of Statistics (ABS) to estimate the domestic ownership share. The most recent updated ABS data gives a range of 0.6 to 0.7 for all equity. The point estimate for the December 2017 quarter indicates a utilisation rate based on domestic ownership of 0.65.

We place significant reliance upon the equity ownership approach when considering estimates of the utilisation rate. This is because:

- it is well aligned with the definition of the utilisation rate in the Monkhouse framework
- it employs a relatively simple and intuitive methodology
- it uses a reliable and transparent source of data
- it provides estimates of the utilisation rate for investors in all equity

In the overview report to the AER, Lally agrees with the AER that in the presence of foreign investors, the utilisation rate is equal to the proportion of Australian equities owned by local investors.¹⁷⁵⁹ He considers one should use the data from the Australian Bureau of Statistics to estimate the utilisation rate if the empirical reality of foreign investors is to be incorporated in the model.¹⁷⁶⁰ Moreover, Lally suggests that the ABS data on all equity should be used in estimating the utilisation rate whereas the distribution rate should be estimated from only a subset of firms.¹⁷⁶¹ Lally considers

APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 15.

¹⁷⁵⁹ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 18.

¹⁷⁶⁰ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 18.

¹⁷⁶¹ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 26.

there is good reason not to match datasets.¹⁷⁶² He suggests a utilisation estimate of around 0.6 based on the most recent updated data.¹⁷⁶³

The CCP in its submission to the AER agrees with Lally that the utilisation rate of the BEE should be estimated from the ABS data for all equity.¹⁷⁶⁴ It recommends a gamma of at least 0.5 based on a distribution rate between 0.75 and 0.83 and a utilisation rate of 0.65 from the equity ownership approach.¹⁷⁶⁵ It considers a gamma of 0.5 will result in an effective adjusted tax allowance of 15 per cent which is closer to the reality of the actual taxation paid by the networks but will still overcompensate the firms given the imputation credits.¹⁷⁶⁶

- We note that ENA submits that the AER should examine the composition of the shareholder base of the BEE in estimating the utilisation rate.¹⁷⁶⁷ In response we note that the domestic Officer Model assumes all the assets in the equity market are owned by domestic investors. This implies the BEE is owned by domestic investors and hence it suggests a utilisation of 1. However, we consider this assumption does not incorporate the existence of foreign ownership. We acknowledge that in reality there is foreign ownership in the Australian market. We consider the utilisation rate estimate from the equity ownership approach that incorporates the existence of foreign ownership approach that utilisation rate from the utilisation rate is a market wide parameter and a utilisation rate from listed equity is not representative of the BEE.¹⁷⁶⁸
- We note that CRG in its submission proposes a utilisation rate of 1 based on the assumption that a BEE would use the most efficient source of funding from Australian sources.¹⁷⁶⁹ Nonetheless, as discussed earlier, we consider that an estimate of the utilisation rate from the equity ownership approach (that incorporates foreign investment in Australian equity) is more reflective of the reality and remains appropriate.
- In the submissions, some businesses argue that the utilisation rate estimate from the equity ownership approach does not take into account investors who receive imputation credits but do not redeem them.¹⁷⁷⁰ Our view is consistent with Lally's

¹⁷⁶² M. Lally, Issues in the estimation of gamma, April 2017, p. 13.

¹⁷⁶³ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 18.

¹⁷⁶⁴ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 132.

¹⁷⁶⁵ A distribution rate of 0.75 is from the ATO tax statistics for all equity and a distribution rate of 0.83 is based on Lally's top 20 listed firms.

CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 119.

¹⁷⁶⁶ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 132.

¹⁷⁶⁷ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 89.

¹⁷⁶⁸ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 95.

¹⁷⁶⁹ CRG, Submission to the AER on its Rate of Return Guideline Review, May 2018, p. 59.

¹⁷⁷⁰ Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6

advice that such matters are only significant to the extent that such non redemptions are significant and there is no evidence showing that.¹⁷⁷¹

- The businesses also argue that the equity ownership estimates of the utilization rate based on ABS data require filtering and adjustment, and may be subject to sampling error.¹⁷⁷² We consider such data issues would also apply to the alternative approaches. The estimates based on ATO data also requires filtering that may be subject to sampling error. Moreover, the large unexplained discrepancies in the ATO data give rise to significant different estimates which led Hathaway to 'urge all caution in using ATO statistics for any estimates of parameters concerned with [imputation] credits'¹⁷⁷³.
- The results from the market value studies are subject to considerable statistical uncertainty, the actions of tax arbitrageurs, a very wide range of results, significant sensitivity to a number of methodological choices, data filtering rules, deletion of outliers and data around ex-dividend dates that are known to be afflicted by anomalous behaviour.¹⁷⁷⁴ Lally considers relative to the problems associated with the estimates based on ATO tax data and market value studies, the concerns expressed about the ABS data are almost inconsequential.¹⁷⁷⁵

ENA also submits that the recent data released by the ABS raises more questions about the reliability of the equity ownership estimates than were apparent at the time of the 2013 Guidelines.¹⁷⁷⁶ We have examined the recent ABS releases which include a technical note published along with the September 2017 Finance and Wealth publication data. We note that the ABS has undertaken some quality assurance work for the historical data through reviews of compilation methods and through source data across the National Accounts. The time series was opened back to 1988 in this review. The Finance and Wealth publication has incorporated the revisions as a result of the historical review.¹⁷⁷⁷

Given the purpose for this revision is to improve the quality of the ABS data, we consider the updated ABS data would be more reliable. As a result, it would produce a more reliable estimate of the utilisation rate than the utilisation rate estimated from the earlier ABS release. ENA argues that the revision to the ABS data is based on a 'backcasting' exercise that could be less reliable than the estimates that it made at the

¹⁷⁷¹ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 28.

¹⁷⁷² Cheung Kong Infrastructure, *Review of the Rate of Return Guideline*, 12 December 2017, pp. 5-6; APGA, *Submission to the AER: Review of rate of return guideline*, 4 May 2018, p. 16.

¹⁷⁷³ N. Hathaway, *Imputation credit redemption ATO data 1988–2011: Where have all the credits gone?*, September 2013, paras. 9 and 99–100.

¹⁷⁷⁴ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 28.M. Lally, *Gamma and the ACT Decision*, May 2016, p. 22.

¹⁷⁷⁵ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 28.

¹⁷⁷⁶ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 91, p. 11.

¹⁷⁷⁷ A technical note which provides details about the major quality assurance work that was undertaken can be found at:

http://www.abs.gov.au/AUSSTATS/abs@.nsf/Latestproducts/5232.0Technical%20Note1Sep%202017?opendocum ent&tabname=Notes&prodno=5232.0&issue=Sep%202017&num=&view

time the historical data was collected.¹⁷⁷⁸ However, the data from the most recent December 2017 release that is not based on 'backcasting' exercise suggests a utilisation rate of 0.65 for all equity, which is within the range of 0.6 to 0.7.

• We note that in the concurrent evidence session, Gray proposed that as the model assumes the equity market is either complete segregation or complete integration, the equity ownership approach in estimating the utilisation rate is not consistent with any equilibrium models.¹⁷⁷⁹ We note these concerns, although are of the view our equity ownership approach appropriately takes into account the empirical reality of foreign investment in the Australian capital market. Lally considers all three of the methods are taking account of the existence of foreign investors, nevertheless, if we incorporated the existence of foreign investors, the ABS type data should be used for estimating the parameter.¹⁷⁸⁰

Gray and some businesses suggest placing some reliance on the equity ownership data while recognising it is an upper bound to the extent that it does not factor in the 45-day rule.¹⁷⁸¹ APGA in its submission proposes that the AER should quantify the impact of this rule.¹⁷⁸² We recognise the equity ownership approach does not take into account the existence of some domestic investors that do not hold their shares for 45 davs at risk over the ex-dividend date (the 45 day rule).¹⁷⁸³ However, we consider this is unlikely to have a material impact on the utilisation of imputation credits by domestic investors. This is supported by Lally.¹⁷⁸⁴ Importantly, no data has been presented that demonstrates a material impact. We also note that 0.6 is at the bottom of the range from ABS all equity data and below the current point estimate from this data. We note for the purposes of consistently estimating the return on equity, we have applied a value for the utilisation rate of 0.6 to post 1988 market return data. Hence, we consider a utilisation rate of 0.6 and our current estimate of MRP that aligns with observed excess market returns since 1988, inclusive of a 60 per cent value for distributed imputation credits, should give the service providers at least adequate compensation to cover their efficient financing costs.

We accept that there are potential disadvantages with the equity ownership approach. Nonetheless, we consider the equity ownership approach provides the most reliable estimate of the utilisation rate given that the alternative approaches are less satisfactory.

¹⁷⁸¹ The AER, Concurrent evidence session 2- Transcript, 5 April 2018, p. 115. APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 16. NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18.

¹⁷⁷⁸ ENA, Response to discussion papers and concurrent expert evidence sessions, 4 May 2018, p. 91.

¹⁷⁷⁹ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 108. NSG, *Submission on the Rate of Return Guideline review*, 4 May 2018, p. 18.

¹⁷⁸⁰ The AER, *Concurrent evidence session 2- Transcript*, 5 April 2018, p. 109.

¹⁷⁸² APGA, Submission to the AER: Review of rate of return guideline, 4 May 2018, p. 17.

¹⁷⁶³ Australian Competition Tribunal, *Application by Public Interest Advocacy Centre Ltd and Ausgrid* [2016] ACompT 1, 26 February 2016, paras. 1048.

¹⁷⁸⁴ M. Lally, *Gamma and the ACT Decision*, May 2016, pp. 20-21.

ATO tax statistics estimates

The ATO publishes aggregate statistics on the tax returns submitted by individuals, superannuation funds and companies, as well as on the imputation credits refunded to certain income tax exempt entities (for example, charities).¹⁷⁸⁵ Theoretically, these statistics can be used to derive a measure of the total amount of imputation credits utilised by eligible investors to offset tax or to be refunded. However, as discussed in relation to the distribution rate, there are underlying data issues with tax statistics. The ATO data gives two distinct estimates for the utilisation rate depending on what data is used. We have had regard to the estimates of the utilisation rate from the ATO tax statistics in determining an estimate of the utilisation rate of 0.6 is appropriate. Lally has also considered the utilisation rate from ATO data should warrant much lesser weight than the ABS equity ownership data.¹⁷⁸⁶ The most recent updated dividend data suggests an estimated utilisation rate of 0.61, while the Franking Account Balance (FAB) data gives an estimated utilisation rate of 0.51.¹⁷⁸⁷

As noted earlier, the ATO has now expressed the view we should not rely on FAB data for undertaking a detailed macro-economic analysis of the Australian imputation system. The ATO considers there are a range of potential issues with the FAB data, particularly the dynamic nature of the taxation system that makes this data unsuitable for time series analysis.¹⁷⁸⁸ Given this, we now place no reliance on the utilisation rate estimate of 0.51 estimated from ATO FAB data. Based on this most recent ATO advice we consider the utilisation rate estimated using ATO dividend data is likely to be somewhat more reliable, and note it supports a utilisation rate of around 0.60.

Implied market value estimates

Implied market value studies seek to infer from market prices of the value of distributed imputation credits. A wide range of such studies have been conducted over time, employing a variety of techniques. A common type of implied market value study is a dividend drop off study. These studies compare the price of a security with and without

¹⁷⁸⁵ These statistics are available at: https://www.ato.gov.au/About-ATO/Research-and-statistics/Taxation-statistics/. Accessed 9 April 2015.

¹⁷⁸⁶ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 18.

¹⁷⁸⁷ We use cumulative redemption rate for the period from 2003-4 to 2015-16 based on Hathaway's findings that the ATO statistics are subject to a number of issues prior to 2004. NERA considers that the cumulative approach gives an estimate that is least likely to be affected by potential distortions in the underlying data set. We adjust the dividend data for the imputation credits recycled within companies based on the number calculated in Hathaway's 2013 report. Hathaway calculated the credits recycled for the period from 2003-4 to 2010-11 was 66 billion. We assume the credits recycled each year is constant, hence the credit recycled in one year is 66/8=8.25 billion. This is because there are changes to the definition of some items in the ATO tax data. The data that was used in calculating credits recycled in Hathaway's 2013 report is no longer available.

N. Hathaway, *Imputation credit redemption ATO data 1988–2011: Where have all the credits gone?*, September 2013, para. 32.

NERA, The payout ratio: A report for the Energy Networks Association, June 2013, p. 11.

¹⁷⁸⁸ AER, *Minute of meeting between AER, ATO, and ENA*, June 2018.

the entitlement to a dividend. Econometric techniques are then used to infer the value of the imputation credits attached to these dividends.

We consider the implied market value studies can be subject to a number of limitations and the estimates of the utilisation rate based on these studies can be unreliable. The limitations of implied market value studies can include:

- These studies can produce nonsensical estimates of the utilisation rate; that is, greater than one or less than zero.
- The results of these studies can be influenced by factors, such as differential
 personal taxes and risk, which are not relevant to the utilisation rate. The utilisation
 rate should be estimated on a post-company pre-personal tax and costs basis
 consistent with the allowed rate of return parameters and the post (corporate) tax
 framework in the NER/NGR. This is supported by the May 2017 Full Federal Court
 decision that the court found the Rules require consistency in the way the relevant
 building blocks interact, that is, on a post-company tax and pre-personal tax and
 personal costs basis.¹⁷⁸⁹
- The results of these studies might not be reflective of the value of imputation credits to investors in the market as a whole. For instance, in dividend drop off studies the value of imputation credits is determined by the marginal investor that trade around the ex-dividend date.¹⁷⁹⁰ There is no reason to assume this reflects the value that long term investors who provide capital to a benchmark efficient entity place on imputation credits in aggregate. There is also no reason to assume this value will show what proportion of company tax is a prepayment of personal tax.¹⁷⁹¹
- These studies can be data intensive and employ complex and sometimes problematic estimation methodologies.
- Regarding dividend drop off studies, it is only the value of the combined package of dividends and imputation credits that can be observed in the market. However, there is no consensus among experts on how to separate the value to the market of dividends from the value to the market of imputation credits (this is referred to as the 'allocation problem').

We note Lally in his recent report to the AER considers the market value studies could give a utilisation rate that is on a post-company pre-personal taxes and personal costs basis so long as these market value studies are correctly interpreted.¹⁷⁹² To illustrate this point, he gives an example of the dividend drop-off studies where he shows that as

 ¹⁷⁸⁹ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC
 79, May 2017, p. 216.

¹⁷⁹⁰ J. Handley, Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits, 29 September 2014, p. 44; A. Ainsworth, G. Partington and G. Warren, Do franking credits matter? Exploring the financial implications of dividend imputation, June 2015, p. 18;

¹⁷⁹¹ J. Handley, Report prepared for the Australian Energy Regulator: Advice on the value of imputation credits, 29 September 2014, p. 44; A. Ainsworth, G. Partington and G. Warren, Do franking credits matter? Exploring the financial implications of dividend imputation, June 2015, p. 18.

¹⁷⁹² M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 24.

long as an appropriate adjustment is made to the estimated coefficients on the dividends and the imputation credits, the dividend drop-off studies could give an estimate of a utilisation rate that is on a post-company pre-personal taxes and costs basis.¹⁷⁹³

We remain of the view that the relevant adjustment proposed by Lally will not necessarily fully account for the potential effect of personal taxes and costs. The adjustment suggested by Lally would only address factors which affect both dividends and imputation credits to the same proportionate degree. However, there are factors identified by SFG which could affect investors' valuation of imputation credits (as reflected in share prices) but would not affect investors' valuation of dividends.¹⁷⁹⁴ Therefore, it does not appear that the proposed adjustment—which only addresses factors which affect both dividends and imputation credits. Moreover, we consider these studies do not clearly measure the utilisation value to long term investors. We therefore consider even the estimated coefficients on the dividends and the imputation credits are adjusted, it still may not give a reliable estimate for the utilisation rate.

Moreover, given the residual concerns about the market value studies and what they are measuring, we have placed limited reliance on estimates of the utilisation rate from these studies. This is consistent with the views we have expressed in all determinations since the 2013 Guidelines, including the final determination for AusNet Services that sets out our detailed considerations on the matter. ¹⁷⁹⁵ Our decision to place limited weight on dividend drop off studies is supported by Lally.¹⁷⁹⁶ The CCP in its submission also considers the market studies should be given no role in the estimation of gamma.¹⁷⁹⁷

The NSG considers that the value of imputation credits should be interpreted as the market value and the utilisation rate should be estimated from market value studies.¹⁷⁹⁸ It argues that one dollar income is not valued the same as one dollar in franking credits.¹⁷⁹⁹ However, we consider our current 'utilisation' approach for this draft instrument is consistent with our estimate of the Vanilla WACC and application of the post-tax revenue model and the Full Federal Court decisions on the value of imputation

¹⁷⁹³ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, pp. 23-25.

¹⁷⁹⁴ Gray for SFG identified some factors that would affect just imputation credits. SFG, *An appropriate regulatory estimate of gamma*, 21 May 2014, para. 65.

¹⁷⁹⁵ The AER, *Final decision: AusNet services transmission determination 2017-2022, Attachment 4- Value of imputation credits, April 2017, p. 42.*

¹⁷⁹⁶ M. Lally, *Review of the AER's views on gearing and gamma*, 7 May 2018, p. 18.

¹⁷⁹⁷ CCP16, Submission to the AER on its Rate of Return Guideline Review Concurrent Evidence Sessions, 4 May 2018, p. 130.

¹⁷⁹⁸ NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18.

 ¹⁷⁹⁹ NSG, Submission on the Rate of Return Guideline review, 4 May 2018, p. 18.
 CEPA, Expert Joint report: Rate of return guideline review- facilitation of concurrent expert evidence, 21 April 2018, p. 69.

credits.¹⁸⁰⁰ The Full Court found that the Rules require consistency in the way the relevant building blocks interact, that is, on a post-company tax and pre-personal tax and personal costs basis.¹⁸⁰¹ As discussed in section A.4.1, we consider the estimates from the market value studies, which is on a post personal tax and personal basis basis, is not consistent with the allowed rate of return parameters and the post (corporate) tax framework in the NER/NGR.

Our current approach for estimating the value of imputation credits is supported by the Full Federal Court. The Full Federal Court found that it was not an error of construction for the AER to focus on utilization rather than on implied market value.¹⁸⁰² It also found the Tribunal erred in concluding that the value of imputation credits is (only) the value claimed or utilised as demonstrated by the behaviour of the shareholder recipients of the imputation credits.¹⁸⁰³

In light of the ongoing concerns with estimates from dividend drop off studies and given Lally's advice, we consider that implied market value studies provide limited guidance on the utilisation rate.

In the 2013 Guidelines we considered that implied market value studies supported an estimate of the utilisation rate between 0 and 0.5.¹⁸⁰⁴ This range was determined with regard to a range of studies, with more weight given to those studies that:

- used longer data periods
- used data since 2000, when the change in tax law entitled eligible investors to a refund of credits that exceeded their tax liability
- encompassed the breadth of the market instead of just selected firms, and
- appeared to use more reasonable and robust econometric treatments.

In this draft instrument we recognise the limitations of a utilisation rate estimated from these studies and propose to place limited weight on the implied market value studies in arriving at our estimate for the utilisation rate.¹⁸⁰⁵

Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC
 79, May 2017

Federal Court of Australia, SA Power Networks v Australian Competition Tribunal (No 2) [2018] FCAFC 3, Jan 2018

¹⁸⁰¹ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, para. 752.

¹⁸⁰² Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, p. 216.

¹⁸⁰³ Federal Court of Australia, Australian Energy Regulator v Australian Competition Tribunal (No 2) [2017] FCAFC 79, May 2017, p. 216.

¹⁸⁰⁴ The AER, AER explanatory statement: Rate of Return guideline, December 2013, p. 176.

¹⁸⁰⁵ We discuss the experts view on implied market value studies and our consideration in detail in our recent decisions for the service providers. For example, see: AER, *Draft decision: ElectraNet transmission determination 2018 to 2013, Attachment4- Value of imputation credits*, October 2017, pp. 40-45, pp. 184-214.

We note that, in coming to our view on the weight to give to the implied market value studies and in determining an estimate for the utilisation rate of 0.6 is appropriate, we have had particular regard to SFG's 2016 dividend drop-off study at the adjusted estimate of the utilisation rate from this study of 2011, which suggests a utilisation rate of 0.4 after making the adjustment suggested by Lally and Handley. The consideration of the SFG work is set out in detail in section A.15.2 of Attachment 4 to our final determination for AusNet Services.¹⁸⁰⁶

¹⁸⁰⁶ The AER, Final decision: AusNet services transmission determination 2017-2022, Attachment 4- Value of imputation credits, April 2017, pp. 177-185.

B CEG return on debt memorandum

The ENA submitted a presentation and a further CEG memorandum on 21 June 2018 setting out methodological recommendations on Chairmont's methodology.¹⁸⁰⁷ ¹⁸⁰⁸

As the material was provided in the final days before we published our draft guideline, we have not had substantial time to review this paper. This section sets out some initial responses to issues raised in CEG's memorandum. Many of these issues were raised in our one-on-one meetings with service providers and we have extended our analysis to consider them.

Overall, we think the ENA and CEG material raises a number of issues worth considering. Most importantly, we agree that it is important to undertake analysis to test the impact of term on the cost of debt. In particular, as discussed in section 10.3, we have maintained the benchmark term of 10 years. In doing so, we have had regard to the submission made by the ENA and other networks that reliance on a simple average of terms across instruments issued from 2013–17 may understate the benchmark term.

Further, we also sought in our discussion paper to account for the term at issuance to isolate the other aspects of our approach. In particular, we included in our discussion paper analysis of spread differences between the service providers' actual debt instruments and our curve providers using matched terms. We have updated this analysis in this draft guideline to accommodate the RBA's revisions to historical bond yield data, and have considered a series of potential adjustments recommended by CEG and the ENA.

Similarly, we agree that it is appropriate to update our data series to reflect the RBA's revisions to its historical F3 data series on which our approach is based in part. We discuss this in greater detail in section 10.4. We have reflected this both in:

- An updated time-series comparison of the EICSI against our approach at the 10 year benchmark term
- Updated analysis of spread differences at matched terms.

In total, this analysis reduces the apparent discrepancy between the AER's estimated cost of debt and the index of actual cost of debt estimated by Chairmont. However, it does not eliminate the discrepancy and a material outperformance remains. Based on our analysis of spread differences at matched terms, reliance on broad-BBB third party yield curves indicates that service providers have raised debt at spreads to swap that are on average 27 basis points below those implied using a combination of BVAL and RBA yield curves at matched terms. We set out this analysis in greater detail in section 10.5.

¹⁸⁰⁷ ENA, AER Debt issues paper: Analysis, 22 June 2016.

¹⁸⁰⁸ CEG, Memorandum: ENA debt data, June 2016.

While we agree with some issues raised by the ENA and in the CEG memorandum, we do not agree with all of these submissions. In particular, we think the material proposing an inflation adjustment misrepresents the regulatory framework we employ. In 2017 we undertook a major review of the treatment of inflation in our regulatory approach and we point to the material developed during that review to address CEG's submission.

The following sections address the major topics raised in the CEG memo.

B.1 Overall role of the EICSI

As identified in our discussion paper, comparison of the AER's approach over 2013-17 compared to actual debt instruments serves as a sense check of:

- Our benchmark characteristics (term and credit rating)-for example, our current benchmark term is 10 years. Collection of information on actual return on debt issuances allows us to evaluate directly whether the service providers have been issuing debt at terms reflecting this assumption.
- How we implement these benchmark characteristics-for example, we rely on a set of third party yield curves with a number of adjustments in order to estimate the yields on AUD BBB+ corporate debt.

That is, we are not undertaking a reconciliation of NSPs actual revenues and costs. Rather, we are reviewing the overall reasonableness of our benchmark allowance consistent with the principles of incentive regulation.

B.2 Adjustment for outturn inflation

CEG submits that, in order to fully consider compensation under the AER's current approach, it is necessary to make an adjustment for the difference between:¹⁸⁰⁹

- The AER's estimates of expected inflation, less
- Outturn inflation.

In our view, CEG's submissions on this topic do not accurately reflect the character of the regulatory framework we employ.

In December 2017 we completed a sector wide review on the regulatory treatment of inflation.¹⁸¹⁰ In this review, we addressed the issue of compensation for inflation outcomes in substantial detail. Having done so, we concluded that:¹⁸¹¹

Our current approach targets the delivery of the initial real rate of return (derived from the initial nominal rate of return and expected inflation) plus actual inflation outcomes over the regulatory period. Targeting the real rate of

¹⁸⁰⁹ CEG, *Memorandum: ENA debt data*, June 2016, p. 3.

¹⁸¹⁰ AER, *Regulatory treatment of inflation–Final position*, December 2017.

¹⁸¹¹ AER, *Regulatory treatment of inflation-Final position*, December 2017, p. 63.

return means that revenues received by the service provider move in the same direction as inflation. If actual inflation outcomes are below expected inflation, service providers recover less revenues than expected; but if actual inflation outcomes are above expected inflation, service providers recover more revenues than expected.

Our final position is to continue to target the initial real rate of return on capital. This approach is consistent with the NER and NGR. We consider that there is a strong economic rationale behind an approach that targets the initial real rate of return. It provides stable real returns to investors and stable real prices to customers.136 We have consistently applied this approach in all our previous electricity and gas decisions (including relevant decisions by the ACCC prior to the formation of the AER). Our method for estimating the rate of return—in particular, the method for estimating equity beta—is consistent with this inflation approach. Hence, we are satisfied that service providers receive the correct overall compensation package.

The ultimate compensation provided in the regulatory framework is a real rate of return. When we estimate total revenue in our determinations we calculated an expected real revenue stream based on our best estimate of expected inflation. In each subsequent year of a regulatory determination we then adjust actual revenue according to a CPI – X formula. Thus, in most cases, actual revenue will deviate from the expected revenue set in our determination. However, this does not imply that service providers are not appropriately compensated for inflation. This is a design feature of the regulatory regime.

CEG's comparison is based on the premise that we should deliver a nominal rate of return. We considered this approach extensively in our inflation review, but concluded that we should continue to provide a real rate of return. For the reasons set out in our review of the regulatory treatment of inflation, we do not agree that CEG's conclusions about compensation arising from out-turn inflation are valid.

B.3 CEG's other recommended adjustments

In the remainder of its memorandum, CEG has proposed a series of other adjustments to Chairmont's methodology for compiling and comparing the EICSI and our subsequent analysis of it. We address the recommended adjustments in this section:

- Incorporating the RBA's data update
- Weighting instruments by tenor
- Weighting only the days on which debt is issued
- Recommendation to include callable and subordinated debt in the sample
- Recommendations about the treatment of fees
- Recommendation to excluded selected instruments from the sample
- Inclusion of debt raised in 2018.

Incorporating the RBA's data update

In its 3 June 2018 update, the RBA revised its historical data series on which our estimates are based.¹⁸¹² We discuss this in section 10.4.

We agree that all analysis and estimates should be updated to reflect the RBA's revised series and have done so for the analysis underlying this draft decision, including our analysis of spread differences at matched terms.

Weighting instruments by tenor

In line with the ENA's submission on our discussion paper, CEG recommends weighting individual instruments by term.¹⁸¹³ We consider the usefulness of this recommended adjustment differs between:

- Estimation of an average term at issuance over the benchmark sample
- Our further analysis of spread differences at matched terms.

Estimation of benchmark term

With respect to estimating the average term at issuance over the full EICSI sample, we agree that a simple average may underestimate benchmark term due to overrepresentation of short term debt if that short term debt is frequently refinanced.

For the reasons set out in section 10.3, our decision is to maintain a 10 year benchmark term. However, we do not agree that the weighting system proposed by the ENA necessarily better reflects the benchmark term because it relies on material assumptions, including:

- That the same annual issuance pattern occurs in equilibrium
- · Shorter term debt never grows in materiality within the portfolio
- Debt is always refinanced with debt of the same term.

Based on our discussions with service providers and our analysis of the data, it is clear that these conditions do not alwayshold over the 2013-17 sample. However, we do recognise that this recommendation highlights a potential shortcoming of a simple average where some or all of those conditions do hold. We have tested the sensitivity of our estimate to one such factor by determining the average term where instruments are issued by size of the debt instrument (face value at issuance). This resulted in an estimate of term issued over 2013–17 of 7.4 years.¹⁸¹⁴ We consider this adjustment relies on fewer material assumptions, but it does not fully address the potential for over-representation of short-term debt instruments.

¹⁸¹² See RBA, Changes to statistical tables, 5 June 2018—Available at: http://rba.gov.au/statistics/tables/changes-totables.html

¹⁸¹³ CEG, Memorandum: ENA debt data, June 2016, p. 4.

¹⁸¹⁴ AER, Discussion paper—Estimating the allowed rate of return on debt, May 2018, p. 31.

In our view, an alternative option to address this issue would be to compile the full sector portfolio using consistent adjustments with those used to compile the EICSI sample over 2013-17. We have data available for this exercise as we also requested all instruments in the starting portfolio at 1 January 2013. However, we have not yet requested Chairmont to undertake this analysis and will consider doing so prior to the final guideline.

Weighting by tenor for analysis of spread differences at matched terms

In contrast, we do not agree that there is a conceptual basis for weighting by terms in our analysis of spread differences at matched terms. In this exercise, for each instrument in the EICSI sample for which BVAL and RBA yield data is available, we have compared:

- Spreads to swap on the service providers' debt instruments on the day they were priced; against
- Spreads to swap using our mix of third party yield curves (broad-BBB only and 2/3 broad-BBB 1:3 broad-A, Bloomberg and RBA curves) at the term at issuance of the corresponding instrument on the date at which it was priced.

Having already accounted for the term at issuance and date of issuance, we do not agree that there is a reason to weight further for the term of debt.

Weighting only the days on which debt is issued

Like the recommendation of weighting by matched terms, we consider the potential impact of this adjustment differs between:

- Estimation of an average term at issuance over the benchmark sample
- Our further analysis of spread differences at matched terms.

On term, we agree that there may be some mismatch arising from comparing a 12 month rolling average of all issued debt in the EICSI sample, noting that instruments are likely to be issued only on a proportion of days within the 12 months, against a 12 month average of the AER's approach. Our decision is to retain the 10 year benchmark term.

In contrast, for analysis of spread differences at matched terms, our analysis already compares differences only on days at which bonds were issued. We consider there is no need for further adjustment.

Recommendation to include callable and subordinated debt in the sample

We do not agree with CEG's recommendation to include callable and subordinated bonds within the sample. We reach this view because:

• Bond optionality can materially affect both the cost of that debt in addition to the interpretation of the term of that debt instrument in a way that is difficult to reliably

control for. For example, a debt instrument may have a long term to maturity, but be callable substantially before that term.

- The RBA, Thomson Reuters, Bloomberg and S&P Global all exclude subordinated debt from their samples for estimating corporate yield curves
- All providers except for RBA exclude debt with embedded options other than makewhole callable debt.
- CEG submits that the impact of including callable and subordinated debt is 3 basis points. We consider this magnitude is immaterial to conclusions that might be reached on the sample.

Where make-whole callable debt has been identified in the EICSI sample, Chairmont has not excluded it.

Recommendations about the treatment of fees

In its analysis, Chairmont has included fees within its estimates of spreads on the following basis:¹⁸¹⁵

Fees associated with debt raising are only included if they act as an additional borrowing margin, such as line fees or commitment fees, as these are constant costs that are sometimes applied in place of a higher lending margin on bank debt. Other fees such as undrawn fees or establishment costs are not considered part of the borrowing margin. The former is a substitute for the borrowing margin, when the loan is undrawn, while the latter is a debt raising expense, which AER treats separately in its allowance consideration.

We agree with Chairmont's approach. As Chairmont observes, we separately make allowance within the building block framework for:

- An allowed return on debt- the ongoing costs (interest rate) of debt
- Debt raising costs for the financing of debt

In contrast, CEG has included all fees (including up-front fees) within its estimates of spreads on the basis that, in CEG's estimation, they exceed the AER's allowance of debt raising costs. We do not agree with this adjustment because:

- Service providers have previously proposed to the AER that, in addition to its standard allowance for debt raising costs, it should also allow further costs for the fees associated with maintaining dedicated liquidity facilities as required by credit ratings agencies.
- We have in the past rejected those fees on the basis that the favourable timing allowances in the PTRM (designed to account for working capital or liquidity) swamp the magnitude of those proposed allowances.

¹⁸¹⁵ Chairmont, Aggregation of return on debt data, April 2018, p. 6.

As we noted in our 2014 draft decision for Transgrid:¹⁸¹⁶

In 2002, Allen Consulting Group (ACG) provided the ACCC with a report on working capital. Working capital is one measure of a service provider's liquidity. It is calculated as current assets minus current liabilities. 'Current' refers to assets/liabilities that will be realised/settled within 12 months. Strictly, TransGrid's proposed allowance is designed to meet S&P Global's definition of liquidity as opposed to working capital. However, while S&P Global's definition of liquidity includes some additional items to that of the strict definition of working capital, the overall concept is the same—that is, that there be enough cashflow and liquid assets to meet short term liabilities over a 12 month period.

The report concluded that, because the PTRM assumes service providers receive revenue on the last day of the year, target revenue would offset any shortfall in the cost of financing operating expenditure (the required return on working capital). The report states:

These results provide no rationale for including an additional allowance in target revenue to provide a return on working capital. Rather, the results suggest that, were further precision to be sought in relation to the within-year timing of cash-flow – which underpins the arguments for a return on working capital – then the likely outcome is that the more precise target revenue would be lower than that derived using the PTRM.

Further, ACG found that:

The results above imply that [a working capital] allowance is unnecessary – while there may be a (small) financing cost associated with operating expenditure, any shortfall from not including an allowance in respect of working capital is likely to be swamped by the favourable allowance provided in respect of capital assets under the PTRM target revenue formula. It follows that if the Commission were to pursue further precision in relation to the assumptions it makes about the within-year timing of cash flow – which underpins the arguments for a return on working capital – then the likely outcome is that more precise target revenue would be lower than that derived using the PTRM.

ACG tested the magnitude of the favourable timing assumptions on a case study of a gas service provider. They found the timing assumptions in the PTRM provided a favourable 'bias' of 1.8 per cent of revenue compared to the revenue required to maintain adequate working capital. ...

In 2007, we identified that the PTRM has been modified since the 2002 ACG report to recognise capex in the middle of each year, while still assuming revenues are received on the last day of the year. In practice, this modification means that we add an additional half year of WACC to all capex in the year that it enters the capital base, in order to adjust for the time value of money. Service providers recover this incremental addition through increased depreciation and by increased return on capital while the capex is being depreciated. While

¹⁸¹⁶ AER, Draft decision- TransGrid transmission determination- Attachment 3: Rate of return, November 2014, pp. 329-330

strictly related to capex timing, this change further benefits the service provider and heightens the favourable cash-flow timing assumptions in the PTRM. This means that the level of the favourable 'bias' in the PTRM is in our view now greater than what was estimated by ACG in 2002. For these reasons we consider there is no need for an additional explicit allowance for liquidity costs, as service providers are already implicitly and sufficiently compensated for such costs.

In our view, this captures the type of fees that CEG proposes to include within its comparison of spreads. For example, undrawn fees are the costs paid by networks to reserve facilities to provide dedicated liquidity support.

Using ACG's initial estimate of 1.8 per cent of revenue as upward bias in the PTRM's timing assumptions is equivalent to increasing the estimated return on debt by approximately 46 basis points applied to the full regulatory asset base as opposed to only a subset of the instruments.¹⁸¹⁷ This estimate does not take account adjustments to the PTRM to add a half year of WACC to all capex, and so may materially understate the favourable upward bias from PTRM timing assumptions.

Recommendation to excluded selected instruments from the sample

CEG proposes to exclude 9 debt instruments with terms at issuance between 0–6 years from its sample on the basis that:¹⁸¹⁸

- CEG has graphically identified these bonds as upward outliers, and
- the instruments were raised within a period of time in which 5 year spreads to swap within the RBA curve exceeded the spread to swap on 10 year debt.

We do not agree with this adjustment, because:

- There is no theoretical reason to conclude that credit spreads at a shorter term cannot exceed credit spreads at a longer term. This could simply reflect investors' forward expectations, or reflect lower perceived credit risk on companies able to raise longer term debt in the corporate bond market.
- Even if we accepted that a downward sloping spread-to-swap curve was evidence
 of anomalous data, neither the available evidence nor Chairmont's quote that CEG
 has referred to supports a conclusion that the 5 year spread to swap is elevated
 over the time period referred to.
- All stakeholders proposing to rely on third party yield curves, including all networks and investors, continue to recommend reliance on the RBA and BVAL curves. Having accepted that the methodology underlying these curves is fit for purpose,

¹⁸¹⁷ We have estimated this using AusNet Services' electricity distribution PTRM for 2016-20.

¹⁸¹⁸ CEG, Memorandum: ENA debt data, June 2016, pp. 13-14.

we consider it is impractical and potentially asymmetrical to selectively remove observations from within those curves.

In particular, even if we agreed this was evidence that the term profile of the RBA curve did not accurately reflect market conditions, it is unclear whether this reflects elevated spreads on 5 year debt or depressed spreads on longer term debt. If we assumed the latter, this would suggest that differences on longer term debt over the corresponding time period are being understated.

Similarly, there may be other periods of time within the sample, not identified by CEG, in which the RBA or BVAL curves produced an inaccurately low estimation of the spreads at a particular term.

Further, even if we concluded that it is not appropriate to rely on that RBA data over the specific time period identified by CEG, it is unclear why instruments issued in this period should be excluded from the sample rather than simply relying on the BVAL data for the period.

Inclusion of debt raised in 2018

The ENA also recommended in its submission that we should include debt instruments issued in 2018.¹⁸¹⁹ CEG appears to have included this debt in its analysis based on its figure 3-1.¹⁸²⁰

We initially sought debt data from service providers in a request dated 6 February 2018. This data was provided at varying times by respondents to our request up to 15 March 2018. As a consequence, we have been provided with a small sample of data for 2018.

Due to the low number of instruments issued in 2018 included in responses to our information request, we have not included this data in our sample. Nonetheless, we consider the inclusion of 2018 data may justify further consideration in developing the final instrument. We may be able to supplement the data already received with a further data request to generate a material sample for 2018.

ENA, Estimating the allowed return on debt—Response to AER discussion paper, May 2018, p. 21.
 CEG, Memorandum: ENA debt data, June 2016, p. 11.

C Illustration of broad credit band weighting

To illustrate our conceptual view on the use of a 2/3 broad-BBB: 1/3 broad-A weighting to implement the benchmark credit rating, we have set out a simple algebraic demonstration below.

This demonstration relies on some simplifying assumptions. The complex curve fitting methodologies employed by various curve providers mean that it is not straightforward to reach conclusions about the ultimate impact of particular bonds on the yield estimates at different points along the term profile. Further, as identified in section 10.5, we consider credit ratings are an informative but imperfect proxy for credit risk.

Nonetheless, to the extent that credit ratings are an informative measure of credit risk, we would expect:

- reliance on a broad-BBB curve only to overestimate the level of credit risk (and ultimately the required yields) of a BBB+ benchmark credit rating- because the benchmark credit rating (BBB+) is the highest rating band amongst the constituents, the inclusion of any of the lower rated bonds in the sample (BBB or BBB-) would, other things held constant, overestimate the required return on debt for the benchmark credit rating
- reliance on a broad-A curve only would underestimate the level of credit risk (and ultimately the required yields) for a BBB+ benchmark credit rating because all constituents (A- ,A ,A+) are higher rated than the BBB+ benchmark credit rating
- some combination of broad-BBB and broad-A curves should therefore provide the best fit to a BBB+ benchmark credit rating. In our view, a 2/3 broad-BBB: 1/3 broad A rating is most likely to match a BBB+ benchmark credit rating.

Credit ratings are discrete variables and cannot directly be 'averaged'. However, credit ratings are primarily a proxy for credit risk (the risk of default).¹⁸²¹

If we assume that:

- credit ratings are equidistant in terms of risk (ie A- is as different from BBB+ as BBB is from BBB-),
- over time, the broad credit rating bands are influenced equally by their credit rating constituents (eg the broad BBB curve is influenced equally by BBB-, BBB and BBB+ bonds).

Then, if we assign a BBB- credit rating some level of risk (y), the level of credit risk for each credit rating within the broad BBB and broad A curves can be defined as set out in Table 49.

¹⁸²¹ See ACCC, Regulatory Economic Unit, Return on debt estimation: a review of the alternative third party data series – Report for the AER, August 2014, p. 23.

Table 49 Credit risk for credit ratings based on an equal spread of riskbetween ratings

Credit rating	Credit risk
BBB-	у
BBB	y+x
BBB+	y+2x
A-	y+3x
A	y+4x
A+	y+5x

Source: AER analysis

The average level of credit risks for the broad bands are then as follows:

Broad BBB only (BBB-, BBB and BBB+ constituents)

$$\mu = \frac{(y) + (y + x) + (y + 2x)}{3}$$
$$\mu = \frac{3y + 3x}{3}$$
$$\mu = y + x$$

This corresponds to an 'average' credit rating of BBB as per Table 49. To the extent that

Broad A only (A-,A,A+ constituents)

$$\mu = \frac{(y+3x) + (y+4x) + (y+5x)}{3}$$
$$\mu = \frac{3y+12x}{3}$$
$$\mu = y+4x$$

This corresponds to an 'average' credit rating of A as per Table 49.

2/3 broad-BBB and 1/3 broad A

$$\mu = \frac{2(y+x)}{3} + \frac{(y+4x)}{3}$$
$$\mu = \frac{2}{3}y + \frac{2}{3}x + \frac{1}{3}y + \frac{4}{3}x$$
$$\mu = y + 2x$$

This corresponds to our benchmark credit rating of BBB+.